MCKOOL SMITH, PC

John J. Sparacino (TX Bar No. 18873700) S. Margie Venus (TX Bar No. 20545900) 600 Travis Street, Suite 7000

Houston, Texas 77002 Telephone: (713) 485-7300 Facsimile: (713) 485-7344

Email: jsparacino@mckoolsmith.com Email: mvenus@mckoolsmith.com

Travis E. DeArman (TX Bar No. 24074117)

300 Crescent Court, Suite 1200

Dallas, Texas 75201

Telephone: (214) 978-4000 Facsimile: (214) 978-4044

Email: tdearman@mckoolsmith.com

Counsel for the Liquidating Trustee, Dundon Advisers LLC

PORZIO, BROMBERG & NEWMAN, P.C.

Warren J. Martin Jr. (admitted pro hac vice) Rachel A. Parisi (admitted *pro hac vice*) 100 Southgate Parkway

P.O. Box 1997

Morristown, New Jersey 07962-1997

Telephone: (973) 538-4006 Facsimile: (973) 538-5146 Email: WJMartin@pbnlaw.com Email: RAParisi@pbnlaw.com

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re: Chapter 11 EIGER BIOPHARMACEUTICALS, INC., et al.1 Case No. 24-80040 (SGJ) Debtors. (Jointly Administered)

THE LIQUIDATING TRUSTEE'S WITNESS AND EXHIBIT LIST FOR APRIL 29, 2025 HEARING

Dundon Advisers, LLC, c/o Joshua Nahas, in its capacity as liquidating trustee (the "Liquidating Trustee") of the liquidating trust of Eiger BioPharmaceuticals, Inc., et al. (the "Liquidating Trust"), by and through its undersigned counsel, in the above-captioned Chapter 11

¹ The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Ave., Dallas, Texas 75201.



bankruptcy proceedings of the above-captioned post-effective date debtors and debtors-in-possession (the "<u>Debtors</u>"), hereby submits this witness and exhibit list (the "<u>Witness and Exhibit</u> <u>List</u>") and designate the following witnesses in connection with the matter scheduled for hearing on **April 29**, **2025**, **at 9:30 a.m.** (**prevailing Central Time**).

WITNESSES

- 1. Any witness listed by any other party;
- 2. Any witness called by any other party; and
- 3. Any impeachment or rebuttal witnesses.

EXHIBITS

| Ex. No. | Description | Off. | Obj. | Adm. |
|---------|--|------|------|------|
| 1. | Order (I) Approving the Sale of the Debtors' Zokinvy Assets, (II) Authorizing Assumption an Assignment of Certain Executory Contracts and Unexpired Leases Related Thereto, and (III) Granting Related Relief [Dkt. No. 162] | | | |
| 2. | Notice of Closing of Zokinvy Sale Transaction [Dkt. No. 214] | | | |
| 3. | Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Remaining Assets Sale Transaction(s) [Dkt. No. 313] | | | |
| 4. | Amended Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Remaining Assets Sale Transaction(s) [Dkt. No. 351] | | | |
| 5. | Notice of Cancellation of Auction(s), Designation of Winning Bid for the Lonafarnib Sale Transaction, and Transition to Private Sale Process for Lonafarnib/Lambda Sale Transactions [Dkt. No. 489] | | | |
| 6. | Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection with the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Dkt. No. 490] | | | |

| 7. | Notice of Hearing for August 20, 2024 Setting [Dkt. No. 509] | | |
|-----|--|--|--|
| 8. | Revised Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection with the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Dkt. No. 558] | | |
| 9. | Notice of Closing of Lonafarnib/Lambda Sale Transactions [Dkt. No. 616] | | |
| 10. | Notice of Occurrence of Effective Date of Fifth Amended Joint Plan of Liquidation of Eiger BioPharmaceuticals, Inc. and its Debtor Affiliates Pursuant to Chapter 11 of the Bankruptcy Code [Dkt. No. 685] | | |
| 11. | Declaration of James Vollins in Support of the Liquidating Trustee and the Plan Administrator's Limited Response to Motion of Sentynl Therapeutics, Inc to (Enforce the Sokinvy Sale Order and (II) for Contempt Against Eiger InnoTherapeutics, Inc. [Dkt. No. 795] | | |
| 12. | Declaration of Joshua Nahas in Support of Objection and Response of the Liquidating Trustee and Plan Administrator to Motion for Allowance of Administrative Expense Claim of Sentynl Therapeutics, Inc. [Dkt. No. 785 (sealed)] | | |
| 13. | Sublicense Agreement [Dkt. No. 785-2 (sealed)] | | |
| 14. | Sentynl email to Plan Administrator dated October 15, 2024 [Dkt. No. 801-2 (sealed)] | | |
| | Any exhibit offered by any other party. | | |
| | Any declaration filed by the Debtors in these bankruptcy cases. | | |
| | Any pleadings, reports, or other documents filed in the above referenced bankruptcy cases, and any transcripts in any such cases. | | |
| | Any exhibit for impeachment of rebuttal purposes. | | |

Dated: April 25, 2025 Respectfully submitted,

MCKOOL SMITH, PC

/s/ S. Margie Venus

John J. Sparacino (SBN 18873700) S. Margie Venus (SBN 20545900) 600 Travis Street, Suite 7000 Houston, Texas 77002 Telephone: (713) 485-7300 Facsimile (713) 485-7344 jsparacino@mckoolsmith.com mvenus@mckoolsmith.com

Travis E. DeArman (SBN 24074117) 300 Crescent Court, Suite 1200 Dallas, Texas 75201 Telephone (214) 978-4000 Facsimile (214) 978-4044 tdearman@mckoolsmith.com

-and-

Warren J. Martin Jr. (admitted *pro hac vice*)
Rachel A. Parisi (admitted *pro hac vice*) **PORZIO, BROMBERG & NEWMAN, P.C.**100 Southgate Parkway
P.O. Box 1997
Morristown, New Jersey 07962-1997
Telephone: (973) 538-4006
Facsimile: (973) 538-5146
WJMartin@pbnlaw.com

Counsel for the Liquidating Trustee, Dundon Advisers LLC

RAParisi@pbnlaw.com

CERTIFICATE OF SERVICE

I hereby certify that on April 25, 2025, I caused a copy of the foregoing redacted document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas, and upon the following via electronic mail:

(I) Sentynl Therapeutics, Inc. and Eiger InnoTherapeutics, Inc and their respective counsel, and counsel for the United States Trsutee, who will all receive both the redacted as well as an unredacted version:

Michael G. Hercz Senior Vice President & General Counsel Sentynl Therapeutics, Inc. mhercz@sentynl.com

Mark Stromberg Stromberg Stock, PLLC mark@strombergstock.com

Counsel to Sentynl Therapeutics, Inc.

Pillsbury Winthrop Shaw Pittman LLP Hugh M. Ray, III hugh.ray@pillsburylaw.com
L. James Dickinson
james.dickinson@pillsburylaw.com
Reed C. Trechter
reed.trechter@pillsburylaw.com
Joshua D. Morse
joshua.morse@pillsburylaw.com
Counsel to Sentynl Therapeutics, Inc.

Dr. Jeffrey Glenn Eiger InnoTherapeutics, Inc. jsglenn@stanford.edu

Goodwin Proctor LLP
Kizzy Jarashow
kjarashow@goodwinlaw.com
Maggie Wong
mwong@goodwinlaw.com
David Chen
davidchen@goodwinlaw.com

Counsel to Eiger InnoTherapeutics, Inc.

Gray Reed
Jason Brookner
jbrookner@grayreed.com
Emily Shanks
eshanks@grayreed.com

Counsel to Eiger InnoTherapeutics, Inc.

Office of the United States Trustee Elizabeth Ziegler Young Elizabeth.a.young@usdoj.gov Counsel for the United States Trustee

and, (II) Lonza, Cordon and PRF (redacted version only):

Lara Crow Lara.crow@lonza.com Stacy Broad Stacy.broad@lonza.com Counsel to Lonza

Naoki Takei & naoki.takei@cordenpharma.com Richard Janovjak richard.janovjak@cordenpharma.com Counsel to Cordon Pharma

Weil, Gotchal & Manges LLP Clifford W. Carlson Clifford.carlson@weil.com Emma Wheeler Emma.wheeler@weil.com

Counsel to Progeria Research Foundation (PRF)

/s/ S. Margie Venus
S. Margie Venus

Exhibit 1



CLERK, U.S. BANKRUPTCY COURT NORTHERN DISTRICT OF TEXAS

> THE DATE OF ENTRY IS ON THE COURT'S DOCKET

The following constitutes the ruling of the court and has the force and effect therein described.

Signed April 24, 2024

United States Bankruptcy Judge

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

ORDER (I) APPROVING THE SALE
OF THE DEBTORS' ZOKINVY ASSETS,
(II) AUTHORIZING ASSUMPTION AND ASSIGNMENT
OF CERTAIN EXECUTORY CONTRACTS AND UNEXPIRED
LEASES RELATED THERETO, AND (III) GRANTING RELATED RELIEF

Upon consideration of the Motion² of the debtors and debtors in possession in the abovecaptioned chapter 11 cases (collectively, the "<u>Debtors</u>") for entry of an order (this "<u>Zokinvy Sale</u>

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainExhibitim1en1PaqePaqef 2.103 112

Order"), pursuant to sections 105(a), 363, and 365 of title 11 of the United States Code (the "Bankruptcy Code") and Rules 2002, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), Rule 9013-1 of the Bankruptcy Local Rules for the Northern District of Texas (the "Bankruptcy Local Rules"), and Section E of the *Procedures for* Complex Chapter 11 Cases in the Northern District of Texas (the "Complex Case Procedures"), authorizing (a) the Debtors' sale of certain of their property free and clear of liens, claims, encumbrances, and interests on the terms set forth in that certain Asset Purchase Agreement by and between Sentynl Therapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated March 31, 2024, (the "Zokinvy Stalking Horse APA") [Docket No. 13, Ex. A-2]; (b) the assumption and assignment of the Designated Contracts in connection with the Amended Zokinvy Stalking Horse APA; and (c) granting related relief, all as more fully set forth in the Motion; and this Court having previously entered the Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III)

Capitalized terms used by not otherwise defined herein shall have the meanings ascribed to such terms in the Debtors' Motion for Entry of an Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 13] (the "Motion") or the Amended Zokinvy Stalking Horse APA (defined herein).

Casse2248800449ssgjj111 DDocc818621 Filitelc0044225225 Hintereclc00442252251171047595 DDess c MainElokoitoutn1en1PaqeP41qef31131112

Granting Related Relief [Docket No. 94] (the "Bid Procedures Order"), authorizing and approving, among other things, the Debtors' designation of Sentynl Therapeutics, Inc. (the "Zokinvy Stalking Horse Purchaser"), as the Zokinvy Stalking Horse Purchaser for the Transferred Assets and certain Bid Protections; and the Debtors having filed the Second Revised Notice of Selection of Winning Bid [Docket No. 139] (the "Notice of Winning Bid"), pursuant to the Bid Procedures Order, announcing the Zokinvy Stalking Horse Purchaser as the highest or otherwise best bidder for the Transferred Assets (the "Purchaser") pursuant to the Zokinvy Stalking Horse APA following the Auction and the Purchaser having submitted the highest or otherwise best offer for assets to be sold to the Purchaser as identified in the Zokinvy Stalking Horse APA (the "Transferred Assets"), as reflected in the Zokinvy Stalking Horse APA and as from time to time amended in accordance with this Zokinvy Sale Order or further order of this Court, including by the Amended Zokinvy Stalking Horse APA, pursuant to which the Debtors have agreed, among other things, to sell the Transferred Assets to the Purchaser, including the Designated Contracts that will be assumed and assigned to the Purchaser on the terms and conditions set forth in the Amended Zokinvy Stalking Horse APA inclusive of the Zokinvy Purchase Price (as defined below) (collectively, the "Zokinvy Sale Transaction"); and the Debtors having filed the Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Zokinvy Sale Transaction [Docket No. 116] (the "Cure Notice") and served the Notice of Assumption and Assignment of Designated Contracts (the "Assignment Notice") in accordance with the Bid Procedures Order; and this Court having conducted the Zokinvy Sale Hearing to consider approval of the Zokinvy Sale Transaction, at which time all interested parties were offered an opportunity to be heard with respect to the Zokinvy Sale Transaction; and this Court having reviewed and considered (i)

Cassc2248800409sggj111 DDocc818821 Fillielc0044225225 Hintercelc00442252251171047505 DDess MairExhibitum1en1PaqePager 4.103 1.12

the Motion and the exhibits thereto, (ii) the First Day Declaration [Docket No. 27], (iii) and the Supplemental Victor Declaration [Docket No. 141], and the arguments and representations of counsel made, and the evidence proffered or adduced at the Zokinvy Sale Hearing; and it appearing that due and proper notice of the Motion, the Zokinvy Stalking Horse APA, the Amended Zokinvy Stalking Horse APA, the Cure Notice, the Assignment Notice, the Bid Procedures Order, and this Zokinvy Sale Order having been provided in accordance with the Bid Procedures Order; and all objections, if any, to approval of the Zokinvy Sale Transaction having been withdrawn, resolved (including by separate agreement between the objecting party and the Debtors), adjourned, or overruled as provided in this Zokinvy Sale Order; and it appearing entry of this Zokinvy Sale Order and consummation of the Zokinvy Sale Transaction are in the best interests of the Debtors, their estates and creditors, and all parties in interest in these chapter 11 cases; and upon the record of the Zokinvy Sale Hearing and these chapter 11 cases; and after due deliberation thereon; and sufficient cause appearing therefor,

IT IS HEREBY FOUND AND DETERMINED THAT:

A. <u>Fed. R. Bankr. P. 7052</u>. The findings and conclusions set forth herein constitute this Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such. This Court's findings shall also include any oral findings of fact and conclusions of law made by this Court during or at the conclusion of the Zokinvy Sale Hearing. To the extent of any conflict, the oral rulings control.

- B. <u>Jurisdiction and Venue</u>. This Court has jurisdiction over the Motion and the Zokinvy Sale Transaction described therein, and in the Amended Zokinvy Stalking Horse APA, including, without limitation, the sale of the Transferred Assets, pursuant to 28 U.S.C. §§ 157 and 1334. The Debtors have asserted that venue for these Chapter 11 Cases is proper pursuant to 28 U.S.C. § 1408.³ This Court may enter a final order consistent with Article III of the United States Constitution. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- C. <u>Statutory Predicates</u>. The statutory authorization for the relief granted herein is found in sections 105(a), 363, and 365 of the Bankruptcy Code, Bankruptcy Rules 2002, 6004, and 9014, Rule 9013-1 of the Bankruptcy Local Rules, and Section E of the Complex Case Procedures.
- D. This Zokinvy Sale Order constitutes a final and appealable order within the meaning of 28 U.S.C. § 158(a). Time is of the essence in closing the Zokinvy Sale Transaction referenced herein, the Debtors and the Purchaser intend to close the Zokinvy Sale Transaction as soon as practicable, and there is no just reason for delay in the implementation of this Zokinvy Sale Order. Specifically, the Zokinvy Sale Transaction must be approved and consummated promptly to preserve the viability of the business in the hands of the Purchaser as a going concern, and to maximize the value to the Debtors, their estates, their creditors, and all other parties in interest. Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), and to any extent necessary under Bankruptcy Rule 9014 and Rule 54(b) of the Federal Rules of Civil Procedure, as made applicable by Bankruptcy Rule 7054, the Court expressly finds that there is no just

On April 11, 2024, The Office of the United States Trustee filed the *United States Trustee's Emergency Motion* to *Transfer Venue or Dismiss under 28 U.S.C. §§ 1406 and 1408 and Fed. R. Bankr. P. 1014(a)(2)* [Docket. No. 111] (the "Venue Motion"). The Venue Motion is set to be heard and determined on May 7, 2024. All parties rights are hereby expressly reserved as to the determination of whether venue is proper in this District.

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainExholicum1en1PaqePaqef 6.103 112

reason for delay in the implementation of this Zokinvy Sale Order, waives any stay, and expressly directs entry of judgment as set forth herein.

- E. <u>Incorporation By Reference</u>. Findings of fact and conclusions of law in the Bid Procedures Order are incorporated herein by reference.
- F. Marketing Process & Auction. The Debtors and their professionals adequately marketed the Transferred Assets to all Potential Bidders in accordance with the Bid Procedures Order. The sale process set forth in the Bid Procedures Order afforded all Potential Bidders a full, fair, and reasonable opportunity to submit a higher or otherwise better offer to purchase the Transferred Assets and participate in the sale process. The Auction was conducted in a reasonable and fair manner in accordance with the Bid Procedures. No other person, or group of persons, has offered to purchase the Transferred Assets for an amount that would give greater value to the Debtors than the value provided by the Purchaser pursuant to the Amended Zokinvy Stalking Horse APA, which reflects the final bid during the Auction of a Base Price in the amount of \$46,100,000 less a credit in the amount of \$900,000 for the Termination Fee resulting in a net Base Price in the amount of \$45,200,000 (assuming a Closing on April 24, 2024) (the "Zokinvy Purchase Price"), which constitutes the highest and best bid for the Transferred Assets. Under the circumstances, the marketing process was robust and sufficiently tested the market to determine the highest and best offer for the Transferred Assets.
- G. <u>Sale Hearing</u>. This Court conducted the Zokinvy Sale Hearing on April 23, 2024, at which time this Court considered the Motion, the evidence and testimony presented, and the statements and argument of counsel, as applicable, in support of the Motion, the Amended Zokinvy Stalking Horse APA, and the Zokinvy Sale Transaction. Except as otherwise expressly provided in this Zokinvy Sale Order, all objections to the Zokinvy Sale Transaction and the relief

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainExholicum1en1PaqePagef71131112

requested in the Motion, whether timely or untimely and whether written or made orally at the Zokinvy Sale Hearing, if any, were heard and considered by this Court. All such objections, if any, were either overruled by this Court, are resolved by the terms hereof or by separate agreement between the objecting party and the Debtors, or were adjourned or withdrawn as a result of an agreement between the objecting party and the Debtors.

- H. <u>Sound Business Purpose</u>. The Debtors have demonstrated good, sufficient, and sound business purposes and justifications for consummation of the Zokinvy Sale Transaction pursuant to the Amended Zokinvy Stalking Horse APA and all other agreements, instruments, certificates, and other documents to be entered into or delivered by any party in connection with the Zokinvy Sale Transaction, including, without limitation, any assumption and assignment agreements entered into in connection therewith (collectively, the "<u>Transaction Documents</u>"), outside of the ordinary course of business and in accordance with the requirements of section 363(b) of the Bankruptcy Code. Consummation of the Zokinvy Sale Transaction prior to and not as part of a chapter 11 plan is (i) justified under the circumstances, (ii) an appropriate exercise of the Debtors' business judgment, and (iii) in the best interests of the Debtors, their estates, and their creditors.
- I. The Debtors' decision to enter into the Zokinvy Stalking Horse APA with the Zokinvy Stalking Horse Purchaser, subject to higher and better offers, was a due and proper exercise of the Debtors' business judgment and was authorized pursuant to the Bid Procedures Order. The Bid Protections contained in the Zokinvy Stalking Horse APA (i) were necessary to preserve the value of the Debtors' estates by inducing the Zokinvy Stalking Horse Purchaser to enter into the Zokinvy Stalking Horse APA and (ii) are in compliance with the Bid Procedures and authorized by the Bid Procedures Order.

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainExholicum1en1PaqePager 8.103 112

J. Following a robust marketing process and Auction consistent with the Bid Procedures Order, the Amended Zokinvy Stalking Horse APA at the Zokinvy Purchase Price, constitutes the highest or otherwise best offer for the Transferred Assets. No other person, or group of persons, has offered to purchase the Transferred Assets for an amount that would give greater value to the Debtors than the value provided by the Zokinvy Purchase Price. The Zokinvy Sale Transaction is the best means available to the Debtors to maximize the return to their creditors and limit the losses to counterparties to the Designated Contracts. No alternative to the Zokinvy Sale Transaction exists that would provide a greater value to the Debtors, their creditors, or other parties in interest.

K. Approval of the Zokinvy Sale Transaction is necessary to maximize the value the Debtors' estates will receive for the Transferred Assets. It is important to the Debtors' customers and suppliers that the transition from the Debtors to the Zokinvy Stalking Horse Purchaser occurs smoothly and without unnecessary delay, so that any customer and vendor issues may be minimized. It is also important that the Zokinvy Sale Transaction be consummated as expeditiously as possible to avoid any disruption to the patients who depend on Zokinvy to treat progeria, a rare and fatal genetic condition that may result from continued uncertainty about the future of the Transferred Assets.

L. Accordingly, the sale of the Transferred Assets pursuant to sections 105(a) and 363 of the Bankruptcy Code upon the terms and conditions set forth in the Amended Zokinvy Stalking Horse APA is the optimal means to create value for the benefit of the Debtors' estates. The Zokinvy Sale Transaction maximizes the value of the Transferred Assets because the Transferred Assets are being sold as part of a going concern, thereby preserving the continuity and remaining goodwill value associated with the Transferred Assets. Unless the sale is

Casse2248800449ssgjj111 DDocc818621 Filitelc0244225225 Hintereclc02442252251171047595 DDess c Mai Extribition the nPageP1601616112

concluded expeditiously, as provided for in the Motion and the Amended Zokinvy Stalking Horse APA, creditor recoveries may be substantially diminished.

M. <u>Fair Purchase Price</u>. The Zokinvy Purchase Price provided by the Purchaser as set at the Auction (i) is fair and adequate; (ii) constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia (including the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act, and similar laws); and (iii) will provide an equal or greater recovery for the Debtors' creditors than would be provided by any other reasonably practicable available alternative. The terms of the Amended Zokinvy Stalking Horse APA, the Transaction Documents, and the Zokinvy Sale Transaction are fair and reasonable under the circumstances of the Debtors' chapter 11 cases, and the Debtors' determination to proceed with such transaction constitutes a valid and sound exercise of the Debtors' business judgment.

N. Adequate and Reasonable Notice. As evidenced by the affidavits of service filed with this Court [Docket Nos. 42, 114, 126, 127, 128, 140], and based upon the record of the Zokinvy Sale Hearing, and as previously determined by this Court in the Bid Procedures Order, (i) due, proper, timely, adequate, and sufficient notice of the Motion, the Zokinvy Auction, the Zokinvy Sale Hearing, the Amended Zokinvy Stalking Horse APA, and the Zokinvy Sale Transaction has been provided to all parties in interest, (ii) such notice was and is good, sufficient, and appropriate under the circumstances, and reasonably calculated to reach and apprise all holders of Liens, claims, encumbrances, and other Interests (as defined herein), including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities, and was provided in accordance with the applicable requirements of the Bankruptcy Code, the Bankruptcy Rules, the Bankruptcy Local Rules, the Complex Case

Casse2248800409ssjýjj11 DDocc83821 Filitelc0244225225 Eintereclc0244225225171047595 DDessc MairEldothichtrien Pagle 1.01:03 112

Procedures, and the procedural due process requirements of the United States Constitution, and (iii) no other or further notice of the Motion, the Zokinvy Auction, the Zokinvy Sale Hearing, the Amended Zokinvy Stalking Horse APA, the Zokinvy Sale Transaction, or of the entry of this Zokinvy Sale Order is necessary or shall be required.

- O. In accordance with the Bid Procedures Order, the Debtors filed with this Court and served the Cure Notice, containing (i) the list of all Designated Contracts to potentially be assigned in connection with the Zokinvy Sale Transaction, (ii) information necessary and appropriate to provide notice of the relevant proposed assumption and assignment of potentially assigned contracts that may be Designated Contracts and rights thereunder, (iii) Cure Amounts, where applicable, and (iv) the procedures for objecting thereto, on all counterparties to such potentially assigned contracts and any party that has requested notice pursuant to Bankruptcy Rule 2002 ("Rule 2002 Notice List"), and caused such notice to be published on the website of the Debtors' noticing agent, Kurtzman Carson Consultants LLC. ("KCC") [Docket No. 116]. The Cure Notice (a) included the Debtors' good faith calculation of the Cure Amounts with respect to each potentially assigned contract; (b) stated that assumption or assignment of any potentially assigned contract is not guaranteed and is subject to this Court's approval; (c) prominently displayed the deadline to file a Cure Objection; and (d) prominently displayed the dates, times, and location of the Sale Hearing. The service and provision of the Cure Notice was good, sufficient, and appropriate under the circumstances and no other or further notice need be given.
- P. In accordance with the Bid Procedures Order, the Debtors also served the Assignment Notices on the counterparties to the Designated Contracts, which contained (i) the list of Designated Contracts selected by the Purchaser, (ii) information necessary and appropriate

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEtholocitritentPagtPagtPagtP1133112

to provide notice of the relevant proposed assumption and assignment of the Designated Contracts and rights thereunder, (iii) the Cure Amounts, and (iv) the procedures for objecting thereto, on all counterparties to the Designated Contracts and all parties on the Rule 2002 Notice List. The service and provision of the Notice of Designated Contracts was good, sufficient, and appropriate under the circumstances and no other or further notice need be given in connection with the assumption and assignment of the Designated Contracts.

- Q. A reasonable opportunity to object and to be heard with respect to the sale of the Transferred Assets, the assumption and assignment of the Designated Contracts, and the determination of defaults and Cure Amounts related thereto, as well as the Amended Zokinvy Stalking Horse APA and the entry of this Zokinvy Sale Order has been given to all interested Persons.
- R. Good Faith Purchaser. The Debtors, the Purchaser, and their respective principals, counsel, and advisors have negotiated, proposed, and entered into the Amended Zokinvy Stalking Horse APA, the Transaction Documents, and each of the transactions contemplated therein in good faith, without collusion and from arm's-length bargaining positions. The Purchaser is a "good faith purchaser" and is acting in good faith within the meaning of section 363(m) of the Bankruptcy Code in closing the Zokinvy Sale Transaction and, as such, is entitled to all the protections afforded thereby. The Purchaser has proceeded in good faith in all respects. The terms of the Zokinvy Sale Transaction, including the Zokinvy Purchase Price, were not controlled by any agreement among Potential Bidders and neither the Debtors nor the Purchaser have engaged in collusion or any conduct that would cause or permit the Amended Zokinvy Stalking Horse APA to be challenged, avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy Code or any other law of the United States, any

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaglP4.Geo12131112

state, territory, possession thereof, or the District of Columbia, or any other applicable law. The Amended Zokinvy Stalking Horse APA was not entered into for the purpose of hindering, delaying, or defrauding creditors under the Bankruptcy Code or under laws of the United States, any state, territory, or possession, or the District of Columbia, or any other applicable law. Neither the Debtors nor the Purchaser entered into the Amended Zokinvy Stalking Horse APA or are consummating the Zokinvy Sale Transaction with any fraudulent or otherwise improper purpose. The Purchaser is not an "insider" or "affiliate" of any of the Debtors, as those terms are defined in section 101 of the Bankruptcy Code, and no common identity of incorporators, directors, or controlling stockholders exists between the Purchaser and the Debtors.

- S. The Zokinvy Sale Transaction, which includes the sale of the Transferred Assets pursuant to the Amended Zokinvy Stalking Horse APA and all covenants in and conditions thereto, is an integrated transaction, meaning that each component is an essential part of every other component and that the Zokinvy Sale Transaction can be consummated only if all of the components are consummated. Accordingly, each component of the Zokinvy Sale Transaction is subject to, and is protected by, the provisions of section 363(m) of the Bankruptcy Code.
- T. Sale Free and Clear under Section 363(f). The Purchaser would not have entered into the Amended Zokinvy Stalking Horse APA and would not consummate the Zokinvy Sale Transaction without entry of this Zokinvy Sale Order approving the Zokinvy Sale Transaction pursuant to section 363(f) of the Bankruptcy Code. Except as expressly provided otherwise in the Amended Zokinvy Stalking Horse APA or this Zokinvy Sale Order, the Debtors have satisfied the standard set forth in section 363(f) of the Bankruptcy Code for selling the Transferred Assets free and clear of all of the following (collectively, "Interests"): Liens, claims (including, but not limited to, those that constitute a "claim" as defined in section 101(5) of the

Casse2248800499sggj111 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDess c MairElatoricutrien Pagle 1.4eo 1.3.103 112

Bankruptcy Code), encumbrances, obligations, liabilities, pledges, charges, demands, guarantees, actions, suits, defenses, deposits, credits, allowances, options, rights, restrictions, limitations, contractual commitments, rights of first refusal, rights of setoff or recoupment, royalties, hypothecations, preferences, debts, easements, suits, licenses, rights of recovery, judgments, orders and decrees of any court or foreign domestic governmental entity, taxes (including foreign, state, and local taxes), covenants, indentures, instruments, leases), claims for reimbursement or subrogation, contribution, indemnity or exoneration, encumbrances, or interests of any kind or nature whatsoever against the Debtors, or any of the Transferred Assets, including, without limitation, any debts arising under or out of, in connection with, or in any way relating to, any acts or omissions, obligations, demands, guaranties, rights, contractual commitments, restrictions, product liability claims, environmental liabilities, employment or labor law claims or liabilities, employee pension or benefit plan claims, multiemployer benefit plan claims, retiree healthcare or life insurance claims or claims for taxes of or against the Debtors or against any property of the Debtors, claims arising under state or federal antitrust laws, any indemnification claim or liabilities relating to any act or omission of the Debtors or any other person prior to the Closing Date or any Excluded Liabilities, any derivative, vicarious, transferee or successor liability claims, alter ego claims, de facto merger claims, rights or causes of action (whether known or unknown, legal or equitable, contingent, matured or unmatured, contingent or non-contingent, liquidated or unliquidated, choate or inchoate, filed or unfiled, scheduled or unscheduled, perfect or unperfected, allowed or disallowed, noticed or unnoticed, recorded or unrecorded, material or non-material, statutory or non-statutory, and asserted or unasserted, whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases (other than Permitted Liens and the Assumed Liabilities), whether imposed by

Casse2248800499sggj111 DDocc82621 Filitelc0244225225 Hintercelc0244225225171047595 DDess c MairElatoricutrien Pagle 1.5eo 1.4 103 112

agreement, understanding, law, equity or otherwise, including without limitation (i) those Interests that purport to give to any party a right or option to effect a setoff against or any forfeiture, modification, or termination of the Debtors' interests in the Transferred Assets, or any similar rights, if any, (ii) those Interests arising under all mortgages, deeds of trust, security interests, conditional sale or other title retention agreements, pledges, hypothecations, liens, judgments, demands, encumbrances, rights of first refusal or charges of any land or nature, if any, (iii) those Interests that are Excluded Liabilities as set forth in the Amended Zokinvy Stalking Horse APA; (iv) those Interests held by the Prepetition Term Loan Secured Parties (as defined in the Interim Cash Collateral Order) in the Transferred Assets, including as provided in the order entered by the Court at Docket No. 93 (the "Interim Cash Collateral Order") and (v) those Interests arising under or out of, in connection with, or in any way related to the Debtors or any of the Debtors' predecessors, Affiliates, or representatives, any of the Sellers' interests in the Transferred Assets, or the operation of any of the Debtors' businesses before the applicable Closing Date, including, without limitation, Interests based on successor liability, transferee liability, derivative liability, vicarious liability, de facto merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. The Prepetition Term Loan Secured Parties have, subject to the terms and conditions of this Zokinvy Sale Order, consented to the relief requested in the Motion with respect to the Zokinvy Sale Transaction. Each other holder of an Interest in the Transferred Assets (a) has, subject to the terms and conditions of this Zokinvy Sale Order, consented or shall be deemed to have consented to the relief requested in the Motion and with respect to the Zokinvy Sale Transaction, (b) could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest, or (c) otherwise falls within the provisions of section 363(f) of the Bankruptcy Code. Those

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc024225225171047595 DDess c MainEtholocitritentPagtPal6eo15113112

holders of Interests that did not object to, or withdrew their objections, if any, to, the relief requested in the Motion, the Amended Zokinvy Stalking Horse APA, the Zokinvy Sale Transaction, or the Assignment Notices are deemed to have consented to the relief requested in the Motion, including, without limitation, the sale of the Transferred Assets and the assumption and assignment of the Designated Contracts to the Purchaser, pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests that did object that have an Interest in the Transferred Assets could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest pursuant to section 363(f)(5) of the Bankruptcy Code or fall within one or more of the other subsections of 363(f) of the Bankruptcy Code and, therefore, are adequately protected by having their Interests that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Zokinvy Sale Transaction ultimately attributable to the property in which they have an Interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Zokinvy Sale Transaction, subject to any defenses of the Debtors.

U. Except as expressly provided otherwise in the Amended Zokinvy Stalking Horse APA or this Zokinvy Sale Order, neither the Purchaser nor any of the Purchasers' Affiliates (including any subsidiary of the Purchasers, any person or entity that could be treated as a single employer with the Purchasers pursuant to Section 4001(b) the Employee Retirement Income Security Act of 1974, as amended ("ERISA") or Section 414(b), (c), (m) or (o) of the Internal Revenue Code of 1986, as amended ("IRC"), and any of their respective managed funds or accounts, any of their respective lenders or investors, and, in each case of the foregoing, each of their respective former, current, or future, shareholders, equity holders, owners, members, managers, employees, representatives, officers, limited or general partners, directors, agents,

Casse2248800499sggj111 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDess c MairElatoricutrien Pagle 1.0131112

professionals, successors, affiliates, or permitted assignees, (collectively with the Purchaser, the "Purchaser Group") shall be responsible for any Interests, including in respect of, based on, relating to, and/or arising under, without limitation, the following: (i) any labor, collective bargaining, or employment agreements; (ii) any mortgages, deeds of trust, or security interests; (iii) any intercompany loans and receivables between one or more of the Seller and any Debtor; (iv) any pension, multiemployer (as such term is defined in Section 3(37) or Section 4001(a)(3) of ERISA), health or welfare plan participation or benefit trust, compensation or other employee benefit plans, agreements, practices and programs (including any Employee Benefit Plan) of or related to any of the Debtors or any of the Debtors' Affiliates or predecessors or any current or former employees of any of the foregoing, including, without limitation, any pension plan of any of the Debtors or any multiemployer plan to which the Debtors have at any time contributed to or had any liability or potential liability; (v) the Debtors' business operations or cessation thereof; (vi) any litigation involving one or more of the Debtors; (vii) any other employee, worker's compensation, occupational disease or unemployment or temporary disability related claim, including, without limitation, claims that might otherwise arise under or pursuant to (a) ERISA, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the Multi-Employer Pension Plan Amendments Act of 1980, including all amendments thereto, (f) the Worker Adjustment and Retraining Notification Act of 1988 or any similar state or local law ("WARN"), (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, including, without limitation, the requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the IRC and of any similar state law (collectively, "COBRA"), (i) the National Labor Relations Act, (j) the Age Discrimination and Employment Act of 1967 and Age Discrimination

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEtholocitritentPagtPagtPage017103112

in Employment Act, as amended, (k) state harassment, discrimination, or retaliation laws, (l) state unemployment compensation laws or any other similar state laws, or (m) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors, or relating to any wages, benefits, employment, or termination of employment with any or all Debtors or any of their predecessors; (viii) any liabilities arising under any Environmental Laws with respect to any assets owned or operated by any of the Debtors or any corporate predecessor of any of the Debtors at any time prior to the applicable Closing Date; (ix) any product liability law; (x) any antitrust laws; (xi) any bulk sales or similar law; (xii) any Employee Benefit Plan) of or related to any of the Debtors or any of the Debtors' Affiliates or tax statutes or ordinances, including, without limitation, the IRC; and (xiii) any Excluded Liabilities.

- V. <u>No Successor, Transferee, or Similar Liability</u>. Except for the Assumed Liabilities, as expressly set forth in the Amended Zokinvy Stalking Horse APA or this Zokinvy Sale Order, the Purchaser has not expressly or impliedly assumed any obligation of the Debtors, or any other party, with respect to the Interests and the Excluded Liabilities, whether at law or in equity, whether by payment, setoff, recoupment, or otherwise, directly or indirectly, and whether from the Transferred Assets or otherwise, including, without limitation, based on successor, transferee, derivative, or vicarious liability.
- W. The Zokinvy Sale Transaction described by the Amended Zokinvy Stalking Horse APA and the Transaction Documents does not amount to a consolidation, merger, or de facto merger of the Purchaser and any of the Debtors and/or any of the Debtors' estates.
- X. There is no continuity between the Purchaser and any of the Debtors. The Purchaser is not holding itself out to the public as a continuation of any of the Debtors or their

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hintereclc0044225225171047595 DDess MairEphototritenPage 1.9 1.13 1.12

respective estates, businesses, or operations. The Purchaser is not a mere continuation of any of the Debtors or their respective estates, businesses, or operations. There is no common identity between any of the Debtors and the Purchaser. The Purchaser does not constitute a successor to any of the Debtors or their estates.

Y. The Purchaser and the Debtors are not entering into the Amended Zokinvy Stalking Horse APA and Transaction Documents or consummating the Zokinvy Sale Transaction for the fraudulent purpose of escaping liability for the Debtors' obligations or to defraud creditors in any way.

Z. Sale Free and Clear and Continuation of Existing Approvals Required by the The Purchaser expressly negotiated for the protection of obtaining the Transferred Purchaser. Assets free and clear of all Interests, including, without limitation, any potential successor liability claims (other than Permitted Liens and the Assumed Liabilities). The total consideration to be provided under the Amended Zokinvy Stalking Horse APA reflects the Purchaser's reliance on this Zokinvy Sale Order to provide it, pursuant to sections 105(a) and 363 of the Bankruptcy Code, with title to and possession of the Transferred Assets free and clear of all Interests of any kind or nature whatsoever (including, without limitation, any potential successor liability claims (other than Permitted Liens and the Assumed Liabilities)). The Purchaser would not have entered into the Amended Zokinvy Stalking Horse APA and would not consummate the Zokinvy Sale Transaction, if the sale of the Transferred Assets to the Purchaser and the assumption and assignment of the Designated Contracts to the Purchaser by the Debtors, were not free and clear of all Interests of any kind or nature whatsoever (other than the Permitted Liens and the Assumed Liabilities), as contemplated by this Zokinvy Sale Order, except as otherwise set forth herein, or if the Purchaser would, or in the future could, be liable for any of Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 20eo 19.103112

the Interests, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities. The Purchaser would not have entered into the Amended Zokinvy Stalking Horse APA and would not consummate the Zokinvy Sale Transaction if the Purchaser would not be authorized, as of the Closing Date, to operate under or renew any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the terms of the Stalking Horse APA); if such licenses, permits, registrations, and governmental authorizations or approvals would not be deemed to have been transferred to the Purchaser as of the Closing Date; or if existing licenses or permits applicable to the business would not remain active and in place for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred.

AA. <u>Assumption and Assignment of the Designated Contracts</u>. The Assumption and Assignment Procedures approved pursuant to the Bid Procedures Order are integral to the Amended Zokinvy Stalking Horse APA, do not constitute unfair discrimination, are in the best interests of the Debtors, their estates and creditors, and all other parties in interest, and are based on the reasonable exercise of sound business judgment by the Debtors. At the Closing and pursuant to Section 365 of the Bankruptcy Code and this Zokinvy Sale Order, the Debtors shall assume and, subject to the terms in the Amended Zokinvy Stalking Horse APA, assign to the Purchaser, and Purchaser shall take assignment from the Debtors of, the Designated Contracts.

BB. On or before the Closing Date, the Purchaser will pay all Cure Amounts with respect to the Designated Contracts proposed to be resolved after the Closing Date in accordance with Paragraph 20 below. Accordingly, the Debtors or the Purchaser, as applicable, will have, to the extent necessary, (i) cured any default existing prior to the Closing with respect to the

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hintereclc0044225225171047595 DDess MairEphototrienPage 20te02013 112

Designated Contracts, and (ii) provided compensation, if any, to each counterparty to a Designated Contract for any actual pecuniary loss to such party resulting from a default prior to the Closing with respect to the Designated Contract with such counterparty, all within the meaning of sections 365(b)(1)(A) and 365(f)(2)(A) of the Bankruptcy Code.

CC. Pursuant to section 365(f) of the Bankruptcy Code, each Designated Contract required to be assumed and assigned under the Amended Zokinvy Stalking Horse APA shall be assigned and transferred to, and remain in full force and effect for the benefit of, the Purchaser, in accordance with their respective terms, notwithstanding any provision in such contract or other restrictions prohibiting its assignment or transfer. No section of any of the Designated Contracts that would directly or indirectly prohibit, restrict, or condition the assumption or assignment of any of the Designated Contracts or would permit termination or modification of such Designated Contracts, or rights and obligations thereunder, by a party other than the Debtors, on account of assignment of such shall have any force or effect in connection with the Transferred Assets.

DD. The assumption and assignment of the Designated Contracts (i) is necessary to sell the Transferred Assets to the Purchaser, (ii) allows the Debtors to sell the Transferred Assets to the Purchaser as a going concern, (iii) limits the losses suffered by counterparties to the Designated Contracts, and (iv) maximizes the recoveries to other creditors of the Debtors by limiting the amount of claims against the Debtors' estates by avoiding the rejection of the Designated Contracts. For these reasons, the Debtors have exercised sound business judgment in assuming and assigning the Designated Contracts and such assumption and assignment is in the best interests of the Debtors' estates.

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hintereclc0044225225171047505 DDess MairENoticitn1enPage 20eo21131112

EE. Adequate Assurance of Future Performance. Counterparties to the Designated Contracts were provided with the Assignment Notice and had the opportunity to request and review information with respect to the Purchaser's adequate assurance of future performance (see Docket No. 94, Ex. 5) and were required to file any objections to the Purchaser's ability to provide adequate assurance of future performance as contemplated under sections 365(b)(1)(C), 365(b)(3) (to the extent applicable) and 365(f)(1) of the Bankruptcy Code (each an "Cure Objection") on or prior to April 16, 2024 at 4:00 P.M. Central Time. Counterparties to Designated Contracts that failed to timely file a Cure Objection are hereby forever barred from objecting to the assumption and assignment of Designated Contracts on the grounds of a failure to provide adequate assurance of future performance. Based on evidence adduced at the Zokinvy Sale Hearing and based on the record in these chapter 11 cases, to the extent necessary, the Debtors have satisfied the requirements of section 365 of the Bankruptcy Code, including sections 365(b)(1)(A), 365(b)(1)(B), 365(b)(1)(C), 365(b)(3) (to the extent applicable) and 365(f) of the Bankruptcy Code, in connection with the sale and assumption and assignment of the Designated Contracts to the extent provided under the Amended Zokinvy Stalking Horse APA. Accordingly, subject to payment of the Cure Amounts, the Designated Contracts may be assumed by the Debtors and assigned to the Purchaser as provided under the Amended Zokinvy Stalking Horse APA and this Zokinvy Sale Order.

FF. Sale Order Required by the Purchaser. Entry of this Zokinvy Sale Order approving the Amended Zokinvy Stalking Horse APA is a requirement of the Amended Zokinvy Stalking Horse APA and such requirement is a reasonable and appropriate condition precedent to the Purchaser's consummation of the Zokinvy Sale Transaction.

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc024225225171047595 DDess c MainEldoiduithien Pagle 2.Geo. 2.2.13 1.12

GG. <u>Transferred Assets Property of the Estates</u>. The Transferred Assets constitute property of the selling Debtors' estates and title thereto is vested in the selling Debtors' estates within the meaning of section 541(a) of the Bankruptcy Code. The selling Debtors have all title, interest, and/or rights in the Transferred Assets required to transfer and to convey the Transferred Assets to the Purchaser, as required by the Amended Zokinvy Stalking Horse APA.

HH. Corporate Authority. Subject to the entry of this Zokinvy Sale Order, (i) the Debtors have full corporate power and authority to perform all of their obligations under the Amended Zokinvy Stalking Horse APA and the Transaction Documents, and the Debtors' prior execution and delivery of, and performance of obligations under, the Amended Zokinvy Stalking Horse APA and the Transaction Documents is hereby ratified, (ii) the Debtors have all of the corporate power and authority necessary to consummate the Zokinvy Sale Transaction, (iii) the Debtors have taken all corporate actions necessary to authorize, approve, execute, and deliver the Amended Zokinvy Stalking Horse APA and the Transaction Documents and to consummate the Zokinvy Sale Transaction, except for the closing conditions expressly provided in the Amended Zokinvy Stalking Horse APA and the Transaction Documents, and (iv) no consents or approvals are required to consummate the Zokinvy Sale Transaction or otherwise perform the obligations under the Amended Zokinvy Stalking Horse APA or the Transaction Documents, except for the closing conditions expressly provided therein.

- II. <u>Sale in Best Interests</u>. The relief requested in the Motion and set forth in this Zokinvy Sale Order is in the best interests of the Debtors, their respective creditors, estates, and all other parties in interest in the Debtors' chapter 11 cases.
- JJ. <u>Prompt Consummation</u>. To maximize the value of the Transferred Assets, it is essential that the Zokinvy Sale Transaction occur within the timeframe set forth in the Amended

Casse2248800409sgijj11 DDocc8B821 Filitelc0044225225 Hinterenclc00442252251171047595 DDess MairElatoricutrii en Pagle 24eo 23.103 112

Zokinvy Stalking Horse APA and Bid Procedures. Time is of the essence in consummating the Zokinvy Sale Transaction. Accordingly, there is cause to lift the stays established by Bankruptcy Rules 6004 and 6006 with regards to the Zokinvy Sale Transaction and the assignment of the Designated Contracts.

NOW, THEREFORE, IT IS ORDERED THAT:

- 1. <u>Motion Is Granted</u>. The Motion and the relief requested therein, and entry into and performance under the Amended Zokinvy Stalking Horse APA, is GRANTED and APPROVED, as set forth herein.
- 2. Objections Overruled. Except as stated otherwise herein, all objections to, or reservation of rights regarding, the relief requested in the Motion, the entry of this Zokinvy Sale Order, or the relief granted herein, including, without limitation, any objections to Cure Amounts or relating to the cure of any defaults under any of the Designated Contracts or to the assumption and assignment of any of the Designated Contracts to the Purchaser by the Debtors, that have not been withdrawn, waived, settled, or adjourned as provided in Paragraph 20 below or otherwise, or that have not otherwise been resolved pursuant to the terms hereof are hereby denied and overruled on the merits with prejudice. All Persons that failed to timely object, or withdrew their objections, to the Motion or the entry of this Zokinvy Sale Order are deemed to consent to the relief granted herein for all purposes, including, without limitation, pursuant to section 363(f)(2) of the Bankruptcy Code. No appeal, motion to reconsider, or similar pleading has been filed with respect to the Bid Procedures Order, and the Bid Procedures Order is a final order of this Court, has not been vacated, withdrawn, rescinded, or amended and remains in full force and effect.

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1Pagle25e024131112

- 3. <u>Notice</u>. Notice of the Motion and Zokinvy Sale Hearing was adequate, appropriate, fair, and equitable under the circumstances and complied in all respects with section 102(1) of the Bankruptcy Code and Bankruptcy Rules 2002, 6004, and 6006, the Bankruptcy Local Rules and the Bid Procedures Order, and as such no further or other notice is required.
- 4. Approval and Authorization. The sale of the Transferred Assets to the Purchaser on the terms and conditions contained in the Amended Zokinvy Stalking Horse APA and the Transaction Documents, including, without limitation, the Closing of the Zokinvy Sale Transaction as required by the Amended Zokinvy Stalking Horse APA, is hereby approved in all respects pursuant to sections 105(a), 363(b) and (f), and 365 of the Bankruptcy Code and Bankruptcy Rule 6004. Pursuant to sections 105, 363, and 365 of the Bankruptcy Code, the Debtors are authorized to perform all obligations under and make all payments required by the Amended Zokinvy Stalking Horse APA and the Transaction Documents as and when due thereunder without further order of this Court. The Debtors, the Purchaser, and each of their respective officers, employees, and agents are hereby authorized to (i) execute the Amended Zokinvy Stalking Horse APA and the Transaction Documents, including that certain Amended Asset Purchase Agreement by and between Sentynl Therapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated April 22, 2024, attached hereto as Exhibit 1 (the "Amended Zokinvy Stalking Horse APA"), and any prior execution of such agreements, documents, and instruments, including the Transaction Documents, is hereby ratified, (ii) perform all obligations under the Amended Zokinvy Stalking Horse APA and the Transaction Documents, to consummate each of the foregoing, including, without limitation, deeds, assignments, and other instruments of transfer, and to consummate the Zokinvy Sale Transaction, and any prior performance of such obligations or any prior consummation of such

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 26e02513 112

Zokinvy Sale Transaction is hereby ratified, (iii) assume and assign the Designated Contracts to the Purchaser, and (iv) take all other and further actions as may be reasonably necessary to consummate and implement the Zokinvy Sale Transaction and to perform all obligations under the Amended Zokinvy Stalking Horse APA and the Transaction Documents and the consummation thereof, without any further corporate action or order of this Court. The Purchaser shall not be obligated to proceed with the Closing under the Amended Zokinvy Stalking Horse APA until all conditions precedent to its obligation to do so thereunder have been satisfied or waived.

- 5. No Sub Rosa Plan. The sale of the Transferred Assets, including, without limitation, the assignment of the Designated Contracts, pursuant to the Amended Zokinvy Stalking Horse APA outside a chapter 11 plan neither impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms of the Debtors' subsequent chapter 11 plan. Neither the Amended Zokinvy Stalking Horse APA nor the Zokinvy Sale Transaction constitutes a sub rosa chapter 11 plan.
- 6. <u>Valid Transfer</u>. As of the Closing, the consummation of the Zokinvy Sale Transaction shall effect a legal, valid, and enforceable sale and transfer of the Transferred Assets to the Purchaser, and shall vest the Purchaser with all legal, equitable, and beneficial right, title, and interest in and to the Transferred Assets free and clear of all Interests of any kind or nature whatsoever. The Amended Zokinvy Stalking Horse APA and the Transaction Documents are valid and binding contracts between the Debtors and the Purchaser and shall be enforceable pursuant to their terms. The Amended Zokinvy Stalking Horse APA, the Transaction Documents, the Zokinvy Sale Transaction itself, and the consummation thereof shall be specifically enforceable against and binding upon (without posting any bond) the Debtors and

Casse2248800409ssjójj11 DDocc83821 Filitelc0244225225 Einteccelc0244225225171047595 DDessc MairEldotlottrien1Pagle27ge216131112

their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Designated Contracts, the Purchaser and its respective successors and assigns, any chapter 11 trustee appointed in these chapter 11 cases or any chapter 7 trustee appointed upon a conversion of these chapter 11 cases to cases under chapter 7 of the Bankruptcy Code, and shall not be subject to rejection or avoidance by the foregoing parties or any other Person.

7. Free and Clear. Except as expressly provided for in the Amended Zokinvy Stalking Horse APA or this Zokinvy Sale Order, pursuant to sections 105(a), 363(b), 363(f), 365(b), and 365(f) of the Bankruptcy Code, the Debtors are authorized and directed to transfer the Transferred Assets to the Purchaser and, upon the Closing, other than the Purchaser's assumption of the Assumed Liabilities and the Purchaser's obligations under the Amended Zokinvy Stalking Horse APA and the Designated Contracts, the Purchaser shall have and take title to and possession of the Transferred Assets free and clear of and shall have no obligation with respect to all Interests (other than Permitted Liens and the Assumed Liabilities) of any kind or nature whatsoever, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities; de facto merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. All holders of Interests fall within one or more of the subsections of section 363(f) of the Bankruptcy Code and are adequately protected by having their Interests attach to the net proceeds ultimately received by the Debtors and attributable to the Transferred Assets against or in which such Interests are asserted, subject to the terms of such Interests, with the same validity, force, and effect, and in

Casse2248800499sggj111 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDess c MairElatoricutrien Pagle 2660271131112

the same order of priority that such Interests now have against the Transferred Assets or their proceeds as of Closing, subject to any rights, claims, and defenses the Debtors or their estates, as applicable, may possess with respect thereto, in addition to any limitations on the use of such proceeds pursuant to any provision of this Zokinvy Sale Order; provided, however, that setoff rights will be extinguished to the extent there is no longer mutuality of the parties after consummation of the Zokinvy Sale Transaction. This Zokinvy Sale Order: (a) is and shall be effective as a determination that other than Assumed Liabilities or as otherwise provided herein, upon the applicable Closing in accordance with the Amended Zokinvy Stalking Horse APA, all claims of any kind or nature whatsoever existing as to Transferred Assets, and any tax liability, prior to the applicable Closing have been unconditionally released, discharged, and terminated, and that the conveyances described herein have been effected, with such Interests and liens attaching in order of priority to the proceeds of the Zokinvy Sale Transaction, and (b) is and shall be binding upon and shall authorize all entities, including without limitation all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies or unites, governmental departments or units, secretaries of state, federal, state and local officials and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to the Transferred Assets conveyed to the Purchaser. All recorded Interests against the Transferred Assets from their records, official and otherwise, shall be deemed stricken upon the Closing in accordance with the Amended Zokinvy Stalking Horse APA and the terms of this Zokinvy Sale Order. The conditions of section 363(f) of the Bankruptcy Code have been satisfied in full; therefore, the Debtor may sell the Transferred

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hintereclc0044225225171047595 DDess MairEphototrienPage 2.9eo.28.108 112

Assets free and clear of any liens, claims, and/or interests (other than Permitted Liens and the Assumed Liabilities).

- 8. The Prepetition Term Loan Secured Parties have, subject to the terms and conditions of this Zokinvy Sale Order, consented to the relief requested in the Motion with respect to the Zokinvy Sale Transaction. Those other holders of Interests or claims who did not object (or who ultimately withdrew their objections, if any) to the Zokinvy Sale Transaction are deemed to have consented pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests or claims who did object that have an interest in the Transferred Assets fall within one or more of sections 363(f)(1), 363(f)(3), 363(f)(4), or 363(f)(5) of the Bankruptcy Code and are therefore adequately protected by having their Interests or claims that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Zokinvy Sale Transaction ultimately attributable to the property in which they have an interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Zokinvy Sale Transaction, subject to any defenses of the Debtors.
- 9. As further adequate protection, the Prepetition Term Loan Agent, on behalf of itself and the other Prepetition Term Loan Secured Parties, shall receive at closing of the sale of the Zokinvy Assets (as defined in the Court's bid procedures order [Docket No. 94] (the "Bid Procedures Order"), the amount of \$15 million from the net sale proceeds from the sale of the Zokinvy Assets (the "Adequate Protection Payment"). Additionally, as further adequate protection, the Debtors shall deposit the net proceeds from the sale of the Zokinvy Assets (less the Adequate Protection Payment) into a segregated bank account which account shall be subject to the liens in favor of the Prepetition Term Loan Secured Parties and shall not be used or

Casse2248800449sggj111 DDocc82621 Filitelc0244225225 Hintercelc0244225225171047595 DDess c MairElatoricutrien Pagle 20eo 29.103 112

expended by the Debtors for any purpose, or otherwise disbursed or transferred, without further notice, hearing (if required), and order of this Court.

10. **Release of Interests.** Any and all Persons that have filed a financing statement, mortgage, mechanic's lien, lis pendens, or other document or agreement evidencing an Interest against or in the Transferred Assets shall deliver to the Debtors prior to the Closing, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, releases, and/or any other similar documents necessary for the purpose of documenting all Interests that such Person has against or in the Transferred Assets. For any Person who has not delivered such termination statements to the Debtors prior to the Closing, then with respect to the holders of the Prepetition Liens, so long as the proceeds of the Zokinvy Sale Transaction shall have attached to the Prepetition Liens in the same order of priority as among such Prepetition Liens that existed prior to the Zokinvy Sale Transaction and with such Prepetition Liens retaining the same validity, force, and effect such Prepetition Liens had prior to the Zokinvy Sale Transaction, (i) the Debtors and/or the Purchaser are hereby authorized to execute and file such statements, instruments, releases, and/or other similar documents on behalf of such Person with respect to the Transferred Assets, provided, however, the Debtors and/or the Purchaser shall request written approval from the Prepetition Term Loan Agent (as defined in the Interim Cash Collateral Order) prior to executing or filing any document on behalf of the Prepetition Term Loan Secured Parties, (ii) the Purchaser is hereby authorized to file, register, or otherwise record a certified copy of this Zokinvy Sale Order that, once filed, registered, or otherwise recorded, shall constitute conclusive evidence of the release of all Interests of any kind or nature against or in the Transferred Assets, and (iii) the Purchaser may seek in this Court, or any other court of appropriate jurisdiction, to compel the appropriate parties to execute

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaqlPaqle30,co3801.13 112

termination statements, instruments of satisfaction, releases, and/or other similar documents with respect to all Interests that such Person has against or in the Transferred Assets. This Zokinvy Sale Order is deemed to be in recordable form sufficient to be placed in the filing or recording system of each and every federal, state, or local government agency, department, or office. Notwithstanding the foregoing, the provisions of this Zokinvy Sale Order authorizing the sale and assignment of the Transferred Assets free and clear of Interests shall be self-executing, and neither the Debtors nor the Purchaser shall be required to execute or file releases, termination statements, assignments, consents, or other instruments in order to effectuate, consummate, and implement the provisions of this Zokinvy Sale Order.

- 11. <u>Surrender of Transferred Assets</u>. All Persons that are presently or on the Closing Date may be in possession of some or all of the Transferred Assets are directed to surrender possession of such Transferred Assets to the Purchaser as of the Closing Date.
- 12. Continuation of Existing Approvals. The Purchaser shall be authorized, as of the Closing Date, to operate under any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the terms of the Amended Zokinvy Stalking Horse APA), and all such licenses, permits, registrations, and governmental authorizations or any other approvals are deemed to have been, and hereby are, directed to be transferred to the Purchaser as of the Closing Date. All existing licenses or permits applicable to the business shall remain active, in place, and, as applicable, shall be renewed for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred in accordance with applicable administrative procedures. To the maximum extent permitted by section 525(a) of the Bankruptcy Code, no governmental unit (as defined in Bankruptcy Code § 101(27)) or any representative thereof may

Casse2248800409ssjjj11 DDocc83821 Filitelc0244225225 Eintereclc0244225225171047595 Diessc MairEldothichtrien Pagle 202081103112

revoke or suspend, or in any way challenge or fail to consent to any renewal of any permit or license relating to the operation of the Transferred Assets because of the filing or pendency of the Debtors' chapter 11 cases or the consummation of the Zokinvy Sale Transaction.

13. **Injunction**. All Persons are hereby prohibited and enjoined from taking any action that would adversely affect or interfere with, or that would be inconsistent with, the ability of the Debtors to sell and transfer the Transferred Assets to the Purchaser in accordance with the terms of the Amended Zokinvy Stalking Horse APA, the Transaction Documents, or this Zokinvy Sale Order. Except as expressly permitted by the Amended Zokinvy Stalking Horse APA with respect to Permitted Liens and Assumed Liabilities or this Zokinvy Sale Order, all Persons (and their respective successors and assigns), including, without limitation, all holders of claims or Interests, lenders, debt security holders, governmental, tax and regulatory authorities, parties to executory contracts and unexpired leases, creditors, contract counterparties, customers, landlords, licensors, employees and former employees, litigation claimants, pension plans, labor unions, trade creditors, and other Persons holding Interests of any kind or nature whatsoever against or in the Debtors or the Transferred Assets (whether known or unknown, legal or equitable, matured or unmatured, contingent or non-contingent, liquidated or unliquidated, asserted or unasserted, whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases, whether imposed by agreement, understanding, law, equity, or otherwise), arising under or out of, in connection with, or in any way relating to, the Debtors, the operation of the Debtors' businesses prior to the Closing, the Transferred Assets, or the transfer of the Transferred Assets to the Purchaser (including, without limitation, any rights or claims based on any successor, transferee, derivative, or vicarious liabilities), shall be and hereby are forever barred, estopped, and permanently enjoined from asserting, prosecuting, or otherwise

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc024225225171047595 DDess c MainEldoiduithien Pagle age 36c082133112

pursuing any Interests against the Purchaser, any of its Affiliates, officers, directors, members, partners, principals, or shareholders, any of their respective representatives, successors, designees, or assigns, the property of the foregoing, and the Transferred Assets transferred to the Purchaser or interests of the Debtors in such Transferred Assets (other than Permitted Liens and the Assumed Liabilities). Following the Closing, no holder of an Interest against the Debtors shall interfere with the Purchaser's title to or use and enjoyment of the Debtors' former interests in the Transferred Assets, including, without limitation, taking any of the following actions with respect to or based on any Interest relating to the Transferred Assets or the transfer of the Transferred Assets to the Purchaser (other than Assumed Liabilities): (a) commencing or continuing in any manner any action or other proceeding against the Purchaser or its successors or assigns, assets or properties; (b) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order against the Purchaser or its successor, or assigns, assets, or properties; (c) creating, perfecting, or enforcing any Interest against the Purchaser, its successors or assigns, assets (including the Transferred Assets), or properties; (d) asserting any Interest as a setoff, right of subrogation, or recoupment of any kind against any obligation due Purchaser or its successors or assigns: (e) commencing or continuing any action in any manner or place that does not comply or is inconsistent with the provisions of this Zokinvy Sale Order or the agreements or actions contemplated or taken in respect thereof; (f) interfering with, preventing, restricting, prohibiting, or otherwise enjoining the consummation of the Sale Transactions; or (g) enforcing any provision of any Designated Contract that prohibits, restricts or conditions, or which purports to terminate or modify, or permits a party other than the Debtors to terminate or modify, any such Designated Contract, or any right or obligation under such Designated Contract, because of the assumption and assignment of such Designated Contract by the Seller to

the Purchaser. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Zokinvy Sale Order, any provision in a Designated Contract, any other document, or any applicable law that prohibits, restricts, or otherwise impairs assignment of the Designated Contracts or the Purchaser's ability to utilize the Transferred Assets in Purchaser's business is hereby void and of no force and effect with respect to the Zokinvy Sale Transaction, including without limitation any provision that (a) terminates or modifies any right or obligation of the Purchaser under such Designated Contract; (b) crossdefaults to or from any other lease or executory contract that is not a Designated Contract; (c) contains operating covenants or "go-dark" provisions that would purport to terminate or modify any Designated Contract before assumption and assignment to the Purchaser; (d) requires a third party's consent prior to assignment of the Designated Contract to the Purchaser; or (e) restricts the Purchaser's use or assignment of any licenses or similar permits if transferred. Notwithstanding the foregoing or any other provision of this Zokinvy Sale Order or the Amended Zokinvy Stalking Horse APA to the contrary, solely with respect to post-Closing claims (and for the avoidance of doubt, other than with respect to pre-Closing claims or defaults or defaults and/or any claims that arise as a result of the Zokinvy Sale Transaction).

14. **General Assignment**. As of the Closing, this Zokinvy Sale Order shall be construed and shall constitute for any and all purposes a full and complete general assignment, conveyance, and transfer of the Transferred Assets and/or a bill of sale or assignment transferring indefeasible title and interest in the Transferred Assets, including the Designated Contracts, to the Purchaser. Each and every federal, state, and local governmental agency or department is hereby authorized and directed to accept any and all documents and instruments

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc024225225171047595 DDess c MainEldoiduitn1en1PaglP35e08413112

necessary and appropriate to consummate the Zokinvy Sale Transaction and to reflect the effectiveness of the Zokinvy Sale Transaction.

No Successor, Transferee, or Similar Liability. The Purchaser, its Affiliates, 15. and any of their respective officers, directors, members, partners, principals, employees, independent contractors, and shareholders (or equivalent) and any of their respective representatives, agents, predecessors, successors, or assigns shall not be and shall not be deemed, as a result of the consummation of the Zokinvy Sale Transaction or otherwise, (i) to be a successor of, successor employer of, successor entity of, to have successorship obligations relating to, or to otherwise be deemed a successor, to the Debtors or the Debtors' estates, including with respect to any labor, employment, employee, personnel, or worker related matter, law, or agreement, including any collective bargaining agreement, works council agreement, union agreement, area labor agreement, multiemployer agreement, project labor agreement, construction agreement, contractor agreement, building agreement, regional agreement, work standards agreement, or other labor Contract (collectively, a "Collective Bargaining Agreement"), any employee benefit plans, any defined benefit pension plan, or any multiemployer plans, and the Purchaser and/or its Affiliates, as applicable, shall instead be, and be deemed to be, a new employer, including with respect to, among other things, any and all federal or state unemployment laws, including the Fair Labor Standards Act, any employee wage and hour law, privacy law, worker classification law, minimum wage law, overtime law, compensation or benefit law, meal or rest break law, time keeping law, employee record or documentation law, workers compensation law, unemployment compensation or tax law, or any other similar federal or state law (provided that the Purchaser shall pay employee-related liabilities solely to the extent expressly included in the Assumed Liabilities); (ii) to have any

Casse2248800449ssgjj111 DDocc818821 Filitelc0244225225 Hintereclc02442252251171047595 DDess c MairEldothichtrien1Pagle266c0851131112

common law successorship liability in relation to any Collective Bargaining Agreement, union, multiemployer organization, employee benefit plan, or multiemployer plan, including with respect to withdrawal liability or contribution obligations; (iii) to have, de facto or otherwise, merged or consolidated with or into any of the Debtors or any of the Debtors' estates, (iv) to be the successor of or a successor employer (as defined under COBRA and applicable regulations thereunder, common law, or otherwise) to the Debtors; (v) to have a common identity with the Debtors; (vi) to be an alter ego, joint employer, single employer, a continuation or substantial continuation, or to be holding itself out as a mere continuation, of any of the Debtors or their respective estates, or any enterprise of any of the Debtors, (vii) to be liable for any acts or omissions of the Seller or Debtors in connection with any Collective Bargaining Agreement, personnel, worker, employee, independent contractor, the conduct of the business, or the operation, funding, or administration of the employee benefit plans or multiemployer plans or arising under or related to the Transferred Assets other than as expressly set forth in the Amended Zokinvy Stalking Horse APA; (viii) to have any successor liability, transferee liability, derivative liability, vicarious liability, or any similar theories of any kind or character including, without limitation, under any theory of foreign, federal, state, or local antitrust, environmental, successor, tax, ERISA, assignee or transferee liability, labor, product liability, employment, de facto merger, substantial continuity, or other law, rule, regulation, or doctrine, whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated; (ix) except as expressly set forth in the Amended Zokinvy Stalking Horse APA, to have any successor liability, transferee liability, derivative, liability, vicarious liability, or any similar theories of any kind or character including under any pending, threatened, or potential claim, litigation, arbitration, settlement, investigation,

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc024225225171047595 DDess c MainEldoiduithien Pagle 276086133112

fact circumstance, or event disclosed in the Transaction Documents; in each case whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated, except to the extent solely and expressly provided for in the Amended Zokinvy Stalking Horse APA. The Purchaser shall not assume, or be deemed to assume, or in any way be responsible for any liability or obligation of any of the Debtors and/or their respective estates, or any of their predecessors or Affiliates. The so-called "bulk sales," "bulk transfer," or other similar laws shall be waived in all necessary jurisdictions, including those relating to Taxes. Except as expressly set forth in the Amended Zokinvy Stalking Horse APA with respect to Assumed Liabilities, the Purchaser, its Affiliates, officers, directors, members, partners, principals, and shareholders (or equivalent) and any of their respective representatives, successors, or assigns, or the Transferred Assets shall have no liability or responsibility whatsoever with respect to, or be required to satisfy in any manner, whether at law or in equity, whether by payment, setoff or otherwise, directly or indirectly (w) any Interest against the Debtors or against an insider of the Debtors, (x) any Interest or Excluded Liabilities, (y) the Debtors except as expressly set forth in the Amended Zokinvy Stalking Horse APA and the Transaction Documents.

Good Faith of the Purchaser. The Zokinvy Sale Transaction specified in the Stalking Horse APA is undertaken by the Purchaser without collusion and in good faith, as that term is defined in section 363(m) of the Bankruptcy Code and, accordingly, the reversal or modification on appeal of the authorization provided herein to consummate the sale shall not affect the validity of the Zokinvy Sale Transaction, including, without limitation, the assumption and assignment of the Designated Contracts, unless such authorization and consummation of the sale are duly and properly stayed pending such appeal. The Purchaser is a good faith purchaser

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaglPagleage3113112

within the meaning of section 363(m) of the Bankruptcy Code and, as such, is entitled to the full protections of section 363(m) of the Bankruptcy Code.

- No Avoidance of Stalking Horse APA. Neither the Debtors nor the Purchaser have engaged in any conduct that would cause or permit the Amended Zokinvy Stalking Horse APA to be avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy Code. Accordingly, the Amended Zokinvy Stalking Horse APA and the Zokinvy Sale Transaction shall not be avoidable under section 363(n) of the Bankruptcy Code, and no party shall be entitled to any damages or other recovery pursuant to section 363(n) of the Bankruptcy Code in respect of the Amended Zokinvy Stalking Horse APA or the Zokinvy Sale Transaction. Specifically, the Purchaser has not acted in a collusive manner with any person or entity and the Zokinvy Purchase Price was not controlled by any agreement among bidders.
- Debtors under the Designated Contracts arising prior to the Closing (without giving effect to any acceleration clauses or any default provisions of the kind specified in section 365(b)(2) of the Bankruptcy Code) as to which no objections were interposed and remain pending as of the date of this Zokinvy Sale Order are deemed satisfied by the payment of the proposed amount necessary, if any, to cure all monetary defaults, if any, under such Designated Contract in those amounts set forth in the Assignment Notice, which was served in compliance with the Bid Procedures Order, and which were satisfied, or shall be satisfied as soon as practicable. For all Designated Contracts for which an Assignment Notice was served, the Purchaser is authorized and directed to pay all Cure Amounts required to be paid by such parties upon the later of (a) the Closing, or (b) for any Designated Contract for which an objection has been filed to the assumption and assignment of such agreement or the Cure Amounts relating thereto and such

objection remains pending as of the date of this Zokinvy Sale Order (a "Cure Dispute"), within ten (10) business days of the resolution of such objection by settlement or order of this Court. Any non-Debtor counterparty to a Designated Contract that has not filed an objection on or before the deadline as set forth in the relevant Assignment Notice, or received an informal extension by the Debtors, shall be barred from objecting or asserting monetary or non-monetary defaults with respect to any such Designated Contract other than the applicable amount set forth in the Assignment Notice, and such Designated Contract shall be deemed assumed by the Debtors and assigned to the Purchaser on the Closing Date. To the extent that any Cure Dispute cannot be consensually resolved by the applicable parties, whether before or after the Closing Date, such Designated Contracts shall be assumed and assigned only upon satisfactory resolution of the Cure Dispute, to be determined in the Zokinvy Stalking Horse Purchaser's reasonable discretion. To the extent a Cure Dispute exists, the Designated Contracts may be conditionally assumed and assigned, subject to the consent of the Purchaser, pending a resolution of the Cure Dispute by agreement of the parties or after notice and a hearing. If a Cure Dispute is not satisfactorily resolved, the Purchaser may determine that such Designated Contracts should not be included on their schedule of Designated Contracts and should be rejected and not assigned, in which case the Purchaser will not be responsible for any Cure Amounts to the contract counterparty. The Debtors may then seek to reject the applicable contract or lease pursuant to Section 365 of the Bankruptcy Code.

19. <u>Determination of Cure Amounts</u>. Unless a counterparty to any Designated Contract has filed a timely Cure Objection which remains subject to an unresolved Cure Dispute as of the entry of this Zokinvy Sale Order, the Cure Amounts set forth on the Assignment Notices shall constitute findings of this Court and shall be final and binding on the counterparties

to the Designated Contracts and their successors and designees upon the Closing and shall not be subject to further dispute or audit based on performance prior to the time of assumption and assignment, irrespective of the terms and conditions of such Designated Contracts. counterparty to a Designated Contract (other than a counterparty who filed a timely Cure Objection) shall be forever barred, estopped, and permanently enjoined from (i) asserting against the Purchaser or its property (including, without limitation, the Transferred Assets), any default arising prior to or existing as of the Closing, or any counterclaim, defense, recoupment, setoff, or any other Interest asserted or assertable against the Debtors (except as otherwise provided herein), and (ii) imposing or charging against the Purchaser or its Affiliates, any accelerations, assignment fees, increases, or any other fees or charges as a result of the Debtors' assumption and assignment to the Purchaser of the Designated Contracts in connection with the Zokinvy Sale Transaction approved by this Zokinvy Sale Order. To the extent a counterparty to any of the Designated Contracts received notice of the Debtors' proposed Cure Amount and fails to file a Cure Objection by the applicable deadline, such party shall be deemed to have (a) consented to the assumption and assignment of the applicable Designated Contract and the payment of the Cure Amount provided in the Assignment Notices and (b) waived any right to assert or collect any other cure amount or enforce any default that may arise or have arisen prior to or as of the Closing.

20. **Payment of Cure Amounts**. With respect to the Designated Contracts, to the extent there are any Cure Amounts unpaid as of the Closing Date, the Purchaser shall be obligated, and is hereby directed, to pay or cause to be paid such Cure Amounts, unless a Cure Amount is subject to an unresolved Cure Dispute, in which case the Purchaser shall pay the Cure Amount in accordance with Paragraph 18 above, or unless the Debtors are otherwise required

Casse2248800409ssjýjj11 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDessc MairEldothichtrien Pagle 40,eo 40,103 112

under applicable law to make such payments prior to the Closing, in which case the Debtors shall obtain both the Prepetition Term Loan Administrative Agent and the Purchaser's written consent before making such payments and Purchaser shall reimburse the Debtors for such amounts paid by the Debtors, provided that the Purchaser shall receive a credit to the Purchase Price for any such Cure Amounts. The Purchaser's promise to perform the obligations under the Designated Contracts arising after their assumption and assignment to the Purchaser shall constitute adequate assurance of future performance within the meaning of sections 365(b) and 365(f)(2) of the Bankruptcy Code. On the Closing Date, subject in all respects to the terms of this Zokinvy Sale Order, the Purchaser shall be deemed to be substituted for the Seller (and/or any other Debtor, to the extent it holds any rights, title, or interests in any of the Designated Contracts) as a party to the applicable Designated Contracts.

21. **Ipso Facto Clauses Ineffective**. Upon the Debtors' assumption and assignment of the Designated Contracts to the Purchaser pursuant to this Zokinvy Sale Order and the payment of the Cure Amounts in accordance with this Zokinvy Sale Order and the Amended Zokinvy Stalking Horse APA, no default shall exist under any Designated Contract and no counterparty to any such Designated Contract shall be permitted to declare or enforce a default by the Debtors or the Purchaser thereunder or otherwise take action against the Purchaser as a result of any Debtor's financial condition, change in control, bankruptcy, or failure to perform any of its obligations under the applicable Designated Contract. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Zokinvy Sale Order, any provision in a Designated Contract that prohibits or conditions, whether directly or indirectly, the assignment of such Designated Contract (including, without limitation, the granting of an Interest therein) or allows the counterparty

Casse2248800409ssjjj11 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDessc MairEldothichtrien Pagle 4Qeo41103112

thereto to terminate, recapture, impose any penalty, condition on renewal or extension, or modify any term or condition upon such assignment shall be deemed an unenforceable anti-assignment provision that is void and of no force and effect with respect to the Zokinvy Sale Transaction as approved by this Zokinvy Sale Order. The failure of the Debtors or the Purchaser to enforce at any time one or more terms or conditions of any Designated Contract shall not be a waiver of such terms or conditions or of the Debtors' or the Purchaser's right, as applicable, to enforce every term and condition of such Designated Contract.

22. **Binding Effect**. This Zokinvy Sale Order and the Amended Zokinvy Stalking Horse APA shall be binding upon and shall govern the acts of all entities, including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other Persons who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to any of the Transferred Assets. The terms and provisions of the Amended Zokinvy Stalking Horse APA, the Transaction Documents, the Bid Procedures Order, and this Zokinvy Sale Order shall be binding in all respects upon the Debtors and their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Designated Contracts, the Purchaser and its respective successors and assigns, and any and all third parties, notwithstanding any subsequent appointment of any trustee, examiners, "responsible persons" or other fiduciaries (collectively, the "Trustee") of the Debtors under any Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 4Geo 42131112

chapter of the Bankruptcy Code, as to which Trustee such terms and provisions likewise shall be binding, and the Amended Zokinvy Stalking Horse APA (including the Designated Contracts) shall not be subject to rejection or avoidance under any circumstances.

- 23. **Release, Discharge, and Termination of Interests**. This Zokinvy Sale Order shall be effective as a determination that, on the Closing, all Interests of any kind or nature whatsoever existing prior to the Closing have been unconditionally released, discharged, and terminated solely as to the Transferred Assets (other than Permitted Liens and the Assumed Liabilities), and that the conveyances described herein have been effected.
- 24. Collaboration and Supply Agreement. For the avoidance of doubt, at the Closing, this Zokinvy Sale Order shall constitute approval of the assumption, assignment, and novation of that certain Amended and Restated Collaboration and Supply Agreement, dated as of February 29, 2024 (the "Collaboration and Supply Agreement") by and between Eiger BioPharmaceuticals, Inc. and the Progeria Research Foundation, Inc. ("PRF"), pursuant to Section 365 of the Bankruptcy Code and subject to the terms of the PRF Novation Agreement (as defined in the Amended Zokinvy Stalking Horse APA). The Collaboration and Supply Agreement is deemed valid and binding and in full force and effect. At the Closing, the Collaboration and Supply Agreement shall be assumed by the Debtors and, subject to the terms in the PRF Novation Agreement, shall be assigned to Purchaser and Purchaser shall take assignment from the Debtors of the Collaboration and Supply Agreement and it shall be deemed valid and binding on Purchaser. Notwithstanding anything to the contrary provided herein or in any cure notice, except as expressly set forth in the Amended Zokinvy Stalking Horse APA, the PRF Novation Agreement, or any of the other Transaction Documents, the assumption, assignment, and novation of the Collaboration and Supply Agreement shall not alter, impair,

Casse2248800409sggj111 DDocc828821 Filitelc0044225225 Hinterenclc0044225225171047595 DDess MairENdoibitrienPage 44eo43.103 112

modify, or otherwise affect any of the parties' respective rights and obligations under the Collaboration and Supply Agreement, whether legal, equitable or contractual. For the avoidance of doubt, Purchaser shall not and is not assuming, and shall not otherwise have, any liability or obligations under the Collaboration and Supply Agreement or with respect to PRF solely to the extent arising prior to the Closing.

- 25. That certain *Guaranty*, dated April 15, 2024, by and between Zydus Pharmaceuticals (USA) Inc. ("Zydus"), an affiliate of Purchaser, and PRF (the "Zydus Guaranty") is hereby approved and ratified. Following Closing, if it occurs, Purchaser and Zydus shall, subject to the terms and conditions of the Zydus Guaranty, be responsible for all post-Closing obligations, whether performance, financial, or otherwise, arising under or in respect of the Collaboration and Supply Agreement. To the extent anything contained in this Zokinvy Sale Order conflicts with a provision in the Zydus Guaranty, the Zydus Guaranty shall govern and control.
- 26. No Material Modifications. The Amended Zokinvy Stalking Horse APA and the Transaction Documents may be modified, amended, or supplemented by the Debtors and the Purchaser, in a writing signed by such parties, and in accordance with the terms thereof, without further order of this Court; *provided*, that (i) any such modification, amendment, or supplement does not have a material adverse effect on the Debtors' estates or its creditors, and (ii) has been agreed to between the Debtors and the Purchaser (with such consent not to be unreasonably withheld) and approved by the Prepetition Term Loan Administrative Agent. Any material modification, amendment, or supplement to the Amended Zokinvy Stalking Horse APA and the Transaction Documents adversely affecting the Debtors' estates must be approved by order of this Court following a motion on notice to all interested parties.

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hintereclc0044225225171047505 DDess MairENholdetritenPage 45e044131112

- 27. <u>Subsequent Orders and Plan Provisions</u>. Nothing contained in any chapter 11 plan confirmed in the Debtors' chapter 11 cases or any subsequent order of this Court, including, without limitation, any order confirming any such chapter 11 plan, any order authorizing the sale of assets of the Debtors pursuant to any section of the Bankruptcy Code, and any order approving wind-down or dismissal of any Debtor's chapter 11 case or any subsequent chapter 7 case shall change, supersede, abrogate, nullify, restrict, or conflict with the provisions of the Amended Zokinvy Stalking Horse APA, the Transaction Documents, or this Zokinvy Sale Order, or in any way prevent or interfere with the consummation or performance of the Zokinvy Sale Transaction.
- 28. **Failure to Specify Provisions**. The failure to specify or include any particular provisions of the Amended Zokinvy Stalking Horse APA or the Transaction Documents in this Zokinvy Sale Order shall not diminish or impair the effectiveness of such provisions, it being the intent of this Court that the Amended Zokinvy Stalking Horse APA, the Transaction Documents, and the Zokinvy Sale Transaction be authorized and approved in their entirety.
- 29. <u>Automatic Stay</u>. The automatic stay pursuant to section 362 of the Bankruptcy Code is hereby lifted solely to the extent necessary to (i) allow the Purchaser to deliver any notice provided for in the Amended Zokinvy Stalking Horse APA and the Transaction Documents, and (ii) allow the Purchaser to take any and all actions permitted under the Amended Zokinvy Stalking Horse APA and the Transaction Documents in accordance with the terms and conditions thereof. The automatic stay imposed by section 362 of the Bankruptcy Code shall be modified solely to the extent necessary to implement the preceding sentence, and this Court shall retain exclusive jurisdiction over any and all disputes with respect thereto.

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEtholocitritentPagtPat6eo45113112

- Bankruptcy Rules Satisfied or Waived. The requirements set forth in Bankruptcy Rules 6004 and 6006 have been satisfied or are otherwise deemed to be waived. As provided by Bankruptcy Rule 9014, the terms of this Zokinvy Sale Order shall be effective and enforceable immediately upon entry, and shall not be subject to stay provisions contained in Bankruptcy Rules 6004(h) and 6004(d). Time is of the essence in closing the Zokinvy Sale Transaction and the Debtors and the Purchaser intend to close the sale as soon as possible.
- 31. <u>Conflicts Between Sale Order and Stalking Horse APA</u>. To the extent anything contained in this Zokinvy Sale Order conflicts with a provision in the Amended Zokinvy Stalking Horse APA or Transaction Documents, this Zokinvy Sale Order shall govern and control. Notwithstanding the foregoing, nothing in this Zokinvy Sale Order shall modify or waive any closing conditions or termination rights in the Amended Zokinvy Stalking Horse APA, and all such conditions and rights shall remain in full force and effect in accordance with their terms.
- 32. **Provisions Nonseverable and Mutually Dependent**. The provisions of this Zokinvy Sale Order, the Amended Zokinvy Stalking Horse APA, and the Transaction Documents are non-severable and mutually dependent.
- 33. Retention of Jurisdiction. This Court shall retain exclusive jurisdiction to, among other things, interpret, implement, and enforce the terms and provisions of the Amended Zokinvy Stalking Horse APA, the Transaction Documents, the Bid Procedures Order, and this Zokinvy Sale Order, and each of the agreements executed in connection therewith to which the Debtors are a party or which has been assigned to the Purchaser by the Debtors, and to adjudicate, if necessary, any and all disputes concerning or relating in any way to the Zokinvy Sale Transaction. This Court retains jurisdiction to compel delivery of the Transferred Assets, to

Casse2248800409sgijj11 DDoc38821 Filitelc0044225225 Eintereclc0044225225171047595 DDessc MairElatoricutriien Pagle 4geo 46.103 112

protect the Purchaser and its assets, including the Transferred Assets, against any Interests or successor or transferee liability and to enter orders, as appropriate, pursuant to sections 105(a), 363, or 365 (or other applicable sections) of the Bankruptcy Code necessary to transfer the Transferred Assets and the Designated Contracts to the Purchaser. In the event this Court abstains from exercising or declines to exercise jurisdiction with respect to any matter referenced in this paragraph or is without jurisdiction, such abstention, refusal, or lack of jurisdiction shall have no effect upon and shall not control, prohibit, or limit the exercise of jurisdiction of any other court having competent jurisdiction with respect to any such matter.

- 34. The Purchaser has standing to seek to enforce any terms of this Zokinvy Sale Order, the Bid Procedures Order, the Amended Zokinvy Stalking Horse APA, and the Transaction Documents in this Court or any other court with competent jurisdiction.
- 35. All time periods set forth in this Zokinvy Sale Order shall be calculated in accordance with Bankruptcy Rule 9006(a).

END OF ORDER

Submitted By:

SIDLEY AUSTIN LLP

Thomas Califano (TX Bar No. 24122825) William Curtin (admitted *pro hac vice*) Anne Wallice (admitted *pro hac vice*) 787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Charles Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

Proposed Attorneys for the Debtors and Debtors in Possession

Exhibit 1

Amended Zokinvy Stalking Horse APA

ASSET PURCHASE AGREEMENT

by and between

SENTYNL THERAPEUTICS, INC., as Purchaser,

and

EIGER BIOPHARMACEUTICALS, INC., as Seller

Dated as of March 31, 2024

Table of Contents

| ARTICLE 1. | DEFINED TERMS | 2 |
|------------|---|----|
| 1.1 | Defined Terms | |
| 1.2 | Other Definitional and Interpretive Matters | |
| ARTICLE 2. | THE PURCHASE AND SALE; CLOSING | 15 |
| 2.1 | Purchase and Sale | 15 |
| 2.2 | Excluded Assets | 16 |
| 2.3 | Assumption of Liabilities | 17 |
| 2.4 | Excluded Liabilities | 18 |
| 2.5 | Excluded Contracts | 18 |
| 2.6 | Nontransferable Assets and Liabilities | 18 |
| 2.7 | <u>Closing</u> | 19 |
| 2.8 | Closing Deliveries of the Parties | 19 |
| 2.9 | Purchase Price; Assumed Liabilities; Deposits | 20 |
| 2.10 | Transfer Taxes | |
| 2.11 | Allocation of Purchase Price | 21 |
| 2.12 | Escrow Accounts | 22 |
| 2.13 | Tax Withholding | 22 |
| ARTICLE 3. | REPRESENTATIONS AND WARRANTIES OF THE SELLER | 22 |
| 3.1 | Organization, Good Standing and Other Matters | 22 |
| 3.2 | Authority and Enforceability | 23 |
| 3.3 | No Conflict; Required Filings and Consents | 23 |
| 3.4 | Compliance With Laws | 23 |
| 3.5 | Permits | |
| 3.6 | <u>Litigation</u> | 24 |
| 3.7 | Real Property; Personal Property | 24 |
| 3.8 | Assigned Contracts | 25 |
| 3.9 | Financial Statements. | 25 |
| 3.10 | Absence of Material Developments. | 25 |
| 3.11 | Customers and Suppliers. | |

Page

Casse22488004409sgijj.11.1 DDocc836821 Ffilitelc0044225225 Hinteredc004422522511/10467595 DDessc MainElonichitm1enPagle5geo51103112

| 3.12 | <u>Intellectual Property</u> | 26 |
|------------|--|----|
| 3.13 | <u>Inventories.</u> | 26 |
| 3.14 | <u>Taxes</u> | 27 |
| 3.15 | Product Liability | 27 |
| 3.16 | Product Warranties; Product Return | 27 |
| 3.17 | Accounts Payable | 27 |
| 3.18 | Brokers and Finders | 27 |
| 3.19 | No Other Representations or Warranties | 27 |
| ARTICLE 4. | REPRESENTATIONS AND WARRANTIES OF PURCHASER | 28 |
| 4.1 | Organization, Good Standing and Other Matters | 28 |
| 4.2 | Authority and Enforceability | 28 |
| 4.3 | No Conflict: Required Filings and Consents | 28 |
| 4.4 | Financing | 29 |
| 4.5 | <u>Solvency</u> | 29 |
| 4.6 | <u>Litigation</u> | 29 |
| 4.7 | Brokers and Finders | 29 |
| 4.8 | Investigation and Agreement by Purchaser; Non-Reliance of Purchaser; No Other Representations and Warranties | 29 |
| 4.9 | No Other Representations or Warranties | |
| ARTICLE 5. | BANKRUPTCY COURT MATTERS | 31 |
| 5.1 | Competing Transaction | 31 |
| 5.2 | Bankruptcy Court Filings | 31 |
| 5.3 | Assumption of Assigned Contracts | 33 |
| ARTICLE 6. | PRE-CLOSING COVENANTS | 34 |
| 6.1 | Conduct of Business. | 34 |
| 6.2 | Access to Information; Confidentiality | 35 |
| 6.3 | Efforts to Consummate. | 36 |
| 6.4 | Notices and Consents | 37 |
| 6.5 | Regulatory Matters | 37 |
| 6.6 | Public Announcements | 38 |
| 6.7 | Update of Schedules; Knowledge of Breach | 38 |
| ARTICLE 7. | POST-CLOSING COVENANTS | 38 |
| 7.1 | Access to Information; Books and Records | 38 |

Casse22488004409sgijj.11.1 DDocc836821 Ffilitelc0044225225 Hinteredc004422522511/10467595 DDessc MainElonichitm1enPagle566052103112

| 7.2 | Post-Closing Receipt and Possession of Assets | 39 |
|-------------|---|----|
| 7.3 | <u>Tax Matters</u> | 39 |
| ARTICLE 8. | CONDITIONS PRECEDENT | 40 |
| 8.1 | Conditions to Each Party's Obligation | 40 |
| 8.2 | Conditions to Obligations of Purchaser | |
| 8.3 | Conditions to Obligations of the Seller | |
| 8.4 | Waiver of Condition; Frustration of Conditions | |
| ARTICLE 9. | TERMINATION | 41 |
| 9.1 | Events of Termination | 41 |
| 9.2 | Effect of Termination | 43 |
| 9.3 | Termination Fee and Expense Reimbursement | 43 |
| ARTICLE 10. | GENERAL PROVISIONS | 45 |
| 10.1 | Survival of Representations, Warranties and Covenants | 45 |
| 10.2 | Entire Agreement | |
| 10.3 | Amendment; No Waiver | 45 |
| 10.4 | Severability; Specific Versus General Provisions | 46 |
| 10.5 | Expenses and Obligations | 46 |
| 10.6 | <u>Notices</u> | 46 |
| 10.7 | <u>Counterparts</u> | 47 |
| 10.8 | Governing Law | 48 |
| 10.9 | Submission to Jurisdiction; Consent to Service of Process | 48 |
| 10.10 | Waiver of Jury Trial | 48 |
| 10.11 | Rights Cumulative | 48 |
| 10.12 | <u>Assignment</u> | 49 |
| 10.13 | Specific Enforcement; Remedies | 49 |
| 10.14 | Third-Party Beneficiaries | 49 |
| 10.15 | No Personal Liability of Directors, Officers and Owners | 50 |
| 10.16 | General Release | 50 |
| 10.17 | Legal Representation | 51 |

SCHEDULES

[forthcoming]

Casse2248800449sgijj.11 Diboc83621 Ffilitelc0044225225 Hintereelc00442252251710467595 Dibosc MainElohibitm1enPagle54c053.103112

EXHIBITS

| Exhibit A | Form of Escrow Agreement |
|-----------|--|
| Exhibit B | Form of Bill of Sale and Assignment and Assumption |
| | Agreement |
| Exhibit C | Merck Side Letter |
| Exhibit D | PRF Novation Agreement |
| Exhibit E | Sublicense Agreement |

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of March 31, 2024 (the "<u>Agreement Date</u>") is entered into by and between Sentynl Therapeutics, Inc., a Delaware Corporation ("<u>Purchaser</u>") and Eiger BioPharmaceuticals, Inc., a Delaware corporation (the "<u>Seller</u>").

RECITALS

WHEREAS, on March 31, 2024 (the "<u>Petition Date</u>") the Seller and certain of its Affiliates (as defined below) filed voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the Northern District of Texas (the "<u>Bankruptcy Court</u>"), thereby commencing chapter 11 cases (collectively, the "<u>Bankruptcy Cases</u>");

WHEREAS, the Seller is a debtor-in-possession under the Bankruptcy Code and manages its properties and assets pursuant to Sections 1107(a) and 1108 of the Bankruptcy Code;

WHEREAS, the Seller is engaged in the Business and owns, directly or indirectly, all of the Transferred Assets;

WHEREAS, the Seller desires to sell (or cause to be sold) to Purchaser, and Purchaser desires to purchase from the Seller, all of the Transferred Assets Free and Clear, and the Seller desires Purchaser to assume, and Purchaser desires to assume from the Seller, all of the Assumed Liabilities, in each case upon the terms and subject to the conditions hereof, pursuant to a Sale Order and Sections 105(a), 363 and 365 of the Bankruptcy Code and Rules 6004 and 6006 of the Federal Rules of Bankruptcy Procedure;

WHEREAS, the Transactions contemplated by this Agreement are subject to approval by the Bankruptcy Court and will only be consummated pursuant, among other things, to the Sale Order to be entered in the Bankruptcy Cases; and

WHEREAS, concurrently with the execution of this Agreement, Purchaser shall deposit (or cause to be deposited) an aggregate amount equal to the Deposit Escrow Amount into an escrow account (the "<u>Deposit Escrow Account</u>") to be established and maintained by Escrow Agent pursuant to the Escrow Agreement.

NOW, **THEREFORE**, in consideration of the premises and the mutual representations, warranties, covenants, agreements and conditions set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE 1. DEFINED TERMS

- 1.1 <u>Defined Terms</u>. The following terms shall have the following meanings in this Agreement:
- "<u>Action</u>" means any action, proceeding, arbitration or litigation (whether civil, criminal or administrative) commenced, brought, conducted or heard by or before any Governmental Authority or arbitrator.
- "Affiliate" of any particular Person means any other Person, directly or indirectly, controlling, controlled by, or under common control with, such particular Person. For the purposes of this definition, "control" (including, with correlative meaning, the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.
 - "Agreement" has the meaning set forth in the preamble.
 - "Agreement Date" has the meaning set forth in the preamble.
 - "Allocation Schedule" has the meaning set forth in Section 2.11(a).
 - "Alternate Transaction" has the meaning set forth in Section 9.1(b).
- "<u>Applicable Law</u>" means, with respect to any Person, any federal, provincial, state, local law, ordinance, principle of common law, code, regulation or statute applicable to such Person or such Person's subsidiaries or to any of their respective securities, assets, properties or businesses.
- "<u>Asset Taxes</u>" means any Taxes with respect to the ownership or operation of the Transferred Assets other than (a) Taxes based on net or gross income, and (b) Transfer Taxes.
 - "Assigned Contracts" has the meaning set forth in Section 2.1(d).
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - "Assumption Notice" has the meaning set forth in Section 5.3(a).
 - "<u>Attorney-Client Information</u>" has the meaning set forth in <u>Section 10.17</u>.
 - "Auction" has the meaning set forth in Section 5.2(i).
- "Avoidance Actions" means any and all avoidance, recovery, subordination, or other claims, actions, rights, or remedies that may be brought by or on behalf of the Seller or its estate or other authorized parties in interest under the Bankruptcy Code or applicable non-bankruptcy law, including, but not limited to, actions or remedies under sections 510, 542, 543, 544, 545, and 547 through and including 553 of the Bankruptcy Code.

Casse2248800409sgijj11 DDocc818621 Filitelc0044225225 Hintercelc00442252251171047595 DDess MainEtaloidaitn1en1Pagle16361138112

- "<u>Back-Up Bid</u>" means the second highest or otherwise best bid if the successful bidder fails to consummate its bid in accordance with the Bid Procedures.
- "<u>Back-up Termination Date</u>" means the first to occur of (a) thirty (30) days after the entry of the Sale Order, (b) consummation of the Transactions with the winning bidder at the Auction, (c) Purchaser's receipt of notice from the Seller of the release by the Seller of Purchaser's obligations under <u>Section 5.2(i)</u> and (d) March 13, 2024.
 - "Bankruptcy Cases" has the meaning set forth in the Recitals.
 - "Bankruptcy Code" has the meaning set forth in the Recitals.
 - "Bankruptcy Court" has the meaning set forth in the Recitals.
- "Base Price" means \$26,000,000 provided, however, that the Base Price shall be reduced by the amount of \$214,285.71 per diem for each calendar day that the Closing occurs between April 24, 2024, and May 31, 2024; provided, further, that, notwithstanding the reduction, the Base Price shall not be less than \$20,000,000 if Closing occurs no later than May 31, 2024.
- "<u>Bid Procedures</u>" means those certain bidding procedures for the Sale of the Seller's assets approved by the Bankruptcy Court.
- "<u>Bid Procedures Motion</u>" means a motion filed by Seller with the Bankruptcy Court to seek approval of the Bid Procedures.
- "<u>Bid Procedures Order</u>" means an Order of the Bankruptcy Court approving the Bid Procedures.
- "<u>Bill of Sale and Assignment and Assumption Agreement</u>" means the bill of sale and assignment and assumption agreement, dated as of the Closing Date, by and between the Seller and Purchaser, substantially in the form attached hereto as <u>Exhibit B</u>.
- "<u>Business</u>" means the business as presently conducted of the Seller Group related to the development, manufacture, sale, maintenance, and commercialization of Zokinvy in the Progeria Field (as such term is defined in the Sublicense Agreement) in the Territory.
- "Business Day" means any day other than (a) a Saturday, Sunday or federal holiday or (b) a day on which commercial banks in Seattle, Washington are authorized or required to be closed.
- "Business Intellectual Property" means all Owned Intellectual Property Assets together with all other Intellectual Property used in, held for use in, or necessary for the conduct of the Business.
- "Buyer's FDA Transfer Letters" means the letter to FDA in form and substance reasonably agreed by Purchaser and the Seller, accepting the transfer of rights to the NDA issued by FDA for Zokinvy in the Progeria Field from Seller.

Casse2248800409sggj111 DDocc8B821 Filitelc0044225225 Hinterenclc00442252251171047505 DDess MainEtatolicutriten Pagle 566c057113 112

- "Closing" has the meaning set forth in Section 2.7.
- "Closing Date" has the meaning set forth in Section 2.7.
- "Code" means the Internal Revenue Code of 1986, as amended, or any successor law.
- "Competing Bid" has the meaning set forth in Section 5.1.
- "Confidentiality Agreement" means that certain Confidentiality Agreement, dated as of July 26, 2023, by and between the Seller and Purchaser.
 - "Consent" means any consent, approval, authorization, waiver or license.
- "Contract" means any written agreement, mortgage, indenture, lease (whether for real or personal property), contract or subcontract.
 - "Contracting Parties" has the meaning set forth in Section 10.15
- "Cure Costs" means any and all costs, expenses or actions that Purchaser is required to pay or perform to assume any of the Assigned Contracts pursuant to section 365(b)(1)(A) and (B) of the Bankruptcy Code.
 - "Deposit Escrow Account" has the meaning set forth in the Recitals.
 - "Deposit Escrow Amount" means \$1,300,000.
 - "Designated Contracts" has the meaning set forth in Section 5.3(b).
 - "<u>Designation Deadline</u>" has the meaning set forth in <u>Section 5.3(b)</u>.
- "<u>Determined Cure Costs</u>" means all Cure Costs for Assigned Contracts, as determined by a final order of the Bankruptcy Court.
- "Enforceability Exceptions" means applicable bankruptcy, insolvency, reorganization, moratorium, receivership and similar Applicable Laws affecting the enforcement of creditors' rights generally and general equitable principles.
- "Environmental Laws" means any Applicable Law relating to pollution or protection of the environment or worker health and safety (in respect of exposure to Hazardous Substances), including such Applicable Laws relating to the use, treatment, storage, disposal, Release or transportation of Hazardous Substances.
 - "Escrow Agent" means Kurtzman Carson Consultants LLC.
- "Escrow Agreement" means the escrow agreement, dated as of the Agreement Date, by and among Purchaser, the Seller and the Escrow Agent in substantially the form attached hereto as Exhibit A.
 - "Excluded Assets" has the meaning set forth in Section 2.2.

"Excluded Books and Records" means the following originals and copies of those books and records, documents, data and information (in whatever form maintained) of the Seller Group and the Business: (i) all corporate minute books (and other similar corporate records) and stock records of the Seller Group, (ii) any books and records relating to the Excluded Assets or (iii) any books, records or other materials that any member of the Seller Group (x) is required by Applicable Law to retain (copies of which, to the extent permitted by Applicable Law, will be made available to Purchaser upon Purchaser's reasonable request), (y) reasonably believes is necessary to enable it to prepare and/or file Tax Returns (copies of which will be made available to Purchaser upon Purchaser's reasonable request) or (z) are prohibited by Applicable Law from delivering to Purchaser.

"Excluded Contracts" has the meaning set forth in Section 2.5.

"Excluded Liabilities" has the meaning set forth in Section 2.4.

"Expense Reimbursement" means the reimbursement by the Seller of Purchaser's actual and reasonable out-of-pocket legal, accounting, and other third-party advisory or service costs and expenses incurred in connection with the Transactions, as evidenced by invoice(s) provided to the Seller, on the terms and subject to the conditions of Section 9.3.

"FDA" means the United States Food and Drug Administration.

"<u>Final Order</u>" means an Order, judgment or other decree of the Bankruptcy Court or any other Governmental Authority of competent jurisdiction that has not been reversed, vacated, modified or amended, is not stayed and remains in full force and effect; <u>provided</u>, that such Order shall be considered a Final Order only after the time period for third parties seeking appeal has expired without the filing of any appeal or motion for reconsideration.

"<u>Free and Clear</u>" means free and clear of all Liens (other than the Permitted Liens and the Assumed Liabilities) to the maximum extent permitted by Section 363(f) of the Bankruptcy Code.

"GAAP" means generally accepted accounting principles in the United States as of the Agreement Date.

"Governmental Authority" means any domestic or foreign national, provincial, state, multi-state or municipal or other local government, any subdivision, agency, commission or authority thereof, any court (including the Bankruptcy Court) or tribunal or any quasi-governmental or private body exercising any regulatory or taxing authority thereunder (including the IRS and the FDA).

"<u>Hazardous Substances</u>" means any substances, materials or wastes which are defined as or included in the definition of "hazardous substances", "hazardous wastes", "hazardous materials", "toxic substances", "pollutants" or "contaminants" under any Environmental Law, including any petroleum or refined petroleum products, radioactive materials, friable asbestos or polychlorinated biphenyls.

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaqlP60e059.133112

"Intellectual Property" means any and all intellectual property and proprietary rights in any jurisdiction throughout the world, including rights arising from the following: (i) patents and patent applications, design rights, industrial design registrations and applications therefor, divisions, continuations, continuations-in-part, reissues, substitutes, renewals, registrations, confirmations, reexaminations, extensions and any provisional applications, and any foreign or international equivalent of any of the foregoing; (ii) trademarks (whether registered, unregistered or applied for), service marks, trade dress, service names, trade names, brand names, product names, slogans, logos, business names, corporate names, and other source or business identifiers, all registrations and applications for registration thereof, and, in each case, together with all of the goodwill associated therewith; (iii) works of authorship, copyrights and all registrations and applications for registration thereof; (iv) trade secrets and know-how; (v) rights in formulae, methods, techniques, processes, assembly procedures, software, software code (in any form, including source code and executable or object code), subroutines, test results, test vectors, user interfaces, protocols, schematics, specifications, drawings, prototypes, molds and models, and other forms of technology (whether or not embodied in any tangible form and including all tangible embodiments of the foregoing), and (vi) social media accounts, social media identifiers, internet domain name registrations.

"<u>Intellectual Property Assignment Agreement</u>" means the assignment agreement assigning the Intellectual Property to Purchaser, in a form reasonably acceptable to Purchaser and the Seller and executed and delivered at Closing.

"Intellectual Property Registrations" means, as to any Owned Intellectual Property Assets, any issuance, registration, application or other filing by, to or with any Governmental Authority or authorized private registrar in any jurisdiction, including domain names, registered trademarks and copyrights, issued and reissued patents and pending applications for any of the foregoing.

"IRS" means the United States Internal Revenue Service.

"Knowledge" means (a) with regard to the Seller, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Seller's chief executive officer, chief financial officer, and general counsel, in each case as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate) and (b) with regard to Purchaser, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Purchaser's chief executive officer as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate).

"Law Firm" means Sidley Austin LLP and its successors.

"<u>Liabilities</u>" shall mean debts, liabilities, duties, obligations or commitments of any nature whatsoever, whether direct or indirect, asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured or otherwise, whenever or however arising (including whether arising out of any Contract or in a tort claim based on negligence or strict liability).

"Lien" means all forms of lien (including mechanic's, contractor's or other similar liens arising under or relating to the provision of goods or services on or to any Transferred Assets, and liens arising under the Bankruptcy Code), encumbrance, defect or irregularity in title, pledge, mortgage, deed of trust, deed to secure debt, security interest, charge, transfer restriction or similar agreement or encumbrance, including any dedication under any gathering, transportation, treating, processing, fractionating, purchase, sale or similar agreements, or any other rights granted or consensual as or against any Transferred Assets including but not limited to easements, encroachments, rights of first refusal, options, or any other interest or right in property that constitutes a lien or interest within the definition or adjudication of such terms under Section 101(37) of the Bankruptcy Code.

"Material Adverse Effect" means a material adverse effect on the business, financial condition or results of operations of the Business (including the Transferred Assets and Assumed Liabilities) taken as a whole; provided, however, that none of the following shall be deemed (either alone or in combination) to constitute, and none of the following shall be taken into account in determining whether there has been or may be, a Material Adverse Effect: (a) any change in, or effects arising from or relating to, general business or economic conditions affecting any industry in which the Business operates; (b) any change in, or effects arising from or relating to, the United States or foreign economies, or securities, banking or financial markets in general, or other general business, banking, financial or economic conditions (including (i) any disruption in any of the foregoing markets, (ii) debt defaults or other restructuring events of any country with respect to which bondholders take a discount to the debt of any country or any increases in the interest rates for any country's debt, (iii) any change in currency exchange rates, (iv) any decline or rise in the price of any security, commodity, contract or index and (v) any increased cost, or decreased availability, of capital or pricing or terms related to any financing for the Transactions); (c) any change from, or effects arising from or relating to, the occurrence, escalation or material worsening of any act of God or other calamity, natural disaster, pandemic or disease, outbreak, hostility, act of war, sabotage, cyber-attack or terrorism or military action; (d) any action taken by Purchaser or its Affiliates with respect to the Transactions or with respect to the Business; (e) any action taken, or failed to be taken, by the Seller at the request of or with the consent of Purchaser or otherwise in compliance with the terms of this Agreement or any change from, or effects arising from or relating to, Purchaser's failure to consent to any action restricted by Section 6.1; (f) any change in, or effects arising from or relating to changes in, Applicable Law or accounting rules (including GAAP) or any interpretation thereof; (g) the failure of the Business to meet any of its projections, forecasts, estimates, plans, predictions, performance metrics or operating statistics or the inputs into such items (whether or not shared with Purchaser or its Affiliates or representatives); (h) national or international political, labor or social conditions; (i) the public announcement of, entry into or pendency of, actions required or contemplated by or performance of obligations under, this Agreement and the Transactions or the identity of the parties to this Agreement; (j) the sale of any assets other than the Transferred Assets to any third parties by a member of the Seller Group or any of their Affiliates; (k) any effect arising or resulting from or related to the filing of the Bankruptcy Cases; (1) any action required to be taken under any Applicable Law or Order or any existing Contract by which any member of the Seller Group's (or any of their properties) are bound; (m) seasonal changes in the results of operations of the Seller Group; (n) any epidemic, pandemic, outbreak of disease or other public health emergency (including COVID-19) or any escalation or worsening of any such conditions or (o) any objections made in the Bankruptcy Court to this Agreement, the

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 62eo61131112

Transactions, the Sale Order or the reorganization, any orders of the Bankruptcy Court and any actions or omissions of the Seller in compliance with any order of the Bankruptcy Court and the assumption or rejection of any Assigned Contract; except in the cause of clauses (a) through (c), (h) and (n), to the extent such conditions, events, changes, crises and disasters, as applicable, do not have a material and disproportionate impact on the Business, taken as a whole, compared to other industry participants (in which case, only the extent of such disproportionate effect shall be taken into account when determining whether there is a Material Adverse Effect).

"Merck" means Merck Sharp & Dohme Corp. (successor-in-interest of Schering Corporation).

"Merck Side Letter" means the letter agreement with Merck substantially in the form attached hereto as Exhibit C duly executed by each of Merck and the Seller.

"New Drug Application" or "NDA" means new drug application as approved by the FDA.

"Non-Transferred Asset" has the meaning set forth in Section 2.6(a).

"Nonparty Affiliates" has the meaning set forth in Section 10.15.

"Open Source Software" means any software that is licensed pursuant to a license approved by the Open Source Initiative and listed at http://www.opensource.org/licenses/alphabetical or that is considered "free" or "open source software" by the Free Software Foundation.

"Order" means any award, decision, injunction, judgment, ruling or verdict entered, issued, made or rendered by any Governmental Authority or arbitrator.

"Organizational Documents" means (a) the articles or certificates of incorporation and the by-laws of a corporation, (b) the partnership agreement and any statement of partnership of a general partnership, (c) the limited partnership agreement and the certificate of limited partnership of a limited partnership, (d) the operating or limited liability company agreement and the certificate of formation of a limited liability company, (e) any charter, joint venture agreement or similar document adopted or filed in connection with the creation, formation or organization of a Person not described in clauses (a) through (d), and (f) any amendment to or equivalent of any of the foregoing.

"Outside Date" has the meaning set forth in Section 9.1(i).

"Owned Intellectual Property Assets" means the Intellectual Property owned or purported to be owned by any of member of the Seller Group that is used in, held for use in, or related to, the conduct of the Business as currently conducted or proposed to be conducted.

"Permit" means all permits, authorizations, certificates, franchises, consents and other approvals from any Governmental Authority.

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1Pagle66e062131112

"Permitted Liens" means (a) Liens for Taxes, assessments or other governmental charges not yet due and payable or being contested in good faith by appropriate proceedings as set forth on Schedule 1.1(a); (b) mechanics', carriers', workers', repairers' and other similar Liens arising or incurred in the ordinary course of business for obligations that are not overdue or are being contested in good faith by appropriate proceedings; (c) zoning, entitlement and building regulations and land use restrictions; (d) purchase money Liens and Liens securing rental payments under capital lease arrangements; (e) Liens arising under leases of property or equipment in favor of the owner thereof; (f) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security; (g) deposits to secure the performance of bids, Contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business; (h) licenses of Intellectual Property granted in the ordinary course of business; (i) [reserved]; (j) Liens arising under or created by this Agreement or any of the Related Documents; (k) Liens arising in the ordinary course of business which would not reasonably be expected to have a Material Adverse Effect; and (1) Liens set forth on Schedule 1.1(b).

"<u>Person</u>" means any individual, corporation (including any non-profit corporation), partnership, limited liability company, joint venture, estate, trust, association, organization, labor union or any other entity or Governmental Authority.

"<u>Personal Information</u>" means any information in the possession or control of the Seller Group (solely as related to the Business) about an identifiable individual other than the name, title or business address, business email address or telephone number of any employee of the Seller Group.

"Petition Date" has the meaning set forth in the Recitals.

"PRF" means the Progeria Research Foundation, Inc.

"PRF Novation Agreement" means the Novation and Assignment Agreement substantially in the form attached hereto as <u>Exhibit D</u> pursuant to which that certain Amended and Restated Collaboration and Supply Agreement, dated as of February 29, 2024, by and between Seller and PRF will be assigned and novated to Purchaser simultaneously with Closing, duly executed by each of the Seller and PRF.

"<u>Pre-Closing Tax Period</u>" means any taxable period ending on or prior to the Closing Date and the portion of any Straddle Period through the Closing Date.

"Progeria Field" has the meaning set forth in the Sublicense Agreement.

"Public Health Measures" means any closures, "shelter-in-place," "stay at home," workforce reduction, social distancing, shut down, closure, curfew or other restrictions or any other Applicable Law, Orders, directives, guidelines or recommendations issued by any Governmental Authority, the Centers for Disease Control and Prevention, the World Health Organization, or any industry group in connection with COVID-19 or any other epidemic, pandemic, or outbreak of disease, or in connection with or in response to any other public health conditions.

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintercelc0244225225171047595 DDess C MainEldoiduithien Pagle 64eo63.13 112

"Purchase Price" means the Base Price less the aggregate amount of Determined Cure Costs.

"Purchaser" has the meaning set forth in the preamble.

"Purchaser Group Members" has the meaning set forth in Section 10.17.

"Purchaser Releasing Party" has the meaning set forth in Section 10.16(b).

"Purchaser Schedules" has the meaning set forth in ARTICLE 4.

"Related Claims" means all claims or causes of action (whether in contract or tort, in law or in equity, or granted by statute or otherwise) that may be based upon, arise out of or relate to this Agreement, the Related Documents and any other document or instrument delivered pursuant to this Agreement or the Related Documents, or the negotiation, execution, termination, validity, interpretation, construction, enforcement, performance or nonperformance of this Agreement or the Related Documents or otherwise arising from the Transactions or the relationship between the parties (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with, or as an inducement to enter into, this Agreement or the Related Documents).

"Related Documents" means the Escrow Agreement, the Bill of Sale and Assignment and Assumption Agreement, Intellectual Property Assignment Agreement, Sublicense Agreement, Merck Side Letter, and PRF Novation Agreement; provided, however, that the Escrow Agreement the Bill of Sale and Assignment and Assumption Agreement, Intellectual Property Assignment Agreement, Sublicense Agreement, Merck Side Letter Agreement, and PRF Novation Agreement shall not be a Related Document solely for purposes of applying the provisions in <u>ARTICLE 10</u> to the extent, and only to the extent, that any such document expressly conflicts with ARTICLE 10.

"Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing into the environment of any Hazardous Substances.

"<u>Sale Motion</u>" means the motion of the Seller seeking entry of the Sale Order approving the terms herein, to be filed on or about March 31, 2024, in the Bankruptcy Cases.

"Sale Order" means an Order of the Bankruptcy Court issued pursuant to sections 105(a), 363 and 365 of the Bankruptcy Code in form and substance acceptable to Purchaser and the Seller, in each party's commercially reasonable discretion, approving this Agreement and all of the terms and conditions hereof and approving and authorizing the Seller to consummate the Transactions contemplated hereby Free and Clear and containing a finding that Purchaser has acted in "good faith" within the meaning of Section 363(m) of the Bankruptcy Code.

"Schedules" has the meaning set forth in ARTICLE 3.

"Seller" has the meaning set forth in the preamble.

"Seller Group" means the Seller and each of its Affiliates.

"Seller Group Members" has the meaning set forth in Section 10.17.

"Seller Group Taxes" means any (i) Liability of Seller Group for Taxes, (ii) any Liability for Asset Taxes attributable to any Pre-Closing Tax Period, and (iii) any Liability of Seller Group for the unpaid Taxes of any Person under Treasury Regulation §1.1502-6 (or any similar provision of state, local, or non-U.S. law), as a transferee or successor, by contract, or otherwise.

"Seller Permits" has the meaning set forth in Section 3.5.

"Seller Releasing Party" has the meaning set forth in Section 10.16(a)

"Solvent" when used with respect to any Person, means that, as of any date of determination, (a) the fair salable value (determined on a going concern basis) of its assets and property will, as of such date, exceed the amounts required to pay its debts as they become absolute and mature, as of such date, (b) such Person will have adequate capital to carry on its business and (c) such Person will be able to pay its debts as they become absolute and mature, in the ordinary course of business, taking into account the timing of and amounts of cash to be received by it and the timing of and amounts of cash to be payable on or in respect of its indebtedness.

"Sublicense Agreement" means the Sublicense Agreement, dated as of the Closing Date, by and among Purchaser and the Seller in substantially the form attached hereto as Exhibit E.

"<u>Straddle Period</u>" means any taxable year or other taxable period beginning on or before and ending after the Closing Date.

"<u>Tax</u>" means any tax of any kind whatsoever (including any income tax, franchise tax, branch profits tax, capital gains tax, value-added tax, unclaimed property, escheat, sales tax, use tax, property tax, transfer tax, payroll tax, social security tax or withholding tax), and any related fine, penalty, interest, or addition to tax with respect thereto, imposed, assessed or collected by or under the authority of any Governmental Authority.

"<u>Tax Return</u>" means any return (including any information return), report, statement, schedule, notice, form, or other document or information (whether in tangible, electronic or other form), including any amendments, schedules attachments, supplements, appendices and exhibits thereto, filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority in connection with the determination, assessment, collection, or payment, of any Tax.

"<u>Technology</u>" means algorithms, applied programming interfaces, apparatus, designs, drawings, data collections, diagrams, systems, procedures, processes, methods, methodologies, models, formulas, inventions (whether or not patentable), discoveries, improvements, know-how, methods, network configurations and architectures, processes, proprietary information, protocols, schematics, specifications, software, software code (in any form, including source code and executable or object code), subroutines, techniques, tools, user interfaces, technical engineering

Casse2248800409sgijj11 DDocc828821 Filitelc0044225225 Eintereclc0044225225171047595 DDessc MairElatoricutrii.en Pagle 66e065.13 112

and manufacturing information and materials including engineering plans and bills of materials, web sites, works of authorship and other forms of technology (whether or not embodied in any tangible form and including all tangible embodiments of the foregoing, such as instruction manuals, laboratory notebooks, prototypes, samples, studies and summaries).

"<u>Termination Fee</u>" means a fee equal to three percent (3.0%) of the initially proposed Base Price in the amount of \$26,000,000 *less* the aggregate reduction in the Base Price resulting from the \$214,285.71 *per diem* deduction for each calendar day that the Closing occurs after April 25, 2024.

"Territory" means the entire world.

"<u>Transactions</u>" means the transactions contemplated by this Agreement and the Related Documents.

"Transfer Taxes" has the meaning set forth in Section 2.10.

"Transferred Assets" has the meaning set forth in Section 2.1.

"Used in the Business" has the meaning set forth in Section 2.1.

"Zokinvy" means that certain commercially available, capsule formulation of lonafarnib as referenced by NDA # N213969.

1.2 Other Definitional and Interpretive Matters.

- (a) Unless otherwise expressly provided, for purposes of this Agreement and the Related Documents, the following rules of interpretation shall apply:
 - (i) <u>Calculation of Time Period</u>. All references to a day or days shall be deemed to refer to a calendar day or days, as applicable, unless otherwise specifically provided. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.
 - (ii) <u>Dollars</u>. Any reference to \$ shall mean U.S. dollars, which is the currency used for all purposes in this Agreement and the Related Documents. The specification of any dollar amount in the representations and warranties or otherwise in this Agreement, the Related Documents or the Schedules is not intended and shall not be deemed to be an admission or acknowledgement of the materiality of such amounts or items, nor shall the same be used in any dispute or controversy between the parties hereto to determine whether any obligation, item or matter (whether or not described herein or included in any schedule) is or is not material for purposes of this Agreement, the Related Documents or the Schedules.

Casse2248800409ssgjj.11 DDoor:33621 Filitelc0044225225 Hintereclc0044225225171047595 DDess c MainEldoidaith1entPagle6geo66.131112

- (iii) Exhibits/Schedules. The Exhibits and Schedules to this Agreement are an integral part of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Schedule shall be deemed to have been disclosed on each other Schedule. Disclosure of any item on any Schedule shall not constitute an admission or indication that any such item is required to be disclosed, or that such item or matter is material or has resulted in or will result in a Material Adverse Effect or that the included items or actions are not in the ordinary course of business. No disclosure on a Schedule relating to a possible breach or violation of any Contract, Applicable Law or Order shall be construed as an admission or indication that a breach or violation exists or has actually occurred. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.
- (iv) <u>Gender and Number</u>. Any reference to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.
- (v) <u>Headings</u>. The provision of a table of contents, the division of this Agreement or Related Documents into articles, sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement or Related Document, as applicable. Unless otherwise specified, all references in this Agreement to any "Section" or other subdivision are to the corresponding section or subdivision of this Agreement, and all references in a Related Document to any "Section" or other subdivision are to the corresponding section or subdivision of such Related Document.
- (vi) <u>Herein</u>. The words such as "herein," "hereinafter," "hereof' and "hereunder" that are used in this Agreement refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires. Uses of such words in the Related Documents shall refer to such Related Document as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
- (vii) Or. The word "or" shall be construed in the inclusive sense of "and/or" unless otherwise specified.
- (viii) <u>Including</u>. The word "including" or any variation thereof means (unless the context of its usage otherwise requires) "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.
- (ix) <u>Successors</u>. A reference to any party to this Agreement, any Related Document or any other agreement or document shall include such party's successors and permitted assigns.

Casse2248800409ssjýj11 DDoc38821 Filitelc0044225225 Eintereclc0044225225171047595 DDessc MairEldoilcitr1en1PaglPaglea651131112

- (x) <u>Legislation</u>. A reference to any legislation or to any provision of any legislation shall include any amendment thereto, and any modification or reenactment thereof, any legislative provision substituted therefor and all regulations and statutory instruments issued thereunder or pursuant thereto.
- (xi) Reflected On or Set Forth In. An item arising with respect to a specific representation or warranty shall be deemed to be "reflected on" or "set forth in" a balance sheet or financial statement, to the extent any such phrase appears in such representation or warranty, if (a) there is a reserve, accrual or other similar item underlying a number on such balance sheet or financial statement that relates to the subject matter of such representation, (b) such item is otherwise specifically set forth on the balance sheet or financial statement or (c) such item is set forth in the notes to the balance sheet or financial statement.
- (xii) <u>Made Available</u>. Any reference in this Agreement to "made available" means a document or other item of information that was provided or made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions.
- (b) All representations and warranties set forth in this Agreement or the Related Documents are contractual in nature only and subject to the sole and exclusive remedies set forth herein. No Person is asserting the truth of any representation and warranty set forth in this Agreement or the Related Documents; rather, the parties have agreed that should any representations and warranties of any party prove untrue, the other parties shall have the specific rights and remedies herein specified as the exclusive remedy therefor, but that no other rights, remedies or causes of action (whether in law or in equity or whether in contract or in tort or otherwise) are permitted to any party hereto as a result of the untruth of any such representation and warranty. The phrase "to Seller's Knowledge" and phrases of similar import or effect are used herein to qualify and limit the scope of any representation or warranty in which they appear and are not affirmations of any Person's "superior knowledge" that the representation or warranty in which they are used is true.
- (c) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and the Related Documents and, in the event an ambiguity or question of intent or interpretation arises, this Agreement and the Related Documents shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement and the Related Documents. The parties hereto agree that changes from earlier drafts to the final version of this Agreement do not necessarily imply that the party agreeing to such change is agreeing to a change in meaning (as the party agreeing to such change may believe the change is stylistic and non-substantive); consequently, no presumption should exist by virtue of a change from a prior draft.

ARTICLE 2. THE PURCHASE AND SALE; CLOSING

- 2.1 <u>Purchase and Sale</u>. Upon the terms and subject to the conditions set forth in this Agreement, the Sublicense Agreement, and the Sale Order, at the Closing, in exchange for an aggregate payment from Purchaser to the Seller equal to the Purchase Price, Purchaser shall purchase, assume and accept from the Seller, and the Seller shall sell, transfer, assign, convey and deliver (or shall cause the sale, transfer, assignment, conveyance and delivery) to Purchaser, Free and Clear (except for Permitted Liens), all of the rights, title and interests in, to and under the following assets and interests used in the Business ("<u>Used in the Business</u>") as the same shall exist on the Closing Date (collectively, the "<u>Transferred Assets</u>"):
- (a) the Transferred Inventory (as such term is defined in the Sublicense Agreement), the sale, transfer, assignment, conveyance and delivery of which are effected through the Sublicense Agreement (or the other transfer instruments contemplated therein) and subject to the terms and conditions thereof;
- (b) the Zokinvy Trademarks and Domain Names (as such terms are defined in the Sublicense Agreement), the sale, transfer, assignment, conveyance and delivery of which are effected through the Sublicense Agreement (or the other transfer instruments contemplated therein) and subject to the terms and conditions thereof;
- (c) the Transferred Regulatory Information (as such term is defined in the Sublicense Agreement), the sale, transfer, assignment, conveyance and delivery of which are effected through the Sublicense Agreement (or the other transfer instruments contemplated therein) and subject to the terms and conditions thereof;
- (d) (i) all Contracts that are listed on <u>Schedule 3.6</u> to the Sublicense Agreement (the sale, transfer, assignment, conveyance and delivery of which are effected through the Sublicense Agreement (or the other transfer instruments contemplated therein) and subject to the terms and conditions thereof), excluding Contracts that expire or are terminated prior to the Closing, and (ii) all Designated Contracts that Purchaser elects to assume pursuant to Section 5.3(b) ((i) and (ii), collectively, the "Assigned Contracts"); and
- (e) the Business Books and Records (as such term is defined in the Sublicense Agreement); <u>provided</u>, <u>however</u>, that the Seller Group shall be entitled to retain copies of any such materials as provided in the Sublicense Agreement;
- (f) all rights to receive mail and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group relating solely to Zokinvy in the Progeria Field (including any such mail and other correspondence and communications (including electronic mail) from the FDA or any other Governmental Authority, customers, advertisers, suppliers, distributors, agents and others and payments with respect to Zokinvy in the Progeria Field);
- (g) all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group with respect to the Business, the Transferred Assets and the Assumed Liabilities (including all guaranties,

Casse2248800409ssjýj11 DDocc83821 Filitelc0244225225 Eintereclc0244225225171047595 DDessc MairEldothichtrien Pagle a Opeo69103 112

warranties, indemnities and similar rights in favor of the Seller Group or any their Affiliates to the extent solely related to the Transferred Assets or the Assumed Liabilities), in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date, except for such rights, claims and causes of related to the Excluded Assets or Excluded Liabilities;

- (h) any other of Seller's assets and/or rights contemplated expressly to be transferred to Purchaser pursuant to the terms and conditions of the Sublicense Agreement; and
- (i) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, choses in action, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent used in or held for use for the Transferred Assets listed in clauses (a) through (h) above or the Assumed Liabilities.
- 2.2 <u>Excluded Assets</u>. Notwithstanding the provisions of <u>Section 2.1</u> or anything to the contrary herein, any and all assets, rights and properties of the Seller Group that are not specifically identified in <u>Section 2.1</u> as Transferred Assets, including the following (collectively, the "<u>Excluded Assets</u>"), shall be retained by the Seller Group, and Purchaser and its designees shall acquire no right, title or interest in the Excluded Assets in connection with the Transaction:
- (a) all (i) cash and cash equivalents, wherever located, including bank balances and bank accounts or safe deposit boxes, monies in the possession of any banks, savings and loans or trust companies and similar cash items, (ii) escrow monies and deposits in the possession of landlords and utility companies, and (iii) investment securities and other shortand medium-term investments;
- (b) all records, documents or other information exclusively relating to current or former employees of the Seller Group that are not hired by Purchaser, and any materials to the extent containing information about any employee, disclosure of which would violate Applicable Law or such employee's reasonable expectation of privacy;
- (c) any interest of the Seller Group under this Agreement or the Related Documents, including the right to receive the Purchase Price and to enforce the Seller's rights and remedies thereunder;
- (d) all Excluded Contracts (including all prepaid assets relating to the Excluded Contracts), other than the Assigned Contracts, to which any member of the Seller Group or any of their respective Affiliates is a party;
- (e) any (i) Attorney-Client Information arising from communications prior to the Closing Date between a member of the Seller Group (including any one or more officers, directors or stockholders of such Seller Group member), on the one hand, and its counsel, on the other hand, and (ii) claims under any director and officer, errors and omissions, fiduciary and commercial crime insurance policies; and
- (f) any rights of the Seller Group to Tax refunds (or credits for overpayment of Taxes in lieu of a refund) attributable to any Pre-Closing Tax Period;

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle a (1,60701.103112

- (g) all Permits (including applications therefor and any trade or import/export Permits) that (i) are not materially related to the Business or (ii) are not transferable to Purchaser under Applicable Law;
 - (h) the Excluded Books and Records;
- (i) any assets not otherwise designated as Transferred Assets or from time to time designated by the parties hereto as Excluded Assets;
- (j) all accounts receivable, intercompany obligations and other amounts receivable by the Seller Group;
 - (k) the Avoidance Actions;
- (l) all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group (including all guaranties, warranties, indemnities and similar rights in favor of the Sellers Group or any of their Affiliates) to the extent arising under the Bankruptcy Code or relating to any of the Excluded Assets or Excluded Liabilities, in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date; and
- (m) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent exclusively related to or exclusively used in or held for use for the Excluded Assets listed in clauses (a) through (l) above.

Notwithstanding anything to the contrary contained in this Agreement or any of the other Related Documents, Purchaser acknowledges and agrees that all of the following are also Excluded Assets, and all right, title and interest in and to all Excluded Assets shall be retained by the Seller Group and shall remain the property of the Seller Group (and shall expressly be excluded from the sale, transfer, assignment and conveyance to Purchaser hereunder), and neither Purchaser nor any of its Affiliates shall have any interest therein: (x) all records and reports prepared or received by the Seller Group or any of their Affiliates in connection with the sale of the Business and the Transactions, including all analyses relating to the Business or Purchaser so prepared or received; and (y) all confidentiality agreements with prospective purchasers of the Business or any portion thereof and all bids and expressions of interest received from third parties with respect thereto.

- 2.3 <u>Assumption of Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement, Purchaser shall, effective as of the Closing, assume and agree to pay, discharge and perform in accordance with their terms the following Liabilities of the Seller Group arising from or related to the Business or the Transferred Assets as the same shall exist on the Closing Date arising only after the Closing Date (collectively, the "<u>Assumed Liabilities</u>"), including:
- (a) all Liabilities relating to the Transferred Assets solely to the extent such Liabilities relate to and arise in periods following the Closing;

Casse2248800499sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle algeo 7 1 103 1 1 2

- (b) subject to <u>Section 2.4</u>, all Liabilities arising under the Assigned Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Determined Cure Costs; and
 - (c) all Taxes for which Purchaser is liable pursuant to this Agreement.
- 2.4 <u>Excluded Liabilities</u>. Notwithstanding <u>Section 2.3</u>, Purchaser is assuming only the Assumed Liabilities of the Seller Group and will not assume or be liable for any Excluded Liabilities (including Seller Group Taxes), and the Seller Group shall retain and shall be responsible for, all Liabilities that are not Assumed Liabilities, including all Liabilities related to Excluded Assets or any other Liabilities of the Business (all such Liabilities not being assumed herein referred to as the "<u>Excluded Liabilities</u>").
- 2.5 <u>Excluded Contracts</u>. Pursuant to <u>Section 5.3(b)</u>, Purchaser shall be entitled, in its sole discretion, by written notice to the Seller up to three Business Days prior to the Closing Date, to elect not to purchase or assume one or more Assigned Contract, in which case, notwithstanding anything in this Agreement or any Related Document to the contrary, such Assigned Contract shall be considered an excluded contract ("<u>Excluded Contract</u>") (and shall constitute an Excluded Asset and not be included in the Transferred Assets) for all purposes of this Agreement and Purchaser shall not have any obligation to satisfy or pay any Cure Costs or other Liabilities with respect to such Excluded Contract. Each assignable Assigned Contract that Purchaser does not elect to remove from the list of Assigned Contracts pursuant to <u>Section 5.3(b)</u> shall be an Assigned Contract.

2.6 Nontransferable Assets and Liabilities.

- (a) Notwithstanding any other provision of this Agreement to the contrary, this Agreement shall not constitute an agreement to assign or transfer any Transferred Asset or any claim, right or benefit arising thereunder or resulting therefrom if an attempted assignment or transfer thereof, without the Consent of a third party (including any Governmental Authority) (after giving effect to the Sale Order or any other applicable order of the Bankruptcy Court that effects such transfer without any required Consents), would constitute a breach or other contravention thereof or a violation of Applicable Law (each, a "Non-Transferred Asset").
- (b) If, on the Closing Date, any third-party Consent is not obtained for a Non-Transferred Asset, or if an attempted transfer or assignment thereof would be ineffective or a violation of Applicable Law, then, until any requisite consent is obtained therefor and the same is transferred and assigned to Purchaser or its designee, each such Non-Transferred Asset shall be held by the Seller as agent for the Purchaser, and the Seller shall, to the extent permitted by Applicable Law, provide to Purchaser the benefits and Purchaser shall assume the obligations and bear the economic burdens associated with such Non-Transferred Asset. The Seller and Purchaser shall use commercially reasonable efforts to enter into agreements (including subcontracting, sublicensing or subleasing, if permitted) by which (i) the Seller shall, at Purchaser's sole expense, without interruption of the Business, provide Purchaser with the economic and operational equivalent of obtaining the requisite third-party Consent and assigning the applicable Non-Transferred Asset to Purchaser (including, with the prior written consent of Purchaser, enforcing for the benefit of Purchaser, and at Purchaser's sole expense, all claims or

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaglPaglPagle072103112

rights arising thereunder) and (ii) Purchaser shall perform, at its sole expense, the obligations and assume the economic burdens of the Seller or its Affiliates to be performed after the Closing with respect to such Non-Transferred Asset. Purchaser shall promptly, upon receipt of a written request therefor from the Seller, reimburse the Seller for all monies paid by the Seller on Purchaser's behalf in connection with any Assumed Liability not assigned or transferred to Purchaser pursuant to this Section 2.6.

2.7 <u>Closing</u>. The closing of the Transactions (the "<u>Closing</u>") will take place remotely by electronic exchange of documents on the date (the "<u>Closing Date</u>") that is the second (2nd) Business Day after the date on which all of the conditions set forth in <u>ARTICLE 8</u> (excluding conditions that, by their terms, are to be satisfied at the Closing, but subject to the satisfaction or waiver of all such conditions at the Closing), have been satisfied or waived by the party hereto entitled the benefit of the same, unless another time or date is agreed to in writing by the parties hereto. Except as otherwise set forth herein, all proceedings to be taken and all documents to be executed and delivered by all parties hereto at the Closing will be deemed to have been taken and executed simultaneously and no proceedings will be deemed to have been taken nor documents executed or delivered until all have been taken, executed, and delivered.

2.8 <u>Closing Deliveries of the Parties</u>. At or prior to the Closing:

- (a) Purchaser and the Seller shall execute and deliver the Bill of Sale and Assignment and Assumption Agreement;
- (b) Purchaser and the Seller shall execute and deliver the Intellectual Property Assignment Agreement, in a form reasonably acceptable to Purchaser and the Seller;
- (c) Purchaser and the Seller shall execute and deliver the Sublicense Agreement;
- (d) Escrow Agent, Purchaser and the Seller shall execute and deliver the Escrow Agreement;
- (e) To the extent not covered in the Sublicense Agreement, on the Closing Date, each Party shall transmit the Purchaser's FDA Transfer Letters to the FDA and shall take any other actions reasonably necessary to effect the transfer of Zokinvy in the Progeria Field from the Seller to Purchaser;
- (f) Purchaser shall deliver, or cause to be delivered, to the Seller or the applicable Person each of the following:
 - (i) a certificate, dated as of the Closing Date, executed by or on behalf of Purchaser as to the satisfaction of the conditions set forth in <u>Section 8.3(a)</u> and Section 8.3(b); and
 - (ii) payment of the closing payments set forth in Section 2.9.
- (g) the Seller shall deliver, or cause to be delivered, to Purchaser or the applicable Person each of the following:

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MairEldoiloutrilentPagle73103112

- (i) the PRF Novation Agreement duly executed by PRF and Seller;
- (ii) the Merck Side Letter duly executed by Merck and Seller;
- (iii) a certificate, dated as of the Closing Date, executed by or on behalf of the Seller as to the satisfaction of the conditions set forth in Section 8.2(a) and Section 8.2(b);
- (iv) an IRS Form W-9 with respect to the Seller, duly completed and executed, dated as of the Closing Date; and
- (v) the deliverables that are required to be delivered to Purchaser on the Effective Date (as defined in the Sublicense Agreement) pursuant to the Sublicense Agreement.

2.9 Purchase Price; Assumed Liabilities; Deposits.

- (a) At the Closing, upon the terms and subject to the conditions set forth herein, in full consideration for the sale, transfer, conveyance, assignment and delivery of the Transferred Assets to Purchaser and assumption of the Assumed Liabilities by Purchaser, Purchaser shall (i) pay to the Seller an aggregate amount equal to the Purchase Price *minus* the Deposit Escrow Amount, which shall be released to the Seller by the Escrow Agent pursuant to Section 2.9(c), by irrevocable wire transfer of immediately available funds in accordance with payment instructions delivered by the Seller to Purchaser prior to the Closing; and (ii) assume the Assumed Liabilities.
- (b) At the Closing, on the terms and subject to the conditions set forth in this Agreement, Purchaser will assume and become responsible for the Assumed Liabilities. Purchaser agrees to pay, perform, honor, and discharge, or cause to be paid, performed, honored and discharged, all Assumed Liabilities in a timely manner in accordance with the terms hereof, including paying or causing to be paid, at or prior to the Closing, all Determined Cure Costs.
 - (c) The Deposit Escrow Amount shall be distributed as follows:
- (i) if the Closing shall occur, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available funds, to an account designated by the Seller to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to the Seller at Closing and credited against the amount required to be paid by Purchaser to the Seller at Closing in accordance with Section 2.9(a);
- (ii) if this Agreement is terminated by the Seller pursuant to Section 9.1(g), (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available funds, to an account designated by the Seller to the Escrow Agent and (B) the Deposit Escrow Amount, which shall constitute liquidated damages (and not a

Casse2248800409ssgjj111 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 25co74131112

penalty), shall be delivered to the Seller within two (2) Business Days following delivery of such joint written instruction; or

(iii) if this Agreement is validly terminated for any reason in accordance with the terms of this Agreement other than by the Seller pursuant to Section 9.1(g) or Purchaser forfeits the Deposit Escrow Amount to the Seller pursuant to Section 8.5, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to Purchaser, by irrevocable wire transfer of immediately available funds, to an account designated by Purchaser to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to Purchaser within two (2) Business Days following delivery of such joint written instruction.

Any issue regarding the entitlement to the Deposit Escrow Amount shall be determined by the Bankruptcy Court, and Purchaser consents to the jurisdiction of the Bankruptcy Court for any issue related to this Agreement.

2.10 <u>Transfer Taxes</u>. Purchaser shall be solely responsible for, and shall indemnify, defend, and hold harmless the Seller Group for, any transfer, documentary, sales, use, excise, stock transfer, value-added, stamp, recording, registration and other similar taxes, levies and fees (including any penalties, fines and interest), together with any conveyance fees, recording charges and other similar fees and charges, incurred in connection with this Agreement and the Transactions (collectively, "<u>Transfer Taxes</u>"). Purchaser and the Seller shall cooperate in good faith to minimize, to the extent permissible under Applicable Law, the amount of any Transfer Taxes due with respect to the Transactions.

2.11 Allocation of Purchase Price.

- (a) The Purchase Price (including all other amounts treated as consideration for U.S. federal income tax purposes) and Assumed Liabilities shall be allocated as set forth on Schedule 2.11 (the "Preliminary Allocation Schedule"). Within 90 days following the final determination of the Purchase Price, Purchaser shall deliver to the Seller a schedule allocating the Purchase Price (and all other amounts treated as consideration for U.S. federal income tax purposes) among the Transferred Assets (the "Allocation Schedule"). The Allocation Schedule shall be reasonable and shall be prepared in accordance with the Preliminary Allocation Schedule, and Purchaser and the Seller shall negotiate in good faith to resolve disputed items, if any, in the Allocation Schedule as promptly as practicable. If Purchaser and the Seller are unable to reach agreement with respect to the Allocation Schedule within 30 days after the delivery of the Allocation Schedule by Purchaser to the Seller, the parties shall be entitled to use their own Purchase Price allocations for Tax reporting purposes.
- (b) To the extent Purchaser and the Seller agree on the Allocation Schedule pursuant to Section 2.11(a), Purchaser and the Seller shall (i) timely file all Tax Returns required to be filed in connection with the Allocation Schedule, and (ii) prepare and file all Tax Returns and determine all Taxes in a manner consistent with the Allocation Schedule, except as may be required by Applicable Law and except as may be necessary to reflect adjustments to the Allocation Schedule resulting from post-Closing payments or events. Purchaser, on the one

hand, and the Seller, on the other hand, shall notify the other if it receives notice that any Governmental Authority proposes any allocation different from Allocation Schedule.

- 2.12 **Escrow Accounts**. At the Closing, the Deposit Escrow Amount shall be used to satisfy a portion of the payment obligations of Purchaser pursuant to Section 2.9(c), otherwise the Deposit Escrow Amount shall be released to Purchaser or the Seller pursuant to Section 2.9(c). Upon the final release of all of the Deposit Escrow Amount pursuant to the terms of this Agreement and the Escrow Agreement, the Escrow Agreement shall automatically terminate. Any fees owed to the Escrow Agent and obligations under the Escrow Agreement shall be borne by Purchaser. The Deposit Escrow Amount shall be held in trust for the benefit of the Seller and shall not be subject to any encumbrance, attachment, trustee process or any other judicial process of any creditor of any party hereto, and shall be held and disbursed solely for the purposes of and in accordance with the terms of this Agreement and the Escrow Agreement.
- 2.13 <u>Tax Withholding</u>. Notwithstanding anything in this Agreement to the contrary, Purchaser shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any Person such amounts as it is required to deduct and withhold from such Person with respect to the making of such payment under the Code and the rules and regulations promulgated thereunder, or any provision of any Law relating to Taxes; <u>provided</u>, <u>however</u>, that the Purchaser shall (i) provide commercially reasonable notice to the Person prior to such deduction and withholding and (ii) afford the Person a reasonable opportunity to provide any additional information, forms or certifications to establish an exemption from, or obtain a reduced rate of, withholding. To the extent that amounts are so withheld and properly remitted by Purchaser, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to such Person in respect of which such deduction and withholding was made by Purchaser.

ARTICLE 3. REPRESENTATIONS AND WARRANTIES OF THE SELLER

Except as disclosed in a document herewith delivered by the Seller to Purchaser (the "<u>Schedules</u>"), the Seller hereby makes the representations and warranties contained in this ARTICLE 3 to Purchaser.

Group is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has, subject to the necessary authority of the Bankruptcy Court, the requisite corporate power and authority to operate the Business and necessary to own, lease or operate the properties and assets owned, leased or operated by it to carry on the Business as now being conducted, except where the failure to be so duly organized, validly existing and in good standing, or to have such power and authority, would not, individually or in the aggregate, have a Material Adverse Effect. Each member of the Seller Group is duly qualified to do business as a foreign company in each jurisdiction in which the nature of the Business as currently conducted by it or the property owned or leased by it makes such qualification necessary, except where the failure to be so qualified would not, individually or in the aggregate, have a Material Adverse Effect.

- Authority and Enforceability. Subject to Bankruptcy Court approval, the Seller has all requisite power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement and the each of the Related Documents to which the Seller is (or at Closing, will be) a party thereto, and the consummation by the Seller of the Transactions, has been duly authorized and approved by all necessary limited liability company action on the part of the Seller and are subject to the approval of the Bankruptcy Court. This Agreement has been, and each Related Document will be, at or prior to the Closing, duly executed and delivered by the Seller and, assuming the due execution and delivery by the other parties hereto or thereto, and subject to the approval of the Bankruptcy Court, constitutes a valid and binding obligation of the Seller, enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.
- 3.3 No Conflict; Required Filings and Consents. Except (a) such filings as may be required in connection with the Transfer Taxes described in Section 2.10 and (b) as otherwise set forth on Schedule 3.3, the execution and delivery of this Agreement by the Seller does not and the execution and delivery of the Related Documents by the Seller will not, and the consummation of the Transactions hereby and thereby will not (i) violate the provisions of the Organizational Documents of any member of the Seller Group, (ii) subject to the entry of the Sale Order, violate any Applicable Law or Order to which any member of the Seller Group is subject or by which its properties or assets are bound, (iii) require any member of the Seller Group to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date (except as required by the Bankruptcy Code or the Sale Order), (iv) subject to the entry of the Sale Order, result in a breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any Assigned Contract or (v) subject to the entry of the Sale Order, result in the imposition or creation of any Lien upon or with respect to any of the assets or properties of the Seller Group; excluding from the foregoing clauses (ii) through (v) any Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, have a Material Adverse Effect.
- Compliance With Laws. To the Seller's Knowledge, (i) the Seller Group is conducting the Business in compliance in all material respects with all material Applicable Laws applicable to the Business and (ii) no member of the Seller Group has received any written notice since the Petition Date of any material violations of any material Applicable Law applicable to their conduct of the Business. As of the Agreement Date, the Seller has and, to the Seller's Knowledge, has obtained all permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals of the FDA or any other Governmental Authority, currently used in, necessary for and material to the operation of sale of Zokinvy in the Progeria Field as presently conducted, all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals are included in the Transferred Assets and Seller has made available to Purchaser true and complete copies of all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. As of the Agreement Date, neither Seller nor, to the Seller's Knowledge, any other Person has received any communication

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1PaglP7866077.103112

from any Governmental Authority that threatens to withdraw or suspend any such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. Seller has filed with the applicable Regulatory Authorities all required filings, declarations, listings, registrations, reports or submissions, including adverse event reports, necessary for and material to the operation of sale of Zokinvy in the Progeria Field as presently conducted. All relevant filings, declarations, listings, registrations, reports or submissions were in material compliance with Applicable Law when filed, and no deficiencies have been asserted by any Governmental Authority with respect to any such filings, declarations, listing, registrations, reports or submissions. As of the Agreement Date, the Seller has not received or been subject to: (1) any FDA Form 483s directly relating to Zokinvy in the Progeria Field; (2) any FDA notices of adverse findings relating to Zokinvy in the Progeria Field; or (3) any warning letters or other correspondence from the FDA or any other Governmental Authority in which the FDA or such other Governmental Authority asserted that the actions of Seller, with respect to Zokinvy in the Progeria Field, were not in compliance with Applicable Laws. There has not been any occurrence of any product recall, market withdrawal or replacement, or post-sale warning conducted by or on behalf of the Seller concerning Zokinvy in the Progeria Field or, to the Seller's Knowledge, any product recall, market withdrawal or replacement conducted by or on behalf of any entity as a result of any alleged defect in Zokinvy in the Progeria Field.

- 3.5 <u>Permits</u>. To the Seller's Knowledge, (i) the Seller Group possess all material Permits required for the operation of the Business as currently conducted (the "<u>Seller Permits</u>") and (ii) no member of the Seller Group has received as of the Agreement Date any written notice of any cancellation, suspension, revocation, invalidation or non-renewal of any Permit since the Petition Date.
- 3.6 <u>Litigation</u>. As of the Agreement Date, there is no Action pending or, to the Seller's Knowledge, formally threatened in writing, against any member of the Seller Group before any Governmental Authority that would have a Material Adverse Effect or affect the Transferred Assets in any material respect after the entry of the Sale Order, if determined adversely and after taking into effect applicable insurance coverage.

3.7 Real Property; Personal Property.

- (a) The Seller Group does not own any real property.
- (b) <u>Schedule 3.6(b)</u> sets forth each parcel of real property leased by the Seller Group.
- (c) <u>Schedule 3.7(b)</u> sets forth a list of all leases of tangible assets and other personal property of the Seller Group as of the Agreement Date involving annual payments in excess of \$50,000. Each member of the Seller Group has good and valid title to, or in the case of leased tangible assets and other personal property, a valid leasehold interest in (or other right to use), all of the material tangible assets and other personal property that are necessary for such member of the Seller Group to conduct the Business, in each case, free and clear of all Liens to the maximum extent permitted by Section 363(f) of the Bankruptcy Code (other than Permitted Liens). All such material tangible assets and other personal property are in good condition and repair, normal wear and tear excepted.

- Assigned Contracts. With respect to the Assigned Contracts, (i) except as a result of, or arising in connection with, the filing of the Bankruptcy Cases, no member of the Seller Group has received any written notice of any default or event that (with due notice or lapse of time or both) would constitute a default by the applicable member of the Seller Group under any Assigned Contract, other than defaults that have been cured or waived in writing or would not reasonably be expected to have a Material Adverse Effect, (ii) to the Seller's Knowledge, each Assigned Contract is a legal, valid and binding obligation of the applicable member of the Seller Group and is in full force and effect (except to the extent subject to, and limited by, the Enforceability Exceptions), (iii) to the Seller's Knowledge, no other party to any Assigned Contract is (with or without the lapse of time or the giving of notice, or both) in material breach of or in material default under any Assigned Contract and (iv) to the Seller's Knowledge, no member of the Seller Group has provided or received any notice of any intention to terminate any Assigned Contract. The Seller has made available to Purchaser true, correct and complete copies of each of the Assigned Contracts listed on Schedule 3.8, together with all amendments thereto.
- 3.9 Financial Statements. Schedule 3.9 sets forth the Seller's (i) balance sheet and the related financial statements of revenue, expenses, retained earnings, and cash flow for the fiscal year ending on December 31, 2022, (ii) balance sheet and the related financial statements of revenue, expenses, retained earnings, and cash flow for the quarter ending on September 30, 2022 and (iii) the Seller's internally prepared statements of revenue, expenses, retained earnings, and cash flow for the months ending October 31, 2023 (collectively, the "Seller Financial Statements"). The Seller Financial Statements have been prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements or, in the case of unaudited financial statements, except as permitted by the SEC on Form 10-Q under the Exchange Act, and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments), have been prepared on a consistent basis throughout the periods covered thereby and presents fairly in all respects the financial condition of the Seller as of such dates and the results of operations of Seller for such periods, and are consistent with the books and records of Seller (which books and records are correct and complete in all material respects).
- 3.10 <u>Absence of Material Developments</u>. Except as disclosed on <u>Schedule 3.10</u>, since the Petition Date, there has occurred no fact, event, condition, change or circumstance which has had or would reasonably be expected to have a Material Adverse Effect.
- 3.11 <u>Customers and Suppliers.</u> Except as disclosed on <u>Schedule 3.11(a)</u>, to the Knowledge of the Seller, since the Petition Date, no customer has or has threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions, purchasing materials, products or services from the Business. Except as disclosed on <u>Schedule 3.11(b)</u>, to the Knowledge of the Seller, no supplier has or has threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions. supplying materials, products or services to the Business.

3.12 **Intellectual Property.**

- (a) A true, correct and complete list of all Intellectual Property Registrations and, material unregistered trademarks, and all material software included in the Owned Intellectual Property Assets is set forth on <u>Schedule 3.12(b)</u>.
- (b) The Seller Group exclusively owns all Owned Intellectual Property Assets. Except as set forth on Schedule 3.12(b), no member of the Seller Group is a party to, or bound by, (i) any license, royalty agreement, or other agreement relating to the use of any material Business Intellectual Property (other than non-exclusive licenses grant to a member of the Seller Group for commercially available, unmodified, off-the-shelf software licensed for aggregate annual fees of less than \$50,000), and (ii) agreements pursuant to which a member of the Seller Group settled any action, litigation, suit or other judicial or administrative proceeding, claim, assertion, or threat with respect to Intellectual Property, including settlement agreements, coexistence agreements, and consent agreements.
- (c) Other than with respect to Excluded Contracts or Assigned Contracts that Purchaser does not ultimately assume, no current or former Affiliate, partner, director, stockholder, officer, member, manager, employee, consultant or contractor of the Seller Group will, after giving effect to the Transactions, own, license or retain any Business Intellectual Property.
- (d) All material Intellectual Property Registrations remain pending or in full force and effect and have not expired or been abandoned or cancelled. To Seller's Knowledge, no interference, opposition, reissue, reexamination, or other proceeding is or has been pending or threatened, in which the scope, validity, or enforceability of any material Owned Intellectual Property Assets is being, has been challenged.
- (e) To the Knowledge of the Seller, the conduct of the Business does not infringe, misappropriate or otherwise violate in any material respect any Person's Intellectual Property.
- (f) To the Knowledge of the Seller's, no Person is currently infringing, misappropriating or otherwise violating any material Owned Intellectual Property Assets.
- (g) The Seller Group has taken commercially reasonable steps to safeguard and maintain the confidentiality of all trade secrets that constitute Owned Intellectual Property Assets, including by using good faith efforts to require all Persons having access thereto to execute written non-disclosure agreements.
- (h) The Seller Group complies with all Applicable Laws, internal policies and contractual obligations relating to privacy, data protection and cybersecurity.
- 3.13 <u>Inventories.</u> Except as disclosed on <u>Schedule 3.13</u>, all inventories of each member of the Seller Group (whether or not reflected on the Seller Financial Statements) consist of a quality and quantity usable and, with respect to finished goods, saleable, in the ordinary

course of business. No member of the Seller Group is in possession of any goods or inventory not owned by a member of the Seller Group, and the inventories (other than goods in transit) of a member of the Seller Group are located on the premises of the Seller Group. The reserve for obsolescence with respect to inventories is adequate and calculated consistent with past practice. Inventories that were purchased after the date of the balance sheet included in the Seller Financial Statements were purchased in the ordinary course of business at a cost not exceeding market prices prevailing at the time of purchase for items of similar quality and quantity. The quantities of each item of inventory are not excessive, but are reasonable for the continued operation of each member of the Seller Group in the ordinary course of business.

- 3.14 <u>Taxes</u>. The Seller Group has timely filed all Tax Returns that it was required to file with respect to Transferred Assets. All such Tax Returns were correct and complete in all material respects. All Taxes owed by the Seller Group (whether or not shown or required to be shown on any Tax Return) with respect to Transferred Assets have been paid. There are no liens on any of the Transferred Assets that arose in connection with any failure (or alleged failure) to pay any Tax. There is no dispute, examination, judicial proceeding or claim concerning any Taxes of the Seller Group with respect to the Transferred Assets.
- 3.15 <u>Product Liability.</u> Except as disclosed on <u>Schedule 3.15</u>, within the three (3) year period prior to the Closing Date there has not been any, and as of the Closing Date there is no pending, material litigation commenced against any member of the Seller Group relating to the sale, distribution or use of any item sold or used in the Business (the "<u>Goods</u>"), including litigation with respect to product liability or recall claims.
- 3.16 **Product Warranties; Product Returns.** Except for warranties arising solely pursuant to Applicable Law or in the ordinary course of business, (a) no member of the Seller Group has made any material warranties, express or implied, written or oral, to any third party with respect to any of the Goods within the three (3) year period prior to the Closing Date, and (b) there is no, and within the three (3) year period prior to the Closing Date there has not been any, material litigation pending or, to the Seller's Knowledge, threatened with respect to any such warranty.
- 3.17 <u>Accounts Payable</u>. The Seller has fully paid all accounts payable and related intercompany obligations of the Seller Group associated with the Business, incurred up to the Petition Date with respect to the suppliers or vendors set forth on <u>Schedule 3.17</u>.
- 3.18 <u>Brokers and Finders</u>. Except for SSG Advisors, LLC, the Seller has not, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.
- 3.19 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in this <u>ARTICLE 3</u> and the Sublicense Agreement and the Related Agreements, the Seller does not, nor do any other Persons on behalf of the Seller, make any other express or implied representation or warranty with respect to itself, the Business, the Transferred Assets or the Assumed Liabilities, or with respect to any other information provided to Purchaser or its representatives, and the Seller disclaims any other representations or warranties, whether made by or on behalf of the Seller or any other Person. The Seller will not, and no other Persons

will, have or be subject to any Liability to Purchaser or any other Person resulting from the distribution to Purchaser, or Purchaser's use of, any such information, including any information, documents, projections, forecasts or other material made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever (electronic or otherwise) or otherwise in expectation of the Transactions.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as disclosed in a document herewith delivered by Purchaser to the Seller (the "<u>Purchaser Schedules</u>"), Purchaser hereby makes the representations and warranties contained in this <u>ARTICLE</u> 4 to the Seller.

- 4.1 Organization, Good Standing and Other Matters. Purchaser is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has all requisite corporate power or other entity power and authority to own its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed to conduct its business as currently conducted and is in good standing in each jurisdiction in which the location of the property owned, leased or operated by it or the nature of its business makes such qualification necessary, except where the failure to be so qualified or licensed would not, individually or in the aggregate, materially impair or delay Purchaser's ability to consummate the Transactions.
- 4.2 Authority and Enforceability. Purchaser has all requisite corporate power or other entity power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party, and the consummation of the Transactions, have been duly authorized and approved by its board of directors (or equivalent governing body) and no other action on the part of Purchaser or its members is necessary to authorize the execution, delivery and performance of this Agreement and the Related Documents by Purchaser and the consummation of the Transactions. This Agreement has been, and each Related Document will be at or prior to Closing, duly executed and delivered by Purchaser and, assuming the due execution and delivery by the other parties hereto or thereto, constitutes a valid and binding obligation of Purchaser enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.
- 4.3 <u>No Conflict: Required Filings and Consents</u>. Except (a) such filings as may be required in connection with the Transfer Taxes described in <u>Section 2.10</u> and (b) as set forth on <u>Schedule 4.3</u>, the execution and delivery of this Agreement and of the Related Documents and the consummation of the Transactions by Purchaser will not (i) violate the provisions of its Organizational Documents, (ii) violate any Applicable Law or Order to which it is subject or by which any of its properties or assets are bound, (iii) require it to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date,

- (iv) result in a material breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any material Contract to which it is a party or (v) result in the imposition or creation of any Lien upon or with respect to any of its assets or properties; excluding from the foregoing clauses (ii) through (v) Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, (A) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (B) otherwise prevent, hinder or delay the consummation of the Transactions.
- 4.4 <u>Financing</u>. Purchaser has, and at the Closing will have, (a) sufficient internal funds (without giving effect to any unfunded financing regardless of whether any such financing is committed) available to pay the Purchase Price in accordance with the terms hereof and any other payments required hereunder and any expenses incurred or required to be paid by Purchaser in connection with the Transactions, and (b) the resources and capabilities (financial or otherwise) to perform its obligations hereunder and under the Related Documents. Purchaser has not incurred any obligation, commitment, restriction, or Liability of any kind, which would impair or adversely affect such resources and capabilities.
- 4.5 <u>Solvency</u>. Purchaser is not entering into this Agreement with the intent to hinder, delay or defraud either present or future creditors. Immediately after giving effect to all of the Transactions, including the making of the payments contemplated by <u>Section 2.9</u>, and assuming satisfaction of the conditions to Purchaser's obligation to consummate the Transactions as set forth herein, the accuracy of the representations and warranties of Purchaser set forth herein and the performance by Purchaser of its obligations hereunder in all material respects, Purchaser will be Solvent.
- 4.6 <u>Litigation</u>. There is no Action pending or, to Purchaser's Knowledge, formally threatened against Purchaser or involving any of its properties or assets that would be reasonably be expected to (a) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (b) otherwise prevent, hinder, or delay the consummation of the Transactions.
- 4.7 <u>Brokers and Finders</u>. None of Purchaser or its Affiliates have, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.

4.8 <u>Investigation and Agreement by Purchaser; Non-Reliance of Purchaser; No Other Representations and Warranties.</u>

(a) Purchaser acknowledges that it and its representatives have received access to such books and records, facilities, equipment, contracts, and other assets of the Business which it and its representatives have desired or requested to review. Purchaser acknowledges and agrees that it has made its own inquiry and investigation into, and, based thereon, have formed an independent judgment concerning the Seller Group, the Business, the Transferred Assets and the Assumed Liabilities.

Casse2248800409ssjýj11 DDocc33621 Filitelc0044225225 Eintereclc0044225225171047595 DDessc MairEldoilcitr1en1Pagle84e083.103112

- Except for the specific representations and warranties expressly (b) made by the Seller in ARTICLE 3 and the Sublicense Agreement and Related Agreements as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, and the representations and warranties made by the Seller in the Sublicense Agreement, Purchaser acknowledges and agrees that (i) the Seller is not making and have not made any representation or warranty, expressed or implied, at law or in equity, in respect of the Business, the Transferred Assets, the Assumed Liabilities, or any of its operations, prospects or condition (financial or otherwise), including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any Liabilities, the prospects of the Business, the effectiveness or the success of any operations, or the accuracy or completeness of any confidential information memoranda, documents, projections, material or other information (financial or otherwise) regarding the Business furnished to Purchaser or its representatives or made available to Purchaser and its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever, and (ii) no officer, director, manager, stockholder, agent, Affiliate, advisor, representative or employee of the Seller Group has any authority, express or implied, to make any representations, warranties or agreements not specifically set forth in ARTICLE 3 and subject to the limited remedies herein provided or in the Sublicense Agreement.
- (c) Other than the specific representations and warranties expressly set forth in ARTICLE 3 as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement or in the Sublicense Agreement, Purchaser specifically disclaims that it is relying upon or has relied upon any such other representations or warranties that may have been made by any Person, and acknowledges and agrees that the Seller and the Seller's Affiliates have specifically disclaimed and do hereby specifically disclaim, and shall not have or be subject to any Liability for reliance on any such other representation or warranty made by any Person. Purchaser specifically waives any obligation or duty by the Seller or the Seller's Affiliates to make any disclosures of fact not required to be disclosed pursuant to the specific representations and warranties expressly set forth in ARTICLE 3 or in the Sublicense Agreement to the specific representations and warranties set forth in ARTICLE 3 or in the Sublicense Agreement.
- (d) Purchaser is acquiring the Business, the Transferred Assets and the Assumed Liabilities subject only to the specific representations and warranties expressly set forth in <u>ARTICLE 3</u> as further limited by the specifically bargained-for exclusive remedies as set forth in <u>Section 9</u> of this Agreement and in the Sublicense Agreement.

4.9 <u>No Other Representations or Warranties</u>. Except for the representations and warranties contained in this <u>ARTICLE 4</u>, neither Purchaser nor any other Person on behalf of Purchaser makes any other express or implied representation or warranty with respect to Purchaser or with respect to any other information provided to the Seller or its representatives, and Purchaser disclaims any other representations or warranties, whether made by Purchaser or any of its Affiliates, officers, directors, employees, agents or representatives.

ARTICLE 5. BANKRUPTCY COURT MATTERS

Sankruptcy Court and the consideration by the Seller of higher or better competing bids in respect of all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) in accordance with the terms of the Bid Procedures Order (each, a "Competing Bid"). From the Agreement Date (and any prior time) and until the Closing, the Seller is permitted to, and to cause its representatives to, initiate contact with, solicit or encourage submission of any inquiries, proposals or offers by, any Person (in addition to Purchaser and its Affiliates and representatives) in connection with any sale or other disposition of the Transferred Assets. In addition, the Seller shall have the authority to respond to any inquiries or offers to purchase all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) and perform any and all other acts related thereto which are required under the Bankruptcy Code, the Bid Procedures Order or other Applicable Law, including supplying information relating to the Business and the assets of the Seller Group to prospective purchasers.

5.2 Bankruptcy Court Filings.

Subject to its right to pursue a Competing Bid in accordance with the Bid Procedures Order, the Seller shall diligently pursue the entry by the Bankruptcy Court of the Sale Order, which Sale Order shall provide for the transfer of the Transferred Assets and the Assumed Liabilities to Purchaser free from all successor or transferee Liability to the fullest extent permitted by Section 363 of the Bankruptcy Code. The Seller shall comply (or obtain an Order from the Bankruptcy Court waiving compliance) with all requirements under the applicable provisions of the Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the Local Bankruptcy Rules for the Bankruptcy Court in obtaining the entry of the Sale Order. The Seller further covenants and agrees that, after entry by the Bankruptcy Court of the Sale Order, and provided, that the Sale Order becomes a Final Order, the terms of any other proposed order submitted by the Seller to the Bankruptcy Court shall not conflict with, supersede, abrogate, nullify or restrict the terms of this Agreement, or in any way prevent or interfere with the consummation or performance of the Transactions. Purchaser agrees that it will promptly take such actions as are reasonably requested by the Seller to assist in obtaining entry of the Sale Order, including by furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Purchaser under this Agreement and demonstrating that Purchaser is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. In the event, if the entry of the Sale Order shall be appealed, the Seller and Purchaser shall use their respective commercially reasonable efforts to defend such appeal.

Casse2248800409ssgjj111 DDoor:33621 Filitelc0044225225 Hintereclc00442252251171047505 DDess c MainEldoidaith1entPagle866co85.131112

- (b) Within one (1) day after the Petition Date, Seller will file the Bid Procedures Motion seeking the Bankruptcy Court's immediate approval and entry of the Bid Procedures Order substantially in the form and substance reasonably agreed to by the Buyer and Seller, among other things, (A) establishing the Bid Procedures, (B) approving payment of the Termination Fee and the Expense Reimbursement, to the extent payable by the terms of this Agreement and the Bid Procedures Order, and (C) providing that the Termination Fee and the Expense Reimbursement shall constitute superpriority administrative expenses of the Seller with priority over any and all administrative expenses pursuant to section 503(b) of the Bankruptcy Code.
- (c) Seller shall use commercially reasonable efforts to provide Purchaser with a reasonable opportunity to review and comment upon all motions, applications, and supporting papers relating to the transactions contemplated by this Agreement prepared by Seller or any Affiliates (including forms of orders and notices to interested parties) prior to the filing thereof in the Bankruptcy Cases; <u>provided</u> that the foregoing shall not require the Seller to take any action that would, in Seller's reasonable business judgment, threaten to harm the overall value to be produced by the Seller's in-court sale process.
- (d) The form of Bid Procedures Order and form of Sale Order submitted by the Seller to the Bankruptcy Court for approval must be reasonably satisfactory in form and substance to the Purchaser; <u>provided</u> that an order approving the form of Bid Procedures Order is, and shall be deemed to be, acceptable to Purchaser.
- (e) Seller shall not seek any modification to the Bid Procedures, Bid Procedures Order, or Sale Order by the Bankruptcy Court that are materially adverse to the Purchaser without the prior written consent of Purchaser, which such consent shall not be unreasonably withheld.
- (f) Each of Purchaser and Seller will promptly take such actions as are reasonably requested by the other party to assist in obtaining entry of the Bid Procedures Order and, subject to the Bid Procedures Order, the Sale Order, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for purposes, among others, of providing necessary assurances of performance by Seller of its obligations under this Agreement and demonstrating that Purchaser is a good faith buyer under section 363(m) of the Bankruptcy Code.
- (g) Seller shall use commercially reasonable efforts to provide appropriate notice of the hearings on the Sale Motion to all Persons entitled to notice, including, but not limited to, all Persons that have asserted Liens on the Transferred Assets, all parties to the Assigned Contracts and all taxing authorities in jurisdictions applicable to Seller and as otherwise required by the Bankruptcy Code and bankruptcy rules.
- (h) Within five (5) Business Days of the Auction (subject to the Bankruptcy Court's availability), if Purchaser is the successful bidder at the Auction (or if there is no Auction), Seller will seek entry of the Sale Order by the Bankruptcy Court.

(i) The Seller and Purchaser agree that, in the event that Purchaser is not the winning bidder at an auction undertaken pursuant to the Bid Procedures Order (the "<u>Auction</u>"), and (i) Purchaser submits the Back-Up Bid at the Auction or (ii) the terms of this Agreement are deemed to constitute a Back-Up Bid, then Purchaser shall be obligated to promptly consummate the Transactions upon the terms and conditions as set forth herein, including the payment of the Purchase Price as the same may be increased by Purchaser at the Auction; <u>provided</u> that, the Seller gives written notice to Purchaser on or before the Back-up Termination Date, stating that the Seller (A) failed to consummate the sale of the Transferred Assets with the winning bidder, and (B) terminated the purchase agreement with the winning bidder.

5.3 Assumption of Assigned Contracts.

- (a) On or before the date that is five (5) Business Days following the date on which the Bid Procedures Order is entered by the Bankruptcy Court, the Seller shall file (or cause to be filed) a notice of assumption (the "Assumption Notice") with the Bankruptcy Court and serve such notice on each counterparty to an Assigned Contract listed thereon. The Assumption Notice shall identify all Assigned Contracts that the Seller and Purchaser believe may be assumed and assigned in connection with the sale of the Transferred Assets and set forth a good faith estimate of the amount of Cure Costs applicable to each such Assigned Contract (and if no Cure Cost is estimated to be applicable with respect to any particular Assigned Contract, the amount of such Cure Cost designated for such Assigned Contract shall be "\$0.00"). In accordance with the Bid Procedures Order, the Seller reserves the right to supplement such list of Assigned Contracts and provide additional notice of assumption, and to remove an Assigned Contract from the list of Assigned Contracts, up to five days prior to the hearing by the Bankruptcy Court with respect to the Sale Motion.
- (b) On or before the date that is five (5) Business Days before the Closing Date (the "Designation Deadline"), Purchaser shall provide to the Seller a list of those Assigned Contracts that Purchaser elects to have assumed and assigned to Purchaser on the Closing Date (the "Designated Contracts"). Purchaser shall be entitled to remove certain Assigned Contracts from the list of Designated Contracts at any time prior to the Designation Deadline by providing the Seller written notice of such removal. In the event that Purchaser removes any of such Assigned Contracts from such list, the Seller will provide the relevant counterparty written notice that the applicable Assigned Contract is no longer identified as a Designated Contract. For the avoidance of doubt, only those executory Assigned Contracts that remain identified as Designated Contracts as of the Closing Date will constitute Assigned Contracts and will be assumed by the Seller and assigned to Purchaser pursuant to the Sale Order. The Seller shall file such motions or pleadings as may be appropriate or necessary to assume and assign the Assigned Contracts and to determine the amount of the Cure Costs; provided, that nothing herein shall preclude the Seller from filing one or more motion to reject any Contracts that are not Assigned Contracts.
- (c) Notwithstanding any provision in this Agreement to the contrary, a Contract shall not be a Designated Contract hereunder and shall not be assigned to, or assumed by, Purchaser to the extent that such Contract is (i) deemed rejected under Section 365 of the Bankruptcy Code, (ii) the subject of an objection to assignment or assumption or requires the consent of any Governmental Authority or other third party (other than, and in addition to, the

Bankruptcy Court) in order to permit the assumption and assignment by the applicable Seller to Purchaser of such Contract pursuant to Section 365 of the Bankruptcy Code, and such objection has not been resolved or such consent has not been obtained prior to the thirtieth day following the Closing Date (as such period may be extended by mutual agreement of Seller and Purchaser), or (iii) is terminated by any party thereto other than Seller, or terminates or expires by its terms, on or prior to such time as it is to be assumed by Purchaser as a Designated Contract hereunder and is not continued or otherwise extended upon assumption. In no event shall the failure to assign to Purchaser any Contract in accordance with subsections (i) through (iii) above reduce the Purchase Price payable to Seller or constitute a failure to satisfy the conditions precedent of Seller under Section 8.3, it being understood that the foregoing shall not relieve Seller of its obligation to deliver, or cause to be delivered, the PRF Novation Agreement, the Merck Side Letter, and the Sublicense Agreement as a condition to Closing pursuant to Section 8.3(c).

- (d) Subject to the terms of <u>Section 2.5</u>, <u>Section 2.8</u>, <u>Section 5.3(a)</u> and <u>Section 5.3(b)</u>, Purchaser shall make provision for the payment of the Determined Cure Costs in cash at Closing in accordance with the Sale Order.
- (e) Notwithstanding any provision in this Agreement to the contrary, from and after the date of the Assumption Notice through the Closing Date, the Seller will not reject or take any action (or fail to take any action that would result in rejection by operation of Applicable Law) to reject, withdraw, repudiate or disclaim any Assigned Contract unless (i) Purchaser has provided its prior written consent; or (ii) Purchaser has removed such Assigned Contract from the list of Designated Contracts.

ARTICLE 6. PRE-CLOSING COVENANTS

- 6.1 <u>Conduct of Business</u>. Except (i) as set forth on <u>Schedule 6.1</u>, (ii) as may be approved by Purchaser (which approval will not be unreasonably withheld, delayed or conditioned; <u>provided</u>, <u>however</u>, that the consent of Purchaser shall be deemed to have been given if Purchaser does not object within 48 hours after written request for such consent is provided by the Seller to Purchaser), (iii) for actions taken or omitted to be taken by any member of the Seller Group in response to any Public Health Measure, or (iv) as is otherwise permitted, contemplated or required by this Agreement, any Assigned Contract, by Applicable Laws or by order of the Bankruptcy Court, from the Agreement Date through the earlier of the Closing Date or the termination of this Agreement in accordance with its terms:
- (a) The Seller Group shall use their commercially reasonable efforts to carry on the Business in all material respects in the ordinary course of business as it has been conducted since the Petition Date; and
 - (b) The Seller shall not, and shall cause its Affiliates not to:
 - (i) sell, license, abandon or otherwise dispose of any material asset or property constituting Transferred Assets other than, in each case, in the ordinary course of business or for the purpose of disposing of obsolete or worthless assets;

Casse2248800409ssjjj111 DDocc818621 Filitelc0244225225 Hintereclc02442252251171047595 DDessc MairEldothichtrien1Pagle869e088113112

- (ii) except in the ordinary course of business, acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of any business or any corporation, partnership or other business organization or otherwise acquire any assets (except inventory), that as of the Closing would constitute Transferred Assets, except for the acquisition of assets in the ordinary course of business;
- (iii) change its present accounting methods or principles in any material respect, except as required by GAAP or Applicable Law; or
- (iv) make or change any Tax election, change an annual accounting period, adopt or change any Tax accounting method, file any amended Tax Return, enter into any closing agreement, settle any material Tax claim or assessment or surrender any right to claim a refund of Taxes, other than in the ordinary course of business or as required by the Code or Applicable Law, and in each case that could have a material effect on the amount of Taxes due from the Business or due as a result of the Transferred Assets for a taxable period (or portion thereof) beginning after the Closing Date.
- (c) Notwithstanding anything to the contrary, nothing contained in this Agreement shall give Purchaser or any of its Affiliates, directly or indirectly, any right to control or direct the Business, assets and operations prior to the Closing. Prior to the Closing, the Seller shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its Business, assets and operations.

6.2 Access to Information; Confidentiality.

From the Agreement Date until the earlier of the Closing Date and the (a) termination of this Agreement, the Seller shall grant Purchaser and its representatives (at Purchaser's sole cost and expense) reasonable access, during normal business hours and upon reasonable notice (and in the event of a facility visit request, at least 48 hours prior notice), and subject to any limitations resulting from any Public Health Measures, to the personnel, facilities, book and records of the Seller Group related to the Business or the Transferred Assets that are in the possession or under the control of the Seller Group; provided, however, that (i) all requests for access shall be directed to such other person(s) as the Seller may designate in writing from time to time (the "Seller Access Contact"), (ii) such activities do not unreasonably interfere with the ongoing business or operations of the Seller Group, (iii) the Seller shall have the right to have one or more of its representatives present at all times during any visits, examinations, discussions or contacts contemplated by this Section 6.2(a), (iv) Purchaser shall have no right to perform invasive or subsurface investigations or conduct any sampling or analysis of environmental media of the nature commonly referred to as a "Phase II Environmental Investigation," such as any soil or groundwater testing, (v) such access or related activities would not cause a violation of any agreement to which any Seller is a party, (vi) no Personal Information shall be disclosed or used other than in compliance with applicable privacy law and (vii) nothing herein shall require any member of the Seller Group or their representatives to furnish to Purchaser or provide Purchaser with access to information that (A) is subject to an attorney-client or an attorney work-product privilege, (B) legal counsel for the Seller Group reasonably concludes may give rise to antitrust or competition law issues or violate a protective order or otherwise may not be disclosed pursuant to Applicable Law (including any Public Health Measure) or (C)

would cause significant competitive harm to the Seller Group if the Transactions are not consummated.

- (b) Notwithstanding anything to the contrary contained in this Agreement, from the Agreement Date until the Closing Date, Purchaser shall not, and shall cause its representatives not to, have any contact or discussions concerning any member of the Seller Group, the Business, the Transaction or any other matters with any lender, borrower, creditor, guarantor, business partner, bank, landlord, tenant, supplier, customer, employee, manager, franchisee, distributer, noteholder, independent contractor, consultant or other material business relation of any Seller, in each case, without the prior written consent of the Seller Access Contact (which consent may be withheld in the Seller's sole discretion and, if given, may be conditioned on the Seller Access Contact or his or her designee having the right to participate in any meeting or discussion).
- (c) Any information provided to or obtained by Purchaser or its representatives, including pursuant to this Section 6.2 is confidential information and subject to the terms of, and the restrictions contained in, the Confidentiality Agreement. Purchaser agrees to be bound by and comply with the provisions set forth in the Confidentiality Agreement as if such provisions were set forth herein, and such provisions are hereby incorporated herein by reference. Effective upon (and only upon) the Closing, the Confidentiality Agreement shall automatically terminate and none of the parties thereto shall have any further Liability or obligation thereunder except with respect to any confidential information provided to or obtained by Purchaser or its representatives concerning the Seller Group, which information shall remain subject to the terms and conditions of the Confidentiality Agreement after the Closing Date. If this Agreement is terminated prior to Closing for any reason, the duration of the confidentiality of the Confidentiality Agreement shall be deemed extended, without any further action by the parties, for a period of time equal to the period of time elapsed between the date such Confidentiality Agreement was initially signed and the date of termination of this Agreement.
- 6.3 Efforts to Consummate. Except as otherwise provided in this Agreement, each of the parties hereto agrees to use its commercially reasonable efforts to cause the Closing to occur as soon as possible after the Agreement Date, including satisfying the conditions precedent set forth in ARTICLE 8 applicable to such party including (a) defending against any Actions, judicial or administrative, challenging this Agreement or the consummation of the Transactions, (b) seeking to have any preliminary injunction, temporary restraining order, stay or other legal restraint or prohibition entered or imposed by any court or other Governmental Authority that is not yet final and nonappealable vacated or reversed, and (c) and executing any additional instruments reasonably requested by another party hereto (without cost or expense to the executing party) necessary to carry out the Transactions and to fully carry out the purposes of this Agreement; provided, however, that, for purposes of "commercially reasonable efforts" standard as required by this Section 6.3, Section 6.4 or Section 6.5, neither the Seller nor its Affiliates or representatives shall be required to offer or grant any accommodation or concession (financial or otherwise) to any third party or to otherwise expend any money or suffer any detriment, to expend any money to remedy any breach of any representation or warranty hereunder, to commence any Action, to waive or surrender any right, to modify any agreement (including any Assigned Contract) or to provide financing to Purchaser for the consummation of the Transactions.

6.4 Notices and Consents. Reasonably promptly following the execution of this Agreement, the Seller will give, or cause to be given, applicable notices to third parties and thereafter will use commercially reasonable efforts (as limited by Section 6.3) to obtain the third-party consents set forth on Schedule 6.4; provided, however, that no representation, warranty, covenant or agreement of the Seller shall be breached or deemed breached, and no condition shall be deemed not satisfied, as a result of (a) the failure to obtain any such third-party consent (unless such consent is part of a closing condition of Seller), (b) any termination of a Contract as a result of the failure to obtain such third-party consent (unless such consent is part of a closing condition of Seller) or (c) any Action commenced or threatened by or on behalf of any Person arising out of or relating to the failure to obtain any such consent or any such termination; provided, further, that nothing in this Section 6.4 shall require the Seller to expend any money or grant any concessions to obtain any such third-party consent (unless Purchaser provides the funds for or reimburses the Seller for such payment).

6.5 **Regulatory Matters**.

- Purchaser and the Seller will establish a mutually acceptable and prompt communication and interaction process to ensure the orderly transfer of the NDA for Zokinvy in the Progeria Field and other similar regulatory approval and authorization documents for jurisdictions outside of the United States. Promptly after Closing, the Parties shall file with the FDA, and any other relevant Governmental Authority all information required in order to transfer the NDA and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, including the information required pursuant to 21 C.F.R. § 314.72, or any successor regulation thereto, any authorization letters or notices, and letters of acceptance. Seller shall file the information required of a former owner, and Purchaser shall file the information required of a new owner, at each Party's own expense. Both Purchaser and the Seller also agree to use all commercially reasonable efforts to take any actions required by the Governmental Authority or other government/health agencies to effect the transfer of the NDAs and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, and hereby further agree to cooperate with each other in order to effectuate the foregoing transfer of Zokinvy in the Progeria Field. The Parties agree to use all commercially reasonable efforts to complete the filing of the transfer of the NDAs and other similar regulatory approval and authorization documents for jurisdictions outside of the United States within ten (10) days from the Closing Date. The Seller may retain an archival copy of the NDAs and other similar regulatory approval and authorization documents for jurisdictions outside of the United States, including supplements and records that are required to be kept under 21 C.F.R. § 314.81 or other similar regulation.
- (b) From and after the Closing Date until the Seller is dissolved, the Seller shall cooperate with Purchaser in preparing, disclosing and providing any relevant records, reports, responses or any other documentation that are required to be made, maintained and reported pursuant to the Governmental Authority in the Territory. The Parties agree to use their commercially reasonable efforts to take any other actions required by the FDA or any other Governmental Authority to effect the transaction.
- (c) Until the completion of the transfer of Zokinvy in the Progeria Field to Purchaser, the Seller shall take all reasonably necessary or advisable actions to maintain the

relevant NDA and other similar regulatory approval and authorization documents for jurisdictions outside of the United States.

- 6.6 <u>Public Announcements</u>. Between the Agreement Date and the Closing Date, except to the extent required by any Applicable Law or Action (including the Bankruptcy Cases), neither Purchaser nor the Seller shall, and Purchaser and the Seller shall cause their respective Affiliates and representatives not to, directly or indirectly, issue any press release or public announcement of any kind without the prior written consent of Purchaser and the Seller; provided, however, that the Seller and its Affiliates may make announcements from time to time to their respective employees, customers, suppliers, and other business relations and otherwise as the Seller may reasonably determine is necessary to comply with Applicable Law or the requirements of this Agreement or any other agreement to which any Seller or any such Affiliate is a party. Purchaser and the Seller shall cooperate in good faith to prepare a joint press release to be issued on the Closing Date, the terms of which shall be mutually agreed upon by the parties.
- 6.7 <u>Update of Schedules; Knowledge of Breach</u>. From time to time prior to the Closing, the Seller may supplement or amend the Schedules with respect to any matter first arising after the Agreement Date that would have been required to be set forth or described in such Schedules. Any such supplemental or amended disclosure shall not be deemed to have cured any such breach of representation or warranty for purposes of determining whether or not the conditions set forth in <u>Section 8.2(a)</u> have been satisfied. From and after the Closing, references to the Schedules shall be references to the Schedules as supplemented, modified and/or updated. If, prior to the Closing, Purchaser shall have reason to believe that any breach of a representation or warranty of the Seller has occurred (other than through notice from the Seller), Purchaser shall promptly so notify the Seller, in reasonable detail. Nothing in this Agreement, including this <u>Section 6.7</u>, shall imply that the Seller is making any representation or warranty as of any date other than the Closing Date (other than representations and warranties that are expressly made as of an earlier date).

ARTICLE 7. POST-CLOSING COVENANTS

7.1 Access to Information; Books and Records. From and after the Closing, Purchaser and its Affiliates shall (i) afford the Seller Group and their respective representatives reasonable access, during normal business hours, upon reasonable advance notice and under reasonable circumstances, to the books and records of Purchaser and the Business shall permit the Seller Group and their respective representatives to examine and copy such books and records to the extent reasonably requested by such party and (ii) cause their representatives to furnish all information reasonably requested by any member of the Seller Group or their representatives in connection with financial or regulatory reporting, audit, third party litigation, preparing or filing of any Tax Return or the defense of any Tax claim or assessment or any other business purpose; provided, however, that nothing in this Section 7.1 shall require Purchaser or its Affiliates to furnish to the Seller Group or their respective representatives any material that is subject to an attorney-client or solicitor-client privilege or an attorney or solicitor work-product privilege or which may not be disclosed pursuant to Applicable Law. For a period of six (6) years following the Closing Date, or such longer period as may be required by Applicable Law or necessitated by applicable statutes of limitations, Purchaser shall, and shall cause its Affiliates

to, maintain all such books and records in the jurisdiction in which such books and records were located prior to the Closing Date and shall not destroy, alter or otherwise dispose of any such books and records. On and after the end of such period, Purchaser shall, and shall cause its Affiliates to, provide the Seller with at least ten Business Days prior written notice before destroying, altering or otherwise disposing any such books and records, during which period the Seller may elect to take possession, at its own expense, of such books and records.

7.2 Post-Closing Receipt and Possession of Assets.

- (a) After the Closing Date, the Seller shall transfer promptly to Purchaser from time to time (but in any event on a monthly basis) any payments constituting Transferred Assets received by the Seller. After the Closing Date, Purchaser shall transfer promptly to the Seller, from time to time (but in any event on a monthly basis), any payments constituting Excluded Assets, including any accounts receivable constituting Excluded Assets, received by Purchaser after the Closing.
- (b) In the event that, after the Closing Date, Purchaser receives or otherwise is in possession of any other Excluded Asset, Purchaser shall promptly notify the Seller of its receipt or possession of such other Excluded Asset and transfer, at the Seller's expense, such Excluded Asset to the Seller. In the event that, after the Closing Date, the Seller receives or otherwise is in possession of any other Transferred Asset, the Seller shall promptly notify Purchaser of its receipt or possession of such other Transferred Asset and transfer, at Purchaser's expense (unless the Seller was required to transfer such Transferred Asset to Purchaser at Closing, in which case, and without limitation of any other remedies available to Purchaser, such transfer will be at the Seller's expense), such Transferred Asset to Purchaser.

7.3 Tax Matters.

- (a) All Taxes with respect to the income or operations of the Business or the ownership of the Transferred Assets that relate to any Straddle Period shall be apportioned between Seller and Purchaser as follows: (i) in the case of ad valorem or other property Taxes, on a per diem basis; and (ii) in the case of income, sales and use and withholding Taxes, employment Taxes, or other Taxes based on or measured by income, receipts or profits, as determined from the closing of the books and records of Seller and the Business at the close of business on the Closing Date.
- (b) After the Closing Date, Purchaser and Seller shall furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance (including access to books, records, work papers and Tax Returns for Pre-Closing Tax Periods) relating to the Business or the Transferred Assets as is reasonably necessary for the preparation of any Tax Return, claim for refund or audit, and the prosecution or defense of any claim, suit or proceeding relating to any proposed Tax adjustment. Upon reasonable notice, Seller and Purchaser shall make its employees and facilities available on a mutually convenient basis to provide reasonable explanation of any documents or information provided hereunder. The other party hereto shall promptly (and in no event later than 30 days after receipt of the request) provide the requested information. The requesting party shall indemnify the other party for any out-of-pocket expenses incurred by such party in connection with providing any

information or documentation pursuant to this <u>Section 7.3(b)</u>. Any information obtained under this <u>Section 7.3(b)</u> shall be kept confidential, except as otherwise reasonably may be necessary in connection with the filing of Tax Returns or claims for refund or in conducting any Tax audit, dispute or contest.

ARTICLE 8. CONDITIONS PRECEDENT

- 8.1 <u>Conditions to Each Party's Obligation</u>. The respective obligations of the parties hereto to effect the Transactions are subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller and Purchaser), at or prior to the Closing, of the following conditions:
- (a) <u>No Injunctions or Restraints</u>. No Order or Applicable Law preventing the consummation of the Transactions shall be in effect.
- (b) <u>Sale Order</u>. The Bankruptcy Court shall have entered the Sale Order and such Sale Order shall be a Final Order (unless such Final Order requirement is waived by the Seller and Purchaser in their respective sole discretion).
- 8.2 <u>Conditions to Obligations of Purchaser</u>. The obligations of Purchaser to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by Purchaser), at or prior to the Closing, of the following conditions:
- (a) <u>Representations and Warranties</u>. Each of the representations and warranties of the Seller set forth in <u>ARTICLE 3</u> shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality", "Material Adverse Effect" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not have, individually or in the aggregate, a Material Adverse Effect.
- (b) <u>Performance of Covenants and Obligations</u>. The Seller shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated expressly by the terms of this Agreement or caused by the Transactions.
- (c) <u>Closing Deliverables</u>. The Seller shall have delivered to Purchaser the closing deliveries required to be delivered by the Seller pursuant to <u>Section 2.8(a)</u>, <u>Section 2.8(b)</u>, <u>Section 2.8(c)</u>, <u>Section 2.8(d)</u>, <u>Section 2.8(e)</u>, and <u>Section 2.8(g)</u>. The Escrow Agent shall have delivered its duly executed signature page of the Escrow Agreement to the Purchaser pursuant to <u>Section 2.8(d)</u>.
- 8.3 <u>Conditions to Obligations of the Seller</u>. The obligation of the Seller to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller), at or prior to the Closing, of the following conditions:

- (a) <u>Representations and Warranties</u>. Each of the representations and warranties of Purchaser set forth in <u>ARTICLE 4</u> shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not, individually or in the aggregate, (i) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (ii) otherwise prevent, hinder or delay the consummation of the Transactions.
- (b) <u>Performance of Covenants and Obligations of Purchaser</u>. Purchaser shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated by the terms of this Agreement or caused by the Transactions.
- (c) <u>Closing Deliverables</u>. Purchaser shall have delivered to the Seller the closing deliveries required to be delivered by Purchaser pursuant to <u>Section 2.8(a)</u>, <u>Section 2.8(b)</u>, <u>Section 2.8(c)</u>, <u>Section 2.8(d)</u>, <u>Section 2.8(e)</u>, and <u>Section 2.8(f)</u>. The Escrow Agent shall have delivered its duly executed signature page of the Escrow Agreement to the Seller pursuant to <u>Section 2.8(d)</u>.
- 8.4 <u>Waiver of Condition; Frustration of Conditions</u>. All conditions to the Closing shall be deemed to have been satisfied or waived from and after the Closing. Neither Purchaser nor the Seller may rely on the failure of any condition set forth in this <u>ARTICLE 8</u>, as applicable, to be satisfied if such failure was caused by such party's failure to use, as required by this Agreement, its reasonable best efforts to consummate the Transactions.
- 8.5 <u>Delivery of a Notice of Readiness to Close</u>. At any time after the Seller's satisfaction of its conditions to Closing in accordance with the terms of <u>Section 8.1</u> and <u>Section 8.3 of this Agreement</u>, the Seller may deliver a notice to the Purchaser (a "<u>Notice of Readiness to Close</u>"). The Purchaser shall have three (3) Business Days from delivery of a Notice of Readiness to Close to satisfy its conditions to Closing in accordance with the terms of <u>Section 8.1</u> and <u>Section 8.2 of this Agreement and consummate the Transactions</u>. If Purchaser does not satisfy its conditions to Closing and consummate the Transaction within three (3) Business Days, Purchaser shall forfeit the entire Deposit Escrow Amount to the Seller.

ARTICLE 9. TERMINATION

- 9.1 <u>Events of Termination</u>. Notwithstanding anything to the contrary, this Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing:
 - (a) by mutual written consent of Purchaser and the Seller;
- (b) automatically, upon (i) the consummation of a sale or other disposition of all or substantially all of the Transferred Assets to a Person other than Purchaser (each,

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduitn1en1Pagle 266c951131112

- an "<u>Alternate Transaction</u>"), (ii) if, at close of the Auction, Purchaser's bid has not been selected as either the winning bid or the Back-Up Bid or (iii) if, at the close of the Auction, Purchaser's bid was selected as the Back-Up Bid, upon the consummation of a Competing Bid or Alternative Transaction;
- (c) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if the Bankruptcy Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code;
- (d) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if Purchaser is not selected as having the winning bid or Back-Up Bid at Auction, if any;
- (e) by Purchaser if the Seller (i) withdraws the motion for the Sale Order, or publicly announces its intention to withdraw such motion, (ii) moves to voluntarily dismiss the Bankruptcy Cases, (iii) moves for conversion of the Bankruptcy Cases to Chapter 7 of the Bankruptcy Code, or (iv) moves for appointment of an examiner with expanded powers pursuant to Section 1104 of the Bankruptcy Code or a trustee in the Bankruptcy Cases;
- (f) by Purchaser, by written notice from Purchaser to the Seller, if there has been a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement, such that the conditions in <u>Section 8.1</u> or <u>Section 8.2</u> are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by the Seller prior to the earlier of (i) 20 Business Days after receipt of written notice from Purchaser requesting such breach be cured or (ii) the Outside Date; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 9.1(f)</u> shall not be available to Purchaser if the failure of Purchaser to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in <u>Section 8.1</u> or <u>Section 8.3</u> are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement;
- (g) by the Seller, by written notice from the Seller to Purchaser, if there has been a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement, such that the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by Purchaser prior to the earlier of (i) 20 Business Days after receipt of written notice from the Seller requesting such breach be cured or (ii) the Outside Date; provided, however, that the right to terminate this Agreement pursuant to this Section 9.1(g) shall not be available to the Seller if the failure of the Seller to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement;
- (h) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if any Governmental Authority of competent jurisdiction shall have issued an Order, enacted any Applicable Law or taken any other action restraining, enjoining or otherwise prohibiting the consummation of the Transactions and, in the case of Orders and other actions,

such Order or other action shall have become Final Orders; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 9.1(h)</u> shall not be available to the party seeking to terminate if any action of such party or any failure of such party to act has contributed to such Order or other action and such action or failure constitutes a breach of this Agreement;

- (i) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if the Closing has not occurred on or prior to June 1, 2024 (the "Outside Date"); provided, however, that the party exercising the right to terminate this Agreement pursuant to this Section 9.1(i) shall not have been responsible for such failure of the Closing to occur through a breach or inaccuracy of a covenant, representation or warranty contained in this Agreement (it being understood, acknowledged, and agreed that if Seller is unable to provide any required Closing deliverable of Seller, then Seller shall be deemed to have been responsible for such failure of the Closing for purposes of this Section 9.1(i)); or
- (j) by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a), Section 9.3(b), and Section 9.3(c).

9.2 **Effect of Termination**.

- (a) In the event that this Agreement shall be terminated pursuant to Section 9.1, (a) Purchaser and its representatives shall promptly return all documents, work papers and other materials of the Seller including any confidential information and (b) all further obligations of the parties hereto under this Agreement shall terminate without further Liability or obligation to the other parties hereto; provided, however, that, notwithstanding the foregoing, the Liabilities and obligations under (i) the Confidentiality Agreement, and (ii) Section 2.9(c), Section 6.2(c), this Section 9.2 and ARTICLE 10 shall continue in full force and effect.
- (b) Notwithstanding anything to the contrary in this Agreement, in the event of valid termination of this Agreement pursuant to Section 9.1, (i) the Seller's Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement shall be capped at an amount equal to the Deposit Escrow Amount, and (ii) no such termination shall relieve Purchaser from any Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement (including if this Agreement is terminated by the Seller pursuant to Section 9.1(g)) and the Seller shall be entitled to all remedies available at law or in equity, including payment of the Deposit Escrow Amount pursuant to Section 2.9(c).

9.3 <u>Termination Fee and Expense Reimbursement.</u>

(a) Subject to limitations set forth in the Bid Procedures Order, in consideration of Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof, and the identification and quantification of assets to be included in the Transferred Assets, and to compensate Purchaser as a stalking-horse bidder, the Seller shall pay in cash to Purchaser, by wire transfer of immediately available funds to the account specified by Purchaser to the Seller in writing, an amount equal to the Termination Fee in the event that this Agreement is terminated pursuant to Section 9.1(b), in which case the

Casse2248800409sgijj11 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MainEldoiduithien Pagle 866097.103112

Termination Fee shall be due and payable simultaneously with any termination of this Agreement; provided, that, Purchaser shall not be entitled to the fee described in this Section 9.3(a) to the extent Purchaser is in material breach of this Agreement at the time this Agreement is terminated pursuant to Section 9.1(b) if Seller has provided notice of such material breach to Purchaser and such material breach has remained uncured for more than five (5) Business Days after Purchaser's receipt of such notice. The Termination Fee shall be payable solely from the proceeds of such Competing Bid or Alternative Transaction. The Seller's obligation to pay the Termination Fee pursuant to this Section 9.3(a) shall survive termination of this Agreement and shall constitute an administrative expense of the Seller under section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind specified in section 503(b) or 507(b) of the Bankruptcy Code.

- Subject to limitations set forth in the Bid Procedures Order, in consideration of Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof, and the identification and quantification of assets to be included in the Transferred Assets, if this Agreement is terminated in accordance with the terms set forth in Section 9.1(b), then the Seller shall pay to Purchaser in cash not later than two (2) Business Days following receipt of documentation supporting the request for reimbursement of costs, fees and expenses, the Expense Reimbursement, in an amount not to exceed \$600,000, by wire transfer of immediately available funds to an account specified by Purchaser to the Seller in writing; provided, that, Purchaser shall not be entitled to the fee described in this Section 9.3(a) to the extent Purchaser is in material breach of this Agreement at the time this Agreement is terminated pursuant to Section 9.1(b) if Seller has provided notice of such material breach to Purchaser and such material breach has remained uncured for more than five (5) Business Days after Purchaser's receipt of such notice. The Expense Reimbursement shall be payable solely from the proceeds of such Competing Bid or Alternative Transaction. Seller's obligation to pay the Expense Reimbursement pursuant to this Section 9.3(b)9.3(b) shall survive termination of this Agreement and shall constitute an administrative expense of Seller under section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind specified in section 503(b) or 507(b) of the Bankruptcy Code.
- (c) The Seller agrees and acknowledges that Purchaser's due diligence, efforts, negotiation, and execution of this Agreement have involved substantial investment of management time and have required significant commitment of financial, legal, and other resources by Purchaser, and that such due diligence, efforts, negotiation, and execution have provided value to the Seller and, in the Seller's reasonable business judgment, is necessary for the preservation of the value of the Seller's estate. The Seller further agrees and acknowledges that the Termination Fee and the Expense Reimbursement are not a penalty, but rather represent liquidated damages that are reasonable in relation to Purchaser's efforts, Purchaser's lost opportunities from pursuing the Transactions, and the magnitude of the Transactions. The provision of the Termination Fee and the Expense Reimbursement is an integral part of this Agreement, without which the Purchaser would not have entered into this Agreement.

ARTICLE 10. GENERAL PROVISIONS

- 10.1 <u>Survival of Representations, Warranties and Covenants.</u> All covenants and agreements contained in this Agreement that by their term are to be performed in whole or in part, or which prohibit actions, subsequent to Closing shall, solely to the extent such covenants and agreements are to be performed, or prohibit actions, subsequent to Closing, survive the Closing in accordance with their terms until fully performed or satisfied. All other covenants and agreements contained herein, and all representations and warranties contained herein or in any certificated deliveries hereunder shall not survive Closing and shall therefor terminate, including any Action for damages in respect of any breach or inaccuracy thereof. Notwithstanding the foregoing, the provisions of Section 2.9(c), Section 6.2, Section 9.2, this Article 10 and the Confidentiality Agreement shall survive the Closing. For the avoidance of doubt, nothing in this Section 10.1 shall affect the survival of the covenants or representations or warranties of Seller under the Sublicense Agreement or its related agreements.
- 10.2 **Entire Agreement.** This Agreement, including the Exhibits and Schedules hereto, the Confidentiality Agreement and the Related Documents, contain the entire understanding of the parties hereto with respect to the subject matter contained herein and therein. This Agreement supersedes all prior and contemporaneous agreements, arrangements, contracts, discussions, negotiations, undertakings and understandings (including any letters of intent or term sheets), whether written or oral, among the parties with respect to such subject matter (other than, for the avoidance of doubt, the Confidentiality Agreement and the Related Documents) or any prior course of dealings. The parties hereto have voluntarily agreed to define their rights, Liabilities and obligations respecting the Transactions exclusively in contract pursuant to the express terms and conditions of this Agreement, the Confidentiality Agreement and the Related Documents, and the parties hereto expressly disclaim that they are owed any duties or entitled to any remedies not expressly set forth in this Agreement, the Confidentiality Agreement and the Related Documents. Furthermore, the parties each hereby acknowledge that this Agreement, the Confidentiality Agreement and the Related Documents embody the justifiable expectations of sophisticated parties derived from arm's-length negotiations, and all parties to this Agreement, the Confidentiality Agreement and the Related Documents specifically acknowledge that no party has any special relationship with another party that would justify any expectation beyond that of an ordinary purchaser and an ordinary seller in an arm's-length transaction. The sole and exclusive remedies for any Related Claims shall be those remedies available at law or in equity for breach of contract only (as such contractual remedies have been further limited or excluded pursuant to the express terms of this Agreement); and the parties hereby agree that neither party hereto shall have any remedies or cause of action (whether in contract or in tort or otherwise) of any statements, communications, disclosures, failures to disclose, representations or warranties not set forth in this Agreement.
- 10.3 <u>Amendment; No Waiver</u>. This Agreement and the Related Documents may be amended, supplemented or changed, and any provision hereof or thereof can be waived, only by a written instrument making specific reference to this Agreement (and, if applicable, the Related Documents) executed by the party against whom enforcement of any such amendment, supplement, modification or waiver is sought. The waiver by any party of a breach of any provision of this Agreement or the Related Documents shall not operate or be construed as a

Casse2248800409ssjýj111 DDocc83821 Filitelc0244225225 Hintereclc024225225171047595 DDessc MaiExDidioituínenPageP1200c991013112

further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any party to exercise, and no delay in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall a single or partial exercise of such right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy.

- Severability; Specific Versus General Provisions. Whenever possible, each provision of this Agreement and the Related Documents shall be interpreted in such manner as to be effective and valid under Applicable Law, but if any term or other provision of this Agreement or the Related Documents is invalid, illegal, or incapable of being enforced by any Applicable Law or public policy, all other terms or provisions of this Agreement and the Related Documents shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, in whole or in part, such term or provision is hereby deemed modified to give effect to the original written intent of the parties to the greatest extent consistent with being valid and enforceable under Applicable Law. No party hereto shall assert, and each party shall cause its respective Affiliates or related parties not to assert, that this Agreement or any part hereof is invalid, illegal or unenforceable. Notwithstanding anything to the contrary, to the extent that a representation, warranty, covenant or agreement of the Seller contained in this Agreement or the Schedules (each, a "Provision") addresses a particular issue with specificity (a "Specific **Provision**"), and no breach by the Seller exists under such Specific Provision, the Seller shall not be deemed to be in breach of any other Provision (with respect to such issue) that addresses such issue with less specificity than the Specific Provision, and if such Specific Provision is qualified or limited by the Seller's Knowledge, or in any other manner, no other Provision shall supersede or limit such qualification in any manner.
- 10.5 <u>Expenses and Obligations</u>. Except as otherwise provided in this Agreement, all costs and expenses incurred by the parties hereto in connection with the Transactions, including the costs, expenses and disbursements of counsel and accountants, shall be borne solely and entirely by the party that has incurred such expenses; <u>provided</u>, <u>however</u>, that Purchaser shall pay, or promptly reimburse the Seller for, any filing fees which relate to any required governmental filing or notification and Purchaser shall pay any Transfer Taxes.
- 10.6 Notices. All notices, consents, waivers, and other communications under this Agreement or the Related Documents must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery, (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service, (c) if delivered by electronic mail (unless the sender receives an automated message that the email has not been delivered) on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day) and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or email addresses set forth below (or to such other addresses as a party may designate by notice to the other parties in accordance with this Section 10.6):

Casse2248800409sggj11 DDocc81821 Filiteck0044225225 Einteccck00442252251171047595 DDessc MairExhibitm1entPage 1.04 100 0.13 112

If to Purchaser:

Sentynl Therapeutics, Inc. 420 Stevens Avenue, Suite 200 Solana Beach, CA 92075 Attn: Matt Heck, Chief Executive Officer

Email: mheck@sentynl.com

with a copy to (which will not constitute notice):

Pillsbury Winthrop Shaw Pittman LLP 11682 El Camino Real, Suite 200 Attn: Christian A. Salaman and Jason Stirling email: christian.salaman@pillsburylaw.com jason.stirling@pillsburylaw.com

If to the Seller:

Eiger BioPharmaceuticals, Inc. 2155 Park Boulevard Palo Alto, CA 94306-1543

Attn: David Apelian, Chief Executive Officer

Email: dapelian@eigerbio.com

with a copy to (which will not constitute notice):

Sidley Austin LLP 2021 McKinney Ave., Suite 2000 Dallas, TX 75201 Attention: Thomas R. Califano William E. Curtin

Email: tom.califano@sidley.com wcurtin@sidley.com anne.wallice@sidley.com

Anne G. Wallice

10.7 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by electronic transmission in .PDF format, or other agreed format shall be sufficient to bind the parties to the terms and conditions of this Agreement. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement or any Related Document, shall be disregarded in determining the party's intent or the effectiveness of such signature.

10.8 <u>Governing Law</u>. This Agreement, the Related Documents and all Related Claims shall be governed by the internal laws of the State of Delaware (including its statute of limitations), without giving effect to any choice or conflict of law principles or rules that would cause the application of the Applicable Laws of any other jurisdiction.

10.9 Submission to Jurisdiction; Consent to Service of Process.

- (a) Without limiting any party's right to appeal any Order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to interpret and/or enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any Related Document, any breach or default hereunder or thereunder, or the Transactions, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 10.6; provided, however, that if the Bankruptcy Cases have closed, the parties agree to irrevocably submit to the exclusive jurisdiction of the United States District Court for the Northern District of Texas over all Related Claims, and each party hereto hereby irrevocably agrees that all Related Claims may be heard and determined in such courts. The parties hereto hereby irrevocably and unconditionally waive, to the fullest extent permitted by Applicable Law, any objection which they may now or hereafter have to the laying of venue of any such Related Claim brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Applicable Law.
- (b) Each of the parties hereto hereby consents to process being served by any party to this Agreement in any Related Claim by the delivery of a copy thereof in accordance with the provisions of Section 10.6 (other than by email) along with a notification that service of process is being served in conformance with this Section 10.9(b). Nothing in this Agreement will affect the right of any party to serve process in any other manner permitted by Applicable Law.
- 10.10 Waiver of Jury Trial. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY EXPRESSLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY PROCEEDING OR RELATED CLAIM BROUGHT BY OR AGAINST IT, DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS.
- 10.11 <u>Rights Cumulative</u>. All rights and remedies of each of the parties under this Agreement and the Related Documents will be cumulative, and the exercise of one or more rights or remedies will not preclude the exercise of any other right or remedy available under this Agreement, the Related Documents or Applicable Law.

- 10.12 **Assignment**. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors by operation of law and permitted assigns of the parties hereto. No assignment of this Agreement or any of the rights, interests or obligations under this Agreement may be made by any party hereto at any time, whether or not by operation of law, without the prior written consent of the Seller and Purchaser, and any attempted assignment without the required consent shall be void; provided, however, that (a) Purchaser may assign (i) any of its rights or delegate any of its duties under this Agreement to any of its Affiliates, and (ii) its rights, but not its duties, under this Agreement to any of its financing sources and (b), the Seller may assign any of its rights or delegate any of its duties under this Agreement (i) to any of its Affiliates, (ii) to any creditor or group of creditors pursuant to an order of the Bankruptcy Court entered in the Bankruptcy Cases, including Seller's rights to payment hereunder and rights and ability to enforce the terms of this Agreement and (iii) for collateral security purposes to any lender of the Seller or its Affiliates; provided, further, however, that, in each case, such assignment shall not release Purchaser from its obligations under this Agreement and the Seller shall have no obligation to pursue remedies against any assignee of Purchaser before proceeding against Purchaser for any breach of Purchaser's obligations hereunder.
- 10.13 Specific Enforcement; Remedies. The parties hereto agree that irreparable damage (for which monetary relief, even if available, would not be an adequate remedy) would occur in the event that any of the provisions of this Agreement were not performed by the parties hereto in accordance with their specific terms or were otherwise breached. It is accordingly agreed that (i) Purchaser, on the one hand, and the Seller, on the other hand, shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of competent jurisdiction without proof of damages or otherwise and that this shall include the right of the Seller to cause Purchaser to fully perform the terms of this Agreement to the fullest extent permissible pursuant to this Agreement and Applicable Laws and to thereafter cause this Agreement and the Transactions to be consummated on the terms and subject to the conditions thereto set forth in this Agreement, and (ii) the right of specific performance and other equitable relief is an integral part of the Transactions and without that right, neither the Seller nor Purchaser would have entered into this Agreement. Remedies shall be cumulative and not exclusive and shall be in addition to any other remedies which any party may have under this Agreement. Each of the parties hereto hereby (A) waives any defenses in any action for specific performance, including the defense that a remedy at law would be adequate, (B) waives any requirement under any Applicable Law to post a bond or other security as a prerequisite to obtaining equitable relief and (C) agrees not to assert that a remedy of specific performance or other equitable relief is unenforceable, invalid, contrary to law or inequitable for any reason, and not to assert that a remedy of monetary damages would provide an adequate remedy or that the parties otherwise have an adequate remedy at law. Notwithstanding anything to the contrary, in no event shall this <u>Section 10.13</u> be used, alone or together with any other provision of this Agreement, to require the Seller to remedy any breach of any representation or warranty of the Seller.
- 10.14 <u>Third-Party Beneficiaries</u>. Except as set forth in <u>ARTICLE 2</u> (with respect to the Seller), <u>Section 10.15</u> (with respect to the Nonparty Affiliates), <u>Section 10.16</u> (with respect to the released parties identified therein), <u>Section 10.17</u> (with respect to the Sellers' Group Members) and the next sentence, nothing in this Agreement, express or implied, is intended to

confer upon any Person other than the parties hereto any rights or remedies of any nature whatsoever under or by reason of this Agreement. From and after the Closing, all of the Persons identified as third-party beneficiaries in the first sentence of this Section 10.14 shall be entitled to enforce such provisions and to avail themselves of the benefits of any remedy for any breach of such provisions, all to the same extent as if such Persons were parties to this Agreement. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with this Agreement without notice or Liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any party hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the Agreement Date or as of any other date.

10.15 No Personal Liability of Directors, Officers and Owners. All Related Claims may be made only against (and are those solely of) the entities that are expressly identified as parties in the preamble to this Agreement (the "Contracting Parties"). No Person who is not a Contracting Party, including any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any Contracting Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any of the foregoing (collectively, "Nonparty Affiliates"), shall have any Liability pursuant to any Related Claim; and, to the maximum extent permitted by Applicable Law, each Contracting Party hereby waives and releases all such Liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Applicable Law, (a) each Contracting Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at Applicable Law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose Liability of a Contracting Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to this Agreement or the Related Documents.

10.16 **General Release**.

(a) Effective as of the Closing, the Seller, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "<u>Seller Releasing Party</u>"), hereby fully, irrevocably and unconditionally releases and forever discharges Purchaser and its respective past and present directors, managers, officers, employees, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, any and all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or

potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Seller Releasing Party has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Seller Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.

(b) Effective as of the Closing, Purchaser, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "Purchaser Releasing Party"), hereby fully, irrevocably and unconditionally releases and forever discharges the Seller, the Seller's Affiliates and its and their respective past and present directors, managers, officers, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Purchaser Releasing Party has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Purchaser Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.

10.17 **Legal Representation**. Purchaser and the Seller acknowledge and agree that the Law Firm has represented the Seller Group in connection with the negotiation, preparation, execution, delivery and performance of this Agreement and the Related Documents and the consummation of the Transactions, and that the Seller, its Affiliates and its partners, officers, directors and representatives (the "Seller Group Members") have a reasonable expectation that the Law Firm will represent them in connection with any Action involving any Seller Group Member, on the one hand, and Purchaser or any of its Affiliates and representatives (the "Purchaser Group Members"), on the other hand, arising under this Agreement, the Related Documents or the Transactions. Purchaser hereby, on behalf of itself and the other Purchaser Group Members, irrevocably: (a) acknowledges and agrees that any attorney-client privilege, solicitor-client privilege, work product or other attorney-client or solicitor-client confidential information ("Attorney-Client Information") arising from communications prior to the Closing between any Seller (including any one or more officers, directors or stockholders of such Seller), on the one hand, and the Law Firm, on the other hand, is not included in the property, rights, privileges, powers, franchises and other interests that are possessed by or vested in the Business or the Transferred Assets, that any such Attorney-Client Information shall be deemed property of, and controlled solely by, such Seller for the benefit and on behalf of the Seller Group Members and, upon request, convey and transfer any Attorney-Client Information to the Seller; (b) acknowledge and agree that the Seller Group Members shall have the right to retain, or cause the Law Firm to retain, any such documentation or information in the possession of the Law Firm or such Seller Group Members at the Closing; (c) agree not to access, retain or use any documentation or information constituting Attorney-Client Information and that no Purchaser Group Member shall have any right to waive any attorney-client privilege or other right to

Casse2248800499sgijj11 DDoc83821 Filitelc0244225225 Hintereclc024225225171047595 DDess c MairExhibitri1entPage 1.04 105 105 113

confidentiality with respect to such Attorney-Client Information; (d) disclaim the right to assert a waiver by any Seller Group Member with regard to the attorney-client privilege, solicitor-client privilege or other right to confidentiality with respect to such Attorney-Client Information solely due to the fact that such documentation or information is physically in the possession of Purchaser after the Closing; (e) consent to the Law Firm's representation after the Closing of any Seller Group Member in any Action that may relate to a Purchaser Group Member or the Transactions and consent to and waive any conflict of interest arising therefrom without the need for any future waiver or consent; and (f) consent to the disclosure by the Law Firm to any Seller Group Member of any documentation or information obtained by the Law Firm during the course of its representation of Seller or any Affiliate prior to the Closing, whether related to this Agreement, the Related Documents, the Transactions or otherwise, whether or not such disclosure is made prior to or after the Closing and whether or not the documentation or information disclosed is subject to any attorney-client privilege, solicitor-client privilege or confidentiality obligation to any Seller, any Affiliate of such Seller or any other Person. In the event that any Action arises after the Closing between any Purchaser Group Member and a Person other than a Seller Group Member, such Purchaser Group Member shall not disclose any documentation or information that is subject to an attorney-client privilege or other rights of confidentiality referenced in this Section 10.17 without the prior written consent of the applicable Seller; provided, however, that if such Purchaser Group Member is required by judicial order or other legal process to make such disclosure, such Purchaser Group Member shall promptly notify the applicable Seller in writing of such requirement (without making disclosure) and shall provide such Seller with such cooperation and assistance as shall be necessary to enable such Seller to prevent disclosure by reason of such attorney-client privilege, solicitor-client privilege or other rights of confidentiality. This Section 10.17 is for the benefit of the Seller Group Members and such Persons are intended third-party beneficiaries of this Section 10.17.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

Casse2248800449sgijj11 DDocc88621 Filitelc0044225225 <u>Hintereclc004422522517104759</u>5 DDessc MairExhibitri1entPage 107 106 13 112

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

PURCHASER:

SENTYNL THERAPEUTICS, INC. Docusigned by:

—DocuSigned by:

Matt Huk

Ву:____с4381FF94B614AС...

Name: Matt Heck Title: President and CEO

SELLER:

EIGER BIOPHARMACEUTICALS, INC.

—DocuSigned by:
Danid Apelian

Name: David Apelian

Title: Chief Executive Officer

AMENDMENT NO. 1 TO ASSET PURCHASE AGREEMENT

THIS AMENDMENT NO. 1 TO ASSET PURCHASE AGREEMENT (this "<u>Amendment</u>"), dated as of April 22, 2024 (the "<u>Amendment Date</u>") is entered into by and between Sentynl Therapeutics, Inc., a Delaware corporation ("<u>Purchaser</u>") and Eiger BioPharmaceuticals, Inc., a Delaware corporation (the "<u>Seller</u>").

RECITALS

WHEREAS, Purchaser and Seller are parties to that certain Asset Purchase Agreement, dated as of March 31, 2024 (the "<u>Agreement</u>" or "<u>Zokinvy Stalking Horse APA</u>");

WHEREAS, capitalized terms not herein defined shall have the meanings ascribed to them in the Agreement;

WHEREAS, Purchaser and Seller desire to amend the Agreement in accordance with and as set forth herein; and

WHEREAS, Section 10.3 of the Agreement provides that the Agreement may be amended by a written instrument making specific reference to the Agreement executed by the party against whom enforcement of such amendment is sought.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth in the Agreement and this Amendment and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. **Amendment of Agreement**. The Agreement is hereby amended as follows:
 - 1.1 The definition of "Base Price" found in Section 1.1 of the Agreement is hereby amended and restated in its entirety to read as follows:

""<u>Base Price</u>" means \$45,200,000 <u>provided</u>, <u>however</u>, that the Base Price shall be reduced by the amount of \$100,000 *per diem* for each calendar day that the Closing occurs between April 24, 2024, and May 31, 2024; <u>provided</u>, <u>further</u>, that, notwithstanding the reduction, the Base Price shall not be less than \$26,000,000 if Closing occurs no later than May 31, 2024."

- 1.2 The definition of "Expense Reimbursement" found in Section 1.1 of the Agreement is hereby deleted in its entirety.
- 1.3 The definition of "Sale Motion" found in Section 1.1 of the Agreement is hereby amended and restated in its entirety to read as follows:

""Sale Motion" means the motion of the Seller seeking entry of the Sale Order approving the terms herein, to be filed on or about April 1, 2024, in the Bankruptcy Cases."

- 1.4 The definition of "Termination Fee" found in Section 1.1 of the Agreement is hereby amended and restated in its entirety to read as follows:
- ""Termination Fee" means a fee equal to three percent (3.0%) of the Base Price."
- 1.5 Section 5.2(b) of the Agreement is hereby amended and restated in its entirety to read as follows:
- "(b) Within one (1) day after the Petition Date, Seller will file the Bid Procedures Motion seeking the Bankruptcy Court's immediate approval and entry of the Bid Procedures Order substantially in the form and substance reasonably agreed to by the Buyer and Seller, among other things, (A) establishing the Bid Procedures, (B) approving payment of the Termination Fee, to the extent payable by the terms of this Agreement and the Bid Procedures Order, and (C) providing that the Termination Fee shall constitute superpriority administrative expenses of the Seller with priority over any and all administrative expenses pursuant to section 503(b) of the Bankruptcy Code."
- 1.6 Section 9.1(j) of the Agreement is hereby amended and restated in its entirety to read as follows:
- "(j) by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a), and Section 9.3(b)."
- 1.7 The title for Section 9.3 of the Agreement is hereby amended and restated in its entirety to read as follows:

"9.3 Termination Fee"

- 1.8 Section 9.3(b) of the Agreement is hereby amended and restated in its entirety to read as follows:
- "(b) The Seller agrees and acknowledges that Purchaser's due diligence, efforts, negotiation, and execution of this Agreement have involved substantial investment of management time and have required significant commitment of financial, legal, and other resources by Purchaser, and that such due diligence, efforts, negotiation, and execution have provided value to the Seller and, in the Seller's reasonable business judgment, is necessary for the preservation of the value of the Seller's estate. The Seller further agrees and acknowledges that the Termination Fee is not a penalty, but rather represent liquidated damages that are reasonable in relation to Purchaser's efforts, Purchaser's lost opportunities from pursuing the Transactions, and the magnitude of the Transactions. The provision of the Termination Fee is an integral part of this Agreement, without which the Purchaser would not have entered into this Agreement."
- 2. <u>Effect</u>. Except as expressly modified by this Amendment, the Agreement shall remain in full force and effect.

Casse2248800440ssjýj 11 DDocc83621 Ffilitelo044225225 Hintercelo044225225171047595 DDessc MairExhibitrileniPage 144 b10131112

3. <u>Miscellaneous</u>. Sections 10.4 (Severability; Specific Versus General Provisions), Section 10.7 (Counterparts), 10.8 (Governing Law), 10.9 (Submission to Jurisdiction; Consent to Service of Process) and 10.10 (Waiver of Jury Trial) of the Agreement shall apply mutatis mutandis to this Amendment.

[The remainder of this page is intentionally left blank]

Casse2248800409sgijj11 DDocc88621 Filitelc0044225225 Eintereclc0044225225171047595 Diessc MairExhibitritentPage 1128 112

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

PURCHASER:

SENTYNL THERAPEUTICS, INC.

Docusigned by:

Matt teck

Name: Matt Heck

Title: President and CEO

SELLER:

EIGER BIOPHARMACEUTICALS, INC.

By: David Apelian

Name: David Apelian

Title: Chief Executive Officer

Exhibit 2

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825)

William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Proposed Attorneys for the Debtors and Debtors in Possession

SIDLEY AUSTIN LLP

Charles M. Persons (TX Bar No. 24060413)

2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al. 1

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF CLOSING OF ZOKINVY SALE TRANSACTION

On March 31, 2024, the Debtors executed an asset purchase agreement (as amended, modified, or supplemented from time to time, the "Zokinvy Stalking Horse APA")² for the sale of the Transferred Assets. The Debtors attached the executed Zokinvy Stalking Horse APA as Exhibit 2 to the proposed order to the Debtors' Motion for Entry of an Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 13] (the "Motion").

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

² Capitalized terms used but not defined herein have the meanings ascribed to them in the Zokinvy Stalking Horse APA, the Bid Procedures Order, or the Zokinvy Sale Order.

Cassc2248800409sgijj.11 DDocc82842 Filiteck054025225 Hinteccck054025225181147055 Dessc MainElokokutr2enPagle 2ccef 2 of 4

On April 5, 2024, the United States Bankruptcy Court for the Northern District of Texas (the "Court") entered an order granting, in part, the Motion and approving the bid procedures (the "Bid Procedures") annexed as Exhibit 1 to the order [Docket No. 94] (the "Bid Procedures Order"). Attached to the Bid Procedures Order as Exhibit 2 was that certain asset purchase agreement between the Debtors and Sentynl Therapeutics, Inc. On April 19, 2024, the Debtors filed the Notice of Filing of Revised Bidding Procedures [Docket No. 119] (the "Revised Bid Procedures").

On April 22, 2024, the Debtors filed the *Notice of Proposed Amendment to Zokinvy Stalking Horse Asset Purchase Agreement and Proposed Form of Zokinvy Sale Order* [Docket No. 148], which included as <u>Exhibit A</u> the proposed first amendment to the asset purchase agreement (the "<u>Zokinvy Stalking Horse APA Amendment</u>" and the asset purchase agreement as amended by the Zokinvy Stalking Horse APA Amendment, the "Amended Zokinvy Stalking Horse APA").

On April 24, 2024, the Court entered an order [Docket No. 162] (the "Zokinvy Sale Order") authorizing and approving entry into the Amended Zokinvy Stalking Horse APA and the Zokinvy Sale Transaction contemplated thereunder. Attached as <u>Exhibit 1</u> to the Zokinvy Sale Order was a copy of the Amended Zokinvy Stalking Horse APA.

On May 3, 2024, the Closing occurred in accordance with the Amended Zokinvy Stalking Horse APA and the Zokinvy Sale Order. Attached hereto as **Exhibit A** is the Debtors' final list of assumed and assigned contracts pursuant to the Closing.³

Copies of the Amended Zokinvy Stalking Horse APA, as well as all related filings and exhibits, are available by: (i) visiting the website of the Debtors' claims, noticing, and solicitation agent, Kurtzman Carson Consultants LLC at: http://www.kccllc.net/eiger, (ii) (888) 733-1544 (Toll-Free) or (310) 751-2638 (International), and/or (iii) emailing https://kccllc.net/eiger/inquiry, or (iv) for a fee via PACER by visiting http://ecf.txnb.uscourts.gov/.

[Remainder of page intentionally left blank]

_

Pursuant to the Amended Zokinvy Stalking Horse APA, the Base Price was calculated in the amount of \$46.1 million less a credit in the amount of \$0.9 million for the termination fee resulting in a net base price in the amount of \$45.2 million, subject to certain purchase price adjustments, including a reduction of \$100,000 per diem if the sale closed after April 24, 2024.

Dated: May 4, 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Proposed Attorneys for the Debtors and Debtors in Possession

Certificate of Service

I certify that on May 4, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ Thomas R. Califano
Thomas R. Califano

Exhibit A

Final List of Debtors' Designated Contracts

Case 24-80040-sgj11 Doc 238-2 Filed 05/25/25 Entered 05/25/25 13:47:05 Desc Exhibit 2 Page 2 of 2

| Contract Counterparty | Contract Counterparty Address | Description of Contract of Lease | | |
|---|---|--|--|--|
| AnGes, Inc. | 7-7-15, Saito Asagi, Ibaraki, Osaka, 567-0085, Japan | Marketing and Distribution Agreement, dated May 10, 2022, as amended by Side Letter, dated May 10, 2022 and Amendment No. 1, dated May 10, 2023 | | |
| xis Clinicals LLC ioanalytical Systems, Inc. harles River Laboratories linigen Inc. con Clinical Research Limited ttsel Chimos eopharm Ltd. RD International, LLC | зарап | Pharmacovigilance Agreement, dated January 11, 2024 | | |
| | | Quality Agreement, dated February 29, 2024 | | |
| Axis Clinicals LLC | Attn Dinkar Sindhu, 1711 Center Ave West, | Clinical Trial Agreement, dated September 5, 2023 | | |
| | Dilworth, MN 56529 | Clinical Trial Agreement, dated October 25, 2023 | | |
| Bioanalytical Systems, Inc. | Stephanie Miller, Director, Client Services, 2701 Kent Avenue, West Lafayette, IN 47906 | Master Independent Contractor Agreement, dated October 24, 2018, as supplemented by Contractor Task Order, dated June 3, 2020 | | |
| Charles River Laboratories | 251 Ballardvale Street, Wilmington, MA 01887- 1096 | Master Services Agreement, dated July 24, 2019, as supplemented by Statement of Work, dated July 29, 2022 and Statement of Work, dated March 11, 2024 | | |
| | Jerome Charton, Chief Executive Officer, Idis | Master Services Agreement, dated April 26, 2018, as | | |
| Clinigen Inc. | House, Churchfield Road, Weybridge Surrey, | supplemented by Letter Agreement, dated November 24, 2022 | | |
| | KT46 8DB, United Kingdom | Quality Technical Agreement, dated October 4, 2021 | | |
| Frontage Laboratories, Inc. | Dr. Abdul Mutlib, CEO, 700 Pennsylvania Drive, Exton, PA 19341 | Project Proposal, dated December 5, 2022 | | |
| ICON Clinical Research Limited | Kyle McAllister, South County Business Park, Leopardstown, Dublin 18, Ireland | Master Services Agreement, dated August 25, 2022, as supplemented by Statement of Work No. 2, dated November 4, 2022 | | |
| Inteal Chimos | Corinne Truffault, Chief Executive Officer, 1 Rue | Agreement, dated June 8, 2023 | | |
| miser Cinnos | Royale- Batiment D, Saint-Cloud, 92210, France | Quality Agreement, dated June 23, 2024 | | |
| Naanharm I td | Neopharm Building, 6 Hashiloach St., Petach- | Distribution Agreement, dated June 4, 2020 | | |
| reopharii Eta. | Tikva, 4951439, Israel | Quality Agreement, dated June 21, 2022 | | |
| RRD International, LLC | Scott Tarrant, Chief Executive Officer, 7361 Calhoun Place, Suite 510, Rockville, MD 20855 | Master Services Agreement, dated March 15, 2015, as supplemented by Work Order No. 19, dated April 22, 2022, Work Order No. 20, dated April 22, 2022, Change Order Form, dated August 28, 2023 and Change Order Form, dated February 1, 2023 | | |
| | | Confidentiality Agreement, dated March 24, 2016 between Yuki Gosei Kogyo Co Ltd and Eiger Biopharmaceuticals, Inc. | | |
| | | Confidential Disclosure Agreement, dated February 1, 2023 between Yuki Gosei Kogyo Co Ltd, Eiger Biopharmaceuticals, Inc., and AnGes, Inc. | | |
| | | Invoice No. EX-72004, dated March 22, 2024 | | |
| Yuki Gosei Kogyo Co Ltd | Seiichiro Matsumoto, President/CEO/Executive Officer, 10-4, Nihonbashi-Ningyocho 3-Chome, Chuo-Ku, Tokyo, 103-0013, Japan | PAA-MPN Stability Test Plan Price Quotation of Analysis Contract No. 103-366 (formerly No. 103-295), dated November 16, 2022, by and between Eiger Biopharmaceuticals, Inc. and Yuki Gosei Kogyo Co., Ltd., as amended by that First Amendment to Price Quotation of Analysis Contract No. 103-366 (formerly No. 103-295), dated February 15, 2023 | | |
| | | Quality Agreement Supplement, dated September 26, 2023 between Yuki Gosei Kogyo Co Ltd, Eiger Biopharmaceuticals, Inc., and AnGes, Inc. | | |
| | | Technical Quality Agreement, dated January 14, 2022 between Yuki Gosei Kogyo Co Ltd and Eiger Biopharmaceuticals, Inc. | | |

Exhibit 3

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et

al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF CURE AMOUNTS AND
POTENTIAL ASSUMPTION AND ASSIGNMENT
OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES
IN CONNECTION WITH THE REMAINING ASSETS SALE TRANSACTION(S)

PLEASE TAKE NOTICE:

On April 1, 2024 (the "<u>Petition Date</u>"), the debtors and debtors in possession in the above-captioned chapter 11 cases (collectively, the "<u>Debtors</u>") filed voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the Northern District of Texas (the "<u>Court</u>").

On April 1, 2024, the Debtors filed a motion [Docket No. 13] (the "Bid Procedures Motion") with the Court seeking entry of an order, among other things: (i)(a) approving the proposed bid procedures (the "Bid Procedures") in connection with the Sale Transaction(s) of the Assets; (b) approving the Bid Protections relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (c) establishing the dates and deadlines relating to the Remaining Asset Bid Deadline, Auction(s), and a Sale Hearing(s); (d) approving the form and manner of the notice of the Sale Transaction(s), the Bid Deadline(s), the Auction(s), and the Sale Hearing(s) (the "Sale Notice"); (e) approving the assumption and assignment procedures for any executory contracts or unexpired lease to be assumed by the Debtors in connection with the Sale Transaction(s), if any (the "Designated Contracts") (the "Assignment and Assumption Procedures"); and (f) approving the form and manner of the potential assignment and assumption notice (the "Assignment Notice"); and (ii) granting related relief.

On April 5, 2024 the Court entered the Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Ave., Dallas, Texas 75201.

Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order").² All recipients of this notice should carefully read the Bid Procedures Order and the Assumption and Assignment Procedures in their entirety.

You are receiving this notice (the "<u>Cure Notice</u>") because you may be a counterparty to an executory contract or unexpired lease of the Debtors that may be assumed and assigned to the Winning Bidder, or as otherwise contemplated in the Winning Bid related to the Remaining Assets Sale Transaction(s) (each, a "<u>Potentially Assigned Contract</u>"). Each of the Potentially Assigned Contracts that may be assumed and assigned in connection with the Remaining Assets Sale Transaction(s) with a Winning Bidder and the Debtors' calculation of the Cure Amounts with respect thereto are set forth on <u>Exhibit A</u> hereto. The Cure Amounts are the only amounts proposed to be paid upon the assumption and/or assignment of the Potentially Assigned Contracts.

The inclusion of any contract or lease on **Exhibit A** shall not constitute or be deemed a determination or admission by the Debtors that such contract or other document is, in fact, an executory contract or unexpired lease within the meaning of the Bankruptcy Code (all rights with respect thereto being expressly reserved).

Notwithstanding the inclusion of any Potentially Assigned Contract on Exhibit A, neither the Debtors nor any Winning Bidder(s) are obligated to assume or assign any Potentially Assigned Contract identified on Exhibit A. The Cure Notice is only advising parties of the Debtors' proposed Cure Amounts should the Potentially Assigned Contract be assumed or assumed and assigned. Subsequent notice will be provided if your Potentially Assigned Contract is proposed to be assumed or assumed and assigned to a Winning Bidder. If you disagree with the proposed Cure Amounts, you must file an objection in accordance with the instructions below.

Objections

A. Cure Objections

Any objection to potential assumption, assignment, or designation of a Potentially Assigned Contract identified on <u>Exhibit A</u>, the subject of which objection is the Debtors' proposed Cure Amounts, if any, must (i) be in writing; (ii) comply with the Bankruptcy Rules and the Bankruptcy Local Rules; (iii) be filed with the Clerk of the Court, by electronic submission through PACER (Public Access to Court Electronic Records at https://ecf.txnb.uscourts.gov/), or if submitted through non-electronic means, by U.S. Mail or other hand delivery system at the following address: United States Bankruptcy Court, 1100 Commerce St, #5300, Dallas, TX 75242, on or before 4:00 p.m. (CT) on June 12, 2024 (the "Remaining Assets Cure Objection Deadline"); (iv) be served, so as to be actually received on or before the Remaining Assets Cure

² Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Bid Procedures Motion, the Bid Procedures, and the Bid Procedures Order, as applicable.

Objection Deadline, upon the following via email: (a) proposed counsel to the Debtors, Sidley Austin LLP (attn: R. Califano (tom.califano@sidley.com), William E. (wcurtin@sidley.com), and Anne G. Wallice (anne.wallice@sidley.com)), (b) counsel to any official committee of unsecured creditors appointed in the Debtors' chapter 11 cases; (c) the Office of the United States Trustee, 1100 Commerce Street, Room 976, Dallas, Texas 75242 (attn: Elizabeth A. Young (elizabeth.a.young@usdoj.gov)); and (d) counsel to Innovatus, Bradley Arant Boult Cummings Roger Jones (rjones@bradley.com) LLP (attn. and (jbender@bradley.com)(collectively, the "Objection Notice Parties"); and (v) state with specificity the legal and factual grounds for such objection, including, without limitation, the Cure Amounts the Counterparty believes is required to cure defaults under the relevant Executory Contracts or Unexpired Leases. Any objections solely on the basis of adequate assurance of future performance must be filed in the manner described in the foregoing sentence and served on the Objection Notice Parties by no later than 4:00 p.m (CT) on June 12, 2024.

IF A COUNTERPARTY FAILS TO FILE WITH THE BANKRUPTCY COURT AND SERVE ON THE OBJECTION NOTICE PARTIES A TIMELY CURE OBJECTION, THE COUNTERPARTY SHALL BE FOREVER BARRED FROM ASSERTING ANY OBJECTION WITH REGARD TO THE AMOUNT TO CURE ANY DEFAULT UNDER THE APPLICABLE POTENTIALLY ASSIGNED CONTRACT. THE CURE AMOUNTS SET FORTH ON EXHIBIT A HERETO SHALL BE CONTROLLING AND WILL BE THE ONLY AMOUNT NECESSARY TO CURE OUTSTANDING DEFAULTS UNDER THE APPLICABLE POTENTIALLY ASSIGNED CONTRACT UNDER BANKRUPTCY CODE SECTION 365(B), NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THE POTENTIALLY ASSIGNED CONTRACT, OR ANY OTHER DOCUMENT, AND THE APPLICABLE COUNTERPARTY SHALL BE FOREVER BARRED FROM ASSERTING ANY ADDITIONAL CURE OR OTHER AMOUNTS WITH RESPECT TO SUCH POTENTIALLY ASSIGNED CONTRACT AGAINST THE DEBTORS, ANY WINNING BIDDER, OR THE PROPERTY OF ANY OF THEM.

B. Adequate Assurance Objection

As stated, this notice does not indicate if your Potentially Assigned Contract will be assumed, rejected, or assumed and assigned to a Winning Bidder. If your Potentially Assigned Contract is proposed to be assumed or assumed and assigned, subsequent notice will be provided. Any objections to an assignee's adequate assurance of future performance are preserved and the procedures for such adequate assurance objections will be set forth in the subsequent notice.

Cure Objection Hearing

Cure Objections that cannot be resolved by the parties will be heard by the Bankruptcy Court on June 26, 2024 at 1:30 p.m. (prevailing Central Time) or as soon thereafter as such Cure Objection may be heard. You may participate in the Hearing either remotely by an audio and video connection or in person.

<u>Remote Participation Instructions</u>. Audio communication will be by use of the Court's dialin facility. You may access the facility at 1.650.479.3207. Video communication will be by use

Casse2248800409sgijj11 DDooc831833 Filitelc0064025225 Hintereelc0064025225161140740.5 DDess MaiExbidioitushenPagePagePage 55of 5

of the Cisco WebEx platform. Connect via the Cisco WebEx application or click the link on Judge Jernigan's home page. The meeting code is 2304-154-2638. Click the settings icon in the upper right corner and enter your name under the personal information setting.

<u>In Person Instructions</u>. If you wish to attend the hearing in person, the hearing will be conducted at Courtroom 1, floor 14, 1100 Commerce Street, Dallas, TX 75242-1496.

<u>Hearing Appearances</u>. If you intend to appear electronically, hearing appearances must be made electronically in advance of electronic hearings. To make your appearance, click the "Electronic Appearance" link on Judge Jernigan's home page. Select the case name, complete the required fields and click "Submit" to complete your appearance.

[Remainder of page intentionally left blank]

Dated: June 4, 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825)
William E. Curtin (admitted *pro hac vice*)
John J. Kuster (admitted *pro hac vice*)
Jon W. Muenz (admitted *pro hac vice*)
Anne G. Wallice (admitted *pro hac vice*)
787 Seventh Avenue

New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com jkuster@sidley.com jmuenz@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413)

2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Exhibit A

Cure Amounts

| | Related | | | | | Lonafarnib |
|-----|--------------------------|------------------------|---|------------------|------------------|------------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Par y | | | | |
| 1 | Lonafarnib | 3D Communications, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | - | | | |
| 2 | Lonafarnib | 3D Communications, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 3 | Lambda | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | \$0.00 | \$0.00 |
| 4 | Lonafarnib | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | ABF Pharmaceutical | | | | |
| 5 | Lonafarnib | Services | Change Order | \$0.00 | \$0.00 | \$0.00 |
| | | ABF Pharmaceutical | | | | |
| 6 | Lonafarnib | Services GmbH | Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| _ | | ABF Pharmaceutical | | 40.00 | 40.00 | * • • • • |
| 7 | Lonafarnib | Services GmbH | Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| 8 | Lonafarnib | Absorption Systems LP | In Vitro Study Proposal | \$0.00 | \$0.00 | \$0.00 |
| 9 | Lambda | Accenture, LLP | Services Agreement | \$0.00 | \$5,399.00 | \$0.00 |
| 10 | T C 1. | A | Amendment One to the Master Services | 60.00 | \$0.00 | ¢0.00 |
| 10 | Lonafarnib | Accenture, LLP | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 11 | Lonafarnib Lonafarnib | Accenture, LLP | Change Order 2 to SOW 3 Change Order 3 to SOW 3 | \$0.00 | \$0.00 \$0.00 | \$0.00 |
| 13 | Lonafarnib | Accenture, LLP | Change Order Form No. 7 | \$0.00 | \$0.00 | \$0.00 \$0.00 |
| 13 | Lonafarnib | Accenture, LLP | Change Order Form No. 7 Change Order Form No. 9 | \$0.00 \$0.00 | \$0.00 | \$0.00 |
| 14 | Lonatarnio | Accenture, LLP | Change Order No. 003 Amending the | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | | Regulatory and Clinical Services Amended and Restated Statement of Work No. 5 | | | |
| 15 | Lonafarnib | Accenture, LLP | | \$0.00 | \$0.00 | \$0.00 |
| 13 | Lonaranno | Accenture, LLF | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change Order No. 004 Amending the | | | |
| | | | Regulatory and Clinical Services Amended and Restated Statement of Work No. 5 | | | |
| 16 | Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 10 | Lonatarnio | Accenture, LLP | Change Order No. 005 Amending the | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | | Regulatory and Clinical Services Amended and Restated Statement of Work No. 5 | | | |
| 17 | Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 1 / | Lonaranno | Accenture, LLF | Change Order No. 006 Amending the | \$0.00 | \$0.00 | \$0.00 |
| | | | Regulatory and Clinical Services Amended | | | |
| | | | and Restated Statement of Work No. 5 | | | |
| 18 | Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 10 | Lonaramo | recenture, EEr | Change Order No. 008 Amending the | ψ0.00 | Ψ0.00 | Ψ0.00 |
| | | | Regulatory and Clinical Services Amended | | | |
| | | | and Restated Statement of Work No. 5 | | | |
| 19 | Lonafarnib | Accenture, LLP | Dated February 1, 2019. | \$0.00 | \$0.00 | \$0.00 |
| 20 | Lonafarnib | Accenture, LLP | Data Privacy and Security Exhibit | \$0.00 | \$0.00 | \$0.00 |
| 21 | Lonafarnib | Accenture, LLP | Scope of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| 22 | Lonafarnib | Accenture, LLP | Scope of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| 23 | Lonafarnib | Accenture, LLP | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | ADAMAS Consulting | | | | |
| 24 | Lonafarnib | LLC | Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| 25 | Lambda | Addison Whitney | Agreement | \$0.00 | \$0.00 | |
| 26 | Lonafarnib | Addison Whitney | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 27 | Lonafarnib | Addison Whitney | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work describes Services and | İ | | |
| 28 | Lonafarnib | Addison Whitney | Deliverables to be | \$0.00 | \$0.00 | \$0.00 |
| | | Addison Whitney LLC, a | | | | |
| | | Syneos Health Group | Peginterferon Lambda Package & Label | | | |
| 29 | Lambda | Company | Design SOW | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------------|---|----------------|--------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| " | resect | Addison Whitney LLC, a | Description of Contract of Bease | Aveature cure | Lambua Curc | Curc |
| | | Syneos Health Group | Additional Submission Document for | | | |
| 30 | Lonafarnib | Company | Lonafarnib Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| - 30 | Lonararino | Addison Whitney LLC, a | Lonafarnib (Progeria and HDV) Global | \$0.00 | \$0.00 | φ0.00 |
| | | Syneos Health Group | Brand Name Development and Validation | | | |
| 31 | Lonafarnib | Company | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| - 31 | Lonararino | Addison Whitney LLC, a | Lonafarnib for HDV EMA co-pack | ψ0.00 | φ0.00 | ψ0.00 |
| | | Syneos Health Group | regulatory Consulting Validation Statement | | | |
| 32 | Lonafarnib | Company | of Work | \$0.00 | \$0.00 | \$0.00 |
| - 32 | Lonararino | Addison Whitney LLC, a | or work | \$0.00 | \$0.00 | \$0.00 |
| | | Syneos Health Group | | | | |
| 33 | Lonafarnib | Company | Webpage Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| 34 | Lonafarnib | Addison Whitney, LLC | Amendment No. 1 to Statement of Work | \$0.00 | · · | \$0.00 |
| 34 | Lonaranno | Agenzia Italiana del | Amendment No. 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 35 | Lonafarnib | Farmaco | Access Request Agreement | \$0.00 | \$0.00 | \$0.00 |
| - 33 | Lonaranno | Tarmaco | Letter re: Declaration on Public | \$0.00 | \$0.00 | \$0.00 |
| | | A conzio Italiano dal | Contributions and Incentives for | | | |
| 36 | Lonafarnib | Agenzia Italiana del Farmaco | Developmental of the Medical Product | \$0.00 | \$0.00 | \$0.00 |
| 30 | Lonaranno | Agenzia Italiana del | Letter re: Developmental of Medical | \$0.00 | \$0.00 | \$0.00 |
| 37 | Lonafarnib | Farmaco | Product | \$0.00 | \$0.00 | \$0.00 |
| 37 | Lonararnio | AIFA – Agenzia Italiana | Product | \$0.00 | \$0.00 | \$0.00 |
| 38 | Lonafarnib | del Farmaco | A a a a a B a muset A amanus ant | \$0.00 | \$0.00 | \$0.00 |
| 39 | Lonafarnib | ALLCARE | Access Request Agreement | \$0.00 | · · | |
| 39 | Lonararmo | ALLCARE PLUS | Statement of Work dated April 14, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 40 | T C 11. | | Martin Carrier Assessment | ¢0.00 | \$0.00 | ¢0.00 |
| 40 | Lonafarnib | PHARMACY LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 41 | T C 1 | AllCare Plus Pharmacy | Ct. 4 CXV 1 | #0.00 | 00.00 | #0.00 |
| 41 | Lonafarnib | LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 42 | T C 11. | AllCare Plus Pharmacy, | A | ¢0.00 | \$0.00 | ¢0.00 |
| 42 | Lonafarnib | LLC | Amendment | \$0.00 | \$0.00 | \$0.00 |
| 42 | T C 11. | A 16 1 | Statement of Work re: Project No. EGE-P2- | ¢0.00 | \$0.00 | ¢0.00 |
| 43 | Lonafarnib | Altasciences | 672 | \$0.00 | \$0.00 | \$0.00 |
| 4.4 | T C 1 | A 10 | Statement of Work re: Project No. EGE-P3- | 60.00 | 00.00 | #0.00 |
| 44 | Lonafarnib | Altasciences | 019 | \$0.00 | \$0.00 | \$0.00 |
| 4.5 | T C 1 | A 10 | Statement of Work re: Project No. EGE-P9- | 60.00 | 00.00 | #0.00 |
| 45 | Lonafarnib | Altasciences | 232 | \$0.00 | \$0.00 | \$0.00 |
| 4.6 | T C 1 | A 10 | Statement of Work re: Project No. RRI-P9- | 60.00 | 60.00 | #0.00 |
| 46 | Lonafarnib | Altasciences | 693 (EIG-LNF-019) | \$0.00 | \$0.00 | \$0.00 |
| | | Altasciences Clinical | | | | |
| | | Kansas, Inc.; Altasciences | Statement of Week as Spanning Dust and FIG | | | |
| 47 | Lonafarnib | Clinical Kansas, P.A. | Statement of Work re: Sponsor Protocol EIG LNF-022/ Project Number EGE-P9-697 | \$0.00 | \$0.00 | \$0.00 |
| 47 | Lonaranno | ALTASCIENCES | LNF-022/ Floject Nulliber EGE-F9-09/ | \$0.00 | \$0.00 | \$0.00 |
| | | COMPANY CANADA | | | | |
| 48 | Lambda | INC. | Agreement For Bioanalytical Services | \$0.00 | \$0.00 | \$0.00 |
| 40 | Lamoua | ALTASCIENCES | Agreement For Bloanarytical Services | \$0.00 | \$0.00 | \$0.00 |
| | | COMPANY CANADA | | | | |
| 49 | Lonafarnib | INC. | Agreement For Bioanalytical Services | \$0.00 | \$0.00 | \$0.00 |
| 47 | Lonaranno | | Agreement For Bloanarytical Services | \$0.00 | \$0.00 | \$0.00 |
| | | ALTASCIENCES | | | | |
| 50 | Lonafarnib | COMPANY CANADA INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 30 | Lonararmo | ALTASCIENCES | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 51 | Lombda | | Change Order 01 | ¢0.00 | ¢0.00 | ድ ስ ስስ |
| 51 | Lambda | COMPANY INC. | Change Order 01 | \$0.00 | \$0.00 | \$0.00 |
| | | ALTACCIENCES | Statement of Walland Street | | | |
| 50 | I | ALTASCIENCES | Statement of Work re: Sponsor Protocol EIG | | 60.00 | \$0.00 |
| 52 | Lonafarnib | COMPANY INC. | LNF-018 / Project Number RRD-P0-895 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|--|---|----------------|---------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | ALTASCIENCES | Task Order re: Contractor Study Number | | | |
| 53 | Lonafarnib | COMPANY INC. | EGE-W8-303 | \$0.00 | \$0.00 | \$0.00 |
| | | ALTREOS RESEARCH | | | | |
| 54 | Lonafarnib | PARTNERS, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | American Association for | | | | |
| | | the Study of Liver | The Liver Meeting 2020 Exhibit Space | | | |
| 55 | Lonafarnib | Diseases | Application and Contract | \$0.00 | \$0.00 | \$0.00 |
| | | American Mobile Health | 11 | ***** | * | * |
| 56 | Lonafarnib | Services, LLC | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 57 | Lambda | AMRI | Laboratory Service Fee Quotation | \$0.00 | \$0.00 | \$0.00 |
| | Zumo uu | 111111 | Amendment Number 1 to Marketing and | \$0.00 | φοίο σ | φοίου |
| 58 | Lonafarnib | ANGES, INC. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| - 50 | Lonaranio | rittees, rite. | MARKETING & DISTRIBUTION | φ0.00 | ψ0.00 | ψο.σσ |
| 59 | Lonafarnib | ANGES, INC. | AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| 60 | Lonafarnib | ANGES, INC. | Side Letter | \$0.00 | \$0.00 | \$0.00 |
| - 00 | Lonaranio | rivold, rive. | Side Better | φ0.00 | ψ0.00 | ψ0.00 |
| 61 | Lonafarnib | Archbow Consulting, LLC | Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| 01 | Lonaranno | ARUP LABORATORIES | Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| 62 | Lonafarnib | INC. | Amendment 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 02 | Lonaranno | ARUP LABORATORIES | Amendment I to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 63 | Lonafarnib | INC. | Amendment 2 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 03 | Lonaranno | IIVC. | Amendment 2 to statement of work | \$0.00 | \$0.00 | \$0.00 |
| 64 | Avexitide | August Bioservices LLC | Proposal: P-1119 | \$0.00 | \$0.00 | \$0.00 |
| 04 | AVEAILIGE | August Dioservices LLC | 110p05a1.1-1117 | \$0.00 | \$0.00 | \$0.00 |
| 65 | Avexitide | August Bioservices LLC | Proposal: P-1126 | \$0.00 | \$0.00 | \$0.00 |
| 0.5 | AVEAILIGE | August Dioservices LLC | 110p05a1. F-1120 | \$0.00 | \$0.00 | \$0.00 |
| 66 | Avexitide | August Bioservices LLC | Proposal: P-1143 | \$0.00 | \$0.00 | \$0.00 |
| 00 | AVCAILIGE | August Dioservices LLC | 110μ0541.1-11τ3 | \$0.00 | \$0.00 | \$0.00 |
| 67 | Avexitide | August Bioservices LLC | Proposal: P-1147 | \$0.00 | \$0.00 | \$0.00 |
| 07 | AVEAILIGE | August Dioscivices LLC | 1 10p0sai. 1 -11 +/ | φ 0.00 | ş0.00 | \$0.00 |
| 68 | Avexitide | August Bioservices LLC | Proposal: P-1160 | \$0.00 | \$0.00 | \$0.00 |
| 00 | AVEAILIGE | August Bioservices LLC Austin Chemical | Letter Agreement re: Offer Ritonavir USP | \$0.00 | \$0.00 | \$0.00 |
| 69 | Lonafarnib | Company, Inc. | (CAS # 155213-67-5) | \$0.00 | \$0.00 | \$0.00 |
| 09 | Lonaranno | Austin Chemical | Letter Agreement re: Offer Ritonavir USP | \$0.00 | \$0.00 | \$0.00 |
| 70 | Longformil | | (CAS # 155213-67-5) | \$0.00 | ¢ ስ ስስ | ¢ስ ስስ |
| /0 | Lonafarnib | Company, Inc. Avayle CTS, LLC; Fisher | (CAS # 133213-07-3) | \$0.00 | \$0.00 | \$0.00 |
| 71 | Longformil | 1 | Quatation | ¢0.00 | ¢0.00 | ¢0.00 |
| 71 | Lonafarnib | Clinical Services | Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | AVISTA PHARMA | | | | |
| 70 | T1. 1 | SOLUTIONS, INC., | Promond FIGR D22 D221 A | #0.00 | φο oo | ΦΩ ΩΩ |
| 72 | Lambda | D/B/A CAMBREX | Proposal EIGR-D22-D001-A | \$0.00 | \$0.00 | \$0.00 |
| | | | Proposal re: Development and Validation of | | | |
| | | A CAR | a Method for Free PEG in IL-29 PEG- | | | |
| 72 | T 1 1 | Avista Pharma Solutions, | Interferon Lambda-1a Drug Conjugate dated | *** | *^ ^^ | * ^ ^^ |
| 73 | Lambda | Inc., d/b/a Cambrex | May 18, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | | Proposal re: Development and Validation of | | | |
| | | | a Method for Free PEG in IL-29 PEG- | | | |
| | L . | Avista Pharma Solutions, | Interferon Lambda-1a Drug Conjugate dated | | | |
| 74 | Lambda | Inc., d/b/a Cambrex | September 30, 2022 | \$0.00 | \$0.00 | \$0.00 |
| 75 | Lonafarnib | Axis Clinicals LLC | Services Agreement | \$0.00 | \$0.00 | \$675.00 |
| | | | Certificate of Analysis (COA) meeting Eiger | | | |
| 76 | Avexitide | BACHEM | Specifications | \$0.00 | \$0.00 | \$0.00 |
| 77 | Avexitide | BACHEM | Change Order 1 to Quote | \$0.00 | \$0.00 | \$0.00 |
| | | | Change Request re: Exendin (9-39), non- | | | |
| 78 | Avexitide | Bachem | GMP | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 79 | Avexitide | BACHEM | Compilation of Validation Plan of Avexitide | \$0.00 | \$0.00 | \$0.00 |
| | | | 1 | | | |

| | Related | | | | | Lonafarnib |
|-----|------------|---------------------------------------|--|----------------|----------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 80 | Avexitide | BACHEM | Quote | \$0.00 | \$0.00 | \$0.00 |
| 81 | Avexitide | BACHEM | Quote 67815 | \$0.00 | \$0.00 | \$0.00 |
| 82 | Avexitide | BACHEM | Quote 69621 | \$0.00 | | \$0.00 |
| 83 | Avexitide | BACHEM | Quote 70720 | \$0.00 | \$0.00 | \$0.00 |
| 84 | Avexitide | BACHEM | Quote 71209 | \$0.00 | \$0.00 | \$0.00 |
| 85 | Avexitide | BACHEM | Quote 72592 | \$0.00 | \$0.00 | \$0.00 |
| 86 | Avexitide | BACHEM | Quote 73445 | \$0.00 | \$0.00 | \$0.00 |
| 87 | Avexitide | Bachem Americas, Inc | Bachem Quote 20005572 | \$0.00 | \$0.00 | \$0.00 |
| 88 | Avexitide | Bachem Americas, Inc | Bachem Quote 58395 | \$0.00 | | \$0.00 |
| 89 | Avexitide | Bachem Americas, Inc | Bachem Quote 63842 | \$0.00 | \$0.00 | \$0.00 |
| 90 | Avexitide | Bachem Americas, Inc | Bachem Quote 64409 | \$0.00 | \$0.00 | \$0.00 |
| 91 | Avexitide | Bachem Americas, Inc | Bachem Quote 65398 | \$0.00 | \$0.00 | \$0.00 |
| 92 | Avexitide | Bachem Americas, Inc | Bachem Quote 65399 | \$0.00 | \$0.00 | \$0.00 |
| 93 | Avexitide | Bachem Americas, Inc | Bachem Quote 65409 | \$0.00 | \$0.00 | \$0.00 |
| 94 | Avexitide | Bachem Americas, Inc | Bachem Quote 65675 | \$0.00 | \$0.00 | \$0.00 |
| 95 | Avexitide | Bachem Americas, Inc | Bachem Quote 66755 | \$0.00 | \$0.00 | \$0.00 |
| 96 | Avexitide | Bachem Americas, Inc | Bachem Quote 67097 | \$0.00 | \$0.00 | \$0.00 |
| 97 | Avexitide | Bachem Americas, Inc | Quotation 20005572 | \$0.00 | \$0.00 | \$0.00 |
| 98 | Avexitide | Bachem Americas, Inc | Quotation 20005959 | \$0.00 | \$0.00 | \$0.00 |
| 99 | Avexitide | Bachem Americas, Inc | Quotation 55597 | \$0.00 | \$0.00 | \$0.00 |
| 100 | Avexitide | Bachem Americas, Inc | Quotation 57429 | \$0.00 | \$0.00 | \$0.00 |
| 101 | Avexitide | Bachem Americas, Inc | Quotation 57583 | \$0.00 | \$0.00 | \$0.00 |
| 102 | Avexitide | Bachem Americas, Inc | Quotation 63034 | \$0.00 | \$0.00 | \$0.00 |
| 102 | Avexitide | Bachem Americas, Inc | Quotation 63583 | \$0.00 | \$0.00 | \$0.00 |
| 103 | Avexitide | Bachem Americas, Inc | Quotation 65372 | \$0.00 | \$0.00 | \$0.00 |
| 104 | Avexitide | BD Medical, | Quotation 00372 | \$0.00 | \$0.00 | \$0.00 |
| | | Pharmaceutical Systems | | | | |
| 105 | Lambda | US | Letter re: Pricing dated April 27, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 103 | Lamoda | BD Medical, | Letter re. 1 Hering dated April 27, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | Pharmaceutical Systems | | | | |
| 106 | Lambda | US | Letter re: Pricing dated May 13, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 100 | Lamoda | Becton, Dickinson & | Letter re. Trieffig dated Way 15, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 107 | Lambda | Company | Letter re: PS-CPQ-636 - BD | \$0.00 | \$0.00 | \$0.00 |
| 107 | Lamoda | BECTON, DICKINSON | Letter 16. 13-c1 Q-030 - BD | \$0.00 | \$0.00 | \$0.00 |
| 108 | Lambda | AND COMPANY | Ouote re: Pharmaceutical Products | \$0.00 | \$0.00 | \$0.00 |
| 108 | Laiiibua | Bend Research Inc., a | Quote re. Filarmaceutical Froducts | \$0.00 | \$0.00 | \$0.00 |
| 109 | Lonafarnib | Lonza Company | Analytical Dronogal | \$0.00 | \$0.00 | \$0.00 |
| 109 | Lonaranno | 1 7 | Analytical Proposal | \$0.00 | \$0.00 | \$0.00 |
| 110 | Lonafarnib | Bend Research Inc., a | Change Order 7 to Statement of West | \$0.00 | \$0.00 | \$0.00 |
| 110 | Lonararnio | Lonza Company Bend Research Inc., a | Change Order 7 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 111 | Lonafarnib | · · · · · · · · · · · · · · · · · · · | Change Order No. 4 to Statement of Wards | \$0.00 | \$0.00 | \$0.00 |
| 111 | Lonararnio | Lonza Company | Change Order No. 4 to Statement of Work Amendment No. 2 to the Master Services | \$0.00 | \$0.00 | \$0.00 |
| 112 | Lonafarnib | BEND RESEARCH, INC. | and Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| 112 | | | Change Order 3 to the Statement of Work | \$0.00 | \$0.00 | |
| | Lonafarnib | Bend Research, Inc. | Change Order 3 to the Statement of Work Change Order 3 to the Statement of Work | | | \$0.00 |
| 114 | Lonafarnib | Bend Research, Inc. | Change Order 5 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 115 | Lonafarnib | DEND DECEMBELL INC | Change Order 4 Dated June 22, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 113 | Lonatarnio | BEND RESEARCH, INC. | Change Order 4 Dated June 22, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 116 | I | DEND DECEARCH BIC | Change Onder 4 Date 1 Mars 2, 2017 | \$0.00 | φ α Δ Δ | \$0.00 |
| 116 | Lonafarnib | BEND RESEARCH, INC. | Change Order 4 Dated May 2, 2017 | \$0.00 | | \$0.00 |
| 117 | Lonafarnib | Bend Research, Inc. | Change Order 5 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 118 | Lonafarnib | Bend Research, Inc. | Change Order 6 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 110 | I 6 | D 1 D 1 . T | Change Order 6 to the Statement of Work | 00.00 | ΦΛ ΛΛ | #0.00 |
| 119 | Lonafarnib | Bend Research, Inc. | dated September 2, 2016 | \$0.00 | \$0.00 | \$0.00 |
| 120 | I am f | DEND DECEMBELL BIG | Clinical Manufacture Acres | 00.00 | Φ0.00 | φο oo |
| 120 | Lonafarnib | BEND RESEARCH, INC. | Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|--------------------------|-----------------------|--|----------------|----------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | 1 0 | Commercial Manufacturing Services and | | | |
| 121 | Lonafarnib | BEND RESEARCH, INC. | _ | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 122 | Lonafarnib | BEND RESEARCH, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 123 | Lonafarnib | BEND RESEARCH, INC. | Memorandum of Understanding | \$0.00 | \$0.00 | \$0.00 |
| | | | Mutual Nondisclosure Agreement dated | | | |
| 124 | Lonafarnib | BEND RESEARCH, INC. | | \$0.00 | \$0.00 | \$0.00 |
| | | | Mutual Nondisclosure Agreement dated | | _ | |
| 125 | Lonafarnib | BEND RESEARCH, INC. | November 22, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 106 | | DENIE DEGE I DOVE DIG | | 00.00 | 40.00 | |
| 126 | Lonafarnib | BEND RESEARCH, INC. | Proposal dated July 9, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 127 | Lanafamil | DEND DESEARCH INC | Promosel dated October 14, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 127 | Lonafarnib Lonafarnib | Bend Research, Inc. | Proposal dated October 14, 2020 Purchase Order dated April 4, 2017 | \$0.00 | | · |
| | | ′ | Purchase Order dated March 6, 2017 | - | | · · |
| 129 | Lonafarnib | Bend Research, Inc. | | \$0.00 | \$0.00 | \$0.00 |
| 120 | | BIOANALYTICAL | Task Order 02926 to the Master | | 40.00 | 40.00 |
| 130 | Lonafarnib | SYSTEMS, INC. | Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | BIOANALYTICAL | Task Order Q-00122 to the Master | | | |
| 131 | Lonafarnib | SYSTEMS, INC. | Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 132 | Lambda | BIORASI, LLC | Change Order 1 to Statement of Work | \$0.00 | | \$0.00 |
| 133 | Lambda | Biorasi, LLC | Clinical Study Agreement | \$0.00 | | \$0.00 |
| 134 | Lambda | BIORASI, LLC | Letter of Intent | \$0.00 | | \$0.00 |
| 135 | Lambda | BIORASI, LLC | Letter of Intent dated September 29, 20201 | \$0.00 | | \$0.00 |
| | | , | Letter of Intent Extension Agreement dated | | | |
| 136 | Lambda | BIORASI, LLC | December 18, 2020 | \$0.00 | \$181,489.74 | \$0.00 |
| 130 | Lumouu | Brore isi, EEC | Letter of Intent Extension Agreement | φ0.00 | | ψ0.00 |
| 137 | Lambda | BIORASI, LLC | September 29, 2021 | \$0.00 | | \$0.00 |
| 138 | Lambda | BIORASI, LLC | Statement of Work | \$0.00 | | \$0.00 |
| 139 | Lambda | BIORASI, LLC | Statement of Work 3 | \$0.00 | | \$0.00 |
| 140 | Lonafarnib | 100 | | \$0.00 | \$0.00 | |
| | | BIORASI, LLC | Change Order 1 to Statement of Work | | | · |
| 141 | Lonafarnib | BIORASI, LLC | Change Order 2 to Statement of Work | \$0.00 | \$0.00 | · |
| 142 | Lonafarnib | BIORASI, LLC | Change Order 3 to Statement of Work | \$0.00 | \$0.00 | · |
| 143 | Lonafarnib | BIORASI, LLC | Change Order 4 to Statement of Work | \$0.00 | | · |
| 144 | Lonafarnib | BIORASI, LLC | Letter of Intent dated June 13, 2020 | \$0.00 | \$0.00 | · |
| 145 | Lonafarnib | BIORASI, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 146 | Lonafarnib | BIORASI, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | Proposal re: Project Manufacture of Exendin | | | |
| 147 | Avexitide | Bioserv Corporation | 9-39 Injection | \$0.00 | \$0.00 | \$0.00 |
| | | | Proposal re: Project Non Aseptic | | | |
| 148 | Avexitide | Bioserv Corporation | Engineering Run | \$0.00 | \$0.00 | \$0.00 |
| 149 | Avexitide | Bioserv Corporation | Proposal re: Project Vial Labeling | \$0.00 | \$0.00 | \$0.00 |
| | | BRISTOL-MYERS | | | | |
| 150 | Lambda | SQUIBB COMPANY | Assignment and Assumption Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | BRISTOL-MYERS | | | | |
| 151 | Lambda | SQUIBB COMPANY | Common Stock Purchase Agreement | \$0.00 | \$0.00 | \$0.00 |
| | 2 | BRISTOL-MYERS | Common Steem 1 washings 1 ignorman | \$0.00 | \$0.00 | ψ0.00 |
| 152 | Lambda | SQUIBB COMPANY | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| 132 | Lamoua | Bristol-Myers Squibb | Diceise Agreement | \$0.00 | φ υ. 00 | \$0.00 |
| 152 | Longformile | • | Ligansa Agraement | ¢0.00 | ቀለ ለላ | \$0.00 |
| 153 | Lonafarnib | Company | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1 | T 6 " | D 11. | First Amendment to Research Funding | *** | ** ** | |
| 154 | Lonafarnib | Brown University | Agreement | \$0.00 | | · |
| 155 | Lonafarnib | Brown University | RESEARCH FUNDING AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| | | | Second Amendment to Research Funding | | | |
| 156 | Lonafarnib | Brown University | Agreement | \$0.00 | · · | · · |
| | Lonafarnib | Bulgarian Drug Agency | Declaration on Funds Ownership | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|-------------|---------------------------|---|----------------|------------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 158 | Avexitide | CALISYM LLC | Services Agreement | \$2,000.00 | \$0.00 | \$0.00 |
| 159 | Lambda | CALISYM LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| 160 | Lonafarnib | CALISYM LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| 161 | Lonafarnib | CALISYM LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| 162 | Lonafarnib | CALISYM LLC | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| 163 | Lonafarnib | CALISYM LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBER | | | | |
| | | PHARMACEUTICALS | | | | |
| 164 | Lonafarnib | INCORPORATED | Nonbinding Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 165 | Lonafarnib | Cambridge Biomarketing | Project Brief V1 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 166 | Lonafarnib | GROUP, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 167 | Lonafarnib | GROUP, LLC | Master Services Agreement Amendment | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 168 | Lonafarnib | GROUP, LLC | Name Change to MSA | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 169 | Lonafarnib | GROUP, LLC | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 170 | Lonafarnib | GROUP, LLC | Statement of Work 10 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 171 | Lonafarnib | GROUP, LLC | Statement of Work 11 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 172 | Lonafarnib | GROUP, LLC | Statement of Work 12 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 173 | Lonafarnib | GROUP, LLC | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| | | BIOMARKETING | | | | |
| 174 | Lonafarnib | GROUP, LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| 1 | | BIOMARKETING | S | 40.00 | * • • • • | 40.00 |
| 175 | Lonafarnib | GROUP, LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| 176 | I | BIOMARKETING | Statement of Worls 5 | 60.00 | φ 0 .00 | \$0.00 |
| 176 | Lonafarnib | GROUP, LLC | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| 177 | Longfar. 11 | BIOMARKETING | Statement of Wards 6 | \$0.00 | # 0.00 | \$0.00 |
| 177 | Lonafarnib | GROUP, LLC | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE | | | | |
| 178 | Lonoformik | BIOMARKETING | Statement of Work 7 | \$0.00 | \$0.00 | 20.00 |
| 1/8 | Lonafarnib | GROUP, LLC | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| | | CAMBRIDGE BIOMARKETING | | | | |
| 179 | Lonafarnib | GROUP, LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| 1/9 | Lonararino | CAMBRIDGE | Statement of WOIK 6 | \$0.00 | \$0.00 | \$0.00 |
| | | BIOMARKETING | | | | |
| 180 | Lonafarnib | GROUP, LLC | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| 100 | Lonaramio | GROUF, LLC | Statement of WOIK 3 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|---------------|------------------------------|--|----------------|----------------|---|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Canadian Life Science; | | | | |
| | | Patheon, by Thermo | | | | |
| 181 | Lonafarnib | Fisher Scientific | PDS Lab Bill Back Request Form | \$0.00 | \$0.00 | \$0.00 |
| 182 | Lonafarnib | Capsugel | Quote re: 1 Color Ink | \$0.00 | \$0.00 | \$0.00 |
| 183 | Lonafarnib | Capsugel | Quote re: Capsules | \$0.00 | \$0.00 | \$0.00 |
| | | | Third Amendment to Statement of Work No. | | | |
| 184 | Lonafarnib | CAREMARK, LLC | 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Agreement for the Establishment of Legal | | | |
| 185 | Lonafarnib | CATO Europe GmbH | Representative in European Union | \$0.00 | \$0.00 | \$0.00 |
| 186 | Lonafarnib | CATO Europe GmbH | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| | | CATO Europe GmbH; | | | | |
| 187 | Lonafarnib | Clinipace GmbH | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| 188 | Lonafarnib | Cenduit LLC | Proposal for the Provision of IRT Services | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | | | | |
| | | Research in Liver | | | | |
| 189 | Lonafarnib | Diseases, LLC (CLDQ) | Fifth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | Ĭ | | | |
| | | Research in Liver | | | | |
| 190 | Lonafarnib | Diseases, LLC (CLDQ) | First Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | | | | |
| | | Research in Liver | | | | |
| 191 | Lonafarnib | Diseases, LLC (CLDQ) | Second Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | Ţ. | | | |
| | | Research in Liver | | | | |
| 192 | Lonafarnib | Diseases, LLC (CLDQ) | Sixth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | | | | |
| | | Research in Liver | | | | |
| 193 | Lonafarnib | Diseases, LLC (CLDQ) | Third Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Center for Outcomes | | | | |
| | | Research in Liver | | | | |
| | | Diseases, LLC d/b/a | | | | |
| 194 | Lonafarnib | CLDQ | Fourth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | 2nd Amnd to 9002088 in 060622 Statement | | | |
| 195 | Avexitide | LABORATORIES | of Work | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amnd to 9002084 in 060622 Statement of | | | |
| 196 | Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amnd to 9002086 in 060622 Statement of | | | |
| 197 | Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amnd to 9002087 in 060622 Statement of | | | |
| 198 | Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amnd to 9002088 in 060622 Statement of | | | |
| 199 | Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amnd to 9002089 in 060622 Statement of | 40100 | 40100 | 40100 |
| 200 | Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | Charles River | | ***** | ***** | * |
| 201 | Avexitide | Laboratories | Archive Storage Letter dated July 12, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | Charles River | | 40100 | 40100 | 40.00 |
| 202 | Avexitide | Laboratories | Archive Storage Letter dated July 20, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | Charles River | | 40100 | 40100 | 40.00 |
| 203 | Avexitide | Laboratories | Letter re: Payment Authorization | \$0.00 | \$0.00 | \$0.00 |
| 05 | - 1. 0 | CHARLES RIVER | | \$5.00 | ψ0.00 | \$5.00 |
| 204 | Avexitide | LABORATORIES | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 207 | 2 I V CAILIGE | CHARLES RIVER | Master Service Agreement | φυ.υυ | φ0.00 | φ0.00 |
| 205 | Avexitide | LABORATORIES | Statement of Work (CRL-210936) | \$0.00 | \$0.00 | \$0.00 |
| 203 | AVEAITIGE | CHARLES RIVER | Statement of Work (CRL-210730) | φ 0.00 | φ υ. 00 | φ υ. 00 |
| 206 | Avexitide | LABORATORIES | Statement of Work (CRL-224818) | \$0.00 | \$0.00 | \$0.00 |
| 200 | AVEXITIES | LADOKATOKIES | Statement of Work (CRL-224010) | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|------------------------------|--|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | CHARLES RIVER | | | | |
| 207 | Avexitide | LABORATORIES | Statement of Work (SFDC-OPP-272991) | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | , , | | | |
| 208 | Avexitide | LABORATORIES | Statement of Work 20220630 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | 1st Amnd to Statement of Work (SFDC OPP- | | | |
| 209 | Lonafarnib | LABORATORIES | 273814) 20230131 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | 2nd Amnd to Statement of Work (SFDC | | | |
| 210 | Lonafarnib | LABORATORIES | OPP-273814) | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amendment 1 to Statement of Work | | | |
| 211 | Lonafarnib | LABORATORIES | 20190821 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Amendment 2 to Statement of Work | | | |
| 212 | Lonafarnib | LABORATORIES | 20200326 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | | | | |
| 213 | Lonafarnib | LABORATORIES | Statement of Work (CRL-237365) | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | | | | |
| 214 | Lonafarnib | LABORATORIES | Statement of Work (SFDC OPP-273814) | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | | | | |
| 215 | Lonafarnib | LABORATORIES | Statement of Work 20190618 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | | | | |
| 216 | Lonafarnib | LABORATORIES | Statement of Work 20220610 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | | | | |
| | | LABORATORIES | | | | |
| 217 | Lonafarnib | MONTREAL ULC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | ARCHIVE STATEMENT OF WORK | | | |
| 218 | Avexitide | LABORATORIES, INC. | CONTINUED STORAGE | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Statement of Work dated December 31, | | | |
| 219 | Lambda | LABORATORIES, INC. | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Statement of Work dated December 31, | | | |
| 220 | Lambda | LABORATORIES, INC. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Statement of Work re: Evaluation of the | | | |
| 221 | Lambda | LABORATORIES, INC. | Stability of E.coli Working Cell Bank | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work re: Preparation and | | | |
| | | CHARLES RIVER | Characterization of an E. coli Master Cell | | | |
| 222 | Lambda | LABORATORIES, INC. | Bank | \$0.00 | \$0.00 | \$0.00 |
| | | CHARLES RIVER | Statement of Work re: Storage of Materials | | | |
| 223 | Lambda | LABORATORIES, INC. | Under Controlled Conditions and Access | \$0.00 | | \$0.00 |
| 224 | Lonafarnib | CHEMTREC | Quote re: Emergency Response Service | \$0.00 | \$0.00 | \$0.00 |
| | | Children's Hospital | | | | |
| | | Corporation dba Boston | Letter re: Confirmation of Regulatory | | | |
| 225 | Lonafarnib | Children's Hospital | Compliance Responsibilities | \$0.00 | \$0.00 | \$0.00 |
| 223 | Lonaranno | CITRINE MEDICINE, | Compilance Responsionities | \$0.00 | \$0.00 | \$0.00 |
| 226 | Lonafarnib | LLC | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| 220 | Lonaranno | LLC | Direct Study Participant Recruitment | \$0.00 | \$0.00 | \$0.00 |
| 227 | Lonafarnib | ClinicalConnection, Inc | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 227 | Lonaramio | Chinical Confidention, the | rgreement | Ψ0.00 | ψ0.00 | Ψ0.00 |
| | | | Single Patient Consent Form for | | | |
| 228 | Lonafarnib | Clinigen | Participation in a Managed Access Program | \$0.00 | \$0.00 | \$0.00 |
| | Zenwianie | omingen. | Amendment No. 1 to Statement of Work No. | φοίου | φοίο σ | φοίου |
| 229 | Lonafarnib | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | Amendment No. 2 to Statement of Work No. | 40.00 | 40100 | 40.00 |
| 230 | Lonafarnib | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | Amendment No. 3 to Statement of Work No. | | * - * * | |
| 231 | Lonafarnib | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | | - | Amendment No. 4 to Statement of Work No. | | | |
| 232 | Lonafarnib | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | • | | | |

| | Related | | | | | Lonafarnib |
|------|------------|----------------------------|---|----------------|--|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | CLINIGEN | | | | |
| 233 | Lonafarnib | HEALTHCARE LTD | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 234 | Lonafarnib | Clinigen Healthcare Ltd. | Manufacturer Statement of Delegation | \$0.00 | \$0.00 | \$0.00 |
| 235 | Lonafarnib | Clinigen Inc. | pursuant to the Master Services Agreement | \$0.00 | \$0.00 | \$79,677.66 |
| | Lonararino | changen me. | Change Order 3 to Work Order C-14 for | ψο.σσ | ψ0.00 | Ψ13,011.00 |
| 236 | Lambda | CLINIPACE, INC. | Eiger | \$0.00 | \$0.00 | \$0.00 |
| | | | Clinical Research | | | |
| | | | | | | |
| | | | Change Order 1 to Work Order C-14 For | | | |
| 237 | Lambda | CLINIPACE, INC. | Eiger | \$0.00 | \$0.00 | \$0.00 |
| | | | Clinical Research | | | |
| | | | | | | |
| | | | Change Order 1 to Work Order C-7 ("CO1") | | | |
| 238 | Lambda | CLINIPACE, INC. | For Eiger | \$0.00 | \$0.00 | \$0.00 |
| 239 | Lambda | Clinipace, Inc. | Work Order C-11 | \$0.00 | \$0.00 | \$0.00 |
| 240 | Lambda | Clinipace, Inc. | Work Order C-12 | \$0.00 | \$0.00 | \$0.00 |
| 241 | Lambda | Clinipace, Inc. | Work Order C-14 | \$0.00 | \$0.00 | \$0.00 |
| 242 | Lambda | Clinipace, Inc. | Work Order C-7 | \$0.00 | \$0.00 | \$0.00 |
| 243 | Lambda | Clinipace, Inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| 244 | Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-1 | \$0.00 | \$0.00 | \$0.00 |
| 245 | Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-4 | \$0.00 | \$0.00 | \$0.00 |
| 246 | Lonafarnib | CLINIPACE, INC. | CO2 to Work Order C-1 | \$0.00 | \$0.00 | \$0.00 |
| 247 | Lonafarnib | Clinipace, Inc. | Work Order C-10 | \$0.00 | \$0.00 | \$0.00 |
| 248 | Lonafarnib | Clinipace, Inc. | Work Order C-10 | \$0.00 | \$0.00 | \$0.00 |
| 249 | Lonafarnib | Clinipace, Inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| 279 | Lonaranno | Cimpace, inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| 250 | Lambda | Clinipace, Inc. dba Caidya | CO2 to Work Order C-7 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 251 | Lonafarnib | Corden Pharma Colorado | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 252 | Lonafarnib | Corden Pharma Colorado | Change Order #6 dated May 19, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 2.52 | | | GI 0 1 46 1 137 5 2010 | 40.00 | * • • • • • • • • • • • • • • • • • • • | 40.00 |
| 253 | Lonafarnib | Corden Pharma Colorado | Change Order #6 dated Nov 5, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 254 | r 6 1 | | GI O I GOA GOO GGOWAHA | 00.00 | 00.00 | # 0.00 |
| 254 | Lonafarnib | Corden Pharma Colorado | Change Order CO4 to CO2 of SOW #2 | \$0.00 | \$0.00 | \$0.00 |
| 255 | T C 11 | CORDEN PHARMA | Channel On Land to Statement of Wash | 60.00 | 60.00 | ¢0.00 |
| 255 | Lonafarnib | COLORADO INC. | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 256 | r 6 1 | CORDEN PHARMA | Change Order 2 to Change Order 1 to | 00.00 | 00.00 | # 0.00 |
| 256 | Lonafarnib | COLORADO INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 255 | T C | CORDEN PHARMA | Change Outline 2 to City of City of City | *^ ^^ | * 0.00 | 00.00 |
| 257 | Lonafarnib | COLORADO INC. | Change Order 3 to Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| 2.50 | | CORDEN PHARMA | a | | | 40.00 |
| 258 | Lonafarnib | COLORADO INC. | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | CORDEN PHARMA | | | | |
| 259 | Lonafarnib | COLORADO INC. | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | | CORDEN PHARMA | | | | |
| 260 | Lonafarnib | COLORADO INC. | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| | | CORDEN PHARMA | | | | |
| 261 | Lonafarnib | COLORADO INC. | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| | | CORDEN PHARMA | | | | |
| 262 | Lonafarnib | COLORADO INC. | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | Corden Pharma Colorado; | | | | |
| | | Corden Pharma | | | | |
| 263 | Lonafarnib | International GmbH | Change Order #5 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|--------------|--------------------------|--|----------------|---------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | P | | | |
| | | Corden Pharma Colorado; | | | | |
| | | Corden Pharma | | | | |
| 264 | Lonafarnib | International GmbH | Change Order 1 to the Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | Corden Pharma Colorado; | | | | |
| | | Corden Pharma | | | | |
| 265 | Lonafarnib | International GmbH | Change Order 5 to the Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | Corden Pharma Colorado; | | | | |
| | | Corden Pharma | | | | |
| 266 | Lonafarnib | International GmbH | Coden Pharma Quote | \$0.00 | \$0.00 | \$0.00 |
| | | Corden Pharma | | | | |
| 267 | Avexitide | International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| | | Corden Pharma | | | | |
| 268 | Lonafarnib | International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| | | CordenPharma; Corden | | | | |
| | | Pharma International | | | | |
| 269 | Lonafarnib | GmbH | Coden Pharma Quotation | \$0.00 | | \$0.00 |
| 270 | Lonafarnib | COREALIS PHARMA | Amendment 1 to Service Agreement - H | \$0.00 | \$0.00 | \$0.00 |
| 271 | Lonafarnib | COREALIS PHARMA | Service Agreement - A | \$0.00 | \$0.00 | \$0.00 |
| 272 | Lonafarnib | COREALIS PHARMA | Service Agreement - B | \$0.00 | \$0.00 | \$0.00 |
| 273 | Lonafarnib | COREALIS PHARMA | Service Agreement - J | \$0.00 | · · | |
| 274 | Lonafarnib | COREALIS PHARMA | Service Agreement - N | \$0.00 | \$0.00 | \$0.00 |
| 275 | Lonafarnib | COREALIS PHARMA | Service Agreement - O | \$0.00 | \$0.00 | \$0.00 |
| 276 | Avexitide | Curia New Jersey, LLC | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 277 | | | Laboratory Service Fee Quotation dated | # 0.00 | 40.00 | 40.00 |
| 277 | Lambda | Curia New Jersey, LLC | 3/26/2022 | \$0.00 | \$0.00 | \$0.00 |
| 270 | | | Laboratory Service Fee Quotation dated | # 0.00 | 40.00 | 40.00 |
| 278 | Lambda | Curia New Jersey, LLC | 5/31/2022 | \$0.00 | \$0.00 | \$0.00 |
| 270 | | | Laboratory Service Fee Quotation dated | # 0.00 | 40.00 | 40.00 |
| 279 | Lambda | Curia New Jersey, LLC | 7/27/2022 | \$0.00 | \$0.00 | \$0.00 |
| 200 | 7 1 1 | C : N I IIC | Laboratory Service Fee Quotation dated | ¢0.00 | \$0.00 | #0.00 |
| 280 | Lambda | Curia New Jersey, LLC | 9/30/2022 | \$0.00 | \$0.00 | \$0.00 |
| 201 | | G : N I IIG | Laboratory Service Fee Quotation dated | #0.00 | # 0.00 | # 0.00 |
| 281 | Lambda | Curia New Jersey, LLC | June 2022 | \$0.00 | \$0.00 | \$0.00 |
| 202 | T C 1. | C | Proposal for Analysis of Active Metabolite | 60.00 | ¢0.00 | ¢0.00 |
| 282 | Lonafarnib | Cyprotex US, LLC | of Lonafarnib (LNF): HM21 | \$0.00 | \$0.00 | \$0.00 |
| | | | Donor and for Annalysis of Anti-en Matalastica | | | |
| 283 | T ama£amaila | Commenters LIC LLC | Proposal for Analysis of Active Metabolites of Lonafarnib (LNF): MH17 and HM21 | \$0.00 | \$0.00 | 00.00 |
| 283 | Lonafarnib | Cyprotex US, LLC | Proposal for CytoChrome P450 Time | \$0.00 | \$0.00 | \$0.00 |
| 284 | T ama£amaila | Commenters LIC LLC | Dependent Inhibition of Lonafarnib (LNF) | \$0.00 | \$0.00 | 00.00 |
| 284 | Lonafarnib | Cyprotex US, LLC | Proposal for MATE Transporter Inhibition | \$0.00 | \$0.00 | \$0.00 |
| 285 | Lonafarnib | Cymnotox US IIC | Testing for Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 |
| 203 | Lonaranno | Cyprotex US, LLC | Proposal for OCT1 and BSEP Inhibition and | \$0.00 | \$0.00 | \$0.00 |
| | | | OCT1 Substrate Testing for Lonafarnib | | | |
| 286 | Lonafarnib | Cyprotex US, LLC | (LNF) | \$0.00 | \$0.00 | \$0.00 |
| 200 | Lonaranno | D. S., PHD [Name on | (LNF) | \$0.00 | \$0.00 | \$0.00 |
| 287 | Avexitide | File] | Consulting Agreement | \$12.750.00 | \$0.00 | \$0.00 |
| 207 | Avexilide | D. S., PHD [Name on | Consulting Agreement | \$12,750.00 | \$0.00 | \$0.00 |
| 288 | Lonafarnib | File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 200 | Lonaranno | DEFENSE HEALTH | Consulting Agreement Department of Defense (DoD) Retail Refund | | \$0.00 | \$0.00 |
| 289 | Lonafarnib | AGENCY | Pricing Agreement | \$0.00 | \$0.00 | \$0.00 |
| 209 | Lonarariiio | Department of Health and | i nemg agreement | \$0.00 | \$0.00 | \$0.00 |
| 290 | Lonafarnib | Human Services | National Drug Rebate Agreement | \$0.00 | \$0.00 | \$0.00 |
| 290 | Lonaraniio | Trainian Belvices | ranonai Diug Kevate Agreement | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|--|--|----------------|---------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | DEPARTMENT OF | P. C. | | | |
| 291 | Lonafarnib | VETERANS AFFAIRS | 1st Amnd to BPA 36E79722A0010 | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | | | | |
| 292 | Lonafarnib | VETERANS AFFAIRS | Amendment | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | | | | |
| 293 | Lonafarnib | VETERANS AFFAIRS | Amendment 2 to 36F79722D0048 | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | | | | |
| 294 | Lonafarnib | VETERANS AFFAIRS | Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | Amendment of Solicitation/Modification of | | | |
| 295 | Lonafarnib | VETERANS AFFAIRS | Contract | \$0.00 | \$0.00 | \$0.00 |
| 206 | | DEPARTMENT OF | Blanket Purchase Agreement dated Jan 1, | 40.00 | 40.00 | 40.00 |
| 296 | Lonafarnib | VETERANS AFFAIRS | 2022 | \$0.00 | \$0.00 | \$0.00 |
| 207 | T C 11. | DEPARTMENT OF | Blanket Purchase Agreement dated Oct 10, | 60.00 | 60.00 | ¢0.00 |
| 297 | Lonafarnib | VETERANS AFFAIRS | 2021 | \$0.00 | \$0.00 | \$0.00 |
| 298 | Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | BPA Cancellation | \$0.00 | \$0.00 | \$0.00 |
| 298 | Lonatarnio | DEPARTMENT OF | BPA Cancellation | \$0.00 | \$0.00 | \$0.00 |
| 299 | Lonafarnib | VETERANS AFFAIRS | Commercial Items | \$0.00 | \$0.00 | \$0.00 |
| 299 | Lonaranno | Department of Veterans | Commercial Items | \$0.00 | \$0.00 | \$0.00 |
| 300 | Lonafarnib | Affairs | D0048 Red A | \$0.00 | \$0.00 | \$0.00 |
| 500 | Eonaranio | Department of Veterans | 200 10 100 11 | ψ0.00 | φ0.00 | φ0.00 |
| 301 | Lonafarnib | Affairs | Facility Code | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | | | | |
| 302 | Lonafarnib | VETERANS AFFAIRS | FSS NFAMP Smoothing Letter | \$0.00 | \$0.00 | \$0.00 |
| | | | I B R8, Amendment 0004 | | | |
| | | | | | | |
| | | DEPARTMENT OF | Amendment of Solicitation/Modification of | | | |
| 303 | Lonafarnib | VETERANS AFFAIRS | Contract | \$0.00 | \$0.00 | \$0.00 |
| | | DEPARTMENT OF | Master Agreements and Pharmaceutical | | | |
| 304 | Lonafarnib | VETERANS AFFAIRS | Pricing Agreement | \$0.00 | \$0.00 | \$0.00 |
| 205 | | DEPARTMENT OF | | 40.00 | # 0.00 | 40.00 |
| 305 | Lonafarnib | VETERANS AFFAIRS | Small Business Subcontracting Plan | \$0.00 | \$0.00 | \$0.00 |
| 206 | T1. 1. | DILISYM SERVICES, | Master Services Agreement dated June 14, | 60.00 | 60.00 | ¢0.00 |
| 306 | Lambda | INC. DILISYM SERVICES, | 2018 | \$0.00 | \$0.00 | \$0.00 |
| 307 | Lonafarnib | INC. | Master Services Agreement May 22, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 307 | Lonaranno | INC. | iviaster Services Agreement iviay 22, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Dom zdravlja Medigroup | | | | |
| 308 | Lonafarnib | dr Ristic Pariske | Donation Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Dr. T. L. M. [Name on | | | | |
| 309 | Avexitide | File] | Proposal to Purchase Assets | \$2,981.00 | \$0.00 | \$0.00 |
| 310 | Lambda | DSG, Inc. | Quotation re: Ongoing Project Management | \$0.00 | \$0.00 | \$0.00 |
| 311 | Lonafarnib | ELEMENT | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | Element Materials | | | | |
| 312 | Lonafarnib | Technology Canada Inc. | Element Quote | \$0.00 | \$0.00 | \$0.00 |
| | | Element Met : 1 | | | | |
| 313 | Lonafarnib | Element Materials Technology Canada Inc. | Floment Quete detect April 7, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 313 | Lonaratino | Element Materials Testing | Element Quote dated April 7, 2020 Quote re: Robustness Study for SOP 20-48- | \$0.00 | \$0.00 | \$0.00 |
| 314 | Lonafarnib | Canada Inc. | SP-2396 | \$0.00 | \$0.00 | \$0.00 |
| 315 | Avexitide | Eurofins | Change Order dated February 14, 2022 | \$0.00 | | |
| 316 | Avexitide | Eurofins | Change Order November 3, 2021 | \$0.00 | | \$0.00 |
| 317 | Lambda | Eurofins | Quote dated December 31, 2017 | \$0.00 | • | • |
| 318 | Lambda | Eurofins | Quote dated December 4, 2020 | \$0.00 | | · |
| | 1 | 1 | , , | | | |

| | Related | | | | | Lonafarnib |
|-----|------------|---------------------------|---|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| π- | Asset | Contract Counterparty | Description of Contract of Lease | Avexitue Cure | Lambua Cure | Cure |
| | | EUROFINS ADVANTAR | | | | |
| 319 | Avexitide | LABORATORIES, INC. | Services Agreement | | \$0.00 | \$0.00 |
| 517 | | El Esta II stands, il (el | Services rigitations | | ψ0.00 | ψ0.00 |
| | | EUROFINS ADVANTAR | | \$25,600.00 | | |
| 320 | Avexitide | LABORATORIES, INC. | AGREEMENT APPROVAL 5 | \$23,000.00 | \$0.00 | \$0.00 |
| | | El Esta II stands, il (el | | | \$0.00 | ψ0.00 |
| | | EUROFINS ADVANTAR | | | | |
| 321 | Avexitide | LABORATORIES, INC. | AGREEMENT APPROVAL 7 | | \$0.00 | \$0.00 |
| | | , | | | · | · |
| | | EUROFINS ADVANTAR | | | | |
| 322 | Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 3, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | , | , | | | |
| | | EUROFINS ADVANTAR | | | | |
| 323 | Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 4, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | EUROFINS ADVANTAR | | | | |
| 324 | Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 5, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins Biopharma | Quote re: Free PEG by HPLC-CAD Method | | | |
| 325 | Lambda | Product Testing | Transfer | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins BioPharma | Quote re: Lambda-1 (Python) FBS | | | |
| 326 | Lambda | Product Testing | Qualification | \$0.00 | \$0.00 | \$0.00 |
| 327 | Avexitide | Eurofins BTP | Quote dated 1/16/2023 | \$0.00 | \$0.00 | \$0.00 |
| 328 | Avexitide | Eurofins BTP | Quote dated 10/30/2020 | \$0.00 | \$0.00 | \$0.00 |
| 329 | Avexitide | Eurofins BTP | Quote dated 2/11/2021 | \$0.00 | \$0.00 | \$0.00 |
| 330 | Avexitide | Eurofins BTP | Quote dated 2/11/2021 | \$0.00 | \$0.00 | \$0.00 |
| 331 | Avexitide | Eurofins BTP | Quote dated 5/5/2021 | \$0.00 | \$0.00 | \$0.00 |
| 332 | Avexitide | Eurofins BTP | Quote dated 7/6/2021 | \$0.00 | \$0.00 | \$0.00 |
| 333 | Avexitide | Eurofins BTP | Quote dated 8/4/2023 | \$0.00 | \$0.00 | \$0.00 |
| 334 | Lambda | Eurofins BTP | Quote dated June 7, 2022 | \$0.00 | \$0.00 | \$0.00 |
| 335 | Lambda | Eurofins BTP | Quote dated October 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins Lancaster | | | | |
| | | Laboratories (Lancaster | | | | |
| 336 | Avexitide | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins Lancaster | | | | |
| | | Laboratories (Lancaster | | | | |
| 337 | Avexitide | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins Lancaster | | | | |
| | | Laboratories (Lancaster | | | | |
| 338 | Lambda | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | Eurofins Lancaster | | | | |
| | | Laboratories (Lancaster | | | | |
| 339 | Lambda | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | EVERSANA LIFE | | | | |
| | | SCIENCE SERVICES, | | | | |
| 340 | Lonafarnib | LLC | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | EXCELLIS EUROPE | | | | |
| 341 | Lonafarnib | LTD. | Work Order No. One | \$0.00 | \$0.00 | \$0.00 |
| 342 | Lonafarnib | F. M., D. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$2,612.50 |
| | | FISHAWACK | | | | |
| 343 | Lonafarnib | FACILITATE LIMITED | Eiger Advisory Board Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | FISHER BIOSERVICES, | | | | |
| 344 | Lambda | INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 345 | Lonafarnib | GmbH | Quote 214873 Order 3 Version 4 20190513 | \$0.00 | \$0.00 | \$0.00 |

| A | | Related | | | | | Lonafarnib |
|--|------|------------|------------------------------|---|-----------------------|---------------|--------------|
| | # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| A | | | Fisher Clinical Services | | | | |
| | 346 | Lonafarnib | GmbH | Quote 214873 Order 7 Version 2 20210318 | \$0.00 | \$0.00 | \$0.00 |
| A | | | Fisher Clinical Services | | | | |
| | 347 | Lonafarnib | | Quote 214873 Order 8 Version 3 20220225 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | | |
| | 348 | Lonafarnib | | Quote 215866 Order 5 Version 1 20210331 | \$0.00 | \$0.00 | \$0.00 |
| Innafamib | 2.40 | T | | O | \$0.00 | \$0.00 | \$0.00 |
| | 349 | Lonatarnio | | ~ | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 214873 Order 4 Version 1 So.00 So.0 | 350 | Lonafarnih | | | \$0.00 | \$0.00 | \$0.00 |
| | 330 | Lonaranno | | | \$0.00 | \$0.00 | φ0.00 |
| Fisher Clinical Services Quote FCS 214873 Order 5 V1 20200218 \$0.00 \$0 | 351 | Lonafarnib | | - | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 214873 Order 5 V2 20200305 \$0.00 \$0 | | Zenwianie | | 20130700 | \$0.00 | φ0.00 | φοίου |
| Avexitide Cambid | 352 | Lonafarnib | GmbH | Quote FCS 214873 Order 5 V1 20200218 | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 214873 Order 6 Version 2 20201229 \$0.00 \$ | | | Fisher Clinical Services | | | | |
| | 353 | Lonafarnib | GmbH | Quote FCS 214873 Order 5 V2 20200305 | \$0.00 | \$0.00 | \$0.00 |
| Signature Fisher Clinical Services Quote FCS 215866 Order 2 Version 1 So.00 | | | Fisher Clinical Services | Quote FCS 214873 Order 6 Version 2 | | | |
| Social Constant | 354 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 215866 Order 2 Version 3 S0.00 S0.00 S0.00 S0.00 | | | | | | | |
| Social Constantib | 355 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 215866 Order 2 Version 4 Quote FCS 215866 Order 3 Version 1 Quote FCS 215866 Order 3 Version 1 Quote FCS 215866 Order 3 Version 1 Quote FCS 215866 Order 4 Version 2 Quote FCS 215866 Order 5 Version 2 Quote FCS 215866 Or | 256 | r 6 1 | | 1 7 | 00.00 | # 0.00 | #0.00 |
| Social Confidence Confidenc | 356 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| Signature Fisher Clinical Services GmbH 20191003 S0.00 S0. | 257 | Lonoformib | | 7 | 00.00 | \$0.00 | 00.00 |
| Somb Combined Co | 337 | Lonaranno | | | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services GmbH 20210121 S0.00 | 358 | Lonafarnih | | 7 | \$0.00 | \$0.00 | \$0.00 |
| Social Constantib | 336 | Lonaranno | | | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Quote FCS 215866 Order 4 Version 2 S0.00 S0.00 S0.00 S0.00 | 359 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| Social Constantible | | | | | | 4 | 40.00 |
| 361 Lonafarnib GmbH 20210705 \$0.00 \$0.00 \$0.00 362 Avexitide Inc. Quote PSG-A-1085163.v2 20231106 \$1,808.02 \$0.00 \$0.00 363 Avexitide Inc. Quote PSG-A-1081156.v1 20230807 \$0.00 \$0.00 364 Avexitide Inc. Quote-PSG-A-1081156.v1 20230807 \$0.00 \$0.00 365 Lambda Inc. CO 1 to PSG-A-1073971 (PSG-A-1076893) 20230524 \$0.00 \$0.00 366 Lambda Inc. Fisher Clinical Services Fisher Clinical Services Fisher Clinical Services Quote PSG-A-1081156.v1 20230807 \$0.00 \$0.00 367 Lambda Inc. Fisher Clinical Services Fisher Clinical Services Quote PSG-A-1081156.v1 20230807 \$0.00 \$0.00 368 Lambda Inc. Fisher Clinical Services Quote PSG-A-1081156.v1 20230807 \$0.00 \$0.00 369 Lambda Inc. Quote 20160927 \$0.00 \$0.00 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 370 Lambda Inc. Quote FCS 55799 Change Order 1 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 373 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 373 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 374 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 375 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 376 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 377 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 378 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 377 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 378 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 378 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 379 Lambda I | 360 | Lonafarnib | GmbH | 1 7 | \$0.00 | \$0.00 | \$0.00 |
| Fisher Clinical Services Inc. Services Agreement S0.00 \$0.00 | | | Fisher Clinical Services | Quote FCS 215866 Order 5 Version 2 | | | |
| 362 Avexitide Inc. | 361 | Lonafarnib | | 20210705 | \$0.00 | \$0.00 | \$0.00 |
| Signature | | | Fisher Clinical Services | | | | |
| 363 Avexitide Inc. Quote PSG-A-1085163.v2 20231106 \$1,808.02 \$0.00 \$0.00 364 Avexitide Inc. Quote-PSG-A-1081156.v1 20230807 \$0.00 \$0.00 365 Lambda Inc. 20230524 \$0.00 \$0.00 366 Lambda Inc. Fisher Clinical Services Inc. Fisher Clinical Quote 20160517 \$0.00 \$0.00 367 Lambda Inc. Fisher Clinical Services Fisher Clinical Services Indantal Services Inda | 362 | Avexitide | | Services Agreement | | \$0.00 | \$0.00 |
| Avexitide Inc. Quote-PSG-A-1081156.v1 20230807 \$0.00 \$0.00 | | | _ | | \$1,808.02 | | |
| 364 Avexitide Inc. Quote-PSG-A-1081156.v1 20230807 \$0.00 \$0.00 365 Lambda Inc. 20230524 \$0.00 \$0.00 366 Lambda Inc. Fisher Clinical Services Fisher Clinical Quote 20160517 \$0.00 \$0.00 367 Lambda Inc. Fisher Clinical Services Quote PSG-A- \$0.00 \$0.00 368 Lambda Inc. Quote 20160927 \$0.00 \$0.00 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 370 Lambda Inc. 20161116 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services Fisher Clinical Services \$0.00 \$0.00 \$0.00 | 363 | Avexitide | | Quote PSG-A-1085163.v2 20231106 | 1 | \$0.00 | \$0.00 |
| Fisher Clinical Services CO 1 to PSG-A-1073971 (PSG-A-1076893) \$0.00 \$0.00 | 264 | A | | O | | \$0.00 | \$0.00 |
| Social Services Sisher Clinical Quote 20160517 Social Services Social Services Sisher Clinical Services Social Services Soci | 304 | Avexitide | | | | \$0.00 | \$0.00 |
| Fisher Clinical Services Fisher Clinical Quote 20160517 \$0.00 \$0.00 | 365 | L ambda | _ | · · · · · · · · · · · · · · · · · · · | | | 20.00 |
| 366 Lambda Inc. Fisher Clinical Quote 20160517 \$0.00 \$0.00 367 Lambda Inc. 1043137.v1 20210812 \$0.00 \$0.00 368 Lambda Inc. Quote 20160927 \$0.00 \$0.00 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 370 Lambda Inc. 20161116 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 | 303 | Lumouu | | 20230321 | \$0.00 | | \$0.00 |
| Fisher Clinical Services Fisher Clinical Services Quote PSG-A- | 366 | Lambda | | Fisher Clinical Quote 20160517 | \$0.00 | | \$0.00 |
| Fisher Clinical Services Quote 20160927 \$0.00 \$0.00 | | | Fisher Clinical Services | Fisher Clinical Services Quote PSG-A- | | | • |
| 368 Lambda Inc. Quote 20160927 \$0.00 \$0.00 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 370 Lambda Inc. Quote FCS 55799 Change Order 1 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services Fisher Clinical Services \$0.00 \$0.00 | 367 | Lambda | Inc. | 1043137.v1 20210812 | \$0.00 | | \$0.00 |
| Fisher Clinical Services Quote FCS 55799 20170221 \$0.00 \$0.00 | | | Fisher Clinical Services | | | | |
| 369 Lambda Inc. Quote FCS 55799 20170221 \$0.00 \$0.00 370 Lambda Inc. Quote FCS 55799 Change Order 1 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services Fisher Clinical Services \$0.00 \$0.00 | 368 | Lambda | Inc. | Quote 20160927 | \$0.00 | | \$0.00 |
| Fisher Clinical Services Quote FCS 55799 Change Order 1 | | | Fisher Clinical Services | | | | |
| 370 Lambda Inc. 20161116 \$0.00 \$0.00 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services Fisher Clinical Services \$0.00 \$0.00 | 369 | Lambda | | | \$0.00 | | \$0.00 |
| Fisher Clinical Services Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 | 2=2 | , , , | | | ** | | ** * * * |
| 371 Lambda Inc. Quote FCS 55799 Order 7 20170629 \$0.00 \$0.00 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services Fisher Clinical Services \$0.00 \$0.00 | 370 | Lambda | | 20161116 | \$0.00 | | \$0.00 |
| Fisher Clinical Services Inc. Quote FCS 55799 Order 8 20180918 \$0.00 \$0.00 Fisher Clinical Services | 271 | I amb 1- | | Oueta ECS 55700 Onder 7 20170620 | \$0.00 | | ¢0.00 |
| 372 Lambda Inc. Quote FCS 55799 Order 8 20180918 \$0.00 Fisher Clinical Services \$0.00 | 3/1 | Lamoda | | Quote FCS 33/99 Order / 201/0029 | \$0.00 | | \$0.00 |
| Fisher Clinical Services | 372 | Lambda | _ | Ouote FCS 55799 Order 8 20180918 | \$0.00 | | \$0.00 |
| | 312 | Lamoda | | Quote 1 05 55777 Order 6 20100710 | φ0.00 | | ψ0.00 |
| | 373 | Lambda | | Quote FCS 58040 20161206 | \$0.00 | | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|-------------------------------|------------------------------------|----------------|---------------|--------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Fisher Clinical Services | | | | |
| 374 | Lambda | Inc. | Quote FCS 62278 20180309 | \$0.00 | | \$0.00 |
| 275 | T1. 1. | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | 60.00 | | \$0.00 |
| 375 | Lambda | Inc. Fisher Clinical Services | 20190720 | \$0.00 | | \$0.00 |
| 376 | Lambda | Inc. | Quote PSG-A-1007253 V2 20190508 | \$0.00 | | \$0.00 |
| 370 | Lamoda | Fisher Clinical Services | Quote 13G-A-100/255 V2 20190508 | \$0.00 | | \$0.00 |
| 377 | Lambda | Inc. | Quote PSG-A-1009306.V1 20190619 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | ,,,,, | | * |
| 378 | Lambda | Inc. | Quote PSG-A-1037571.v1 20210414 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 379 | Lambda | Inc. | Quote PSG-A-1037572.v1_20210420 | \$0.00 | | \$0.00 |
| 200 | T1. 1. | Fisher Clinical Services | O | 00.00 | \$36,483.85 | Φ0.00 |
| 380 | Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1037587.v1 20210420 | \$0.00 | . , | \$0.00 |
| 381 | Lambda | Inc. | Quote PSG-A-1038183.v1 20210422 | \$0.00 | | \$0.00 |
| 301 | Lamoda | Fisher Clinical Services | Quote 15G-A-1030103.V1 20210422 | \$0.00 | | \$0.00 |
| 382 | Lambda | Inc. | Quote PSG-A-1040820.V3 20210713 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 383 | Lambda | Inc. | Quote PSG-A-1041275.v4 20210819 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 384 | Lambda | Inc. | Quote PSG-A-1044658.v1 20210902 | \$0.00 | | \$0.00 |
| 205 | , , , | Fisher Clinical Services | D | 00.00 | | #0.00 |
| 385 | Lambda | Inc. | Quote PSG-A-1045926.v1 20210927 | \$0.00 | | \$0.00 |
| 386 | Lambda | Fisher Clinical Services Inc. | Ouote PSG-A-1053812.v2 20220323 | \$0.00 | | \$0.00 |
| 360 | Lambda | Fisher Clinical Services | Quote 13G-A-1033812.V2 20220323 | \$0.00 | | \$0.00 |
| 387 | Lambda | Inc. | Quote PSG-A-1056697.v1 20221205 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | 44111 | | 40100 |
| 388 | Lambda | Inc. | Quote PSG-A-1060127.v1 20221205 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 389 | Lambda | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | | \$0.00 |
| 200 | T 1 1 | Fisher Clinical Services | O + PGG + 1072071 20220420 | #0.00 | | Φ0.00 |
| 390 | Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1073971 20230429 | \$0.00 | | \$0.00 |
| 391 | Lambda | Inc. | Ouote PSG-A-1086699.v1 20231128 | \$0.00 | | \$0.00 |
| 391 | Lamoda | Fisher Clinical Services | Quote 13G-A-1080077.V1 20231128 | \$0.00 | | \$0.00 |
| 392 | Lambda | Inc. | Quote re: Protocol No. EIG-LMD-001 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 393 | Lambda | Inc. | Quote-PSG-A-1037570.v2 20210423 | \$0.00 | | \$0.00 |
| | | Fisher Clinical Services | | | | |
| 394 | Lambda | Inc. | Quote-PSG-A-1045938.v1 20210927 | \$0.00 | | \$0.00 |
| 205 | , , , | Fisher Clinical Services | Dag 4 1060005 1 20220204 | | | |
| 395 | Lambda | Inc. | Quote-PSG-A-1069905.v1 20230204 | \$0.00 | | \$0.00 |
| 396 | Lonafarnib | Fisher Clinical Services Inc. | Letter of Intent 20220517 | \$0.00 | \$0.00 | |
| 390 | Lonaranno | Fisher Clinical Services | Letter of intent 20220317 | \$0.00 | \$0.00 | |
| 397 | Lonafarnib | Inc. | Quote FCS 63502 20180613 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | \$3.00 | \$0.00 | |
| 398 | Lonafarnib | Inc. | Quote FCS 64018 20180705 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 399 | Lonafarnib | Inc. | Quote FCS 64045 20180629 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | Quote FCS 64045 Order 3 Version 2 | | | |
| 400 | Lonafarnib | Inc. | 20180803 | \$0.00 | \$0.00 | |
| 401 | I | Fisher Clinical Services | Quote FCS 64045 Order 4 Version 2 | 60.00 | # 0.00 | |
| 401 | Lonafarnib | Inc. | 20190110 | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|-----|------------|-------------------------------|--|----------------|------------------|-------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Fisher Clinical Services | | | | |
| 402 | Lonafarnib | Inc. | Quote FCS 64624 20180821 | \$0.00 | \$0.00 | |
| 402 | r c 1 | Fisher Clinical Services | Quote FCS 64624 Order 1 Version 2 | 00.00 | # 0.00 | |
| 403 | Lonafarnib | Inc. Fisher Clinical Services | 20190128 | \$0.00 | \$0.00 | |
| 404 | Lonafarnib | Inc. | Quote FCS 65165 20180919 | \$0.00 | \$0.00 | |
| 404 | Lonaranno | Fisher Clinical Services | Quote FCS 03103 20180319 | \$0.00 | \$0.00 | |
| 405 | Lonafarnib | Inc. | Quote FCS 65165 v2 20181002 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | (| - | 40100 | |
| 406 | Lonafarnib | Inc. | Quote FCS 65165 v3 20181023 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 407 | Lonafarnib | Inc. | Quote FCS 66521 20190128 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | | | |
| 408 | Lonafarnib | Inc. | 20190720 | \$0.00 | \$0.00 | |
| 400 | | Fisher Clinical Services | DOG 1 1001000 XXI 20100210 | 00.00 | * • • • • | |
| 409 | Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1004098.V1 20190219 | \$0.00 | \$0.00 | |
| 410 | Lonafarnib | Inc. | Ouote PSG-A-1004100.V1 20190219 | \$0.00 | \$0.00 | |
| 410 | Lonaranno | Fisher Clinical Services | Quote 13G-A-1004100. V1 20190219 | \$0.00 | \$0.00 | |
| 411 | Lonafarnib | Inc. | Quote PSG-A-1004911.V1 20190311 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | Quete 12 0 11 100 () 11 1 1 2 0 1 7 0 0 11 | \$0.00 | \$0.00 | |
| 412 | Lonafarnib | Inc. | Quote PSG-A-1004912.V1 20190311 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 413 | Lonafarnib | Inc. | Quote PSG-A-1006224 20190403 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 414 | Lonafarnib | Inc. | Quote PSG-A-1006225 20190404 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | ** | |
| 415 | Lonafarnib | Inc. | Quote PSG-A-1006225.V2 20190610 | \$0.00 | \$0.00 | |
| 416 | Lonafarnib | Fisher Clinical Services Inc. | Ouote PSG-A-1009322.V1 20190627 | \$0.00 | \$0.00 | |
| 410 | Lonaranno | Fisher Clinical Services | Quote FSG-A-1009322.V1 20190027 | \$0.00 | \$0.00 | |
| 417 | Lonafarnib | Inc. | Quote PSG-A-1011449.V4 20190910 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | 7.111 | 40100 | |
| 418 | Lonafarnib | Inc. | Quote PSG-A-1011991 20190827 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 419 | Lonafarnib | Inc. | Quote PSG-A-1011991.V2 20190909 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 420 | Lonafarnib | Inc. | Quote PSG-A-1019194.v1 20200311 | \$0.00 | \$0.00 | |
| 421 | T C 11. | Fisher Clinical Services | O | 60.00 | ¢0.00 | \$11,961.26 |
| 421 | Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1020030.v2 20200406 | \$0.00 | \$0.00 | |
| 422 | Lonafarnib | Inc. | Quote PSG-A-1028287.v3 20201124 | \$0.00 | \$0.00 | |
| 122 | Lonaramio | Fisher Clinical Services | Quote 150 11 1020207.13 20201121 | ψ0.00 | ψ0.00 | |
| 423 | Lonafarnib | Inc. | Quote PSG-A-1034823.v2 20210303 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 424 | Lonafarnib | Inc. | Quote PSG-A-1035048.v2 20210308 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 425 | Lonafarnib | Inc. | Quote PSG-A-1035056.v1 20210224 | \$0.00 | \$0.00 | |
| 126 | T C 1 | Fisher Clinical Services | O + PGG + 1035555 1 20210200 | 40.00 | # 0.00 | |
| 426 | Lonafarnib | Inc. | Quote PSG-A-1035575.v1 20210308 | \$0.00 | \$0.00 | |
| 427 | Lonafarnib | Fisher Clinical Services Inc. | Quote PSG-A-1036207.v2 20210722 | \$0.00 | \$0.00 | |
| 74/ | Lonaranno | Fisher Clinical Services | Quote 150-A-1030207.V2 20210722 | φυ.00 | φ 0. 00 | |
| 428 | Lonafarnib | Inc. | Quote PSG-A-1038320.v1 20210421 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | \$3.00 | \$0.00 | |
| 429 | Lonafarnib | Inc. | Quote PSG-A-1042064.v1 20210709 | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|-------|------------|---------------------------------------|---|----------------|---|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Fisher Clinical Services | | | | |
| 430 | Lonafarnib | Inc. | Quote PSG-A-1043391.v1 20210806 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 431 | Lonafarnib | Inc. | Quote PSG-A-1045137.v1 20210913 | \$0.00 | \$0.00 | |
| 432 | Lonafarnib | Fisher Clinical Services Inc. | Ouote PSG-A-1048146.v2 20220106 | \$0.00 | \$0.00 | |
| 432 | Lonaranno | Fisher Clinical Services | Quote FSG-A-1048140.V2 20220100 | \$0.00 | \$0.00 | |
| 433 | Lonafarnib | Inc. | Ouote PSG-A-1051190.v1 20220114 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | 4 | |
| 434 | Lonafarnib | Inc. | Quote PSG-A-1051277.v3 20220225 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 435 | Lonafarnib | Inc. | Quote PSG-A-1051408.v1 20220118 | \$0.00 | \$0.00 | |
| 12.6 | | Fisher Clinical Services | Dag + 1055000 1 2000000 | 00.00 | # 0.00 | |
| 436 | Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1055223.v1 20220330 | \$0.00 | \$0.00 | |
| 437 | Lonafarnib | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | \$0.00 | |
| 737 | Lonaranno | Fisher Clinical Services | Quote 1 5G-A-10/2043 V1 20230420 | \$0.00 | \$0.00 | |
| 438 | Lonafarnib | Inc. | Ouote PSG-A-1075062.v1 20230414 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | * | |
| 439 | Lonafarnib | Inc. | Quote PSG-A-1077951.v1 20230608 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 440 | Lonafarnib | Inc. | Quote PSG-A-1080106.v2 20230814 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 441 | Lonafarnib | Inc. | Quote-PSG-A-1024294.v1 20200707 | \$0.00 | \$0.00 | |
| 442 | Lonafarnib | Fisher Clinical Services Inc. | Quote-PSG-A-1027328.v1 20200915 | \$0.00 | \$0.00 | |
| 442 | Lonatarnio | Fisher Clinical Services | Quote-PSG-A-102/328.V1 20200913 | \$0.00 | \$0.00 | |
| 443 | Lonafarnib | Inc. | Quote-PSG-A-1037585.v1 20210412 | \$0.00 | \$0.00 | |
| - 1.0 | | Fisher Clinical Services | | 4 | 4 | |
| 444 | Lonafarnib | Inc. | Quote-PSG-A-1058344.v1 20220710 | \$0.00 | \$0.00 | |
| | | Fisher Clinical Services | | | | |
| 445 | Lonafarnib | Inc. | Quote-PSG-A-1058998.v1 20230120 | \$0.00 | \$0.00 | |
| 116 | T C 11 | Fisher Clinical Services | D | 00.00 | # 0.00 | |
| 446 | Lonafarnib | Inc. | Quote-PSG-A-1072091.v2 20230302 | \$0.00 | \$0.00 | |
| 447 | Lonafarnib | Fisher Clinical Services U.K. Limited | Ouote PSG-A-1007765.V1 20190514 | \$0.00 | \$0.00 | \$0.00 |
| 44/ | Lonaranno | Frontage Laboratories, | Quote 1 50-A-1007/05. V1 20190514 | \$0.00 | \$0.00 | \$0.00 |
| 448 | Avexitide | Inc. | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| | | Frontage Laboratories, | | | * | * |
| 449 | Avexitide | Inc. | Proposal 1005-30837-112.1 20210308 | \$0.00 | \$0.00 | \$0.00 |
| | | Frontage Laboratories, | | | | |
| 450 | Avexitide | Inc. | Proposal 1005-31914-111.0 2021323 | \$0.00 | \$0.00 | \$0.00 |
| 45. | | Frontage Laboratories, | D 11005 22072 122 2 22212722 | 00.00 | ** ** | ** ** |
| 451 | Avexitide | Inc. | Proposal 1005-32862-122.0 20210520 | \$0.00 | \$0.00 | \$0.00 |
| 452 | Avexitide | Frontage Laboratories, | Proposal 1005 40000 122 0 20220902 | \$0.00 | \$0.00 | \$0.00 |
| 432 | Avexiliae | Inc. Frontage Laboratories, | Proposal 1005-40909-122.0 20230802 | \$0.00 | \$0.00 | \$0.00 |
| 453 | Lonafarnib | Inc. | Proposal 1005.38448.400 20220830 | \$0.00 | \$0.00 | \$0.00 |
| | | | Stability Studies Termination, | \$3.00 | \$5.00 | \$0.00 |
| 454 | Lambda | Fujifilm Diosynth | Accountability and Reconciliation Memo | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| | | BIOCTECHNOLOGIES | 2020 ("SOW9") and the Master | | | |
| 455 | Lambda | USA, INC. | Bioprocessing Services Agreement | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| 150 | T am 1. 1. | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | 40.00 | | # 0.00 |
| 456 | Lambda | USA, INC. | January 15, 2019 | \$0.00 | | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|--------------------------------|--|----------------|-------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | FUJIFILM DIOSYNTH | | | | |
| | | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | | | |
| 457 | Lambda | USA, INC. | July 15, 2019 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | 1 | ***** | 1 | ****** |
| | | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | | | |
| 458 | Lambda | USA, INC. | October 15, 2019 | \$0.00 | | \$0.00 |
| | 2 | FUJIFILM DIOSYNTH | 2010001 10, 2019 | ψ0.00 | | Ψ0.00 |
| | | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | | | |
| 459 | Lambda | USA, INC. | October 5, 2016 | \$0.00 | | \$0.00 |
| 137 | Lumouu | FUJIFILM DIOSYNTH | 3, 2010 | ψ0.00 | | ψ0.00 |
| | | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | | | |
| 460 | Lambda | USA, INC. | September 22, 2016 | \$0.00 | | \$0.00 |
| 100 | Lamota | FUJIFILM DIOSYNTH | September 22, 2010 | ψ0.00 | | ψ0.00 |
| | | BIOCTECHNOLOGIES | Bioprocessing Services Agreement dated | | | |
| 461 | Lambda | USA, INC. | September 22, 2016 | \$0.00 | | \$0.00 |
| 401 | Lamoda | FUJIFILM DIOSYNTH | September 22, 2010 | \$0.00 | 1 | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 462 | Lambda | USA, INC. | Change Order 2 | \$0.00 | | \$0.00 |
| 702 | Lamoua | FUJIFILM DIOSYNTH | Change Order 2 | \$0.00 | 1 | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 463 | Lambda | USA, INC. | Change Order 3 | \$0.00 | | \$0.00 |
| 403 | Lamoda | FUJIFILM DIOSYNTH | Change Order 5 | \$0.00 | ł | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 464 | Lambda | USA, INC. | Change Order 4 | \$0.00 | | \$0.00 |
| 404 | Lamoua | FUJIFILM DIOSYNTH | Change Order 4 | \$0.00 | | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 465 | Lambda | USA, INC. | Change Order 5 to Statement of Work 2 | \$0.00 | | \$0.00 |
| 403 | Lamoua | FUJIFILM DIOSYNTH | Change Order 5 to Statement of Work 2 | \$0.00 | | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 466 | Lambda | USA, INC. | Letter Agreement dated April 22, 2021 | \$0.00 | | \$0.00 |
| 400 | Lamoua | FUJIFILM DIOSYNTH | Letter Agreement dated April 22, 2021 | \$0.00 | | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 467 | Lambda | USA, INC. | Letter Agreement dated June 04, 2021 | \$0.00 | | \$0.00 |
| 407 | Lamoua | FUJIFILM DIOSYNTH | Letter Agreement dated Julie 04, 2021 | \$0.00 | \$25,000.00 | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 468 | Lambda | USA, INC. | Master Services Agreement | \$0.00 | | \$0.00 |
| 408 | Lamoda | FUJIFILM DIOSYNTH | Master Services Agreement Scope of Work #2 (Manufacturing) | \$0.00 | | \$0.00 |
| | | BIOCTECHNOLOGIES | ("SOW2") and the Master Bioprocessing | | | |
| 469 | Lambda | USA, INC. | Services Agreement | \$0.00 | | \$0.00 |
| 409 | Lamoda | FUJIFILM DIOSYNTH | Services Agreement | \$0.00 | | \$0.00 |
| | | | | | | |
| 470 | T amala da | BIOCTECHNOLOGIES | Statement of West 11 | \$0.00 | | \$0.00 |
| 470 | Lambda | USA, INC. FUJIFILM DIOSYNTH | Statement of Work 11 | \$0.00 | | \$0.00 |
| | | | | | | |
| 471 | T 1 . 1 . | BIOCTECHNOLOGIES | State | 60.00 | | \$0.00 |
| 471 | Lambda | USA, INC. | Statement of Work 14 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| 472 | I amb Ja | BIOCTECHNOLOGIES | Statement of Worls 2 | ¢0.00 | | ¢0.00 |
| 472 | Lambda | USA, INC. | Statement of Work 2 | \$0.00 | ļ | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| 472 | Lorel 1. | BIOCTECHNOLOGIES | Statement of Warls 2 to Change Out of | # 0.00 | | \$0.00 |
| 473 | Lambda | USA, INC. | Statement of Work 2 to Change Order 1 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| 474 | I amil 1: | BIOCTECHNOLOGIES | Statement of World 2 to Change Out 2 | \$0.00 | | \$0.00 |
| 474 | Lambda | USA, INC. | Statement of Work 2 to Change Order 3 | \$0.00 | | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|-----------|---|---|-----------------|-------------|---|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| " | 713300 | FUJIFILM DIOSYNTH | Description of Contract of Lease | Tivesitiae care | Zambaa care | Cure |
| | | BIOCTECHNOLOGIES | | | | |
| 475 | Lambda | USA, INC. | Statement of Work 3 | \$0.00 | | \$0.00 |
| | Zumo uu | FUJIFILM DIOSYNTH | | \$0.00 | 1 | \$0.00 |
| | | BIOCTECHNOLOGIES | | | | |
| 476 | Lambda | USA, INC. | Statement of Work 5 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | 4 | 1 | 4 |
| | | BIOCTECHNOLOGIES | | | | |
| 477 | Lambda | USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| | † | FUJIFILM DIOSYNTH | | , | 1 | **** |
| | | BIOCTECHNOLOGIES | | | | |
| 478 | Lambda | USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| | | BIOCTECHNOLOGIES | | | | |
| 479 | Lambda | USA, INC. | Statement of Work 8 | \$0.00 | | \$0.00 |
| | | FUJIFILM DIOSYNTH | | | | |
| | | BIOCTECHNOLOGIES | | | | |
| 480 | Lambda | USA, INC. | Statement of Work 9 | \$0.00 | | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| 481 | Avexitide | Biotechnologies (FDB) | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| 482 | Lambda | Biotechnologies (FDB) | Letter re: Lamba Program on Hold | \$0.00 | \$0.00 | \$0.00 |
| | 1 | FUJIFILM Diosynth | | | | |
| | | Biotechnologies U.S.A., | | | | |
| 483 | Lambda | Inc. | Change Order 2 to Python SOW 17 | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| | | Biotechnologies U.S.A., | Change Order 5 re: MCB and WCB | | | |
| 484 | Lambda | Inc. | Bioassay Characterization | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | Change Order 6 re: Establishment of | | | |
| | | Biotechnologies U.S.A., | Degraded SEC and Degraded Issi-Asp, CEX | | | |
| 485 | Lambda | Inc. | & RP Purity Assay Controls | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| | | Biotechnologies U.S.A., | Change Order 6 re: Positional Isomer | | | |
| 486 | Lambda | Inc. | Feasibility | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| | | Biotechnologies U.S.A., | | | | |
| 487 | Lambda | Inc. | Change Order 9 re:Update | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| | | Biotechnologies U.S.A., | | | | |
| 488 | Lambda | Inc. | Purchase Order No. 5134 | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| 400 | | Biotechnologies U.S.A., | | | | |
| 489 | Lambda | Inc. | Scope of Work | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| 400 | r 1 1 | Biotechnologies U.S.A., | G 637 1 112 | #0.00 | ¢0.00 | ¢0.00 |
| 490 | Lambda | Inc. | Scope of Work #13 | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth | | | | |
| 401 | I omk de | Biotechnologies U.S.A., | Same of Work #17 | \$0.00 | ¢0.00 | \$0.00 |
| 491 | Lambda | Inc. | Scope of Work #17 | \$0.00 | \$0.00 | \$0.00 |
| | | FUJIFILM Diosynth Biotechnologies U.S.A., | | | | |
| 492 | Lambda | = | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| 492 | Lambua | Inc. FUJIFILM Diosynth | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| | | Biotechnologies U.S.A., | | | | |
| 493 | Lambda | Inc. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| 773 | Lamoua | 1110. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|------------------------------|--|-----------------------|----------------|---|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | FUJIFILM Diosynth | 1 | | | |
| | | Biotechnologies U.S.A., | | | | |
| 494 | Lambda | Inc. | Scope of Work 23 | \$0.00 | \$0.00 | \$0.00 |
| .,, | Zumo uu | Gemini Staffing | Exhibit A to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| 495 | Lonafarnib | Consultants, Inc. | Contractors | \$0.00 | \$0.00 | \$0.00 |
| 496 | Lonafarnib | General Synco Inc. | Quotation | \$0.00 | \$0.00 | · · |
| 497 | Lambda | GENSCRIPT USA INC | Quote U532XIC280 | \$0.00 | \$0.00 | \$0.00 |
| 498 | Lambda | GENSCRIPT USA INC | Ouote U7278IC280 | \$0.00 | \$0.00 | \$0.00 |
| 499 | Lambda | Gilero, LLC | Proposal re: Covid PFS EUA | \$0.00 | \$0.00 | \$0.00 |
| 122 | Lamoda | Glicio, EEC | Proposal re: Lambda Interferon DHF Gap | Ψ0.00 | ψ0.00 | ψ0.00 |
| 500 | Lambda | Gilero, LLC | Analysis | \$0.00 | \$0.00 | \$0.00 |
| 300 | Lamoda | Glicio, EEC | Proposal re: Lambda Interferon DV&V | Ψ0.00 | \$0.00 | Φ0.00 |
| 501 | Lambda | Gilero, LLC | Execution | \$0.00 | \$0.00 | \$0.00 |
| 502 | Lonafarnib | GKV Spitzenverband | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| 503 | Lonafarnib | GLSYNTHESIS, INC. | Certificate of Analysis | \$0.00 | \$0.00 | \$0.00 |
| 303 | Lonaranno | · · | Certificate of Analysis | \$0.00 | \$0.00 | \$0.00 |
| | | GND GEIGER NITZ | | | | |
| | | DAUNDERER | | | | |
| 504 | T | RECHTSANWALTE | T 1 A 1 A | ¢0.00 | \$0.00 | ¢0.00 |
| 504 | Lonafarnib | PARTG MBB | Legal Advisory Agreement | \$0.00 | \$0.00 | \$0.00 |
| 505 | r 1 1 | GOETHE-UNIVERSITY | T CT . | #0.00 | #0.00 | Φ0.00 |
| 505 | Lambda | FRANKFURT | Letter of Intent | \$0.00 | - | \$0.00 |
| 506 | Lonafarnib | H. B. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$962.50 |
| | L | Halloran Consulting | | | | |
| 507 | Lambda | Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 508 | Lambda | Group, Inc. | SOW #5 – Inspection Readiness Assessment | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 509 | Lonafarnib | Group, Inc. | Amendment #1 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 510 | Lonafarnib | Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 511 | Lonafarnib | Group, Inc. | Amendment #1 to Statement of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 512 | Lonafarnib | Group, Inc. | Amendment #2 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | EMA Site Inspection Preparation and | | | |
| 513 | Lonafarnib | Group, Inc. | Support – (SOW #4) | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 514 | Lonafarnib | Group, Inc. | Site Inspection Readiness – (SOW #3) | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 515 | Lonafarnib | Group, Inc. | SOW #1 | \$0.00 | \$0.00 | \$0.00 |
| | | Halloran Consulting | | | | |
| 516 | Lonafarnib | Group, Inc. | Statement of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| 517 | Lonafarnib | HD Biosciences | Study Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | | Amendment #1 to HDB-EigerBio-01- | | | |
| 518 | Lonafarnib | HD Biosciences, Inc. | 05282019 | \$0.00 | \$0.00 | \$0.00 |
| 519 | Lonafarnib | HD Biosciences, Inc. | Study Quotation | \$0.00 | | \$0.00 |
| | | HEALTH | , , | ***** | ***** | * |
| | | CORPORATION OF | | | | |
| | | GALILEE MEDICAL | | | | |
| 520 | Lonafarnib | CENTER | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| 320 | | Healthcare at Home | | ψ0.00 | ψ0.00 | ψ0.00 |
| 521 | Lonafarnib | (Netherlands) B.V. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| J41 | Lonaranno | HEALTHCARE AT | Distribution Agreement | \$0.00 | φ υ. 00 | φυ.00 |
| | | HOME | | | | |
| 522 | Lonoformil | | Distribution Agreement | ድስ ስስ | ድ ስ ስስ | ድስ ስስ |
| 522 | Lonafarnib | (NETHERLANDS) B.V. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|------------------------------|--|----------------|---|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | HEALTHCARE AT | • | | | |
| | | HOME | | | | |
| 523 | Lonafarnib | (NETHERLANDS) B.V. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 524 | Lonafarnib | HH Consultancy | Quote 1203202a: Statement of Work | \$0.00 | \$0.00 | \$2,066.00 |
| | | Icon Clinical Research | | | | · |
| 525 | Avexitide | Limited | Consulting Services Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | ICON CLINICAL | | | | |
| 526 | Lonafarnib | RESEARCH LIMITED | Start-up Statement of Work No. 1 | \$0.00 | \$0.00 | \$0.00 |
| | | ICON CLINICAL | 1 | | | |
| 527 | Lonafarnib | RESEARCH LIMITED | Start-up Statement of Work No. 2 | \$0.00 | \$0.00 | \$0.00 |
| | | Icon Clinical Research | 1 | | | |
| 528 | Lonafarnib | Limited | Statement of Work EIG22720 | \$0.00 | \$0.00 | \$0.00 |
| | | Icon Clinical Research | | | | |
| 529 | Lonafarnib | Limited | Statement of Work EIG22721 | \$0.00 | \$0.00 | \$0.00 |
| 530 | Lonafarnib | IDEA REGULATORY | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| 531 | Lambda | Idem Translations, Inc. | Quotation | \$0.00 | \$0.00 | \$0.00 |
| | 1 | Inova Health Care | | 1 | | |
| 532 | Lonafarnib | Services | Statement of Work 20181101 | \$0.00 | \$0.00 | \$0.00 |
| | 1 | Inova Health Care | - | 1 | ,,,,, | 7 |
| 533 | Lonafarnib | Services | Statement of Work 20220930 | \$0.00 | \$0.00 | \$0.00 |
| | | INTEGRATED | | | | · |
| | | MEDICAL | Statement of work #1 to Eiger-IMD Master | | | |
| 534 | Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |
| | | INTEGRATED | , , , | , , , , , | * | * |
| | | MEDICAL | Statement of work #3 to Eiger-IMD Master | | | |
| 535 | Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |
| | | INTERNATIONAL | , , , | , , , , , | * | **** |
| | | TOXICOLOGY | | | | |
| | | RESEARCH | | | | |
| | | LABORATORIES | | | | |
| 536 | Avexitide | CANADA, INC. | Statement of Work 42827 | \$0.00 | \$0.00 | \$0.00 |
| | | INTERNATIONAL | | | | |
| | | TOXICOLOGY | | | | |
| | | RESEARCH | | | | |
| | | LABORATORIES | | | | |
| 537 | Avexitide | CANADA, INC. | Statement of Work 42828 | \$0.00 | \$0.00 | \$0.00 |
| | † | INTERNATIONAL | | +==== | 42.00 | Ţ 3 V |
| | | TOXICOLOGY | | 1 | | |
| | | RESEARCH | | 1 | | |
| | | LABORATORIES | | | | |
| 538 | Avexitide | CANADA, INC. | Statement of Work 74705 | \$0.00 | \$0.00 | \$0.00 |
| | | Intrado Digital Media | | 72.00 | 42.00 | 7 0 0 |
| 539 | Lambda | LLC | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| 540 | Avexitide | INTRINSIK CORP | Statement of Work 11 | \$5.00 | \$0.00 | · |
| 541 | Avexitide | INTRINSIK CORP | Statement of Work 13 | 1 | \$0.00 | \$0.00 |
| 542 | Avexitide | INTRINSIK CORP | Statement of Work 15 | \$2,595.00 | | |
| 543 | Avexitide | INTRINSIK CORP | Statement of Work 2 | Ψ2,373.00 | \$0.00 | · |
| 544 | Avexitide | INTRINSIK CORP | Statement of Work Avx | 1 | \$0.00 | * |
| 545 | Lambda | INTRINSIK CORP | Statement of Work 10 | \$0.00 | | |
| 546 | Lambda | INTRINSIK CORP | Statement of Work 3 | \$0.00 | | |
| 547 | Lambda | INTRINSIK CORP | Statement of Work 5 | \$0.00 | · | |
| 548 | Lambda | INTRINSIK CORP | Statement of Work 6 | \$0.00 | | |
| 549 | Lambda | INTRINSIK CORP | Statement of Work 9 | \$0.00 | | \$1,935.00 |
| 550 | Lonafarnib | INTRINSIK CORP | Change Order 1 to Statement of Work | \$0.00 | · | |
| 551 | Lonafarnib | INTRINSIK CORP | Master Service Agreement | \$0.00 | - | |
| 221 | Lonaranno | II VI KII VOIK COKI | iviasiei service Agreement | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|-------|-------------------|------------------------------|--|----------------|-------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 552 | Lonafarnib | INTRINSIK CORP | Statement of Work 14 | \$0.00 | \$0.00 | |
| 553 | Lonafarnib | Intrinsik Corp. | Statement of Work AMCP | \$0.00 | \$0.00 | \$0.00 |
| 554 | Lonafarnib | Intrinsik Corp. | Statement of Work dated July 10, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 555 | Lonafarnib | Intrinsik Corp. | Statement of Work dated March 29, 2020 | \$0.00 | \$0.00 | \$0.00 |
| 556 | Lonafarnib | Intrinsik Corp. | Statement of Work dated March 4, 2016 | \$0.00 | \$0.00 | \$0.00 |
| 557 | Lonafarnib | Intrinsik Corp. | Statement of Work dated May 1,2020 | \$0.00 | \$0.00 | \$0.00 |
| 337 | Eonaramo | Intrinsik Health Sciences | Statement of Work dated May 1,2020 | φ0.00 | ψ0.00 | ψ0.00 |
| 558 | Lambda | Inc. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 559 | Lonafarnib | INTSEL CHIMOS | AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| 560 | Lonafarnib | INTSEL CHIMOS | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 300 | Lonaramio | inVentiv Health Clinique | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 561 | Lonafarnib | Inc. | Bioanalytical Services Quotation | \$0.00 | \$0.00 | \$0.00 |
| 301 | Lonaranno | INVENTIV HEALTH | Bloanarytical Services Quotation | \$0.00 | \$0.00 | \$0.00 |
| 562 | Lonafarnib | | Master Comings Assessment | \$0.00 | \$0.00 | \$0.00 |
| 302 | Lonatarnio | CLINIQUE, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | inVentiv Health Public | | | | |
| | | | | | | |
| 562 | I amafamil | Relations LLC, a Syneos | Statement of Worls | \$0.00 | \$0.00 | \$0.00 |
| 563 | Lonafarnib | Heath Company | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | INVENTIV HEALTH | | | | |
| - C 4 | l | PUBLIC RELATIONS, | | | 40.00 | 40.00 |
| 564 | Avexitide | LLC | Statement of Work dated March 19, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | INVENTIV HEALTH | | | | |
| | | PUBLIC RELATIONS, | | | | |
| 565 | Lonafarnib | LLC | Change Order | \$0.00 | \$0.00 | \$0.00 |
| | | INVENTIV HEALTH | | | | |
| | | PUBLIC RELATIONS, | | | | |
| 566 | Lonafarnib | LLC | Statement of Work dated July 16, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | INVENTIV HEALTH | | | | |
| | | PUBLIC RELATIONS, | | | | |
| 567 | Lonafarnib | LLC | Statement of Work dated March 19, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change Notification Form dated May 15, | | | |
| 568 | Lonafarnib | IQVIA | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change Notification Form dated October 26, | | | |
| 569 | Lonafarnib | IQVIA | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | ` | Warranty Letter re: Protocol number EIG- | , , , , , | * | **** |
| 570 | Lonafarnib | IQVIA | LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| 571 | Lambda | IQVIA Biotech LLC | Change Notification Form 2 | \$0.00 | 40.00 | \$0.00 |
| 572 | Lambda | IQVIA Biotech LLC | Change Notification Form 3 | \$0.00 | | \$0.00 |
| 573 | Lambda | IQVIA Biotech LLC | Change Notification Form 4 | \$0.00 | | \$0.00 |
| 574 | Lambda | IQVIA Biotech LLC | Change Proposal No. 1 | \$0.00 | \$6,151.93 | \$0.00 |
| 575 | Lambda | IQVIA Biotech LLC | Statement of Work | \$0.00 | | \$0.00 |
| 576 | Lambda | IQVIA Biotech LLC | Work Order | \$0.00 | | \$0.00 |
| 577 | Lambda Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 15 | \$0.00 | \$0.00 | \$0.00 |
| 311 | Lonarafillo | IN A TY DIOIECH FFC | Letter of Delegation re: Protocol Number | \$0.00 | \$0.00 | \$0.00 |
| 578 | Lonafarnib | IOVIA Clinical AD | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| 310 | Lonaranno | IQVIA Clinical AB | Consent for Disclosure of Study Information | | \$0.00 | \$0.00 |
| 570 | I one form the | IQVIA Holdings, Inc.; | | | ΦΩ ΩΩ | \$0.00 |
| 579 | Lonafarnib | ClinAudits, LLC | to Auditors | \$0.00 | \$0.00 | \$0.00 |
| | | | TSQM and TSQM Scoring Algorithm | | | |
| | | | License (collectively, "Licensed Materials") | | | |
| | | YOY 7. Y | for 150 participants - in connection with | | | |
| 580 | Lambda | IQVIA Inc. | protocol EIG-LMD | \$0.00 | \$0.00 | \$0.00 |
| 581 | Lonafarnib | IQVIA Inc. | 22-Jun-20 General Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 582 | Lonafarnib | IQVIA Inc. | 6/22/2020 General Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Boulevard, Durham, NC 27703 ("IQVIA") | | | |
| | | | and relates to the Master Services | | | |
| 583 | Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |

| | Related | | | | | Lonafarnib |
|-----|------------|--------------------------|--|----------------|-------------|---|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 584 | Lonafarnib | IQVIA Inc. | Change Order #1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 585 | Lonafarnib | IQVIA Inc. | Change Order #1 Work Order #1 | \$0.00 | \$0.00 | \$0.00 |
| 586 | Lonafarnib | IQVIA Inc. | Change Order #2 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 587 | Lonafarnib | IQVIA Inc. | Change Order #2 Work Order #1 | \$0.00 | \$0.00 | \$0.00 |
| 588 | Lonafarnib | IQVIA Inc. | Change Order #3 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | Change Order 1 to General Services | 40100 | 44114 | 40.00 |
| 589 | Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 590 | Lonafarnib | IQVIA Inc. | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 591 | Lonafarnib | IQVIA Inc. | Change Order 3 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 592 | Lonafarnib | IQVIA Inc. | Details of MSA and Work Order: | \$0.00 | \$0.00 | \$0.00 |
| 593 | Lonafarnib | IQVIA Inc. | Executed Work Order | \$0.00 | \$0.00 | \$0.00 |
| 594 | Lonafarnib | IQVIA Inc. | General Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 595 | Lonafarnib | IQVIA Inc. | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 596 | Lonafarnib | IQVIA Inc. | Novation Agreement | \$0.00 | \$0.00 | \$0.00 |
| 597 | Lonafarnib | IQVIA Inc. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | Technical Agreement for the Provision of | ***** | * | * |
| 598 | Lonafarnib | IQVIA Inc. | EU Qualified Person | \$0.00 | \$0.00 | \$0.00 |
| | | | Work Order #2: SOP Training | 40.00 | 40.00 | 40.00 |
| | | | Work Gradi #2. 501 Training | | | |
| 599 | Lonafarnib | IQVIA Inc. | Work Order #2 | \$0.00 | \$0.00 | \$0.00 |
| 600 | Lonafarnib | IQVIA Inc. | Work Order dated May 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 601 | Lonafarnib | IQVIA Inc. | Work Order dated May 3, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 001 | Editaranio | IQ VIII IIIC. | Work Gradi auted May 3, 2017 | φ0.00 | ψ0.00 | ψ0.00 |
| | | IQVIA Ltd.; IQVIA RDS | | | | |
| | | Eastern Holdings GmbH; | | | | |
| | | IQVIA Tibbi Istatistik | | | | |
| | | Ticaret ve Musavirlik | | | | |
| | | Limited Sirketi; IQVIA | Letter of Delegation re: Protocol Number | | | |
| 602 | Lonafarnib | RDS Ges.m.b.H. | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| 002 | Lonaranno | KD5 GCs.III.U.II. | Authorization to Proceed Agreement dated | \$0.00 | φ0.00 | \$0.00 |
| 603 | Lonafarnib | IQVIA RDS INC. | June 30, 2022 | \$0.00 | \$0.00 | \$0.00 |
| 003 | Lonaranno | IQVIA RDS INC. | Authorization to Proceed Agreement dated | \$0.00 | \$0.00 | \$0.00 |
| 604 | Lonafarnib | IQVIA RDS INC. | October 29, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 605 | Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | \$0.00 | \$0.00 | \$0.00 |
| 606 | Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA Change Order 3 to MSA | \$0.00 | \$0.00 | \$0.00 |
| 607 | Lonafarnib | IQVIA RDS Inc. | Change Order No. 2 | \$0.00 | \$0.00 | \$0.00 |
| 608 | Lonafarnib | IQVIA RDS Inc. | Change Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 609 | Lonafarnib | IQVIA RDS INC. | Change Order No. 5 Change Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| | Lonafarnib | IQVIA RDS INC. | | \$0.00 | | \$0.00 |
| 610 | Lonararnio | IQVIA KDS INC. | Details of General Services Agreement: | \$0.00 | \$0.00 | \$0.00 |
| (11 | T C 1. | IOVIA DDC DIC | Details of Master Services Agreement and | ¢0.00 | ¢0.00 | ¢0.00 |
| 611 | Lonafarnib | IQVIA RDS INC. | Authorization to Proceed: | \$0.00 | \$0.00 | \$0.00 |
| 612 | Lonafarnib | IQVIA RDS INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| (12 | T C '1 | IQVIA RDS Ireland | EU Legal Representative of Non-EU | 00.00 | 00.00 | # 0.00 |
| 613 | Lonafarnib | Limited | Sponsor | \$0.00 | \$0.00 | \$0.00 |
| | | YOUNG DESCRIPTION OF | Sponsor's Commitment Form - Clinical Trial | | 40.00 | 40.00 |
| 614 | Lonafarnib | IQVIA RDS LTD, Israel | with an Investigational Product | \$0.00 | \$0.00 | \$0.00 |
| | | ITR Laboratories Canada | | | | |
| 615 | Avexitide | Inc. | Revised (4) Supplementary Quotation | \$0.00 | \$0.00 | \$0.00 |
| | 1 | ITR Laboratories Canada | | | | |
| 616 | Avexitide | Inc. | Supplementary Quotation | \$0.00 | \$0.00 | \$0.00 |
| 617 | Lambda | J. A. [Name onf File] | Services Agreement | \$0.00 | \$400.00 | \$0.00 |
| | | | Letter of Support for the University of | | | |
| | | Jordan Feld M.D., MPH; | Toronto COVID-19 Action Initiative Grant | | | |
| 618 | Lambda | Toronto General Hospital | Application | \$0.00 | \$0.00 | \$0.00 |
| | | JUSTIN STINDT | | | | |
| 619 | Lonafarnib | CONSULTING | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |

| | Related | | | | | Lonafarnib |
|------|--------------------------|--------------------------------------|--|------------------|------------------|-------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| π | ASSCI | JUSTIN STINDT | Description of Contract of Bease | Aveattlac Cure | Lambua Curc | Curc |
| 620 | Lonafarnib | CONSULTING | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| 020 | Lonaranno | JUSTIN STINDT | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| 621 | Lonafarnib | CONSULTING | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| 021 | Lonaranno | JUSTIN STINDT | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| (22 | T C 11. | | Statement of West-2 | \$0.00 | \$0.00 | \$0.00 |
| 622 | Lonafarnib | CONSULTING JUSTIN STINDT | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| 622 | Lonafarnib | CONSULTING | Statement of Wesley | \$0.00 | \$0.00 | \$0.00 |
| 623 | Lonaranno | JUSTIN STINDT | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| 624 | Lonafarnib | CONSULTING | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| 024 | Lonaranno | JUSTIN STINDT | Statement of Work o | \$0.00 | \$0.00 | \$0.00 |
| 625 | Lonafarnib | CONSULTING | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| 023 | Lonararnio | JUSTIN STINDT | Statement of Work / | \$0.00 | \$0.00 | \$0.00 |
| 626 | Lonafarnib | CONSULTING | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| 020 | Lonaranno | JUSTIN STINDT | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| 627 | Lonafarnib | CONSULTING | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| 027 | Lonatarnio | CONSULTING | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| | | Justin Stindt of Justin | | | | |
| 628 | Lonafarnib | Stindt Consulting GmbH | Paprocentation Mandata | \$0.00 | \$0.00 | \$0.00 |
| 028 | Lonaranno | Kaiser Foundation | Representation Mandate | \$0.00 | \$0.00 | \$0.00 |
| 629 | Lonafarnib | Research Institute | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| 029 | Lonaranno | KAROLINSKA | Chilical That Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | UNIVERSITY | | | | |
| 630 | Lonafarnib | HOSPITAL | Contract Period | \$0.00 | \$0.00 | \$0.00 |
| 631 | Lonafarnib | KPS Life, LLC | Exhibit A- Work Order No. 2 | \$0.00 | \$0.00 | · · |
| 632 | Lonafarnib | KPS LIFE, LLC | | \$0.00 | \$0.00 | \$0.00 |
| 032 | Lonaranno | KROLL CYBER | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 633 | Lonafarnib | SECURITY, LLC | LOE | \$0.00 | \$0.00 | \$0.00 |
| 033 | Lonaranno | SECURITI, LLC | Statement of Work for Information Security | \$0.00 | \$0.00 | \$0.00 |
| 634 | Lonafarnib | V nall Cylean Sagymity II C | and Computer Forensics Services | \$0.00 | \$0.00 | \$0.00 |
| 034 | Lonaranno | KRYOCAL, LLC DBA | and Computer Forensics Services | \$0.00 | \$0.00 | \$0.00 |
| 635 | Avexitide | KYROSPHERE | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 033 | Avexilide | KRYOCAL, LLC DBA | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 636 | Lambda | KYROSPHERE | Statement of Understanding | \$0.00 | \$6,562.50 | \$0.00 |
| 637 | Lonafarnib | Kymanox Corporation | Statement of Understanding Master Services Agreement | \$0.00 | \$0,302.30 | \$0.00 |
| 037 | Lonaranno | Kymanox Corporation | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 638 | Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated Jul 20, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 038 | Lonaranno | Laviana Filannia Co., Liu | Quotation dated Jul 20, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 639 | Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated July 31, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 039 | Lonaranno | Liver Care Centre | Zaoudon dated July 31, 2010 | φ0.00 | φ0.00 | φ0.00 |
| | | | Indemnification Letter re: Protocol No. EIG- | | | |
| 640 | Lonafarnib | Liver Centre | LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| 040 | Lonaranno | Liver Centre | Amendment No. 3 to Master Services and | ş0.00 | \$0.00 | \$0.00 |
| 641 | Avexitide | LONZA BEND, INC. | Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| 041 | Avexiliae | LONZA DEND, INC. | Amendment No. 1 to the Commercial | \$0.00 | \$0.00 | \$0.00 |
| 642 | Lonafarnib | LONZA BEND, INC. | Supply Agreement | \$0.00 | \$0.00 | |
| 042 | Lonaranno | LONZA DEND, INC. | Amendment No. 2 to the Commercial | \$0.00 | \$0.00 | |
| 6/12 | Lonofornih | LONZA BEND, INC. | | \$0.00 | ¢ስ ስስ | |
| 643 | Lonafarnib Lonafarnib | LONZA BEND, INC. | Supply Agreement Change Order 2 | \$0.00 \$0.00 | \$0.00 \$0.00 | |
| 645 | Lonafarnib | LONZA BEND, INC. | Change Order 2 to Statement of Work | \$0.00 | \$0.00 | |
| 646 | Lonafarnib | LONZA BEND, INC. | Change Order 4 to Statement of Work Change Order 4 to Statement of Work | \$0.00 | \$0.00 | |
| 647 | | · · | _ | | \$0.00 | |
| 04/ | Lonafarnib | LONZA BEND, INC. | Change Order 8 to Statement of Work Services and Clinical Manufacture | \$0.00 | \$0.00 | |
| 610 | I one form:1 | LONZA DENID, INC. | | ¢0.00 | ቀ ስ ስስ | |
| 648 | Lonafarnib Lonafarnib | LONZA BEND, INC. LONZA BEND, INC. | Agreement dated 15 December, 2022 Statement of Work | \$0.00 \$0.00 | \$0.00 \$0.00 | \$21,000.00 |
| | | | | · · | · · | · |
| 650 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 10 April 2023 | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|-----|------------|--|---|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 651 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |
| 652 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |
| 653 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 14 August 2020 | \$0.00 | \$0.00 | |
| 654 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 22 April 2019 | \$0.00 | \$0.00 | |
| 655 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 4 December 2018 | \$0.00 | \$0.00 | |
| 656 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 5 April 2018 | \$0.00 | \$0.00 | |
| 657 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 8 May 2018 | \$0.00 | \$0.00 | |
| 658 | Lonafarnib | LONZA BEND, INC. | Statement of Work dated 9 January 2019 | \$0.00 | \$0.00 | |
| | | Lonza Bend; Patheon | Total Transportation Management ("TTM") | | | |
| 659 | Lonafarnib | Canada | Freight Quote | \$0.00 | \$0.00 | \$0.00 |
| 660 | Lonafarnib | Lonza Pharma & BioTech | Change Order #7 to E141-8598 | \$0.00 | \$0.00 | \$0.00 |
| 661 | Lonafarnib | Lonza Pharma & Biotech | Change Order No.1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 662 | Lonafarnib | Lonza Pharma & Biotech | Lonza Amendment 1 to SOW - PN 03706 | \$0.00 | \$0.00 | \$0.00 |
| 663 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| 664 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 05119 | \$0.00 | \$0.00 | \$0.00 |
| 665 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 6 to SOW E141-8083 | \$0.00 | \$0.00 | \$0.00 |
| 666 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form | \$0.00 | \$0.00 | \$0.00 |
| 667 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05119 | \$0.00 | \$0.00 | \$0.00 |
| 668 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05281 | \$0.00 | \$0.00 | \$0.00 |
| 669 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05895 | \$0.00 | \$0.00 | \$0.00 |
| 670 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001073 | \$0.00 | \$0.00 | \$0.00 |
| 671 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001256 | \$0.00 | \$0.00 | \$0.00 |
| 672 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| 673 | Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order to SOW PN0405 | \$0.00 | \$0.00 | \$0.00 |
| 674 | Lonafarnib | Lonza Pharma & BioTech | Proposed Statement of Work (SOW) | \$0.00 | \$0.00 | \$0.00 |
| 675 | Lonafarnib | Lonza Pharma & BioTech | Stability Proposal | \$0.00 | \$0.00 | \$0.00 |
| 676 | Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-092310 | \$0.00 | \$0.00 | \$0.00 |
| 677 | Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-136736 | \$0.00 | \$0.00 | \$0.00 |
| 678 | Lonafarnib | Lonza Pharma & BioTech | Validation Proposal dated 29 January 2020 | \$0.00 | \$0.00 | \$0.00 |
| 679 | Lonafarnib | Lonza Pharma & BioTech | Validation Proposal dated 6 April 2020 | \$0.00 | \$0.00 | \$0.00 |
| 680 | Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-030137" | \$0.00 | \$0.00 | \$0.00 |
| 681 | Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-031335" | \$0.00 | \$0.00 | \$0.00 |
| | | i | | 1 | | |

| Asset | | Related | | | | | Lonafarnib |
|--|-----|--------------|------------------------|---|----------------|---|------------|
| | # | | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | |
| BioTech; Bend Research Change Order No. 1 to Statement of Work So.00 S | | | | | | | |
| | | | | Change Order No. 1 to Statement of Work | | | |
| Lonza Pharma & BioTech; Hend Research Change Order No. 1 to Statement of Work Discussion | 682 | Lonafarnib | , , | _ | \$0.00 | \$0.00 | \$0.00 |
| Bot Teck, Bend Research Change Order No. 1 to Statement of Work So.00 | | | | | ***** | 4 | 4 |
| | | | | Change Order No. 1 to Statement of Work | | | |
| Conza Pharma & Bio Tech; Bend Research Inc. Conza Pharma & Bio Tech; Bend Research Inc. Conza Pharma & Bio Tech; Bend Research Change Order No. 2 to Statement of Work PN-030137" S0.00 S0.0 | 683 | Lonafarnib | · · | _ | \$0.00 | \$0.00 | \$0.00 |
| Change Order No. 2 to Statement of Work So.00 So | | | | | ***** | 40.00 | 4 |
| | | | | Change Order No. 2 to Statement of Work | | | |
| Conza Pharma & BioTech; Bend Research Change Order No. 2 to Statement of Work PN-031335" S0.00 | 684 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Change Order No. 2 to Statement of Work So.00 S | | | | | ***** | ***** | ***** |
| Constantib Inc. PN-031335" S0.00 S0. | | | | Change Order No. 2 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Change Order No. 2 to Statement of Work PN-05281" S0.00 | 685 | Lonafarnib | · · | | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. Change Order No. 2 to Statement of Work PN-05281" So.00 | | | | | ***** | 40.00 | 4 |
| Lonaramib Inc. "PN-05281" S0.00 S0.0 | | | | Change Order No. 2 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Change Order No. 2 to Statement of Work BioTech; Bend Research Change Order No. 3 to Statement of Work So.00 | 686 | Lonafarnib | · · | _ | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research PN-033823 S0.00 | | Zonaranio | | 11, 00201 | \$0.00 | φ0.00 | \$0.00 |
| | | | | Change Order No. 2 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Inc. PN-04064" S0.00 S0. | 687 | Lonafarnib | · · | _ | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. Change Order No. 3 to Statement of Work The No. 4064" So. 00 S | | 201101111111 | | 11, 000020 | \$0.00 | \$0.00 | \$0.00 |
| Lonza Pharma & Bio Tech; Bend Research Change Order No. 3 to Statement of Work So.00 S | | | | Change Order No. 3 to Statement of Work | | | |
| Lonza Pharma & Bio Tech; Bend Research Change Order No. 3 to Statement of Work So.00 S | 688 | Lonafarnib | · · | 6 | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. Change Order No. 3 to Statement of Work So.00 So | | Zonaranio | | 11,01001 | \$0.00 | φ0.00 | \$0.00 |
| Lonza Pharma & BioTech; Bend Research Inc. | | | | Change Order No. 3 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Inc. Change Order No. 3 to Statement of Work PN-033823 \$0.00 | 689 | Lonafarnib | | e | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Change Order No. 3 to Statement of Work PN-033823 S0.00 S0.00 S0.00 S0.00 | | Zonaranio | | 11, 00201 | \$0.00 | φ0.00 | \$0.00 |
| Constarnib Inc. PN-033823 S0.00 S0.00 S0.00 S0.00 | | | | Change Order No. 3 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Change Order No. 5 to Statement of Work So.00 So.00 So.00 So.00 So.00 | 690 | Lonafarnib | | 6 | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. | 0,0 | Zonaranio | | 11, 000020 | \$0.00 | ψ0.00 | \$0.00 |
| Constantib Inc. | | | | Change Order No. 5 to Statement of Work | | | |
| Lonza Pharma & BioTech; Bend Research Inc. Development Proposal \$0.00 \$0.00 \$0.00 \$0.00 | 691 | Lonafarnib | · · | _ | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. Development Proposal \$0.00 | | | | | , , , , | * | *** |
| Contain Con | | | | | | | |
| Lonza Pharma & BioTech; Bend Research Inc. Proposed Change Order #1 to Eiger SOW \$0.00 \$0.00 \$0.00 | 692 | Lonafarnib | · · | Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| Constantib Inc. Proposed Change Order #1 to Eiger SOW \$0.00 \$0.00 \$0.00 | | | Lonza Pharma & | 1 1 | , , , , | * | *** |
| Constantib Inc. Proposed Change Order #1 to Eiger SOW \$0.00 \$0.00 \$0.00 | | | BioTech; Bend Research | | | | |
| Lonza Pharma & BioTech; Bend Research Proposed Statement of Work (SOW) dated August 20, 2018 \$0.00 \$ | 693 | Lonafarnib | | Proposed Change Order #1 to Eiger SOW | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Proposed Statement of Work (SOW) dated August 20, 2018 \$0.00 | | | Lonza Pharma & | | | | |
| Lona | | | | Proposed Statement of Work (SOW) dated | | | |
| Lonza Pharma & BioTech; Bend Research Proposed Statement of Work (SOW) dated August 8, 2018 \$0.00 \$0 | 694 | Lonafarnib | | - | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Inc. August 8, 2018 \$0.00 | | | Lonza Pharma & | | | | |
| Lonza Pharma & BioTech; Bend Research Proposed Statement of Work (SOW) dated So.00 So. | | | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonza Pharma & Bio Tech; Bend Research Proposed Statement of Work (SOW) dated So.00 So | 695 | Lonafarnib | · · | . , | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Proposed Statement of Work (SOW) dated | | | Lonza Pharma & | | | | |
| Lonza Pharma & BioTech; Bend Research Proposed Statement of Work (SOW) dated So.00 So. | | | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| BioTech; Bend Research Proposed Statement of Work (SOW) dated | 696 | Lonafarnib | Inc. | May 30, 2018 | \$0.00 | \$0.00 | \$0.00 |
| BioTech; Bend Research Proposed Statement of Work (SOW) dated | | | Lonza Pharma & | | | | |
| 697 Lonafarnib Inc. May 8,2018 \$0.00 \$0.00 \$0.00 Lonza Pharma & BioTech; Bend Research BioTech; Bend Research 80.00 \$0.00 | | | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonza Pharma & BioTech; Bend Research Validation Proposal \$0.00 \$0.00 \$0.00 | 697 | Lonafarnib | · · | 1 · · · · · · · · · · · · · · · · | \$0.00 | \$0.00 | \$0.00 |
| 698 Lonafarnib Inc. Validation Proposal \$0.00 \$0.00 \$0.00 699 Lonafarnib Luto Research Limited Quote dated 16^ January 2020 \$0.00 \$0.00 \$0.00 700 Lonafarnib Luto Research Limited Quote dated 1st October 2019 \$0.00 \$0.00 \$0.00 701 Lonafarnib Luto Research Limited Quote dated 7th April 2021 \$0.00 \$0.00 \$0.00 | | | Lonza Pharma & | | | | |
| 698 Lonafarnib Inc. Validation Proposal \$0.00 \$0.00 \$0.00 699 Lonafarnib Luto Research Limited Quote dated 16^ January 2020 \$0.00 \$0.00 \$0.00 700 Lonafarnib Luto Research Limited Quote dated 1st October 2019 \$0.00 \$0.00 \$0.00 701 Lonafarnib Luto Research Limited Quote dated 7th April 2021 \$0.00 \$0.00 \$0.00 | | | | | | | |
| 699LonafarnibLuto Research LimitedQuote dated 16^ January 2020\$0.00\$0.00\$0.00700LonafarnibLuto Research LimitedQuote dated 1st October 2019\$0.00\$0.00\$0.00701LonafarnibLuto Research LimitedQuote dated 7th April 2021\$0.00\$0.00\$0.00 | 698 | Lonafarnib | · · | Validation Proposal | \$0.00 | \$0.00 | \$0.00 |
| 700LonafarnibLuto Research LimitedQuote dated 1st October 2019\$0.00\$0.00\$0.00701LonafarnibLuto Research LimitedQuote dated 7th April 2021\$0.00\$0.00\$0.00 | 699 | Lonafarnib | Luto Research Limited | | | | \$0.00 |
| 701 Lonafarnib Luto Research Limited Quote dated 7th April 2021 \$0.00 \$0.00 \$0.00 | 700 | | Luto Research Limited | _ | \$0.00 | | \$0.00 |
| | 701 | | | | | | \$0.00 |
| | 702 | Lonafarnib | Luto Research Ltd. | Quote for User Testing | | | |

| | Related | | | | | Lonafarnib |
|------------|------------|---------------------------|--|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | M. C., MD [Name on | | | | |
| 703 | Lambda | File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 704 | Avexitide | M. T., MD [Name on File] | | \$232.00 | \$0.00 | \$0.00 |
| 705 | Lambda | M. B. C. [Name on File] | Statement of Work Hepatitis Delta Virus (HDV) | \$0.00 | \$0.00 | \$0.00 |
| | | Mary Bordeaux | Statement of Work Hepatitis Delta Virus | | | |
| 706 | Lonafarnib | Consulting | (HDV) | \$0.00 | \$0.00 | \$0.00 |
| 707 | Lonafarnib | MC2 | Change Order No. 1 | \$0.00 | \$0.00 | \$0.00 |
| 708 | Lonafarnib | MC2 | Standard Service Agreement dated 2/14/2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Standard Service Agreement dated | | | |
| 709 | Lonafarnib | MC2 | 2/27/2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | Standard Service Agreement dated | | | |
| 710 | Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Standard Service Agreement dated | | | |
| 711 | Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| 712 | Avexitide | MEDPACE INC. | 1st Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| 713 | Avexitide | MEDPACE INC. | 1st Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| 714 | Avexitide | MEDPACE INC. | 1st Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| 715 | Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| 716 | Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| 717 | Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| 718 | Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| 719 | Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| 720 | Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| 720 | Avexitide | MEDI ACE INC. | Amendment 1 to LOI AVX-005 | | \$0.00 | · |
| 722 | Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-006 | \$17.79 | \$0.00 | \$0.00 |
| 723 | Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-000 Amendment 1 to LOI AVX-007 | | \$0.00 | \$0.00 |
| 723 | Avexitide | MEDI ACE INC. | Amendment 2 to LOI AVX-007 | | \$0.00 | \$0.00 |
| 725 | Avexitide | MEDPACE INC. | Amendment 2 to LOI AVX-005 Amendment 2 to LOI AVX-006 | | \$0.00 | \$0.00 |
| | Avexitide | MEDPACE INC. | | | \$0.00 | \$0.00 |
| 726 727 | Avexitide | MEDPACE INC. MEDPACE INC. | LOI AVX007 LOI CHI | | \$0.00 | · |
| | | | | | | \$0.00 |
| 728 | Avexitide | MEDPACE INC. | LOI CHI Children | | \$0.00 | \$0.00 |
| 729 | Avexitide | MEDPACE INC. | Master Service Agreement | | \$0.00 | \$0.00 |
| 730 | Avexitide | MEDPACE INC. | Task Order 2 | | \$0.00 | \$0.00 |
| 731 | Avexitide | MEDPACE INC. | Task Order 3 | | \$0.00 | \$0.00 |
| | | Merck Sharp & Dohme | | | | |
| 732 | Lonafarnib | LLC | Amendment 6 | \$0.00 | \$0.00 | \$0.00 |
| | | Merck Sharp & Dohme | | | | |
| 733 | Lonafarnib | LLC | Amendment 7 | \$0.00 | \$0.00 | \$0.00 |
| | | Merck Sharp & Dohme | Amendment to the Pharmacovigilance | | | |
| 734 | Lonafarnib | LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Merck Sharp & Dohme | Memo of Understanding (MOU) the | | | |
| 735 | Lonafarnib | LLC | Pharmacovigilance Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Merck Sharp & Dohme | | | | |
| 736 | Lonafarnib | LLC | Termination Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | MillBridge | Cost Proposal - API Supplier Qualification | | | |
| 737 | Lambda | Pharmaceutical Science | Audit | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | MillBridge | | | | |
| 738 | Lonafarnib | Pharmaceutical Science | Cost Proposal - Supplier Qualification Audit | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | MillBridge | Cost Proposal - Supplier Qualification | | | |
| 739 | Lonafarnib | Pharmaceutical Science | Audits | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |

| | Related | | | | | Lonafarnib |
|-----|------------|-----------------------|--|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 740 | Lonafarnib | MIND+MATTER LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 741 | Lonafarnib | MIND+MATTER LLC | Statement of Work 14 | \$0.00 | \$0.00 | \$0.00 |
| 742 | Lonafarnib | MIND+MATTER LLC | Statement of Work 15 | \$0.00 | \$0.00 | \$0.00 |
| 743 | Lonafarnib | MIND+MATTER LLC | Statement of Work 16 | \$0.00 | \$0.00 | \$0.00 |
| 744 | Lonafarnib | MIND+MATTER LLC | Statement of Work 17 | \$0.00 | \$0.00 | \$0.00 |
| 745 | Lonafarnib | MIND+MATTER LLC | Statement of Work 22 | \$0.00 | \$0.00 | \$0.00 |
| 746 | Lonafarnib | MKO Global Partners | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 747 | Avexitide | MPI Research | Archived Materials Report | \$0.00 | \$0.00 | \$0.00 |
| 748 | Lonafarnib | Myoderm | Quote re: Destruction of Drug | \$0.00 | \$0.00 | \$0.00 |
| 749 | Lambda | N/A | Letter of Legal Representative | \$0.00 | \$0.00 | \$0.00 |
| 750 | Lonafarnib | NDA GROUP AB | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | NDA REGULATORY | 5 | | · | |
| 751 | Lonafarnib | SCIENCE LTD | Work Order | \$0.00 | \$0.00 | \$0.00 |
| 752 | Lonafarnib | NEOPHARM LTD. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Terms for an Exclusive Distribution | | · | • |
| 753 | Lonafarnib | NEOPHARM LTD. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work Refolding Studies of IL- | · | · | |
| 754 | Lambda | Neurozon LLC | 29 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work Refolding Studies of IL- | · | | |
| 755 | Lambda | Neurozon LLC | 29 (Supplement) | \$0.00 | \$0.00 | \$0.00 |
| 756 | Lonafarnib | NIH | Author Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | NORTH RIM | 5 | · | | |
| 757 | Lambda | CONSULTING, INC | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 758 | Lambda | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| 759 | Lambda | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| 760 | Lambda | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |
| 761 | Lambda | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| 762 | Lambda | Novella Clinical LLC | Change Proposal 7 | \$0.00 | \$0.00 | \$0.00 |
| 763 | Lambda | Novella Clinical LLC | Change Proposal No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 764 | Lonafarnib | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| 765 | Lonafarnib | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| 766 | Lonafarnib | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |
| 767 | Lonafarnib | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| 768 | Lonafarnib | Novella Clinical LLC | Change Proposal 7 | \$0.00 | \$0.00 | \$0.00 |
| 769 | Avexitide | Novocol Pharma | Development Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| | | OneSource Regulatory | | | | |
| 770 | Lonafarnib | LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Origami Evenements | | | | |
| 771 | Lonafarnib | SARL | Contrat de Participation a un Evenement | \$0.00 | \$0.00 | \$0.00 |
| | | Origami Evènements | - | | | |
| 772 | Lonafarnib | SARL | Contract of Participation in an Event | \$0.00 | \$0.00 | \$0.00 |
| | | | - | | | |
| | | ORPHAN DRUG | | | | |
| 773 | Lonafarnib | CONSULTING LIMITED | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | ORPHAN DRUG | | | | |
| 774 | Lonafarnib | | Services and Confidentiality Agreement | \$0.00 | \$0.00 | \$0.00 |
| 775 | Lambda | Oxford Finance | Amendment | \$0.00 | \$0.00 | \$0.00 |
| | | | Approval of Second Amendment to Oxford | | | |
| 776 | Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| 777 | Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Second Amendment to Loan and Security | | | |
| 778 | Lambda | Oxford Finance | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 779 | Lonafarnib | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Third Amendment to Loan and Security | | | |
| | Lonafarnib | Oxford Finance | Agreement | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------|---------------------------|--|----------------|---|-------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 781 | Lambda | Oxford Finance LLC | Amendment Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | | Oxford Finance LLC, as | | * | * | * |
| 782 | Lambda | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| | | Oxford Finance LLC, as | , | *** | * | * |
| 783 | Lambda | Collateral Agent | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Oxford Finance LLC, as | Second Amendment to Loan and Security | *** | * | *** |
| 784 | Lambda | Collateral Agent | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Oxford Finance LLC, as | Second Amendment to Loan and Security | · | · | |
| 785 | Lambda | Collateral Agent | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Oxford Finance LLC, as | | | | |
| 786 | Lonafarnib | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| 787 | Lonafarnib | Oxford Finance, LLC | Success Fee Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | PACIFIC PHARMA | Statement of Work describes Services and | | | |
| 788 | Lonafarnib | GROUP, LLC | Deliverables to be performed and | \$0.00 | \$0.00 | \$0.00 |
| | | Pacific Pharmaceutical | · | | | |
| | | Services and Alcami | | | | |
| 789 | Avexitide | Company | Services Agreement | \$25,061.25 | \$0.00 | \$0.00 |
| | | Palmetto GBA; Ed | | . , | | |
| 790 | Lonafarnib | Gazarkh | Letter re: TPA Authorization | \$0.00 | \$0.00 | \$0.00 |
| | | PARTNERS4ACCESS | | | | |
| 791 | Lonafarnib | B.V. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - EU / Access & | | | |
| | | PARTNERS4ACCESS | Commercialisation Support dated Dec 23rd, | | | |
| 792 | Lonafarnib | B.V. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - EU / Access & | | | |
| | | PARTNERS4ACCESS | Commercialisation Support dated Dec 23rd, | | | |
| 793 | Lonafarnib | B.V. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - Lonafarnib EU / | | | |
| | | PARTNERS4ACCESS | Access & Commercialisation Support dated | | | |
| 794 | Lonafarnib | B.V. | April 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - Lonafarnib EU / | | | |
| | | PARTNERS4ACCESS | Access & Commercialisation Support dated | | | |
| 795 | Lonafarnib | B.V. | July 30th, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - Lonafarnib EU / | | | |
| | | PARTNERS4ACCESS | Access & Commercialisation Support dated | | | |
| 796 | Lonafarnib | B.V. | March 10th, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work - Lonafarnib EU | | | |
| | | PARTNERS4ACCESS | Commercialisation Planning dated March 5, | | | |
| 797 | Lonafarnib | B.V. | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | Patheon Inc, part of | | | | |
| 798 | Lonafarnib | Thermo Fisher Scientific | Change of Scope 10 | \$0.00 | \$0.00 | |
| | | | | | | |
| | | Patheon Inc, part of | | | | |
| 799 | Lonafarnib | Thermo Fisher Scientific | Change of Scope 11 | \$0.00 | \$0.00 | \$68,895.30 |
| | | D d T | | | | 400,000 |
| 000 | T C 1 | Patheon Inc, part of | | 00.00 | Φ0.00 | |
| 800 | Lonafarnib | Thermo Fisher Scientific | Change of Scope 13 | \$0.00 | \$0.00 | |
| | | Patheon Inc, part of | | | | |
| 801 | Lonafarnib | Thermo Fisher Scientific | Change of Scope 14 | \$0.00 | \$0.00 | |
| 802 | Avexitide | PATHEON INC. | GRP Agreement | \$0.00 | \$0.00 | \$0.00 |
| 802 | Avexitide | PATHEON INC. PATHEON INC. | Project Proposal dated Dec 13, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 803 | Avexitide | PATHEON INC. PATHEON INC. | Validation Project Plan | \$0.00 | \$0.00 | \$0.00 |
| 804 | Lambda | PATHEON INC. PATHEON INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 803 | Lonafarnib | Patheon Inc. | Change of Scope 1 | \$0.00 | \$0.00 | φυ.υυ |
| 000 | Lonararino | i auteon inc. | Change of Scope 1 | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|-----|-------------|---------------------------|--|----------------|----------------|-------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 807 | Lonafarnib | PATHEON INC. | Change of Scope 11 | \$0.00 | \$0.00 | |
| 808 | Lonafarnib | PATHEON INC. | Change of Scope 12 | \$0.00 | \$0.00 | |
| 809 | Lonafarnib | PATHEON INC. | Change of Scope 13 | \$0.00 | \$0.00 | |
| 810 | Lonafarnib | PATHEON INC. | Change of Scope 14 | \$0.00 | \$0.00 | |
| 811 | Lonafarnib | PATHEON INC. | Change of Scope 15 | \$0.00 | \$0.00 | |
| 812 | Lonafarnib | PATHEON INC. | Change of Scope 16 | \$0.00 | \$0.00 | |
| 813 | Lonafarnib | PATHEON INC. | Change of Scope 17 | \$0.00 | \$0.00 | |
| 814 | Lonafarnib | PATHEON INC. | Change of Scope 18 | \$0.00 | \$0.00 | |
| 815 | Lonafarnib | PATHEON INC. | Change of Scope 19 | \$0.00 | \$0.00 | |
| 816 | Lonafarnib | PATHEON INC. | Change of Scope 2 | \$0.00 | \$0.00 | |
| 817 | Lonafarnib | PATHEON INC. | Change of Scope 20 | \$0.00 | \$0.00 | |
| 818 | Lonafarnib | PATHEON INC. | Change of Scope 21 | \$0.00 | \$0.00 | |
| 819 | Lonafarnib | PATHEON INC. | Change of Scope 22 | \$0.00 | \$0.00 | |
| 820 | Lonafarnib | PATHEON INC. | Change of Scope 22 Change of Scope 23 | \$0.00 | \$0.00 | |
| 821 | Lonafarnib | PATHEON INC. | Change of Scope 24 | \$0.00 | \$0.00 | |
| 822 | | PATHEON INC. | Change of Scope 25 | \$0.00 | \$0.00 | |
| | Lonafarnib | PATHEON INC. PATHEON INC. | | | | |
| 823 | Lonafarnib | | Change of Scope 27 | \$0.00 | \$0.00 | |
| 824 | Lonafarnib | PATHEON INC. | Change of Scope 28 | \$0.00 | \$0.00 | |
| 825 | Lonafarnib | PATHEON INC. | Change of Scope 29 | \$0.00 | \$0.00 | |
| 826 | Lonafarnib | PATHEON INC. | Change of Scope 3 | \$0.00 | \$0.00 | |
| 827 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC | \$0.00 | \$0.00 | |
| 828 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 1 | \$0.00 | \$0.00 | |
| 829 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 2 | \$0.00 | \$0.00 | |
| 830 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 3 | \$0.00 | \$0.00 | |
| 831 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 4 | \$0.00 | \$0.00 | |
| 832 | Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 5 | \$0.00 | \$0.00 | |
| 833 | Lonafarnib | PATHEON INC. | Change of Scope 30 | \$0.00 | \$0.00 | |
| 834 | Lonafarnib | PATHEON INC. | Change of Scope 31 | \$0.00 | \$0.00 | |
| 835 | Lonafarnib | PATHEON INC. | Change of Scope 32 | \$0.00 | \$0.00 | |
| 836 | Lonafarnib | PATHEON INC. | Change of Scope 34 | \$0.00 | \$0.00 | |
| 837 | Lonafarnib | PATHEON INC. | Change of Scope 35 | \$0.00 | \$0.00 | |
| 838 | Lonafarnib | PATHEON INC. | Change of Scope 36 | \$0.00 | \$0.00 | |
| 839 | Lonafarnib | PATHEON INC. | Change of Scope 37 | \$0.00 | \$0.00 | |
| 840 | Lonafarnib | PATHEON INC. | Change of Scope 4 | \$0.00 | \$0.00 | |
| 841 | Lonafarnib | Patheon Inc. | Change of Scope 40 | \$0.00 | \$0.00 | |
| 842 | Lonafarnib | Patheon Inc. | Change of Scope 41 | \$0.00 | \$0.00 | \$68,895.30 |
| 843 | Lonafarnib | Patheon Inc. | Change of Scope 43 | \$0.00 | \$0.00 | |
| 844 | Lonafarnib | PATHEON INC. | Change of Scope 44 | \$0.00 | \$0.00 | |
| 845 | Lonafarnib | Patheon Inc. | Change of Scope 45 | \$0.00 | \$0.00 | |
| 846 | Lonafarnib | Patheon Inc. | Change of Scope 46 | \$0.00 | \$0.00 | |
| 847 | Lonafarnib | Patheon Inc. | Change of Scope 47 | \$0.00 | \$0.00 | |
| 848 | Lonafarnib | Patheon Inc. | Change of Scope 5 | \$0.00 | \$0.00 | |
| 849 | Lonafarnib | Patheon Inc. | Change of Scope 52 | \$0.00 | \$0.00 | |
| 850 | Lonafarnib | PATHEON INC. | Change of Scope 6 | \$0.00 | \$0.00 | |
| 851 | Lonafarnib | Patheon Inc. | Change of Scope 7 | \$0.00 | \$0.00 | |
| 852 | Lonafarnib | Patheon Inc. | Change of Scope 8 | \$0.00 | \$0.00 | |
| 853 | Lonafarnib | Patheon Inc. | Change of Scope 9 | \$0.00 | \$0.00 | |
| | | | Change of Scope COS-55-R0 to Proposal | | | |
| 854 | Lonafarnib | Patheon Inc. | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | |
| - | | | Change of Scope Proposal For Eiger | | | |
| 855 | Lonafarnib | Patheon Inc. | Biopharmaceuticals | \$0.00 | \$0.00 | |
| 856 | Lonafarnib | Patheon Inc. | Change of Scope to Proposal | \$0.00 | \$0.00 | |
| | 1 | 1 | Change of Scope | 1.00 | 44.30 | |
| 857 | Lonafarnib | Patheon Inc. | Proposal For Eiger Biopharmaceuticals | \$0.00 | \$0.00 | |
| 858 | Lonafarnib | PATHEON INC. | Change Order | \$0.00 | \$0.00 | |
| 859 | Lonafarnib | PATHEON INC. | COS 1 | \$0.00 | \$0.00 | |
| 059 | Lonaraillio | I IIIILON IIIC. | 0001 | \$0.00 | φ 0. 00 | |

| | Related | | | | | Lonafarnib |
|-----|-------------|------------------------------|---|----------------|------------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 860 | Lonafarnib | Patheon Inc. | Phase II CTM Manufacturing Proposal | \$0.00 | \$0.00 | |
| | | | Product Agreement for Lonafarnib - | | | |
| 861 | Lonafarnib | PATHEON INC. | Progeria | \$0.00 | \$0.00 | |
| | | | Product Agreement") for Lonafarnib | , , , , , | * | |
| | | | Capsules 50 mg and 7S mg ("Product") to | | | |
| 862 | Lonafarnib | PATHEON INC. | the Master Service | \$0.00 | \$0.00 | |
| 863 | Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | \$0.00 | |
| 864 | Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | \$0.00 | |
| 865 | Lonafarnib | PATHEON INC. | Project Proposal dated May 21, 2021 | \$0.00 | \$0.00 | |
| 866 | Lonafarnib | PATHEON INC. | Project Proposal dated September 27, 2021 | \$0.00 | \$0.00 | |
| 867 | Lonafarnib | PATHEON INC. | Project Proposal March 30, 2020 | \$0.00 | \$0.00 | |
| 868 | Lonafarnib | PATHEON INC. | Quote dated April 2, 2019 | \$0.00 | \$0.00 | |
| 869 | Lonafarnib | PATHEON INC. | Quote dated July 09, 2019 | \$0.00 | \$0.00 | |
| 870 | Lonafarnib | PATHEON INC. | Quote dated March 25,2019 | \$0.00 | \$0.00 | |
| 871 | Lonafarnib | PATHEON INC. | Quote dated March 27, 2019 | \$0.00 | \$0.00 | |
| 872 | Lonafarnib | PATHEON INC. | Quote for Size 4 dated July 09, 2019 | \$0.00 | \$0.00 | |
| 0,2 | Zenwimie | Patheon Inc.; Gerteis | Quete let 5120 : auteu eury 65, 2015 | \$0.00 | ψ0.00 | |
| | | Maschinen + | | | | |
| | | Processengineering AG; | | | | |
| 873 | Lonafarnib | Joseph Lannen | Quotation No 106954 | \$0.00 | \$0.00 | \$0.00 |
| 874 | Avexitide | Patheon Italia Spa | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 07. | 11701111111 | Patheon Manufacturing | Services rigidement | ψ0.00 | ψ0.00 | ψ0.00 |
| 875 | Avexitide | Services LLC | Change of Scope to Proposal 1 | | \$0.00 | \$0.00 |
| 075 | rivexities | Patheon Manufacturing | Change of Scope to Froposar 1 | 1 | ψ0.00 | Ψ0.00 |
| 876 | Avexitide | Services LLC | Change of Scope to Proposal 10 | | \$0.00 | \$0.00 |
| 070 | Tivexitide | Patheon Manufacturing | Change of Scope to Froposar 10 | - | ψ0.00 | Ψ0.00 |
| 877 | Avexitide | Services LLC | Change of Scope to Proposal 14 | | \$0.00 | \$0.00 |
| 077 | Tivexitide | Patheon Manufacturing | Change of Scope to Froposar 11 | - | ψ0.00 | Ψ0.00 |
| 878 | Avexitide | Services LLC | Change of Scope to Proposal 14-RO | | \$0.00 | \$0.00 |
| 070 | Tivexitide | Patheon Manufacturing | Change of Scope to Froposar 1 1 10 | - | ψ0.00 | Ψ0.00 |
| 879 | Avexitide | Services LLC | Patheon Change of Scope 1 R0 | | \$0.00 | \$0.00 |
| 017 | Aveature | Patheon Manufacturing | Tatheon Change of Scope 1 Ro | - | ψ0.00 | φ0.00 |
| 880 | Avexitide | Services LLC | Patheon Change of Scope 11 | | \$0.00 | \$0.00 |
| 880 | Avexitiue | Patheon Manufacturing | Tatheon Change of Scope 11 | \$111,092.41 | φ0.00 | \$0.00 |
| 881 | Avexitide | Services LLC | Patheon Change of Scope 12 | | \$0.00 | \$0.00 |
| 001 | Aveature | Patheon Manufacturing | Tatheon Change of Scope 12 | - | \$0.00 | \$0.00 |
| 882 | Avexitide | Services LLC | Patheon Change of Scope 13 | | \$0.00 | \$0.00 |
| 002 | Avexitiue | Patheon Manufacturing | 1 attreon Change of Scope 13 | - | φ0.00 | \$0.00 |
| 883 | Avexitide | Services LLC | Patheon Change of Scope 2 | | \$0.00 | \$0.00 |
| 003 | Avexitiue | Patheon Manufacturing | 1 atticon Change of Scope 2 | - | \$0.00 | \$0.00 |
| 884 | Avexitide | Services LLC | Patheon Change of Scope 2-R2 | | \$0.00 | \$0.00 |
| 004 | Avexitide | Patheon Manufacturing | r attreon Change of Scope 2-K2 | - | \$0.00 | \$0.00 |
| 885 | Avexitide | Services LLC | Patheon Change of Scope 3 | | \$0.00 | \$0.00 |
| 883 | Avexitide | | Patheon Change of Scope 3 | | \$0.00 | \$0.00 |
| 006 | A | Patheon Manufacturing | Poth on Chance of Some 1 | | \$0.00 | \$0.00 |
| 886 | Avexitide | Services LLC | Patheon Change of Scope 4 | \$0.00 | \$0.00 \$0.00 | \$0.00 |
| 887 | Lambda | Patheon UK Limited | Change of Scope | \$0.00 | * | \$0.00 |
| 888 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 10 | \$0.00 | \$0.00 | \$0.00 |
| 889 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 11 | \$0.00 | \$0.00 | \$0.00 |
| 890 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 15 | \$0.00 | \$0.00 | \$0.00 |
| 891 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 16 | \$0.00 | \$0.00 | \$0.00 |
| 892 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 18 | \$0.00 | \$0.00 | \$0.00 |
| 893 | Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 9 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope COS P-MNC-101564-R3- | | | |
| 894 | Lambda | Patheon UK Limited | COS-07-R0_20200424 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope COS P-MNC-101564-R3- | | | |
| 895 | Lambda | Patheon UK Limited | COS-11-R0_20210712 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-----|------------------|---|--|----------------|---------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | I | Change of Scope COS P-MNC-101564-R3- | | | |
| 896 | Lambda | Patheon UK Limited | COS-11-R0 20210712 (1) | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope COS P-MNC-101564-R3- | | | |
| 897 | Lambda | Patheon UK Limited | COS-02-R0_20170119 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope COS-17-R0 to P-MNC- | | | |
| 898 | Lambda | Patheon UK Limited | 101564-R3_20220324 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope COS-P-MNC-101564-R3- | | | |
| 899 | Lambda | Patheon UK Limited | COS-08-R3_20210309 | \$0.00 | \$0.00 | \$0.00 |
| | | | Change of Scope P-MNC-101564-R3-COS- | | | |
| 900 | Lambda | Patheon UK Limited | 11-R1_20211103 | \$0.00 | \$0.00 | \$0.00 |
| 001 | T 1 1 | D 41 THZ I ' 4 1 | Purchase Order Patheon UK_COS 3 P-MNC | #0.00 | \$0.00 | ¢0.00 |
| 901 | Lambda | Patheon UK Limited | 101564-R3_20171129 | \$0.00 | \$0.00 | \$0.00 |
| 902 | Lambda | Patheon UK Limited | Purchase Order Patheon UK_COS 4 P-MNC | \$0.00 | \$0.00 | \$0.00 |
| 902 | Lambda | Patheon UK Limited | 101564-R3_20171115 | \$0.00 | \$0.00 | \$0.00 |
| 903 | Lambda | PATHEON UK LIMITED | Validation Master Plan | \$0.00 | \$0.00 | \$0.00 |
| 703 | Lamoda | Patheon UK Limited; | Change of Scope Patheon UK COS 14 P- | \$0.00 | \$0.00 | \$0.00 |
| 904 | Lambda | Fisher Clinical Services | MNC-101564-R3 2022 | \$0.00 | \$0.00 | \$0.00 |
| 701 | Lumouu | Patheon UK Limited; | Change of Scope Patheon UK COS 20 P- | φ0.00 | ψ0.00 | φ0.00 |
| 905 | Lambda | Fisher Clinical Services | MNC-101564-R4 20220722 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon UK Limited; | Change of Scope Patheon UK COS 21 P- | 4 | 4 | 4 |
| 906 | Lambda | Fisher Clinical Services | MNC-101564-R4 20221025 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon UK Limited; | Change of Scope Patheon UK COS 24 P- | | | |
| 907 | Lambda | Fisher Clinical Services | MNC-101564-R4_20230209 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon UK Limited; | Change of Scope Patheon UK_COS 25 P- | | | |
| 908 | Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon UK Limited; | Change of Scope Patheon UK_COS 26 P- | | | |
| 909 | Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon UK Limited; | Change of Scope Patheon UK_COS 29-R0 | | | |
| 910 | Lambda | Fisher Clinical Services | to P-MNC-101564-R4_20240213 | \$0.00 | \$0.00 | \$0.00 |
| 011 | | Patheon UK Limited; | Change of Scope: Prefilled Syringes COS 12 | # 0.00 | # 0.00 | Φο οο |
| 911 | Lambda | Fisher Clinical Services | to P-MNC-101564-R3_20220125 | \$0.00 | \$0.00 | \$0.00 |
| | | Dath and LIV Limited. | Change of Scope: Prefilled Syringes Patheon | | | |
| 912 | Lambda | Patheon UK Limited; Fisher Clinical Services | UK_P-MNC-101564-R4-COS-23- R0 20220922 | \$0.00 | \$0.00 | \$0.00 |
| 912 | Lamoda | Patheon UK Limited; | Change of Scope: Proposal COS 12 to C- | \$0.00 | \$0.00 | \$0.00 |
| 913 | Lonafarnib | Fisher Clinical Services | TRC-121992-R4 20230429 | \$0.00 | \$0.00 | \$0.00 |
| 713 | Lonararino | i isher Chinear Services | TRC-121772-R4_20230427 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon, Part of Thermo | | | | |
| | | Fischer Scientific; | Element Quote Element Quote 20-012- | | | |
| 914 | Lonafarnib | Element Toronto | 162900 Rev 1_20200420 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon, Part of | Total Transportation Management ("TTM") | | | |
| 915 | Lonafarnib | Thermofisher | Freight Quote 453021_20201202 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-03-R2 to Project | | | |
| 916 | Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-04-R1 to Project | | | |
| 917 | Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| 010 | | Patheon; Thermo Fisher | Change of Scope COS-05-R0 to Project | ** ** | *^ ^ ^ | ** ** |
| 918 | Avexitide | Scientific Fig. | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| 010 | A supposited it. | Patheon; Thermo Fisher | Change of Scope COS-06-R0 to Project | 60.00 | \$0.00 | \$0.00 |
| 919 | Avexitide | Scientific Patheon; Thermo Fisher | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| 920 | Avexitide | Scientific | Change of Scope COS-07-R0 to Project Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| 920 | AVEAINGE | Patheon; Thermo Fisher | Change of Scope COS-08-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| 921 | Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| 741 | 2 I V CAITIUC | Scientific | 1 10p0541 110. 1 -OKI -203333-KI | φυ.00 | φυ.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------------|------------------------|-----------------------------------|---|----------------|------------------|---|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Patheon; Thermo Fisher | Change of scope COS-09-R1 to Project | | | |
| 922 | Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-19-R0 to Project | | | |
| 923 | Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-38-R0 to Proposal | | | |
| 924 | Lonafarnib | Scientific | No. P-TRP- 114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| 025 | T C 1 | Patheon; Thermo Fisher | Change of Scope COS-39-R0 to Proposal | # 0.00 | # 0.00 | 00.00 |
| 925 | Lonafarnib | Scientific Patheon; Thermo Fisher | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| 926 | Lonafarnib | Scientific | Change of Scope COS-42-R1 to Proposal No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| 920 | Lonaranno | Patheon; Thermo Fisher | Change of Scope COS-48-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| 927 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| 721 | Lonararino | Patheon; Thermo Fisher | Change of Scope COS-49-R0 to Proposal | ψ0.00 | ψ0.00 | ψ0.00 |
| 928 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-50-R0 to Proposal | , , , , , | * | * |
| 929 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-51-R0 to Proposal | | | |
| 930 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-53-R0 to Proposal | | | |
| 931 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-54-R2 to Proposal | | | |
| 932 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-57-R1 to Proposal | | | |
| 933 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | Patheon; Thermo Fisher | Change of Scope COS-58-R0 to Proposal | | | |
| 934 | Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| 00.5 | | Pharma Solutions USA, | | | | |
| 935 | Lonafarnib | Inc. | Services Agreement | \$0.00 | \$0.00 | \$745.49 |
| | | Pharmaceutical Research | | | | |
| 936 | Lonafarnib | Associates, Inc. | Services Agreement | \$0.00 | \$0.00 | \$658.79 |
| 937 | Lambda | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 | \$0.00 | \$0.00 | · |
| 731 | Lamoda | i narmaDirections, me. | WKO-EIG-1225 Lambda Interferon CMC | \$0.00 | \$0.00 | \$0.00 |
| 938 | Lambda | PharmaDirections, Inc. | Oversight | \$0.00 | \$0.00 | \$0.00 |
| | | | WKO-EIG-1361 CMC Support: Interferon | 41111 | 44144 | 4 |
| 939 | Lambda | PharmaDirections, Inc. | Lambda | \$0.00 | \$0.00 | \$0.00 |
| 940 | Lambda | PharmaDirections, Inc. | WKO-EIG-991 Ad hoc Consulting | \$0.00 | \$0.00 | \$0.00 |
| 941 | Lonafarnib | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 | \$0.00 | \$0.00 | |
| 942 | Lonafarnib | PharmaDirections, Inc. | Amendment #2 WKO-EIG-879 | \$0.00 | \$0.00 | |
| | | | WKO-EIG-1362 CMC Support: Lonafarnib | | | \$2,902.50 |
| 943 | Lonafarnib | PharmaDirections, Inc. | Progeria Commercial | \$0.00 | \$0.00 | |
| 944 | Lonafarnib | PharmaDirections, Inc. | WKO-EIG-879 Ad hoc Consulting | \$0.00 | \$0.00 | |
| | | PharmaLex UK Services | | | | |
| 945 | Lonafarnib | Limited | Estimate | \$0.00 | \$0.00 | \$0.00 |
| 0.46 | | PHARMALEX US | | | | |
| 946 | Lonafarnib | CORPORATION | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 0.45 | | Pharmalytical Consulting | | 04.500.00 | | |
| 947 | Avexitide | LLC | Services Agreement | \$1,520.00 | \$0.00 | \$0.00 |
| 948 949 | Avexitide | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 6 | 4 | \$0.00 \$0.00 | |
| | Avexitide | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 4 | 4 | \$0.00 \$0.00 | \$0.00 |
| 950 951 | Avexitide Avexitide | PharmaPace, Inc. PharmaPace, Inc. | Change Order No. 2 to Work Order No. 4 Change Order No. 2 to Work Order No. 6 | 4 | \$0.00 \$0.00 | |
| 951 | Avexitide | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 4 | 1 | \$0.00 | \$0.00 |
| 934 | AVEAILIGE | i narmar acc, mc. | Third Amendment to t he Statement of Work | 1 | φυ.00 | \$0.00 |
| 953 | Avexitide | PharmaPace, Inc. | #1 Agreement | \$7,848.75 | \$0.00 | \$0.00 |
| 954 | Avexitide | PharmaPace, Inc. | Work Order No. 4 | 1 | \$0.00 | · |
| 254 | 11 VEAILIUE | 1 marmar acc, mc. | II OIA OIGGI NO. T | l | φ υ. 00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|--------------------------|--|---|--|----------------------------|----------------------------|----------------------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 955 | Avexitide | PharmaPace, Inc. | Work Order No. 6 | | \$0.00 | \$0.00 |
| 956 | Avexitide | PharmaPace, Inc. | Work Order No. 7 | † | \$0.00 | \$0.00 |
| 957 | Avexitide | PharmaPace, Inc. | Work Order No. 8 | - | \$0.00 | \$0.00 |
| 958 | Avexitide | PharmaPace, Inc. | Work Order No. 9 | - | \$0.00 | \$0.00 |
| 959 | Lonafarnib | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 960 | Lonafarnib | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 961 | Lonafarnib | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 962 | Lonafarnib | PharmaPace, Inc. | Change Order No. 4 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 963 | Lonafarnib | PharmaPace, Inc. | Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| 703 | Lonararmo | PINEX COMERCIAL | Work Gradi No. 3 | Ψ0.00 | ψ0.00 | ψ0.00 |
| | | IMPORTADORA E | | | | |
| 964 | Lonafarnib | SERVICOS LTDA | Letter for Authorization | \$0.00 | \$0.00 | \$0.00 |
| 704 | Lonaranno | PLATFORM | Letter for Authorization | \$0.00 | \$0.00 | \$0.00 |
| 965 | Lambda | LIFESCIENCES INC. | Duafassianal Cauriass Assamant | \$0.00 | \$0.00 | \$0.00 |
| 903 | Lamoua | POLARIS | Professional Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 966 | Avexitide | CONSULTING, INC. | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 900 | Avexilide | POLARIS | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 067 | A | | Mostor Comico A successor | \$0.00 | \$0.00 | 20.00 |
| 967 | Avexitide | CONSULTING, INC. | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| 0.60 | 1 | POLARIS | G : A | #0.00 | #0.00 | ¢0.00 |
| 968 | Avexitide | CONSULTING, INC. | Service Agreement August 15, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 0.60 | | POLARIS | | | 40.00 | 40.00 |
| 969 | Avexitide | CONSULTING, INC. | Service Agreement dated 24 Jul 2023 | \$0.00 | \$0.00 | \$0.00 |
| | 1 | POLARIS | | | ** | |
| 970 | Avexitide | CONSULTING, INC. | Service Agreement dated August 8, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | L | | | | ** | |
| 971 | Lambda | PPD Development, L.P. | LMD Agreement | \$0.00 | \$0.00 | \$0.00 |
| 0.70 | | | | | 40.00 | 40.00 |
| 972 | Lambda | PPD Development, L.P. | Proposal | \$0.00 | \$0.00 | \$0.00 |
| 973 | Avexitide | PPS | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 974 | Lonafarnib | Proforma Element 3 | Order Acknowledgement | \$0.00 | \$0.00 | \$0.00 |
| 0.7.5 | | D 0 51 10 | Order Acknowledgement dated October 23, | | 40.00 | 40.00 |
| 975 | Lonafarnib | Proforma Element 3 | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | 1 Totorma Element 3 | | | | |
| | | | | | | |
| | | ProPharma Group | | | | |
| | | ProPharma Group Holdings, LLC; | Project Assignment #3 Under Consulting | | | |
| 976 | Lambda | ProPharma Group | Project Assignment #3 Under Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 976 | Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC | | \$0.00 | \$0.00 | \$0.00 |
| 976 | Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; | Agreement Project Assignment #4 Under Consulting | | | |
| 976 977 | Lambda Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group | Agreement | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| | | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC | Agreement Project Assignment #4 Under Consulting | | | |
| | | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC | Agreement Project Assignment #4 Under Consulting Agreement | | | |
| 977 | Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting | \$0.00 | \$0.00 | \$0.00 |
| | | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group | Agreement Project Assignment #4 Under Consulting Agreement | | | |
| 977 | Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group LCC Q SQUARED | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 977 | Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Coup Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 | Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group LC Q SQUARED SOLUTIONS HOLDINGS, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 977 | Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group LC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 978 979 | Lambda Lonafarnib Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 | Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 978 979 | Lambda Lonafarnib Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 978 979 980 | Lambda Lonafarnib Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August 2016 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 |
| 977 978 979 | Lambda Lonafarnib Lambda | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 977 978 979 980 | Lambda Lonafarnib Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August 2016 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 |
| 977 978 979 980 | Lambda Lonafarnib Lambda Lonafarnib | ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group, LLC ProPharma Group Holdings, LLC; ProPharma Group Holdings, LLC; ProPharma Group, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC Q SQUARED SOLUTIONS HOLDINGS, LLC | Agreement Project Assignment #4 Under Consulting Agreement Project Assignment #6 Under Consulting Agreement Hold Harmless Agreement dated April 7, 2021 Hold Harmless Agreement dated 17 August 2016 Work Order dated May 3rd, 2019 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 | \$0.00 \$0.00 \$0.00 |

| | Related | | | | | Lonafarnib |
|------------|------------|------------------------------|--|-----------------------|-------------|---------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| " | Isset | Contract Counterparty | Description of Contract of Bease | Aveatine cure | Lambua Cure | Curc |
| | | Ouest Diagnostics Clinical | Change Order No. 2 to SOW Type 4: | | | |
| 983 | Lonafarnib | Laboratories, Inc. | Combined Services | \$0.00 | \$0.00 | \$0.00 |
| 705 | Lonaramo | Eucoratories, me. | Comonica Services | ψ0.00 | φ0.00 | ψ0.00 |
| | | QUEST DIAGNOSTICS | | | | |
| 984 | Lonafarnib | INCORPORATED | Amendment 1 to Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | 7 | 40.00 | ***** |
| | | QUEST DIAGNOSTICS | | | | |
| 985 | Lonafarnib | INCORPORATED | Amendment 2 to Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | QUEST DIAGNOSTICS | | | | |
| 986 | Lonafarnib | INCORPORATED | Master Data Licensing Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | QUEST DIAGNOSTICS | | | | |
| 987 | Lonafarnib | INCORPORATED | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | QUEST DIAGNOSTICS | | | | |
| 988 | Lonafarnib | INCORPORATED | Statement of Work SOW#2 | \$0.00 | \$0.00 | \$0.00 |
| | | Quest Diagnostics | | | | |
| 989 | Lonafarnib | Incorporated | Statement of Work SOW#3 | \$0.00 | \$0.00 | \$0.00 |
| 000 | r 6 1 | | Letter of Delegation re: Protocol Number | #0.00 | ФО ОО | # 0.00 |
| 990 991 | Lonafarnib | Quintiles Vietnam LLC | EIG-LNF-011 Statement of Work SOW#3 | \$0.00 | | \$0.00 |
| 991 | Lambda | R3LS, LLC | Statement of Work SOW#3 | \$0.00 | \$0.00 | \$0.00 |
| 002 | T1. 1. | RAMBAM HEALTH | Clinia 1 Tai-1 A amand | \$0.00 | \$0.00 | ¢0.00 |
| 992 | Lambda | CORPORATION | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| 993 | Lonafarnib | RAMBAM HEALTH CORPORATION | Clinical Trial Assessment | \$0.00 | \$0.00 | \$0.00 |
| 993 | Lonaranno | REGULATORY | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| 994 | Lonafarnib | PHARMA NET SRL | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 777 | Lonaranno | THARMA NET SKE | Project Assignment #1 Under Consulting | φ0.00 | \$0.00 | \$0.00 |
| 995 | Lonafarnib | Regulatory Pharma Net srl | | \$0.00 | \$0.00 | \$0.00 |
| 773 | Lonaranio | Research Triangle | rigicement | ψ0.00 | φ0.00 | ψ0.00 |
| | | Institute DBA RTI | | | | |
| 996 | Lonafarnib | International | Proposal 0282201.557 | \$0.00 | \$0.00 | \$0.00 |
| 997 | Avexitide | RHO, INC. | LOI | \$0.00 | | \$0.00 |
| 998 | Avexitide | RHO, INC. | LOI Extension | \$0.00 | · · | \$0.00 |
| 999 | Avexitide | RHO, INC. | LOI Second Extension | \$0.00 | | \$0.00 |
| | Avexitide | RHO, INC. | LOI Third Extension | \$0.00 | | |
| 1001 | Avexitide | RHO, INC. | Master Services Agreement | \$0.00 | | \$0.00 |
| | | ROTE LISTE Service | | | | |
| 1002 | Lonafarnib | GmbH | Contract for the Publication | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1003 | Avexitide | LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1004 | Lambda | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1005 | Lambda | LLC | Letter for Authorization PA 3 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Project Agreement 3 to the Product | | | |
| 1006 | Lambda | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1007 | Lambda | LLC | Second Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1008 | Lambda | LLC | Work Order 15, February 28, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1009 | Lambda | LLC | Work Order 15, January 25, 2021 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------|--|----------------|---------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | RRD INTERNATIONAL, | | | | |
| 1010 | Lambda | LLC | Work Order 16, December 9, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1011 | Lambda | LLC | Work Order 16, March 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1012 | Lambda | LLC | Work Order 16, March 11, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1013 | Lambda | LLC | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1014 | Lambda | LLC | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 1015 | | RRD INTERNATIONAL, | W 10 1 0 % i G . 1 5 2017 | # 0.00 | #0.00 | Φ0.00 |
| 1015 | Lambda | LLC | Work Order 9, effective September 5, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 1016 | Lambda | RRD INTERNATIONAL, | Wash Order O effective Sentember 5 2017 | \$0.00 | \$0.00 | 00.00 |
| 1010 | Lambda | LLC | Work Order 9, effective September 5, 2017 Work Order No. 10 provided for Strategic | \$0.00 | \$0.00 | \$0.00 |
| | | PPD INTERNATIONAL | Regulatory Consultation and FDA Liaison | | | |
| 1017 | Lambda | LLC | Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| 1017 | Lamoda | RRD INTERNATIONAL, | • | \$0.00 | φ0.00 | φ0.00 |
| 1018 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| 1010 | Zame aa | RRD INTERNATIONAL, | Work Order No. 16 to the Master Services | φοίου | φοίου | \$0.00 |
| 1019 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 17 to the Master Services | * | * | , |
| 1020 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 18 to the Master Services | | | |
| 1021 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 8 to the Master Services | | | |
| 1022 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1023 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1024 | Lambda | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Amended and Restated Work Order No. 11 | | | |
| 1025 | Lonafarnib | LLC | to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1006 | r c 1 | RRD INTERNATIONAL, | 1 1 1 1 20101210 | \$0.00 | #0.00 | Φ0.00 |
| 1026 | Lonafarnib | LLC RRD INTERNATIONAL, | Amendment 1 to 1 20191210 | \$0.00 | \$0.00 | \$0.00 |
| 1027 | Lonafarnib | LLC | Amendment 1 to 2 20191002 | \$0.00 | \$0.00 | \$0.00 |
| 1027 | Lonaranno | RRD INTERNATIONAL, | Amendment 1 to 2 20191002 | \$0.00 | \$0.00 | \$0.00 |
| 1028 | Lonafarnib | LLC | Amendment 1 to 3 20191022 | \$0.00 | \$0.00 | \$0.00 |
| 1020 | Zenwianie | RRD INTERNATIONAL, | 20171022 | φοίου | ψ0.00 | \$0.00 |
| 1029 | Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 1 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | · | · |
| 1030 | Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 2 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1031 | Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 3 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1032 | Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 4 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1033 | Lonafarnib | LLC | Amendment 2 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 |
| 100. | | RRD INTERNATIONAL, | | ** | | ** |
| 1034 | Lonafarnib | LLC | Amendment 2 to TO 2 202003003 | \$0.00 | \$0.00 | \$0.00 |
| 1025 | T C | RRD INTERNATIONAL, | A 1 | 00.00 | 00.00 | 00.00 |
| 1035 | Lonafarnib | LLC | Amendment 2 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 |
| 1026 | Lonoformil | RRD INTERNATIONAL, | Amondment 2 to DA 1 Interior CO 5 | ¢0.00 | ድ ስ ስስ | ¢0.00 |
| 1036 | Lonafarnib | LLC | Amendment 3 to PA 1 - Interim CO 5 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------|---|----------------|--------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | RRD INTERNATIONAL, | | | | |
| 1037 | Lonafarnib | LLC | Amendment 3 to PA 1 - Interim CO 6 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1038 | Lonafarnib | LLC | Amendment 3 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Amendment 3 to Project Agreement 1 | | | |
| 1039 | Lonafarnib | LLC | Interim CO 4 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1040 | Lonafarnib | LLC | Amendment 3 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1041 | Lonafarnib | LLC | Amendment 3 to TO 2 20200303 | \$0.00 | \$0.00 | \$0.00 |
| 1042 | T C 1 | RRD INTERNATIONAL, | 1 | ¢0.00 | #0.00 | Φ0.00 |
| 1042 | Lonafarnib | LLC | Amendment 3 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 |
| 1043 | Lonafarnib | RRD INTERNATIONAL, LLC | Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1043 | Lonaranno | LLC | Contractor Agreement Development Advisory Committee | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Development Advisory Committee | | | |
| 1044 | Lonafarnib | LLC | Amendment 2 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 |
| 1011 | Editaranio | EEC | Development Advisory Committee | ψ0.00 | ψ0.00 | ψ0.00 |
| 1 | | RRD INTERNATIONAL, | 20. stoping in 110. isoty committee | | | |
| 1045 | Lonafarnib | LLC | Amendment 2 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | Eiger Progeria PA2 Amendment 1 | | | |
| 1 | | RRD INTERNATIONAL, | | | | |
| 1046 | Lonafarnib | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Eiger Progeria PA2 Amendment 1 Interim | | | |
| | | | CO2 | | | |
| | | RRD INTERNATIONAL, | | | | |
| 1047 | Lonafarnib | LLC | Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Eiger Progeria PA2 Amendment 1 Interim | | | |
| | | RRD INTERNATIONAL, | | | | |
| 1048 | Lonafarnib | LLC | Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1049 | Lonafarnib | RRD INTERNATIONAL, LLC | Eiger Progeria PA2 Amendment 2 Interim | \$0.00 | \$0.00 | 00.00 |
| 1049 | Lonatarnio | RRD INTERNATIONAL, | CO1 Eiger Project Agreement Eighth Amendment to Task Order 1 to the | \$0.00 | \$0.00 | \$0.00 |
| 1050 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1030 | Lonaranno | RRD INTERNATIONAL, | Waster independent Contractor Agreement | \$0.00 | φ0.00 | \$0.00 |
| 1051 | Lonafarnib | LLC | Fifth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1001 | Zenaranie | RRD INTERNATIONAL, | Fifth Amendment to Task Order 1 to the | \$0.00 | φοίου | \$0.00 |
| 1052 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1053 | Lonafarnib | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | First Amendment to Task Order 1 to the | | | |
| 1054 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | · · | First Amendment to Task Order 1 to the | | | |
| 1055 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | | | |
| 1056 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1057 | T 6 1 | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | #0.00 | #0.00 | Φ0.00 |
| 1057 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1058 | Lonafarnib | LLC | First Amendment to Task Order 10 to the | \$0.00 | \$0.00 | \$0.00 |
| 1036 | Lonaranno | RRD INTERNATIONAL, | Master Independent Contractor Agreement First Amendment to Task Order 2 to the | \$0.00 | \$0.00 | \$0.00 |
| 1059 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1007 | | RRD INTERNATIONAL, | First Amendment to Task Order 2 to the | ψ0.00 | ψ0.00 | φ0.00 |
| 1060 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Task Order 5 to the | \$0.00 | \$0.00 | \$0.00 |
| 1061 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|-------|------------|---------------------------|---|----------------|-------------|--|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | First Amendment to Task Order 6 to the | | | |
| 1062 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Task Order 6 to the | | | |
| 1063 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Task Order 8 to the | | | |
| 1064 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to the Master Independent | | | |
| 1065 | Lonafarnib | LLC | Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Work Order No. 12 to | | | |
| 1066 | Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | First Amendment to Work Order No. 13 to | | | |
| 1067 | Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 10.00 | | RRD INTERNATIONAL, | First Amendment to Work Order No. 14 to | # 0.00 | 40.00 | * • • • • • • • • • • • • • • • • • • • |
| 1068 | Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 10.00 | | RRD INTERNATIONAL, | | 40.00 | 40.00 | * • • • • |
| 1069 | Lonafarnib | LLC | Fourth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1070 | T C 1 | RRD INTERNATIONAL, | Fourth Amendment to Task Order 1 to the | ¢0.00 | ¢0.00 | Φ0.00 |
| 1070 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1071 | T C 11. | RRD INTERNATIONAL, | Fourth Amendment to the Amended and | 60.00 | \$0.00 | ¢0.00 |
| 1071 | Lonafarnib | LLC RRD INTERNATIONAL, | Restated Work Order | \$0.00 | \$0.00 | \$0.00 |
| 1072 | Lonafarnib | LLC | Interim CO 1 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| 1072 | Lonararino | RRD INTERNATIONAL, | internii CO 1 to FA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| 1073 | Lonafarnib | LLC | Interim CO 1 to PA 2 (MM TO 12) | \$0.00 | \$0.00 | \$0.00 |
| 1073 | Lonaranno | RRD INTERNATIONAL, | internit CO 1 to 1 A 2 (WIW 10 12) | \$0.00 | \$0.00 | \$0.00 |
| 1074 | Lonafarnib | LLC | Interim CO 2 to PA 2 (MM TO 14) | \$0.00 | \$0.00 | \$0.00 |
| 1074 | Lonaranno | RRD INTERNATIONAL, | Internit CO 2 to 1 A 2 (IVIIVI 10 14) | \$0.00 | \$0.00 | \$0.00 |
| 1075 | Lonafarnib | LLC | Interim CO 3 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| 1073 | Lonararino | RRD INTERNATIONAL, | internii 60 3 to 1 A 1 Amendment 3 | \$0.00 | \$0.00 | φ0.00 |
| 1076 | Lonafarnib | LLC | Interim CO 3 to PA 2 (MM TO 22) | \$0.00 | \$0.00 | \$0.00 |
| 1070 | Lonararino | RRD INTERNATIONAL, | internit de 3 to 1712 (MM 1 e 22) | φ0.00 | ψ0.00 | ψ0.00 |
| 1077 | Lonafarnib | LLC | Interim CO 4 to PA 2_20200701 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | 4 | 4 | 40100 |
| 1078 | Lonafarnib | LLC | Letter for Authorization for PA 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | RRD INTERNATIONAL, | Letter for Authorization for RRD to | | | |
| 1079 | Lonafarnib | LLC | Commence Activities Under Work Order 11 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1080 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Master Independent Contractor Agreement | | | |
| 1081 | Lonafarnib | LLC | effective December 5, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Master Independent Contractor Agreement | | | |
| 1082 | Lonafarnib | LLC | effective February 22, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Master Independent Contractor Agreement | | | |
| 1083 | Lonafarnib | LLC | effective March 1, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Master Independent Contractor Agreement | | | |
| 1084 | Lonafarnib | LLC | effective May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | L | | Master Independent Contractor Agreement | | | |
| 1085 | Lonafarnib | LLC | effective May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 1000 | | RRD INTERNATIONAL, | Master Independent Contractor Agreement | ** | ** | |
| 1086 | Lonafarnib | LLC | effective October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 100= | | RRD INTERNATIONAL, | | *** | ** ** | ** |
| 1087 | Lonafarnib | LLC | Product Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1000 | T C '' | RRD INTERNATIONAL, | Project Agreement 1 Amendment 2 | #A AA | #A AA | |
| 1088 | Lonafarnib | LLC | ("Project Agreement 1 Amendment 2") | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|-----------------------|---|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | RRD INTERNATIONAL, | Project Agreement 1 to the Product | | | |
| 1089 | Lonafarnib | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Project Agreement 2 to the Product | | | |
| 1090 | Lonafarnib | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Project Agreement 3 to the Product | | | |
| 1091 | Lonafarnib | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Second Amendment to Task Order 1 to the | | | |
| 1092 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Second Amendment to the Amended and | | | |
| 1093 | Lonafarnib | LLC | Restated Work Order | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Second Amendment to Work Order No. 12 | | | |
| 1094 | Lonafarnib | LLC | to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Sixth Amendment to Task Order 1 to the | | | |
| 1095 | Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Statement of Work 1 to the Master | | | |
| 1096 | Lonafarnib | LLC | Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 1 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| 1097 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 1 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated February 22, | | | |
| 1098 | Lonafarnib | LLC | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 1 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated March 11, | | | |
| 1099 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Task Order 1 to the Master Independent | | | |
| 1100 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 1 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1101 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 10 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated effective | | | |
| 1102 | Lonafarnib | LLC | October 29, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 11 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1103 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 12 to the Master Independent | | | |
| | | | Contractor Agreement dated October 24, | | | |
| 1104 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 13 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1105 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 2 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | | 4 - | ± - | ± - |
| 1106 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 2 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated February 22, | | | |
| 1107 | Lonafarnib | LLC | 2019 | \$0.00 | \$0.00 | \$0.00 |
| 4400 | | RRD INTERNATIONAL, | Task Order 2 to the Master Independent | ** | | |
| 1108 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 2 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1109 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 3 to the Master Independent | _ | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| 1110 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|-----------------------|---|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | RRD INTERNATIONAL, | | | | |
| 1111 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 3 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1112 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Task Order 4 to the Master Independent | | | |
| 1113 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 4 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1114 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Task Order 5 to the Master Independent | | | |
| 1115 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 5 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1116 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Task Order 6 to the Master Independent | | | |
| 1117 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 6 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1118 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Task Order 7 to the Master Independent | | | |
| 1119 | Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 7 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1120 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 8 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1121 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Task Order 9 to the Master Independent | | | |
| | | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| 1122 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Third Amendment to Project Agreement 3 | | | |
| 1123 | Lonafarnib | LLC | dated March 1, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Third Amendment to Project Agreement | | | |
| 1124 | Lonafarnib | LLC | dated March 1, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | Third Amendment to Task Order 1 to the | | | |
| | | RRD INTERNATIONAL, | Master Independent Contractor Agreement | | | |
| 1125 | Lonafarnib | LLC | dated March 1, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | Third Amendment to the Amended and | | | |
| | | RRD INTERNATIONAL, | | | | |
| 1126 | Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | | | |
| 1127 | Lonafarnib | LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| I | L . T | RRD INTERNATIONAL, | | | | |
| 1128 | Lonafarnib | LLC | Work Order 19, April 22, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | L | RRD INTERNATIONAL, | | | | |
| 1129 | Lonafarnib | LLC | Work Order 19, August 28, 2023 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | 4 - | <u> </u> | |
| 1130 | Lonafarnib | LLC | Work Order 20, April 22, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | 4 - | <u> </u> | |
| 1131 | Lonafarnib | LLC | Work Order 20, April 22, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | W. 1 0 1 21 17 2 2022 | ** ** | ** * - | ** |
| 1132 | Lonafarnib | LLC | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | | ** | | |
| 1133 | Lonafarnib | LLC | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|-------------------------|---|-----------------|--------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| " | 113500 | RRD INTERNATIONAL, | Description of Contract of Lease | Tivexitiae Saic | Eambaa care | Curc |
| 1134 | Lonafarnib | LLC | Work Order 22, December 6, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | , , , , | ***** | * | , |
| 1135 | Lonafarnib | LLC | Work Order 5, effective March 31, 2016 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | -, | **** | * | * |
| 1136 | Lonafarnib | LLC | Work Order 5, effective March 31, 2016 | \$0.00 | \$0.00 | \$0.00 |
| | | | Work Order No. 10 provided for Strategic | | | |
| | | RRD INTERNATIONAL, | Regulatory Consultation and FDA Liaison | | | |
| 1137 | Lonafarnib | LLC | Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 12 to the Master Services | | | |
| 1138 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 13 to the Master Services | | | |
| 1139 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 14 to the Master Services | | | |
| 1140 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 19 to the Master Services | | | |
| 1141 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 2 to the Master Services | | | |
| 1142 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 20 to the Master Services | | | |
| 1143 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 21 to the Master Services | | | |
| 1144 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 22 to the Master Services | | | |
| 1145 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 24 to the Master Services | | | |
| 1146 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 4 to the Master Services | | | |
| 1147 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD INTERNATIONAL, | Work Order No. 5 to the Master Services | | | |
| 1148 | Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | RRD International, LLC; | | | | |
| 1149 | Lonafarnib | AltaSciences | Quotation Description | \$0.00 | · | 4 |
| 1150 | Lonafarnib | RTI INTERNATIONAL | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1151 | | SAFEBRIDGE | | #0.00 | #0.00 | . |
| 1151 | Avexitide | CONSULTANTS, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | SafeBridge Regulatory | | | | |
| 1152 | Lambda | and Life Sciences Group | Quotation 220503.0560 | \$0.00 | \$0.00 | \$0.00 |
| 1132 | Lamoua | and Life Sciences Group | Quotation 220303.0300 | \$0.00 | \$0.00 | \$0.00 |
| | | SafeBridge Regulatory | | | | |
| 1153 | Lonafarnib | and Life Sciences Group | Quotation 220503.0560 | \$0.00 | \$0.00 | \$0.00 |
| 1133 | Lonararmo | SAFEGUARD WORLD | Quotation 220303.0300 | ψ0.00 | ψ0.00 | ψ0.00 |
| | | INTERNATIONAL | | | | |
| 1154 | Lonafarnib | LIMITED | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1155 | Avexitide | S. B. I. [Name on File] | Sam Brown Proposal | \$0.00 | | |
| | | Sard Verbinnen & Co., | | \$3.00 | \$3.00 | \$2.00 |
| 1156 | Lambda | LLC | Engagement Letter | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | | 72.00 | 42.00 | 42.00 |
| 1157 | Lonafarnib | INTERNATIONAL B.V. | Change Control Note 1 | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | <u> </u> | 11111 | | |
| 1158 | Lonafarnib | INTERNATIONAL B.V. | Change Control Note 3 | \$0.00 | \$0.00 | \$0.00 |
| | | 1 | | Ţ | 40 | 45.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------|--|----------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | Change Control Note Number 1 | | | |
| | | | | | | |
| | | | Amendments To Eiger Contract To Reflect | | | |
| | | SCIENSUS | Agreed Logistics & Risk Arrangements on | | | |
| 1159 | Lonafarnib | INTERNATIONAL B.V. | Early Access | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | - | | | |
| 1160 | Lonafarnib | INTERNATIONAL B.V. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | | | | |
| 1161 | Lonafarnib | INTERNATIONAL B.V. | Letter of Distribution | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | | | | |
| 1162 | Lonafarnib | INTERNATIONAL B.V. | Main Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | SCIENSUS | | | | |
| 1163 | Lonafarnib | INTERNATIONAL B.V. | Pharmacovigilance Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Sciensus International | | | | |
| 1164 | Lonafarnib | B.V. | Schedule 1 to CCN 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 1165 | Lonafarnib | Sciensus International BV | Variation Letter | \$0.00 | \$0.00 | \$0.00 |
| | | Sciensus Rare | | | | |
| 1166 | Lonafarnib | International B.V. | Services Specification | \$0.00 | \$0.00 | \$0.00 |
| | | Scientific Committee of | | | | |
| | | General Hospital of | | | | |
| 1167 | Lonafarnib | Athens "Laiko" | GDPR Statement | \$0.00 | \$0.00 | \$0.00 |
| | | SECRETARY OF | | | | |
| 1168 | Lonafarnib | VETERAN AFFAIRS | Acknowledgement Letter | \$0.00 | \$0.00 | \$0.00 |
| | | SECRETARY OF | - | | | |
| 1169 | Lonafarnib | VETERAN AFFAIRS | Interim Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | SECRETARY OF | | | | |
| 1170 | Lonafarnib | VETERAN AFFAIRS | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | SENTYNL | | | | |
| 1171 | Lonafarnib | THERAPEUTICS, INC. | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | | SGS Analytics Germany | | | | |
| 1172 | Lonafarnib | GmbH | Quotation No. 210409-01 | \$0.00 | \$0.00 | \$0.00 |
| | | SGS Analytics Germany | | | | |
| 1173 | Lonafarnib | GmbH | Quotation No. 210514-01 | \$0.00 | \$0.00 | \$0.00 |
| | | SGS Analytics Germany | | | | |
| 1174 | Lonafarnib | GmbH | Quotation No. 210602-01 | \$0.00 | \$0.00 | \$0.00 |
| | | Shaare Zedek Medical | | | | |
| 1175 | Lambda | Center; Dr. Lurie Yoav | Form 4a | \$0.00 | \$0.00 | \$0.00 |
| 1176 | Avexitide | Sharp Corporation | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1177 | Lambda | Sharp Corporation | Quote dated 3/3/22 | \$0.00 | \$0.00 | \$0.00 |
| 1178 | Lambda | Sharp Corporation | Quote dated 8/11/21 | \$0.00 | \$0.00 | \$0.00 |
| 1179 | Lambda | Sharp Corporation | Quote re: Concept | \$0.00 | \$0.00 | \$0.00 |
| 1180 | Lambda | Sharp Corporation | Quote re: Groninger Tooling | \$0.00 | \$0.00 | \$0.00 |
| | | Sharp Global Design | | | | |
| 1181 | Lambda | Center | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Sheba Fund for Health | | | | |
| | | Services and Research | | | | |
| 1182 | Lambda | (R.A.) | Clinical Study Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Sheba Fund for Health | | | | |
| | | Services and Research | | | | |
| | | (R.A.); IQVIA RDS Israel | Letter of Indemnification re: Protocol No. | | | |
| 1183 | Lonafarnib | Ltd. | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| | | | Statement of Work re: ENDO 2022 - | | | |
| 1184 | Avexitide | Slate360, Inc. | Graphics Refresh | \$0.00 | \$0.00 | \$0.00 |
| | | | Quote N21-11679 re: Screening and | | | |
| | - | i | Quantitation of Nitrosamine DIPNA | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|--------------|----------------------|--|---|------------------|------------------|------------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 1186 | Lonafarnib | Solvias AG | Quote N21-11680 re: Lonafarnib API | \$0.00 | \$0.00 | \$0.00 |
| 1187 | Lonafarnib | Solvias AG | Quote N21-11681 re: Lonafarnib API | \$0.00 | \$0.00 | \$0.00 |
| 1188 | Lonafarnib | Solvias AG | Quote N21-12606 re: Lonafarnib API | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 4400 | | Soroka Medical Center; | | | ** ** | |
| 1189 | Lambda | Dr. Etzion Ohad | Form 4a | \$0.00 | \$0.00 | \$0.00 |
| 1100 | T 1.1 | Soroka University | | ¢0.00 | Φ0.00 | #0.00 |
| 1190 1191 | Lambda Lonafarnib | Medical Center Southwood Research | Clinical Trial Agreement Proposal | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| 1171 | Lonaranno | Southwood Research | Порова | \$0.00 | \$0.00 | \$0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1192 | Lambda | - GLOBAL, LLC | Amendment #1 to Work Order 1 LMD001 | \$0.00 | \$0.00 | \$0.00 |
| | | , | | | | |
| | | SPRI CLINICAL TRAILS | | | | |
| 1193 | Lambda | - GLOBAL, LLC | Amendment #3 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 1104 | T1. 1. | SPRI CLINICAL TRAILS | A second access 2 to Wester Online 1 | \$0.00 | \$0.00 | \$0.00 |
| 1194 | Lambda | - GLOBAL, LLC | Amendment 2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1195 | Lambda | - GLOBAL, LLC | Amendment 5 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| 1170 | Zumio um | 0202:12, 220 | a monument of the system of the s | \$0.00 | \$0.00 | ψ0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1196 | Lambda | - GLOBAL, LLC | Amendment to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | SPRI CLINICAL TRAILS | | | | |
| 1197 | Lambda | - GLOBAL, LLC | Statement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1198 | Lonafarnib | - GLOBAL, LLC | Amendment #1 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| 1170 | Zonaranio | 0202:12, 220 | | φοισσ | \$0.00 | \$0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1199 | Lonafarnib | - GLOBAL, LLC | Amendment #1 to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 1200 | T 6 '1 | SPRI CLINICAL TRAILS | | #0.00 | Φ0.00 | 00.00 |
| 1200 | Lonafarnib | - GLOBAL, LLC | Amendment #2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | | SPRI CLINICAL TRAILS | | | | |
| 1201 | Lonafarnib | - GLOBAL, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | 4 | 40100 | ***** |
| | | SPRI CLINICAL TRAILS | | | | |
| 1202 | Lonafarnib | - GLOBAL, LLC | Statement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | CDDI CLERTICIT | | | | |
| 1202 | T 1 | SPRI CLINICAL TRAILS | Ctatana at a CW at I NEOOA | Φ0.00 | # 0.00 | # 0.00 |
| 1203 | Lonafarnib | - GLOBAL, LLC Stanford University - | Statement of Work LNF004 | \$0.00 | \$0.00 | \$0.00 |
| | | Office of Technology | | | | |
| 1204 | Avexitide | Licensing | Services Agreement | \$1,116.00 | \$0.00 | \$0.00 |
| | - 1 . 5 | | Quote re: SG-Lab Mechanical or | ψ1,110.00 | φ0.00 | ψ0.00 |
| 1205 | Avexitide | Stevanato Group S.p.A. | Performance Test | \$0.00 | \$0.00 | \$0.00 |
| 1206 | Lambda | Syneos | Statement of Work Dated August 7, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Syneos Health | | | | |
| | | Communications, Inc. | | | | |
| | | f/k/a inVentiv Health | | | | |
| 1207 | Lonafarnib | Public Relations, LLC | Statement of Work re: Submission | \$0.00 | \$0.00 | \$0.00 |
| 1208 | Lonafarnib | The Aga Khan University | Data Transfer Agreement V | \$0.00 | \$0.00 | \$0.00 |
| 1200 | Lonaranno | The Aga Khali Ulliveisity | Data Halistel Agreement v | \$0.00 | Φ 0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------------|---|-------------------|---------------|-------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | 120000 | Convener Country purity | Letter re: Change to the Data Transfer | TIVE MICHAEL CALL | | |
| 1209 | Lonafarnib | The Aga Khan University | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | The Children's Hospital of | | | | |
| 1210 | Avexitide | Philadelphia | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | THE CHILDRENG | | | | |
| | | THE CHILDRENS HOSPITAL | | | | |
| | | CORPORATION, D/B/A | | | | |
| | | BOSTON CHILDRENS | | | | |
| 1211 | Lonafarnib | HOSPITAL | Data Sharing Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | THE CHILDRENS | 5 6 | | * | * |
| | | HOSPITAL OF | | | | |
| 1212 | Avexitide | PHILADELPHIA | Indemnification Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | The Doctor's Laboratory | | | | |
| 1213 | Lambda | Limited | Change Order 1 NZ | \$0.00 | \$0.00 | \$0.00 |
| | | The Doctor's Laboratory | et e t t pyr | ** | | |
| 1214 | Lambda | Limited | Change Order 1 PK | \$0.00 | \$0.00 | \$0.00 |
| 1215 | Lambda | The Doctor's Laboratory | Change Order 2 ID | \$0.00 | \$0.00 | ¢0.00 |
| 1215 | Lambda | Limited The Doctor's Laboratory | Change Order 2 IR | \$0.00 | \$0.00 | \$0.00 |
| 1216 | Lambda | Limited | Change Order 2 NZ | \$0.00 | \$0.00 | \$0.00 |
| 1210 | Lamoda | The Doctor's Laboratory | Change Order 2 142 | φυ.00 | \$0.00 | \$0.00 |
| 1217 | Lambda | Limited | Change Order 2 PK | \$0.00 | \$0.00 | \$0.00 |
| | | The Doctor's Laboratory | <u> </u> | Ţ3.00 | 44300 | 44300 |
| 1218 | Lambda | Limited | Pricing Agreement NZ | \$0.00 | \$0.00 | \$0.00 |
| | | The Doctor's Laboratory | | | | |
| 1219 | Lambda | Limited | Pricing Agreement PK | \$0.00 | \$0.00 | \$0.00 |
| | | The Doctor's Laboratory | | | | |
| 1220 | Lonafarnib | Limited | Pricing / Supplemental Agreement | \$0.00 | \$0.00 | |
| 1001 | T C 1 | The Doctor's Laboratory | D:: /G 11 | #0.00 | \$0.00 | |
| 1221 | Lonafarnib | Limited The Doctor's Laboratory | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| 1222 | Lonafarnib | Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| 1222 | Eonaramio | The Doctor's Laboratory | Themg/supplemental Agreement | \$0.00 | φ0.00 | |
| 1223 | Lonafarnib | Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| | | The Doctor's Laboratory | Pricing/Supplemental Agreement dated 10th | Ţ3.00 | 44300 | |
| 1224 | Lonafarnib | Limited | September 2018 | \$0.00 | \$0.00 | |
| | | | Pricing/Supplemental Agreement dated 14 | | | |
| 1225 | Lonafarnib | Limited | September 2018 | \$0.00 | \$0.00 | |
| | | The Doctor's Laboratory | Pricing/Supplemental Agreement dated 14 | | | |
| 1226 | Lonafarnib | Limited | September 2018 | \$0.00 | \$0.00 | \$10,000.00 |
| 1007 | T C 1 | The Doctor's Laboratory | Pricing/Supplemental Agreement dated 18th | 40.00 | # 0.00 | \$10,000.00 |
| 1227 | Lonafarnib | Limited The Destaria Laboratory | January 2017 | \$0.00 | \$0.00 | |
| 1228 | Longfornih | The Doctor's Laboratory | Pricing/Supplemental Agreement dated 18th | \$0.00 | \$0.00 | |
| 1220 | Lonafarnib | Limited The Doctor's Laboratory | January 2017 Pricing/Supplemental Agreement dated | \$0.00 | \$0.00 | |
| 1229 | Lonafarnib | Limited | 2021 | \$0.00 | \$0.00 | |
| 1227 | | The Doctor's Laboratory | Pricing/Supplemental Agreement dated | φ0.00 | ψ0.00 | |
| 1230 | Lonafarnib | Limited | December 2014 | \$0.00 | \$0.00 | |
| | | The Doctor's Laboratory | | | | |
| 1231 | Lonafarnib | Limited | Pricing/Supplemental Agreement Trials | \$0.00 | \$0.00 | |
| | | The Doctor's Laboratory | | | | |
| 1232 | Lonafarnib | Limited | Trials Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| | | The Doctor's Laboratory | | | | |
| 1233 | Lonafarnib | Limited | Trials Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |

| | Related | | | | | Lonafarnib |
|------|------------|--|--|----------------|---------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | The Johns Hopkins | | | | |
| 1234 | Lonafarnib | University | Sponsored Research Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | The Liver Center and | | | | |
| 1235 | Lonafarnib | Naranjargal Dashdorj | ADDITIONAL BUDGET AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| | | The Liver Unit Hadassah | | | | |
| | | University Hospital, Ein | | | | ** ** |
| 1236 | Lambda | Kerem | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| | | The Liver Unit Hadassah | | | | |
| | | University Hospital; The | | | | |
| | | Research Fund of | | | | |
| | | Hadassah Medical | | | | |
| 1237 | Lambda | Organization (R.A.) | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| | | The Progeria Research | | | | |
| 1238 | Lambda | Foundation | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| | | The Progeria Research | | | | |
| 1239 | Lonafarnib | Foundation | Agreement to Stay Arbitration | \$0.00 | \$0.00 | \$0.00 |
| | | The Progeria Research | Amended and Restated Collaboration and | | | |
| 1240 | Lonafarnib | Foundation | Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1011 | | The Progeria Research | Amendment No. 1 to the Collaboration and | 00.00 | 40.00 | |
| 1241 | Lonafarnib | Foundation | Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1242 | Lonafarnib | The Progeria Research Foundation | Callahanatian and Swamby Assessment | \$0.00 | \$0.00 | \$0.00 |
| 1242 | Lonatarnio | The Progeria Research | Collaboration and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1243 | Lonafarnib | Foundation | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| 1243 | Lonaranno | The Progeria Research | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| 1244 | Lonafarnib | Foundation | CTS Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | The Progeria Research | | ***** | 40100 | ***** |
| 1245 | Lonafarnib | Foundation | Tern Sheet | \$0.00 | \$0.00 | \$0.00 |
| | | The Progeria Research | | | | |
| 1246 | Lonafarnib | Foundation, Inc. | Collaboration and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | The University of Texas | | | | |
| | | MD Anderson Cancer | | | | |
| 1247 | Lonafarnib | Center | Drug Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1240 | · | The District of the Control of the C | 0 | 00.00 | # 0.00 | ФО ОО |
| 1248 | Lambda | Thermo Fisher Scientific Thermo Fisher Scientific. | Quotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| 1249 | Lonafarnib | Inc. | Quote re: Sourcing of Medicinal Product | \$0.00 | \$0.00 | \$0.00 |
| 1249 | Lonaranno | Thermo Fisher Scientific, | Quote re. Sourcing of Wedicinal Froduct | \$0.00 | \$0.00 | \$0.00 |
| | | Inc.; Fisher Clinical | | | | |
| 1250 | Lonafarnib | Services | Quote | \$0.00 | \$0.00 | \$0.00 |
| | † | | | **** | * | * |
| | | Thermo Fisher Scientific; | | | | |
| | | Patheon UK Limited, Part | | | | |
| | | of Thermo Fisher | | | | |
| 1251 | Lambda | Scientific | Quotation #200616-01-SF | \$0.00 | \$0.00 | \$0.00 |
| | | The same of First and Control of Con- | | | | |
| | | Thermo Fisher Scientific; Patheon UK Limited, Part | | 1 | | |
| | | of Thermo Fisher | | | | |
| 1252 | Lambda | Scientific | Ouotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| | | | | \$3.00 | \$0.00 | \$0.00 |
| | | Thermo Fisher Scientific; | | 1 | | |
| | | Patheon UK Limited, Part | | | | |
| | | of Thermo Fisher | | | | |
| 1253 | Lambda | Scientific | Quotation #220328-01-SF | \$0.00 | \$0.00 | \$0.00 |

| Asset | | Related | | | | | Lonafarnib |
|--|------|------------|-------------------------------|--|----------------|--------------|---------------|
| Total Transport Netherlands Hub Freight Quote dated 06 So.00 S | # | | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | |
| | | 125500 | | | | | Caro |
| Total Transport Netherlands Hub Freight Quote dated 15 S0.00 S | 1254 | Lambda | | | \$0.00 | \$0.00 | \$0.00 |
| 1255 Lambda Management March 2022 S0.00 S0.00 S0.00 S0.00 1256 Lambda Management 4/10/2012 S0.00 S0.00 S0.00 S0.00 1257 Lambda Management 4/10/2012 S0.00 S0.00 S0.00 S0.00 1258 Lambda Management 13, 2021 Netherlands Hub Freight Quote dated Dec S0.00 S0.00 S0.00 S0.00 1258 Lambda Management 14, 2022 S0.00 S0.00 S0.00 S0.00 1259 Lambda Management 14, 2022 S0.00 S0.00 S0.00 1259 Lambda Management 27, 2012 S0.00 S0.00 S0.00 1260 Lambda Management 27, 2012 S0.00 S0.00 S0.00 1261 Lambda Management 27, 2012 S0.00 S0.00 S0.00 1261 Lambda Management 27, 2012 S0.00 S0.00 S0.00 1261 Lambda Management 27, 2011 S0.00 S0.00 S0.00 1262 Lonafarnib Management 27, 2011 S0.00 S0.00 S0.00 1263 Lonafarnib Management 27, 2011 S0.00 S0.00 S0.00 1264 Lonafarnib LC Statement of Work dated August 12, 2021 S0.00 S0.00 S0.00 1265 Lonafarnib LC Statement of Work dated August 12, 2021 S0.00 S0.00 S0.00 1266 Lonafarnib LC Statement of Work dated August 12, 2021 S0.00 S0.00 S0.00 1268 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-put-22 S0.00 S0.00 S0.00 1270 Lonafarnib LC Statement of Work dated October 29, 2019 S0.00 S0.00 S0.00 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-put-22 S0.00 S0.00 S0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-put-22 S0.00 S0.00 S0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1276 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-put-22 S0.00 S0.00 S0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Price Quotation dated 20-put-22 S0.00 S0.00 S0.00 1278 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 | | | | * | 40100 | 40100 | 40100 |
| Total Transport Netherlands Hub Freight Quote dated So.00 So.0 | 1255 | Lambda | _ | 9 1 | \$0.00 | \$0.00 | \$0.00 |
| 1256 Lambda Management 4/10/2022 S0.00 S0. | | | · · | Netherlands Hub Freight Ouote dated | ***** | * | * |
| 1257 Lambda | 1256 | Lambda | _ | 9 1 | \$0.00 | \$0.00 | \$0.00 |
| 1257 Lambda Management 13, 2021 So.00 So.0 | | | | | ***** | * | * |
| Lambda | 1257 | Lambda | • | | \$0.00 | \$0.00 | \$0.00 |
| 1258 | | | | · · | | | |
| Total Transport Netherlands Hub Freight Quote dated Jan | 1258 | Lambda | - | 14, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lambda Management 27, 2022 S0.00 S0.00 S0.00 S0.00 S0.00 | | | _ | Netherlands Hub Freight Quote dated Jan | | | |
| 1260 Lambda Management 27, 2021 S0.00 \$0.00 \$0.00 1261 Lambda Management 27, 2021 S0.00 \$0.00 \$0.00 1262 Lonafamib Management Metherlands Hub Freight Quote dated S0.00 \$0.00 \$0.00 1263 Lonafamib Management August 18, 2021 S0.00 \$0.00 \$0.00 1264 Lonafamib Total Transport Netherlands Hub Freight Quote dated S0.00 \$0.00 \$0.00 1265 Lonafamib LLC Total Transport Netherlands Hub Freight Quote dated Nov 2, 201 S0.00 \$0.00 \$0.00 1265 Lonafamib LLC Statement of Work S0.00 \$0.00 \$0.00 1266 Lonafamib LLC Statement of Work dated August 12, 2021 S0.00 \$0.00 \$0.00 1266 Lonafamib International LLC Statement of Work dated August 12, 2021 S0.00 \$0.00 \$0.00 1267 Lonafamib International LLC Statement of Work dated October 29, 2019 S0.00 \$0.00 \$0.00 1268 Lambda TransPerfect Translations International LLC Statement of Work dated October 29, 2019 S0.00 \$0.00 \$0.00 1270 Lonafamib International LLC Statement of Work dated October 29, 2019 S0.00 S0.00 \$0.00 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 S0.00 S0.00 \$0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1276 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 S0.00 S0.00 S0.00 1277 Lonafamib Trialog Clinical Trials Ltd Statement of Work dated April 2020 S0.00 S0.00 S0.00 1278 Lonafamib Trialog Clinical Trials Ltd Statement of Work dated Dure 7, 2021 S0.00 S0.00 S0.00 1278 Lonafamib Trialog Clinical Tri | 1259 | Lambda | _ | 27, 2022 | \$0.00 | \$0.00 | \$0.00 |
| 1261 Lambda Management 27, 2021 S0.00 \$0.00 \$0.00 \$0.00 | | | Total Transport | Netherlands Hub Freight Quote dated June | | | |
| 1261 Lambda Management 27, 2021 S0.00 S0.00 S0.00 S0.00 1262 Lonafarnib | 1260 | Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 1262 | | | Total Transport | Netherlands Hub Freight Quote dated June | | | |
| 1262 Lonafarnib Management August 18, 2021 \$0.00 \$0.00 \$0.00 | 1261 | Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 1263 | | | Total Transport | Netherlands Hub Freight Quote dated | | | |
| 1263 | 1262 | Lonafarnib | Management | August 18, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 1264 Lonafarnib LLC Statement of Work S0.00 | | | Total Transport | Netherlands Hub Freight Quote dated Nov 2, | | | |
| 1264 Lonafarnib LLC Statement of Work S0.00 \$0.00 \$0.00 | 1263 | Lonafarnib | Management | 2021 | \$0.00 | \$0.00 | \$0.00 |
| TransPerfect International LLC Statement of Work dated August 12, 2021 \$0.00 \$ | | | TransPerfect International | | | | |
| 1265 Lonafarnib LLC | 1264 | Lonafarnib | LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 1266 Lonafarnib TransPerfect Translations International LLC Statement of Work dated June 7, 2019 \$0.00 \$ | | | TransPerfect International | | | | |
| 1266 Lonafarnib International LLC Statement of Work dated June 7, 2019 \$0.00 \$0.00 \$0.00 | 1265 | Lonafarnib | LLC | Statement of Work dated August 12, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 1266 Lonafarnib International LLC Statement of Work dated June 7, 2019 \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1267 Lonafarnib TransPerfect Translations International LLC Statement of Work dated October 29, 2019 \$0.00 \$0.00 \$0.00 \$0.00 \$1.00 | | | | | | | |
| 1267 Lonafarnib International LLC Statement of Work dated October 29, 2019 \$0.00 \$0.00 \$0.00 | 1266 | Lonafarnib | International LLC | Statement of Work dated June 7, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 1267 Lonafarnib International LLC Statement of Work dated October 29, 2019 \$0.00 \$0.00 \$0.00 | | | m | | | | |
| TRANSPERFECT TRANSLATIONS INTERNATIONAL, INC. Solutidns Agreement \$0.00 \$0 | 1067 | T C 11 | | G | 00.00 | #0.00 | # 0.00 |
| TRANSLATIONS INTERNATIONAL, INC. Solutidns Agreement S0.00 \$0.00 \$0.00 \$0.00 | 1267 | Lonafarnib | | Statement of Work dated October 29, 2019 | \$0.00 | \$0.00 | \$0.00 |
| 1268 Lambda INTERNATIONAL, INC. Solutions Agreement \$0.00 \$0.00 \$0.00 \$0.00 | | | | | | | |
| TRANSPERFECT TRANSLATIONS Deliverables 1 PDF File1 Certificate of \$0.00 \$0.00 \$0.00 \$0.00 | 1260 | T 1. 1. | | Color di dina di Anno and | £0.00 | \$0.00 | ¢0.00 |
| TRANSLATIONS INTERNATIONAL, INC. TRG Communications LLC Services Agreement \$0.00 \$0.00 \$36,000,00 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1268 | Lambda | - | Solutions Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1269 Lonafarnib INTERNATIONAL, INC. Translation \$0.00 \$0.00 \$0.00 1270 Lonafarnib LLC Services Agreement \$0.00 \$0.00 \$36,000.00 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$0.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1279 Lonafarnib Trialog Clinical Trials Ltd \$0.00 \$0.00 1270 Lonafarnib Trialog Clinical Trials Ltd \$0.00 \$0.00 1270 Lonafarnib Trialog Clinical Trials Ltd \$0.00 \$0.00 1270 Lonafarnib Trialog Clinical Trials Ltd \$0.00 \$0.00 \$0.00 1271 Lonafarnib | | | | Deliverables 1 DDE Eile1 Centificate of | | | |
| 1270 Lonafarnib LLC Services Agreement \$0.00 \$0.00 \$36,000.00 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$0.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd \$0.00 \$0.00 1 | 1260 | Lanafamila | | | 20.00 | \$0.00 | \$0.00 |
| Lonafarnib LLC Services Agreement \$0.00 \$0.00 \$36,000.00 | 1209 | Lonaranno | | Translation | \$0.00 | \$0.00 | \$0.00 |
| 1271 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$0.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1270 | Lanafamih | | Saminas Agraamant | \$0.00 | ¢0.00 | \$26,000,00 |
| 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$3,760.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 | 1270 | Lonatarnio | LLC | Services Agreement | \$0.00 | \$0.00 | \$36,000.00 |
| 1272 Lambda Trialog Clinical Trials Ltd Price Quotation dated 20-Jul-22 \$0.00 \$0.00 1273 Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$3,760.00 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 | 1271 | Lambda | Triolog Clinical Triols I td | Price Ouetation dated 20 Jul 22 | \$0.00 | | 00.02 |
| Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$3,760.00 \$ | 12/1 | Lamoua | Thatog Chilical Thats Ltu | Tice Quotation dated 20-Jui-22 | \$0.00 | | \$0.00 |
| Lambda Trialog Clinical Trials Ltd Price Quotation dated 21-Feb-23 \$0.00 \$3,760.00 \$ | 1272 | Lambda | Trialog Clinical Trials I td. | Price Ouotation dated 20 Jul 22 | \$0.00 | | \$0.00 |
| 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 12/2 | Lamoda | Thatog Chilical Thats Ltd | Tree Quotation dated 20-Jui-22 | \$0.00 | | \$0.00 |
| 1274 Lambda Trialog Clinical Trials Ltd Price Quotation dated 3-Mar-22 \$0.00 \$0.00 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1273 | Lambda | Trialog Clinical Trials I td. | Price Quotation dated 21 Feb 23 | \$0.00 | | \$0.00 |
| 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1273 | Lamoda | Thaiog Chilical Thais Liu | Thee Quotation dated 21-1-60-25 | \$0.00 | \$3,760.00 | \$0.00 |
| 1275 Lambda Trialog Clinical Trials Ltd Price Quotation dated 7-Dec-22 \$0.00 \$0.00 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$0.00 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1274 | L ambda | Trialog Clinical Trials Ltd. | Price Quotation dated 3-Mar-22 | \$0.00 | | \$0.00 |
| 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 \$0.00 \$0.00 \$1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$ | 1271 | Eumoua | Thurs Chinear Thuis Eta | The Qualition dated 5 Wait 22 | φ0.00 | | ψο.σσ |
| 1276 Lambda Trialog Clinical Trials Ltd Statement of Work dated April 2020 \$0.00 \$0.00 \$0.00 \$0.00 \$1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 \$ | 1275 | L ambda | Trialog Clinical Trials Ltd. | Price Quotation dated 7-Dec-22 | \$0.00 | | \$0.00 |
| 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 | 1273 | Eumoua | Thurs Chinear Thuis Eta | Thee Quotation dated / Bee 22 | φ0.00 | | ψ0.00 |
| 1277 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated 2019 \$0.00 | 1276 | Lambda | Trialog Clinical Trials Ltd | Statement of Work dated April 2020 | \$0.00 | | \$0.00 |
| 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | -2,0 | | Eta | ·· | \$2.00 | | ψ3.00 |
| 1278 Lonafarnib Trialog Clinical Trials Ltd Statement of Work dated June 7, 2021 \$0.00 \$0.00 \$0.00 | 1277 | Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | 5 200 | | 72.00 | 42.30 | 72.00 |
| | 1278 | Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated June 7, 2021 | \$0.00 | \$0.00 | \$0.00 |
| 1279 Lonafarnib Trialog Clinical Trials Ltd Trialog Clinical Trials Quote 1 \$0.00 \$0.00 | | | | , | | | |
| | 1279 | Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote 1 | \$0.00 | \$0.00 | \$0.00 |

| Asset | | Related | | | | | Lonafarnib |
|---|------|------------|-----------------------------|---|----------------|-------------|------------|
| 1280 Lonafarmib Trialog Clinical Trials Ltd Trialog Clinical Trials Quote 2 \$0.00 \$0.00 \$0.00 \$0.00 \$1.00 | # | | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | |
| 1281 Lonafarmib Trialog Clinical Trials Lot Trialog Clinical Trials Quote dated So.00 So.0 | | risset | Contract Counterparty | Description of Contract of Lease | Aveature cure | Lambda Curc | Curc |
| 1281 Lonafarmib Trialog Clinical Trials Lot Trialog Clinical Trials Quote dated So.00 So.0 | 1280 | Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Ouote 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafamib Trialog Clinical Trials Ltd 1/30/2021 #3 S0.00 S0.00 S0.00 S0.00 | | | 8 | | 44444 | 40.00 | ***** |
| Trialog Clinical Trials Quote dated | 1281 | Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| 1282 Lonafarnib Trialog Clinical Trials Ltd 1/30/2021 #2 S0.00 S0.00 S0.00 S0.00 1284 Lonafarnib Trialog Clinical Trials Trialog Clinical Trials Consultation | | | | | | | |
| Trialog Clinical Trials Ltd Trialog Clinical Trials Ltd Trialog Clinical Trials Quote dated So.00 So | 1282 | Lonafarnib | Trialog Clinical Trials Ltd | - | \$0.00 | \$0.00 | \$0.00 |
| 1284 Lonafamib Trialog Clinical Trials Ltd Trialog Clinical Trials Quote dated | | | | | | | |
| | 1283 | Lonafarnib | Trialog Clinical Trials Ltd | 11/30/2021 #3 | \$0.00 | \$0.00 | \$0.00 |
| 1285 | | | | Trialog Clinical Trials Quote dated | | | |
| 1285 | 1284 | Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Trialog Clinical Trials Ltd 2021 S0.00 | | | | Trialog Clinical Trials Quote dated Jan 24, | | | |
| 1286 | 1285 | Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Trialog Clinical Trials Ltd 2021 | | | | | | | |
| 1287 Lonafarnib Trialog Clinical Trials Ltd 2021 S0.00 \$0.00 \$0.00 \$0.00 \$1.288 Lonafarnib LLC Services Agreement S0.00 \$0 | 1286 | Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Triangle Biostatistics, Work Order: ElG-300 Under Master S0.00 | | | | | | | |
| Lambda | 1287 | Lonafarnib | | | \$0.00 | \$0.00 | \$0.00 |
| 1289 | | | | | | | |
| 1290 | | | | | | · · | \$0.00 |
| 1290 Avexitide LLC | 1289 | Lambda | | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Triangle Insights Group, LLC Master Services Agreement S0.00 | | | | | | | |
| 1291 Lambda LLC | 1290 | Avexitide | | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1292 Lonafarnib TriangleInsights Statement of Work \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$1.00 \$ | | | | | | | |
| 1293 | | | | _ | | · | · |
| 1293 | 1292 | Lonafarnib | | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| TRILIGENT INTERNATIONAL Consulting Agreement S0.00 \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1294 Lonafarnib INTERNATIONAL Consulting Agreement \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$1.00 \$ | 1293 | Lambda | | Project Assignment 5-1 | \$0.00 | \$0.00 | \$0.00 |
| TRILIGENT INTERNATIONAL Project Assignment 3 \$0.00 \$0.00 \$0.00 \$0.00 \$1.00 | | | | | | | |
| 1295 | 1294 | Lonafarnib | | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1296 Lonafarnib INTERNATIONAL Project Assignment 5 \$0.00 \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1296 Lonafarnib INTERNATIONAL Project Assignment 5 \$0.00 \$0.00 \$0.00 | 1295 | Lonafarnib | | Project Assignment 3 | \$0.00 | \$0.00 | \$0.00 |
| Trustees of the University of Pennsylvania ("Penn") and the Children's Open Market Sale Agreement \$0.00 \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1297 Avexitide Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | 1296 | Lonafarnib | INTERNATIONAL | Project Assignment 5 | \$0.00 | \$0.00 | \$0.00 |
| 1297 Avexitide Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | The Colonia is | | | | |
| 1297 Avexitide | | | | | | | |
| 1297 Avexitide | | | | | | | |
| Trustees of the University of Pennsylvania ("Penn") and the Children's Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0. | 1207 | A141 1. | | O Marlant Sala A | 00.00 | \$0.00 | 90.00 |
| 1298 Lambda Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | 1297 | Avexitide | Hospital of Philadelphia | Open Market Sale Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1298 Lambda Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | Trustees of the University | | | | |
| 1298 Lambda Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1298 Lambda Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | | | | | |
| Trustees of the University of Pennsylvania ("Penn") and the Children's Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0. | 1298 | I ambda | | Onen Market Sale Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1299 Lonafarnib Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 \$0.00 | 1270 | Lamoda | 1103pital of 1 iniaccipina | open warket bate Agreement | ψ0.00 | ψ0.00 | ψ0.00 |
| 1299 Lonafarnib Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 \$0.00 | | | Trustees of the University | | | | |
| 1299 Lonafarnib Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | | | | | |
| 1299 Lonafarnib Hospital of Philadelphia Open Market Sale Agreement \$0.00 \$0.00 \$0.00 | | | | | | | |
| Statement of Work 1 to the Master | 1299 | Lonafarnib | Hospital of Philadelphia | Open Market Sale Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1300 Lonafarnib TWO LABS, LLC Independent Contractor Agreement \$0.00 \$0.00 \$0.00 1301 Lonafarnib TWO LABS, LLC Statement of Work 2 \$0.00 \$0.00 \$0.00 1302 Lonafarnib TWO LABS, LLC Statement of Work 3 \$0.00 \$0.00 \$0.00 1303 Lonafarnib TWO LABS, LLC Statement of Work 4 \$0.00 \$0.00 \$0.00 1304 Lambda Tx Evidence, Inc SAP Statistical Analysis Plan 3 \$0.00 \$0.00 \$0.00 1305 Lonafarnib Tx Evidence, Inc Master Service Agreement \$0.00 \$0.00 \$0.00 1306 Lonafarnib Tx Evidence, Inc SAP Statistical Analysis Plan 1 \$0.00 \$0.00 \$0.00 | | 1 | | | | | |
| 1301 Lonafarnib TWO LABS, LLC Statement of Work 2 \$0.00 \$0.00 \$0.00 1302 Lonafarnib TWO LABS, LLC Statement of Work 3 \$0.00 \$0.00 \$0.00 1303 Lonafarnib TWO LABS, LLC Statement of Work 4 \$0.00 \$0.00 \$0.00 1304 Lambda Tx Evidence, Inc SAP Statistical Analysis Plan 3 \$0.00 \$0.00 \$0.00 1305 Lonafarnib Tx Evidence, Inc Master Service Agreement \$0.00 \$0.00 \$0.00 1306 Lonafarnib Tx Evidence, Inc SAP Statistical Analysis Plan 1 \$0.00 \$0.00 \$0.00 | 1300 | Lonafarnib | TWO LABS, LLC | | \$0.00 | \$0.00 | \$0.00 |
| 1302 Lonafarnib TWO LABS, LLC Statement of Work 3 \$0.00 \$0.00 \$0.00 1303 Lonafarnib TWO LABS, LLC Statement of Work 4 \$0.00 \$0.00 \$0.00 1304 Lambda Tx Evidence, Inc SAP Statistical Analysis Plan 3 \$0.00 \$0.00 \$0.00 1305 Lonafarnib Tx Evidence, Inc Master Service Agreement \$0.00 \$0.00 \$0.00 1306 Lonafarnib Tx Evidence, Inc SAP Statistical Analysis Plan 1 \$0.00 \$0.00 \$0.00 | 1301 | Lonafarnib | | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| 1303 Lonafarnib TWO LABS, LLC Statement of Work 4 \$0.00 \$0.00 \$0.00 1304 Lambda Tx Evidence, Inc SAP Statistical Analysis Plan 3 \$0.00 \$0.00 \$0.00 1305 Lonafarnib Tx Evidence, Inc Master Service Agreement \$0.00 \$0.00 \$0.00 1306 Lonafarnib Tx Evidence, Inc SAP Statistical Analysis Plan 1 \$0.00 \$0.00 \$0.00 | 1302 | Lonafarnib | TWO LABS, LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| 1305LonafarnibTx Evidence, IncMaster Service Agreement\$0.00\$0.00\$0.001306LonafarnibTx Evidence, IncSAP Statistical Analysis Plan 1\$0.00\$0.00\$0.00 | 1303 | Lonafarnib | TWO LABS, LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| 1306 Lonafarnib Tx Evidence, Inc SAP Statistical Analysis Plan 1 \$0.00 \$0.00 \$0.00 | 1304 | Lambda | Tx Evidence, Inc | SAP Statistical Analysis Plan 3 | \$0.00 | \$0.00 | \$0.00 |
| | 1305 | Lonafarnib | Tx Evidence, Inc | | \$0.00 | \$0.00 | \$0.00 |
| | 1306 | Lonafarnib | Tx Evidence, Inc | | \$0.00 | \$0.00 | \$0.00 |
| | 1307 | Lonafarnib | Tx Evidence, Inc | SAP Statistical Analysis Plan 2 | \$0.00 | \$0.00 | \$0.00 |

| | Related | | | | | Lonafarnib |
|------|------------|---------------------------|---|---|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 1308 | Lonafarnib | Tx Evidence, Inc | Term of Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1500 | Editaranio | TA Bytachee, me | Eiger Study EIG-LMD-001 Submission | ψ0.00 | ψ0.00 | ψ0.00 |
| 1309 | Lambda | Tx Evidence, Inc. | New Work Proposal | \$0.00 | \$0.00 | \$0.00 |
| 1310 | Lambda | Tx Evidence, Inc. | Statical Data Analysis Proposal | \$0.00 | \$0.00 | \$0.00 |
| | | | Eiger Study EIG-300 Submission New | 4 | + 4 1 1 0 1 | 40100 |
| 1311 | Lonafarnib | Tx Evidence, Inc. | Work Proposal | \$0.00 | \$0.00 | \$0.00 |
| | | United Biosource LLC | 1 | * | * | * |
| 1312 | Lonafarnib | (UBC) | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | University Health | - | | | |
| 1313 | Lambda | Network | Data Use Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | - | | | |
| | | University Medical Center | | | | |
| | | of Johannes Gutenberg- | | | | |
| 1314 | Lambda | University Mainz | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| | | University of Texas MD | First Amendment to Drug Supply | | | |
| 1315 | Lonafarnib | Anderson Cancer Center | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | US BIOSERVICES | First Amendment to Statement of Work No. | | | |
| 1316 | Lonafarnib | CORPORATION | 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | US BIOSERVICES | | | | |
| 1317 | Lonafarnib | CORPORATION | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | US BIOSERVICES | Second Amendment to Statement of Work | | | |
| 1318 | Lonafarnib | CORPORATION | No. 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | US BIOSERVICES | | | | |
| 1319 | Lonafarnib | CORPORATION | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | US BIOSERVICES | | | | |
| 1320 | Lonafarnib | CORPORATION | Specialty Pharmacy Provider Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | Consent for Photography / Authorization for | | | |
| 1321 | Avexitide | Vito Vento | Publication | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 1322 | Lonafarnib | Walgreens Boots Alliance | Global Estimated Quotation | \$0.00 | \$0.00 | \$0.00 |
| 1323 | Lonafarnib | WCCT GLOBAL, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| 1324 | Avexitide | WCG Clinical Inc. | Services Agreement | \$0.00 | \$0.00 | \$1,312.50 |
| 1325 | Lonafarnib | WOVEN DATA, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1326 | Lonafarnib | WOVEN DATA, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| 1327 | Lonafarnib | WuXi AppTec | Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | WUXI APPTEC | | | | |
| 1328 | Lonafarnib | (HONGKONG) LTD. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | WuXi | | | | |
| | | AppTec(HongKong)Limit | | | | |
| 1329 | Lonafarnib | ed | Quotation #1 | \$0.00 | \$0.00 | \$0.00 |
| | | WuXi | | | | |
| | | AppTec(HongKong)Limit | | | | |
| 1330 | Lonafarnib | ed | Quotation #2 | \$0.00 | \$0.00 | \$0.00 |
| 1331 | Lonafarnib | Xcelience | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Xenotech LLC dba | | | | |
| 1332 | Lonafarnib | Sekisui Xenotech, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Yuki Gosei Kogyo Co., | Price Quotation and Trade Terms dated | | | |
| 1333 | Lonafarnib | Ltd. | December 10, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Yuki Gosei Kogyo Co., | Price Quotation and Trade Terms dated July | | | |
| 1334 | Lonafarnib | Ltd. | 2, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | L | Yuki Gosei Kogyo Co., | Price Quotation and Trade Terms dated June | | | |
| 1335 | Lonafarnib | Ltd. | 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | L | Yuki Gosei Kogyo Co., | Price Quotation and Trade Terms dated May | | | |
| 1336 | Lonafarnib | Ltd. | 23, 2018 | \$0.00 | \$0.00 | \$0.00 |

Case 24-80040-sgj11 Doc 838-3 Filed 06/05/25 Entered 06/05/25 16:40:05 Desc Exhibit & Sex Page 39 of 39

| | Related | | | | | Lonafarnib |
|------|------------|-------------------------|------------------------------------|-----------------------|-------------|------------|
| # | Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | YUKI GOSEI KOGYO | Yuki Gosei 1st Amnd to SOW 103-366 | | | |
| 1337 | Lonafarnib | CO., LTD. | 20230407 | \$0.00 | \$0.00 | \$0.00 |
| | | YUKI GOSEI KOGYO | | | | |
| 1338 | Lonafarnib | CO., LTD. | Yuki Gosei SOW 102-054 20210702 | \$0.00 | \$0.00 | \$0.00 |
| | | YUKI GOSEI KOGYO | | | | |
| 1339 | Lonafarnib | CO., LTD. | Yuki Gosei SOW 103-093 20220530 | \$0.00 | \$0.00 | \$0.00 |
| | | YUKI GOSEI KOGYO | | | | |
| 1340 | Lonafarnib | CO., LTD. | Yuki Gosei SOW 103-101 20220614 | \$0.00 | \$0.00 | \$0.00 |
| | | YUKI GOSEI KOGYO | | | | |
| 1341 | Lonafarnib | CO., LTD. | Yuki Gosei SOW 103-295 20221122 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | | |
| 1342 | Lonafarnib | Zokinvy Isi Development | Statement of Work | \$0.00 | \$0.00 | \$0.00 |

Exhibit 4

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

AMENDED NOTICE OF CURE AMOUNTS
AND POTENTIAL ASSUMPTION AND ASSIGNMENT
OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES
IN CONNECTION WITH THE REMAINING ASSETS SALE TRANSACTION(S)

PLEASE TAKE NOTICE:

On April 1, 2024 (the "<u>Petition Date</u>"), the debtors and debtors in possession in the above-captioned chapter 11 cases (collectively, the "<u>Debtors</u>") filed voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the Northern District of Texas (the "<u>Court</u>").

On April 1, 2024, the Debtors filed a motion [Docket No. 13] (the "Bid Procedures Motion") with the Court seeking entry of an order, among other things: (i)(a) approving the proposed bid procedures (the "Bid Procedures") in connection with the Sale Transaction(s) of the Assets; (b) approving the Bid Protections relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (c) establishing the dates and deadlines relating to the Remaining Asset Bid Deadline, Auction(s), and a Sale Hearing(s); (d) approving the form and manner of the notice of the Sale Transaction(s), the Bid Deadline(s), the Auction(s), and the Sale Hearing(s) (the "Sale Notice"); (e) approving the assumption and assignment procedures for any executory contracts or unexpired lease to be assumed by the Debtors in connection with the Sale Transaction(s), if any (the "Designated Contracts") (the "Assignment and Assumption Procedures"); and (f) approving the form and manner of the potential assignment and assumption notice (the "Assignment Notice"); and (ii) granting related relief.

On April 5, 2024 the Court entered the Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Ave., Dallas, Texas 75201.

Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order").² All recipients of this notice should carefully read the Bid Procedures Order and the Assumption and Assignment Procedures in their entirety.

On June 4, 2024, the Debtors filed the Amended Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Remaining Assets Sale Transaction(s) [Docket No. 313] (the "Initial Cure Notice").

You are receiving this notice (the "Amended Cure Notice") because you may be a counterparty to an executory contract or unexpired lease of the Debtors that may be assumed and assigned to the Winning Bidder, or as otherwise contemplated in the Winning Bid related to the Remaining Assets Sale Transaction(s) (each, a "Potentially Assigned Contract"). Each of the Potentially Assigned Contracts that may be assumed and assigned in connection with the Remaining Assets Sale Transaction(s) with a Winning Bidder and the Debtors' calculation of the Cure Amounts with respect thereto are set forth on Exhibit A hereto (the "Cure Schedule"). The Cure Amounts are the only amounts proposed to be paid upon the assumption and/or assignment of the Potentially Assigned Contracts. A redline of the Cure Schedule as against the cure amounts set forth in the Initial Cure Notice is set forth on Exhibit B hereto.

The inclusion of any contract or lease on **Exhibit A** shall not constitute or be deemed a determination or admission by the Debtors that such contract or other document is, in fact, an executory contract or unexpired lease within the meaning of the Bankruptcy Code (all rights with respect thereto being expressly reserved).

Notwithstanding the inclusion of any Potentially Assigned Contract on Exhibit A, neither the Debtors nor any Winning Bidder(s) are obligated to assume or assign any Potentially Assigned Contract identified on Exhibit A. The Amended Cure Notice is only advising parties of the Debtors' proposed Cure Amounts should the Potentially Assigned Contract be assumed or assumed and assigned. Subsequent notice will be provided if your Potentially Assigned Contract is proposed to be assumed or assumed and assigned to a Winning Bidder. If you disagree with the proposed Cure Amounts, you must file an objection in accordance with the instructions below.

Objections

A. Cure Objections

Any objection to potential assumption, assignment, or designation of a Potentially Assigned Contract identified on **Exhibit A**, the subject of which objection is the Debtors' proposed Cure Amounts, if any, must (i) be in writing; (ii) comply with the Bankruptcy Rules and the Bankruptcy Local Rules; (iii) be filed with the Clerk of the Court, by electronic submission through

² Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Bid Procedures Motion, the Bid Procedures, and the Bid Procedures Order, as applicable.

PACER (Public Access to Court Electronic Records at https://ecf.txnb.uscourts.gov/), or if submitted through non-electronic means, by U.S. Mail or other hand delivery system at the following address: United States Bankruptcy Court, 1100 Commerce St, #5300, Dallas, TX 75242, on or before 4:00 p.m. (CT) on June 12, 2024 (the "Remaining Assets Cure Objection Deadline"); (iv) be served, so as to be actually received on or before the Remaining Assets Cure Objection Deadline, upon the following via email: (a) proposed counsel to the Debtors, Sidley Austin R. Califano (tom.califano@sidley.com), William LLP Thomas (wcurtin@sidley.com), and Anne G. Wallice (anne.wallice@sidley.com)), (b) counsel to any official committee of unsecured creditors appointed in the Debtors' chapter 11 cases; (c) the Office of the United States Trustee, 1100 Commerce Street, Room 976, Dallas, Texas 75242 (attn: Elizabeth A. Young (elizabeth.a.young@usdoi.gov)); and (d) counsel to Innovatus, Bradley Arant Boult (rjones@bradley.com) Cummings LLP (attn. Roger Jones and (jbender@bradley.com)(collectively, the "Objection Notice Parties"); and (v) state with specificity the legal and factual grounds for such objection, including, without limitation, the Cure Amounts the Counterparty believes is required to cure defaults under the relevant Executory Contracts or Unexpired Leases. Any objections solely on the basis of adequate assurance of future performance must be filed in the manner described in the foregoing sentence and served on the Objection Notice Parties by no later than 4:00 p.m (CT) on June 12, 2024.

IF A COUNTERPARTY FAILS TO FILE WITH THE BANKRUPTCY COURT AND SERVE ON THE OBJECTION NOTICE PARTIES A TIMELY CURE OBJECTION, THE COUNTERPARTY SHALL BE FOREVER BARRED FROM ASSERTING ANY OBJECTION WITH REGARD TO THE AMOUNT TO CURE ANY DEFAULT UNDER THE APPLICABLE POTENTIALLY ASSIGNED CONTRACT. THE CURE AMOUNTS SET FORTH ON EXHIBIT A HERETO SHALL BE CONTROLLING AND WILL BE THE ONLY AMOUNT NECESSARY TO CURE OUTSTANDING DEFAULTS UNDER THE APPLICABLE POTENTIALLY ASSIGNED CONTRACT UNDER BANKRUPTCY CODE SECTION 365(B), NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THE POTENTIALLY ASSIGNED CONTRACT, OR ANY OTHER DOCUMENT, AND THE APPLICABLE COUNTERPARTY SHALL BE FOREVER BARRED FROM ASSERTING ANY ADDITIONAL CURE OR OTHER AMOUNTS WITH RESPECT TO SUCH POTENTIALLY ASSIGNED CONTRACT AGAINST THE DEBTORS, ANY WINNING BIDDER, OR THE PROPERTY OF ANY OF THEM.

B. Adequate Assurance Objection

As stated, this notice does not indicate if your Potentially Assigned Contract will be assumed, rejected, or assumed and assigned to a Winning Bidder. If your Potentially Assigned Contract is proposed to be assumed or assumed and assigned, subsequent notice will be provided. Any objections to an assignee's adequate assurance of future performance are preserved and the procedures for such adequate assurance objections will be set forth in the subsequent notice.

Cure Objection Hearing

Cure Objections that cannot be resolved by the parties will be heard by the Bankruptcy Court on June 26, 2024 at 1:30 p.m. (prevailing Central Time) or as soon thereafter as such

Cure Objection may be heard. You may participate in the Hearing either remotely by an audio and video connection or in person.

Remote Participation Instructions. Audio communication will be by use of the Court's dialin facility. You may access the facility at 1.650.479.3207. Video communication will be by use of the Cisco WebEx platform. Connect via the Cisco WebEx application or click the link on Judge Jernigan's home page. The meeting code is 2304-154-2638. Click the settings icon in the upper right corner and enter your name under the personal information setting.

<u>In Person Instructions</u>. If you wish to attend the hearing in person, the hearing will be conducted at Courtroom 1, floor 14, 1100 Commerce Street, Dallas, TX 75242-1496.

<u>Hearing Appearances</u>. If you intend to appear electronically, hearing appearances must be made electronically in advance of electronic hearings. To make your appearance, click the "Electronic Appearance" link on Judge Jernigan's home page. Select the case name, complete the required fields and click "Submit" to complete your appearance.

[Remainder of page intentionally left blank]

Dated: June 21, 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825)
William E. Curtin (admitted *pro hac vice*)
John J. Kuster (admitted *pro hac vice*)
Jon W. Muenz (admitted *pro hac vice*)
Anne G. Wallice (admitted *pro hac vice*)
787 Seventh Avenue

New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com jkuster@sidley.com jmuenz@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413)

2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Cure Amounts

| Related | | | | | Lonafarnib |
|------------|------------------------|--|-----------------------|--------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | | | | | |
| Lonafarnib | 3D Communications, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | 3D Communications, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lambda | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | | |
| Lonafarnib | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | \$0.00 | |
| | | 8 | 4 | ***** | 40.00 |
| Lonafarnib | Absorption Systems LP | In Vitro Study Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Accenture, LLP | Services Agreement | \$0.00 | \$5,399.00 | \$0.00 |
| | | Amendment One to the Master Services | | | |
| Lonafarnib | Accenture, LLP | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | 40.00 |
| Lonafarnib | Accenture, LLP | Change Order 2 to SOW 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Change Order 3 to SOW 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Accenture, LLP | Change Order 3 to SOW 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Change Order Form No. 7 | \$0.00 | \$0.00 | \$0.00 |
| Londianio | riccontaire, EEr | Change Graci Form 1(c) | \$0.00 | ψ0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Change Order Form No. 9 | \$0.00 | \$0.00 | \$0.00 |
| | · | Change Order No. 003 Amending the | | | |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Order No. 004 Amending the | | | |
| | | Regulatory and Clinical Services Amended | | | |
| T 0 " | | and Restated Statement of Work No. 5 | #0.00 | #0.00 | #0.00 |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Order No. 005 Amending the Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Accenture, LLI | Change Order No. 006 Amending the | \$0.00 | \$0.00 | \$0.00 |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Order No. 008 Amending the | * | * | , , , , , |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019. | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Accenture, LLP | Data Privacy and Security Exhibit | \$0.00 | \$0.00 | \$0.00 |
| T C '1 | | G CANY 1 A | Ф0.00 | #0.00 | Φο οο |
| Lonafarnib | Accenture, LLP | Scope of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Scope of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Accenture, LLF | Beope of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | ADAMAS Consulting | <i>G</i> | \$3.00 | \$3.00 | \$0.50 |
| Lonafarnib | LLC | Project Proposal | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|--|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lambda | Addison Whitney | Agreement | \$0.00 | | |
| | j | Statement of Work describes Services and | | | · |
| Lonafarnib | Addison Whitney | Deliverables to be | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Addison Whitney LLC, a Syneos Health Group Company | Peginterferon Lambda Package & Label Design SOW | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Additional Submission Document for Lonafarnib Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Company | Bondarino Statement of Work | \$0.00 | φ0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Lonafarnib (Progeria and HDV) Global Brand Name Development and Validation Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Lonafarnib for HDV EMA co-pack regulatory Consulting Validation Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Webpage Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Сопрану | webpage Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney, LLC | Amendment No. 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Agenzia Italiana del Farmaco | Access Request Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Agenzia Italiana del Farmaco | Letter re: Declaration on Public Contributions and Incentives for Developmental of the Medical Product | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Agenzia Italiana del | Letter re: Developmental of Medical | \$0.00 | φ0.00 | \$0.00 |
| Lonafarnib | Farmaco | Product | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | AIFA – Agenzia Italiana del Farmaco ALTASCIENCES | Access Request Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | COMPANY CANADA INC. | Agreement For Bioanalytical Services | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | ALTASCIENCES | rigiteement 1 of Biodinary fied Betvices | \$0.00 | ψ0.00 | \$0.00 |
| Lambda | COMPANY INC. | Change Order 01 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | ALTREOS RESEARCH PARTNERS, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | American Association for the Study of Liver Diseases | The Liver Meeting 2020 Exhibit Space Application and Contract | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | American Mobile Health Services, LLC | Service Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------------------|-------------------------------|--|-----------------------|--------------|---------------|
| Asset | | Description of Contract of Lease | Avexitide Cure | | Cure |
| Lambda | AMRI | Laboratory Service Fee Quotation | \$0.00 | \$0.00 | \$0.00 |
| I | Anablani Canavitina II C | Statement of Worls #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Archbow Consulting, LLC | Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| | ARUP LABORATORIES | | | | |
| Lonafarnib | INC. | Amendment 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | · | |
| | ARUP LABORATORIES | | | | |
| Lonafarnib | INC. | Amendment 2 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | August Bio Quotes covering batches | | | |
| | | AJ110001 (Diluent, Jun 2021) and | | | |
| Avexitide | August Bioservices LLC | AJ111001 (Jun 2021 225 Mg /ml). | \$0.00 | \$0.00 | \$0.00 |
| | | D 1 D 1110 | Ф0.00 | фо. о о | # 0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1119 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1126 | \$0.00 | \$0.00 | \$0.00 |
| Aveatude | August Dioscivices LLC | 110p0sai. 1-1120 | \$0.00 | \$0.00 | φ0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1143 | \$0.00 | \$0.00 | \$0.00 |
| | | • | | | |
| Avexitide | August Bioservices LLC | Proposal: P-1147 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | August Bioservices LLC | Proposal: P-1160 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Austin Chemical | Letter Agreement re: Offer Ritonavir USP (CAS # 155213-67-5) | \$0.00 | \$0.00 | \$0.00 |
| Lonatarnio | Company, Inc. Austin Chemical | Letter Agreement re: Offer Ritonavir USP | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Company, Inc. | (CAS # 155213-67-5) | \$0.00 | \$0.00 | \$0.00 |
| | ,, , | (| 4 | 4,333 | , , , , , , |
| | Avayle CTS, LLC; Fisher | | | | |
| Lonafarnib | Clinical Services | Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | AVISTA PHARMA | | | | |
| | SOLUTIONS, INC., | | | | |
| Lambda | D/B/A CAMBREX | Proposal EIGR-D22-D001-A | \$0.00 | \$0.00 | \$0.00 |
| | B/B/II GIII/IBIEII | Proposal re: Development and Validation of | \$0.00 | ψο.σσ | \$0.00 |
| | | a Method for Free PEG in IL-29 PEG- | | | |
| | Avista Pharma Solutions, | Interferon Lambda-1a Drug Conjugate dated | | | |
| Lambda | Inc., d/b/a Cambrex | May 18, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal re: Development and Validation of | | | |
| | Avista Pharma Solutions, | a Method for Free PEG in IL-29 PEG- Interferon Lambda-1a Drug Conjugate dated | | | |
| Lambda | Inc., d/b/a Cambrex | September 30, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | mo., arora Camorex | Certificate of Analysis (COA) meeting Eiger | | φυ.υυ | \$0.00 |
| Avexitide | BACHEM | Specifications | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | BACHEM | Change Order 1 to Quote | \$0.00 | | |
| | | Change Request re: Exendin (9-39), non- | | | |
| Avexitide | Bachem | GMP | \$0.00 | \$0.00 | \$0.00 |
| A 1,1.4 | DACHEM | | #0.00 | #0.00 | # 0.00 |
| Avexitide | BACHEM | Compilation of Validation Plan of Avexitide | | | |
| | | | | | |
| Avexitide Avexitide | BACHEM BACHEM | Quote Quote 67815 | \$0.00 \$0.00 | | |

| Related | Contract Countownauty | Description of Contract of Lease | Avexitide Cure | Lambda Cuus | Lonafarnib | |
|--------------------|---|---|----------------|-------------|-------------|--|
| Asset Avexitide | Contract Counterparty BACHEM | Quote 69621 | \$0.00 | \$0.00 | Cure \$0.00 | |
| Avexitide | BACHEM | Quote 70720 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 71209 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 72592 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 73445 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 20005572 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 58395 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 63842 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 64409 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65398 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65399 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65409 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65675 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 66755 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 67097 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 20005572 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 20005959 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 55597 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 57429 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 57583 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 63034 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 63583 | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Quotation 66372 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | BD Medical, Pharmaceutical Systems US | Letter re: Pricing dated April 27, 2020 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | BD Medical, Pharmaceutical Systems US | Letter re: Pricing dated May 13, 2020 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Becton, Dickinson & Company | Letter re: PS-CPQ-636 - BD | \$0.00 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|---|----------------|-------------|--------------------|
| Lambda | BECTON, DICKINSON AND COMPANY | Quote re: Pharmaceutical Products | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research Inc., a Lonza Company | Analytical Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research Inc., a Lonza Company | Change Order 7 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research Inc., a Lonza Company | Change Order No. 4 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Amendment No. 2 to the Master Services and Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 3 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 3 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Change Order 4 Dated June 22, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Change Order 4 Dated May 2, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 5 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 6 to the Statement of Work Change Order 6 to the Statement of Work | \$0.00 | | |
| Lonafarnib | Bend Research, Inc. | dated September 2, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Commercial Manufacturing Services and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Memorandum of Understanding | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Mutual Nondisclosure Agreement dated February 23, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Mutual Nondisclosure Agreement dated November 22, 2017 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---|---|----------------|--------------|--------------------|
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Proposal dated July 9, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BEND RESEARCH, INC. | Proposal dated October 14, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Purchase Order dated April 4, 2017 | \$0.00 | | |
| | , | 1 , , , , | , | • | * |
| Lonafarnib | Bend Research, Inc. | Purchase Order dated March 6, 2017 | \$0.00 | \$0.00 | \$0.00 |
| | BIOANALYTICAL | Task Order Q-00122 to the Master | | | |
| Lonafarnib | SYSTEMS, INC. | Independent Contractor Agreement | \$0.00 | | |
| Lambda | BIORASI, LLC ¹ | Change Order 1 to Statement of Work | \$0.00 | 1 | \$0.00 |
| Lambda | Biorasi, LLC | Clinical Study Agreement | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Letter of Intent | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Letter of Intent dated September 29, 20201 | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Letter of Intent Extension Agreement dated December 18, 2020 | \$0.00 | \$181,489.74 | \$0.00 |
| | | Letter of Intent Extension Agreement | 40100 | | 40100 |
| Lambda | BIORASI, LLC | September 29, 2021 | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Statement of Work | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Statement of Work 3 | \$0.00 | | \$0.00 |
| Lonafarnib | BIORASI, LLC | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | BIORASI, LLC | Change Order 2 to Statement of Work | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Change Order 3 to Statement of Work | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Change Order 4 to Statement of Work | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Letter of Intent dated June 13, 2020 | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Master Services Agreement | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Statement of Work | \$0.00 | \$0.00 | |
| Avexitide | Bioserv Corporation | Proposal re: Project Manufacture of Exendin 9-39 Injection | \$0.00 | | |
| Avexitide | Bioserv Corporation | Proposal re: Project Non Aseptic Engineering Run | \$0.00 | | |
| Aveatude | Biosciv Corporation | Engineering Run | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bioserv Corporation | Proposal re: Project Vial Labeling | \$0.00 | \$0.00 | \$0.00 |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | Assignment and Assumption Agreement | \$0.00 | \$0.00 | \$0.00 |
| T 11 | BRISTOL-MYERS | | Ф0.00 | Ф0.00 | #0.00 |
| Lambda | SQUIBB COMPANY | Common Stock Purchase Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY Bristol-Myers Squibb | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Company | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Brown University | First Amendment to Research Funding Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Brown University | RESEARCH FUNDING AGREEMENT | \$0.00 | | |
| _onaranino | DIOWII OIII OIBILY | ILLELING TOTALINE | ψ0.00 | ψ0.00 | ψ0.00 |

¹ The Debtors, Biorasi, LLC and its affiliates (collectively "Biorasi") are engaged in ongoing discussions to consensually resolve the cure amounts for each of the contracts between the Debtors and Biorasi. All parties' rights are reserved.

| Related | | | | | Lonafarnib |
|------------|------------------------------|---|-----------------------|----------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Second Amendment to Research Funding | | | |
| Lonafarnib | Brown University | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Bulgarian Drug Agency | Declaration on Funds Ownership | \$0.00 | | |
| Avexitide | CALISYM LLC | Services Agreement | \$2,000.00 | \$0.00 | |
| Lambda | CALISYM LLC | Statement of Work 4 | \$0.00 | | |
| Lonafarnib | CALISYM LLC | Statement of Work 3 | \$0.00 | \$0.00 | |
| Lonafarnib | CALISYM LLC | Statement of Work 4 | \$0.00 | \$0.00 | |
| Lonafarnib | CALISYM LLC | Statement of Work 7 | \$0.00 | \$0.00 | |
| Lonafarnib | CALISYM LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBER | | | | |
| | PHARMACEUTICALS | | | | |
| Lanafamil | | Nanhinding Town Short | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | INCORPORATED | Nonbinding Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Cambridge Biomarketing | Project Brief V1 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Master Services Agreement Amendment | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Name Change to MSA | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 10 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 11 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 12 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| т с ч | BIOMARKETING | Grand CW 1.2 | #0.00 | # 0.00 | 00.00 |
| Lonafarnib | GROUP, LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| I f '1 | BIOMARKETING | Statement of W. J. A | Φ0.00 | #A AA | φο ο ο |
| Lonafarnib | GROUP, LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| Longfamil | BIOMARKETING | Statement of Wards 5 | \$0.00 | φ α α α | ቀለ ለሳ |
| Lonafarnib | GROUP, LLC | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|--|-----------------------|---------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | 40.00 | | 40.00 |
| Lonafarnib | GROUP, LLC | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Canadian Life Science; | | | | |
| | Patheon, by Thermo | | | | |
| Lonafarnib | Fisher Scientific | DDC Lab Dill Dook Dagwagt Form | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Capsugel | PDS Lab Bill Back Request Form Quote re: 1 Color Ink | \$0.00 | | |
| Lonafarnib | Capsugel | Quote re: Capsules | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Capsuger | Agreement for the Establishment of Legal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CATO Europe GmbH | Representative in European Union | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | CATO Europe Omori | Representative in European Onion | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CATO Europe GmbH | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| | GATTO E | | | | |
| T C 1 | CATO Europe GmbH; | T CA d | Ф0.00 | Ф0.00 | #0.00 |
| Lonafarnib | Clinipace GmbH | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Can duit II C | Durant for the Duranisian of IDT Commission | \$0.00 | \$0.00 | \$0.00 |
| Lonatarnio | Cenduit LLC | Proposal for the Provision of IRT Services | \$0.00 | \$0.00 | \$0.00 |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Fifth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Diseases, LLC (CLDQ) | 1 Ittil Amendment to License Agreement | \$0.00 | \$0.00 | φ0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | First Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Second Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Discases, LLC (CLDQ) | Second Americanion to License Agreement | φ0.00 | φυ.00 | \$0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| I | Research in Liver | Circle Amondance to Time A | 00.00 | # 0.00 | Φ0.00 |
| Lonafarnib | Diseases, LLC (CLDQ) | Sixth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |

| D.I. (.) | | | | | T 6 1 |
|------------------|-------------------------------|--|----------------|--------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| | | · | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Third Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC d/b/a CLDQ | Fourth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | CLDQ | Tourth Amendment to Electise Agreement | ψ0.00 | ψ0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002084 in 060622 Statement of | #0.00 | #0.00 | #0.00 |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002086 in 060622 Statement of | | | |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002087 in 060622 Statement of | | | |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002089 in 060622 Statement of | | | |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | Charles River | | ФО ОО | ФО ОО | #0.00 |
| Avexitide | Laboratories Charles River | Archive Storage Letter dated July 12, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Laboratories | Archive Storage Letter dated July 20, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Ai4: .1 - | Charles River | Latter and Developed Angle aringtion | 90.00 | \$0.00 | ¢0.00 |
| Avexitide | Laboratories | Letter re: Payment Authorization | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Avexitide | LABORATORIES | Statement of Work (CRL-210936) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Avexitide | LABORATORIES | Statement of Work (CRL-224818) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Avexitide | LABORATORIES | Statement of Work (SFDC-OPP-272991) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Avexitide | LABORATORIES | Statement of Work 20220630 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | CHARLES RIVER LABORATORIES | 1st Amnd to Statement of Work (SFDC OPP-273814) 20230131 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LADORATORIES | 273017) 20230131 | \$0.00 | φ0.00 | \$0.00 |
| T C " | CHARLES RIVER | 2nd Amnd to Statement of Work (SFDC | 40.00 | 40.00 | 40.00 |
| Lonafarnib | LABORATORIES | OPP-273814) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amendment 1 to Statement of Work | | | |
| Lonafarnib | LABORATORIES | 20190821 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|----------------------|-------------------------------|---|------------------|------------------|------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | CHARLES RIVER | Amendment 2 to Statement of Work | | | |
| Lonafarnib | LABORATORIES | 20200326 | \$0.00 | \$0.00 | \$0.00 |
| | CHADLEC DIVED | | | | |
| Lonafarnib | CHARLES RIVER LABORATORIES | Statement of Work (CRL-237365) | \$0.00 | \$0.00 | \$0.00 |
| | | Same ment of went (end 20,000) | \$0.00 | \$0.00 | \$ 0.00 C |
| T C | CHARLES RIVER | St. t SW. 1- 20100619 | ¢0.00 | #0.00 | ¢0.00 |
| Lonafarnib | LABORATORIES | Statement of Work 20190618 | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Lonafarnib | LABORATORIES | Statement of Work 20220610 | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| | LABORATORIES | | | | |
| Lonafarnib | MONTREAL ULC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CHARLES RIVER | ARCHIVE STATEMENT OF WORK | | | |
| Avexitide | LABORATORIES, INC. | CONTINUED STORAGE | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CHARLES RIVER | Statement of Work dated December 31, | | | |
| Lambda | LABORATORIES, INC. | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CHARLES RIVER | Statement of Work dated December 31, | | | |
| Lambda | LABORATORIES, INC. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CHARLES RIVER | Statement of Work re: Evaluation of the | | | |
| Lambda | LABORATORIES, INC. | Stability of E.coli Working Cell Bank | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work re: Preparation and | | | |
| | CHARLES RIVER | Characterization of an E. coli Master Cell | #0.00 | #0.00 | #0.00 |
| Lambda | LABORATORIES, INC. | Bank | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T 1.1 | CHARLES RIVER | Statement of Work re: Storage of Materials | #0.00 | #A AA | #0.00 |
| Lambda Lonafarnib | LABORATORIES, INC. CHEMTREC | Under Controlled Conditions and Access Quote re: Emergency Response Service | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Lonararmo | CITRINE MEDICINE, | Quote re. Emergency Response Service | ψ0.00 | ψ0.00 | φυ.υυ |
| Lonafarnib | LLC | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | ClinicalConnection, Inc | Direct Study Participant Recruitment Agreement | \$0.00 | \$0.00 | \$0.00 |
| 201141411110 | emiliar connection, inc | Change Order 3 to Work Order C-14 for | | ψ0.00 | ψυ.υυ |
| Lambda | CLINIPACE, INC. | Eiger | \$0.00 | \$0.00 | \$0.00 |
| | | Clinical Research | | | |
| | | Change Order 1 to Work Order C-14 For | | | |
| Lambda | CLINIPACE, INC. | Eiger | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|--|----------------|--------------|---------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Clinical Research | | | |
| | | Change Order 1 to Work Order C-7 ("CO1") | | | |
| Lambda | CLINIPACE, INC. | For Eiger | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Clinipace, Inc. | Work Order C-11 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Clinipace, Inc. | Work Order C-12 | \$0.00 | \$0.00 | 4 |
| Lambda | Clinipace, Inc. | Work Order C-14 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Clinipace, Inc. | Work Order C-7 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Clinipace, Inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CLINIPACE, INC. | CO2 to Work Order C-1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinipace, Inc. | Work Order C-10 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinipace, Inc. | Work Order C-11 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinipace, Inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| T 1 1 | C1: . I 11 C : 1 | CO2+ W 1 O 1 C 7 | #0.00 | #0.00 | фо о о |
| Lambda | Clinipace, Inc. dba Caidya Colleen M. Craig | CO2 to Work Order C-/ | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Consulting, LLC | Services Agreement | \$1,650.00 | \$0.00 | \$0.00 |
| Avexilide | Consuming, LLC | Services Agreement | \$1,030.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Corden i narma Colorado | Change Graci #1 | ψ0.00 | \$0.00 | ψ0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 dated May 19, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | 40100 | 4000 | 40.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 dated Nov 5, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Corden Pharma Colorado | Change Order CO4 to CO2 of SOW #2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CODDEN DILADAA | Change Only 24. Change Only 14. | | | |
| I C | CORDEN PHARMA COLORADO INC. | Change Order 2 to Change Order 1 to | ¢0.00 | ¢0.00 | ¢0.00 |
| Lonafarnib | COLORADO INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Change Order 3 to Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | COLOR DO INC. | change order 5 to Statement of Work 2 | φ0.00 | ψο.σσ | ψ0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | CORDEN PHARMA | a a a a a a a a a a a a a a a a a a a | <u> </u> | ±= // / | |
| Lonafarnib | COLORADO INC. | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|---|----------------|---------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| Lonafarnib | CORDEN PHARMA COLORADO INC. | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | COLORADO INC. | Statement of Work 3 | \$0.00 | \$0.00 | ψ0.00 |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Corden Pharma Colorado; | | | | |
| Lonafarnib | Corden Pharma International GmbH | Change Order #5 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | International Oniori | Change Graci #9 | \$0.00 | \$0.00 | ψ0.00 |
| | Corden Pharma Colorado; | | | | |
| | Corden Pharma | | | | |
| Lonafarnib | International GmbH | Change Order 1 to the Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Corden Pharma Colorado; Corden Pharma | | | | |
| Lonafarnib | International GmbH | Change Order 5 to the Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Corden Pharma Colorado; | | | | |
| Lonafarnib | Corden Pharma International GmbH | Coden Pharma Quote | \$0.00 | \$0.00 | \$0.00 |
| | | couch I minim Quest | Ψ0100 | \$0100 | \$0.00 |
| Avexitide | Corden Pharma International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| TIVENITIAE | | Quote 20210323 | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Corden Pharma International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| | | (| ***** | ***** | ****** |
| | CordenPharma; Corden Pharma International | | | | |
| Lonafarnib | GmbH | Coden Pharma Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Amendment 1 to Service Agreement - H | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | COREALIS PHARMA | Service Agreement - A | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - B | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - J | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - N | \$0.00 | \$0.00 | \$0.00 |
| | | Service Agreement - IV | | | |
| Lonafarnib | COREALIS PHARMA | Service Agreement - O | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure | |
|------------------|--|---|----------------|-------------|--------------------|--|
| Avexitide | Curia New Jersey, LLC | Services Agreement | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 3/26/2022 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 5/31/2022 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 7/27/2022 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 9/30/2022 | \$0.00 | \$0.00 | \$0.00 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated June 2022 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Cyprotex US, LLC | Proposal for Analysis of Active Metabolite of Lonafarnib (LNF): HM21 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Cyprotex US, LLC | Proposal for Analysis of Active Metabolites of Lonafarnib (LNF): MH17 and HM21 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Cyprotex US, LLC | Proposal for CytoChrome P450 Time Dependent Inhibition of Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Cyprotex US, LLC | Proposal for MATE Transporter Inhibition Testing for Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Cyprotex US, LLC | Proposal for OCT1 and BSEP Inhibition and OCT1 Substrate Testing for Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 | |
| Avexitide | D. S., PHD [Name on File] | Consulting Agreement | \$12,750.00 | \$0.00 | \$0.00 | |
| Lonafarnib | D. S., PHD [Name on File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | DEFENSE HEALTH AGENCY | Department of Defense (DoD) Retail Refund Pricing Agreement | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Department of Health and Human Services | National Drug Rebate Agreement | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | 1st Amnd to BPA 36E79722A0010 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Amendment | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Amendment 2 to 36F79722D0048 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Amendment 3 | \$0.00 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|----------------------|--|---|------------------|-------------|--------------------|
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Amendment of Solicitation/Modification of Contract | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Blanket Purchase Agreement dated Jan 1, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Blanket Purchase Agreement dated Oct 10, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | BPA Cancellation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Commercial Items | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Department of Veterans Affairs | D0048 Red A | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Department of Veterans Affairs | Facility Code | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | FSS NFAMP Smoothing Letter I B R8, Amendment 0004 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Amendment of Solicitation/Modification of Contract | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Master Agreements and Pharmaceutical Pricing Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DEPARTMENT OF VETERANS AFFAIRS | Small Business Subcontracting Plan | \$0.00 | \$0.00 | \$0.00 |
| Lambda | DILISYM SERVICES, INC. | Master Services Agreement dated June 14, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | DILISYM SERVICES, INC. | Master Services Agreement May 22, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Dom zdravlja Medigroup dr Ristic Pariske Dr. T. L. M. [Name on | Donation Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | File] | Proposal to Purchase Assets | \$2,981.00 | \$0.00 | \$0.00 |
| Lambda Lonafarnib | DSG, Inc. ELEMENT | Quotation re: Ongoing Project Management Master Service Agreement | \$0.00 \$0.00 | | |
| Lonafarnib | Element Materials Technology Canada Inc. | Element Quote | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--------------------------------------|---|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | Element Materials | | | | |
| I C 11 | Element Materials | F1 | ¢0.00 | £0.00 | £0.00 |
| Lonafarnib | Technology Canada Inc. | Element Quote dated April 7, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | Flement Materials Testing | Quote re: Robustness Study for SOP 20-48- | | | |
| Lonafarnib | Canada Inc. | SP-2396 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Eurofins | Change Order dated February 14, 2022 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins | Change Order November 3, 2021 | \$0.00 | | \$0.00 |
| Lambda | Eurofins | Quote dated December 31, 2017 | \$0.00 | | \$0.00 |
| Lambda | Eurofins | Quote dated December 4, 2020 | \$0.00 | | \$0.00 |
| | | | | | |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Avexitide | LABORATORIES, INC. | Services Agreement | | \$0.00 | \$0.00 |
| | | | | | |
| | | | \$25,600.00 | | |
| | EUROFINS ADVANTAR | | 4-2,000 | | |
| Avexitide | LABORATORIES, INC. | AGREEMENT APPROVAL 5 | | \$0.00 | \$0.00 |
| | | | | | |
| | ELIDOEINIC ADVANITAD | | | | |
| Avexitide | EUROFINS ADVANTAR LABORATORIES, INC. | AGREEMENT APPROVAL 7 | | \$0.00 | \$0.00 |
| Avexitide | LABORATORIES, INC. | AGREEMENT AFFROVAL / | | \$0.00 | \$0.00 |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 3, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | , | -, | * 1 1 1 | , | * |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 4, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 5, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | E C D' 1 | DEGI UNI COLDALI I | | | |
| T 1.1 | Eurofins Biopharma | Quote re: Free PEG by HPLC-CAD Method | #0.00 | #0.00 | ФО ОО |
| Lambda | Product Testing | Transfer | \$0.00 | \$0.00 | \$0.00 |
| | Eurofins BioPharma | Quote re: Lambda-1 (Python) FBS | | | |
| Lambda | Product Testing | Qualification | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 1/16/2023 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 10/30/2020 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 2/11/2021 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 2/11/2021 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 5/5/2021 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 7/6/2021 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 8/4/2023 | \$0.00 | | \$0.00 |
| Lambda | Eurofins BTP | Quote dated June 7, 2022 | \$0.00 | | \$0.00 |
| Lambda | Eurofins BTP | Quote dated October 1, 2021 | \$0.00 | \$0.00 | \$0.00 |

| | • | | • | | |
|--------------|--------------------------------|--|----------------|--------------|------------|
| | | | | | |
| Related | Contract Constant | Description of Contract of Land | A | I amb la Com | Lonafarnib |
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Eurofins Lancaster | | | | |
| | Laboratories (Lancaster | | | | |
| Avexitide | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Eurofins Lancaster | | | | |
| Avexitide | Laboratories (Lancaster PA) | Ovete | \$0.00 | \$0.00 | \$0.00 |
| Avexinde | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | Eurofins Lancaster | | | | |
| | Laboratories (Lancaster | | | | |
| Lambda | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Eurofins Lancaster | | | | |
| T1. 1. | Laboratories (Lancaster | Onest | ¢0.00 | ¢0.00 | £0.00 |
| Lambda | PA) EVERSANA LIFE | Quote | \$0.00 | \$0.00 | \$0.00 |
| | SCIENCE SERVICES, | | | | |
| Lonafarnib | LLC | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | EXCELLIS EUROPE | | · | | · |
| Lonafarnib | LTD. | Work Order No. One | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | F. M., D. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$2,612.50 |
| | FISHAWACK | | | | |
| Lonafarnib | FACILITATE LIMITED | Eiger Advisory Board Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | THEIDITHE ENVITED | Eiger Havisory Board Service rigidement | ψ0.00 | φ0.00 | \$0.00 |
| | FISHER | | | | |
| Lambda | BIOSERVICES, INC. ² | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote 214873 Order 3 Version 4 20190513 | \$0.00 | \$0.00 | \$0.00 |
| I | Fisher Clinical Services | Overto 214972 Orden 7 Vension 2 20210219 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH Fisher Clinical Services | Quote 214873 Order 7 Version 2 20210318 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | Quote 214873 Order 8 Version 3 20220225 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | C | 40.00 | 40.00 | 44144 |
| Lonafarnib | GmbH | Quote 215866 Order 5 Version 1 20210331 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote FCS 214873 20180809 | \$0.00 | \$0.00 | \$0.00 |
| I C 11. | Fisher Clinical Services | Quote FCS 214873 Order 3 Version 5 | ¢0.00 | ¢0.00 | £0.00 |
| Lonafarnib | GmbH Fisher Clinical Services | 20191024 Ouote FCS 214873 Order 4 Version 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | 20190703 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | 40.00 | 40.00 | 44144 |
| Lonafarnib | GmbH | Quote FCS 214873 Order 5 V1 20200218 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote FCS 214873 Order 5 V2 20200305 | \$0.00 | \$0.00 | \$0.00 |
| Longfare '1. | Fisher Clinical Services | Quote FCS 214873 Order 6 Version 2 | ΦΩ ΩΩ | ቀለ ለላ | ቁለ ለለ |
| Lonafarnib | GmbH Fisher Clinical Services | 20201229 Quote FCS 215866 Order 2 Version 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | 20190827 | \$0.00 | \$0.00 | \$0.00 |
| _01141411110 | J1011 | -01/0001 | Ψ0.00 | Ψ0.00 | ψυ.υυ |

² The Debtors, Fisher Clinical Services Inc., and its affiliates (collectively "<u>TFS</u>") are engaged in ongoing discussions to consensually resolve the cure amounts for each of the contracts between the Debtors and TFS. All parties' rights are reserved.

| Related | | | | | Lonafarnib |
|------------|------------------------------|---------------------------------------|-----------------------|-------------|---------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Fisher Clinical Services | Quote FCS 215866 Order 2 Version 3 | | | |
| Lonafarnib | GmbH | 20190918 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 2 Version 4 | | | |
| Lonafarnib | GmbH | 20191003 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 3 Version 1 | | | |
| Lonafarnib | GmbH | 20191003 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 4 Version 1 | | | |
| Lonafarnib | GmbH | 20210121 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 4 Version 2 | | | |
| Lonafarnib | GmbH | 20210504 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 5 Version 2 | | | |
| Lonafarnib | GmbH | 20210705 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | | | | |
| Avexitide | Inc. | Services Agreement | | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | \$1,405.02 | | |
| Avexitide | Inc. | Quote PSG-A-1085163.v2 20231106 | | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Avexitide | Inc. | Quote-PSG-A-1081156.v1 20230807 | | \$0.00 | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | CO 1 to PSG-A-1073971 (PSG-A-1076893) | | | |
| Lambda | Inc. | 20230524 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Fisher Clinical Quote 20160517 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Fisher Clinical Services Quote PSG-A- | | | |
| Lambda | Inc. | 1043137.v1 20210812 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote 20160927 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 55799 20170221 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Quote FCS 55799 Change Order 1 | | | |
| Lambda | Inc. | 20161116 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 55799 Order 7 20170629 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 55799 Order 8 20180918 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | 777 700 10 701 1100 1 | 40.00 | | 40.00 |
| Lambda | Inc. | Quote FCS 58040 20161206 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | 40.00 | | 40.00 |
| Lambda | Inc. | Quote FCS 62278 20180309 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | 40.00 | | 40.00 |
| Lambda | Inc. | 20190720 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Dag + 1005050 VO 00100500 | #0.00 | | # 0.00 |
| Lambda | Inc. | Quote PSG-A-1007253 V2 20190508 | \$0.00 | | \$0.00 |
| | E' 1 C'' ' 1C ' | | | | |
| T 1 1 | Fisher Clinical Services | O + DOO + 1000000 (371 00100010 | 00.00 | | 40.00 |
| Lambda | Inc. | Quote PSG-A-1009306.V1 20190619 | \$0.00 | ļ | \$0.00 |
| | Elabora Olivia de C | | | | |
| T amala 1- | Fisher Clinical Services | Oneta DCC A 10275711 20210414 | φ α α α | | ቀለ ለለ |
| Lambda | Inc. | Quote PSG-A-1037571.v1 20210414 | \$0.00 | | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|-------------------------------|---|----------------|-------------|--------------------|
| risset | Contract Counterparty | Description of Contract of Lease | Aveature Cure | Lambua Curc | Curc |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1037572.v1_20210420 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1037587.v1 20210420 | \$0.00 | \$36,483.85 | \$0.00 |
| | Fisher Clinical Services | | | \$30,103.03 | |
| Lambda | Inc. | Ouote PSG-A-1038183.v1 20210422 | \$0.00 | | \$0.00 |
| Zumouu | Fisher Clinical Services | Quant 12 0 11 1000100 11 20210 122 | \$0.00 | | φσισσ |
| Lambda | Inc. | Quote PSG-A-1040820.V3 20210713 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1041275.v4 20210819 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1044658.v1 20210902 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1045926.v1 20210927 | \$0.00 | | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | Dag 4 1052012 2 2020022 | #0.00 | | Ф0.00 |
| Lambda | Inc. | Quote PSG-A-1053812.v2 20220323 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1056697.v1 20221205 | \$0.00 | | \$0.00 |
| T amalada | Fisher Clinical Services | Overta DSC A 1060127 vil 20221205 | \$0.00 | | \$0.00 |
| Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1060127.v1 20221205 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1073971 20230429 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1086699.v1 20231128 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote re: Protocol No. EIG-LMD-001 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote-PSG-A-1037570.v2 20210423 | \$0.00 | | \$0.00 |
| | E. I. Cl. : I.C. : | | | | |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1045938.v1 20210927 | \$0.00 | | \$0.00 |
| Lamoda | me. | Quote-1 5G-11-10+3/36.V1 20210/2/ | ψ0.00 | | ψ0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote-PSG-A-1069905.v1 20230204 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Letter of Intent 20220517 | \$0.00 | \$0.00 |] |
| | Fisher Clinical Services | | | |] |
| Lonafarnib | Inc. Fisher Clinical Services | Quote FCS 63502 20180613 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote FCS 64018 20180705 | \$0.00 | \$0.00 | |
| _01141411110 | 1 | 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 | ψ0.00 | ψ0.00 | I |

| Related | | | | | Lonafarnib |
|------------|-------------------------------|---|----------------|--------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote FCS 64045 20180629 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quote FCS 64045 Order 3 Version 2 | | | |
| Lonafarnib | Inc. | 20180803 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quote FCS 64045 Order 4 Version 2 | фо. о о | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | 20190110 | \$0.00 | \$0.00 | |
| T | | Overta FCS 64624 20190921 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote FCS 64624 20180821 Quote FCS 64624 Order 1 Version 2 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | 20190128 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | 20170120 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote FCS 65165 20180919 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quete 1 02 00 100 20100717 | \$0.00 | ψ0.00 | |
| Lonafarnib | Inc. | Quote FCS 65165 v2 20181002 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote FCS 65165 v3 20181023 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote FCS 66521 20190128 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | | | |
| Lonafarnib | Inc. | 20190720 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1004098.V1 20190219 | \$0.00 | \$0.00 | |
| T C 1 | Fisher Clinical Services | O | ФО ОО | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1004100.V1 20190219 | \$0.00 | \$0.00 | |
| Lonafarnib | _ | Oveta BSC A 1004011 V1 20100211 | \$0.00 | \$0.00 | |
| Lonararino | Inc. Fisher Clinical Services | Quote PSG-A-1004911.V1 20190311 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Ouote PSG-A-1004912.V1 20190311 | \$0.00 | \$0.00 | |
| Lonararmo | Fisher Clinical Services | Quote 15G-11-100+512. V1 20150511 | ψ0.00 | ψ0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1006224 20190403 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quete 1 2 3 11 100022 : 2017 0 :00 | \$0.00 | ψ0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1006225 20190404 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1006225.V2 20190610 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1009322.V1 20190627 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1011449.V4 20190910 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Dag 4 1011001 20100025 | фо. о о | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1011991 20190827 | \$0.00 | \$0.00 | |
| Lonafarnib | - | Ouote PSG-A-1011991.V2 20190909 | \$0.00 | 90.00 | |
| Lonaranno | Inc. Fisher Clinical Services | Quote PSG-A-1011991. V2 20190909 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1019194.v1 20200311 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | Quote 1 50 11 1017177.V1 20200511 | ψ0.00 | φ0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1020030.v2 20200406 | \$0.00 | \$0.00 | |
| | | | ψ0.00 | \$5.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1028287.v3 20201124 | \$0.00 | \$0.00 | Ø11 0C1 2C |
| | Fisher Clinical Services | | | | \$11,961.26 |
| Lonafarnib | Inc. | Quote PSG-A-1034823.v2 20210303 | \$0.00 | \$0.00 | |

| Related | | | | | |
|------------|-------------------------------|---|----------------|---|--|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1035048.v2 20210308 | \$0.00 | \$0.00 | |
| | | | | | |
| | Fisher Clinical Services | DOG 1 1005056 1 00010001 | #0.00 | # 0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1035056.v1 20210224 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Ouote PSG-A-1035575.v1 20210308 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote 15G-71-1033373.V1 20210300 | ψ0.00 | ψ0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1036207.v2 20210722 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1038320.v1 20210421 | \$0.00 | \$0.00 | |
| | | | | | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1042064.v1 20210709 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1043391.v1 20210806 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1045137.v1 20210913 | \$0.00 | \$0.00 | |
| Zonaranno | Fisher Clinical Services | Quote 15G-A-10+3137.V1 20210713 | φ0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1048146.v2 20220106 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | , , , , , | * | |
| Lonafarnib | Inc. | Quote PSG-A-1051190.v1 20220114 | \$0.00 | \$0.00 | |
| | | | | | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1051277.v3 20220225 | \$0.00 | \$0.00 | |
| C 1 | Fisher Clinical Services | O + DOC + 1051400 1 20220110 | ФО ОО | #0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1051408.v1 20220118 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Ouote PSG-A-1055223.v1 20220330 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | \$0.00 | \$5.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1075062.v1 20230414 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1077951.v1 20230608 | \$0.00 | \$0.00 | |
| | Fight on Climic of Commit | | | | |
| onofom:1 | Fisher Clinical Services | Oueta BSG A 1000106 2 20220014 | ¢0.00 | _የ ለ ለላ | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1080106.v2 20230814 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Ouote-PSG-A-1024294.v1 20200707 | \$0.00 | \$0.00 | |
| Containino | 1110. | Quota 1 00 11 102 1277.V1 20200707 | ψ0.00 | ψ0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1027328.v1 20200915 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1037585.v1 20210412 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1058344.v1 20220710 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|--|----------------|-------------|--------------------|
| Lonafarnib | Fisher Clinical Services Inc. | Quote-PSG-A-1058998.v1 20230120 | \$0.00 | \$0.00 | |
| Lonafarnib | Fisher Clinical Services Inc. | Quote-PSG-A-1072091.v2 20230302 | \$0.00 | \$0.00 | |
| Lonafarnib | Fisher Clinical Services U.K. Limited | Quote PSG-A-1007765.V1 20190514 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Frontage Laboratories, Inc. | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Frontage Laboratories, Inc. Frontage Laboratories, | Proposal 1005-30837-112.1 20210308 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Inc. Frontage Laboratories, | Proposal 1005-31914-111.0 2021323 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Inc. Frontage Laboratories, | Proposal 1005-32862-122.0 20210520 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Inc. | Proposal 1005-40909-122.0 20230802 Stability Studies Termination, | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Fujifilm Diosynth ³ | Accountability and Reconciliation Memo | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | 2020 ("SOW9") and the Master Bioprocessing Services Agreement | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated January 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated July 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated October 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated October 5, 2016 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated September 22, 2016 | \$0.00 | | \$0.00 |

³ The Debtors, Fujifilm Diosynth Bioctechnologies USA, Inc., and its affiliates (collectively "<u>Fuji</u>") are engaged in ongoing discussions to consensually resolve the cure amounts for each of the contracts between the Debtors and Fuji. All parties' rights are reserved.

| | | | | Lonafarnib |
|--|--|--|---|---|
| Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| FUJIFILM DIOSYNTH | | | | |
| | Bioprocessing Services Agreement dated September 22, 2016 | \$0.00 | | \$0.00 |
| | | | | |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| USA, INC. | Change Order 2 | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH | | | | |
| USA, INC. | Change Order 3 | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH | | | | |
| USA, INC. | Change Order 4 | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | 40.00 | | **** |
| USA, INC. | Change Order 5 to Statement of Work 2 | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Letter Agreement dated April 22, 2021 | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Letter Agreement dated June 04, 2021 | \$0.00 | \$25,000.00 | \$0.00 |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA INC | Master Services Agreement | \$0.00 | | \$0.00 |
| JOSA, INC. | iviasici scrvices Agreement | \$0.00 | | \$0.00 |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Scope of Work #2 (Manufacturing) ("SOW2") and the Master Bioprocessing Services Agreement | \$0.00 | | \$0.00 |
| | Ü | ,,,,,, | | |
| FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | Statement of Work 11 | 90.00 | | \$0.00 |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 2 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 3 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 4 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 Change Order 4 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 Letter Agreement dated April 22, 2021 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Letter Agreement dated June 04, 2021 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Master Services Agreement FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 2 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 3 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 4 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 to Statement of Work 2 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 to Statement of Work 2 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Letter Agreement dated April 22, 2021 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Letter Agreement dated June 04, 2021 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Scope of Work #2 (Manufacturing) FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 2 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 3 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 4 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 to Statement of Work 2 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Change Order 5 to Statement of Work 2 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Letter Agreement dated April 22, 2021 S0.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Letter Agreement dated June 04, 2021 S0.00 \$25,000.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Master Services Agreement S0.00 \$25,000.00 FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Soope of Work #2 (Manufacturing) ("SOW2") and the Master Bioprocessing USA, INC. FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. Soope of Work #2 (Manufacturing) ("SOW2") and the Master Bioprocessing Services Agreement S0.00 |

| D.I. (I | | | | | T 6 11 |
|------------------|---------------------------------------|---|----------------|-------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| Lambda | BIOCTECHNOLOGIES USA, INC. | Statement of Work 14 | \$0.00 | | \$0.00 |
| Lumota | Cort, free. | Statement of Work 11 | ψ0.00 | | ψ0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| T 1. 1. | BIOCTECHNOLOGIES | Statement of West-2 | Ф0.00 | | ¢0.00 |
| Lambda | USA, INC. | Statement of Work 2 | \$0.00 | | \$0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| | BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 2 to Change Order 1 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 2 to Change Order 3 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| Lambda | BIOCTECHNOLOGIES USA, INC. | Statement of Work 3 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| Lambda | BIOCTECHNOLOGIES USA, INC. | Statement of Work 5 | \$0.00 | | \$0.00 |
| 241110 444 | | Surrentens of World | \$0.00 | | \$0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| T 11 | BIOCTECHNOLOGIES | Co CW. 1 C | ФО ОО | | фо ОО |
| Lambda | USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| | BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 8 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 9 | \$0.00 | | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|--|----------------|-------------|--------------------|
| Avexitide | FUJIFILM Diosynth Biotechnologies (FDB) | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies (FDB) | Letter re: Lamba Program on Hold | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Change Order 2 to Python SOW 17 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Change Order 5 re: MCB and WCB Bioassay Characterization | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Change Order 6 re: Establishment of Degraded SEC and Degraded Issi-Asp, CEX & RP Purity Assay Controls | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Change Order 6 re: Positional Isomer Feasibility | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Change Order 9 re:Update | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Purchase Order No. 5134 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Scope of Work | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Scope of Work #13 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Scope of Work #17 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|--------------------------|--|--|------------------|------------------|--------------------|
| Lambda | FUJIFILM Diosynth Biotechnologies U.S.A., Inc. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| | FUJIFILM Diosynth Biotechnologies U.S.A., | | | | |
| Lambda | Inc. | Scope of Work 23 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Gemini Staffing Consultants, Inc. | Exhibit A to Statement of Work #1 Contractors | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | Consultants, me. | Contractors | ψ0.00 | ψ0.00 | ψ0.00 |
| Lambda | GENSCRIPT USA INC | Quote U532XIC280 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | GENSCRIPT USA INC | Quote U7278IC280 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Gilero, LLC | Proposal re: Covid PFS EUA | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Gilero, LLC | Proposal re: Lambda Interferon DHF Gap Analysis | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Gilero, LLC | Proposal re: Lambda Interferon DV&V Execution | \$0.00 | \$0.00 | \$0.00 |
| | | | | | 4 |
| Lonafarnib Lonafarnib | GKV Spitzenverband Global Project Partner | Distribution Agreement Change Order | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Lonaranno | Global Floject Farther | Change Order | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GLSYNTHESIS, INC. | Certificate of Analysis | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GND GEIGER NITZ DAUNDERER RECHTSANWALTE PARTG MBB | Legal Advisory Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | GOETHE-UNIVERSITY FRANKFURT | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | H. B. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$962.50 |
| Lambda | Halloran Consulting Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Halloran Consulting Group, Inc. | SOW #5 – Inspection Readiness Assessment | | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #1 to Statement of Work #2 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|--------------------------|------------------------------|---|------------------|--------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Halloran Consulting | | | | |
| Lonafarnib | Group, Inc. | Amendment #2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 1 | Halloran Consulting | EMA Site Inspection Preparation and | Φ0.00 | #0.00 | #0.00 |
| Lonafarnib | Group, Inc. | Support – (SOW #4) | \$0.00 | \$0.00 | \$0.00 |
| | Halloran Consulting | | | | |
| Lonafarnib | Group, Inc. | Site Inspection Readiness – (SOW #3) | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Halloran Consulting | SOW #1 | \$0.00 | \$0.00 | 00.00 |
| Lonatarnio | Group, Inc. | SOW #1 | \$0.00 | \$0.00 | \$0.00 |
| | Halloran Consulting | | | | |
| Lonafarnib | Group, Inc. | Statement of Work #2 | \$0.00 | | |
| Lonafarnib | HD Biosciences | Study Quotation | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | IID D' ' I | Amendment #1 to HDB-EigerBio-01- | Φ0.00 | Φ0.00 | ФО ОО |
| Lonafarnib | HD Biosciences, Inc. | 05282019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | HD Biosciences, Inc. | Study Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | TID BIOSCIONOUS, INC. | Study Quounton | \$0.00 | ψ0.00 | \$0.00 |
| | HEALTH | | | | |
| | CORPORATION OF | | | | |
| - 0 " | GALILEE MEDICAL | | | | |
| Lonafarnib Lonafarnib | CENTER | Letter of Intent Quote 1203202a: Statement of Work | \$0.00 \$0.00 | | |
| Lonatarnio | HH Consultancy | Quote 1203202a: Statement of work | \$0.00 | \$0.00 | \$2,066.00 |
| | Icon Clinical Research | | | | |
| Avexitide | Limited | Consulting Services Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | ICON CLINICAL | | | | |
| Lonafarnib | RESEARCH LIMITED | Start-up Statement of Work No. 1 | \$0.00 | \$0.00 | \$0.00 |
| | Icon Clinical Research | | | | |
| Lonafarnib | Limited | Statement of Work EIG22720 | \$0.00 | \$0.00 | \$0.00 |
| | | | | * | , , , , , |
| | Icon Clinical Research | | | | |
| Lonafarnib | Limited | Statement of Work EIG22721 | \$0.00 | \$0.00 | \$0.00 |
| I | IDEA DECLILATORY | Letter of Intent | \$0.00 | \$0.00 | 00.00 |
| Lonafarnib | IDEA REGULATORY | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Idem Translations, Inc. | Ouotation | \$0.00 | \$0.00 | \$0.00 |
| | Inova Health Care | | | * | , , , , , |
| Lonafarnib | Services | Statement of Work 20181101 | \$0.00 | \$0.00 | \$0.00 |
| | Inova Health Care | a | * • • • | ** | * 0 |
| Lonafarnib | Services | Statement of Work 20220930 | \$0.00 | \$0.00 | \$0.00 |
| | INTEGRATED | | | | |
| | MEDICAL | Statement of work #1 to Eiger-IMD Master | | | |
| Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|--|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | INTEGRATED | | | | |
| | MEDICAL | Statement of work #3 to Eiger-IMD Master | | | |
| Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |
| | INTERNATIONAL | | 400,000 | 43333 | 7 |
| | TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| | LABORATORIES | | | | |
| Avexitide | CANADA, INC. | Statement of Work 42827 | \$0.00 | \$0.00 | \$0.00 |
| | INTERNATIONAL TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| | LABORATORIES | | | | |
| Avexitide | CANADA, INC. | Statement of Work 42828 | \$0.00 | \$0.00 | \$0.00 |
| TiveAttrac | INTERNATIONAL | Statement of Work 12020 | \$0.00 | φ0.00 | ψ0.00 |
| | TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| | LABORATORIES | | | | |
| Avexitide | CANADA, INC. | Statement of Work 74705 | \$0.00 | \$0.00 | \$0.00 |
| T 1.1 | Intrado Digital Media | | ФО ОО | ФО ОО | ФО ОО |
| Lambda | LLC | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 11 | | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 13 | | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 15 | \$2,595.00 | \$0.00 | \$0.00 |
| | | |] | | |
| Avexitide | INTRINSIK CORP | Statement of Work 2 | 4 | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work Avx | | \$0.00 | \$0.00 |
| Lambda | INTRINSIK CORP | Statement of Work 10 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 3 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 5 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 6 | \$0.00 | | |
| | | | | | \$1,933.00 |
| Lambda | INTRINSIK CORP | Statement of Work 9 | \$0.00 | \$0.00 | |
| Lonafarnib | INTRINSIK CORP | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | INTRINSIK CORP | Master Service Agreement | \$0.00 | \$0.00 | 1 |
| Lonafarnib | INTRINSIK CORP | Statement of Work 14 | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work AMCP | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated July 10, 2020 | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated March 29, 2020 | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated March 4, 2016 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|----------------------|---|---|--------------------------|---------------|------------------|
| Asset Lonafarnib | Contract Counterparty Intrinsik Corp. | Description of Contract of Lease Statement of Work dated May 1,2020 | Avexitide Cure \$0.00 | | Cure \$0.00 |
| Lonaranno | Intrinsik Corp. Intrinsik Health Sciences | Statement of work dated May 1,2020 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Inc. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Zume uu | inVentiv Health Clinique | Substitution of the state | \$0.00 | Ψ0.00 | φσ.σσ |
| Lonafarnib | Inc. | Bioanalytical Services Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | INVENTIV HEALTH | | | | |
| Lonafarnib | CLINIQUE, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | ' 37 ' ' II 1/1 D 1 1' | | | | |
| | inVentiv Health Public Relations LLC, a Syneos | | | | |
| Lonafarnib | Heath Company | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Heath Company | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Avexitide | LLC | Statement of Work dated March 19, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Lonafarnib | LLC | Change Order | \$0.00 | \$0.00 | \$0.00 |
| | INIX/ENITIX/ HE AT TH | | | | |
| | INVENTIV HEALTH PUBLIC RELATIONS, | | | | |
| Lonafarnib | LLC | Statement of Work dated July 16, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LLC | Statement of Work dated July 10, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Lonafarnib | LLC | Statement of Work dated March 19, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Notification Form dated May 15, | | | |
| Lonafarnib | IQVIA | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Notification Form dated October 26, | | | |
| Lonafarnib | IQVIA | 2020 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | TOTAL A | Warranty Letter re: Protocol number EIG- | #A AA | #A AA | ¢0.00 |
| Lonafarnib Lambda | IQVIA IQVIA Biotech LLC | LNF-011 Change Notification Form 2 | \$0.00 \$0.00 | | \$0.00 \$0.00 |
| Lambda | IQVIA Biotech LLC | Change Notification Form 3 | \$0.00 | | \$0.00 |
| Lambda | IQVIA Biotech LLC | Change Notification Form 4 | \$0.00 | | \$0.00 |
| <u> </u> | 14 + 11 12 10 10 11 12 12 | change I terment of the | φοισσ | 1 | φσ.σσ |
| Lambda | IQVIA Biotech LLC | Change Proposal No. 1 | \$0.00 | \$6,151.93 | \$0.00 |
| | | | | 1 | |
| Lambda | IQVIA Biotech LLC | Statement of Work | \$0.00 | | \$0.00 |
| | | | | | |
| Lambda | IQVIA Biotech LLC | Work Order | \$0.00 | | \$0.00 |
| I£ '1. | IOVIA Diete 1 II C | Change Dranges No. 15 | \$0.00 | \$0.00 | # 0.00 |
| Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 15 Letter of Delegation re: Protocol Number | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Clinical AB | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | IQ VIA CIIIICAI AD | PIO-PIM -011 | \$0.00 | φυ.υυ | φυ.υυ |
| | IQVIA Holdings, Inc.; | Consent for Disclosure of Study Information | | | |
| Lonafarnib | ClinAudits, LLC | to Auditors | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|--|-----------------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | TSQM and TSQM Scoring Algorithm | | | |
| | | License (collectively, "Licensed Materials") | | | |
| | | for 150 participants - in connection with | | | |
| Lambda | IQVIA Inc. | protocol EIG-LMD | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | 22-Jun-20 General Services Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | 6/22/2020 General Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Boulevard, Durham, NC 27703 ("IQVIA") | | | |
| | | and relates to the Master Services | | | |
| Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | | * |
| Lonafarnib | IQVIA Inc. | Change Order #1 to Statement of Work | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Change Order #1 Work Order #1 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Change Order #2 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Change Order #2 Work Order #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Change Order #3 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Change Order 1 to General Services | | | |
| Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Change Order 3 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Details of MSA and Work Order: | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Executed Work Order | \$0.00 | \$0.00 | * |
| Lonafarnib | IQVIA Inc. | General Services Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Master Service Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Novation Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Technical Agreement for the Provision of | | | |
| Lonafarnib | IQVIA Inc. | EU Qualified Person | \$0.00 | \$0.00 | \$0.00 |
| | | Work Order #2: SOP Training | | | |
| | | | | | |
| Lonafarnib | IQVIA Inc. | Work Order #2 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Work Order dated May 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Work Order dated May 3, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | IQVIA Ltd.; IQVIA RDS | | | | |
| | Eastern Holdings GmbH; | | | | |
| | IQVIA Tibbi Istatistik | | | | |
| | Ticaret ve Musavirlik | | | | |
| | Limited Sirketi; IQVIA | Letter of Delegation re: Protocol Number | | | |
| Lonafarnib | RDS Ges.m.b.H. | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| | | Authorization to Proceed Agreement dated | | | |
| Lonafarnib | IQVIA RDS INC. | June 30, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | Authorization to Proceed Agreement dated | | | |
| Lonafarnib | IQVIA RDS INC. | October 29, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS Inc. | Change Order No. 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS Inc. | Change Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | Change Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | Details of General Services Agreement: | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|--|------------------|-------------|--------------------|
| 113500 | Some act Sounter party | Details of Master Services Agreement and | Tivestillae Care | Zumsuu cure | Curc |
| Lonafarnib | IQVIA RDS INC. | Authorization to Proceed: | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | Master Services Agreement | \$0.00 | \$0.00 | |
| | IQVIA RDS Ireland | EU Legal Representative of Non-EU | 40.00 | 40.00 | 40.00 |
| Lonafarnib | Limited | Sponsor | \$0.00 | \$0.00 | \$0.00 |
| | | Sponsor's Commitment Form - Clinical Trial | 40.00 | 40.00 | 40.00 |
| Lonafarnib | IQVIA RDS LTD, Israel | with an Investigational Product | \$0.00 | \$0.00 | \$0.00 |
| | ITR Laboratories Canada | S | • | | |
| Avexitide | Inc. | Revised (4) Supplementary Quotation | \$0.00 | \$0.00 | \$0.00 |
| | ITR Laboratories Canada | () 11 | • | | |
| Avexitide | Inc. | Supplementary Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lambda | J. A. [Name onf File] | Services Agreement | \$0.00 | | |
| Lambda | Jordan Feld M.D., MPH; Toronto General Hospital | Letter of Support for the University of Toronto COVID-19 Action Initiative Grant Application | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Kaiser Foundation Research Institute KAROLINSKA | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | UNIVERSITY HOSPITAL | Contract Period | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | KPS Life, LLC | Exhibit A- Work Order No. 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | KPS LIFE, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | KROLL CYBER | | | | |
| Lonafarnib | SECURITY, LLC | LOE | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Kroll Cyber Security, LLC | Statement of Work for Information Security and Computer Forensics Services | \$0.00 | \$0.00 | \$0.00 |
| | KRYOCAL, LLC DBA | | | | |
| Avexitide | KYROSPHERE | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | KRYOCAL, LLC DBA KYROSPHERE | Statement of Understanding | \$0.00 | \$6,562.50 | \$0.00 |
| Lonafarnib | Kymanox Corporation | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated Jul 20, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated July 31, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Liver Care Centre Corporation d/b/a Toronto Liver Centre | LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | LONZA BEND, INC. | Amendment No. 3 to Master Services and Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LONZA BEND, INC. | Amendment No. 1 to the Commercial Supply Agreement | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|-------------|--------------------------------------|--|-----------------------|--------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Amendment No. 2 to the Commercial | | | |
| Lonafarnib | LONZA BEND, INC. | Supply Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 2 | \$0.00 | \$0.00 | |
| | | | | | |
| Lonafarnib | LONZA BEND, INC. | Change Order 2 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 4 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 8 to Statement of Work | \$0.00 | \$0.00 | |
| Lonaranno | LONZA BEND, INC. | Services and Clinical Manufacture | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Agreement dated 15 December, 2022 | \$0.00 | \$0.00 | |
| Lonaranno | ECIVER BEIGH, INC. | rigiteement dated 13 December, 2022 | ψ0.00 | ψ0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 10 April 2023 | \$0.00 | \$0.00 | \$21,000.00 |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |
| I | LONZA DENID INC | Ctatamant of Walls dated 12 June 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 14 August 2020 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 22 April 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 4 December 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 5 April 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 8 May 2018 | \$0.00 | \$0.00 | |
| | LONGA DENID DIG | G | Φ0.00 | ФО ОО | |
| Lonafarnib | LONZA BEND, INC. Lonza Bend; Patheon | Statement of Work dated 9 January 2019 Total Transportation Management ("TTM") | \$0.00 | \$0.00 | |
| Lonafarnib | Canada | Freight Quote | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Change Order #7 to E141-8598 | \$0.00 | \$0.00 | \$0.00 |
| | | | | #0.00 | |
| Lonafarnib | Lonza Pharma & Biotech | Change Order No.1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Amendment 1 to SOW - PN 03706 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 05119 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 6 to SOW E141-8083 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | | Lonza Change Order Form | \$0.00 | \$0.00 | \$0.00 |
| Lonararillo | Lonza i naima & Diotecii | Lonza Change Older Pollii | \$0.00 | \$0.00 | φυ.υυ |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05119 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|---|-----------------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05281 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05895 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001073 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001256 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | | Lonza Change Order to SOW PN0405 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | | Proposed Statement of Work (SOW) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Stability Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-092310 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-136736 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Validation Proposal dated 29 January 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | | Validation Proposal dated 6 April 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-030137" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-031335" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work PN-014163 | \$0.00 | | |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work PN-033823 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-030137" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-031335" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-05281" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work PN-033823 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 3 to Statement of Work "PN-04064" | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|-----------------------------|--|-----------------------|--------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Lonza Pharma & | | | | |
| | BioTech; Bend Research | Change Order No. 3 to Statement of Work | | | |
| Lonafarnib | Inc. | "PN-05281" | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | | | | |
| | BioTech; Bend Research | Change Order No. 3 to Statement of Work | | | |
| Lonafarnib | Inc. | PN-033823 | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | | | | |
| | BioTech; Bend Research | Change Order No. 5 to Statement of Work | | | |
| Lonafarnib | Inc. | "PN-05119" | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | | | | |
| | BioTech; Bend Research | | | | |
| Lonafarnib | Inc. | Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | | | | |
| | BioTech; Bend Research | | | | |
| Lonafarnib | Inc. | Proposed Change Order #1 to Eiger SOW | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | D 10 10 10 10 10 10 10 10 10 10 10 10 10 | | | |
| | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonafarnib | Inc. | August 20, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | B 10t t CW 1 (COW) 1 t 1 | | | |
| T C 1 | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | \$0.00 | #0.00 | #0.00 |
| Lonafarnib | Inc. Lonza Pharma & | August 8, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Dung and Statement of Worls (SOW) dated | | | |
| Lonafarnib | BioTech; Bend Research Inc. | Proposed Statement of Work (SOW) dated | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Lonza Pharma & | May 30, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonafarnib | Inc. | May 8,2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Lonza Pharma & | Way 6,2016 | \$0.00 | \$0.00 | \$0.00 |
| | BioTech; Bend Research | | | | |
| Lonafarnib | Inc. | Validation Proposal | \$0.00 | \$0.00 | \$0.00 |
| Zonaramo | | · unduvion i repesur | φοισσ | φοισσ | \$0.00 |
| Lonafarnib | Luto Research Limited | Quote dated 16 [^] January 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | 7 | 40.00 | 4444 |
| Lonafarnib | Luto Research Limited | Quote dated 1st October 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | , | * | * |
| Lonafarnib | Luto Research Limited | Quote dated 7th April 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | • | | | |
| Lonafarnib | Luto Research Ltd. | Quote for User Testing | \$0.00 | \$0.00 | \$0.00 |
| | M. C., MD [Name on | | | | |
| Lambda | File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | M. T., MD [Name on File] | | \$232.00 | \$0.00 | \$0.00 |
| | | Statement of Work Hepatitis Delta Virus | | | |
| Lambda | M. B. C. [Name on File] | (HDV) | \$0.00 | \$0.00 | \$0.00 |
| | Mary Bordeaux | Statement of Work Hepatitis Delta Virus | | <u>.</u> | <u>.</u> |
| Lonafarnib | Consulting | (HDV) | \$0.00 | | \$0.00 |
| Lonafarnib | MC2 | Change Order No. 1 | \$0.00 | \$0.00 | \$0.00 |
| T C " | 1402 | Standard Service Agreement dated | 00.00 | #0.00 | 40.00 |
| Lonafarnib | MC2 | 2/14/2019 | \$0.00 | \$0.00 | \$0.00 |
| I | MC2 | Standard Service Agreement dated | 00.00 | Φ0.00 | Φ0.00 |
| Lonafarnib | MC2 | 2/27/2020 | \$0.00 | \$0.00 | \$0.00 |

| | 1 | | | | |
|------------|--|---|----------------|--------------|---------------|
| | | | | | |
| Related | | | | | Lonafarnib |
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Į | Standard Service Agreement dated | | | |
| Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Standard Service Agreement dated | | | |
| Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-005 | \$17.79 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-006 | Ψ17.77 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-007 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 2 to LOI AVX-005 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 2 to LOI AVX-006 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | LOI AVX007 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | LOI CHI | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | LOI CHI Children | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Master Service Agreement | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Task Order 3 | | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | | | |
| | LLC; The University of | | \$0.00 | \$0.00 | \$0.00 |
| | Texas M.D. Anderson | | 4 | 4 | 40.00 |
| Lonafarnib | Cancer Center | Amendment dated September 18, 2006 | | | |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 1 | | | |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 2 | | | |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 3 | | | |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 4 | | | |
| T C '1 | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 5 | | | |
| T C '1 | Merck Sharp & Dohme | 1 10 | Φ0.00 | #0.00 | #0.00 |
| Lonafarnib | LLC | Amendment 6 | \$0.00 | \$0.00 | \$0.00 |
| T C 11. | Merck Sharp & Dohme | A | \$0.00 | ¢0.00 | ድ ለ ለለ |
| Lonafarnib | LLC | Amendment 7 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Merck Sharp & Dohme | Amendment to the Pharmacovigilance | \$0.00 | ¢0.00 | ቀ ለ ለላ |
| Lonaiarnio | LLC Merck Sharp & Dohme | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | _ | | | | |
| | LLC; The University of Texas M.D. Anderson | Drug Supply Agragment dated Assessed 20 | \$0.00 | \$0.00 | \$0.00 |
| Lonofomik | | Drug Supply Agreement dated August 29, 2017 | | | |
| Lonafarnib | Cancer Center Merck Sharp & Dohme | License Agreement dated September 30, | + | | |
| Lonafarnib | | 2010 | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | LLC | 2010 | 1 | | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|----------------------|--|--|------------------|-------------|--------------------|
| 113500 | Merck Sharp & Dohme | Description of Contract of Lease | TIVESTORIC CUTC | | Curc |
| | LLC; The University of | | | | |
| | Texas Health Center at | | | | |
| | Tyler; The University of | | | | |
| | Texas Health Science | | | | |
| | Center at Houston; The | | | | |
| | University of Texas Health Science Center at | | \$0.00 | \$0.00 | \$0.00 |
| | San Antonio; The | | | | |
| | University of Texas M.D. | | | | |
| | Anderson Cancer Center; | | | | |
| | The University of Texas | | | | |
| | Health Southwestern | Investigator Initiated Master Clinical Trail | | | |
| Lonafarnib | Medical Center at Dallas | Research Agreement dated July 1, 2004 | | | |
| | Merck Sharp & Dohme | Memo of Understanding (MOU) the | | | |
| Lonafarnib | LLC | Pharmacovigilance Agreement | \$0.00 | \$0.00 | \$0.00 |
| T C '1. | Merck Sharp & Dohme | S. S. to. A | ¢0.00 | £0.00 | ¢0.00 |
| Lonafarnib | LLC Merck Sharp & Dohme | Safety Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | Termination Agreement | \$0.00 | \$0.00 | \$0.00 |
| | M:IID =: d == | Coat Duon coal A DI Sugalian Opplification | | | |
| Lambda | MillBridge Pharmaceutical Science | Cost Proposal - API Supplier Qualification Audit | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Thatmaceancar science | 110010 | ψ0.00 | \$0.00 | φο.σσ |
| | MillBridge | | | | |
| Lonafarnib | Pharmaceutical Science | Cost Proposal - Supplier Qualification Audit | \$0.00 | \$0.00 | \$0.00 |
| | MillBridge | Cost Proposal - Supplier Qualification | | | |
| Lonafarnib | Pharmaceutical Science | Audits | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | MIND+MATTER LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 14 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 15 | \$0.00 | \$0.00 | \$0.00 |
| | | - | , , , , , | | , , , , , |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 16 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 17 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 22 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | MKO Global Partners | Statement of Work | \$0.00 | | |
| Avexitide | MPI Research | Archived Materials Report Quote re: Destruction of Drug | \$0.00 \$0.00 | | |
| Lonafarnib Lambda | Myoderm N/A | Letter of Legal Representative | \$0.00 | | |
| Lonafarnib | NDA GROUP AB | Master Service Agreement | \$0.00 | | |
| | | 20 | \$5.00 | ψ5.00 | \$3.00 |
| I onafarnih | NDA REGULATORY | Work Order | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SCIENCE LTD | Work Order | \$0.00 | \$0.00 | \$0.0 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|----------------------|------------------------------|---|------------------|--------------|--------------------|
| | | Statement of Work Refolding Studies of IL- | | | |
| Lambda | Neurozon LLC | 29 | \$0.00 | \$0.00 | \$0.00 |
| T 1.1 | N HC | Statement of Work Refolding Studies of IL- | #0.00 | #0.00 | #0.00 |
| Lambda Lonafarnib | Neurozon LLC NIH | 29 (Supplement) Author Agreement | \$0.00 \$0.00 | | |
| Lonararnio | NIH | Author Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | NORTH RIM CONSULTING, INC | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 7 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 7 | \$0.00 | | |
| Avexitide | Novocol Pharma | Development Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | OneSource Regulatory LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Oregon Health and | Statement of work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Science University | Notice of Assignment | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Origami Evenements SARL | Contrat de Participation a un Evenement | \$0.00 | \$0.00 | \$0.00 |
| Lanafamih | Origami Evènements | Contract of Posticination in an Event | \$0.00 | \$0.00 | 00.00 |
| Lonafarnib Lambda | SARL Oxford Finance | Contract of Participation in an Event Amendment | \$0.00 \$0.00 | | |
| Lamoua | OATOIU FIIIAIICE | Approval of Second Amendment to Oxford | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | | |
| | | Second Amendment to Loan and Security | 42700 | 42100 | 41.00 |
| Lambda | Oxford Finance | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Oxford Finance | Third Amendment to Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|--|----------------|-------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lambda | Oxford Finance LLC | Amendment Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | | | | |
| Lambda | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | | | | |
| Lambda | Collateral Agent | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance LLC, as Collateral Agent | Second Amendment to Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance LLC, as Collateral Agent | Second Amendment to Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | | | | |
| Lonafarnib | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Oxford Finance, LLC | Success Fee Agreement | \$0.00 | \$0.00 | \$0.00 |
| | PACIFIC PHARMA | Statement of Work describes Services and | | | |
| Lonafarnib | GROUP, LLC | Deliverables to be performed and | \$0.00 | \$0.00 | \$0.00 |
| | Pacific Pharmaceutical | | | | |
| Avexitide | Services and Alcami Company | Services Agreement | \$25,061.25 | \$0.00 | \$0.00 |
| Lanafamih | Palmetto GBA; Ed Gazarkh | Letter re: TPA Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PARTNERS4ACCESS | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | B.V. | Master Services Agreement Statement of Work - EU / Access & | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PARTNERS4ACCESS B.V. | Commercialisation Support dated Dec 23rd, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Statement of Work - EU / Access & | · | · | |
| Lonafarnib | B.V. | Commercialisation Support dated Dec 23rd, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Statement of Work - Lonafarnib EU / Access & Commercialisation Support dated | | | |
| Lonafarnib | B.V. | April 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Statement of Work - Lonafarnib EU / Access & Commercialisation Support dated | | | |
| Lonafarnib | B.V. | July 30th, 2021 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | PARTNERS4ACCESS | Statement of Work - Lonafarnib EU / Access & Commercialisation Support dated | Ф0.00 | Ф0.00 | #0.00 |
| Lonafarnib | B.V. | March 10th, 2021 Statement of Work - Lonafarnib EU | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PARTNERS4ACCESS B.V. | Commercialisation Planning dated March 5, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | | 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Patheon Inc, part of Thermo Fisher Scientific | Change of Scope 10 | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|--------------------------|--|---------------------------------------|------------------|-------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Dath and Inc. most of | | | | |
| Lanafamil | Patheon Inc, part of Thermo Fisher Scientific | Change of Spans 11 | \$0.00 | 90.00 | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 11 | \$0.00 | \$0.00 | \$68,895.30 |
| | Patheon Inc, part of | | | | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 13 | \$0.00 | \$0.00 | |
| Lonaramie | Thermo Tisher Scientific | change of scope 15 | ψ0.00 | ψ0.00 | |
| | Patheon Inc, part of | | | | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 14 | \$0.00 | \$0.00 | |
| Avexitide | PATHEON INC. | GRP Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | PATHEON INC. | Project Proposal dated Dec 13, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | PATHEON INC. | Validation Project Plan | \$0.00 | \$0.00 | \$0.00 |
| Lambda | PATHEON INC. | Statement of Work | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 1 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 11 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 12 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 13 | \$0.00 | · · | 4 |
| Lonafarnib | PATHEON INC. | Change of Scope 14 | \$0.00 | | 4 |
| Lonafarnib | PATHEON INC. | Change of Scope 15 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 16 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 17 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 18 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 19 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 2 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 20 | \$0.00 \$0.00 | | 1 |
| Lonafarnib Lonafarnib | PATHEON INC. | Change of Scope 21 | \$0.00 | | |
| Lonafarnib | PATHEON INC. PATHEON INC. | Change of Scope 22 Change of Scope 23 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 24 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 25 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 27 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 28 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 29 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 1 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 2 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 3 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 4 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 5 | \$0.00 | | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 30 | \$0.00 | | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 31 | \$0.00 | | 4 |
| Lonafarnib | PATHEON INC. | Change of Scope 32 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 34 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 35 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 36 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 37 | \$0.00 | |] |
| Lonafarnib | PATHEON INC. | Change of Scope 4 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 40 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 41 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 43 | \$0.00 | \$0.00 | \$00,075.50 |

| Related | | | | | Lonafarnib |
|--------------------------|---------------------------|--|------------------|---------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | PATHEON INC. | Change of Scope 44 | \$0.00 | | Curc |
| Lonafarnib | Patheon Inc. | Change of Scope 45 | \$0.00 | * | |
| Lonafarnib | Patheon Inc. | Change of Scope 46 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 47 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 5 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 52 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 6 | \$0.00 | \$0.00 | |
| Lonafarnib | Patheon Inc. | Change of Scope 7 | \$0.00 | \$0.00 | |
| Lonafarnib | Patheon Inc. | Change of Scope 8 | \$0.00 | \$0.00 | |
| Lonafarnib | Patheon Inc. | Change of Scope 9 | \$0.00 | \$0.00 | |
| | | Change of Scope COS-55-R0 to Proposal | | | |
| Lonafarnib | Patheon Inc. | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | |
| | | Change of Scope Proposal For Eiger | | | |
| Lonafarnib | Patheon Inc. | Biopharmaceuticals | \$0.00 | \$0.00 | |
| Lonafarnib | Patheon Inc. | Change of Scope to Proposal | \$0.00 | \$0.00 | |
| | | Change of Scope | | | |
| Lonafarnib | Patheon Inc. | Proposal For Eiger Biopharmaceuticals | \$0.00 | * | |
| Lonafarnib | PATHEON INC. | Change Order | \$0.00 | | |
| Lonafarnib | PATHEON INC. | COS 1 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Phase II CTM Manufacturing Proposal | \$0.00 | \$0.00 | |
| | | Product Agreement for Lonafarnib - | | | |
| Lonafarnib | PATHEON INC. | Progeria | \$0.00 | \$0.00 | |
| | | Product Agreement") for Lonafarnib | | | |
| | | Capsules 50 mg and 7S mg ("Product") to | | | |
| Lonafarnib | PATHEON INC. | the Master Service | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Project Proposal dated May 21, 2021 | \$0.00 | \$0.00 | |
| T C '1 | DATELEON DIG | D : . D 1 | ФО ОО | # 0.00 | |
| Lonafarnib | PATHEON INC. | Project Proposal dated September 27, 2021 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Project Proposal March 30, 2020 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Quote dated April 2, 2019 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Quote dated July 09, 2019 | \$0.00 | | |
| Lonafarnib Lonafarnib | PATHEON INC. PATHEON INC. | Quote dated March 25,2019 Quote dated March 27, 2019 | \$0.00 \$0.00 | | |
| Lonafarnib | PATHEON INC. | Quote for Size 4 dated July 09, 2019 | \$0.00 | \$0.00 | |
| Lonararnio | PATHEON INC. | Quote for Size 4 dated July 09, 2019 | \$0.00 | \$0.00 | |
| | | | | | |
| | Patheon Inc.; Gerteis | | | | |
| | Maschinen + | | | | |
| | Processengineering AG; | | | | |
| Lonafarnib | Joseph Lannen | Quotation No 106954 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Joseph Laimen | Quotation No 100934 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Patheon Italia Spa | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 2 I V CAILIUC | i amcon nana spa | Services rigitediffett | \$0.00 | \$0.00 | φυ.υυ |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Change of Scope to Proposal 1 | | \$0.00 | \$0.00 |
| 2 I V CAILIGE | Services ELEC | change of beope to Hoposai i | † | ψ0.00 | ψ0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Change of Scope to Proposal 10 | | \$0.00 | \$0.00 |
| 02111140 | 201 11000 EEC | change of beope to Hoposai 10 | | Ψ0.00 | Ψ0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------------------|--|----------------|-------------|--------------------|
| | Patheon Manufacturing | | | 40.00 | 40.00 |
| Avexitide | Services LLC | Change of Scope to Proposal 14 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Change of Scope to Proposal 14-RO | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 1 R0 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 11 | \$111,092.41 | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 12 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 13 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 2 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 2-R2 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 3 | | \$0.00 | \$0.00 |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 4 | | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 10 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 11 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 15 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 16 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 18 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 9 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope COS P-MNC-101564-R3-COS-07-R0 20200424 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope COS P-MNC-101564-R3- COS-11-R0_20210712 | \$0.00 | | |
| Lambda | Patheon UK Limited | Change of Scope COS P-MNC-101564-R3-COS-11-R0_20210712 (1) | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|---|--|----------------|-------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Change of Scope COS_P-MNC-101564-R3- | | | |
| Lambda | Patheon UK Limited | COS-02-R0_20170119 | \$0.00 | \$0.00 | \$0.00 |
| | | Change of Scope COS-17-R0 to P-MNC- | | | |
| Lambda | Patheon UK Limited | 101564-R3 20220324 | \$0.00 | \$0.00 | \$0.00 |
| | D 4 THZ I ' 4 1 | Change of Scope COS-P-MNC-101564-R3- | Ф0.00 | Ф0.00 | #0.00 |
| Lambda | Patheon UK Limited | COS-08-R3_20210309 | \$0.00 | \$0.00 | \$0.00 |
| T1. 1. | D-41 IIIZ I ''4- 1 | Change of Scope P-MNC-101564-R3-COS- | £0.00 | £0.00 | ¢0.00 |
| Lambda | Patheon UK Limited | 11-R1_20211103 Purchase Order Patheon UK_COS 3 P-MNC- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | 101564-R3 20171129 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patiteon UK Limited | Purchase Order Patheon UK_COS 4 P-MNC- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | 101564-R3 20171115 | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | I atticon OK Limited | 101304-R3_201/1113 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | PATHEON UK LIMITED | Validation Master Plan | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | THIRD OF CIVE ENVIRTED | variation master ran | φσ.σσ | ψ0.00 | ψ0.00 |
| | Patheon UK Limited; | Change of Scope Patheon UK COS 14 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R3 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | | , | * | * |
| | Patheon UK Limited; | Change of Scope Patheon UK COS 20 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4 20220722 | \$0.00 | \$0.00 | \$0.00 |
| | | _ | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 21 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20221025 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 24 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20230209 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 25 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 26 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | D. d. THE T. C. d. | | | | |
| | • | Change of Scope Patheon UK_COS 29-R0 | ФО ОО | Ф0.00 | #0.00 |
| Lambda | Fisher Clinical Services | to P-MNC-101564-R4_20240213 | \$0.00 | \$0.00 | \$0.00 |
| | Doth on LIV Limited | Change of Spane, Bustillad Syminger COS 12 | | | |
| T amalada | Patheon UK Limited; Fisher Clinical Services | Change of Scope: Prefilled Syringes COS 12 to P-MNC-101564-R3 20220125 | | \$0.00 | \$0.00 |
| Lambda | Fisher Clinical Services | Change of Scope: Prefilled Syringes | \$0.00 | \$0.00 | \$0.00 |
| | Patheon UK Limited; | Patheon UK P-MNC-101564-R4-COS-23- | | | |
| Lambda | Fisher Clinical Services | R0 20220922 | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | risher Chinear Services | K0_20220922 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon UK Limited; | Change of Scope: Proposal COS 12 to C- | | | |
| Lonafarnib | Fisher Clinical Services | TRC-121992-R4 20230429 | \$0.00 | \$0.00 | \$0.00 |
| | 20111003 | <u>-</u> | \$5.50 | \$5.50 | \$0.00 |
| | Patheon, Part of Thermo | | | | |
| | Fischer Scientific; | Element Quote Element Quote 20-012- | | | |
| Lonafarnib | Element Toronto | 162900 Rev 1 20200420 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon, Part of | Total Transportation Management ("TTM") | | | |
| Lonafarnib | Thermofisher | Freight Quote 453021 20201202 | \$0.00 | \$0.00 | \$0.00 |

| Related | Combinant Common outre | Description of Contract of Lease | Ameritide Come | Lambda Cuna | Lonafarnib |
|------------|--|--|----------------|-------------------|---------------|
| Asset | Contract Counterparty Patheon; Thermo Fisher | Change of Scope COS-03-R2 to Project | Avexitide Cure | Lambda Cure | Cure |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Patheon; Thermo Fisher | Change of Scope COS-04-R1 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexilide | Patheon; Thermo Fisher | Change of Scope COS-05-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Patheon; Thermo Fisher | Change of Scope COS-06-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexilide | Patheon; Thermo Fisher | Change of Scope COS-07-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexilide | Patheon; Thermo Fisher | Change of Scope COS-08-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexilide | Patheon; Thermo Fisher | Change of scope COS-09-R1 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Patheon; Thermo Fisher | Change of Scope COS-19-R0 to Project | \$0.00 | \$0.00 | \$0.00 |
| Avavitida | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Patheon; Thermo Fisher | Change of Scope COS-38-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP- 114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | Patheon; Thermo Fisher | Change of Scope COS-39-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | 00.00 |
| Lonaranno | Patheon; Thermo Fisher | Change of Scope COS-42-R1 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonararnio | Patheon; Thermo Fisher | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | Change of Scope COS-48-R0 to Proposal No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Patheon; Thermo Fisher | Change of Scope COS-49-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonararnio | | | \$0.00 | \$0.00 | \$0.00 |
| T C 11. | Patheon; Thermo Fisher | Change of Scope COS-50-R0 to Proposal | \$0.00 | #0.00 | ¢0.00 |
| Lonafarnib | Scientific Patheon; Thermo Fisher | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| T C 11. | · · | Change of Scope COS-51-R0 to Proposal | \$0.00 | #0.00 | ¢0.00 |
| Lonafarnib | Scientific Patheon; Thermo Fisher | No. P-TRP-114750-R2 Change of Scope COS-53-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| T C '1 | · · | | ФО ОО | #0.00 | фо о о |
| Lonafarnib | Scientific Patheon; Thermo Fisher | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| T C 11. | , | Change of Scope COS-54-R2 to Proposal | \$0.00 | #0.00 | ¢0.00 |
| Lonafarnib | Scientific Patheon; Thermo Fisher | No. P-TRP-114750-R2 Change of Scope COS-57-R1 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| I | Scientific | No. P-TRP-114750-R2 | \$0.00 | ድስ ሰብ | \$0.00 |
| Lonafarnib | Patheon; Thermo Fisher | Change of Scope COS-58-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lanafamil | Scientific | | \$0.00 | ድስ ሰብ | 00.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | Pharma Solutions USA, | | | | |
| I | · · · · · · · · · · · · · · · · · · · | Camina Aamana | \$0.00 | ድስ ሰሰ | \$745.40 |
| Lonafarnib | Inc. | Services Agreement | \$0.00 | \$0.00 | \$745.49 |
| | Pharmaceutical Research | | | | |
| I amofamil | | Sarvings Agreement | ¢0.00 | _የ ለ ለለ | ¢ (5 0 7 0 |
| Lonafarnib | Associates, Inc. | Services Agreement | \$0.00 | \$0.00 | \$658.79 |
| Lombde | Dharma Directions Inc | Amandment #1 WVO EIC 970 | ¢0.00 | ቀለ ለላ | ቀ ስ ስስ |
| Lambda | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 WKO-EIG-1225 Lambda Interferon CMC | \$0.00 | \$0.00 | \$0.00 |
| T al. 1. | Dhama Dina sti su a Iu | | \$0.00 | ቀለ ለለ | ቀለ ለለ |
| Lambda | PharmaDirections, Inc. | Oversight WKO EIC 1261 CMC Symposity Interferon | \$0.00 | \$0.00 | \$0.00 |
| I aml-J. | Dhommo Dina atia I | WKO-EIG-1361 CMC Support: Interferon | Φ0.00 | ቀለ ለሳ | ሰ ስ ስስ |
| Lambda | PharmaDirections, Inc. | Lambda | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|----------------------------------|---|----------------|-------------|--------------------|
| Lambda | PharmaDirections, Inc. | WKO-EIG-991 Ad hoc Consulting | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 | \$0.00 | \$0.00 | |
| Lonafarnib | PharmaDirections, Inc. | Amendment #2 WKO-EIG-879 | \$0.00 | \$0.00 | \$2,902.50 |
| Lonafarnib | PharmaDirections, Inc. | WKO-EIG-1362 CMC Support: Lonafarnib Progeria Commercial | \$0.00 | \$0.00 | , |
| Lonafarnib | PharmaDirections, Inc. | WKO-EIG-879 Ad hoc Consulting | \$0.00 | \$0.00 | |
| Lonafarnib | PharmaLex UK Services Limited | Estimate | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PHARMALEX US CORPORATION | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Pharmalytical Consulting LLC | Services Agreement | \$3,515.00 | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order 1 to Work Order No. 6 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 6 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Third Amendment to t he Statement of Work #1 Agreement | \$7,848.75 | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 6 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 7 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 8 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 9 | | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 4 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|---------------------|-----------------------------------|---|------------------|-------------|------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | PINEX COMERCIAL | | | | |
| | IMPORTADORA E | | #0.00 | фо оо | # 0.00 |
| Lonafarnib | SERVICOS LTDA | Letter for Authorization | \$0.00 | \$0.00 | \$0.00 |
| | PLATFORM | | | | |
| Lambda | LIFESCIENCES INC. | Professional Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | POLARIS | | | | |
| Avexitide | CONSULTING, INC. | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| | POLARIS | | | | |
| Avexitide | CONSULTING, INC. | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | POLARIS | | | | |
| Avexitide | CONSULTING, INC. | Service Agreement August 15, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | DOL A DIG | | | | |
| Avexitide | POLARIS CONSULTING, INC. | Service Agreement dated 24 Jul 2023 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | · |
| Avexitide | POLARIS CONSULTING, INC. | Service Agreement dated August 8, 2019 | \$0.00 | \$0.00 | \$0.00 |
| TiveAttide | CONSOLITIVO, IIVE. | Service Agreement dated August 6, 2017 | ψ0.00 | ψ0.00 | \$0.00 |
| Lambda | PPD Development, L.P. | LMD Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | PPD Development, L.P. | LIMD Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda Avexitide | PPD Development, L.P. PPS | Proposal Services Agreement | \$0.00 \$0.00 | | \$0.00 \$0.00 |
| TiveAttide | | Services rigreement | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Proforma Element 3 | Order Acknowledgement Order Acknowledgement dated October 23, | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Proforma Element 3 | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | n ni c | | | | |
| | ProPharma Group Holdings, LLC; | Project Assignment #3 Under Consulting | | | |
| Lambda | ProPharma Group, LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | ProPharma Group | | | | |
| | Holdings, LLC; | Project Assignment #4 Under Consulting | | | |
| Lambda | ProPharma Group, LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | ProPharma Group | | | | |
| | Holdings, LLC; | Project Assignment #6 Under Consulting | | | |
| Lonafarnib | ProPharma Group, LLC Q SQUARED | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | SOLUTIONS | Hold Harmless Agreement dated April 7, | | | |
| Lambda | HOLDINGS, LLC | 2021 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset Contract Counterparty Description of C | | | |
|--|--------------------------------------|----------------|------------|
| Asset Contract Counterparty Description of C | | | |
| | 1 4 CT 4 201 C | 1 11 6 | Lonafarnib |
| Q DQ CI II LLD | ontract of Lease Avexitide Cu | re Lambda Cure | Cure |
| | greement dated 17 August | | |
| Lonafarnib HOLDINGS, LLC 2016 | - | .00 \$0.00 | \$250.00 |
| Q SQUARED | | | |
| SOLUTIONS | | | |
| Lonafarnib HOLDINGS, LLC Work Order date | d May 3rd, 2019 \$0 | .00 \$0.00 | \$250.00 |
| Quest Diagnostics Clinical | | | |
| Lonafarnib Laboratories, Inc. Change Order N | b. 1 to Statement of Work \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| Quest Diagnostics Clinical Change Order N | • • | | |
| Lonafarnib Laboratories, Inc. Combined Service | es \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| QUEST DIAGNOSTICS | | | |
| Lonafarnib INCORPORATED Amendment 1 to | Statement of Work 1 \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| QUEST DIAGNOSTICS | | | |
| Lonafarnib INCORPORATED Amendment 2 to | Statement of Work 2 \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| QUEST DIAGNOSTICS | | | |
| | ensing Agreement \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| QUEST DIAGNOSTICS | | | |
| Lonafarnib INCORPORATED Master Services | A greement \$0 | .00 \$0.00 | \$0.00 |
| Estatable International Intern | rgreement | ψυ.υς | ψ0.00 |
| OUEST DIA CNOSTICS | | | |
| QUEST DIAGNOSTICS Lonafarnib INCORPORATED Statement of Wo | rk SOW#2 | .00 \$0.00 | \$0.00 |
| Editatalino invegri ora i Editatalino invegr | 1K 3O W π2 | .00 \$0.00 | \$0.00 |
| Quest Diagnostics | | | |
| Lonafarnib Incorporated Statement of Wo | | .00 \$0.00 | \$0.00 |
| Lonafarnib Quintiles Vietnam LLC EIG-LNF-011 | ion re: Protocol Number | .00 \$0.00 | \$0.00 |
| Lambda R3LS, LLC Statement of Wo | | .00 \$0.00 | |
| Tio 25, 226 | Ψ0 | , | \$0.00 |
| RAMBAM HEALTH | | | |
| Lambda CORPORATION Clinical Trial Ag | reement \$0 | .00 \$0.00 | \$0.00 |
| RAMBAM HEALTH | | | |
| Lonafarnib CORPORATION Clinical Trial Ag | reement \$0 | .00 \$0.00 | \$0.00 |
| | | | |
| REGULATORY Longfornib DHARMA NET SPL Congulting Agree | am ant | 00 | \$0.00 |
| Lonafarnib PHARMA NET SRL Consulting Agre | ement \$0 ent #1 Under Consulting | .00 \$0.00 | \$0.00 |
| Lonafarnib Regulatory Pharma Net srl Agreement | 9 | .00 \$0.00 | \$0.00 |

| Dolotod | | | | | Lonafarnib |
|------------------------|------------------------|--|------------------|------------------|------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| 115500 | Contract Counterparty | Description of Contract of Bouse | Tivesiciae Gare | Zamsua cure | Cure |
| | Research Triangle | | | | |
| | Institute DBA RTI | | | | |
| Lonafarnib | International | Proposal 0282201.557 | \$0.00 | | * |
| Avexitide | RHO, INC. | LOI | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | RHO, INC. | LOI Extension | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | RHO, INC. | LOI Second Extension | \$0.00 | \$0.00 | \$0.00 |
| Avexitide Avexitide | RHO, INC. RHO, INC. | LOI Third Extension Master Services Agreement | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Avexide | ROTE LISTE Service | Iviaster Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | Contract for the Publication | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | Gillett | Contract for the Fubication | ψ0.00 | ψ0.00 | Ψ0.00 |
| | RRD INTERNATIONAL, | | | | |
| Avexitide | LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T 11 | RRD INTERNATIONAL, | | фо оо | #0.00 | Ф0.00 |
| Lambda | LLC | Letter for Authorization PA 3 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNATIONAL | Project Agreement 3 to the Product | | | |
| Lambda | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| Laiiioua | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Second Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 15, February 28, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 15, January 25, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 16, December 9, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | LLC | Work Order 10, December 7, 2022 | \$0.00 | ψ0.00 | Ψ0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 16, March 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 16, March 11, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | LLC | work Order 6, effective August 3, 2017 | φυ.υυ | φυ.υυ | φυ.υυ |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 9, effective September 5, 2017 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------|--|----------------|-------------|--------------------|
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 9, effective September 5, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 10 provided for Strategic Regulatory Consultation and FDA Liaison Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 15 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 16 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 17 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 18 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 8 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 9 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 9 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amended and Restated Work Order No. 11 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 1 to 1 20191210 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 1 to 2 20191002 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 1 to 3 20191022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to PA 1 - Interim CO 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to PA 1 - Interim CO 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to PA 1 - Interim CO 3 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure | |
|------------------|---------------------------|--|----------------|-------------|--------------------|--|
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to PA 1 - Interim CO 4 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 2 to TO 2 202003003 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 2 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to PA 1 - Interim CO 5 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to PA 1 - Interim CO 6 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to Project Agreement 1 Interim CO 4 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to TO 2 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 3 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Contractor Agreement | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Development Advisory Committee Amendment 2 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Development Advisory Committee Amendment 2 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Eiger Progeria PA2 Amendment 1 First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | Eiger Progeria PA2 Amendment 1 Interim CO2 | 73,000 | 7.700 | 44.00 | |
| Lonafarnib | LLC | Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 | |

| B. I I | | | | | |
|------------------|---------------------------|---|------------------|-------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| 113500 | contract counterparty | Eiger Progeria PA2 Amendment 1 Interim | TIVERITURE CUITE | | Curc |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Eiger Progeria PA2 Amendment 2 Interim | | | |
| Lonafarnib | LLC | CO1 Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Fighth Amondayant to Tools Orden 1 to the | | | |
| Lonafarnib | LLC | Eighth Amendment to Task Order 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | BBC | Transfer independent Constactor rigiteement | ψ0.00 | \$0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | | | | 40.00 |
| Lonafarnib | LLC | Fifth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Fifth Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDNATIONAL | | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | LLC | I list Amendment to I Toject Agreement | \$0.00 | ψ0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Task Order 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | LLC | Master independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL | First Amendment to Task Order 10 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Task Order 2 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | LLC | Master independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 2 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 5 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 1 | RRD INTERNATIONAL, | | Φο οο | ФО ОО | #0.00 |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 6 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNIATIONIAT | First Amondment to Teels Order 9 to the | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Task Order 8 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararnio | LLC | iviasiei independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | ~ | | Lonafarnib |
|------------|-----------------------|--|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | First Amendment to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | , | First Amendment to Work Order No. 12 to | | | |
| Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | First Amendment to Work Order No. 13 to | | | |
| Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | First Amendment to Work Order No. 14 to | фо. о о | фо. о о | #0.00 |
| Lonafarnib | LLC | the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | фо. о о | фо. о о | #0.00 |
| Lonafarnib | LLC | Fourth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Fourth Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Fourth Amendment to the Amended and | | | |
| Lonafarnib | LLC | Restated Work Order | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Interim CO 1 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | The state of the page of the state of the st | фо. о о | фо. о о | #0.00 |
| Lonafarnib | LLC | Interim CO 1 to PA 2 (MM TO 12) | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Interim CO 2 to PA 2 (MM TO 14) | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Interim CO 3 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Interim CO 3 to PA 2 (MM TO 22) | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | T | фо. о о | фо. о о | #0.00 |
| Lonafarnib | LLC | Interim CO 4 to PA 2_20200701 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Letter for Authorization for PA 2 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Letter for Authorization for RRD to | | | |
| Lonafarnib | LLC | Commence Activities Under Work Order 11 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | * | * | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Master Independent Contractor Agreement | | | |
| Lonafarnib | LLC | effective December 5, 2018 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------|---|----------------|-------------|--------------------|
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective February 22, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | | Master Independent Contractor Agreement effective March 1, 2018 | \$0.00 | | |
| Lonafarnib | | Master Independent Contractor Agreement effective May 10, 2019 | \$0.00 | | |
| Lonafarnib | | Master Independent Contractor Agreement effective May 24, 2018 | \$0.00 | | |
| Lonafarnib | | Master Independent Contractor Agreement effective October 24, 2018 | \$0.00 | | |
| Lonafarnib | RRD INTERNATIONAL, | Product Development Agreement | \$0.00 | | |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | |
| Lonafarnib | RRD INTERNATIONAL, | Project Agreement 1 to the Product Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Project Agreement 2 to the Product Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Project Agreement 3 to the Product Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Second Amendment to Task Order 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Second Amendment to the Amended and Restated Work Order | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Second Amendment to Work Order No. 12 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Sixth Amendment to Task Order 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Statement of Work 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Task Order 1 to the Master Independent Contractor Agreement dated December 5, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Task Order 1 to the Master Independent Contractor Agreement dated February 22, 2019 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|---|-----------------------|-------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | DDD DITEDNIATIONAL | Task Order 1 to the Master Independent | | | |
| T C 11 | RRD INTERNATIONAL, | Contractor Agreement dated March 11, | Ф0.00 | Ф0.00 | #0.00 |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Task Order 1 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | LLC | Task Order 1 to the Master Independent | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | LLC | Task Order 10 to the Master Independent | Ψ0.00 | Ψ0.00 | \$0.00 |
| 1 | RRD INTERNATIONAL, | Contractor Agreement dated effective | | | |
| Lonafarnib | LLC | October 29, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Londianio | EEC | Task Order 11 to the Master Independent | ψ0.00 | ψ0.00 | ψ0.00 |
| 1 | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 12 to the Master Independent | 40.00 | 40.00 | 44144 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 13 to the Master Independent | * | , | * |
| 1 | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 2 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 2 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated February 22, | | | |
| Lonafarnib | LLC | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | Task Order 2 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 2 to the Master Independent | | | |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 3 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDNIATIONAL | | | | |
| T C 11 | RRD INTERNATIONAL, | Task Order 3 to the Master Independent | ФО ОО | Ф0.00 | #0.00 |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNATIONAL | Task Order 3 to the Master Independent | | | |
| I | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNATIONAL | Task Order 4 to the Master Independent | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LLC | Task Order 4 to the Master Independent | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LLC | 2010 | φ0.00 | φ0.00 | φυ.υυ |
| | RRD INTERNATIONAL, | Task Order 5 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LLC | Contractor Agreement dated iviay 24, 2016 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------|---|----------------|-------------|--------------------|
| Lonafarnib | RRD INTERNATIONAL, LLC | Task Order 5 to the Master Independent Contractor Agreement dated October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Task Order 6 to the Master Independent Contractor Agreement dated October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Task Order 7 to the Master Independent Contractor Agreement dated May 24, 2018 Task Order 7 to the Master Independent | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | - | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Contractor Agreement dated October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Task Order 9 to the Master Independent Contractor Agreement dated October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Third Amendment to Project Agreement 3 dated March 1, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Third Amendment to Project Agreement dated March 1, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | dated March 1, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Third Amendment to the Amended and Restated Work Order dated February 15, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Work Order 5, effective March 31, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Work Order 5, effective March 31, 2016 | \$0.00 | | |

| Related | | | | | Lonafarnib |
|-------------|---|---|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Work Order No. 10 provided for Strategic | | | |
| | RRD INTERNATIONAL, | Regulatory Consultation and FDA Liaison | | | |
| Lonafarnib | LLC | Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 12 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 13 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 14 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 2 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 21 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 24 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 4 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 5 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD International, LLC; | | | | |
| Lonafarnib | AltaSciences | Quotation Description | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RTI INTERNATIONAL | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | SAFEBRIDGE | | | | |
| Avexitide | CONSULTANTS, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | SafeBridge Regulatory | | | | |
| Lambda | and Life Sciences Group | Quotation 220503.0560 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | SafeBridge Regulatory and Life Sciences Group | Quotation 220503.0560 | \$0.00 | \$0.00 | \$0.00 |
| Lonararillo | • | Quotation 220505.0500 | \$0.00 | \$0.00 | \$0.00 |
| | SAFEGUARD WORLD INTERNATIONAL | | | | |
| Lonafarnib | LIMITED | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | S. B. I. [Name on File] | Sam Brown Proposal | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|--------------------------|--|--|------------------|------------------|--------------------|
| | Sard Verbinnen & Co., | · | | | |
| Lambda | LLC | Engagement Letter | \$0.00 | \$0.00 | \$0.00 |
| | Scientific Committee of General Hospital of | | | | |
| Lonafarnib | Athens "Laiko" | GDPR Statement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SECRETARY OF VETERAN AFFAIRS | Acknowledgement Letter | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SECRETARY OF VETERAN AFFAIRS | Interim Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SECRETARY OF VETERAN AFFAIRS | - | \$0.00 | \$0.00 | |
| Lonararmo | SENTYNL | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | THERAPEUTICS, INC. | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SGS Analytics Germany GmbH SGS Analytics Germany | Quotation No. 210409-01 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | Quotation No. 210514-01 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SGS Analytics Germany GmbH | Quotation No. 210602-01 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Shaare Zedek Medical Center; Dr. Lurie Yoav | Form 4a | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Sharp Corporation | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote dated 3/3/22 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote dated 8/11/21 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote re: Concept | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote re: Groninger Tooling | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Global Design Center | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | Sheba Fund for Health Services and Research | | Ф0.00 | Ф0.00 | #0.00 |
| Lambda | ` /: ~ | Clinical Study Agreement Letter of Indemnification re: Protocol No. | \$0.00 | \$0.00 | |
| Lonafarnib Lonafarnib | Ltd. Sherpa | EIG-LNF-011 Change Order dated 12/5/2016 | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Lonafarnib | Sherpa | Change Order dated 12/3/2016 Change Order dated 12/12/2016 | φυ.00 | φυ.00 | \$0.00 |

| nterparty Do | | | | |
|-----------------|---|--|--|---|
| ntownouty D | | | | |
| matawa a wither | | | | Lonafarnib |
| | escription of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | tatement of Work re: ENDO 2022 - | #0.00 | #0.00 | фо о о |
| | raphics Refresh | \$0.00 | \$0.00 | \$0.00 |
| | | \$0.00 | ¢0.00 | \$0.00 |
| | | | · | |
| | | · · | - | |
| | | | | |
| Α, | dote 1/21 12000 fe. Eonatarino / 11 1 | ψ0.00 | ψ0.00 | ψ0.00 |
| al Center: | | | | |
| - | orm 4a | \$0.00 | \$0.00 | \$0.00 |
| | | 44144 | 4 3 1 3 2 | ****** |
| sity | | | | |
| • | linical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| search Pr | roposal | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| | | | | |
| LC A1 | mendment #1 to Work Order 1 LMD001 | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| | | | | |
| LC A1 | mendment #3 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| | 1 424 W 1 0 1 1 | #0.00 | #0.00 | фо о о |
| LC A1 | mendment 2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| AI TD AII C | | | | |
| | mandmant 5 to Wark Order 1 | 00.02 | \$0.00 | \$0.00 |
| LC A | mendment 3 to work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| AL TRAILS | | | | |
| | mendment to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| 20 71 | mendinent to Work Gradi | ψ0.00 | ψ0.00 | ψ0.00 |
| AL TRAILS | | | | |
| LC St | tatement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | <u>v</u> | | | |
| AL TRAILS | | | | |
| LC A1 | mendment #1 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| | | | | |
| LC A1 | mendment #1 to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | |
| | 1 | #0.00 | Φ0.00 | #0.00 |
| LC A1 | mendment #2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| AL TRAILC | | | | |
| | tatament of Worls | 00.00 | \$0.00 | \$0.00 |
| LC Si | tatement of work | \$0.00 | \$0.00 | \$0.00 |
| AL TRAILS | | | | |
| | tatement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| 51 | miomoni of work Duted August 27, 2013 | φυ.υυ | φ0.00 | φυ.υυ |
| AL TRAILS | | | | |
| | tatement of Work LNF004 | \$0.00 | \$0.00 | \$0.00 |
| | Q Q Q al Center; ad Formula Sity er C esearch Properties AL TRAILS LC A AL TRAILS LC Si AL TRAILS | resity r Clinical Trial Agreement Search Proposal AL TRAILS LC Amendment #1 to Work Order 1 LMD001 AL TRAILS LC Amendment #3 to Work Order 1 AL TRAILS LC Amendment 5 to Work Order 1 AL TRAILS LC Amendment to Work Order 1 AL TRAILS LC Amendment to Work Order 1 AL TRAILS LC Amendment to Work Order 5 AL TRAILS LC Statement of Work Dated August 24, 2015 AL TRAILS LC Amendment #1 to Work Order 1 AL TRAILS LC Amendment #1 to Work Order 1 AL TRAILS LC Amendment #2 to Work Order 1 AL TRAILS LC Statement of Work AL TRAILS LC Statement of Work Dated August 24, 2015 AL TRAILS LC Statement of Work Dated August 24, 2015 | Quantitation of Nitrosamine DIPNA \$0.00 | Quantitation of Nitrosamine DIPNA \$0.00 \$0.00 Quote N21-11680 re: Lonafamib API \$0.00 \$0.00 Quote N21-11681 re: Lonafamib API \$0.00 \$0.00 Quote N21-12606 re: Lonafamib API \$0.00 \$0.00 Quote N21-12606 re: Lonafamib API \$0.00 \$0.00 al Center; ad Form 4a \$0.00 \$0.00 sity rr Clinical Trial Agreement \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 LMD001 \$0.00 \$0.00 AL TRAILS LC Amendment #3 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment 2 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment 5 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment 40 \$0.00 \$0.00 AL TRAILS LC Amendment 41 to Work Order 5 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Amendment #1 to Work Order 1 \$0.00 \$0.00 AL TRAILS LC Statement of Work Order 1 \$0.00 \$0.00 AL TRAILS LC Statement of Work Order 1 \$0.00 \$0.00 AL TRAILS LC Statement of Work Order 1 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work Dated August 24, 2015 \$0.00 \$0.00 AL TRAILS LC Statement of Work |

| Related | | | | | Lonafarnib |
|------------|---|--|-----------------------|-------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Quote re: SG-Lab Mechanical or | фо оо | фо оо | #0.00 |
| Avexitide | Stevanato Group S.p.A. | Performance Test | \$0.00 | | |
| Lambda | Syneos | Statement of Work Dated August 7, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Syneos Health Communications, Inc. f/k/a inVentiv Health Public Relations, LLC | Statement of Work re: Submission | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Aga Khan University | Data Transfer Agreement V | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | The Aga Khan Oniversity | Letter re: Change to the Data Transfer | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Aga Khan University | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | University; Colleen Craig; | Asset Purchase Agreement by and between Colleen Craig and Tracey McLaughlin (the "Original Licensees"), on the one hand, and Eiger BioPharmaceuticals, Inc. ("Eiger"), dated 25 September 2015 (the "APA"), by which Eiger became party to that certain License Agreement by and between Colleen Craig and Tracey McLaughlin on the one hand and The Board of Trustees of the Leland Stanford Junior University ("Stanford"), dated May 4, 2015 The Amendment to the APA License Agreement among Stanford, the Original Licensees, and Eiger dated as of 25 September 2015 ("Amendment No. 1") (this License Agreement Amendment No. 1 was executed together with the APA as Exhibit | \$1,116.00 | | |
| Avexitide | the Leland Stanford Junior University; Colleen Craig; | , , | | \$0.00 | |
| Avexitide | Tracey McLaughlin The Board of Trustees of the Leland Stanford Junior University; Colleen Craig; | May 11, 2018 ("Amendment No. 2") The Second Amendment to the APA between the Original Licensees and Eiger, dated 22 May, 2018 (the "APA Amendment 2" and, together with the APA, Amendment No. 1, and Amendment No. 2, the | | \$0.00 | \$0.00 |
| Avexitide | Tracey McLaughlin | "Agreements") | | \$0.00 | \$0.00 |
| Avexitide | The Children's Hospital of Philadelphia | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | THE CHILDRENS HOSPITAL OF PHILADELPHIA | Indemnification Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|------------------------------------|---|----------------|-------------|--------------------|
| Lambda | The Doctor's Laboratory Limited | Change Order 1 NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 1 PK | \$0.00 | | |
| | The Doctor's Laboratory | | | | |
| Lambda | Limited The Doctor's Laboratory | Change Order 2 IR | \$0.00 | | |
| Lambda | Limited | Change Order 2 NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 2 PK | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Pricing Agreement NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Pricing Agreement PK | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing / Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 10th September 2018 | \$0.00 | | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 14 September 2018 | \$0.00 | | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 14 September 2018 | \$0.00 | | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 18th January 2017 | \$0.00 | | \$10,000.00 |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 18th January 2017 | \$0.00 | | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 2021 | \$0.00 | | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|--|----------------|-------------|--------------------|
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated December 2014 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement Trials | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Trials Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory | Trials Pricing/Supplemental Agreement | \$0.00 | | |
| Lonafarnib | The Johns Hopkins University | Sponsored Research Agreement | \$0.00 | | \$0.00 |
| Lonafarnib | The Liver Center and Naranjargal Dashdorj | ADDITIONAL BUDGET AGREEMENT | \$0.00 | | \$0.00 |
| Lambda | The Liver Unit Hadassah University Hospital, Ein Kerem | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Liver Unit Hadassah University Hospital; The Research Fund of Hadassah Medical Organization (R.A.) | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Progeria Research Foundation | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific Thermo Fisher Scientific, | Quotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Inc. Thermo Fisher Scientific, | Quote re: Sourcing of Medicinal Product | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Inc.; Fisher Clinical Services | Quote | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #200616-01-SF | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|---|--|-----------------------|--------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | Thermo Fisher Scientific; | | | | |
| | Patheon UK Limited, Part | | | | |
| | of Thermo Fisher | | | | |
| Lambda | Scientific | Quotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | Thermo Fisher Scientific; | | | | |
| | Patheon UK Limited, Part of Thermo Fisher | | | | |
| Lambda | Scientific | Ouotation #220328-01-SF | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Total Transport | Netherlands Hub Freight Quote dated 06 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Management | April 2022 | \$0.00 | \$0.00 | \$0.00 |
| <u> </u> | Total Transport | Netherlands Hub Freight Quote dated 15 | \$0.00 | Ψ0.00 | \$0.00 |
| Lambda | Management | March 2022 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated | | | |
| Lambda | Management | 4/10/2022 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated Dec | | | |
| Lambda | Management | 13, 2021 | \$0.00 | \$0.00 | \$0.00 |
| T 1.1 | Total Transport | Netherlands Hub Freight Quote dated Feb | ФО ОО | #0.00 | ФО ОО |
| Lambda | Management | 14, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Total Transport Management | Netherlands Hub Freight Quote dated Jan 27, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Total Transport | Netherlands Hub Freight Quote dated June | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| <u> </u> | Total Transport | Netherlands Hub Freight Quote dated June | \$0.00 | φοισσ | \$0.00 |
| Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated | | | |
| Lonafarnib | Management | August 18, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated Nov | | | |
| Lonafarnib | Management | 2, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | TRANSPERFECT | | | | |
| | TRANSLATIONS | | | | |
| Lambda | | Solutidns Agreement | \$0.00 | \$0.00 | \$0.00 |
| <u> </u> | ii (TEIG WIII o WiE, II (e) | 2 of avious 11g. common | \$0.00 | Ψ0.00 | \$0.00 |
| | TRG Communications | | | | |
| Lonafarnib | LLC | Services Agreement | \$0.00 | \$0.00 | \$36,000.00 |
| | | | | | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 20-Jul-22 | \$0.00 | | \$0.00 |
| T 1 1 | mil of the training | D: 0 4/2 14 120 1 122 | #0.00 | | ф0.00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 20-Jul-22 | \$0.00 | | \$0.00 |
| Lambda | Triples Clinical Triple I +4 | Price Quotation dated 21-Feb-23 | \$0.00 | | \$0.00 |
| Lambda | Thaiog Chinical Thais Ltd | 1 The Quotation dated 21-Feb-23 | \$0.00 | \$3,760.00 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 3-Mar-22 | \$0.00 | | \$0.00 |
| | Etailo Eta | | ψ0.00 | | \$0.00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 7-Dec-22 | \$0.00 | | \$0.00 |

| Related | | | | | Lonafarnib |
|----------------------|--------------------------------|--|------------------|------------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| Lambda | Trialog Clinical Trials Ltd | Statement of Work dated April 2020 | \$0.00 | | \$0.00 |
| | | • | | | · |
| | | | | | 40.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated June 7, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote 2 | \$0.00 | \$0.00 | \$0.00 |
| | B | Trialog Clinical Trials Quote dated | 40.00 | 4 | 40000 |
| Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| - 2 " | | Trialog Clinical Trials Quote dated | | | |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Thurs Chinear Thurs Eta | Trialog Clinical Trials Quote dated | ψ0.00 | ψ0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | 7/31/2021 | \$0.00 | \$0.00 | \$0.00 |
| | | Trialog Clinical Trials Quote dated Jan 24, | | | |
| Lonafarnib | Trialog Clinical Trials Ltd | 2021 Trialog Clinical Trials Quote dated Jan 26, | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | - | \$0.00 | \$0.00 | \$0.00 |
| Lonarano | Thurs Chinear Thurs Eta | Trialog Clinical Trials Quote dated June 27, | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | W. L. O. L. EVG 200 V. L. M. | | | |
| Lonafarnib | Triangle Biostatistics, LLC | Work Order: EIG-300 Under Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Triangle Insights | Services Agreement | \$0.00 | | |
| Lamoua | Triangle Insights Group | Services rigitedment | ψ0.00 | ψ0.00 | \$0.00 |
| Avexitide | LLC | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Triangle Insights Group, | | | | |
| Lambda Lonafarnib | LLC TriangleInsights | Master Services Agreement Statement of Work | \$0.00 \$0.00 | \$0.00 \$0.00 | |
| Lonararino | TRILIGENT | Statement of work | \$0.00 | \$0.00 | \$0.00 |
| Lambda | INTERNATIONAL | Project Assignment 5-1 | \$0.00 | \$0.00 | \$0.00 |
| | TRILIGENT | | | | |
| Lonafarnib | INTERNATIONAL | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lanafamih | TRILIGENT | Project Assignment 3 | \$0.00 | \$0.00 | 00.00 |
| Lonafarnib | INTERNATIONAL TRILIGENT | Project Assignment 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | INTERNATIONAL | Project Assignment 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | Trustees of the University | | | | |
| | of Pennsylvania ("Penn") | | | | |
| | and the Children's | | | | |
| Avexitide | Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cura | Lonafarnib Cure |
|------------------|--|---|------------------|-------------|--------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitude Cure | Lambua Cure | Cure |
| | | | | | |
| | | | | | |
| | Trustees of the University | | | | |
| | of Pennsylvania ("Penn") and the Children's | | | | |
| T 1 . 1 . | una una cumaran s | L'anna Amanant Istalia (Mass 10, 2010 | ¢0.00 | ¢0.00 | ¢0.00 |
| Lambda | Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | Trustees of the University | | | | |
| | of Pennsylvania ("Penn") | | | | |
| | and the Children's | | | | |
| Lonafarnib | Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work 1 to the Master | | | |
| Lonafarnib | TWO LABS, LLC | Independent Contractor Agreement | \$0.00 | | |
| Lonafarnib | TWO LABS, LLC | Statement of Work 2 | \$0.00 | | |
| Lonafarnib | TWO LABS, LLC | Statement of Work 3 | \$0.00 | | |
| Lonafarnib | TWO LABS, LLC | Statement of Work 4 | \$0.00 | | |
| Lambda | Tx Evidence, Inc | SAP Statistical Analysis Plan 3 | \$0.00 | | |
| Lonafarnib | Tx Evidence, Inc | Master Service Agreement | \$0.00 | · | |
| Lonafarnib | Tx Evidence, Inc | SAP Statistical Analysis Plan 1 | \$0.00 | | |
| Lonafarnib | Tx Evidence, Inc | SAP Statistical Analysis Plan 2 | \$0.00 | | |
| Lonafarnib | Tx Evidence, Inc | Term of Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Tu Evidence Inc | Eiger Study EIG-LMD-001 Submission | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Tx Evidence, Inc. Tx Evidence, Inc. | New Work Proposal Statical Data Analysis Proposal | \$0.00 \$0.00 | | |
| Lambda | 1x Evidence, inc. | Eiger Study EIG-300 Submission New Work | | \$0.00 | \$0.00 |
| Lonafarnib | Tx Evidence, Inc. | Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | United Biosource LLC | Порова | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | (UBC) | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Londidinio | University Health | Waster Services rigicement | ψ0.00 | ψ0.00 | ψ0.00 |
| Lambda | Network | Data Use Agreement | \$0.00 | \$0.00 | \$0.00 |
| Zumouu | | Daw Court Statement | φοισσ | ψ0.00 | \$0.00 |
| | | | | | |
| | University Medical Center | | | | |
| | of Johannes Gutenberg- | | | | |
| Lambda | University Mainz | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | University of Oxford | Notice of Assignment | \$0.00 | \$0.00 | \$0.00 |
| | | Clinical Trial Agreement, dated October 4, | | | |
| | | 2016, by and among Seller, Vince & | | | |
| | | Associates Clinical Research, P.A., and | | | |
| | Vince & Associates | Vince & Associates Clinical Research, Inc. | | | |
| Avexitide | Clinical Research | (the "Vince CTA"). | \$0.00 | \$0.00 | \$0.00 |
| | | Consent for Photography / Authorization for | ψο.00 | \$3.00 | \$0.00 |
| Avexitide | Vito Vento | Publication | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Walgreens Boots Alliance | Global Estimated Quotation | \$0.00 | \$0.00 | \$0.00 |
| | Washington University in | | | | |
| Lonafarnib | St Louis | Notice of Assignment | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---|----------------------------------|----------------|-------------|--------------------|
| Lonafarnib | WCCT GLOBAL, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | WCG Clinical Inc. | Services Agreement | \$0.00 | \$0.00 | \$1,312.50 |
| Lonafarnib | WOVEN DATA, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | WOVEN DATA, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Xcelience | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Xenotech LLC dba Sekisui Xenotech, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |

Redline to Initial Cure Schedule

| Related | | | | | Lonafarnib |
|-----------------------|------------------------|---|-------------------|-------------------|--------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cura | Cure |
| Asset | Contract Counterparty | Description of Contract of Bease | Aveature Cure | Lambua Curc | Curc |
| Lonafarnib | 3D Communications IIC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | 3D Communications, EEC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | 3D Communications, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lambda | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | | |
| Lonafarnib | ABBVIE INC. | Asset Purchase Agreement | \$0.00 | \$0.00 | • |
| | | 8 | 40.00 | 4 4 1 4 4 | 40100 |
| | ABF Pharmaceutical | | | | |
| Lonafarnib | Services | Change Order | \$0.00 | \$0.00 | \$0.00 |
| | | | * | * | , , , , , |
| | ABF Pharmaceutical | | | | |
| Lonafarnib | Services GmbH | Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | ABF Pharmaceutical | | | | |
| Lonafarnib | Services GmbH | Project Proposal | \$0.00 | \$0.00 | \$ 0.00 |
| | | | | | |
| Lonafarnib | Absorption Systems LP | In Vitro Study Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Accenture, LLP | Services Agreement | \$0.00 | \$5,399.00 | \$0.00 |
| | | Amendment One to the Master Services | | | |
| Lonafarnib | Accenture, LLP | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Accenture, LLP | Change Order 2 to SOW 3 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Accenture, LLP | Change Order 3 to SOW 3 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | I I I I | | ФО ОО | #A AA | Ф0.00 |
| Lonafarnib | Accenture, LLP | Change Order Form No. 7 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | A acomtuma I I D | Change Order Form No. 0 | \$0.00 | \$0.00 | \$0.00 |
| Lonatarnio | Accenture, LLP | Change Order Form No. 9 Change Order No. 003 Amending the | \$0.00 | \$0.00 | \$0.00 |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Accenture, LLI | Change Order No. 004 Amending the | φ0.00 | φ0.00 | \$0.00 |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | recenture, EEr | Change Order No. 005 Amending the | ψ0.00 | Ψ0.00 | ψ0.00 |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | 11000110010, 221 | Change Order No. 006 Amending the | ψ0.00 | \$0.00 | \$0.00 |
| | | Regulatory and Clinical Services Amended | | | 1 |
| | | and Restated Statement of Work No. 5 | | | 1 |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Change Order No. 008 Amending the | \$3.00 | \$5.00 | \$0.00 |
| | | Regulatory and Clinical Services Amended | | | |
| | | and Restated Statement of Work No. 5 | | | |
| Lonafarnib | Accenture, LLP | Dated February 1, 2019. | \$0.00 | \$0.00 | \$0.00 |
| | , .==- | J / | 4.100 | 45.00 | |

| Related | | | | | Lonafarnib |
|-----------------------|--|--|-------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | Accenture, LLP | Data Privacy and Security Exhibit | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Scope of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP | Scope of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Accenture, ELI | Scope of work 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Accenture, LLP ADAMAS Consulting | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | Project Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Addison Whitney | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work describes Services and | | | |
| Lonafarnib | Addison Whitney | Deliverables to be | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | Addison Whitney | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Addison Whitney LLC, a Syneos Health Group Company | Peginterferon Lambda Package & Label Design SOW | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Additional Submission Document for Lonafarnib Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Lonafarnib (Progeria and HDV) Global Brand Name Development and Validation Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Lonafarnib for HDV EMA co-pack regulatory Consulting Validation Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Addison Whitney LLC, a Syneos Health Group Company | Webpage Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Company | Weepinge Development Proposur | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Addison Whitney, LLC Agenzia Italiana del | Amendment No. 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Farmaco | Access Request Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Agenzia Italiana del | Letter re: Declaration on Public Contributions and Incentives for | | | |
| Lonafarnib | Farmaco | Developmental of the Medical Product | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Agenzia Italiana del Farmaco | Letter re: Developmental of Medical Product | \$0.00 | \$0.00 | \$0.00 |
| | | | \$3.00 | \$3.00 | \$3.00 |
| I and from 11 | AIFA – Agenzia Italiana | A coord Downset A construct | Φ0.00 | # 0.00 | #0.00 |
| Lonafarnib | del Farmaco | Access Request Agreement Statement of Work dated April 14, 2020 | \$0.00 \$0.00 | \$0.00 | |
| Lonafarnib | ALLCARE | Statement of work dated April 14, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | ALLCARE PLUS- PHARMACY LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------------------|------------------------------|--|-------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib Lonafarnib | AllCare Plus Pharmacy | | | | |
| | LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | AllCare Plus Pharmacy, | | | | |
| | LLC | Amendment | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work re: Project No. EGE P2 | #0.00 | #0.00 | Ф0.00 |
| Lonafarnib | Altasciences | 672 Statement of Work re: Project No. EGE P3- | \$0.00 | \$0.00 | \$0.00 |
| Lonoformile | Altasciences | 019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | Antasciences | Statement of Work re: Project No. EGE P9 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Altasciences | 232 | \$0.00 | \$0.00 | \$0.00 |
| | Titusciciees | Statement of Work re: Project No. RRI-P9 | \$0.00 | \$0.00 | Ψ0.00 |
| Lonafarnib | Altasciences | 693 (EIG LNF 019) | \$0.00 | \$0.00 | \$0.00 |
| Contrarino | | | ***** | ***** | 4 |
| | | | | | |
| | | | | | |
| | Altasciences Clinical | | | | |
| | Kansas, Inc.; Altasciences | A | | | |
| Lonafarnib Lonafarnib | Clinical Kansas, P.A. | LNF-022/ Project Number EGE-P9-697 | \$0.00 | \$0.00 | \$0.00 |
| | ALTASCIENCES | | | | |
| | COMPANY CANADA | | | | |
| | INC. | Agreement For Bioanalytical Services | \$0.00 | \$0.00 | \$0.00 |
| | ALTASCIENCES | | | | |
| Lonafarnib | COMPANY CANADA | Chatago and a CW a la | ¢0.00 | ¢0.00 | \$0.00 |
| Lonatarnio | INC. ALTASCIENCES | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | COMPANY CANADA | | | | |
| Lambda | INC. | Agreement For Bioanalytical Services | \$0.00 | \$0.00 | \$0.00 |
| | ALTASCIENCES | regreement 1 of Bloanary fieur bervices | \$0.00 | \$0.00 | \$0.00 |
| Lambda | COMPANY INC. | Change Order 01 | \$0.00 | \$0.00 | \$0.00 |
| | | | * | * | * |
| | ALTASCIENCES | Statement of Work re: Sponsor Protocol EIG | | | |
| Lonafarnib | COMPANY INC. | LNF-018 / Project Number RRD-P0-895 | \$0.00 | \$0.00 | \$0.00 |
| | ALTASCIENCES- | Task Order re: Contractor Study Number | | | |
| Lonafarnib- | COMPANY INC. | EGE-W8-303 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | ALTREOS RESEARCH | | | | |
| Lonafarnib | PARTNERS, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | American Association for | | | | |
| | the Study of Liver | The Liver Meeting 2020 Exhibit Space | | | |
| Lonafarnib | Diseases | Application and Contract | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Discuses | Application and contract | \$0.00 | \$0.00 | \$0.00 |
| | American Mobile Health | | | | |
| Lonafarnib | Services, LLC | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | AMRI | Laboratory Service Fee Quotation | \$0.00 | | |
| | | Amendment Number 1 to Marketing and | , , , , , | , , , , , | , , , , , |
| Lonafarnib- | ANGES, INC. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | MARKETING & DISTRIBUTION | | | |
| Lonafarnib- | ANGES, INC. | AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | ANGES, INC. | Side Letter | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------------------|---|--|-------------------|-------------------|---------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | Archbow Consulting, LLC | Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| | ARUP LABORATORIES | | | | |
| Lonafarnib | INC. | Amendment 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | ARUP LABORATORIES | | | | |
| Lonafarnib | INC. | Amendment 2 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | August Bio Quotes covering batches AJ110001 (Diluent, Jun 2021) and | | | |
| Avexitide | August Bioservices LLC | AJ111001 (Jun 2021 225 Mg/ml). | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1119 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1126 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | August Bioservices LLC | Proposal: P-1143 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | August Bioservices LLC | Proposal: P-1147 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | August Bioservices LLC Austin Chemical | Proposal: P-1160 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Company, Inc. | Letter Agreement re: Offer Ritonavir USP (CAS # 155213-67-5) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Austin Chemical Company, Inc. | Letter Agreement re: Offer Ritonavir USP (CAS # 155213-67-5) | \$0.00 | \$0.00 | \$0.00 |
| | | (612 188218 8, 8) | \$0.00 | φσισσ | φσισσ |
| Lonafarnib | Avayle CTS, LLC; Fisher Clinical Services | Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | AVISTA PHARMA | | | | |
| Lambda | SOLUTIONS, INC., D/B/A CAMBREX | Proposal EIGR-D22-D001-A | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal re: Development and Validation of a Method for Free PEG in IL-29 PEG- | | | |
| | Avista Pharma Solutions, | Interferon Lambda-1a Drug Conjugate dated | | | |
| Lambda | Inc., d/b/a Cambrex | May 18, 2022 Proposal re: Development and Validation of | \$0.00 | \$0.00 | \$0.00 |
| | Avista Pharma Solutions, | a Method for Free PEG in IL-29 PEG- | | | |
| Lambda | Inc., d/b/a Cambrex | Interferon Lambda-1a Drug Conjugate dated September 30, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Axis Clinicals LLC | Services Agreement | \$0.00 | \$0.00 | \$675.00 |
| | | Certificate of Analysis (COA) meeting Eiger | | | |
| Avexitide Avexitide | BACHEM BACHEM | Specifications Change Order 1 to Quote | \$0.00 \$0.00 | | \$0.00 \$0.00 |
| | | Change Request re: Exendin (9-39), non- | | | |
| Avexitide | Bachem | GMP | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | BACHEM | Compilation of Validation Plan of Avexitide | | | \$0.00 |
| Avexitide | BACHEM | Quote | \$0.00 | \$0.00 | \$0.00 |

| Related | | D : :: : : : : : : : : : : : : : : : : | 4 201 G | | Lonafarnib |
|--------------------|---|--|--------------------------|--------|-------------|
| Asset Avexitide | Contract Counterparty BACHEM | Description of Contract of Lease Quote 67815 | Avexitide Cure \$0.00 | \$0.00 | Cure \$0.00 |
| Avexitide | BACHEM | Quote 67813 Quote 69621 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 70720 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | BACHEM | Quote 71209 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 72592 | \$0.00 | \$0.00 | |
| Avexitide | BACHEM | Quote 73445 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 20005572 | \$0.00 | \$0.00 | |
| Avexitide | Bachem Americas, Inc | Bachem Quote 58395 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 63842 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 64409 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65398 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65399 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65409 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 65675 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 66755 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Bachem Quote 67097 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 20005572 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 20005959 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 55597 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 57429 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 57583 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 63034 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc | Quotation 63583 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Bachem Americas, Inc BD Medical, | Quotation 66372 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Pharmaceutical Systems US | Letter re: Pricing dated April 27, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | BD Medical, Pharmaceutical Systems US | Letter re: Pricing dated May 13, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | Becton, Dickinson & | | | | |
| Lambda | Company | Letter re: PS-CPQ-636 - BD | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|--------------|--|---|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | BECTON, DICKINSON | | | | |
| Lambda | AND COMPANY | Quote re: Pharmaceutical Products | \$0.00 | \$0.00 | \$0.00 |
| | Bend Research Inc., a | | | | |
| Lonafarnib | Lonza Company | Analytical Proposal | \$0.00 | \$0.00 | \$0.00 |
| | Bend Research Inc., a | | | | |
| Lonafarnib | Lonza Company | Change Order 7 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Bend Research Inc., a Lonza Company | Change Order No. 4 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | 4000 | | ***** |
| Lonafarnib | BEND RESEARCH, INC. | Amendment No. 2 to the Master Services and Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | BEND RESEARCH, INC. | and Chinear Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 3 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 3 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Change Order 4 Dated June 22, 2017 | \$0.00 | \$0.00 | \$0.00 |
| 201141111110 | | | \$0.00 | φοιου | \$6.60 |
| Lonafarnib | REND DESEARCH INC | Change Order 4 Dated May 2, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | BEND RESEARCH, INC. | Change Order 4 Dated Way 2, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 5 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bend Research, Inc. | Change Order 6 to the Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | , | Change Order 6 to the Statement of Work | | | |
| Lonafarnib | Bend Research, Inc. | dated September 2, 2016 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Commercial Manufacturing Services and | | | |
| Lonafarnib | BEND RESEARCH, INC. | Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Memorandum of Understanding | \$0.00 | \$0.00 | \$0.00 |
| | | Mutual Nandicalogues Agreement dated | | | |
| Lonafarnib | BEND RESEARCH, INC. | Mutual Nondisclosure Agreement dated February 23, 2016 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Mutual Nondisclosure Agreement dated November 22, 2017 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|--|------------------|-------------------|---------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| Lonafarnib | DENID DECEADOR INC | Proposal dated July 9, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | DEND RESEARCH, INC. | 1 Toposai dated July 9, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | BEND RESEARCH, INC. | Proposal dated October 14, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Bend Research, Inc. | Purchase Order dated April 4, 2017 | \$0.00 | \$0.00 | \$0.00 |
| T 0 "1 | D 1D 1 I | D 1 0 1 1 1 1 1 6 2017 | #0.00 | #0.00 | #0.00 |
| Lonafarnib | Bend Research, Inc. | Purchase Order dated March 6, 2017 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | BIOANALYTICAL | Task Order Q-00122 to the Master | Ф0.00 | ФО ОО | ФО ОО |
| Lonafarnib | SYSTEMS, INC. BIOANALYTICAL | Independent Contractor Agreement Task Order 02926 to the Master | \$0.00 | \$0.00 | \$0.00 |
| т с и | | | #0.00 | #0.00 | #0.00 |
| Lonafarnib | SYSTEMS, INC. | Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | BIORASI, LLC | Change Order 1 to Statement of Work | \$0.00 | 1 | \$0.00 |
| Lambda | Biorasi, LLC | Clinical Study Agreement | \$0.00 \$0.00 | 1 | \$0.00 |
| Lambda | BIORASI, LLC | Letter of Intent | \$0.00 | 1 | \$0.00 |
| Lambda | DIODACLILC | Latter of Intent dated Sentember 20, 20201 | \$0.00 | | \$0.00 |
| Lamoda | BIORASI, LLC | Letter of Intent dated September 29, 20201 Letter of Intent Extension Agreement dated | \$0.00 | \$181,489.74 | \$0.00 |
| Lambda | BIORASI, LLC | December 18, 2020 | \$0.00 | \$101,409.74 | \$0.00 |
| Lamoua | BIORASI, LLC | Letter of Intent Extension Agreement | \$0.00 | 1 | \$0.00 |
| Lambda | BIORASI, LLC | September 29, 2021 | \$0.00 | | \$0.00 |
| Lambda | BIORASI, LLC | Statement of Work | \$0.00 | 1 | \$0.00 |
| Lambda | BIORASI, LLC | Statement of Work 3 | \$0.00 | 1 | \$0.00 |
| Lonafarnib | BIORASI, LLC | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BIORASI, LLC | Change Order 2 to Statement of Work | \$0.00 | | \$0.00 |
| Lonafarnib | BIORASI, LLC | Change Order 3 to Statement of Work | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Change Order 4 to Statement of Work | \$0.00 | | \$0.00 |
| Lonafarnib | BIORASI, LLC | Letter of Intent dated June 13, 2020 | \$0.00 | | \$0.00 |
| Lonafarnib | BIORASI, LLC | Master Services Agreement | \$0.00 | | |
| Lonafarnib | BIORASI, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal re: Project Manufacture of Exendin | | \$0.00 | \$0.00 |
| Avexitide | Bioserv Corporation | 9-39 Injection | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal re: Project Non Aseptic | 40.00 | 40100 | 44100 |
| Avexitide | Bioserv Corporation | Engineering Run | \$0.00 | \$0.00 | \$0.00 |
| | | | | | · |
| Avexitide | Bioserv Corporation | Proposal re: Project Vial Labeling | \$0.00 | \$0.00 | \$0.00 |
| | • | | | | |
| | BRISTOL-MYERS | | | | |
| Lambda | SQUIBB COMPANY | Assignment and Assumption Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | BRISTOL-MYERS | | | | |
| Lambda | SQUIBB COMPANY | Common Stock Purchase Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | BRISTOL-MYERS | | | | |
| Lambda | SQUIBB COMPANY | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Bristol-Myers Squibb | | | | |
| Lonafarnib | Company | License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | First Amendment to Research Funding | | | |
| Lonafarnib | Brown University | Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|---------------------|------------------------|---|----------------|--------------|--------------|
| Asset Lonafarnib | Contract Counterparty | Description of Contract of Lease RESEARCH FUNDING AGREEMENT | Avexitide Cure | | Cure |
| Lonafarnib | Brown University | | \$0.00 | \$0.00 | \$0.00 |
| T:1- | Danis IIaisanita | Second Amendment to Research Funding | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Brown University | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Bulgarian Drug Agency | Declaration on Funds Ownership | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | CALISYM LLC | Services Agreement | \$2,000.00 | \$0.00 | \$0.00 |
| Lambda | CALISYM LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CALISYM LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CALISYM LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CALISYM LLC | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CALISYM LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBER | | | | |
| T C 11 | PHARMACEUTICALS | N. I. I. T. GI | ¢0.00 | #0.00 | #0.00 |
| Lonafarnib | INCORPORATED | Nonbinding Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Cambridge Biomarketing | Project Brief V1 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | CAMBRIDGE | Project Brief v i | \$0.00 | \$0.00 | \$0.00 |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | CAMBRIDGE | Waster Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Master Services Agreement Amendment | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | CAMBRIDGE | Widster Services rigidement rimenament | ψ0.00 | ψ0.00 | ψ0.00 |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Name Change to MSA | \$0.00 | \$0.00 | \$0.00 |
| Zenaranie | CAMBRIDGE | Trume change to marr | \$0.00 | \$0.00 | \$0.00 |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | , | * | * |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 10 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 11 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 12 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-----------------------|------------------------------|--|-----------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| | CAMBRIDGE | | | | |
| | BIOMARKETING | | | | |
| Lonafarnib | GROUP, LLC | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | Canadian Life Science; | | | | |
| | Patheon, by Thermo | | | | |
| Lonafarnib | Fisher Scientific | PDS Lab Bill Back Request Form | \$0.00 | | \$0.00 |
| Lonafarnib | Capsugel | Quote re: 1 Color Ink | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Capsugel | Quote re: Capsules | \$0.00 | \$0.00 | \$0.00 |
| | | Third Amendment to Statement of Work No. | | | |
| Lonafarnib | CAREMARK, LLC | 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Agreement for the Establishment of Legal | | | |
| Lonafarnib | CATO Europe GmbH | Representative in European Union | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | CATO Europe GmbH | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| | CATO Europe GmbH; | | | | |
| Lonafarnib | Clinipace GmbH | I attan of Anthonication | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | Cillipace Gillon | Letter of Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Cenduit LLC | Proposal for the Provision of IRT Services | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Cenduit LLC | Troposar for the Frovision of fix i Services | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Fifth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Londidinio | Discuses, EEC (CEDQ) | Then I mineralized to Electibe I igreement | ψ0.00 | ψ0:00 | ψ0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | First Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | Discuses, ELEC (CLDQ) | This Timenament to Electica Tigreement | ψ0.00 | ψ0.00 | ψ0.00 |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Second Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | , , , | 6 | , , , , , | , , , , | , , , , |
| | | | | | |
| | Center for Outcomes | | | | |
| | Research in Liver | | | | |
| Lonafarnib | Diseases, LLC (CLDQ) | Sixth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|---|-----------------------|--------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | Center for Outcomes | | | | |
| Lonafarnib | Research in Liver Diseases, LLC (CLDQ) | Third Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Diseases, LLC (CLDQ) | Third Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Center for Outcomes | | | | |
| | Research in Liver Diseases, LLC d/b/a | | | | |
| Lonafarnib | CLDQ | Fourth Amendment to License Agreement | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002084 in 060622 Statement of | | | |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | CHARLES RIVER LABORATORIES | Amnd to 9002086 in 060622 Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Aveaude | LABORATORIES | WOIK | ψ0.00 | ψ0.00 | \$0.00 |
| A :: 1 | CHARLES RIVER | Amnd to 9002087 in 060622 Statement of | ФО ОО | #0.00 | #0.00 |
| Avexitide | LABORATORIES | Work | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amnd to 9002089 in 060622 Statement of | | | |
| Avexitide | LABORATORIES Charles River | Work | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Laboratories | Archive Storage Letter dated July 12, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | Charles River | | фо оо | #0.00 | #0.00 |
| Avexitide | Laboratories Charles River | Archive Storage Letter dated July 20, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Laboratories | Letter re: Payment Authorization | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | | | | |
| Avexitide | LABORATORIES | Statement of Work (CRL-210936) | \$0.00 | \$0.00 | \$0.00 |
| | CHADLEC DIVED | | | | |
| Avexitide | CHARLES RIVER LABORATORIES | Statement of Work (CRL-224818) | \$0.00 | \$0.00 | \$0.00 |
| | | , | | | |
| Avexitide | CHARLES RIVER LABORATORIES | Statement of Work (SFDC-OPP-272991) | \$0.00 | \$0.00 | \$0.00 |
| Aveatude | LABORATORIES | Statement of Work (Si De-Oil -2/2771) | \$0.00 | \$0.00 | \$0.00 |
| A :: 1 | CHARLES RIVER | G CW. 1 20220720 | Φ0.00 | #0.00 | #0.00 |
| Avexitide | LABORATORIES | Statement of Work 20220630 | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | 1st Amnd to Statement of Work (SFDC OPP- | | | |
| Lonafarnib | LABORATORIES | 273814) 20230131 | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | 2nd Amnd to Statement of Work (SFDC | | | |
| Lonafarnib | LABORATORIES | OPP-273814) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Amendment 1 to Statement of Work | | | |
| Lonafarnib | LABORATORIES | 20190821 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-------------|-------------------------------------|--|-------------------|-------------------|--------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | CHARLES RIVER LABORATORIES | Amendment 2 to Statement of Work 20200326 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | CHARLES RIVER LABORATORIES | Statement of Work (CRL-237365) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CHARLES RIVER LABORATORIES | Statement of Work 20190618 | \$0.00 | \$0.00 | \$0.00 |
| I C 1 | CHARLES RIVER | G | Ф0.00 | Ф0.00 | Ф0.00 |
| Lonafarnib | LABORATORIES | Statement of Work 20220610 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide- | CHARLES RIVER- LABORATORIES | 2nd Amnd to 9002088 in 060622 Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | CHARLES RIVER- LABORATORIES | Amnd to 9002088 in 060622 Statement of Work | \$0.00 | \$0.00 | \$ 0.00 |
| Aveatide | LABORATORIES | WOIK | ψ0.00 | ψ0.00 | ψ0.00 |
| Avexitide | CHARLES RIVER LABORATORIES | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | CHARLES RIVER- LABORATORIES | Statement of Work (SFDC OPP-273814) | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER LABORATORIES | | | | |
| Lonafarnib | MONTREAL ULC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | CHARLES RIVER LABORATORIES, INC. | ARCHIVE STATEMENT OF WORK CONTINUED STORAGE | \$0.00 | \$0.00 | \$0.00 |
| | , | | 4 0.00 | 4 0.00 | 4000 |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work dated December 31, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | Bridge Grant Grade, nive | | φσισσ | φσισσ | \$0.00 |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work dated December 31, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | , | | ,, | , | |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work re: Evaluation of the Stability of E.coli Working Cell Bank | \$0.00 | \$0.00 | \$0.00 |
| | CHARLES RIVER | Statement of Work re: Preparation and Characterization of an E. coli Master Cell | | | |
| Lambda | LABORATORIES, INC. | Bank | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-----------------------|--|--|-------------------|-------------------|------------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | CHARLES RIVER | Statement of Work re: Storage of Materials | | | |
| Lambda | LABORATORIES, INC. | Under Controlled Conditions and Access | \$0.00 | \$0.00 | \$0.00 |
| | , | | | | |
| | | | | | |
| | Children to II and hal | | | | |
| | Children's Hospital Corporation dba Boston | Letter re: Confirmation of Regulatory | | | |
| Lonafarnib | Children's Hospital | Compliance Responsibilities | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CHEMTREC | Quote re: Emergency Response Service | \$0.00 | | |
| | CITRINE MEDICINE, | (<u></u> | ***** | 4 | 40.00 |
| Lonafarnib | LLC | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | | Direct Study Participant Recruitment | | | |
| Lonafarnib | ClinicalConnection, Inc | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Single Patient Consent Form for | | | |
| Lonafarnib | Clinigen | Participation in a Managed Access Program | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | Cimigen | Amendment No. 1 to Statement of Work No. | \$0.00 | Ψ0.00 | \$0.00 |
| Lonafarnib- | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | | Amendment No. 2 to Statement of Work No. | | | |
| Lonafarnib- | Clinigen Healthcare | 4 | \$0.00 | \$0.00 | \$0.00 |
| T 0 11 | CULT TO 14 | Amendment No. 3 to Statement of Work No. | #0.00 | Ф0.00 | #0.00 |
| Lonafarnib- | Clinigen Healthcare | Amendment No. 4 to Statement of Work No. | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | Clinigen Healthcare | 1 | \$0.00 | \$0.00 | \$0.00 |
| | 8 | | ****** | 4 | ***** |
| | CLINIGEN- | | | | |
| Lonafarnib | HEALTHCARE LTD | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | CILL II III III | M. C. | #A AA | #0.00 | #0.00 |
| Lonafarnib | Clinigen Healthcare Ltd. | Manufacturer Statement of Delegation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Clinigen Inc. | pursuant to the Master Services Agreement | \$0.00 | \$0.00 | \$79,677.66 |
| 20114111110 | ogen mer | Change Order 3 to Work Order C-14 for | \$0.00 | φο.σσ | \$12,011.00 |
| Lambda | CLINIPACE, INC. | Eiger | \$0.00 | \$0.00 | \$0.00 |
| | | Clinical Research | | | |
| | | | | | |
| T amb da | CLINIDA CE INC | Change Order 1 to Work Order C-14 For | \$0.00 | \$0.00 | \$0.00 |
| Lambda | CLINIPACE, INC. | Eiger Clinical Research | \$0.00 | \$0.00 | \$0.00 |
| | | Chineur research | | | |
| | | Change Order 1 to Work Order C-7 ("CO1") | | | |
| Lambda | CLINIPACE, INC. | For Eiger | \$0.00 | | \$0.00 |
| Lambda | Clinipace, Inc. | Work Order C-11 | \$0.00 | | |
| Lambda | Clinipace, Inc. | Work Order C-12 | \$0.00 | | |
| Lambda | Clinipace, Inc. | Work Order C-14 | \$0.00 | | |
| Lambda Lambda | Clinipace, Inc. Clinipace, Inc. | Work Order C-7 Work Order C-9 | \$0.00 \$0.00 | | |
| Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-1 | \$0.00 | | |
| Lonafarnib | Clinipace, Inc. | Change Order #1 to Work Order C-4 | \$0.00 | | |
| Lonafarnib | CLINIPACE, INC. | CO2 to Work Order C-1 | \$0.00 | | |

| Related | | | | | Lonafarnib |
|---------------------|----------------------------------|--|------------------|--------|--------------|
| Asset Lonafarnib | Clinings Inc | Description of Contract of Lease | Avexitide Cure | | Cure |
| Lonafarnib | Clinipace, Inc. Clinipace, Inc. | Work Order C-10 Work Order C-11 | \$0.00 \$0.00 | | * |
| Lonafarnib | Clinipace, Inc. | Work Order C-9 | \$0.00 | \$0.00 | \$0.00 |
| 20114141110 | eminpute, mer | Work Grade C y | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Clinipace, Inc. dba Caidya | CO2 to Work Order C-7 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Colleen M. Craig Consulting, LLC | Services Agreement | \$1,650.00 | \$0.00 | \$0.00 |
| Tivexitide | Consuming, EDC | Services regreement | ψ1,030.00 | φ0.00 | <u>φυ.υυ</u> |
| Lonafarnib | Corden Pharma Colorado | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 dated May 19, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 dated Nov 5, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order CO4 to CO2 of SOW #2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORPENIANTA | | | | |
| Lonafarnib | CORDEN PHARMA COLORADO INC. | Change Order 2 to Change Order 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | COLORADO INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 1 | CORDEN PHARMA | CI O I A COM I A | Φ0.00 | Ф0.00 | ФО ОО |
| Lonafarnib | COLORADO INC. | Change Order 3 to Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | COLORADO IIVO. | Statement of Work 1 | ψ0.00 | ψ0.00 | φ0.00 |
| | | | | | |
| T C " | CORDEN PHARMA | G | *** | 40.00 | 40.00 |
| Lonafarnib | COLORADO INC. | Statement of Work 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | CORDEN PHARMA | | | | |
| Lonafarnib | COLORADO INC. | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |

| D.I. (I | | | | | T 6 11 |
|------------------|--|--|----------------|-------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| | , | | | | |
| | Corden Pharma Colorado; Corden Pharma | | | | |
| Lonafarnib | International GmbH | Change Order #5 | \$0.00 | \$0.00 | \$0.00 |
| | Corden Pharma Colorado; Corden Pharma | | | | |
| Lonafarnib | International GmbH | Change Order 1 to the Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado; Corden Pharma International GmbH | Change Order 5 to the Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado; Corden Pharma International GmbH | Coden Pharma Quote | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Corden Pharma International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Corden Pharma International GmbH | Quote 20210329 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | CordenPharma; Corden Pharma International GmbH | Coden Pharma Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Amendment 1 to Service Agreement - H | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - A | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - B | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - J | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - N | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | COREALIS PHARMA | Service Agreement - O | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Curia New Jersey, LLC | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 3/26/2022 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 5/31/2022 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation dated 7/27/2022 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cura | Lonafarnib Cure |
|------------------|---|---|----------------|-------------|--------------------|
| Asset | Contract Counterparty | Laboratory Service Fee Quotation dated | Avexitide Cure | Lambua Cure | Cure |
| Lambda | Curia New Jersey, LLC | 9/30/2022 | \$0.00 | \$0.00 | \$0.00 |
| | , | Laboratory Service Fee Quotation dated | | | |
| Lambda | Curia New Jersey, LLC | June 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal for Analysis of Active Metabolite | | | |
| Lonafarnib | Cyprotex US, LLC | of Lonafarnib (LNF): HM21 | \$0.00 | \$0.00 | \$0.00 |
| | | Day and the state of Astina Matabalita | | | |
| Lonafarnib | Crommatary LIC LLC | Proposal for Analysis of Active Metabolites | \$0.00 | \$0.00 | \$0.00 |
| Lonaiarnio | Cyprotex US, LLC | of Lonafarnib (LNF): MH17 and HM21 | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal for CytoChrome P450 Time | | | |
| Lonafarnib | Cyprotex US, LLC | Dependent Inhibition of Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 |
| 20114111110 | ejpreten es, EEe | Proposal for MATE Transporter Inhibition | φ0.00 | φοισσ | Ψ0.00 |
| Lonafarnib | Cyprotex US, LLC | Testing for Lonafarnib (LNF) | \$0.00 | \$0.00 | \$0.00 |
| | | Proposal for OCT1 and BSEP Inhibition and | | | |
| | | OCT1 Substrate Testing for Lonafarnib | | | |
| Lonafarnib | Cyprotex US, LLC | (LNF) | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| A 1 | D. S., PHD [Name on | | #12 750 00 | Φ0.00 | Ф0.00 |
| Avexitide | File] | Consulting Agreement | \$12,750.00 | \$0.00 | \$0.00 |
| | D. S., PHD [Name on | | | | |
| Lonafarnib | File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | i nej | Consulting rigiteement | ψ0.00 | ψ0.00 | Ψ0.00 |
| | DEFENSE HEALTH | Department of Defense (DoD) Retail | | | |
| Lonafarnib | AGENCY | Refund Pricing Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Department of Health and | | | | |
| Lonafarnib | Human Services | National Drug Rebate Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DED A DED CENTER OF | | | | |
| T C '1. | DEPARTMENT OF | 1 A 1 DDA 2/E70722 A 0010 | ¢0.00 | ¢0.00 | ¢0.00 |
| Lonafarnib | VETERANS AFFAIRS | 1st Amnd to BPA 36E79722A0010 | \$0.00 | \$0.00 | \$0.00 |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | Amendment | \$0.00 | \$0.00 | \$0.00 |
| 20114111110 | V E I E I I I I I I I I I I I I I I I I | | φ0.00 | φοισσ | Ψ0.00 |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | Amendment 2 to 36F79722D0048 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| | DEDARTMENT OF | Amondment of Calinitation (N. f. 116 and a | | | |
| Lanafamil | DEPARTMENT OF VETERANS AFFAIRS | Amendment of Solicitation/Modification of | የ ስ ስስ | \$0.00 | የ ስ ስስ |
| Lonafarnib | VEIERANS AFFAIKS | Contract | \$0.00 | \$0.00 | \$0.00 |
| | DEPARTMENT OF | Blanket Purchase Agreement dated Jan 1, | | | |
| Lonafarnib | VETERANS AFFAIRS | 2022 | \$0.00 | \$0.00 | \$0.00 |
| | . ETERTION THE THREE | | ψ0.00 | ψ0.00 | Ψ0.00 |
| | DEPARTMENT OF | Blanket Purchase Agreement dated Oct 10, | | | |
| Lonafarnib | VETERANS AFFAIRS | 2021 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|-------------------------|---------------------------|---|------------------|---------------|--------------------|
| 113300 | Contract Counterparty | Description of Contract of Lease | Tivexitide Cure | Lambua cure | cure |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | BPA Cancellation | \$0.00 | \$0.00 | \$0.00 |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | Commercial Items | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Department of Veterans | Commercial Items | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Affairs | D0048 Red A | \$0.00 | \$0.00 | \$0.00 |
| | Department of Veterans | | | | |
| Lonafarnib | Affairs | Facility Code | \$0.00 | \$0.00 | \$0.00 |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | FSS NFAMP Smoothing Letter | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | VETERATIO ATTAINS | I B R8, Amendment 0004 | \$0.00 | \$0.00 | ψ0.00 |
| | | , | | | |
| | DEPARTMENT OF | Amendment of Solicitation/Modification of | | | |
| Lonafarnib | VETERANS AFFAIRS | Contract | \$0.00 | \$0.00 | \$0.00 |
| | DEPARTMENT OF | Master Agreements and Pharmaceutical | | | |
| Lonafarnib | VETERANS AFFAIRS | Pricing Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | VETERATIO THITTING | Themg regreement | ψ0.00 | ψ0.00 | ψ0.00 |
| | DEPARTMENT OF | | | | |
| Lonafarnib | VETERANS AFFAIRS | Small Business Subcontracting Plan | \$0.00 | \$0.00 | \$0.00 |
| | DILISYM SERVICES, | Master Services Agreement dated June 14, | фо оо | фо оо | фо оо |
| Lambda | INC. DILISYM SERVICES, | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | INC. | Master Services Agreement May 22, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | n (c. | France Services rigidement may 22, 2010 | \$0.00 | ψ0.00 | ψ0.00 |
| | Dom zdravlja Medigroup | | | | |
| Lonafarnib | dr Ristic Pariske | Donation Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Dr. T. L. M. [Name on | | Φ2 001 00 | фо оо | фо оо |
| Avexitide | File] | Proposal to Purchase Assets | \$2,981.00 | \$0.00 | \$0.00 |
| Lambda | DSG, Inc. | Ouotation re: Ongoing Project Management | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | ELEMENT | Master Service Agreement | \$0.00 | | |
| | | _ | | | |
| | | | | | |
| T C 1 | Element Materials | | Φο οο | #0.00 | #0.00 |
| Lonafarnib | Technology Canada Inc. | Element Quote | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Element Materials | | | | |
| Lonafarnib | Technology Canada Inc. | Element Quote dated April 7, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| I | Element Materials Testing | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib Avexitide | Canada Inc. Eurofins | SP-2396 Change Order dated February 14, 2022 | \$0.00 \$0.00 | | |
| Avexitide | Eurofins | Change Order November 3, 2021 | \$0.00 | | |
| Lambda | Eurofins | Quote dated December 31, 2017 | \$0.00 | | |
| Lambda | Eurofins | Quote dated December 4, 2020 | \$0.00 | | |

| Related | | | | | Lonafarnib |
|---------------------|--|--|------------------|-------------|------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Avexitide | LABORATORIES, INC. | Services Agreement | | \$0.00 | \$0.00 |
| | | | | | |
| | ELIDOEDIG ADVANTAD | | \$25,600.00 | | |
| Avexitide | EUROFINS ADVANTAR LABORATORIES, INC. | AGREEMENT APPROVAL 5 | | \$0.00 | \$0.00 |
| Avexitiue | LABORATORIES, INC. | AGREEMENT AFFROVAL 3 | | \$0.00 | \$0.00 |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Avexitide | LABORATORIES, INC. | AGREEMENT APPROVAL 7 | | \$0.00 | \$0.00 |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | | Eurofins Contact Dated June 3, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Zonaranio | | Date in the contract of the co | φοισσ | Ψ0.00 | \$0.00 |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 4, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | EUROFINS ADVANTAR | | | | |
| Lonafarnib | LABORATORIES, INC. | Eurofins Contact Dated June 5, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | , | -, - | , , , , , | * | * |
| | Eurofins Biopharma | Quote re: Free PEG by HPLC-CAD Method | | | |
| Lambda | Product Testing | Transfer | \$0.00 | \$0.00 | \$0.00 |
| | Eurofins BioPharma | Quote re: Lambda-1 (Python) FBS | | | |
| Lambda | Product Testing | Qualification | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 1/16/2023 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 10/30/2020 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 2/11/2021 | \$0.00 | | \$0.00 |
| Avexitide | | Quote dated 2/11/2021 | \$0.00 | | |
| Avexitide | Eurofins BTP | Quote dated 5/5/2021 | \$0.00 | | \$0.00 |
| Avexitide | Eurofins BTP | Quote dated 7/6/2021 | \$0.00 | | \$0.00 |
| Avexitide Lambda | Eurofins BTP Eurofins BTP | Quote dated 8/4/2023 Quote dated June 7, 2022 | \$0.00 \$0.00 | | \$0.00 \$0.00 |
| Lambda | Eurofins BTP | Quote dated October 1, 2021 | \$0.00 | | \$0.00 |
| Lumouu | Editing B11 | Quote union detable 1, 2021 | ψ0.00 | ψ0.00 | \$0.00 |
| | Eurofins Lancaster | | | | |
| | Laboratories (Lancaster | | | | |
| Avexitide | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | Even fina I amanata | | | | |
| | Eurofins Lancaster Laboratories (Lancaster | | 1 | | |
| Avexitide | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| . I , CAILIGE | | | ψ0.00 | Ψ0.00 | \$0.00 |
| | Eurofins Lancaster | | | | |
| | Laboratories (Lancaster | | | | |
| Lambda | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|----------------------------------|---|----------------|--------------|---------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | _ | | | | |
| | Eurofins Lancaster | | | | |
| | Laboratories (Lancaster | | | | |
| Lambda | PA) | Quote | \$0.00 | \$0.00 | \$0.00 |
| | EVERSANA LIFE | | | | |
| T C 1 | SCIENCE SERVICES, | | #0.00 | #A AA | фо о о |
| Lonafarnib | LLC EXCELLIS EUROPE | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LTD. | Work Order No. One | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | LID. | Work Order No. One | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | F. M., D. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$2,612.50 |
| Lonaramo | 1. W., D. [Tunic on The] | Services regreement | ψ0.00 | ψ0.00 | Ψ2,012.30 |
| | FISHAWACK | | | | |
| Lonafarnib | FACILITATE LIMITED | Eiger Advisory Board Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | FISHER BIOSERVICES, | | | | |
| Lambda | INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote 214873 Order 3 Version 4 20190513 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote 214873 Order 7 Version 2 20210318 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote 214873 Order 8 Version 3 20220225 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | Fisher Clinical Services | 0 4 215966 0 1 5 37 1 1 20210221 | #0.00 | #A AA | фо о о |
| Lonafarnib | GmbH Fisher Clinical Services | Quote 215866 Order 5 Version 1 20210331 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | Quote FCS 214873 20180809 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Fisher Clinical Services | Quote FCS 214873 Order 3 Version 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | 20191024 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | Fisher Clinical Services | Ouote FCS 214873 Order 4 Version 1 | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | GmbH | 20190703 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | , | , | * |
| Lonafarnib | GmbH | Quote FCS 214873 Order 5 V1 20200218 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lonafarnib | GmbH | Quote FCS 214873 Order 5 V2 20200305 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 214873 Order 6 Version 2 | | | |
| Lonafarnib | GmbH | 20201229 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 2 Version 1 | | | |
| Lonafarnib | GmbH | 20190827 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 2 Version 3 | | | |
| Lonafarnib | GmbH | 20190918 | \$0.00 | \$0.00 | \$0.00 |
| T C 1 | Fisher Clinical Services | Quote FCS 215866 Order 2 Version 4 | #0.00 | #0.00 | фо о о |
| Lonafarnib | GmbH | 20191003 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Fisher Clinical Services GmbH | Quote FCS 215866 Order 3 Version 1 20191003 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Fisher Clinical Services | Quote FCS 215866 Order 4 Version 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | 20210121 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Fisher Clinical Services | Quote FCS 215866 Order 4 Version 2 | \$0.00 | φυ.00 | φυ.υυ |
| Lonafarnib | GmbH | 20210504 | \$0.00 | \$0.00 | \$0.00 |
| | Fisher Clinical Services | Quote FCS 215866 Order 5 Version 2 | \$5.50 | \$5.00 | ψυ.συ |
| Lonafarnib | GmbH | 20210705 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|-------------------------------|---------------------------------------|-----------------------|-------------|--------------------|
| | E. 1 Cl. : 1C | | | | |
| Avavitida | Fisher Clinical Services | Sawings Agreement | | \$0.00 | 00.00 |
| Avexitide | Inc. Fisher Clinical Services | Services Agreement | <u>\$1,405.02</u> | \$0.00 | \$0.00 |
| Avexitide | Inc. | Quote PSG-A-1085163.v2 20231106 | \$1,808.02 | \$0.00 | \$0.00 |
| Tivexitide | Fisher Clinical Services | Quote 150 11 1005105.72 20251100 | 1 | ψ0.00 | Ψ0.00 |
| Avexitide | Inc. | Quote-PSG-A-1081156.v1 20230807 | | \$0.00 | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | CO 1 to PSG-A-1073971 (PSG-A-1076893) | | | |
| Lambda | Inc. | 20230524 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Fisher Clinical Quote 20160517 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Fisher Clinical Services Quote PSG-A- | фо. о о | | # 0.00 |
| Lambda | Inc. Fisher Clinical Services | 1043137.v1 20210812 | \$0.00 | | \$0.00 |
| Lambda | | Quote 20160927 | \$0.00 | | \$0.00 |
| Lamoua | Inc. Fisher Clinical Services | Quote 20100927 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote FCS 55799 20170221 | \$0.00 | | \$0.00 |
| Lamoda | Fisher Clinical Services | Quote FCS 55799 Change Order 1 | Ψ0.00 | | \$0.00 |
| Lambda | Inc. | 20161116 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | 40100 | | 44144 |
| Lambda | Inc. | Quote FCS 55799 Order 7 20170629 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 55799 Order 8 20180918 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 58040 20161206 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote FCS 62278 20180309 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | фо. о о | | # 0.00 |
| Lambda | Inc. Fisher Clinical Services | 20190720 | \$0.00 | | \$0.00 |
| Lambda | _ | Quote PSG-A-1007253 V2 20190508 | \$0.00 | | \$0.00 |
| Lambda | lnc. | Quote PSG-A-100/233 V2 20190308 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1009306.V1 20190619 | \$0.00 | | \$0.00 |
| Zum uu | 11101 | Quete 12 d 11 100,0001 (1 201,001) | ψοίου | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1037571.v1 20210414 | \$0.00 | | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1037572.v1_20210420 | \$0.00 | | \$0.00 |
| | | | | | |
| | Fisher Clinical Services | Dag 4 102555 1 20210120 | * • • • | | * |
| Lambda | Inc. | Quote PSG-A-1037587.v1 20210420 | \$0.00 | \$36,483.85 | \$0.00 |
| | Elabora Cilinia 1 C | | | | |
| I amb 1. | Fisher Clinical Services | Onata DCC A 10201021 20210422 | # 0.00 | | ф л 00 |
| Lambda | Inc. Fisher Clinical Services | Quote PSG-A-1038183.v1 20210422 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1040820.V3 20210713 | \$0.00 | | \$0.00 |
| Lamoua | me. | Quote F3O-A-1040020. V3 20210/13 | \$0.00 | | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|-------------------------------|--|----------------|-------------|--------------------|
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1041275.v4 20210819 | \$0.00 | | \$0.00 |
| Lamoda | Fisher Clinical Services | Quote 15G-A-10+12/3.v+ 2021001) | φ0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1044658.v1 20210902 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1045926.v1 20210927 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1053812.v2 20220323 | \$0.00 | | \$0.00 |
| | Fight of Clinian I Commission | | | | |
| Lambda | Fisher Clinical Services Inc. | Ouote PSG-A-1056697.v1 20221205 | \$0.00 | | \$0.00 |
| Lamoda | Fisher Clinical Services | Quote 13G-A-1030077.V1 20221203 | \$0.00 | | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1060127.v1 20221205 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | | \$0.00 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1073971 20230429 | \$0.00 | | \$0.00 |
| Lamoua | Fisher Clinical Services | Quote F3G-A-10/39/1 20230429 | \$0.00 | ł | \$0.00 |
| Lambda | Inc. | Quote PSG-A-1086699.v1 20231128 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote re: Protocol No. EIG-LMD-001 | \$0.00 | | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote-PSG-A-1037570.v2 20210423 | \$0.00 | | \$0.00 |
| | | | | | |
| T 1 1 | Fisher Clinical Services | O + DCC + 1045030 1 20210037 | ФО ОО | | ФО ОО |
| Lambda | Inc. | Quote-PSG-A-1045938.v1 20210927 | \$0.00 | - | \$0.00 |
| | Fisher Clinical Services | | | | |
| Lambda | Inc. | Quote-PSG-A-1069905.v1 20230204 | \$0.00 | | \$0.00 |
| | | | | | |
| I | Fisher Clinical Services | Letter of Intent 20220517 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Letter of intent 20220317 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | Inc. | Quote FCS 63502 20180613 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote FCS 64018 20180705 | \$0.00 | \$0.00 | |
| T C 11 | Fisher Clinical Services | 0 - FGG (4045 20100(20 | #0.00 | ФО ОО | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote FCS 64045 20180629 Quote FCS 64045 Order 3 Version 2 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | Inc. | 20180803 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote FCS 64045 Order 4 Version 2 | ψ0.00 | ψ0.00 | 1 |
| Lonafarnib | Inc. | 20190110 | \$0.00 | \$0.00 |] |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote FCS 64624 20180821 | \$0.00 | \$0.00 | |
| Lonafarnib | Fisher Clinical Services Inc. | Quote FCS 64624 Order 1 Version 2 20190128 | \$0.00 | \$0.00 | |
| Lonarafillo | Fisher Clinical Services | 20170120 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote FCS 65165 20180919 | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|------------|-------------------------------|-----------------------------------|----------------|--------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| T C 1 | Fisher Clinical Services | O | #0.00 | Ф0.00 | |
| Lonafarnib | Inc. | Quote FCS 65165 v2 20181002 | \$0.00 | \$0.00 | |
| I | Fisher Clinical Services | Out FCS (5165 v2 20191022 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote FCS 65165 v3 20181023 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote FCS 66521 20190128 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote FCS 68128 Order 1 Version 1 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | 20190720 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | * | * | |
| Lonafarnib | Inc. | Quote PSG-A-1004098.V1 20190219 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1004100.V1 20190219 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1004911.V1 20190311 | \$0.00 | \$0.00 | |
| T C '1 | Fisher Clinical Services | DOC 4 1004012 VI 20100211 | #0.00 | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1004912.V1 20190311 | \$0.00 | \$0.00 | |
| Lonafarnib | _ | Ouote PSG-A-1006224 20190403 | \$0.00 | \$0.00 | |
| Lonaranno | Inc. Fisher Clinical Services | Quote FSG-A-1000224 20190403 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1006225 20190404 | \$0.00 | \$0.00 | |
| Lonaramo | Fisher Clinical Services | Quete 136 11 1000223 20170 10 1 | \$0.00 | ψ0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1006225.V2 20190610 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | · | | |
| Lonafarnib | Inc. | Quote PSG-A-1009322.V1 20190627 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1011449.V4 20190910 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1011991 20190827 | \$0.00 | \$0.00 | |
| T C 1 | Fisher Clinical Services | O + PGC + 1011001 V/2 20100000 | #0.00 | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1011991.V2 20190909 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Ouote PSG-A-1019194.v1 20200311 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote 15G-A-1017174.V1 20200311 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1020030.v2 20200406 | \$0.00 | \$0.00 | |
| | | C | 7,111 | 40.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1028287.v3 20201124 | \$0.00 | \$0.00 | \$11,961.26 |
| | Fisher Clinical Services | | | | \$11,901.20 |
| Lonafarnib | Inc. | Quote PSG-A-1034823.v2 20210303 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1035048.v2 20210308 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | _ | Oueta BSG A 1025056 v1 20210224 | \$0.00 | \$0.00 | |
| Lonaranino | Inc. | Quote PSG-A-1035056.v1 20210224 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1035575.v1 20210308 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1036207.v2 20210722 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1038320.v1 20210421 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cura | Lonafarnib Cure |
|------------------|---------------------------------------|--------------------------------------|----------------|--------------|--------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambua Cure | Cure |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1042064.v1 20210709 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1043391.v1 20210806 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Ouote PSG-A-1045137.v1 20210913 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote 13G-A-1043137.V1 20210913 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1048146.v2 20220106 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1051190.v1 20220114 | \$0.00 | \$0.00 | |
| | | | | | |
| T C '1 | Fisher Clinical Services | O + POC + 1051277 2 2222227 | 40.00 | #0.00 | |
| Lonafarnib | Inc. Fisher Clinical Services | Quote PSG-A-1051277.v3 20220225 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Ouote PSG-A-1051408.v1 20220118 | \$0.00 | \$0.00 | |
| Lonaranno | mc. | Quote 13G-A-1031408.V1 20220118 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1055223.v1 20220330 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1072645 v1 20230420 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote PSG-A-1075062.v1 20230414 | \$0.00 | \$0.00 | |
| T | Fisher Clinical Services | Overta DSC A 1077051 v.1 20220609 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote PSG-A-1077951.v1 20230608 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Ouote PSG-A-1080106.v2 20230814 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1024294.v1 20200707 | \$0.00 | \$0.00 | |
| | | | | | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1027328.v1 20200915 | \$0.00 | \$0.00 | |
| Lonafarnib | Fisher Clinical Services Inc. | Quote-PSG-A-1037585.v1 20210412 | \$0.00 | \$0.00 | |
| Lonaranno | Fisher Clinical Services | Quote-13G-A-103/363.V1 20210412 | \$0.00 | \$0.00 | |
| Lonafarnib | Inc. | Quote-PSG-A-1058344.v1 20220710 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1058998.v1 20230120 | \$0.00 | \$0.00 | |
| | Fisher Clinical Services | | | | |
| Lonafarnib | Inc. | Quote-PSG-A-1072091.v2 20230302 | \$0.00 | \$0.00 | |
| | Eighan Climical Carria | | | | |
| Lonafarnib | Fisher Clinical Services U.K. Limited | Quote PSG-A-1007765.V1 20190514 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Frontage Laboratories, | Quote 1 50-A-100 / /05. V 1 20170514 | \$0.00 | φυ.00 | φυ.υυ |
| Avexitide | Inc. | Change Order #1 | \$0.00 | \$0.00 | \$0.00 |
| | Frontage Laboratories, | | 73.00 | Ţ3.00 | Ţ3.00 |
| Avexitide | Inc. | Proposal 1005-30837-112.1 20210308 | \$0.00 | \$0.00 | \$0.00 |
| | Frontage Laboratories, | | | | |
| Avexitide | Inc. | Proposal 1005-31914-111.0 2021323 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|-----------------------|--|--|-------------------|-------------------|--------------------|
| Avexitide | Frontage Laboratories, Inc. | Proposal 1005-32862-122.0 20210520 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Frontage Laboratories, Inc. | Proposal 1005-40909-122.0 20230802 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Frontage Laboratories, Inc. | Proposal 1005.38448.400 20220830 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Fujifilm Diosynth | Stability Studies Termination, Accountability and Reconciliation Memo | \$0.00 | \$0.00 | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | 2020 ("SOW9") and the Master Bioprocessing Services Agreement | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated January 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated July 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated October 15, 2019 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated October 5, 2016 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated September 22, 2016 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement dated September 22, 2016 | \$0.00 | | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 2 | \$0.00 | | \$0.00 |

| Related | | | | | Lonafarnib | |
|---------|--|---|----------------|-------------|------------|--------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 3 | \$0.00 | | 0 | \$0.00 |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 4 | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 5 to Statement of Work 2 | \$0.00 | <u>00</u> | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Letter Agreement dated April 22, 2021 | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Letter Agreement dated June 04, 2021 | \$0.00 | \$25,000.00 | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Master Services Agreement | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Scope of Work #2 (Manufacturing) ("SOW2") and the Master Bioprocessing Services Agreement | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Statement of Work 11 | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Statement of Work 14 | \$0.00 | | \$0.00 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Statement of Work 2 | \$0.00 | | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------------------|---------------------------------------|----------------|-------------|--------------------|
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 2 to Change Order 1 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| Lambda | BIOCTECHNOLOGIES USA, INC. | Statement of Work 2 to Change Order 3 | \$0.00 | | \$0.00 |
| Lumouu | our, nve. | Statement of Work 2 to Change Order 5 | ψ0.00 | | ψ0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| | BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 3 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 5 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| Lambda | BIOCTECHNOLOGIES USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| Lamoua | OSA, IIVC. | Statement of Work o | \$0.00 | | ψ0.00 |
| | FUJIFILM DIOSYNTH | | | | |
| | BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 6 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES | | | | |
| Lambda | USA, INC. | Statement of Work 8 | \$0.00 | | \$0.00 |
| | | | | | |
| | FUJIFILM DIOSYNTH | | | | |
| T1. 1 - | BIOCTECHNOLOGIES | Statement of West-O | £0.00 | | ¢0.00 |
| Lambda | USA, INC. | Statement of Work 9 | \$0.00 | | \$0.00 |
| | FUJIFILM Diosynth | | | | |
| Avexitide | Biotechnologies (FDB) | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | FUJIFILM Diosynth | | | | |
| Lambda | Biotechnologies (FDB) | Letter re: Lamba Program on Hold | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure | |
|------------------|---|---|-----------------|---------------|--------------------|--|
| 113500 | | Description of Contract of Bease | Tivexitide Sure | Eumoui cure | Curc | |
| | FUJIFILM Diosynth Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Change Order 2 to Python SOW 17 | \$0.00 | \$0.00 | \$0.00 | |
| | | | | | | |
| | FUJIFILM Diosynth Biotechnologies U.S.A., | Change Order 5 re: MCB and WCB | | | | |
| Lambda | Inc. | Bioassay Characterization | \$0.00 | \$0.00 | \$0.00 | |
| | ETHEN V.D. 4 | | | | | |
| | FUJIFILM Diosynth Biotechnologies U.S.A., | Change Order 6 re: Establishment of Degraded SEC and Degraded Issi-Asp, CEX | | | | |
| Lambda | Inc. | & RP Purity Assay Controls | \$0.00 | \$0.00 | \$0.00 | |
| | ELHEH M.D.; | | | | | |
| | FUJIFILM Diosynth Biotechnologies U.S.A., | Change Order 6 re: Positional Isomer | | | | |
| Lambda | Inc. | Feasibility | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Change Order 9 re:Update | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Purchase Order No. 5134 | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Scope of Work | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Scope of Work #13 | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | | | | | |
| Lambda | Inc. | Scope of Work #17 | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| | Biotechnologies U.S.A., | S 0337 1 1/2 | #0.00 | #0.00 | фо оо | |
| Lambda | Inc. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| т 11 | Biotechnologies U.S.A., | C CW 1 1/2 | #A AA | #A AA | ФО ОО | |
| Lambda | Inc. | Scope of Work #2 | \$0.00 | \$0.00 | \$0.00 | |
| | FUJIFILM Diosynth | | | | | |
| Lambde | Biotechnologies U.S.A., | Scope of Work 22 | ቀ ለ ለላ | ቀ ለ ለላ | ቀ ለ ለላ | |
| Lambda | Inc. Gemini Staffing | Scope of Work 23 Exhibit A to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 | |
| Lonafarnib | Consultants, Inc. | Contractors | \$0.00 | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|-------------|--|--|-------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib- | General Synco Inc. | Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lambda | GENSCRIPT USA INC | Quote U532XIC280 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | GENSCRIPT USA INC | Quote U7278IC280 | \$0.00 | | |
| Lambda | Gilero, LLC | Proposal re: Covid PFS EUA | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Gilero, LLC | Proposal re: Lambda Interferon DHF Gap Analysis | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Gilero, LLC | Proposal re: Lambda Interferon DV&V Execution | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GKV Spitzenverband | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Global Project Partner | <u>Change Order</u> | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GLSYNTHESIS, INC. | Certificate of Analysis | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GND GEIGER NITZ DAUNDERER RECHTSANWALTE PARTG MBB | Legal Advisory Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | - garage | 40.00 | 44144 | 7,111 |
| Lambda | GOETHE-UNIVERSITY FRANKFURT | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | H. B. [Name on File] | Services Agreement | \$0.00 | \$0.00 | \$962.50 |
| Lambda | Halloran Consulting Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Halloran Consulting | SOW #5 – Inspection Readiness Assessment | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Group, Inc. | SOW #3 – Inspection Readiness Assessment | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #1 | \$0.00 | \$0.00 | \$0.00 |
| | Halloran Consulting | | | | |
| Lonafarnib | Group, Inc. | Amendment #1 to Statement of Work #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #1 to Statement of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Amendment #2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | EMA Site Inspection Preparation and Support – (SOW #4) | \$0.00 | \$0.00 | \$0.00 |
| Lonaraniio | Halloran Consulting | Jouppoit (UO II π7) | \$0.00 | \$0.00 | 90.00 |
| Lonafarnib | Group, Inc. | Site Inspection Readiness – (SOW #3) | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|-----------------------|---|--|-------------------|-------------------|--------------------|
| | | · | | | |
| Lonafarnib | Halloran Consulting Group, Inc. | SOW #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | | 56 11 11 | \$0.00 | \$0.00 | ψ0.00 |
| Lonafarnib | Halloran Consulting Group, Inc. | Statement of Work #2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | HD Biosciences | Study Quotation | \$0.00 | | |
| T 6 1 | 11D D; ; 1 | Amendment #1 to HDB-EigerBio-01- | фо оо | ФО ОО | #0.00 |
| Lonafarnib | HD Biosciences, Inc. | 05282019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | HD Biosciences, Inc. | Study Quotation | \$0.00 | \$0.00 | \$0.00 |
| | HEALTH CORPORATION OF | | | | |
| Lonafarnib | GALILEE MEDICAL CENTER | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Healthcare at Home | | | | |
| Lonafarnib- | (Netherlands) B.V. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| | HEALTHCARE AT HOME | | | | |
| Lonafarnib- | (NETHERLANDS) B.V. | Distribution Agreement | \$0.00 | \$0.00 | \$0.00 |
| | HEALTHCARE AT HOME | | | | |
| Lonafarnib Lonafarnib | (NETHERLANDS) B.V. HH Consultancy | Letter of Intent Quote 1203202a: Statement of Work | \$0.00 \$0.00 | \$0.00 \$0.00 | |
| Lonaramio | TH Consultancy | Quote 1203202a: Statement of Work | \$0.00 | \$0.00 | \$2,000.00 |
| T 0 " | ICON CLINICAL | | ф. 0.0 | # 0.00 | #0.00 |
| Lonafarnib | RESEARCH LIMITED | Start up Statement of Work No. 2 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Icon Clinical Research Limited | Consulting Services Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | ICON CLINICAL | | | | |
| Lonafarnib | RESEARCH LIMITED | Start-up Statement of Work No. 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Icon Clinical Research Limited | Statement of Work EIG22720 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | Icon Clinical Research Limited | Statement of Work EIG22721 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IDEA REGULATORY | Letter of Intent | \$0.00 | | |
| | | | | | |
| Lambda | Idem Translations, Inc. Inova Health Care | Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Services | Statement of Work 20181101 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|-------------------------------|--|-----------------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Inova Health Care | | | | |
| Lonafarnib | Services | Statement of Work 20220930 | \$0.00 | \$0.00 | \$0.00 |
| | INTEGRATED | | | | |
| | MEDICAL | Statement of work #1 to Eiger-IMD Master | | | |
| Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |
| | | | 400,000 | ***** | ****** |
| | INTEGRATED | | | | |
| | MEDICAL | Statement of work #3 to Eiger-IMD Master | | | |
| Avexitide | DEVELOPMENT LLC | Services Agreement effective July 12, 2022 | \$83,081.41 | \$0.00 | \$0.00 |
| | INTERNATIONAL | | | | |
| | TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| A:4: d - | LABORATORIES | Statement of Worls 12927 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | CANADA, INC. INTERNATIONAL | Statement of Work 42827 | \$0.00 | \$0.00 | \$0.00 |
| | TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| | LABORATORIES | | | | |
| Avexitide | CANADA, INC. | Statement of Work 42828 | \$0.00 | \$0.00 | \$0.00 |
| | INTERNATIONAL | | | | |
| | TOXICOLOGY | | | | |
| | RESEARCH | | | | |
| | LABORATORIES | | | | |
| Avexitide | CANADA, INC. | Statement of Work 74705 | \$0.00 | \$0.00 | \$0.00 |
| T1. 1. | Intrado Digital Media | Markey Assessment | ¢0.00 | ¢0.00 | £0.00 |
| Lambda | LLC | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 11 | | \$0.00 | \$0.00 |
| | | 2222 1 12 | | | |
| Avexitide | INTRINSIK CORP | Statement of Work 13 | _ | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 15 | \$2,595.00 | \$0.00 | \$0.00 |
| Avexiliae | INTRINSIK CORI | Statement of Work 13 | - | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work 2 |] | \$0.00 | \$0.00 |
| Avexitide | INTRINSIK CORP | Statement of Work Avx | | \$0.00 | \$0.00 |
| T and da | INITRINICIE CORR | Statement of Wests 10 | ¢0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 10 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 3 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 5 | \$0.00 | \$0.00 | |
| Lamoda | INTRINSIK CORI | Statement of Work 3 | \$0.00 | \$0.00 | |
| Lambda | INTRINSIK CORP | Statement of Work 6 | \$0.00 | \$0.00 | \$1,935.00 |
| Lambda | INTRINSIK CORP | Statement of Work 9 | \$0.00 | \$0.00 | |
| Lamoua | II TRIINDIK COKE | Satement of WOIK 7 | \$0.00 | \$0.00 | |
| Lonafarnib | INTRINSIK CORP | Change Order 1 to Statement of Work | \$0.00 | \$0.00 | |
| Lonafarnib | INTRINSIK CORP | Master Service Agreement | \$0.00 | \$0.00 | |

| Related | Contract Countainants | Description of Contract of Logo | Avexitide Cure | Lambda Cura | Lonafarnib |
|-----------------------|---------------------------|---|-------------------|---|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexidue Cure | Lambua Cure | Cure |
| Lonafarnib | INTRINSIK CORP | Statement of Work 14 | \$0.00 | \$0.00 | |
| Lonafarnib | Intrinsik Corp. | Statement of Work AMCP | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated July 10, 2020 | \$0.00 | | |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated March 29, 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated March 4, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Intrinsik Corp. | Statement of Work dated May 1,2020 | \$0.00 | \$0.00 | \$0.00 |
| | Intrinsik Health Sciences | | | | |
| Lambda | Inc. | Statement of Work | \$0.00 | * | |
| Lonafarnib | INTSEL CHIMOS | AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | INTSEL CHIMOS | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | inVentiv Health Clinique | | | | |
| Lonafarnib | Inc. | Bioanalytical Services Quotation | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 1 | INVENTIV HEALTH | | фо о о | #0.00 | #0.00 |
| Lonafarnib | CLINIQUE, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | inVentiv Health Public | | | | |
| | Relations LLC, a Syneos | | | | |
| Lonafarnib | Heath Company | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | Treath Company | Statement of Work | Ψ0.00 | \$0.00 | \$0.00 |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Avexitide | LLC | Statement of Work dated March 19, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | , , , , , | * | , , , , , |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Lonafarnib | LLC | Change Order | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | | | | |
| Lonafarnib | LLC | Statement of Work dated July 16, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | INVENTIV HEALTH | | | | |
| | PUBLIC RELATIONS, | a | #0.00 | #0.00 | фо оо |
| Lonafarnib | LLC | Statement of Work dated March 19, 2019 Change Notification Form dated May 15, | \$0.00 | \$0.00 | \$0.00 |
| Lanafamil | IOMA | 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA | Change Notification Form dated October 26, | | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA | 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | IQVIA | Warranty Letter re: Protocol number EIG- | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA | LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | IQVIA Biotech LLC | Change Notification Form 2 | \$0.00 | | \$0.00 |
| Lambda | IQVIA Biotech LLC | Change Notification Form 3 | \$0.00 | | \$0.00 |
| Lambda | IQVIA Biotech LLC | Change Notification Form 4 | \$0.00 | 1 | \$0.00 |
| | | | 72.00 | 1 | 73.30 |
| Lambda | IQVIA Biotech LLC | Change Proposal No. 1 | \$0.00 | \$6,151.93 | \$0.00 |
| | | | | 1 | |
| Lambda | IQVIA Biotech LLC | Statement of Work | \$0.00 | | \$0.00 |
| | | | | | |
| Lambda | IQVIA Biotech LLC | Work Order | \$0.00 | | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|------------------------|--|----------------|-------------|--------------------|
| Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 15 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | IQVIA Biotecti LLC | Letter of Delegation re: Protocol Number | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Clinical AB | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | IQVIA Clinical AB | EIG-LNF-VII | \$0.00 | \$0.00 | \$0.00 |
| | IQVIA Holdings, Inc.; | Consent for Disclosure of Study Information | | | |
| Lonafarnib | ClinAudits, LLC | to Auditors | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Cilifaudits, EEC | TSQM and TSQM Scoring Algorithm | \$0.00 | \$0.00 | \$0.00 |
| | | License (collectively, "Licensed Materials") | | | |
| | | for 150 participants - in connection with | | | |
| Lambda | IQVIA Inc. | protocol EIG-LMD | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | 22-Jun-20 General Services Agreement | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | 6/22/2020 General Services Agreement | \$0.00 | \$0.00 | |
| Lonaramo | TQ VII TINC. | Boulevard, Durham, NC 27703 ("IQVIA") | ψ0.00 | ψ0.00 | ψ0.00 |
| | | and relates to the Master Services | | | |
| Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Change Order #1 to Statement of Work | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Change Order #1 Work Order #1 | \$0.00 | · · | |
| Lonafarnib | IQVIA Inc. | Change Order #2 to Statement of Work | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Change Order #2 Work Order #1 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA Inc. | Change Order #3 to Statement of Work | \$0.00 | | |
| <u> </u> | 14 / 11 1 11101 | Change Order 1 to General Services | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Change Order 1 to Statement of Work | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Change Order 3 to Statement of Work | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Details of MSA and Work Order: | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Executed Work Order | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | General Services Agreement | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Master Service Agreement | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Novation Agreement | \$0.00 | | |
| Lonafarnib | IQVIA Inc. | Statement of Work | \$0.00 | | |
| | | Technical Agreement for the Provision of | , | , | * |
| Lonafarnib | IQVIA Inc. | EU Qualified Person | \$0.00 | \$0.00 | \$0.00 |
| | | Work Order #2: SOP Training | · | · | |
| | | Ī | | | |
| Lonafarnib | IQVIA Inc. | Work Order #2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Work Order dated May 1, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA Inc. | Work Order dated May 3, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Londrarino | JOYNA LAL JOYNA DDG | | | | |
| | IQVIA Ltd.; IQVIA RDS | | | | |
| | Eastern Holdings GmbH; | | | | |
| | IQVIA Tibbi Istatistik | | | | |
| | Ticaret ve Musavirlik | | | | |
| | Limited Sirketi; IQVIA | Letter of Delegation re: Protocol Number | | | |
| Lonafarnib | RDS Ges.m.b.H. | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| | | Authorization to Proceed Agreement dated | | | |
| Lonafarnib | IQVIA RDS INC. | June 30, 2022 | \$0.00 | \$0.00 | \$0.00 |

| D.1.4.3 | | | | | T |
|-----------------------|------------------------------|---|-------------------|-------------------|------------------------|
| Related | Control Constant | Description of Contract of Least | A | Lemb de Com | Lonafarnib |
| Asset | Contract Counterparty | Description of Contract of Lease Authorization to Proceed Agreement dated | Avexitide Cure | Lambda Cure | Cure |
| Lanafamil | IOVIA DDC INC | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | IQVIA RDS INC. | October 29, 2021 | \$0.00 | | |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | · | · · | |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA RDS Inc. | Change Order No. 2 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA RDS Inc. | Change Order No. 5 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA RDS INC. | Change Order No. 5 | \$0.00 | \$0.00 | |
| Lonafarnib | IQVIA RDS INC. | Details of General Services Agreement: | \$0.00 | \$0.00 | \$0.00 |
| T 0 " | TOTAL BEGINS | Details of Master Services Agreement and | #0.00 | 40.00 | 40.00 |
| Lonafarnib | IQVIA RDS INC. | Authorization to Proceed: | \$0.00 | | |
| Lonafarnib | IQVIA RDS INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | IQVIA RDS Ireland | EU Legal Representative of Non-EU | | | |
| Lonafarnib | Limited | Sponsor | \$0.00 | \$0.00 | \$0.00 |
| | | Sponsor's Commitment Form - Clinical Trial | | | |
| Lonafarnib | IQVIA RDS LTD, Israel | with an Investigational Product | \$0.00 | \$0.00 | \$0.00 |
| | ITR Laboratories Canada | | | | |
| Avexitide | Inc. | Revised (4) Supplementary Quotation | \$0.00 | \$0.00 | \$0.00 |
| | ITR Laboratories Canada | | | | |
| Avexitide | Inc. | Supplementary Quotation | \$0.00 | | |
| Lambda | J. A. [Name onf File] | Services Agreement | \$0.00 | \$400.00 | \$0.00 |
| | | | | | |
| | | Letter of Support for the University of | | | |
| | Jordan Feld M.D., MPH; | Toronto COVID-19 Action Initiative Grant | | | |
| Lambda | Toronto General Hospital | Application | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | |
| Lonafarnib- | CONSULTING | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | |
| Lonafarnib | CONSULTING | Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | |
| Lonafarnib- | CONSULTING | Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | |
| Lonafarnib- | CONSULTING | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | · |
| Lonafarnib- | CONSULTING | Statement of Work 4 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | | | |
| Lonafarnib | CONSULTING | Statement of Work 6 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | , | , | , |
| Lonafarnib- | CONSULTING | Statement of Work 7 | \$0.00 | \$0.00 | \$0.00 |
| | JUSTIN STINDT | | 40.00 | 40.00 | 40.00 |
| Lonafarnib- | CONSULTING | Statement of Work 8 | \$0.00 | \$0.00 | \$0.00 |
| Zonaramo | JUSTIN STINDT | Satement of Work o | φ0.00 | φσ.σσ | \$0.00 |
| Lonafarnib- | CONSULTING | Statement of Work 9 | \$0.00 | \$0.00 | \$0.00 |
| 201141411110 | CONSOLING | Substitute of more y | Ψ0.00 | Ψ0.00 | Ψ0.00 |
| | | | | | |
| | Justin Stindt of Justin | | | | |
| Lonafarnib | Stindt Consulting GmbH | Representation Mandate | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | Strict Consulting OmbH | representation wandate | \$0.00 | \$0.00 | \$\displays |
| | | | | | |
| | Kaiser Foundation | | | | |
| Lanafamik | Research Institute | Clinical Trial Agreement | \$0.00 | \$0.00 | ¢ስ ስስ |
| Lonafarnib | Research institute | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|--|---|----------------|---------------|-------------|
| Asset | Contract Counterparty KAROLINSKA | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | UNIVERSITY | | | | |
| Lonafarnib | HOSPITAL | Contract Period | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | KPS Life, LLC | Exhibit A- Work Order No. 2 | \$0.00 | | |
| Lonafarnib | KPS LIFE, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | KROLL CYBER | | | | |
| Lonafarnib | SECURITY, LLC | LOE | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work for Information Security | | | |
| Lonafarnib | Kroll Cyber Security LLC | and Computer Forensics Services | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Kion Cyber Security, EEC | and Computer Forensies Services | ψ0.00 | ψ0.00 | ψ0.00 |
| | KRYOCAL, LLC DBA | | | | |
| Avexitide | KYROSPHERE | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | KRYOCAL, LLC DBA | | | | |
| Lambda | KYROSPHERE | Statement of Understanding | \$0.00 | \$6,562.50 | \$0.00 |
| Lonafarnib | Kymanox Corporation | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Kymanox Corporation | Waster Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated Jul 20, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | , | 3, 1 | , , , , | * | , , , , , |
| Lonafarnib | Laviana Pharma Co., Ltd | Quotation dated July 31, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Liver Care Centre | | | | |
| Lonafarnib | Corporation d/b/a Toronto Liver Centre | Indemnification Letter re: Protocol No. EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Lonaiarnio | Liver Centre | Amendment No. 3 to Master Services and | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | LONZA BEND, INC. | Clinical Manufacture Agreement | \$0.00 | \$0.00 | \$0.00 |
| Tivexinae | EGIVERI BEIVE, IIVE. | Amendment No. 1 to the Commercial | φ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | LONZA BEND, INC. | Supply Agreement | \$0.00 | \$0.00 | |
| | | Amendment No. 2 to the Commercial | | | |
| Lonafarnib | LONZA BEND, INC. | Supply Agreement | \$0.00 | \$0.00 | |
| | LONG A DENIE DIG | | #0.00 | # 0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 2 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 2 to Statement of Work | \$0.00 | \$0.00 | |
| Lonaranno | LONZA BLIND, INC. | Change order 2 to Statement of Work | ψ0.00 | ψ0.00 | |
| Lonafarnib | LONZA BEND, INC. | Change Order 4 to Statement of Work | \$0.00 | \$0.00 | |
| | · | | | |] |
| Lonafarnib | LONZA BEND, INC. | Change Order 8 to Statement of Work | \$0.00 | \$0.00 | |
| | | Services and Clinical Manufacture | | | |
| Lonafarnib | LONZA BEND, INC. | Agreement dated 15 December, 2022 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA DEND INC | Statement of Work | \$0.00 | 90.00 | |
| Lonaranno | LONZA BEND, INC. | Statement of Work | \$0.00 | \$0.00 | 1 |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 10 April 2023 | \$0.00 | \$0.00 | \$21,000.00 |
| | | | \$0.00 | \$0.00 | 1 |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |
| | | | | | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 13 June 2018 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|-------------------------------|---|----------------|-------------|--------------------|
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 14 August 2020 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 22 April 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 4 December 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 5 April 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 8 May 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work dated 9 January 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | Lonza Bend; Patheon Canada | Total Transportation Management ("TTM") Freight Quote | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Change Order #7 to E141-8598 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Change Order No.1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Amendment 1 to SOW - PN 03706 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 2 to SOW PN 05119 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order 6 to SOW E141-8083 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05119 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05281 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order Form PN 05895 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001073 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 001256 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order No.1 to Statement of Work PN 04064 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Lonza Change Order to SOW PN0405 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Proposed Statement of Work (SOW) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Stability Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-092310 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & Biotech | Validation Proposal - PN-136736 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|---|----------------|-------------|--------------------|
| Lonafarnib | Lonza Pharma & BioTech | Validation Proposal dated 29 January 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Validation Proposal dated 6 April 2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-030137" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work "PN-031335" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work PN-014163 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 1 to Statement of Work PN-033823 | \$0.00 | | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-030137" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-031335" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work "PN-05281" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 2 to Statement of Work PN-033823 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 3 to Statement of Work "PN-04064" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 3 to Statement of Work "PN-05281" | \$0.00 | | |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 3 to Statement of Work PN-033823 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Change Order No. 5 to Statement of Work "PN-05119" | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Development Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Proposed Change Order #1 to Eiger SOW | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech; Bend Research Inc. | Proposed Statement of Work (SOW) dated August 20, 2018 | \$0.00 | | |

| Related | Contract Country | Description of Contract of Logo | Avexitide Cure | Lambda Cuus | Lonafarnib |
|------------|--------------------------------------|--|----------------|-------------|------------|
| Asset | Contract Counterparty Lonza Pharma & | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonafarnib | Inc. | August 8, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Lonza Pharma & | August 6, 2016 | \$0.00 | \$0.00 | \$0.00 |
| | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonafarnib | Inc. | May 30, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Lonza Pharma & | 171ay 30, 2010 | ψ0.00 | ψ0.00 | ψ0.00 |
| | BioTech; Bend Research | Proposed Statement of Work (SOW) dated | | | |
| Lonafarnib | Inc. | May 8,2018 | \$0.00 | \$0.00 | \$0.00 |
| | Lonza Pharma & | | 7 | 40.00 | 44444 |
| | BioTech; Bend Research | | | | |
| Lonafarnib | Inc. | Validation Proposal | \$0.00 | \$0.00 | \$0.00 |
| | | • | | | |
| Lonafarnib | Luto Research Limited | Quote dated 16^ January 2020 | \$0.00 | \$0.00 | \$0.00 |
| T C 11 | T . D . 1 T 1 | 0 . 1 . 11 . 0 . 1 . 2010 | ФО ОО | ФО ОО | ФО ОО |
| Lonafarnib | Luto Research Limited | Quote dated 1st October 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Luto Research Limited | Quote dated 7th April 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | Euro Researen Eminea | Quote duted /til / ipili 2021 | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Luto Research Ltd. | Quote for User Testing | \$0.00 | \$0.00 | \$0.00 |
| | M. C., MD [Name on | <u> </u> | 40.00 | 40.00 | 44444 |
| Lambda | File] | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| | - | | | | |
| Avexitide | M. T., MD [Name on File] | | \$232.00 | \$0.00 | \$0.00 |
| | | Statement of Work Hepatitis Delta Virus | | | |
| Lambda | M. B. C. [Name on File] | (HDV) | \$0.00 | \$0.00 | \$0.00 |
| - 0 " | Mary Bordeaux | Statement of Work Hepatitis Delta Virus | | | |
| Lonafarnib | Consulting | (HDV) | \$0.00 | · · | \$0.00 |
| Lonafarnib | MC2 | Change Order No. 1 | \$0.00 | \$0.00 | \$0.00 |
| T C 11. | MC2 | Standard Service Agreement dated | ¢0.00 | £0.00 | ¢0.00 |
| Lonafarnib | MC2 | 2/14/2019 Standard Service Agreement dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MC2 | 2/27/2020 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | IVIC2 | Standard Service Agreement dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | IVIC2 | Standard Service Agreement dated | Ψ0.00 | Ψ0.00 | Ψ0.00 |
| Lonafarnib | MC2 | 8/23/2019 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 1 | ψ0.00 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 1st Amendment to Task Order 3 | _ | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 2nd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 1 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | 3rd Amendment to Task Order 3 | | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-005 | \$17.79 | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-006 | ψ1/./ <i>9</i> | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 1 to LOI AVX-007 | _ | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 2 to LOI AVX-005 | _ | \$0.00 | \$0.00 |
| Avexitide | MEDPACE INC. | Amendment 2 to LOI AVX-006 | | \$0.00 | \$0.00 |

| D 1 / 1 | | | | | T C 1 |
|--------------------|------------------------------------|--|----------------|------------------|--------------------|
| Related | Contract Countamout | Description of Contract of Lease | Avexitide Cure | Lambda Cura | Lonafarnib |
| Asset Avexitide | Contract Counterparty MEDPACE INC. | Description of Contract of Lease LOI AVX007 | Avexitude Cure | \$0.00 | Cure \$0.00 |
| Avexitide | MEDPACE INC. | LOI CHI | _ | \$0.00 | |
| Avexitide | MEDI ACE INC. MEDPACE INC. | LOI CHI Children | | \$0.00 | |
| Avexitide | MEDPACE INC. | Master Service Agreement | _ | \$0.00 | |
| Avexitide | MEDPACE INC. | Task Order 2 | | \$0.00 | \$0.00 |
| Avexitide | MEDI ACE INC. | Task Order 3 | _ | \$0.00 | |
| Avexitide | Merck Sharp & Dohme | Task Older 5 | | \$0.00 | \$0.00 |
| | LLC; The University of | | | | |
| | Texas M.D. Anderson | | \$0.00 | <u>\$0.00</u> | \$0.00 |
| Lonafarnib | Cancer Center | Amendment dated September 18, 2006 | | | |
| <u> Lonararino</u> | Merck Sharp & Dohme | 7 michament dated September 10, 2000 | | | |
| Lonafarnib | LLC | Amendment 1 | \$0.00 | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 2 | <u>\$0.00</u> | \$0.00 | <u>\$0.00</u> |
| | Merck Sharp & Dohme | | #0.00 | ФО ОО | фо. o o |
| Lonafarnib | LLC | Amendment 3 | <u>\$0.00</u> | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | \$0.00 | #0.00 | фо оо |
| Lonafarnib | LLC | Amendment 4 | <u>\$0.00</u> | \$0.00 | <u>\$0.00</u> |
| | Merck Sharp & Dohme | | \$0.00 | #0.00 | фо оо |
| Lonafarnib | LLC | Amendment 5 | \$0.00 | \$0.00 | <u>\$0.00</u> |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 6 | \$0.00 | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | | | |
| Lonafarnib | LLC | Amendment 7 | \$0.00 | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | Amendment to the Pharmacovigilance | | | |
| Lonafarnib | LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | | | |
| | LLC; The University of | | \$0.00 | \$0.00 | \$0.00 |
| | Texas M.D. Anderson | Drug Supply Agreement dated August 29, | 40.00 | 40100 | - |
| <u>Lonafarnib</u> | Cancer Center | 2017 | | | |
| - 0 " | Merck Sharp & Dohme | License Agreement dated September 30, | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | 2010 | | | |
| | Merck Sharp & Dohme | | | | |
| | LLC; The University of | | | | |
| | Texas Health Center at | | | | |
| | Tyler; The University of | | | | |
| | Texas Health Science | | | | |
| | Center at Houston; The | | | | |
| | University of Texas | | \$0.00 | \$0.00 | \$0.00 |
| | Health Science Center at | | \$0.00 | \$0.00 | \$0.00 |
| | San Antonio; The | | | | |
| | University of Texas M.D. | | | | |
| | Anderson Cancer Center; | | | | |
| | The University of Texas | | | | |
| | Health Southwestern | Investigator Initiated Master Clinical Trail | | | |
| Lonafarnib | Medical Center at Dallas | Research Agreement dated July 1, 2004 | | | |
| | Merck Sharp & Dohme | Memo of Understanding (MOU) the | | | |
| Lonafarnib | LLC | Pharmacovigilance Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Merck Sharp & Dohme | | ψ0.00 | \$5.50 | ψ0.00 |
| Lonafarnib | LLC | Safety Agreement | \$0.00 | \$0.00 | \$0.00 |
| _onarailil0 | | Salety rigidellicity | ψοίου | ψυισο | ψ0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|-----------------------|--------------------------------------|--|-------------------|-------------------|--------------------|
| | Merck Sharp & Dohme | | #0.00 | #0.00 | # 0.00 |
| Lonafarnib | LLC | Termination Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | MillBridge Pharmaceutical Science | Cost Proposal - API Supplier Qualification Audit | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Filatifiaceutical Science | Audit | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MillBridge Pharmaceutical Science | Cost Proposal - Supplier Qualification Audit | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MillBridge Pharmaceutical Science | Cost Proposal - Supplier Qualification Audits | \$0.00 | \$0.00 | \$0.00 |
| I | MIND I MATTED LLC | Statement of West | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 14 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 15 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 16 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 17 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MIND+MATTER LLC | Statement of Work 22 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | MKO Global Partners | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | MPI Research | Archived Materials Report | \$0.00 | | |
| Lonafarnib | Myoderm | Quote re: Destruction of Drug | \$0.00 | | |
| Lambda | N/A | Letter of Legal Representative | \$0.00 | | |
| Lonafarnib | NDA GROUP AB | Master Service Agreement | \$0.00 | | |
| Lonafarnib | NDA REGULATORY SCIENCE LTD | Work Order | \$0.00 | | |
| Lonafarnib | NEOPHARM LTD. | Distribution Agreement Terms for an Exclusive Distribution | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | NEOPHARM LTD. | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Neurozon LLC | Statement of Work Refolding Studies of IL- 29 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Neurozon LLC | Statement of Work Refolding Studies of IL-29 (Supplement) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | NIH | Author Agreement | \$0.00 | | |
| Lambda | NORTH RIM CONSULTING, INC | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|---------------------------------|---|--|--------------------|-------------------|--------------------|
| Lambda | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal 7 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Novella Clinical LLC | Change Proposal No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 6 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 7 | \$0.00 | \$0.00 | |
| Avexitide | Novocol Pharma OneSource Regulatory | Development Project Proposal | \$0.00 | \$0.00 | |
| Lonafarnib | Coregon Health and | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Science University | Notice of Assignment | \$0.00 | \$0.00 | <u>\$0.00</u> |
| Lonafarnib | Origami Evenements SARL | Contrat de Participation a un Evenement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Origami Evènements SARL | Contract of Participation in an Event | \$0.00 | \$0.00 | \$0.00 |
| | ORPHAN DRUG | Constant of Function in an 27 vite | φσισσ | φοισσ | \$0.00 |
| Lonafarnib | | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| T C 7 | ORPHAN DRUG | | \$ 0.00 | #0.00 | Ф0.00 |
| Lonafarnib Lambda | Oxford Finance | Services and Confidentiality Agreement Amendment | \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| | | Approval of Second Amendment to Oxford | | | |
| Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | | |
| Lambda | Oxford Finance | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| I amalada | Oufand Finance | Second Amendment to Loan and Security | \$0.00 | \$0.00 | \$0.00 |
| Lambda Lonafarnib | Oxford Finance Oxford Finance | Agreement Loan and Security Agreement | \$0.00 \$0.00 | \$0.00 \$0.00 | |
| Lonaranno | Oxford Finance | Third Amendment to Loan and Security | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Oxford Finance | Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance LLC | Amendment Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | | | | |
| Lambda | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Oxford Finance LLC, as Collateral Agent | Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-------------|---|---|----------------|-------------|-------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | 0.0.15 | | | | |
| Lambda | Oxford Finance LLC, as Collateral Agent | Second Amendment to Loan and Security Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Conateral Agent | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | Second Amendment to Loan and Security | | | |
| Lambda | Collateral Agent | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Oxford Finance LLC, as | | | | |
| Lonafarnib | Collateral Agent | Consent to Sale of Priority Review Voucher | \$0.00 | \$0.00 | \$0.00 |
| 20114141110 | - Commercial Ligent | Constitution Switch of Filtering Teachers | φσ.σσ | \$0.00 | \$0.00 |
| Lonafarnib | Oxford Finance, LLC | Success Fee Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DA CIEIC DILADAGA | | | | |
| Lonafarnib | PACIFIC PHARMA GROUP, LLC | Statement of Work describes Services and Deliverables to be performed and | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | GROUF, LLC | Deriverables to be performed and | \$0.00 | \$0.00 | \$0.00 |
| | Pacific Pharmaceutical | | | | |
| | Services and Alcami | | | | |
| Avexitide | Company | Services Agreement | \$25,061.25 | \$0.00 | \$0.00 |
| Lanafamih | Palmetto GBA; Ed Gazarkh | Latter no. TDA Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PARTNERS4ACCESS | Letter re: TPA Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | B.V. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work - EU / Access & | | | |
| | PARTNERS4ACCESS | Commercialisation Support dated Dec 23rd, | | | |
| Lonafarnib | B.V. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Statement of Work - EU / Access & Commercialisation Support dated Dec 23rd, | | | |
| Lonafarnib | B.V. | 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | B. V. | Statement of Work - Lonafarnib EU / | ψ0.00 | ψ0.00 | Ψ0.00 |
| | PARTNERS4ACCESS | Access & Commercialisation Support dated | | | |
| Lonafarnib | B.V. | April 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Statement of Work - Lonafarnib EU / | | | |
| Lonafarnib | B.V. | Access & Commercialisation Support dated July 30th, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | D. V. | Statement of Work - Lonafarnib EU / | \$0.00 | \$0.00 | \$0.00 |
| | PARTNERS4ACCESS | Access & Commercialisation Support dated | | | |
| Lonafarnib | B.V. | March 10th, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work - Lonafarnib EU | | | |
| I fo | PARTNERS4ACCESS | Commercialisation Planning dated March 5, | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | B.V. | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon Inc, part of | | | | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 10 | \$0.00 | \$0.00 | |
| | | | | | |
| T C '1 | Patheon Inc, part of | | #0.00 | #A AA | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 11 | \$0.00 | \$0.00 | \$68,895.30 |
| | Patheon Inc, part of | | | | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 13 | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|------------|---|---|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | part of the state | | | | |
| | Patheon Inc, part of | | | | |
| Lonafarnib | Thermo Fisher Scientific | Change of Scope 14 | \$0.00 | \$0.00 | |
| Avexitide | PATHEON INC. | GRP Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | PATHEON INC. | Project Proposal dated Dec 13, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | PATHEON INC. | Validation Project Plan | \$0.00 | \$0.00 | \$0.00 |
| Lambda | PATHEON INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Patheon Inc. | Change of Scope 1 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 11 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 12 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 13 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 14 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 15 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 16 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 17 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 18 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 19 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 2 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 20 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 21 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 22 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 23 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 24 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 25 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 27 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 28 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 29 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 1 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 2 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 3 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 4 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 3 C-TRC 5 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PATHEON INC. | Change of Scope 30 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Change of Scope 31 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 32 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 34 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Change of Scope 35 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 36 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 37 | \$0.00 | \$0.00 |] |
| Lonafarnib | PATHEON INC. | Change of Scope 4 | \$0.00 | \$0.00 |] |
| Lonafarnib | Patheon Inc. | Change of Scope 40 | \$0.00 | \$0.00 | |
| Lonafarnib | Patheon Inc. | Change of Scope 41 | \$0.00 | | 1 |
| Lonafarnib | Patheon Inc. | Change of Scope 43 | \$0.00 | | 36X X93 30 |
| Lonafarnib | PATHEON INC. | Change of Scope 44 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 45 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 46 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 47 | \$0.00 | | 4 |
| Lonafarnib | Patheon Inc. | Change of Scope 5 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 52 | \$0.00 | | |

| Related | | | | | Lonafarnib |
|--------------------------|------------------------------|---|-----------------------|--------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | | Cure |
| Lonafarnib | PATHEON INC. | Change of Scope 6 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 7 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 8 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Change of Scope 9 | \$0.00 | \$0.00 | |
| - 2 " | | Change of Scope COS-55-R0 to Proposal | 40.00 | | |
| Lonafarnib | Patheon Inc. | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | |
| T C '1 | D d I | Change of Scope Proposal For Eiger | Φ0.00 | #0.00 | |
| Lonafarnib Lonafarnib | Patheon Inc. | Biopharmaceuticals | \$0.00 \$0.00 | | |
| Lonaiarnib | Patheon Inc. | Change of Scope to Proposal Change of Scope | \$0.00 | \$0.00 | |
| I | Dath and Inc | | \$0.00 | \$0.00 | |
| Lonafarnib Lonafarnib | Patheon Inc. PATHEON INC. | Proposal For Eiger Biopharmaceuticals | \$0.00 \$0.00 | | |
| Lonafarnib | PATHEON INC. PATHEON INC. | Change Order COS 1 | \$0.00 | | |
| Lonafarnib | Patheon Inc. | Phase II CTM Manufacturing Proposal | \$0.00 | * | |
| Lonararino | rauleon inc. | Product Agreement for Lonafarnib - | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Progeria | \$0.00 | \$0.00 | |
| Lonaranno | FATHEON INC. | Product Agreement") for Lonafarnib | \$0.00 | \$0.00 | |
| | | Capsules 50 mg and 7S mg ("Product") to | | | |
| Lonafarnib | PATHEON INC. | the Master Service | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Project Proposal dated July 27, 2019 | \$0.00 | | |
| Lonafarnib | PATHEON INC. | Project Proposal dated May 21, 2021 | \$0.00 | \$0.00 | |
| Lonaranno | TATHLON INC. | Troject Proposar dated Way 21, 2021 | ψ0.00 | Ψ0.00 | |
| Lonafarnib | PATHEON INC. | Project Proposal dated September 27, 2021 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Project Proposal March 30, 2020 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Quote dated April 2, 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Quote dated July 09, 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Quote dated March 25,2019 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Quote dated March 27, 2019 | \$0.00 | \$0.00 | |
| Lonafarnib | PATHEON INC. | Quote for Size 4 dated July 09, 2019 | \$0.00 | \$0.00 | |
| | | | | | |
| | | | | | |
| | Patheon Inc.; Gerteis | | | | |
| | Maschinen + | | | | |
| | Processengineering AG; | | | | |
| Lonafarnib | Joseph Lannen | Quotation No 106954 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | Patheon Italia Spa | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Change of Scope to Proposal 1 | 4 | \$0.00 | \$0.00 |
| | Detheron Mannet | | | | |
| | Patheon Manufacturing | | | Ф0.00 | #0.00 |
| Avexitide | Services LLC | Change of Scope to Proposal 10 | 4 | \$0.00 | \$0.00 |
| | Dath on Mounts strains | | | | |
| Avoriti de | Patheon Manufacturing | Change of Coope to Decree 114 | | ¢0.00 | ¢0.00 |
| Avexitide | Services LLC | Change of Scope to Proposal 14 | - | \$0.00 | \$0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Change of Scope to Proposal 14-RO | | \$0.00 | \$0.00 |
| AVEXITIE | Del vices LLC | Change of Scope to Froposal 14-KO | i | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-------------|---------------------------------------|---|----------------|---------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | D d M C d | | | | |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 1 R0 | | \$0.00 | \$0.00 |
| Avexilide | Services LLC | Patneon Change of Scope 1 RU | - | \$0.00 | \$0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Patheon Change of Scope 11 | \$111,092.41 | \$0.00 | \$0.00 |
| | | | \$111,072.41 | | |
| | Patheon Manufacturing | | | #0.00 | ФО ОО |
| Avexitide | Services LLC | Patheon Change of Scope 12 | - | \$0.00 | \$0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Patheon Change of Scope 13 | | \$0.00 | \$0.00 |
| | | | 1 | | |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Patheon Change of Scope 2 | | \$0.00 | \$0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Patheon Change of Scope 2-R2 | | \$0.00 | \$0.00 |
| 11101111111 | Services EEC | Tambén change of scope 2 1t2 | 1 | φ0.00 | \$0.00 |
| | Patheon Manufacturing | | | | |
| Avexitide | Services LLC | Patheon Change of Scope 3 | | \$0.00 | \$0.00 |
| | | | | | |
| Avexitide | Patheon Manufacturing Services LLC | Patheon Change of Scope 4 | | \$0.00 | \$0.00 |
| Avexilide | Services LLC | Patheon Change of Scope 4 | | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 10 | \$0.00 | \$0.00 | \$0.00 |
| | D. d | al as pro H 11 | Ф0.00 | # 0.00 | фо оо |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 11 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 15 | \$0.00 | \$0.00 | \$0.00 |
| Lumoda | Tuncon of Linned | Change of Scope TEG TE 13 | ψ0.00 | ψ0.00 | ψ0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 16 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 18 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | Change of Scope - PEG-rIL- 9 | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | Tatheon OK Limited | Change of Scope COS P-MNC-101564-R3- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | COS-07-R0 20200424 | \$0.00 | \$0.00 | \$0.00 |
| | | Change of Scope COS P-MNC-101564-R3- | | | |
| Lambda | Patheon UK Limited | COS-11-R0_20210712 | \$0.00 | \$0.00 | \$0.00 |
| T 1 | Ded and IIIZ I 1 2 1 | Change of Scope COS P-MNC-101564-R3- | #A AA | φο oo | #0.00 |
| Lambda | Patheon UK Limited | COS-11-R0_20210712 (1) Change of Scope COS P-MNC-101564-R3- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Patheon UK Limited | COS-02-R0 20170119 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | I dilleon of Emilieu | Change of Scope COS-17-R0 to P-MNC- | ψ0.00 | Ψ0.00 | Ψ0.00 |
| Lambda | Patheon UK Limited | 101564-R3_20220324 | \$0.00 | \$0.00 | \$0.00 |
| | | Change of Scope COS-P-MNC-101564-R3- | | | |
| Lambda | Patheon UK Limited | COS-08-R3_20210309 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|---|--|----------------|---------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | Change of Scope P-MNC-101564-R3-COS- | | | |
| Lambda | Patheon UK Limited | 11-R1_20211103 | \$0.00 | \$0.00 | \$0.00 |
| | D. d. THE T. C. d. | Purchase Order Patheon UK_COS 3 P-MNC | # 0.00 | # 0.00 | #0.00 |
| Lambda | Patheon UK Limited | 101564-R3 20171129 | \$0.00 | \$0.00 | \$0.00 |
| T1. 1. | Dedice a HIZ I built 1 | Purchase Order Patheon UK_COS 4 P-MNC | ¢0.00 | ¢0.00 | \$0.00 |
| Lambda | Patheon UK Limited | 101564-R3_20171115 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | PATHEON UK LIMITED | Validation Master Plan | \$0.00 | \$0.00 | \$0.00 |
| | D. d. THE TANK I I I | | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 14 P- | фо. оо | фо. оо | #0.00 |
| Lambda | Fisher Clinical Services | MNC-101564-R3_2022 | \$0.00 | \$0.00 | \$0.00 |
| | Doth and LIV Limited. | Change of Same Both and LIV COS 20 B | | | |
| T amala da | Patheon UK Limited; | Change of Scope Patheon UK_COS 20 P- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20220722 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon UK Limited; | Change of Scope Patheon UK COS 21 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4 20221025 | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | I isher Chinear Services | MINO-101304-R4_20221023 | ψ0.00 | ψ0.00 | ψ0.00 |
| | Patheon UK Limited; | Change of Scope Patheon UK COS 24 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4 20230209 | \$0.00 | \$0.00 | \$0.00 |
| | | | 40100 | 40100 | 4000 |
| | Patheon UK Limited; | Change of Scope Patheon UK COS 25 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | | _ | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 26 P- | | | |
| Lambda | Fisher Clinical Services | MNC-101564-R4_20221118 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon UK Limited; | Change of Scope Patheon UK_COS 29-R0 | | | |
| Lambda | Fisher Clinical Services | to P-MNC-101564-R4_20240213 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Patheon UK Limited; | Change of Scope: Prefilled Syringes COS 12 | | | |
| Lambda | Fisher Clinical Services | to P-MNC-101564-R3_20220125 | \$0.00 | \$0.00 | \$0.00 |
| | Dealers IIIZ I for to de | Change of Scope: Prefilled Syringes | | | |
| T amala da | Patheon UK Limited; Fisher Clinical Services | Patheon UK_P-MNC-101564-R4-COS-23- | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Fisher Clinical Services | R0_20220922 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon UK Limited; | Change of Scope: Proposal COS 12 to C- | | | |
| Lonafarnib | Fisher Clinical Services | TRC-121992-R4 20230429 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | I isher Chinear Services | TKC-121772-K4_20230427 | Ψ0.00 | Ψ0.00 | ψ0.00 |
| | Patheon, Part of Thermo | | | | |
| | Fischer Scientific; | Element Quote Element Quote 20-012- | | | |
| Lonafarnib | Element Toronto | 162900 Rev 1 20200420 | \$0.00 | \$0.00 | \$0.00 |
| | | | * | * | , |
| | Patheon, Part of | Total Transportation Management ("TTM") | | | |
| Lonafarnib | Thermofisher | Freight Quote 453021_20201202 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-03-R2 to Project | | | |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-04-R1 to Project | | | |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-05-R0 to Project | | | |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|---|---|---------------|--------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Patheon; Thermo Fisher | Change of Scope COS-06-R0 to Project | | | |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-07-R0 to Project | * | * | * |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-08-R0 to Project | * | * | , |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of scope COS-09-R1 to Project | * | * | , |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-19-R0 to Project | 40.00 | 44100 | 40.00 |
| Avexitide | Scientific | Proposal No. P-GRP-283533-R1 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-38-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP- 114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Patheon; Thermo Fisher | Change of Scope COS-39-R0 to Proposal | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Patheon; Thermo Fisher | Change of Scope COS-42-R1 to Proposal | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | Patheon; Thermo Fisher | Change of Scope COS-48-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | Patheon; Thermo Fisher | Change of Scope COS-49-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | Patheon; Thermo Fisher | Change of Scope COS-50-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | Patheon; Thermo Fisher | Change of Scope COS-51-R0 to Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | የ ስ ስስ | \$0.00 |
| Lonararmo | Patheon; Thermo Fisher | | \$0.00 | \$0.00 | \$0.00 |
| I C 11. | - | Change of Scope COS-53-R0 to Proposal | ¢0.00 | #0.00 | £0.00 |
| Lonafarnib | Scientific Fig. 1 | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-54-R2 to Proposal | Ф0.00 | Ф0.00 | Ф0.00 |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-57-R1 to Proposal | фо. оо | фо. оо | фо. о о |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | Patheon; Thermo Fisher | Change of Scope COS-58-R0 to Proposal | | | |
| Lonafarnib | Scientific | No. P-TRP-114750-R2 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Pharma Solutions USA, | | | | *** ******* |
| Lonafarnib | Inc. | Services Agreement | \$0.00 | \$0.00 | \$745.49 |
| | | | | | |
| | Pharmaceutical Research | | | | |
| Lonafarnib | Associates, Inc. | Services Agreement | \$0.00 | \$0.00 | \$658.79 |
| | | | | | |
| Lambda | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 | \$0.00 | \$0.00 | \$0.00 |
| | | WKO-EIG-1225 Lambda Interferon CMC | | | |
| Lambda | PharmaDirections, Inc. | Oversight | \$0.00 | \$0.00 | \$0.00 |
| | | WKO-EIG-1361 CMC Support: Interferon | | | |
| Lambda | PharmaDirections, Inc. | Lambda | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | PharmaDirections, Inc. | WKO-EIG-991 Ad hoc Consulting | \$0.00 | \$0.00 | \$0.00 |
| I | Dhama Dina stissas Is | Amondment #1 WWO FIG 970 | #0.00 | ቀለ ላላ | |
| Lonafarnib | PharmaDirections, Inc. | Amendment #1 WKO-EIG-879 | \$0.00 | \$0.00 | 1 |
| Lonafarnib | PharmaDirections, Inc. | Amendment #2 WKO-EIG-879 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---|---|--------------------------|-------------|--------------------|
| Lonafarnib | PharmaDirections, Inc. | WKO-EIG-1362 CMC Support: Lonafarnib Progeria Commercial | \$0.00 | \$0.00 | \$2,702.30 |
| Lonafarnib | PharmaDirections, Inc. | WKO-EIG-879 Ad hoc Consulting | \$0.00 | \$0.00 | |
| Lonafarnib | PharmaLex UK Services Limited | Estimate | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PHARMALEX US CORPORATION | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Pharmalytical Consulting LLC | Services Agreement | \$3,515.00 \$1,520.00 | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order 1 to Work Order No. 6 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 6 | \$7,848.75 | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Third Amendment to the Statement of Work #1 Agreement | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 4 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 6 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 7 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 8 | | \$0.00 | \$0.00 |
| Avexitide | PharmaPace, Inc. | Work Order No. 9 | | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 1 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 2 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 3 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Change Order No. 4 to Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PharmaPace, Inc. | Work Order No. 5 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | PINEX COMERCIAL IMPORTADORA E SERVICOS LTDA | Letter for Authorization | \$0.00 | \$0.00 | \$0.00 |
| Lambda | PLATFORM LIFESCIENCES INC. | Professional Services Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|---------------------|---|---|------------------|------------------|------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | POLARIS | | Φ0.00 | ФО ОО | ФО ОО |
| Avexitide | CONSULTING, INC. | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| | POLARIS | | | | |
| Avexitide | CONSULTING, INC. | Master Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | POLARIS | | Ф0.00 | # 0.00 | фо оо |
| Avexitide | CONSULTING, INC. | Service Agreement August 15, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | POLARIS | | | | |
| Avexitide | CONSULTING, INC. | Service Agreement dated 24 Jul 2023 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | POLARIS | | Ф0.00 | # 0.00 | Ф0.00 |
| Avexitide | CONSULTING, INC. | Service Agreement dated August 8, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | PPD Development, L.P. | LMD Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | - | | | |
| | | | Ф0.00 | # 0.00 | Ф0.00 |
| Lambda Avexitide | PPD Development, L.P. PPS | Proposal Services Agreement | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Avexitide | 113 | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Proforma Element 3 | Order Acknowledgement | \$0.00 | \$0.00 | \$0.00 |
| | | Order Acknowledgement dated October 23, | | | |
| Lonafarnib | Proforma Element 3 | 2020 | \$0.00 | \$0.00 | \$0.00 |
| | ProPharma Group | | | | |
| | Holdings, LLC; | Project Assignment #3 Under Consulting | | | |
| Lambda | ProPharma Group, LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | n ni c | | | | |
| | ProPharma Group Holdings, LLC; | Project Assignment #4 Under Consulting | | | |
| Lambda | ProPharma Group, LLC | Agreement | \$0.00 | \$0.00 | \$0.00 |
| 241110 44 | 1707 11411114 010 449, 220 | | \$0.00 | φσ.σσ | \$0.00 |
| | ProPharma Group | | | | |
| T C 1 | Holdings, LLC; | Project Assignment #6 Under Consulting | ФО ОО | ФО ОО | ФО ОО |
| Lonafarnib | ProPharma Group, LLC Q SQUARED | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | SOLUTIONS | Hold Harmless Agreement dated April 7, | | | |
| Lambda | HOLDINGS, LLC | 2021 | \$0.00 | \$0.00 | \$0.00 |
| | Q SQUARED | | | | |
| | SOLUTIONS | Hold Harmless Agreement dated 17 August | Ф0.00 | # 0.00 | #2.5 0.00 |
| Lonafarnib | HOLDINGS, LLC Q SQUARED | 2016 | \$0.00 | \$0.00 | \$250.00 |
| | SOLUTIONS | | | | |
| Lonafarnib | HOLDINGS, LLC | Work Order dated May 3rd, 2019 | \$0.00 | \$0.00 | \$250.00 |
| | | j | | | |
| | 0 (D) | | | | |
| Lonafarnib | Quest Diagnostics Clinical Laboratories, Inc. | Change Order No. 1 to Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| LOHAIAITHD | Laudiatories, IIIC. | Change Order No. 1 to Statement of work | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------------------|--------------------------------|---|------------------|--------------|--------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | Quest Diagnostics Clinical | Change Order No. 2 to SOW Type 4: | | | |
| Lonafarnib | Laboratories, Inc. | Combined Services | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Eurorium s, me. | Comomed Services | ψο.σσ | ψ0.00 | ψ0.00 |
| | | | | | |
| | QUEST DIAGNOSTICS | | | | |
| Lonafarnib | INCORPORATED | Amendment 1 to Statement of Work 1 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | QUEST DIAGNOSTICS | | | | |
| Lonafarnib | INCORPORATED | Amendment 2 to Statement of Work 2 | \$0.00 | \$0.00 | \$0.00 |
| | | | , , , , , | * | * |
| | | | | | |
| | QUEST DIAGNOSTICS | | | | |
| Lonafarnib | INCORPORATED | Master Data Licensing Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | QUEST DIAGNOSTICS | | | | |
| Lonafarnib | INCORPORATED | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | <u> </u> | | | |
| | | | | | |
| | QUEST DIAGNOSTICS | G | Ф0.00 | Ф0.00 | фо оо |
| Lonafarnib | INCORPORATED | Statement of Work SOW#2 | \$0.00 | \$0.00 | \$0.00 |
| | Quest Diagnostics | | | | |
| Lonafarnib | Incorporated | Statement of Work SOW#3 | \$0.00 | \$0.00 | \$0.00 |
| | 1 | Letter of Delegation re: Protocol Number | | | |
| Lonafarnib | Quintiles Vietnam LLC | EIG-LNF-011 | \$0.00 | | \$0.00 |
| Lambda | R3LS, LLC | Statement of Work SOW#3 | \$0.00 | \$0.00 | \$0.00 |
| | DAMDAM HEAT TH | | | | |
| Lambda | RAMBAM HEALTH CORPORATION | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lamoda | CORTORATION | Chinical That Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RAMBAM HEALTH | | | | |
| Lonafarnib | CORPORATION | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 11 | REGULATORY | | ФО ОО | #0.00 | #0.00 |
| Lonafarnib | PHARMA NET SRL | Consulting Agreement Project Assignment #1 Under Consulting | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Regulatory Pharma Net srl | | \$0.00 | \$0.00 | \$0.00 |
| Lonaranio | Tregulatory i marina i vet bir | rgreement | \$0.00 | ψ0.00 | ψ0.00 |
| | Research Triangle | | | | |
| | Institute DBA RTI | | | | |
| Lonafarnib | International | Proposal 0282201.557 | \$0.00 | | |
| Avexitide | RHO, INC. | LOI | \$0.00 | | |
| Avexitide Avexitide | RHO, INC. | LOI Extension LOI Second Extension | \$0.00 \$0.00 | | |
| Avexitide | RHO, INC. RHO, INC. | LOI Second Extension LOI Third Extension | \$0.00 | | |
| Avexitide | RHO, INC. | Master Services Agreement | \$0.00 | | |

| Doloted | | | | | Lonafarnib |
|---|---------------------------|---|-------------------|-------------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | ROTE LISTE Service | | | | |
| Lonafarnib | GmbH | Contract for the Publication | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Work Order 19, April 22, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | | , , , | | | |
| T 0 11 | RRD INTERNATIONAL, | W. 1 0 1 10 4 4 20 2022 | #0.00 | Ф0.00 | Ф0.00 |
| Lonafarnib | LLC | Work Order 19, August 28, 2023 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib- | LLC | Work Order 20, April 22, 2022 | \$0.00 | \$0.00 | \$ 0.00 |
| | RRD INTERNATIONAL | | | | |
| Lonafarnib | LLC | Work Order 20, April 22, 2022 | \$0.00 | \$0.00 | \$0.00 |
| 201141111111111111111111111111111111111 | | | \$ | \$0.00 | \$ |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Work Order 22, December 6, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL. | Work Order No. 19 to the Master Services | | | |
| Lonafarnib | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib- | RRD INTERNATIONAL, | Work Order No. 20 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | 550 | Agreement dated Waren 13, 2013 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Work Order No. 22 to the Master Services | | | |
| Lonafarnib- | LLC | Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Avexitide | LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T amala da | RRD INTERNATIONAL, | First Amoundurant to Ducinet Amound | \$0.00 | \$0.00 | \$0.00 |
| Lambda | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Letter for Authorization PA 3 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Project Agreement 3 to the Product | | | |
| Lambda | LLC | Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | , , , , , | , , , , | |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Second Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 15, February 28, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNIATIONAL | | | | |
| Lambda | RRD INTERNATIONAL, LLC | Work Order 15, January 25, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lumouu | | 511 51461 15, validary 25, 2021 | \$0.00 | \$0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 16, December 9, 2022 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------|--|----------------|-------------|--------------------|
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 16, March 1, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order 16, March 11, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order 8, effective August 3, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order 9, effective September 5, 2017 | \$0.00 | | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lambda | LLC | Work Order 9, effective September 5, 2017 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 10 provided for Strategic Regulatory Consultation and FDA Liaison Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 15 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 16 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 17 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 18 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 8 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | RRD INTERNATIONAL, | Work Order No. 9 to the Master Services Agreement dated March 15, 2015 | \$0.00 | | \$0.00 |
| Lambda | RRD INTERNATIONAL, LLC | Work Order No. 9 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Amended and Restated Work Order No. 11 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 1 to 1 20191210 | \$0.00 | | \$0.00 |

| Related | | | | | Lonafarnib | |
|------------|------------------------------|--|----------------|-------------|----------------|--|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 1 to 2 20191002 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 1 to 3 20191022 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 1 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 2 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 2 to PA 1 - Interim CO 3 | \$0.00 | \$0.00 | \$0.00 | |
| | DDD DITEDMATIONAL | | | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to PA 1 - Interim CO 4 | \$0.00 | \$0.00 | \$0.00 | |
| | | | | · | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 2 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| | | | | ***** | Q 0.100 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 2 to TO 2 202003003 | \$0.00 | \$0.00 | \$0.00 | |
| Lonaramo | | Time trainers 2 to 10 2 202003003 | ψ0.00 | \$0.00 | ψ0.00 | |
| Lonafarnib | RRD INTERNATIONAL, | Amendment 2 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| Lonararino | EDC | Amendment 2 to 10 3 20200303 | ψ0.00 | ψ0:00 | φυ.σσ | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amendment 3 to PA 1 - Interim CO 5 | \$0.00 | \$0.00 | \$0.00 | |
| Lonaranno | LLC | Amendment 3 to FA 1 - Internit CO 3 | \$0.00 | \$0.00 | \$0.00 | |
| T | RRD INTERNATIONAL, | Amondon and 2 to DA 1. Interior CO 6 | 00.00 | \$0.00 | \$0.00 | |
| Lonafarnib | LLC | Amendment 3 to PA 1 - Interim CO 6 | \$0.00 | \$0.00 | \$0.00 | |
| T C '1. | RRD INTERNATIONAL, | A manufacture 2 to Day in the American 1 | ¢0.00 | ¢0.00 | ¢0.00 | |
| Lonafarnib | LLC | Amendment 3 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 | |
| T C 1 | RRD INTERNATIONAL, | Amendment 3 to Project Agreement 1 | фо оо | Φ0.00 | ФО ОО | |
| Lonafarnib | LLC | Interim CO 4 | \$0.00 | \$0.00 | \$0.00 | |
| - 0 " | RRD INTERNATIONAL, | | 40.00 | | | |
| Lonafarnib | LLC | Amendment 3 to TO 1 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 3 to TO 2 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Amendment 3 to TO 3 20200303 | \$0.00 | \$0.00 | \$0.00 | |
| | RRD INTERNATIONAL, | | | | | |
| Lonafarnib | LLC | Contractor Agreement | \$0.00 | \$0.00 | \$0.00 | |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---------------------------|--|----------------|-------------|--------------------|
| | | Development Advisory Committee | | | |
| - 2 " | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Amendment 2 to Project Agreement 1 Development Advisory Committee | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Development Advisory Committee | | | |
| Lonafarnib | LLC | Amendment 2 to Project Agreement 1 | \$0.00 | \$0.00 | \$0.00 |
| | | Eiger Progeria PA2 Amendment 1 | \$0.00 | φοισσ | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Eiger Progeria PA2 Amendment 1 Interim | | | |
| | DDD DITEDNIATIONAL | CO2 | | | |
| Lonafarnib | RRD INTERNATIONAL, | Figur Drainet Agreement | \$0.00 | 90.00 | \$0.00 |
| Lonalarnio | LLC | Eiger Project Agreement Eiger Progeria PA2 Amendment 1 Interim | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | Eiger Progeria PA2 Amendment 2 Interim | | | |
| Lonafarnib | LLC | CO1 Eiger Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDMATIONAL | | | | |
| T | | Eighth Amendment to Task Order 1 to the | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Fifth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | 3 6 | | | |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDMATIONAL | | | | |
| T | RRD INTERNATIONAL, | First Amondment to Desirat Assessment | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | First Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL. | First Amendment to Task Order 1 to the | | | |
| Lonafarnib | | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDNIATIONAL | | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | | \$0.00 | 90.00 | \$0.00 |
| Lonarariio | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | First Amendment to Task Order 1 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | First Amendment to Task Order 10 to the | | | |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITERMATION A | | | | |
| I anafamil | • | First Amendment to Task Order 2 to the | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|------------------------------|---|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Task Order 2 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | First Amendment to Task Order 5 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | First Amendment to Task Order 6 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | First Amendment to Task Order 8 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | First Amendment to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | First Amendment to Work Order No. 12 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Work Order No. 13 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | First Amendment to Work Order No. 14 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Fourth Amendment to Project Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Fourth Amendment to Task Order 1 to the Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Fourth Amendment to the Amended and Restated Work Order | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 1 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 1 to PA 2 (MM TO 12) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 2 to PA 2 (MM TO 14) | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 3 to PA 1 Amendment 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 3 to PA 2 (MM TO 22) | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|------------|---------------------------|--|----------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | RRD INTERNATIONAL, | Interim CO 4 to PA 2_20200701 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Letter for Authorization for PA 2 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Letter for Authorization for RRD to Commence Activities Under Work Order 11 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Master Independent Contractor Agreement effective December 5, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective February 22, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective March 1, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Master Independent Contractor Agreement effective October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Product Development Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | | Project Agreement 1 Amendment 2 ("Project Agreement 1 Amendment 2") | \$0.00 | | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | | \$0.00 | | \$0.00 |
| Lonafarnib | | Second Amendment to the Amended and Restated Work Order | \$0.00 | | |

| | | T | | | |
|------------------|-----------------------|--|----------------|--------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | L amhda Cure | Lonafarnib Cure |
| Tisset | Contract Counterparty | Description of Contract of Lease | Avealude Cure | Lambua Curc | Curc |
| | RRD INTERNATIONAL, | Second Amendment to Work Order No. 12 | | | |
| Lonafarnib | LLC | to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 11 | RRD INTERNATIONAL, | | ФО ОО | Φ0.00 | ФО ОО |
| Lonafarnib | LLC | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Statement of Work 1 to the Master | | | |
| Lonafarnib | LLC | Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| 20114111110 | | Task Order 1 to the Master Independent | \$0.00 | Ψ0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 1 to the Master Independent | | | |
| | RRD INTERNATIONAL, | e , | | | |
| Lonafarnib | LLC | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDNIATIONAL | Task Order 1 to the Master Independent | | | |
| T C 1 | RRD INTERNATIONAL, | Contractor Agreement dated March 11, | #0.00 | #0.00 | фо о о |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDMATIONAL | Task Order 1 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | LLC | Task Order 1 to the Master Independent | ψ0.00 | ψ0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 10 to the Master Independent | · | | · |
| | RRD INTERNATIONAL, | Contractor Agreement dated effective | | | |
| Lonafarnib | LLC | October 29, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 11 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITEDNIATIONAL | Task Order 12 to the Master Independent | | | |
| T C 1 | RRD INTERNATIONAL, | , | #0.00 | #0.00 | фо о о |
| Lonafarnib | LLC | 2018 Task Order 13 to the Master Independent | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | • | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | LLC | Task Order 2 to the Master Independent | ψ0.00 | ψ0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated December 5, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 2 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated February 22, | | | |
| Lonafarnib | LLC | 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| L | RRD INTERNATIONAL, | Task Order 2 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD DITERNATION A | Task Order 2 to the Master Independent | | | |
| Lanafamil | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | \$0.00 | ¢0.00 | \$0.00 |
| Lonafarnib | LLC | 2018 Task Order 3 to the Master Independent | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | <u> </u> | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| -Onaraniii0 | LLLC | 2010 | ψ0.00 | φυ.υυ | ψ0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|---|---------------------------|--|----------------|-------------|--------------------|
| | | | | | |
| L | RRD INTERNATIONAL, | * | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNIATIONIAI | Task Order 3 to the Master Independent | | | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Contractor Agreement dated October 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Task Order 4 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| 201141111111111111111111111111111111111 | | Task Order 4 to the Master Independent | \$0.00 | φοισσ | \$0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | Task Order 5 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 5 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| T C 1 | RRD INTERNATIONAL, | • | Ф0.00 | ФО ОО | # 0.00 |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNIATIONIAI | Task Order 6 to the Master Independent | | | |
| Lonafarnib | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | \$0.00 | \$0.00 | \$0.00 |
| Lonatarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Task Order 7 to the Master Independent | | | |
| Lonafarnib | LLC | Contractor Agreement dated May 24, 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaramio | LLC | Task Order 7 to the Master Independent | ψ0.00 | Ψ0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Zonaramo | EEC | Task Order 8 to the Master Independent | ψ0.00 | ψ0.00 | ψο.σσ |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | Task Order 9 to the Master Independent | | | |
| | RRD INTERNATIONAL, | Contractor Agreement dated October 24, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | ş ç | | | |
| Lonafarnib | LLC | dated March 1, 2020 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | RRD INTERNATIONAL, | Third Amendment to Project Agreement | #0.00 | фо. оо | фо. о о |
| Lonafarnib | LLC | dated March 1, 2020 Third Amendment to Task Order 1 to the | \$0.00 | \$0.00 | \$0.00 |
| | DDD INTEDNIATIONIAI | | | | |
| r | RRD INTERNATIONAL, | Master Independent Contractor Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | LLC | dated March 1, 2018 Third Amendment to the Amended and | \$0.00 | \$0.00 | \$0.00 |
| | RRD INTERNATIONAL, | Restated Work Order dated February 15, | | | |
| Lonafarnib | LLC | 2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | | 2010 | ψ0.00 | ψ0.00 | ψ0.00 |
| | RRD INTERNATIONAL, | | | | |
| Lonafarnib | LLC | Work Order 1, March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|---|--|----------------|-------------|--------------------|
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 21, May 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 5, effective March 31, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order 5, effective March 31, 2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, | Work Order No. 10 provided for Strategic Regulatory Consultation and FDA Liaison Activities for the dated February 19,2018 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 12 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 13 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 14 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 2 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 21 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 24 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 4 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Work Order No. 5 to the Master Services Agreement dated March 15, 2015 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RRD International, LLC; AltaSciences | Quotation Description | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | RTI INTERNATIONAL | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | SAFEBRIDGE CONSULTANTS, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |

| ease Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|---|---|---|
| ease Avexitide Cure | Lambda Cure | |
| Aveature Cure | Lambua Cure | |
| | | Cure |
| | | |
| Ф0.00 | ФО ОО | Ф0.00 |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| | | |
| \$0.00 | | |
| \$0.00 | \$0.00 | \$0.00 |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | 20.00 | \$0.00 |
| ₩₩₩ | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| ' | | |
| et To Reflect | | |
| ngements on | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| 00.02 | 00.02 | \$0.00 |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| 40.00 | | |
| \$0.00 | \$0.00 | \$0.00 |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| | | |
| \$0.00 | \$0.00 | \$0.00 |
| 1 | \$0.00 \$0.00 \$0.00 \$0.00 Let To Reflect- ngements on \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 \$0.00 | \$0.00 |

| | 1 | | | | |
|------------------|--|---|----------------|-------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| | CECDETA DV OF | | | | |
| Lonafarnib | SECRETARY OF VETERAN AFFAIRS | Interim Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | VETERATIVATIANS | mermi Agreement | ψ0.00 | ψ0.00 | \$0.00 |
| | SECRETARY OF | | | | |
| Lonafarnib | VETERAN AFFAIRS | Master Agreement | \$0.00 | \$0.00 | \$0.00 |
| | SENTYNL | | | | |
| Lonafarnib | THERAPEUTICS, INC. | Term Sheet | \$0.00 | \$0.00 | \$0.00 |
| | SGS Analytics Germany | | | | |
| Lonafarnib | GmbH | Quotation No. 210409-01 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | SGS Analytics Germany GmbH | Quotation No. 210514-01 | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | SGS Analytics Germany | Quotation ivo. 210314-01 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | GmbH | Quotation No. 210602-01 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | Shaare Zedek Medical Center; Dr. Lurie Yoav | Form 4a | \$0.00 | \$0.00 | \$0.00 |
| Laiiibua | Center, Dr. Lurie 10av | 1 01111 4a | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Sharp Corporation | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | 1.00 | | | 40.00 |
| Lambda | Sharp Corporation | Quote dated 3/3/22 | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote dated 8/11/21 | \$0.00 | \$0.00 | \$0.00 |
| | | | | * | |
| Lambda | Sharp Corporation | Quote re: Concept | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Sharp Corporation | Quote re: Groninger Tooling | \$0.00 | \$0.00 | \$0.00 |
| Laiiioda | Sharp Global Design | Quote re. Gronniger rooming | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Center | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Sheba Fund for Health Services and Research | | | | |
| Lambda | (R.A.) | Clinical Study Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Sheba Fund for Health Services and Research | Letter of Indemnification re: Protocol No. | | | |
| Lonafarnib | Ltd. | EIG-LNF-011 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Sherpa | Change Order dated 12/5/2016 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Sherpa | Change Order dated 12/12/2016 | | | |
| Avexitide | Slate360, Inc. | Statement of Work re: ENDO 2022 - Graphics Refresh | \$0.00 | \$0.00 | \$0.00 |
| AVEAILIGE | State 300, IIIc. | Quote N21-11679 re: Screening and | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Solvias AG | Quantitation of Nitrosamine DIPNA | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Solvias AG | Quote N21-11680 re: Lonafarnib API | \$0.00 | · | |
| Lonafarnib | Solvias AG | Quote N21-11681 re: Lonafarnib API | \$0.00 | | |
| Lonafarnib | Solvias AG | Quote N21-12606 re: Lonafarnib API | \$0.00 | \$0.00 | \$0.00 |
| | Soroka Medical Center; | | | | |
| Lambda | Dr. Etzion Ohad | Form 4a | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|----------------|---------------------------------------|--|-----------------------|-------------|------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Soroka University | | | | |
| Lambda | Medical Center | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| T C 11 | C 4 1D 1 | D 1 | ФО ОО | ФО ОО | ФО ОО |
| Lonafarnib | Southwood Research | Proposal | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lambda | - GLOBAL, LLC | Amendment #1 to Work Order 1 LMD001 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lambda | - GLOBAL, LLC | Amendment #3 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lambda | - GLOBAL, LLC | Amendment 2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lambda | - GLOBAL, LLC | Amendment 5 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lambda | - GLOBAL, LLC | Amendment to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lambda | SPRI CLINICAL TRAILS - GLOBAL, LLC | Statement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | | | *** | , , , , , | |
| Lonafarnib | SPRI CLINICAL TRAILS - GLOBAL, LLC | Amendment #1 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonararmo | - GEODIAL, ELEC | Amendment #1 to work order 1 | ψ0.00 | Ψ0.00 | ψ0.00 |
| T C 11 | SPRI CLINICAL TRAILS | | #0.00 | #0.00 | #0.00 |
| Lonafarnib | - GLOBAL, LLC | Amendment #1 to Work Order 5 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lonafarnib | - GLOBAL, LLC | Amendment #2 to Work Order 1 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lonafarnib | - GLOBAL, LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lonafarnib | - GLOBAL, LLC | Statement of Work Dated August 24, 2015 | \$0.00 | \$0.00 | \$0.00 |
| | SPRI CLINICAL TRAILS | | | | |
| Lonafarnib | - GLOBAL, LLC | Statement of Work LNF004 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | Stanford University | | | | |
| A committee of | Office of Technology | Samilar A many sut | Ø1 11C 00 | Φ0.00 | Φ0.00 |
| Avexitide | Licensing | Services Agreement Quote re: SG-Lab Mechanical or | \$1,116.00 | \$0.00 | \$0.00 |
| Avexitide | Stevanato Group S.p.A. | Performance Test | \$0.00 | | |
| Lambda | Syneos | Statement of Work Dated August 7, 2018 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | 6 | | Lonafarnib |
|------------|------------------------------|---|-------------------|---------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | Syneos Health | | | | |
| | Communications, Inc. | | | | |
| | f/k/a inVentiv Health | | | | |
| Lonafarnib | Public Relations, LLC | Statement of Work re: Submission | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| Lonafarnib | The Aga Khan University | Data Transfer Agreement V | \$0.00 | \$0.00 | \$0.00 |
| T | The Ass When Heisensites | Letter re: Change to the Data Transfer | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Aga Khan University | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | Asset Purchase Agreement by and between | | | |
| | | Colleen Craig and Tracey McLaughlin (the | | | |
| | | "Original Licensees"), on the one hand, and | | | |
| | | Eiger BioPharmaceuticals, Inc. ("Eiger"), | | | |
| | | dated 25 September 2015 (the "APA"), by | | | |
| | | which Eiger became party to that certain | | | |
| | | License Agreement by and between Colleen | | | |
| | The Board of Trustees of | Craig and Tracey McLaughlin on the one | | | |
| | the Leland Stanford Junior | hand and The Board of Trustees of the | | | |
| | University; Colleen Craig; | Leland Stanford Junior University | | | |
| Avexitide | Tracey McLaughlin | ("Stanford"), dated May 4, 2015 | | <u>\$0.00</u> | <u>\$0.00</u> |
| | | The Amendment to the APA License | | | |
| | | Agreement among Stanford, the Original | | | |
| | | Licensees, and Eiger dated as of 25 | \$1,116.00 | | |
| | The Board of Trustees of | September 2015 ("Amendment No. 1") (this | | | |
| | the Leland Stanford Junior | License Agreement Amendment No. 1 was | | | |
| A141 1 - | University; Colleen Craig; | executed together with the APA as Exhibit | | \$0.00 | ¢0.00 |
| Avexitide | Tracey McLaughlin | C to the APA) | - | \$0.00 | <u>\$0.00</u> |
| | The Board of Trustees of | The Second Amendment to the License | | | |
| | | Agreement by and among Stanford, the | | | |
| | University; Colleen Craig; | | | | |
| Avexitide | Tracey McLaughlin | May 11, 2018 ("Amendment No. 2") | | \$0.00 | \$0.00 |
| | | The Second Amendment to the APA | 1 | | |
| | | between the Original Licensees and Eiger, | | | |
| | The Board of Trustees of | dated 22 May, 2018 (the "APA Amendment | | | |
| | the Leland Stanford Junior | 2" and, together with the APA, Amendment | | | |
| | University; Colleen Craig; | No. 1, and Amendment No. 2, the | | | |
| Avexitide | Tracey McLaughlin | "Agreements") | | \$0.00 | \$0.00 |
| | | | | | |
| | THE CHILDRENS | | | | |
| | HOSPITAL | | | | |
| | CORPORATION, D/B/A | | | | |
| | BOSTON CHILDRENS | | | | |
| Lonafarnib | HOSPITAL | Data Sharing Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | \$5.00 | \$5.00 | \$0.00 |
| | The Children's Hospital of | | | | |
| Avexitide | Philadelphia | License Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|---|----------------|-------------|--------------------|
| Avexitide | THE CHILDRENS HOSPITAL OF PHILADELPHIA | Indemnification Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 1 NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 1 PK | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 2 IR | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 2 NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Change Order 2 PK | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Pricing Agreement NZ | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Doctor's Laboratory Limited | Pricing Agreement PK | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing / Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 10th September 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 14 September 2018 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 14 September 2018 | \$0.00 | \$0.00 | \$10,000.00 |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 18th January 2017 | \$0.00 | \$0.00 | |

| Related | | | | | Lonafarnib |
|-----------------------|--|---|-------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 18th January 2017 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated 2021 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement dated December 2014 | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Pricing/Supplemental Agreement Trials | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Trials Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Doctor's Laboratory Limited | Trials Pricing/Supplemental Agreement | \$0.00 | \$0.00 | |
| Lonafarnib | The Johns Hopkins University | Sponsored Research Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Liver Center and Naranjargal Dashdorj | ADDITIONAL BUDGET AGREEMENT | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Liver Unit Hadassah University Hospital, Ein Kerem | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| Lambda | The Liver Unit Hadassah University Hospital; The Research Fund of Hadassah Medical Organization (R.A.) | Letter of Indemnification | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Progeria Research Foundation | Agreement to Stay Arbitration | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Progeria Research Foundation | Amended and Restated Collaboration and Supply Agreement | \$0.00 \$0.00 | \$0.00 \$0.00 | \$0.00 \$0.00 |
| Lonafarnib | The Progeria Research Foundation | Amendment No. 1 to the Collaboration and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | The Progeria Research Foundation | Collaboration and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |

| | | | _ | | |
|-----------------------|---|--|-------------------|-------------------|--------------------|
| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
| | | · | | | |
| - 0 " | The Progeria Research | | | | |
| Lonafarnib | Foundation | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| | The Progeria Research | | | | |
| Lonafarnib- | Foundation | CTS Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | The Progeria Research | | | | |
| Lonafarnib | Foundation | Tern Sheet | \$0.00 | \$0.00 | \$0.00 |
| | The Progeria Research | | | | |
| Lonafarnib- | Foundation, Inc. | Collaboration and Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaramo | r ouridation, me. | Condoration and Supply Figreement | ψ0.00 | ψ0.00 | ψ0.00 |
| | The Progeria Research | | | | |
| Lambda | Foundation | Consent Letter | \$0.00 | \$0.00 | \$0.00 |
| | The University of Texas | | | | |
| T | MD Anderson Cancer | Days Consults A sussessed | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Center | Drug Supply Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific | Quotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| | Thermo Fisher Scientific, | Character and the second secon | 7,000 | 40.00 | 7,110 |
| Lonafarnib | Inc. | Quote re: Sourcing of Medicinal Product | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Thermo Fisher Scientific, Inc.; Fisher Clinical Services | Quote | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #200616-01-SF | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #200901-01-SF | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #220328-01-SF | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Total Transport Management | Netherlands Hub Freight Quote dated 06 April 2022 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Total Transport | Netherlands Hub Freight Quote dated 15 | φυ.υυ | \$0.00 | \$0.00 |
| Lambda | Management | March 2022 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated | | | |
| Lambda | Management | 4/10/2022 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|-----------------------|------------------------------|--|-------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | Total Transport | Netherlands Hub Freight Quote dated Dec | Ф0.00 | #0.00 | фо. о о |
| Lambda | Management | 13, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | Total Transport | Netherlands Hub Freight Quote dated Feb | Ф0.00 | # 0.00 | ФО ОО |
| Lambda | Management Total Transport | 14, 2022 Netherlands Hub Freight Quote dated Jan | \$0.00 | \$0.00 | \$0.00 |
| T amalada | _ | Ę į | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Management Total Transport | 27, 2022 Netherlands Hub Freight Quote dated June | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lambua | Total Transport | Netherlands Hub Freight Quote dated June | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Management | 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lamoua | Total Transport | Netherlands Hub Freight Quote dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Management | August 18, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonararino | Total Transport | Netherlands Hub Freight Quote dated Nov | Ψ0.00 | \$0.00 | φ0.00 |
| Lonafarnib | Management | 2, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Londramo | TransPerfect International | 2, 2021 | ψ0.00 | ψο.σσ | φ0.00 |
| Lonafarnib- | LLC | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | TransPerfect International | | , | * | * |
| Lonafarnib | LLC | Statement of Work dated August 12, 2021 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | TransPerfect Translations | | | | |
| Lonafarnib- | International LLC | Statement of Work dated June 7, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | TransPerfect Translations | | | | |
| Lonafarnib- | International LLC | Statement of Work dated October 29, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | TRANSPERFECT- | | | | |
| | TRANSLATIONS | Deliverables 1 PDF File1 Certificate of | | | |
| Lonafarnib | INTERNATIONAL, INC. | <u>Translation</u> | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | TD ANGDEDERGT | | | | |
| | TRANSPERFECT | | | | |
| T 1.1 | TRANSLATIONS | | #0.00 | #0.00 | #0.00 |
| Lambda | INTERNATIONAL, INC. | Solutions Agreement | \$0.00 | \$0.00 | \$0.00 |
| | TRG Communications | | | | |
| Lonafarnib | LLC | Services Agreement | \$0.00 | \$0.00 | \$36,000.00 |
| Lonaranno | LLC | Services Agreement | \$0.00 | \$0.00 | \$30,000.00 |
| Lambda | Trialog Clinical Trials I td | Price Quotation dated 20-Jul-22 | \$0.00 | | \$0.00 |
| Lamoda | Thatog Chinear Thats Eta | Trice Quotation dated 20-3ui-22 | φ0.00 | | \$0.00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 20-Jul-22 | \$0.00 | | \$0.00 |
| Lamoua | Thatog chinear that Eta | Theo Quotation dated 20 var 22 | ψο.σσ | | ψ0.00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 21-Feb-23 | \$0.00 | | \$0.00 |
| | <i>5</i> 2 | | 72.00 | \$3,760.00 | 73,00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 3-Mar-22 | \$0.00 | | \$0.00 |
| | | | | | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation dated 7-Dec-22 | \$0.00 | | \$0.00 |
| | - | | | | |
| | | | | | |
| Lambda | Trialog Clinical Trials Ltd | Statement of Work dated April 2020 | \$0.00 | | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|------------------|--|---|----------------|-------------|--------------------|
| Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated 2019 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Statement of Work dated June 7, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote 1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote 2 Trialog Clinical Trials Quote dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | 11/30/2021 #1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote dated 11/30/2021 #3 Trialog Clinical Trials Quote dated | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | 7/31/2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Trialog Clinical Trials Quote dated June 27, 2021 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Triangle Biostatistics, | Work Order: EIG-300 Under Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Triangle Insights | Services Agreement | \$0.00 | | \$0.00 |
| Avexitide | Triangle Insights Group LLC | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Triangle Insights Group, LLC | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | TriangleInsights TRILIGENT | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| Lambda | INTERNATIONAL TRILIGENT | Project Assignment 5-1 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | INTERNATIONAL | Consulting Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | TRILIGENT INTERNATIONAL | Project Assignment 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | TRILIGENT INTERNATIONAL | Project Assignment 5 | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Trustees of the University of Pennsylvania ("Penn") and the Children's Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |

| Related | | | | | Lonafarnib |
|--------------------------|-----------------------------------|--|-----------------------|-------------------|-------------------|
| Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Cure |
| | | | | | |
| | | | | | |
| | Trustees of the University | | | | |
| | of Pennsylvania ("Penn") | | | | |
| | and the Children's | | | | |
| Lambda | Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | | | | |
| | | | | | |
| | Trustees of the University | | | | |
| | of Pennsylvania ("Penn") | | | | |
| | and the Children's | | | | |
| Lonafarnib | Hospital of Philadelphia | License Agreement dated as of May 10, 2019 | \$0.00 | \$0.00 | \$0.00 |
| | | Statement of Work 1 to the Master | | | |
| Lonafarnib | TWO LABS, LLC | Independent Contractor Agreement | \$0.00 | 4 | \$0.00 |
| Lonafarnib | TWO LABS, LLC | Statement of Work 2 | \$0.00 | | \$0.00 |
| Lonafarnib | TWO LABS, LLC | Statement of Work 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | TWO LABS, LLC | Statement of Work 4 | \$0.00 | | \$0.00 |
| Lambda | Tx Evidence, Inc | SAP Statistical Analysis Plan 3 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib Lonafarnib | Tx Evidence, Inc | Master Service Agreement | \$0.00 | \$0.00 \$0.00 | \$0.00 |
| Lonafarnib | Tx Evidence, Inc Tx Evidence, Inc | SAP Statistical Analysis Plan 1 SAP Statistical Analysis Plan 2 | \$0.00 \$0.00 | | \$0.00 \$0.00 |
| Lonafarnib | Tx Evidence, Inc | Term of Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | TX Evidence, inc | Eiger Study EIG-LMD-001 Submission | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Tx Evidence, Inc. | New Work Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lambda | Tx Evidence, Inc. | Statical Data Analysis Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lumouu | TX Evidence, me. | Eiger Study EIG-300 Submission New Work | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | Tx Evidence, Inc. | Proposal | \$0.00 | \$0.00 | \$0.00 |
| Lonaranno | United Biosource LLC | Toposui | ψ0.00 | ψ0.00 | ψ0.00 |
| Lonafarnib | (UBC) | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | University Health | The state of the s | φοισσ | \$0.00 | \$0.00 |
| Lambda | Network | Data Use Agreement | \$0.00 | \$0.00 | \$0.00 |
| | | | · | | · |
| | | | | | |
| | University Medical Center | | | | |
| | of Johannes Gutenberg- | | | | |
| Lambda | University Mainz | Clinical Trial Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | University of Oxford | Notice of Assignment | \$0.00 | \$0.00 | \$0.00 |
| | | | | | |
| | | L | | | |
| L | University of Texas MD | First Amendment to Drug Supply | | | |
| Lonafarnib | Anderson Cancer Center | Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Ha Broarby are | T | | | |
| T C " | US BIOSERVICES | First Amendment to Statement of Work No. | #0.00 | 40.00 | #0.00 |
| Lonafarnib | CORPORATION | 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | HG DIOGEDANGEG | | | | |
| Lonofomil | US BIOSERVICES CORPORATION | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib- | CURTURA HUN | iviasici Services Agreement | \$0.00 | \$0.00 | \$0.00 |

| Related Asset (| Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|--------------------|---|--|-------------------|-------------------|--------------------|
|] | US BIOSERVICES | Second Amendment to Statement of Work | | | |
| | CORPORATION | No. 1 to the Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| 1 | US BIOSERVICES | | | | |
| | CORPORATION | Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | US BIOSERVICES CORPORATION | Specialty Pharmacy Provider Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Vince & Associates | Clinical Trial Agreement, dated October 4, 2016, by and among Seller, Vince & Associates Clinical Research, P.A., and Vince & Associates Clinical Research, Inc. | | | |
| Avexitide | Clinical Research | (the "Vince CTA"). | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | Vito Vento | Consent for Photography / Authorization for Publication | \$0.00 | \$0.00 | \$0.00 |
| | Walgreens Boots Alliance | Global Estimated Quotation | \$0.00 | \$0.00 | \$0.00 |
| l l | Washington University in St Louis | Notice of Assignment | <u>\$0.00</u> | <u>\$0.00</u> | <u>\$0.00</u> |
| Lonafarnib | WCCT GLOBAL, INC. | Letter of Intent | \$0.00 | \$0.00 | \$0.00 |
| Avexitide | WCG Clinical Inc. | Services Agreement | \$0.00 | \$0.00 | \$1,312.50 |
| Lonafarnib | WOVEN DATA, INC. | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | WOVEN DATA, INC. | Statement of Work | \$0.00 | \$0.00 | \$0.00 |
| | WuXi AppTee | Quotation | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib (| WUXI APPTEC (HONGKONG) LTD: WuXi | Master Services Agreement | \$0.00 | \$0.00 | \$0.00 |
| | AppTec(HongKong)Limit | Quotation #1 | \$0.00 | \$0.00 | \$0.00 |
| 2 | WuXi AppTec(HongKong)Limit | Quounon #1 | \$0.00 | | φυ.υυ |
| Lonafarnib | ed Xcelience | Quotation #2 | \$0.00 \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib 1 | Acenence | Service Agreement | \$0.00 | \$0.00 | \$0.00 |
| | Xenotech LLC dba | | | | |
| | Sekisui Xenotech, LLC Yuki Gosei Kogyo Co., | Master Services Agreement Price Quotation and Trade Terms dated | \$0.00 | \$0.00 | \$0.00 |
| | Yuki Gosei Kogyo Co., Ltd. | December 10, 2018 | \$0.00 | \$0.00 | \$0.00 |
| | Yuki Gosei Kogyo Co., | Price Quotation and Trade Terms dated July- | | | |
| | Ltd. Yuki Gosei Kogyo Co., | 2, 2021 Price Quotation and Trade Terms dated June | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib l | Ltd. | 2, 2022 | \$0.00 | \$0.00 | \$0.00 |
| | Yuki Gosei Kogyo Co., Ltd. | Price Quotation and Trade Terms dated May 23, 2018 | \$0.00 | \$0.00 | \$0.00 |

| Related Asset | Contract Counterparty | Description of Contract of Lease | Avexitide Cure | Lambda Cure | Lonafarnib Cure |
|-----------------------|--------------------------------|---|-------------------|-------------------|--------------------|
| Lonafarnib | YUKI GOSEI KOGYO CO., LTD. | Yuki Gosei 1st Amnd to SOW 103-366-20230407 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | YUKI GOSEI KOGYO CO., LTD. | Yuki Gosei SOW 102-054 20210702 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | YUKI GOSEI KOGYO CO., LTD. | Yuki Gosei SOW 103-093-20220530 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | YUKI GOSEI KOGYO- CO., LTD. | Yuki Gosei SOW 103-101-2022061 4 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | YUKI GOSEI KOGYO- CO., LTD. | Yuki Gosei SOW 103-295-20221122 | \$0.00 | \$0.00 | \$0.00 |
| Lonafarnib | Zokinvy Isi Development | Statement of Work | \$0.00 | \$0.00 | \$0.00 |

Exhibit 5

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al., 1

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF CANCELLATION OF AUCTION(S), DESIGNATION OF WINNING BID FOR THE LONAFARNIB SALE TRANSACTION, AND TRANSITION TO PRIVATE SALE PROCESS FOR LONAFARNIB/LAMBDA SALE TRANSACTIONS

PLEASE TAKE NOTICE that, on April 5, 2024 the Court entered the *Order* (*I*)(*A*) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if Any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order"),² which, among other things, establishes key dates and deadlines related to the Auction for, and the Sale of, the Assets.

PLEASE TAKE FURTHER NOTICE that, on April 8, 2024, and June 3, 2024, the Debtors served the *Notice of Sale, Bid Procedures, Auction, and Sale Hearing* on all known parties in interest. *See* Docket Nos. 128, 320.

PLEASE TAKE FURTHER NOTICE that, on April 15, 2024, the Debtors filed the *Notice of Filing of Revised Bidding Procedures* [Docket No. 119], which included the revised bidding procedures (the "<u>Bid Procedures</u>") attached as <u>Exhibit A</u>.

PLEASE TAKE FURTHER NOTICE that, on June 12, 2024, the Debtors filed and served the *Revised Notice of Sale, Bid Procedures, Auction, and Sale Hearing* [Docket No. 331] on all known parties in interest. *See* Docket Nos. 374, 431.

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

² Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Bid Procedures Motion, the Bid Procedures, and the Bid Procedures Order, as applicable.

- **PLEASE TAKE FURTHER NOTICE** that, on July 13, 2024, Debtors filed the *Further Revised Notice of Bid Deadlines* [Docket No. 422], which included revised dates and deadlines related to the Bid Deadline for the Lonafarnib sale transaction (the "Lonafarnib Sale Transaction") and the Lambda sale transaction (the "Lambda Sale Transaction").
- **PLEASE TAKE FURTHER NOTICE** that, as of July 19, 2024 at 4:00 p.m. (prevailing Central Time), which was the Bid Deadline for the Lonafarnib Sale Transaction, the Debtors only received one Qualified Bid for the Lonafarnib Sale Transaction from Eiger InnoTherapeutics, Inc. ("Inno").
- **PLEASE TAKE FURTHER NOTICE** that, in accordance with the Bid Procedures the Debtors have <u>cancelled</u> the Auction and hereby designate Inno as the highest and best bid with a Base Price in the amount of \$5,200,000 (the "Winning Lonafarnib Bid").
- **PLEASE TAKE FURTHER NOTICE** that Inno seeks to purchase the Lonafarnib Assets free and clear of liens, claims, encumbrances, and other interests on the terms set forth in the Asset Purchase Agreement by and between Inno and Eiger BioPharmaceuticals, Inc. (the "<u>Lonafarnib APA</u>").
- **PLEASE TAKE FURTHER NOTICE** that, the Debtors have designated Inno as the highest and best bid with a Base Price in the amount of \$1,000,000 (the "Winning Lambda Bid").
- **PLEASE TAKE FURTHER NOTICE** that the Debtors have <u>cancelled</u> the Auction for the Lambda Sale Transaction and, on August 1, 2024, executed an asset purchase agreement by and between Inno and Eiger BioPharmaceuticals, Inc. for the Lambda Sale Transaction (the "<u>Lambda APA</u>").
- PLEASE TAKE FURTHER NOTICE that, under the Lonafarnib APA and the Lambda APA, Inno has agreed to pay certain cure costs in an amount up to \$2,650,000 in the aggregate if certain conditions are met.
- **PLEASE TAKE FURTHER NOTICE** that Inno seeks to purchase the Lambda Assets free and clear of liens, claims, encumbrances, and other interests on the terms set forth in the Lambda APA.
- **PLEASE TAKE FURTHER NOTICE** that the Debtors will file a motion seeking approval of the Lonafarnib Sale Transaction and the Lambda Sale Transaction (the "Lonafarnib/Lambda Sale Motion"). The Debtors will seek approval of the Lonafarnib Sale Transaction, the Lambda Sale Transaction, and the Lonafarnib/Lambda Sale Motion on an expedited basis at a to-be-scheduled hybrid hearing.
- PLEASE TAKE FURTHER NOTICE that attached as <u>Exhibit A</u> and <u>Exhibit B</u> is a list of the Assigned Contracts and applicable Cure Amounts for each respective Assigned Contract that Inno has indicated will be cured by, and assigned to, Inno pursuant to the Lonafarnib Sale Transaction and the Lambda Sale Transaction, respectively. As more fully described in the Lonafarnib APA and the Lambda APA, Inno retains the right to modify the list of Assigned Contracts prior to Closing. The Debtors hereby certify that the Debtors will provide, in

Casse2248800409sgijj111 DDooc84895 Filitelc084025225 Hintereclc08402522511913497305 DDess Mai Extribitorius in en PagePage 20 of 5

coordination with the proposed assignee, Inno's Adequate Assurance Information to each affected Counterparty on a confidential basis.

PLEASE TAKE FURTHER NOTICE that copies of foregoing pleadings, including of the Bid Procedures, are available by: (i) visiting the website of the Debtors' claims, noticing, and solicitation agent, Kurtzman Carson Consultants LLC dba Verita Global ("Verita") at https://www.veritaglobal.net/Eiger, (ii) (888)733-1544 (Toll-Free) or (310 751-2638 (International), and/or (iii) emailing https://www.veritaglobal.net/Eiger/inquiry or (iv) for a fee via PACER at https://ecf.txnb.gov/.

PLEASE TAKE FURTHER NOTICE that you may obtain additional information concerning these Chapter 11 Cases on the Case Website.

[Remainder of page intentionally left blank]

Casse2248800409sgijj11 DDooc84895 Filitelc084025225 Hintereclc08402522511913497305 DDess Mai Extraction in the control of the

Dated: August 2 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Certificate of Service

I certify that on August 2, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ *Thomas R. Califano* Thomas R. Califano

Exhibit A

Lonafarnib Assigned Contracts and Cure Amounts

| Asset | Counterparty | Description of Contract | Cure Amounts |
|------------|---|--|---------------------|
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Change Order 7 to Statement of Work | |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Amendment No. 2 to the Master Services and Clinical Manufacture Agreement | \$0.00 |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Master Services and Clinical Manufacture Agreement | |
| Lonafarnib | BIORASI, LLC | Master Services Agreement, dated June 23, 2020 | |
| Lonafarnib | BIORASI, LLC | Statement of Work #157-1, dated July 10, 2020, as governed by Master Services Agreement, dated June 23, 2020 | |
| Lonafarnib | BIORASI, LLC | Change Order 1 to Statement of Work #157-1 | \$[•]¹ |
| Lonafarnib | BIORASI, LLC | Change Order 2 to Statement of Work #157-1 | |
| Lonafarnib | BIORASI, LLC | Change Order 3 to Statement of Work #157-1 | |
| Lonafarnib | BIORASI, LLC | Change Order 4 to Statement of Work #157-1, dated October 29, 2021 | |
| Lonafarnib | CHARLES RIVER LABORATORIES | 1st Amendment to Statement of Work (SFDC OPP 273814) [20230131] | \$0.00 |
| Lonafarnib | CHARLES RIVER LABORATORIES | 2nd Amendment to Statement of Work (SFDC OPP-273814) | |
| Lonafarnib | CHARLES RIVER LABORATORIES MONTREAL ULC | Master Services Agreement | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 to Statement of Work, dated May 19, 2021 | |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 to Statement of Work, dated November 5, 2019 | \$0.00 |
| Lonafarnib | Corden Pharma Colorado | Statement of Work 6 | |
| Lonafarnib | Corden Pharma Colorado; Corden Pharma International GmbH | Change Order 1 to the Statement of Work 6 | \$0.00 |

¹ The Biorasi, LLC ("<u>Biorasi</u>") contract cure amounts are subject to resolution among the Debtors, Biorasi, and the Purchaser.

| Lonafarnib | Cyprotex US, LLC | Proposal for Analysis of Active Metabolites of Lonafarnib (LNF): MH17 and HM21 | \$0.00 |
|------------|---------------------------------------|--|-------------------|
| Lonafarnib | Fiona McPhee, DPhil | Services Agreement | \$2,612.50 |
| Lonafarnib | Fisher Clinical Services GmbH | Quote 214873 Order 8 Version 3 20220225 | \$0.00 |
| Lonafarnib | Fisher Clinical Services Inc. | Quote PSG-A-1051277.v3 20220225 | \$5.650.50 |
| Lonafarnib | Fisher Clinical Services Inc. | Quote-PSG-A-1072091.v2 20230302 | \$5,673.53 |
| Lonafarnib | Fisher Clinical Services U.K. Limited | Quote PSG-A-1007765.v1 20190514 | \$0.00 |
| Lonafarnib | INTRINSIK CORP | Statement of Work #8, dated July 9, 2022, as governed by Master Services Agreement, dated March 6, 2020 | \$55.00 |
| Lonafarnib | LONZA BEND, INC. | Amendment No. 1 to the Commercial Supply Agreement | |
| Lonafarnib | LONZA BEND, INC. | Amendment No. 2 to the Commercial Supply Agreement | |
| Lonafarnib | LONZA BEND, INC. | Change Order 8 to Statement of Work | \$21,000.00 |
| Lonafarnib | LONZA BEND, INC. | Master Services and Clinical Manufacture Agreement, dated 15 December, 2022 | |
| Lonafarnib | LONZA BEND, INC. | Statement of Work, dated 10 April 2023 | |
| Lonafarnib | Lonza Bend; Patheon Canada | Total Transportation Management ("TTM") Freight Quote | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Change Order #7 to E141- 8598 | |
| Lonafarnib | Lonza Pharma & BioTech | Change Order No.1 to Statement of Work | \$0.00 |
| Lonafarnib | Lonza Pharma & BioTech | Stability Proposal | + 3.00 |
| Lonafarnib | Lonza Pharma & BioTech | Validation Proposal, dated 6 April 2020 | |
| Lonafarnib | Patheon, Inc. | Solely the extent related to the 25mg strength, Change of Scope COS-55-R0 to Proposal No. P-TRP-114750-R2 | |

| Lonafarnib | Patheon, Inc. | Project Proposal, dated July 27, 2019 | |
|------------|---|---|---------------|
| Lonafarnib | Patheon, Inc. | Project Proposal, dated July 27, 2019 | |
| Lonafarnib | Patheon, Inc. | Project Proposal # C-TRC- 270507-R4, dated September 27, 2021 | $[\bullet]^2$ |
| Lonafarnib | Patheon, Inc. | Change of Scope # C-TRC- 270507-R4-COS-01-R0 | |
| Lonafarnib | Patheon, Inc. | Master Manufacturing Services Agreement, dated January 9, 2020 | |
| Lonafarnib | Patheon UK Limited; Fisher Clinical Services | Change of Scope: Proposal COS 12 to C-TRC-121992- R4_20230429 | \$0.00 |
| Lonafarnib | Patheon, Part of Thermo Fischer Scientific; Element Toronto | Element Quote Element_Quote 20-012162900 Rev 1_20200420 | \$0.00 |
| Lonafarnib | Patheon, Part of Thermofisher | Total Transportation Management ("TTM") Freight Quote 453021_20201202 | \$0.00 |
| Lonafarnib | PharmaDirections, Inc | WKO-EIG-879 Ad hoc Consulting | \$2,767.50 |
| Lonafarnib | Q SQUARED SOLUTIONS HOLDINGS, LLC | Work Order, dated October 20, 2023, under that certain Master Laboratory Services Agreement, dated May 3, 2019 | \$0.00 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Project Agreement 1 to the Product Development Agreement, dated July 1, 2018, as governed by Product Development Agreement, dated July 1, 2018 | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Amended and Restated Work Order No. 11 to the Master Services Agreement | \$45,368.75 |
| Lonafarnib | RRD INTERNATIONAL, LLC | Interim CO 1 to PA 1 Amendment 3 | |
| Lonafarnib | RRD INTERNATIONAL, LLC | Product Development Agreement, dated July 1, 2018 | |

_

 $^{^2}$ The Patheon, Inc. ("<u>Patheon</u>") contract cure amounts are subject to resolution among the Debtors, Patheon, and the Purchaser.

| Lonafarnib | Trialog Clinical Trials Ltd | Study Protocol No.: EIG-LNF-011, dated July 18, 2019 | |
|------------|---|---|--------|
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2019120, dated August 14, 2019 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #201989, dated December 3, 2019 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020017, dated January 27, 2020 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020082, dated March 30, 2020 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020191, dated July 28, 2020 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020201, dated August 9, 2020 | \$0.00 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020348, dated December 31, 2020 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2021-028, dated January 25, 2021 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2021-210, dated June 8, 2021 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Study Protocol No.: SCRC20042, dated June 7, 2021 | |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #20221259, dated July 20, 2022 | |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Commercial Manufacturing Services and Supply Agreement, dated October 9, 2019* | |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Commercial Quality Agreement, dated October 17, 2019, as amended by Amendment No. 1 to Quality Agreement, dated February 15, 2023 | \$0.00 |
| Lonafarnib | CordenPharma | Master Services Agreement, dated March 22, 2016 | |
| Lonafarnib | CordenPharma | Commercial Quality Agreement, dated February 19, 2020 | \$0.00 |

| Lonafarnib | Fisher Clinical Services, Inc. | Master Services Agreement, Inc. dated May 6, 2016 | \$0.00 |
|------------|--|---|--------------------|
| Lonafarnib | Fisher Clinical Services, Inc. | First Amendment and Restated Quality Agreement, dated February 23, 2021 | \$0.00 |
| Lonafarnib | INSERM U1110, Université de Strasbourg, France | Project Proposal 1 | \$0.00 |
| Lonafarnib | U1111, Centre International de Recherche en Infectiologie, Lyon, France, team HepVir | Project Proposal V-2023-03- 16, dated March 16, 2023 | \$0.00 |
| N/A | Eiger Group International, Inc. | Asset Purchase Agreement, dated December 8, 2010 | \$0.00 |
| Lonafarnib | EZUS LYON (Subsidiary of the Universite Claude Bernard Lyon 1), Subsidiary of the Universite Claude Bernard Lyon 1, Centre National de la Recherche Scientifique, Ecole Nomale Superieure de Lyon, and Inserm Transfert SA | Research Agreement, dated February 15, 2024. | \$173,200.51 |
| Lonafarnib | SATT Conectus Alsace, University of Strasbourg, French National Institute of Health and Medical Research, and Institute for Viral and Liver Diseases | Sponsored Research Agreement, dated January 12, 2024. | \$0.00 |
| Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 15 | |
| Lonafarnib | IQVIA Clinical AB | Letter of Delegation re: Protocol Number EIG-LNF- 011 | \$[•] ³ |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to MSA | · L J |
| Lonafarnib | IQVIA RDS INC. | General Services Agreement: | |

_

³ The IQVIA Biotech LLC, IQVIA Clinical AB, and IQVIA RDS INC., (collectively, "IQVIA") contract cure amounts are subject to resolution among the Debtors, IQVIA, and the Purchaser.

Case 24-80040-sgj11 Doc \$89-5 Filed 0\$/02/25 Entered 0\$/02/25 19:39:00 Desc Exhibiti €A PRage 4.3 of 20

| Lonafarnib | Novella Clinical LLC | Change Proposal 2 | |
|------------|----------------------|-------------------|--------|
| Lonafarnib | Novella Clinical LLC | Change Proposal 3 | |
| Lonafarnib | Novella Clinical LLC | Change Proposal 4 | \$0.00 |
| Lonafarnib | Novella Clinical LLC | Change Proposal 6 | |
| Lonafarnib | Novella Clinical LLC | Change Proposal 7 | |

Exhibit B

Lambda Assigned Contracts and Cure Amounts

Assigned Contracts and Cure Amounts

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|--|---|---------------------|
| Lambda | BECTON, DICKINSON AND COMPANY | Quote re: Pharmaceutical Products | \$0.00 |
| Lambda | BIORASI, LLC | Statement of Work #157-2, dated April 16, 2021, as governed by Master Services Agreement, dated June 23, 2020 | |
| Lambda | BIORASI, LLC | Change Order 1 to Statement of Work #157-2, dated December 29, 2021 | фг. ц |
| Lambda | BIORASI, LLC | Change Order 2 to Statement of Work #157-2, dated December 20, 2021 | \$[•]¹ |
| Lambda | BIORASI, LLC | Statement of Work #157-3, dated January 21, 2021, as governed by Master Services Agreement, dated June 23, 2020 | |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | Assignment and Assumption Agreement | |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | Common Stock Purchase Agreement | \$0.00 |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | License Agreement, dated April 20, 2016 | |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work, dated December 31, 2020 | |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work, dated December 31, 2021 | |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work re: Evaluation of the Stability of E.coli Working Cell Bank | \$0.00 |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work re: Preparation and Characterization of an E. coli Master Cell Bank | φ3.00 |
| Lambda | CHARLES RIVER LABORATORIES, INC. | Statement of Work re: Storage of Materials Under Controlled Conditions and Access | |

¹ The Biorasi, LLC ("<u>Biorasi</u>") contract cure amounts are subject to resolution among the Debtors, Biorasi, and the Purchaser.

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|--|--|---------------------|
| Lambda | Eurofins Biopharma Product Testing | Quotation #HEY2PH220237-01 re: Establishment of a Method for Free PEG by HPLC- CAD, dated May 26, 2022 | |
| Lambda | Eurofins Biopharma Product Testing | Quotation #HEY2PH220237-02 re: Establishment of a Method for Free PEG by HPLC- CAD, dated November 9, 2022 | \$0.00 |
| Lambda | Eurofins BioPharma Product Testing | Quotation #VFK8PH210375-01 re: FBS Qualification for for Lambda-1 (Python), dated September 14, 2021 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #W9MYPH200689-02, dated December 4, 2020 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #W9MYPH200689-05, dated April 18, 2022 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #NQ-0143063, dated December 5, 2016 | \$0.00 |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #NQ-0148470, dated March 15, 2017 | |
| Lambda | FISHER BIOSERVICES, INC. | Statement of Work | \$[•] ² |
| Lambda | Fisher Clinical Services Inc. | CO 1 to PSG-A-1073971 (PSG-A-1076893) 20230524 | |
| Lambda | Fisher Clinical Services Inc. | Quote 20160517 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A- 1043137.v1 20210812 | 420.557.77 |
| Lambda | Fisher Clinical Services Inc. | Quote 20160927 | \$29,097.55 |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 20170221 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Change Order 1 20161116 | |

_

 $^{^{2}}$ The Fisher BioServices, Inc. ("<u>Fisher</u>") contract cure amount is subject to resolution among the Debtors, Fisher, and the Purchaser.

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|-------------------------------|---|---------------------|
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Order 7 20170629 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Order 8 20180918 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 58040 20161206 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 62278 20180309 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 68128 Order 1 Version 1 20190720 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1007253 V2 20190508 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1009306.V1 20190619 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1037571.v1 20210414 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A- 1037572.v1_20210420 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1037587.v1 20210420 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1038183.v1 20210422 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1040820.V3 20210713 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1041275.v4 20210819 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1044658.v1 20210902 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1045926.v1 20210927 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1053812.v2 20220323 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1056697.v1 20221205 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1060127.v1 20221205 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1072645 v1 20230420 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1073971 20230429 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1086699.v1 20231128 | |

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|--|--|---------------------|
| Lambda | Fisher Clinical Services Inc. | Quote re: Protocol No. EIG- LMD-001 | |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1037570.v2 20210423 | |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1045938.v1 20210927 | |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1069905.v1 20230204 | |
| Lambda | Fujifilm Diosynth | Stability Studies Termination, Accountability and Reconciliation Memo | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | 2020 ("SOW9") and the Master Bioprocessing Services Agreement | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement, dated October 15, 2019 | \$[•] ³ |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Bioprocessing Services Agreement, dated September 22, 2016 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Master Services Agreement | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 5 re: MCB and WCB Bioassay Characterization | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 6 re: Establishment of Degraded SEC and Degraded Issi-Asp, CEX & RP Purity Assay Controls | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Change Order 6 re: Positional Isomer Feasibility | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGIES USA, INC. | Scope of Work 23 | |
| Lambda | INTRINSIK CORP | Statement of Work 9, dated October 12, 2022, as governed by Master Services | \$0.00 |

_

 $^{^3}$ The Fujifilm Diosynth Bioctechnologies USA, Inc. ("<u>Fuji</u>") contract cure amounts are subject to resolution among the Debtors, Fuji, and the Purchaser.

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|---|---|---------------------|
| | | Agreement, dated March 6, 2020 | |
| Lambda | INTRINSIK CORP | SOW #3, dated February 9, 2022; | |
| Lambda | INTRINSIK CORP | SOW #5, dated February 23, 2022; and | |
| Lambda | INTRINSIK CORP | SOW #6, dated March 26, 2022. | |
| Lambda | Intrinsik Health Sciences Inc. | Proposal Re: Canadian Regulatory Services for Phase II Study for PEG- Interferon Lambda, dated March 4, 2016 | |
| Lambda | KRYOCAL, LLC DBA KYROSPHERE | Statement of Understanding | \$13,125.00 |
| Lambda | Patheon UK Limited | Change of Scope COS-17- R0 to P-MNC- 101564- R3_20220324 | |
| Lambda | Patheon UK Limited | Change of Scope COS-P- MNC-101564-R3- COS-08- R3_20210309 | \$0.00 |
| Lambda | Patheon UK Limited | Validation Master Plan | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 20 P-MNC- 101564-R4_20220722 | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 24 P-MNC- 101564-R4_20230209 | 40.00 |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 29-R0 to P-MNC- 101564-R4_20240213 | \$0.00 |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope: Prefilled Syringes Patheon UK_P- MNC-101564-R4-COS-23- R0_20220922 | |
| Lambda | RRD INTERNATIONAL, LLC | First Amendment to Project Agreement 3, dated October 1, 2019 | |
| Lambda | RRD INTERNATIONAL, LLC | Project Agreement 3 to the Product Development Agreement, dated April 1, 2019, as governed by Product Development Agreement, dated July 1, 2018 | \$0.00 |

Case 24-80040-sgj11 Doc \$89-2 Filed 0\$/02/25 Entered 0\$/02/25 19:39:00 Desc Exhibiti6B PRgg €0 of 20

| Asset | Counterparty | Description of Contract | Cure Amounts |
|--------|--|---|---------------------|
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #220328-01-SF | \$0.00 |
| Lambda | Total Transport Management | Netherlands Hub Freight Quote, dated March 15, 2022 | \$0.00 |
| Lambda | Trialog Clinical Trials Ltd | Study Protocol No.: SCRC20006 Agreement, dated April 14, 2020 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, dated March 3, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG- LMD-002, dated July 20, 2022 | \$3,760.00 |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG- LMD-002, dated December 7, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG- LMD-002, dated February 21, 2023 | |

Exhibit 6

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*)

Anne G. Wallice (admitted pro hac vice)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Attorneys for the Debtors and Debtors in Possession

SIDLEY AUSTIN LLP

Charles M. Persons (TX Bar No. 24060413)

2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et

 $al.^1$

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

DEBTORS' EMERGENCY MOTION FOR THE ENTRY OF AN ORDER (I) AUTHORIZING THE SALE OF THE LONAFARNIB AND LAMBDA ASSETS FREE AND CLEAR OF LIENS, CLAIMS, ENCUMBRANCES, AND OTHER INTERESTS, (II) AUTHORIZING THE ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES, (III) GRANTING THE PURCHASER THE PROTECTIONS AFFORDED TO A GOOD FAITH PURCHASER, (IV) APPROVING PURCHASER PROTECTIONS IN CONNECTION WITH THE SALE OF THE LONAFARNIB AND LAMBDA ASSETS, AND (V) GRANTING RELATED RELIEF

Emergency relief has been requested. Relief is requested not later than 9:30 a.m. prevailing Central Time on August 20, 2024.

If you object to the relief requested or you believe that emergency consideration is not warranted, you must appear at the hearing if one is set, or file a written response prior to the date that relief is requested in the preceding paragraph. Otherwise, the Court may treat the pleading as unopposed and grant the relief requested.

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

A hearing will be conducted on this matter on August 20, 2024 at 9:30 a.m. prevailing Central Time in Courtroom 1, Floor 14, 1100 Commerce Street, Dallas, TX 75242-1496.

You may participate in the hearing either in person or by an audio and video connection.

Audio communication will be by use of the Court's dial-in facility. You may access the facility at 1.650.479.3207. Video communication will be by use of the Cisco WebEx platform. Connect via the Cisco WebEx application or click the link on Judge Jernigan's home page. The meeting code is 2304-154-2638. Click the settings icon in the upper right corner and enter your name under the personal information setting.

Hearing appearances must be made electronically in advance of electronic hearings. To make your appearance, click the "Electronic Appearance" link on Judge Jernigan's home page. Select the case name, complete the required fields and click "Submit" to complete your appearance.

The debtors and debtors in possession in the above-captioned chapter 11 cases (collectively, the "<u>Debtors</u>"), hereby submit this motion (this "<u>Motion</u>") for entry of an order approving the sale of the Lonafarnib/Lambda Assets to Eiger InnoTherapeutics, Inc. (the "<u>Lonafarnib/Lambda Sale Order</u>"). In support of this Motion, the Debtors respectfully state as follows:

Relief Requested

1. The Debtors seek entry of an order (i) authorizing the Debtors to sell assets in connection with Lonafarnib (the "Lonafarnib Assets") and Lambda (the "Lambda Assets" and together with the Lonafarnib Assets, the "Lonafarnib/Lambda Assets") to Eiger InnoTherapeutics, Inc. and/or its assigns (the "Purchaser") free and clear of all liens, claims, encumbrances, and other interests, pursuant to that (a) certain Asset Purchase Agreement for the sale of the Lonafarnib Assets, substantially in the form attached as Exhibit 1 to the Lonafarnib/Lambda Sale Order (the "Lonafarnib APA"), and (b) certain Asset Purchase Agreement for the sale of the Lambda Assets, substantially in the form attached as Exhibit 2 to the Lonafarnib/Lambda Sale Order (the "Lambda APA" and together with the Lonafarnib APA, the "Lonafarnib/Lambda APAs"); (ii) authorizing the assumption and assignment of executory contracts and unexpired leases (the "Assigned Contracts"); (iii) granting the Purchaser the protections afforded to a good faith purchaser pursuant to Section 363(m) of the Bankruptcy Code (as defined below); (iv) approving Purchaser's

protections in the form of a termination fee and expense reimbursement in connection with the sale of the Lonafarnib/Lambda Assets; and (v) granting related relief as described herein. In support of this Motion, the Debtors respectfully submit the Declaration of J. Scott Victor in Support of the Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear Of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief (the "Victor Declaration") filed contemporaneously herewith.

Jurisdiction and Venue

- 2. The United States Bankruptcy Court for the Northern District of Texas (the "Court") has jurisdiction over this matter pursuant to 28 U.S.C. § 1334. This matter is a core proceeding pursuant to 28 U.S.C. § 157(b). The Debtors confirm their consent, pursuant to rule 7008 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), to the Court's entry of a final order in connection with this motion.
 - 3. Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409.
- 4. The bases for the relief requested herein are Sections 105, 363(b), (f), and (m), and 365 of the Bankruptcy Code and Bankruptcy Rules 2002 and 6004.

Background

I. General Background

5. The Debtors were a commercial-stage biopharmaceutical company focused on the development of innovative therapies for hepatitis delta virus (HDV) and other serious diseases. All of the Debtors' rare disease programs have FDA Breakthrough Therapy designation.

- 6. On April 1, 2024 (the "Petition Date"), each Debtor filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code. The Debtors are operating their business and managing their property as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. These chapter 11 cases have been consolidated for procedural purposes only and are being administered jointly. No party has requested the appointment of a trustee or examiner in these cases.
- 7. A detailed description of the Debtors and their business, and the facts and circumstances supporting this motion and the Debtors' chapter 11 cases, are set forth in greater detail in the *Declaration of David Apelian in Support of Chapter 11 Petitions and First Day Pleadings* [Docket No. 19] (the "First Day Declaration").

II. Previous Sale Transactions

- 8. Prior to the Petition Date and following the commencement of these chapter 11 cases, the Debtors and their advisors have actively marketed the Debtors' assets. This marketing process enabled the Debtors to conduct two successful auctions: one for the sale of the Zokinvy Assets and another for the sale of the Avexitide Assets. Following these Auctions, the Debtors have successfully consummated two Sale Transactions for the sale of the Zokinvy Assets and the Avexitide Assets. On April 24, 2024, the Court entered the *Order (I) Approving the Sale of the Debtors' Zokinvy Assets, (II) Authorizing Assumption and Assignment of Certain Executory Contracts and Unexpired Leases Related Thereto, and (III) Granting Related Relief [Docket No. 162], approving the sale of the Zokinvy Assets to Sentynl Therapeutics, Inc. ("Sentynl"). The Zokinvy Sale Transaction closed on May 3, 2024 [Docket No. 214].*
- 9. On June 26, 2024, the Court entered the Order (I) Approving the Sale of the Debtors' Avexitide Assets, (II) Authorizing Assumption and Assignment of Certain Executory

Contracts and Unexpired Leases Related Thereto, and (III) Granting Related Relief [Docket No. 376]. The Avexitide Sale Transaction closed on July 10, 2024 [Docket No. 416].

The Marketing Process for the Sale of the Lonafarnib/Lambda Assets

- During the weeks prior to the Petition Date, the Debtors began an extensive marketing process for the Debtors' assets, including the Lonafarnib/Lambda Assets. The Debtors engaged SSG Advisors, LLC ("SSG"), an investment banking firm, in March 2024, to assist in: (i) identifying and evaluating potential counterparties for a potential sale process; (ii) preparing a marketing plan and information materials describing the Debtors to distribute to potential buyers on a confidential basis; (iii) assisting the Debtors in contacting potential buyers, arranging meetings with such parties, and coordinating the due diligence investigation of the Debtors; (iv) developing a strategy to effectuate both a sale of the Debtors' Assets; (v) structuring and negotiating a potential sale transaction(s) and the related logistics surrounding a potential sale transaction(s); and (vi) assisting the Debtors with the coordination of a data room and due diligence efforts.
- 11. Following the Petition Date, SSG continued the marketing process for the Debtors' Assets, including the Lonafarnib/Lambda Assets. SSG worked closely with the Debtors' senior management and other advisors, to identify potential bidders for the Lonafarnib/Lambda Assets.
- 12. Through this process, SSG interfaced with potential purchasers to solicit the highest or otherwise best proposal from third parties and attempted to facilitate an auction process. As of August 5, 2024, SSG contacted approximately 342 potential purchasers for the Lonafarnib Assets and approximately 376 potential purchasers for the Lambda Assets. Following this initial engagement, 24 potential purchasers entered into nondisclosure agreements with the Debtors in order to gain access to the virtual data room for the sale of the Lonafarnib/Lambda Assets. SSG shared teasers, financial models, and other related financial and operational information. Relevant

information regarding the Debtors' business has been made available in the virtual data room, which includes more than 3,000 pages of diligence for the Lonafarnib Assets and more than 22,500 pages of diligence for the Lambda Assets.

- 13. After extensive discussions with potential purchasers and engaging with their key stakeholders, the Debtors, in an exercise of their business judgment, identified one potential purchaser for the Lonafarnib/Lambda Assets and decided to sell the Lonafarnib/Lambda Assets to the Purchaser free and clear of all liens, claims, encumbrances, and other interests, pursuant to the Lonafarnib/Lambda APAs. The Marketing Process was robust and appropriate to identify potential bidders for the Lonafarnib/Lambda Assets and, at this juncture, entry into the Lonafarnib/Lambda APAs presents the best means to achieve the highest and best available value of the Debtors' estates for all stakeholders.
- 14. This Motion is an integral component to maximize the value of the Lonafarnib/Lambda Assets. Given that these assets have been extensively marketed prior to the filing of these chapter 11 cases and throughout these chapter 11 cases, the Debtors believe that the relief requested herein is the best means to achieve the highest and best value of the Debtors' estates for all stakeholders. Moreover, completion of the sale process in a timely manner will further maximize the value of the Lonafarnib/Lambda Assets and value available to the estate for the benefit of all stakeholders. Thus, the Debtors have determined that pursuing the sale in the manner proposed herein is in the best interest of the Debtors' estates.

An Expedited Sale is Both Necessary and Appropriate

15. As noted above and in further detail in the Victor Declaration, the Debtors commenced these chapter 11 cases to facilitate orderly sales of the Debtors' assets during a liquidity crisis. Victor Decl. ¶ 8. After already successfully completing the sale of the Zokinvy

Casse22448800449ssjýjj11.1 DDocc84896 Filitelc08*4025*225 <u>Hintercel</u>c08*4025*225201247235 DDessc Mai Exbritoriu finen PagePage 2705 f 33

Assets and the sale of the Avexitide Assets, the Debtors now have the opportunity to complete the Lonafarnib/Lambda Sale Transactions for the final significant group of assets, bringing at least \$6,200,000 in additional sale proceeds into the Debtors' estate for the benefit of its stakeholders and eliminating future costs and expenses relating to the sales process and operations and maintenance relating to the Lonafarnib/Lambda Assets. An expedited sale timeline for the Lonafarnib/Lambda Sale Transactions is necessary and appropriate where, as here (1) the Debtors have extensively marketed the assets for sale over a period of approximately four months; (2) the transaction terms were negotiated in good faith and at arms-length; (3) the transaction is the highest and best offer for the Lonafarnib/Lambda Assets and is value-maximizing for the benefit of the Debtors' estates and its stakeholders; (4) no other bidders have made an actionable proposal to buy the Lonafarnib/Lambda Assets; (5) the estates will realize significant savings by having the Lonafarnib/Lambda Transactions close promptly, including reducing go-forward legal spend and shifting expenses relating to upcoming FDA response deadlines related to the Lambda Assets to Purchaser; (6) counterparties to Assigned Contracts have already received timely notice of the potential assignment and of the Debtors' asserted Cure Amounts; (7) the sale is supported by the Debtors, the Official Committee of Unsecured Creditors (the "UCC") and the Official Equity Security Holders' Committee (the "ESHC" and together with the UCC, the "Committees"); and (8) the resolution of the sale of the remaining assets will allow the Debtors to focus their efforts towards confirming their Plan and making distributions of the estates' assets to creditors and stakeholders.

16. The Lonafarnib/Lambda APAs were reached following months of arm's-length, good faith negotiations, and are expected to close as expeditiously as possible after the receipt of Court approval. Victor Decl. ¶ 9. The proceeds of the sale of the Lonafarnib/Lambda Assets, and

Casse2**24880049**9sgijj**1**1.1 DDoor8**39**96 Filitiek00**8402522**5 Eintereck08**402522**52**0**12**4723**5 DDess Mai Exbritoitu tõnen PaqeP3ropé 2805 f 33

the elimination of ongoing expenses associated with the Lonafarnib/Lambda Assets from the Debtors' balance sheet, are critical to maximizing value for the estates and for the Debtors to focus their limited resources on the remainder of these chapter 11 cases, including confirmation of their plan and distribution of estate assets to creditors and other stakeholders. Victor Decl. ¶ 10. Simply put, the sooner this transaction can close, the greater recovery there will be for all estate stakeholders.

- Assets, the Debtors would be required to expend significant time, money and resources including clinical and laboratory review and reporting responding to time-sensitive inquiries from the FDA relating to the Lambda Assets. That process would minimize estate assets and take away focus from the Debtors' Plan process. The costs and burden associated with the FDA reporting can be assumed by the Purchaser, and transitioned away from the estates—provided that the Lonafarnib/Lambda Sale Transactions are promptly closed.
- The Lonafarnib/Lambda Assets have already been marketed for over four months, and the Debtors believe that the probability that a competing bidder will emerge with a higher and better offer for the Lonafarnib/Lambda Assets that can close within the necessary time frame is extremely low. Victor Decl. ¶ 22. Notably, no other bids were received for the Lonafarnib Assets by the July 19, 2024 bid deadline. *Further Revised Notice of Bid Deadline* [Docket. 422]. And, importantly, the Purchaser's offer for the Lonafarnib Assets is expressly contingent upon the purchase of the Lambda Assets. The Debtors simply cannot risk loss of a transaction that, in the exercise of their reasonable business judgment, maximizes value for the Debtors' estates and all stakeholders. Victor Decl. ¶ 12.

- 19. Contract counterparties to each of the Assigned Contracts timely received notice of any proposed Cure Amounts. The Cure Notices (defined below) related to the Lonafarnib/Lambda Assets were served out on June 10th and June 21st, respectively, and the objection deadline to object to cure has already passed. The Lonafarnib/Lambda APA provides for the resolution of any remaining cure objection and the payment, in full, of Cure Amounts for the vast majority of the Debtors' remaining contract counterparties. Accordingly, no contract counterparty will be prejudiced by the expedited sale timeline.
- 20. Under these circumstances, the Debtors believe that the expedited sale of the Lonafarnib/Lambda Assets to the Purchaser pursuant to the terms and conditions of the Lonafarnib/Lambda APAs is both appropriate and in the best interest of the Debtors, the estates, their creditors, and all stakeholders. Given the extensive marketing process of all assets, including the Lonafarnib/Lambda Assets, and other efforts to date, the Debtors believe a sale process on an expedited timeline is appropriate. Victor Decl. ¶ 8-12.

The Lonafarnib APA

- 21. As more fully described in the Lonafarnib APA, and as summarized below, the Debtors propose to enter into a transaction whereby the Purchaser will acquire the Lonafarnib Assets free and clear of all liens, claims, encumbrances, and other interests (the "Lonafarnib Sale Transaction").
- 22. The material terms of the Lonafarnib Sale Transaction are found in the Lonafarnib APA,² and are summarized in the following table:

SUMMARY OF LONAFARNIB APA

² To the extent of any conflict between the terms of the Lonafarnib APA and this Motion, the terms of the Lonafarnib APA shall govern.

| Parties | The Lonafarnib APA is entered into by and between Eiger InnoTherapeutics, Inc. (the "Purchaser") and Eiger BioPharmaceuticals, Inc., a Delaware corporation (the "Seller"). |
|--|---|
| Purchase Price (Lonafarnib APA, Section 2.9) | At Closing, Purchaser will pay to Seller an amount equal to US \$5,200,000 (the "Base Price"), plus (a) if Purchaser does not assume any Cross-Over Contract, then up to \$180,000 in the aggregate, (b) if Purchaser assumes the IQVIA Contracts, then up to \$2,180,000, (c) if Purchaser assumes the Biorasi Contracts and the IQVIA Contracts, then up to \$2,380,000, or (d) if Purchaser assumes the Biorasi Contracts but not the IQVIA Contracts, then up to \$380,000 in the aggregate. |
| Transferred Assets (Lonafarnib APA, Section 2.1) | As more fully described in the Lonafarnib APA attached hereto, the Transferred Assets consists of the following assets, properties and rights of Seller Used in the Business: • the Assigned Contracts; • the Owned Intellectual Property Assets; • the Transferred Regulatory Information; • the Business Books and Records exclusively relating to any Lonafarnib Antiviral Product; • all rights to receive mail and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group relating to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field (including any such mail and other correspondence and communications (including electronic mail) from the FDA or any other Governmental Authority, customers, advertisers, suppliers, distributors, agents and others and payments with respect to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field; • all of the Seller Group's rights, claims or causes of action, whether class, individual or otherwise in nature, under contract or in law or in equity, against third parties relating to the assets, properties, business or operations of the Seller Group with respect to the Business, the Transferred Assets and the Assumed Liabilities (including all guaranties, warranties, indemnities and similar rights in favor of the Seller Group or any their Affiliates to the extent solely related to the Transferred Assets or the Assumed Liabilities), in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date, except for such rights, claims and causes of action related to the Excluded Assets or Excluded Liabilities; • all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, choses in action, rights |

| | of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent used in or held for use for the Transferred Assets listed in clauses (a) through (f) in Section 2.1 of the Lonafarnib APA or the Assumed Liabilities; Inventory; Transferred Data; Transferred Studies; All advertising, marketing, market research, sale and promotional files and materials (including any television, radio and print content and materials), pricing lists, consulting deliverables and other related literature, catalogs, point of sale materials and website content, including all intellectual property therein, relating to any transferred asset and assumed liability that are within the seller group's control or reasonably accessible to the seller group; and To the extent not covered, any goodwill associated with or symbolized by any of the foregoing Transferred Assets described in clauses (a) through (k) in Section 2.1 of the Lonafarnib APA and any properties, rights and interests of every kind and nature, whether tangible or intangible, real, personal or mixed, known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, wherever located, associated with or appurtenant to the above-referenced Transferred Assets |
|----------------------------------|---|
| Excluded Assets (Lonafarnib APA, | As more fully described in the Lonafarnib APA attached hereto, the Excluded Assets consists of the following assets, properties and rights of Seller Used in the Business: |
| Section 2.2) | all (i) cash and cash equivalents, wherever located, including bank balances and bank accounts or safe deposit boxes, monies in the possession of any banks, savings and loans or trust companies and similar cash items, (ii) escrow monies and deposits in the possession of landlords and utility companies, and (iii) investment securities and other short- and medium-term investments; all records, documents or other information exclusively relating to current or former employees of the Seller Group that are not hired by Purchaser, and any materials to the extent containing information about any employee, disclosure of which would violate Applicable Law or such employee's reasonable expectation of privacy; any interest of the Seller Group under the Lonafarnib APA or the Related Documents, including the right to receive the Purchase Price and to enforce the Seller's rights and remedies thereunder; all Excluded Contracts (including all prepaid assets relating to the Excluded Contracts), other than the Assigned Contracts, to |

- which any member of the Seller Group or any of their respective Affiliates is a party;
- any (i) Attorney-Client Information arising from communications prior to the Closing Date between a member of the Seller Group (including any one or more officers, directors or stockholders of such Seller Group member), on the one hand, and its counsel, on the other hand, and (ii) claims under any director and officer, errors and omissions, fiduciary and commercial crime insurance policies; and
- any rights of the Seller Group to Tax refunds (or credits for overpayment of Taxes in lieu of a refund) attributable to any Pre-Closing Tax Period;
- all Permits (including applications therefor and any trade or import/export Permits) that (i) are not materially related to the Business or (ii) are not transferable to Purchaser under Applicable Law;
- the Excluded Books and Records;
- any assets not otherwise designated as Transferred Assets or from time to time designated by the parties hereto as Excluded Assets:
- all accounts receivable, intercompany obligations and other amounts receivable by the Seller Group;
- the Avoidance Actions;
- all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group (including all guaranties, warranties, indemnities and similar rights in favor of the Sellers Group or any of their Affiliates) to the extent arising under the Bankruptcy Code or relating to any of the Excluded Assets or Excluded Liabilities, in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date; and
- all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent exclusively related to or exclusively used in or held for use for the Excluded Assets listed in clauses (a) through (l) in Section 2.2 of the Lonafarnib APA.

Notwithstanding anything to the contrary contained in the Lonafarnib APA or any of the other Related Documents, Purchaser acknowledges and agrees that all of the following are also Excluded Assets, and all right, title and interest in and to all Excluded Assets shall be retained by the Seller Group and shall remain the property of the Seller Group (and shall expressly be excluded from the sale,

| | transfer, assignment and conveyance to Purchaser hereunder), and neither Purchaser nor any of its Affiliates shall have any interest therein: (x) all records and reports prepared or received by the Seller Group or any of their Affiliates in connection with the sale of the Business and the Transactions, including all analyses relating to the Business or Purchaser so prepared or received; and (y) all confidentiality agreements with prospective purchasers of the Business or any portion thereof and all bids and expressions of interest received from third parties with respect thereto. |
|--|--|
| Assumed Liability (Lonafarnib APA, Section 2.3) | Purchaser shall, effective as of the Closing, assume and agree to pay, discharge and perform in accordance with their terms the following Liabilities of the Seller Group arising from or related to the Business or the Transferred Assets as the same shall exist on the Closing Date arising only after the Closing Date (collectively, the "Assumed Liabilities"), including: |
| | all Liabilities relating to the Transferred Assets other than the Assigned Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing; subject to Section 2.4, all Liabilities arising under the Assigned Contracts other than the Existing Manufacturing Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Costs; subject to Section 2.4 of the Lonafarnib APA, all Liabilities arising under each Existing Manufacturing Contract solely to the extent such Liabilities relate to and arise (i) in connection with the transition activities under Section 7.6 of the Lonafarnib APA performed by the Seller pursuant to Purchaser's instructions following the Closing and before the applicable Existing Manufacturing Contract Transfer Date and (ii) in periods following the applicable Existing Manufacturing Contract Transfer Date, and all of the Purchaser Cure Amounts; and all Taxes for which Purchaser is liable pursuant to the Lonafarnib APA |
| Excluded Liability (Lonafarnib APA, Section 2.4) | Notwithstanding Section 2.3 of the Lonafarnib APA, Purchaser is assuming only the Assumed Liabilities of the Seller Group and will not assume or be liable for any Excluded Liabilities (including Seller Group Taxes), and the Seller Group shall retain and shall be responsible for, all Liabilities that are not Assumed Liabilities, including all Liabilities related to Excluded Assets or any other Liabilities of the Business (all such Liabilities not being assumed herein referred to as the "Excluded Liabilities"). The Excluded Liabilities shall exclude any amounts payable or due to Merck for the assignment by Seller to Purchaser of the Merck License |

| | Agreement, respectively, whether arising in periods before or |
|---|---|
| | following the Closing, which shall be solely borne by Purchaser. |
| Purchaser Protections (Lonafarnib APA, Section 9.3) | The Purchaser Protections consist of (i) a Termination Fee equal to \$36,000; and (ii) the Expense Reimbursement for reimbursement of costs, fees and expenses, in an amount not to exceed \$224,000. |
| Termination Events (Lonafarnib APA, | Notwithstanding anything to the contrary, the Lonafarnib APA may be terminated, and the Transactions may be abandoned at any time prior to the Closing: |
| Section 9.1) | by mutual written consent of Purchaser and the Seller; automatically, upon (i) the consummation of a sale or other disposition of all or substantially all of the Transferred Assets to a Person other than Purchaser (each, an "Alternate Transaction"), (ii) if, at close of the Auction, Purchaser's bid has not been selected as either the winning bid or the Back-Up Bid or (iii) if, at the close of the Auction, Purchaser's bid was selected as the Back-Up Bid, upon the consummation of a Competing Bid or Alternate Transaction;³ by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if the Bankruptcy Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code; by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if Purchaser is not selected as having the winning bid or Back-Up Bid at Auction, if any; by Purchaser if the Seller (i) withdraws the motion for the Sale Order, or publicly announces its intention to withdraw such motion, (ii) moves to voluntarily dismiss the Bankruptcy Cases, (iii) moves for conversion of the Bankruptcy Cases to Chapter 7 of the Bankruptcy Code, or (iv) moves for appointment of an examiner with expanded powers pursuant to Section 1104 of the Bankruptcy Code or a trustee in the Bankruptcy Cases; by Purchaser, by written notice from Purchaser to the Seller, if there has been a breach or inaccuracy of a covenant, representation or warranty made by the Seller in the Lonafarnib APA, such that the conditions in Section 8.1 or Section 8.2 of the Lonafarnib APA are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by the Seller prior to the earlier of (i) twenty (20) Business Days after receipt of written notice from Purchaser requesting such breach be cured or (ii) the Outside Date; provided, however, that the right to terminate the Lonafarnib APA pursuant to Section 9.1(f) of the Lon |

³ For the avoidance of doubt, the Auction was canceled and, as a result, the Termination Events relating to the Auction are no longer applicable.

- be available to Purchaser if the failure of Purchaser to fulfill any of its obligations under the Lonafarnib APA has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 of the Lonafarnib APA are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in the Lonafarnib APA;
- by the Seller, by written notice from the Seller to Purchaser, if there has been a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in the Lonafarnib APA, such that the conditions in Section 8.1 or Section 8.3 of the Lonafarnib APA are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by Purchaser prior to the earlier of (i) 20 Business Days after receipt of written notice from the Seller requesting such breach be cured or (ii) the Outside Date; provided, however, that the right to terminate the Lonafarnib APA pursuant to this Section 9.1(g) of the Lonafarnib APA shall not be available to the Seller if the failure of the Seller to fulfill any of its obligations under the Lonafarnib APA has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 of the Lonafarnib APA are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by the Seller in the Lonafarnib APA;
- by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if any Governmental Authority of competent jurisdiction shall have issued an Order, enacted any Applicable Law or taken any other action restraining, enjoining or otherwise prohibiting the consummation of the Transactions and, in the case of Orders and other actions, such Order or other action shall have become Final Orders; provided, however, that the right to terminate the Lonafarnib APA pursuant to this Section 9.1(h) of the Lonafarnib APA shall not be available to the party seeking to terminate if any action of such party or any failure of such party to act has contributed to such Order or other action and such action or failure constitutes a breach of the Lonafarnib APA;
- by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if the Closing has not occurred on or prior to October 5, 2024 (the "Outside Date"); provided, however, that the party exercising the right to terminate the Lonafarnib APA pursuant to Section 9.1(i) of the Lonafarnib APA shall not have been responsible for such failure of the Closing to occur through a breach or inaccuracy of a covenant, representation or warranty contained in the Lonafarnib APA (it being understood,

| | acknowledged, and agreed that if Seller is unable to provide any required Closing deliverable of Seller, then Seller shall be deemed to have been responsible for such failure of the Closing for purposes of this Section 9.1(i) of the Lonafarnib APA); or by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a), Section 9.3(b), and Section 9.3(c) of the Lonafarnib APA. |
|-------------------------------------|---|
| Closing Condition | A closing condition for the Closing of the Lonafarnib APA includes that the Closing of the Lambda APA takes place. |
| (Lonafarnib APA, Section 2.8(h)) | |

The Lambda APA

- 23. As more fully described in the Lambda APA, and as summarized below, the Debtors propose to enter into a transaction whereby the Purchaser will acquire the Lambda Assets free and clear of all liens, claims, encumbrances, and other interests (the "Lambda Sale Transaction" and together with the Lonafarnib Sale Transaction, the "Lonafarnib/Lambda Sale Transactions").
- 24. The material terms of the Lambda Sale Transaction are found in the Lambda APA, and are summarized in the following table:⁴

| SUMMARY OF LAMBDA APA | |
|---------------------------|--|
| Parties | The Lambda APA is entered into by and between Eiger InnoTherapeutics, Inc. (the "Purchaser") and Eiger |
| | BioPharmaceuticals, Inc., a Delaware corporation (the "Seller"). |
| Purchase Price | At Closing, Purchaser will pay to Seller an amount equal to US \$1,000,000 (the "Base Price"), plus up to \$269,000 in the aggregate |
| (Lambda APA, Section 2.9) | in Cure Costs. |

⁴ To the extent of any conflict between the terms of the Lambda APA and this Motion, the terms of the Lambda APA shall govern.

Transferred Assets

(Lambda APA, Section 2.1)

As more fully described in the Lambda APA attached hereto, the Transferred Assets consists of the following assets, properties and rights of Seller Used in the Business:

- the Assigned Contracts;
- the Owned Intellectual Property Assets;
- the Transferred Regulatory Information;
- the Business Books and Records exclusively relating to the Business or any Licensed Compound or Licensed_Product;
- all rights to receive mail and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group relating to any Licensed Compound or Licensed Product (including any such mail and other correspondence and communications (including electronic mail) from the FDA or any other Governmental Authority, customers, advertisers, suppliers, distributors, agents and others) and payments with respect to any Licensed Compound or Licensed Product;
- all of the Seller Group's rights, claims or causes of action, whether class, individual or otherwise in nature, under contract or in law or in equity, against third parties relating to the assets, properties, business or operations of the Seller Group with respect to the Business, the Transferred Assets and the Assumed Liabilities (including all guaranties, warranties, indemnities and similar rights in favor of the Seller Group or any their Affiliates to the extent solely related to the Transferred Assets or the Assumed Liabilities), in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date, except for such rights, claims and causes of action related to the Excluded Assets or Excluded Liabilities;
- all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, choses in action, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent used in or held for use for the Transferred Assets listed in clauses (a) through (f) in Section 2.1 of the Lambda APA or the Assumed Liabilities;
- Inventory;
- Transferred Data:
- Transferred Studies;
- All advertising, marketing, market research, sale and promotional files and materials (including any television, radio and print content and materials), pricing lists, consulting deliverables and other related literature, catalogs, point of sale materials and website content, including all intellectual property therein, relating to any transferred asset and assumed liability

| | that are within the seller group's control or reasonably accessible to the seller group; and To the extent not covered, any goodwill associated with or symbolized by any of the foregoing transferred assets described in clauses (a) through (k) in Section 2.1 of the Lambda APA and any properties, rights and interests of every kind and nature, whether tangible or intangible, real, personal or mixed, known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, wherever located, associated with or appurtenant to the above-referenced transferred assets |
|---------------------------|--|
| Excluded Assets | As more fully described in the Lambda APA attached hereto, the |
| (Lambda APA, Section 2.2) | Excluded Assets consists of the following assets, properties and rights of Seller Used in the Business: all (i) cash and cash equivalents, wherever located, including bank balances and bank accounts or safe deposit boxes, monies in the possession of any banks, savings and loans or trust companies and similar cash items, (ii) escrow monies and deposits in the possession of landlords and utility companies, and (iii) investment securities and other short- and medium-term investments; all records, documents or other information exclusively relating to current or former employees of the Seller Group that are not hired by Purchaser, and any materials to the extent containing information about any employee, disclosure of which would violate Applicable Law or such employee's reasonable expectation of privacy; any interest of the Seller Group under the Lambda APA or the Related Documents, including the right to receive the Purchase Price and to enforce the Seller's rights and remedies thereunder; all Excluded Contracts (including all prepaid assets relating to the Excluded Contracts), other than the Assigned Contracts, to which any member of the Seller Group or any of their respective Affiliates is a party; any (i) Attorney-Client Information arising from communications prior to the Closing Date between a member of the Seller Group (including any one or more officers, directors or stockholders of such Seller Group member), on the one hand, and its counsel, on the other hand, and (ii) claims under any director and officer, errors and omissions, fiduciary and commercial crime insurance policies; and any rights of the Seller Group to Tax refunds (or credits for overpayment of Taxes in lieu of a refund) attributable to any Pre-Closing Tax Period; all Permits (including applications therefor and any trade or import (overst Pormits) that (i) are not metarcially related to the import (overst Pormits) that (i) are not metarcially related to the |

import/export Permits) that (i) are not materially related to the

Business or (ii) are not transferable to Purchaser under Applicable Law;

- the Excluded Books and Records;
- any assets not otherwise designated as Transferred Assets or from time to time designated by the parties hereto as Excluded Assets:
- all accounts receivable, intercompany obligations and other amounts receivable by the Seller Group;
- the Avoidance Actions;
- all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group (including all guaranties, warranties, indemnities and similar rights in favor of the Sellers Group or any of their Affiliates) to the extent arising under the Bankruptcy Code or relating to any of the Excluded Assets or Excluded Liabilities, in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date; and
- all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent exclusively related to or exclusively used in or held for use for the Excluded Assets listed in clauses (a) through (l) in Section 2.2 of the Lambda APA.

Notwithstanding anything to the contrary contained in the Lambda APA or any of the other Related Documents, Purchaser acknowledges and agrees that all of the following are also Excluded Assets, and all right, title and interest in and to all Excluded Assets shall be retained by the Seller Group and shall remain the property of the Seller Group (and shall expressly be excluded from the sale, transfer, assignment and conveyance to Purchaser hereunder), and neither Purchaser nor any of its Affiliates shall have any interest therein: (x) all records and reports prepared or received by the Seller Group or any of their Affiliates in connection with the sale of the Business and the Transactions, including all analyses relating to the Business or Purchaser so prepared or received; and (y) all confidentiality agreements with prospective purchasers of the Business or any portion thereof and all bids and expressions of interest received from third parties with respect thereto.

Assumed Liabilities

(Lambda APA, Section 2.3)

Purchaser shall, effective as of the Closing, assume and agree to pay, discharge and perform in accordance with their terms the following Liabilities of the Seller Group arising from or related to the Business or the Transferred Assets as the same shall exist on the Closing Date

| | arising only after the Closing Date (collectively, the "Assumed Liabilities"), including: all Liabilities relating to the Transferred Assets other than the Assigned Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing; subject to Section 2.4, all Liabilities arising under the Assigned Contracts other than the Existing Manufacturing Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Costs; subject to Section 2.4 of the Lambda APA, all Liabilities arising under each Existing Manufacturing Contract solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Amounts; and all Taxes for which Purchaser is liable pursuant to the Lambda APA. |
|---|--|
| Excluded Liability (Lambda APA, Section 2.4) | Notwithstanding Section 2.3 of the Lambda APA, Purchaser is assuming only the Assumed Liabilities of the Seller Group and will not assume or be liable for any Excluded Liabilities (including Seller Group Taxes), and the Seller Group shall retain and shall be responsible for, all Liabilities that are not Assumed Liabilities, including all Liabilities related to Excluded Assets or any other Liabilities of the Business (all such Liabilities not being assumed herein referred to as the "Excluded Liabilities"). The Excluded Liabilities shall exclude any amounts payable or due to BMS for the assignment by Seller to Purchaser of the BMS License Agreement, respectively, whether arising in periods before or following the Closing, which shall be solely borne by Purchaser. |
| Purchaser Protections (Lambda APA, Section 9.3) | The Purchaser Protections consist of the Expense Reimbursement for reimbursement of costs, fees and expenses, in an amount not to exceed \$50,000. |
| Termination Events (Lambda APA, Section 9.1) | Notwithstanding anything to the contrary, the Lambda APA may be terminated, and the Transactions may be abandoned at any time prior to the Closing: by mutual written consent of Purchaser and the Seller; automatically, upon (i) the consummation of a sale or other disposition of all or substantially all of the Transferred Assets to a Person other than Purchaser (each, an "Alternate Transaction"), (ii) if, at close of the Auction, Purchaser's bid has not been selected as either the winning bid or the Back-Up Bid or (iii) if, at the close of the Auction, Purchaser's bid was selected as the Back-Up Bid, upon the consummation of a Competing Bid or Alternate Transaction; |

- by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if the Bankruptcy Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code;
- by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if Purchaser is not selected as having the winning bid or Back-Up Bid at Auction, if any;
- by Purchaser if the Seller (i) withdraws the motion for the Sale Order, or publicly announces its intention to withdraw such motion, (ii) moves to voluntarily dismiss the Bankruptcy Cases, (iii) moves for conversion of the Bankruptcy Cases to Chapter 7 of the Bankruptcy Code, or (iv) moves for appointment of an examiner with expanded powers pursuant to Section 1104 of the Bankruptcy Code or a trustee in the Bankruptcy Cases;
- by Purchaser, by written notice from Purchaser to the Seller, if there has been a breach or inaccuracy of a covenant, representation or warranty made by the Seller in the Lambda APA, such that the conditions in Section 8.1 or Section 8.2 of the Lonafarnib APA are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by the Seller prior to the earlier of (i) twenty (20) Business Days after receipt of written notice from Purchaser requesting such breach be cured or (ii) the Outside Date; provided, however, that the right to terminate the Lambda APA pursuant to this Section 9.1(f) of the Lambda APA shall not be available to Purchaser if the failure of Purchaser to fulfill any of its obligations under the Lambda APA has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 of the Lambda APA are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in the Lambda APA;
- by the Seller, by written notice from the Seller to Purchaser, if there has been a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in the Lambda APA, such that the conditions in Section 8.1 or Section 8.3 of the Lambda APA are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by Purchaser prior to the earlier of (i) 20 Business Days after receipt of written notice from the Seller requesting such breach be cured or (ii) the Outside Date; provided, however, that the right to terminate the Lambda APA pursuant to Section 9.1(g) of the Lambda APA shall not be available to the Seller if the failure of the Seller to fulfill any of its obligations under the Lambda APA has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 of the Lambda APA are not capable of being satisfied because there is

| | then a breach or inaccuracy of a covenant, representation or warranty made by the Seller in the Lambda APA; by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if any Governmental Authority of competent jurisdiction shall have issued an Order, enacted any Applicable Law or taken any other action restraining, enjoining or otherwise prohibiting the consummation of the Transactions and, in the case of Orders and other actions, such Order or other action shall have become Final Orders; provided, however, that the right to terminate the Lambda APA pursuant to Section 9.1(h) of the Lambda APA shall not be available to the party seeking to terminate if any action of such party or any failure of such party to act has contributed to such Order or other action and such action or failure constitutes a breach of the Lambda APA; by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if the Closing has not occurred on or prior to October 5, 2024 (the "Outside Date"); provided, however, that the party exercising the right to terminate the Lambda APA pursuant to Section 9.1(i) of the Lambda APA shall not have been responsible for such failure of the Closing to occur through a breach or inaccuracy of a covenant, representation or warranty contained in the Lambda APA (it being understood, acknowledged, and agreed that if Seller is unable to provide any required Closing deliverable of Seller, then Seller shall be deemed to have been responsible for such failure of the Closing for purposes of Section 9.1(i) of the Lambda APA); or by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a), Section 9.3(b), and Section 9.3(c) of the Lambda APA. |
|--|---|
| Closing Condition (Lambda APA, Section 2.8(g)) | A closing condition for the Closing of the Lambda APA includes that the Closing of the Lonafarnib APA takes place. |

Assumption and Assignment of Designated Contracts

25. The Debtors have previously provided notice of the potential assumption and assignment of executory contracts and unexpired leases in connection with the Remaining Assets Sale Transaction(s), including for the Lonafarnib/Lambda Sale Transactions [Docket Nos. 313 and

351] (the "Cure Notices"). The Cure Notices provided counterparties of potential executory contracts and unexpired leases notice of the Debtors' calculation of the Cure Amounts with respect thereto for the Lonafarnib Assets. The deadline to object to the potential assumption, assignment, or designation of a Potentially Assigned Contract (as defined in the Cure Notices) was June 12, 2024 at 4:00 p.m. (CT). Accordingly, the Debtors seek authorization of the assumption and assignment of executory contracts and unexpired leases in connection with the Lonafarnib/Lambda Sale Transaction. For the avoidance of doubt, the Debtors will file a Designated Contracts List for the Lonafarnib/Lambda Sale Transactions, in accordance with the Bidding Procedures.

Arguments

- I. The Court Should Approve the Sale of the Lonafarnib Assets Pursuant to Section 363(b) of the Bankruptcy Code
- 26. Section 363(b)(1) of the Bankruptcy Code provides that a debtor, "after notice and a hearing, may use, sell, or lease, other than in the ordinary course of business, property of the estate." 11 U.S.C. § 363(b)(1). To approve the use, sale or lease of property outside the ordinary course of business, this Court need only determine that the Debtors' decision is supported by "some articulated business justification," as established by the Second Circuit in *Committee of Equity Sec. Holders v. Lionel Corp.* (*In re Lionel Corp.*), 722 F.2d 1063, 1070 (2nd Cir. 1983), which decision has been adopted in this circuit. *Institutional Creditors of Continental Air Lines, Inc. v. Continental Air Lines, Inc., et al.* (*In re Continental Air Lines, Inc.*), 780 F.2d 1223, 1226 (5th Cir. 1986); *see also Fulton State Bank v. Schipper*, 933 F.2d 513, 515 (7th Cir. 1991); *In re San Jacinto*

Glass Indus., Inc., 93 B.R. 934, 944 (Bankr. S.D. Tex. 1988); In re Condere Corp., 228 B.R. 615, 628-69 (Bankr. S.D. Miss. 1998).

- 27. The business judgment rule shields a debtor's management from judicial second-guessing. See In re Johns-Manville Corp., 60 B.R. 612, 615-16 (Bankr. S.D.N.Y. 1986) ("a presumption of reasonableness attaches to a debtor's management decisions"). Once a debtor articulates a valid business justification, "[t]he business judgment rule 'is a presumption that in making a business decision the directors of a corporation acted on an informed basis, in good faith and in the honest belief that the action was in the best interests of the company." In re Integrated Resources, Inc., 147 B.R. 650, 656 (Bankr. S.D.N.Y. 1992) (quoting Smith v. Van Gorkom, 488 A.2d 858, 872 (Del. 1985)).
- Thus, if a debtor's actions satisfy the business judgment rule, then the transaction in question should be approved under Section 363(b)(1). When applying the business judgment standard, courts show great deference to a debtor's business decisions. *See GBL Holding Co., Inc. v. Blackburn/Travis/Cole, Ltd.*, 331 B.R. 251, 254 (N.D. Tex. 2005); *In re First Wellington Canyon Assocs.*, 1989 U.S. Dist. LEXIS 10687, at *8-9 (N.D. Ill. September 8, 1989) ("Under this test, the debtor's business judgment . . . must be accorded deference unless shown that the bankrupt's decision was taken in bad faith or in gross abuse of the bankrupt's retained discretion.").
- 29. Here, the Debtors respectfully submit that sound business reasons support their decision to enter into the Lonafarnib/Lambda APAs. The Debtors extensively marketed their assets during successive marketing processes for more than four months. Based upon the Debtors' extensive marketing efforts and knowledge of this industry, the Debtors do not foresee obtaining a higher or better offer from a competing purchaser. The purchase price agreed upon for the Lonafarnib/Lambda Assets in the Lonafarnib/Lambda APAs which provides for at least

\$6,200,000 in value to the Debtors' estates – is the result of extensive, good-faith, arm's-length negotiations, and is currently the highest and best proposal. Additionally, an expeditious closing of the sale is essential to enabling the Debtors to expeditiously move towards completing their chapter 11 liquidation without incurring additional operating or administrative estate resources. Therefore, the Debtors' proposed entry into the Lonafarnib/Lambda APAs for the Lonafarnib/Lambda Assets is in the best interest of the Debtors' estates and constitutes a good exercise of the Debtors' business judgment.

II. The Court Should Authorize the Sale of the Lonafarnib/Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests Pursuant to Sections 363(f) and 105 of the Bankruptcy Code

- 30. The Debtors respectfully submit that it is appropriate to sell the Lonafarnib/Lambda Assets free and clear of all interests, pursuant to Section 363(f) of the Bankruptcy Code, with all such interests attaching to the net proceeds of the Lonafarnib/Lambda Assets to the extent applicable. Section 363(f) of the Bankruptcy Code authorizes a debtor to sell assets free and clear of liens, claims, interests and encumbrances if:
 - i. applicable non-bankruptcy law permits sale of such property free and clear of such interests;
 - ii. such entity consents;
 - iii. such interest is a lien and the price at which such property is to be sold is greater than the value of all liens on such property;
 - iv. such interest is in bona fide dispute; or
 - v. such entity could be compelled, in a legal or equitable proceeding, to accept a money satisfaction of such interest.
- 11 U.S.C. § 363(f). This provision is supplemented by Section 105(a) of the Bankruptcy Code, which provides that "[t]he Court may issue any order, process or judgment that is necessary or appropriate to carry out the provisions of [the Bankruptcy Code]." 11 U.S.C. § 105(a).

- 31. Because Section 363(f) of the Bankruptcy Code is drafted in the disjunctive, satisfaction of any one of its five requirements will suffice to permit the sale of the Subject Assets free and clear of the interests. *In re Nature Leisure Times, LLC*, 2007 WL 4554276, *3 (Bankr. E.D. Tex. Dec. 19, 2007) ("The language of § 363(f) is in the disjunctive such that a sale free and clear of an interest can be approved if any one of the aforementioned conditions contained in § 363(f) are satisfied."); *In re Wolverine Radio Co.*, 930 F.2d 1132, 1147 n.24 (6th Cir. 1991) (stating that Section 363(f) of the Bankruptcy Code is written in the disjunctive; holding that the court may approve the sale "free and clear" provided at least one of the subsections of § 363(f) is met).
- 32. The Debtors believe that one or more of the tests under Section 363(f) will be satisfied with respect to the transfer of the Lonafarnib/Lambda Assets pursuant to a Lonafarnib/Lambda Sale Order, and therefore should be approved by the Court pursuant to Section 363(f) and 105 of the Bankruptcy Code.

III. The Court Should Authorize the Protections for the Purchaser

- 33. Courts have acknowledged that the approval of expenses in connection with substantial sales in bankruptcy is warranted to compensate an unsuccessful, potential acquirer whose initial offer served as the basis and catalyst for higher or better offers, which can only inure to the benefit of the debtor's estate. *See Calpine Corp. v. O'Brien Envtl. Energy, Inc.*, 181 F.3d 527, 533 (3d Cir. 1999) (holding that even though bidding incentives are measured against a business judgment standard in non-bankruptcy transactions, the administrative expense provisions of 503(b) of the Bankruptcy Code govern in the bankruptcy context); *see also In re Hupp Indus., Inc.*, 140 B.R. 191, 195 (Bankr. N.D. Ohio 1992).
- 34. The proposed buyer protections are appropriate in these chapter 11 cases in light of the circumstances. Bid Procedures Order, ¶ 10 (approving the Remaining Asset Bid Protections

up to a maximum of five percent for any purchaser who establishes a baseline bid for the remaining assets). Here, the proposed buyer protections are consistent with the Bid Procedures Order and therefore should be approved. Moreover, the Purchaser agreed to serve as the purchaser of the Lonafarnib/Lambda Assets, which is helping to preserve the value of the Debtors' assets and benefit the Debtors' estates. In the unlikely event the Lonafarnib/Lambda Sale Transactions do not close after entry of the Lonafarnib/Lambda Sale Order, the Purchaser should be entitled to the Expense Reimbursement for its costs and expenses for these efforts. The Termination Fee equal to \$36,000; and the Expense Reimbursement for reimbursement of costs, fees and expenses, in an amount not to exceed \$274,000 in the aggregate for the Lonafarnib/Lambda Sale Transactions are designed to protect the Purchaser, are reasonable under the circumstances, and are within the parameters of fees commonly approved in this District.

IV. The Court Should Grant the Purchaser the Full Protections Afforded to a Good Faith Purchaser Pursuant to Section 363(m) of the Bankruptcy Code

35. Pursuant to Section 363(m) of the Bankruptcy Code, a good faith purchaser is one who purchases assets for value, in good faith, and without notice of adverse claims. *See Miami Ctr. Ltd. P'ship v. Bank of New York*, 838 F.2d 1547, 1554 (11th Cir. 1988); *In re Mark Bell Furniture Warehouse, Inc.*, 992 F.2d 7, 9 (1st Cir. 1993); *In re Willemain v. Kivitz*, 764 F.2d 1019, 1023 (4th Cir. 1985); *In re Congoleum Corp.*, Case No. 03-51524, 2007 WL 1428477 (Bankr. D.N.J. May 11, 2007). Due to the absence of a bright line test for good faith, the determination is based on the facts of each case, concentrating on the "integrity of [an actor's] conduct in the course of the sale proceedings." *In re Pisces Leasing Corp.*, 66 B.R. 671, 673 (E.D.N.Y. 1986) (*quoting In re Rock Indus. Mach. Corp.*, 572 F.2d 1195, 1998 (7th Cir. 1978)). Further, while the Bankruptcy Code does not define "good faith," the court in *In re Sullivan Central Plaza I, Ltd.*, 106 B.R. 934 (Bankr. N.D. Tex. 1998) stated that:

[t]he type of conduct of a Buyer which would destroy its good faith status under § 363(m) involves fraud, collusion between the Buyer and other bidders of the trustee, or an attempt to take grossly unfair advantages of other bidders.

106 R. at 938 (citing *Matter of Bleaufontaine, Inc.*, 634 F.2d 1383, 1388 (5th Cir. 1981)).

36. Here, there is no fraud or collusion between the Purchaser or any of its affiliates and the Debtors. The Debtors, led by their restructuring professionals, and the Purchaser engaged in extensive arm's-length discussions to reach a deal. Further, the Debtors attest that neither the proposed Purchaser of the Lonafarnib/Lambda Assets nor any of its assigns or affiliates are "insiders" of the Debtors within the meaning of Section 101(31) of the Bankruptcy Code, and are not controlled by, or acting on behalf of, any insider of the Debtors. Accordingly, the Debtors request the Court find that the Purchaser is a "good faith" buyer under Section 363(m) of the Bankruptcy Code.

V. A Private Sale of the Lonafarnib/Lambda Assets Is Appropriate Under Bankruptcy Rule 6004

37. Bankruptcy Rule 6004(f) permits a debtor to conduct a private sale pursuant to Section 363 of the Bankruptcy Code. Specifically, Bankruptcy Rule 6004(f) provides that "[a]ll sales not in the ordinary course of business may be by private sale or by public auction." FED. R. BANKR. P. 6004(f)(1) (emphasis added). *See In re Alisa P'ship*, 15 B.R. 802, 802 (Bankr. D. Del. 1981) (holding that manner of sale is within the debtor's discretion); *In re Bakalis*, 220 B.R. 525, 531-32 (Bankr. E.D.N.Y. 1998) (stating that debtor has authority to conduct public or private sales of estate property).

⁵ For the avoidance of doubt, the Purchaser is a former director of the Debtors. The Purchaser was not a director at the time the Purchaser began discussions with the Debtors or their advisors for the sale of the Lonafarnib/Lambda Assets.

- 38. Accordingly, in light of Bankruptcy Rule 6004(f) and case law regarding Section 363 sales, a debtor may conduct a private sale if a good business reason exists. *See*, *e.g.*, *In re Pritam Realty. Inc.*, 233 B.R. 619 (D. P.R. 1999) (upholding the bankruptcy court's approval of a private sale conducted by a chapter 11 debtor); *In re Condere Corp.*, 228 B.R. 615, 629 (Bankr. S.D. Miss. 1998) (authorizing private sale of debtors' tire company where "[d]ebtor has shown a sufficient business justification for the sale of the assets to the [p]urchaser"); *In re Embrace Sys. Corp.*, 178 B.R. 112, 123 (Bankr. W.D. Mich. 1995) ("A large measure of discretion is available to a bankruptcy court in determining whether a private sale should be approved. The court should exercise its discretion based upon the facts and circumstances of the proposed sale."); *In re Wieboldt Stores. Inc.*, 92 B.R. 309 (N.D. Ill. 1988) (affirming right of chapter 11 debtor to transfer assets by private sale).
- 39. The Debtors submit that the proposed private sale of the Lonafarnib/Lambda Assets is appropriate in light of the facts and circumstances of these chapter 11 cases. Specifically, and as more fully described herein and in the Victor Declaration, given that the Lonafarnib/Lambda Assets have already been marketed for over four months, and that the Purchaser is the only bidder that provided a bid for the Lonafarnib/Lambda Assets in their entirety, the Debtors believe that the probability that a competing bidder will emerge with a higher and better offer that can close within the necessary time frame is extremely low; and that extreme probability does not justify incurring the costs and risks associated with a delay of the sale.
- 40. The Lonafarnib/Lambda Sale Transactions will allow the Debtors to maximize the value of the Lonafarnib/Lambda Assets and provide a significant benefit to the Debtors' estates. Because private sales are specifically authorized under Bankruptcy Rule 6004 and the Debtors believe that the Purchaser's offer represents the highest and best offer for the Lonafarnib/Lambda

Assets, the Debtors request that the Court approve the proposed private sale on the terms described herein and in the Lonafarnib/Lambda APA.

VI. The Form, Manner, and Extent of Notice of the Motion Is Appropriate and Adequate Under the Circumstances

- 41. The Debtors will serve the Sale Motion as set forth herein. Such notice to be provided by the Debtors as set forth herein sufficiently describes the terms and conditions of the proposed Lonafarnib/Lambda Sale Transactions.
- 42. Several sections of the Bankruptcy Code and Bankruptcy Rules dictate the sufficiency of notice and adequacy of service. As discussed below, the content and manner of service of this Motion and the related notices satisfy all such requirements:
 - i. Section 363 Notice Section 363 of the Bankruptcy Code provides that a trustee may sell property "after notice and hearing." Under Section 102(1) of the Bankruptcy Code, the phrase "after notice and hearing" means "notice as is appropriate in the particular circumstances, and such opportunity for a hearing as is appropriate in the particular circumstances." 11 U.S.C. § 102(1)(A). As set forth above, creditors have been provided notice of the salient details regarding this Motion and the Sale Hearing. Accordingly, notice is sufficient under Section 363 of the Bankruptcy Code.
 - ii. <u>Emergency Relief</u> The Procedures For Complex Cases in The Northern District of Texas (the "<u>Complex Case Rules</u>") permit the Debtors to file this Motion on an emergency basis. The Debtors request this Motion be heard on an emergency basis in accordance with the Local Rules and Complex Case Rules. As set forth above, the notice of this Motion that has been and will be provided by the Debtors satisfies each of the requirements under the Complex Case Rules.
 - iii. <u>Bankruptcy Rule 6004</u> Bankruptcy Rule 6004 requires that notice of sales of property outside of the ordinary course of business complies with Bankruptcy Rule 2002. As set forth above, the Debtors have complied with Bankruptcy Rule 2002.
 - iv. <u>Procedural Due Process</u> The notices of this Motion that are being provided as described herein, including the notice being provided by publication as set forth above, are "reasonably calculated" to apprise interested parties of the pendency of the matter and to afford them an opportunity to object. *See Mullane v. Central Hanover Bank & Trust Co.*, 339 U.S. 306 (1950). Parties in interest have been and should be found to have been afforded adequate notice of this Motion, the Lonafarnib/Lambda Sale Transactions, the Bidding Procedures and the other relief requested herein.

43. The Debtors submit that the notice they have provided, as outlined above, with respect to the Sale Motion, is reasonable and appropriate and constitutes good and adequate notice of the Lonafarnib/Lambda Sale Transactions of the Lonafarnib/Lambda Assets and therefore should be approved by this Court.

VI. The Stay of the Lonafarnib/Lambda Sale Order Should Be Waived

44. Pursuant to Bankruptcy Rule 6004(h) an order authorizing the Sale of the Lonafarnib/Lambda Assets is stayed for fourteen (14) days after the entry of an order unless the Court orders otherwise. To preserve the value of the Lonafarnib/Lambda Assets and limit the costs of administering and preserving such assets, it is critical that the Debtors promptly and expeditiously close the Lonafarnib/Lambda Sale Transactions as soon as possible after all closing conditions have been met or waived. Accordingly, the Debtors hereby request that the Court waive the 14-day stay period under Bankruptcy Rule 6004(h). The Debtors submit that the relief requested is necessary and appropriate, is in the best interests of the Debtors and their estate and should be granted in all respects.

Notice

45. The Debtors will provide notice of this motion to the following: (a) the U.S. Trustee; (b) the holders of the thirty (30) largest unsecured claims against the Debtors (on a consolidated basis); (c) Innovatus; (d) the Unsecured Creditors Committee; (e) the Equity Committee; (f) the United States Attorney's Office for the Northern District of Texas; (g) the Food and Drug Administration; (h) the Internal Revenue Service; (i) the United States Securities and Exchange Commission; (j) the state attorneys general for the states in which the Debtors conduct business; and (k) any party that has requested notice pursuant to Bankruptcy Rule 2002. No other or further notice is needed in light of the nature of the relief requested.

[Remainder Of Page Intentionally Left Blank.]

Casse22488004409sgijj.11 DDooc84996 Filitielc084025225 Hintereelc0840252252012427205 DDess Mai Exbritation en Page P340 e 1320 f 33

The Debtors respectfully requests entry of the Lonafarnib/Lambda Sale Order granting the relief requested herein and granting such other relief as is just and proper.

Dated: August 5, 2024 SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*) 787 Seventh Avenue

New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000 Dallas, Texas 75201

Telephone: (214) 981-3300

Facsimile: (214) 981-3400

Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Casse2248800440ssggJ11 DDocc84896 Filitelc084025245 Eintereclc0840252252012427205 DDessc MailExbridationenPagePage13205f 33

Certificate of Service

I certify that on August 5, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ Thomas R. Califano

Thomas R. Califano

Exhibit A

Proposed Lonafarnib/Lambda Sale Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re: Chapter 11

EIGER BIOPHARMACEUTICALS, INC., et al. 1

Debtors.

Case No. 24-80040 (SGJ)

(Jointly Administered)

ORDER (I) AUTHORIZING

THE SALE OF THE LONAFARNIB AND LAMBDA ASSETS FREE AND CLEAR OF LIENS, CLAIMS, ENCUMBRANCES, AND OTHER INTERESTS, (II) AUTHORIZING THE ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES, (III) GRANTING THE PURCHASER THE PROTECTIONS AFFORDED TO A GOOD FAITH PURCHASER, (IV) APPROVING PURCHASER PROTECTIONS IN CONNECTION WITH THE SALE OF THE LONAFARNIB AND LAMBDA ASSETS, AND (V) GRANTING RELATED RELIEF

_

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A), and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

Upon consideration of the Motion² of the debtors and debtors in possession in the abovecaptioned chapter 11 cases (collectively, the "Debtors") for entry of an order (this "Lonafarnib/Lambda Sale Order"), pursuant to sections 105(a), 363, and 365 of title 11 of the United States Code (the "Bankruptcy Code") and Rules 2002, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), Rule 9013-1 of the Bankruptcy Local Rules for the Northern District of Texas (the "Bankruptcy Local Rules"), and Section E of the Procedures for Complex Chapter 11 Cases in the Northern District of Texas (the "Complex Case Procedures"), authorizing (a) the Debtors' sale of certain of their property free and clear of liens, claims, encumbrances, and interests on the terms set forth in that certain (i) Asset Purchase Agreement by and between Eiger InnoTherapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated as of August 1, 2024 for the sale of the Lonafarnib Assets (the "Lonafarnib APA") attached hereto as Exhibit A and (ii) Asset Purchase Agreement by and between Eiger InnoTherapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated as of August 1, 2024 for the sale of the Lambda Assets (the "Lambda APA" and together with the Lonafarnib APA, the "Lonafarnib/Lambda APAs") attached hereto as **Exhibit B**; (b) the assumption and assignment of the Assigned Contracts in connection with the Lonafarnib/Lambda APAs; and (c) granting related relief, all as more fully set forth in the Motion; and this Court having previously entered the Order (I)(A) Approving the Bid Procedures;

² Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the *Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V)_Granting Related Relief [Docket No. [•]] (the "Motion") or the Lonafarnib/Lambda APAs. If there are any inconsistencies between the defined terms in the Lonafarnib/Lambda APAs and the Motion, the defined terms of the Lonafarnib/Lambda APAs shall control.*

(B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; and (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order"); and the Debtors having filed the Notice of Cancellation of Auction(s), Designation of Winning Bid for the Lonafarnib Sale Transaction, and Transition To Private Sale Process for Lonafarnib/Lambda Sale Transactions [Docket No. 489] (the "Lonafarnib/Lambda Sale Notice") selecting Eiger InnoTherapeutics, Inc. ("Inno") as the highest and best bidder for the Transferred Assets (the "Purchaser") pursuant to the Lonafarnib/Lambda APAs with the Purchaser having submitted the highest and best offer for the Transferred Assets to be sold to the Purchaser as identified in the Lonafarnib/Lambda APAs, as reflected in the Lonafarnib/Lambda APAs and as from time to time may be amended in accordance with this Lonafarnib/Lambda Sale Order or further order of this Court, including by the Lonafarnib/Lambda APAs, pursuant to which the Debtors have agreed, among other things, to sell the Transferred Assets to the Purchaser, including the Assigned Contracts that will be assumed and assigned to the Purchaser on the terms and conditions set forth in the Lonafarnib/Lambda APAs inclusive of the Lonafarnib Purchase Price (the "Lonafarnib Sale Transaction") and Lambda Purchase Price (the "Lonafarnib Sale Transaction" and together with the Lonafarnib Sale Transaction, the "Lonafarnib/Lambda Sale Transactions"); and the Debtors having filed the Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired

Leases in Connection with the Remaining Sale Transaction(s) [Docket No. 313] (the "Cure Notice") and the Amended Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Remaining Assets Sale Transaction(s) [Docket No. 351] (the "Amended Cure Notice") and served the Lonafarnib Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. A] and the Lambda Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. B] (the "Assignment Notice"); and this Court having conducted the hearing to consider approval of the Lonafarnib/Lambda Sale Transactions (the "Lonafarnib/Lambda Sale Hearing"), at which time all interested parties were offered an opportunity to be heard with respect to the Lonafarnib/Lambda Sale Transactions; and this Court having reviewed and considered (i) the Motion and the exhibits thereto, (ii) the First Day Declaration [Docket No. 27], (iii) and the Declaration of J. Scott Victor [4] in Support of the Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Docket No. [___]], and the arguments and representations of counsel made, and the evidence proffered or adduced at the Lonafarnib/Lambda Sale Hearing; and it appearing that due and proper notice of the Motion, the Lonafarnib/Lambda APAs, the Cure Notice, the Amended Cure Notice, and the Assignment Notice having been provided; and all objections, if any, to approval of the Lonafarnib/Lambda Sale Transactions having been withdrawn, resolved (including by separate agreement between the objecting party and the Debtors), adjourned, or overruled as provided in this Lonafarnib/Lambda Sale Order; and

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit 6A Prage 46 off 1205

it appearing entry of this Lonafarnib/Lambda Sale Order and consummation of the Lonafarnib/Lambda Sale Transactions are in the best interests of the Debtors, their estates and creditors, and all parties in interest in these chapter 11 cases; and upon the record of the Lonafarnib/Lambda Sale Hearing and these chapter 11 cases; and after due deliberation thereon; and sufficient cause appearing therefor,

IT IS HEREBY FOUND AND DETERMINED THAT:

- A. <u>Fed. R. Bankr. P. 7052</u>. The findings and conclusions set forth herein constitute this Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such. This Court's findings shall also include any oral findings of fact and conclusions of law made by this Court during or at the conclusion of the Lonafarnib/Lambda Sale Hearing. To the extent of any conflict, the oral rulings control.
- B. <u>Jurisdiction and Venue</u>. This Court has jurisdiction over the Motion and the Lonafarnib/Lambda Sale Transactions described therein, and in the Lonafarnib/Lambda APAs, including, without limitation, the sale of the Transferred Assets, pursuant to 28 U.S.C. §§ 157 and 1334. Venue for these Chapter 11 Cases is proper pursuant to 28 U.S.C. § 1408. This Court may enter a final order consistent with Article III of the United States Constitution. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- C. <u>Statutory Predicates</u>. The statutory authorization for the relief granted herein is found in sections 105(a), 363, and 365 of the Bankruptcy Code; Bankruptcy Rules 2002, 6004,

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit 6A Prage 47 off 1205

and 9014; Rule 9013-1 of the Bankruptcy Local Rules; and Section E of the Complex Case Procedures.

- D. This Lonafarnib/Lambda Sale Order constitutes a final and appealable order within the meaning of 28 U.S.C. § 158(a). Time is of the essence in closing the Lonafarnib/Lambda Sale Transactions referenced herein, and the Debtors and the Purchaser intend to close the Lonafarnib/Lambda Sale Transactions as soon as practicable in accordance with the Lonafarnib/Lambda APAs, and there is no just reason for delay in the implementation of this Lonafarnib/Lambda Sale Order. Specifically, the Lonafarnib/Lambda Sale Transactions must be approved and consummated promptly in accordance with the Lonafarnib/Lambda APAs to preserve the viability of the business in the hands of the Purchaser as a going concern, and to maximize the value to the Debtors, their estates, their creditors, and all other parties in interest. Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), and to any extent necessary under Bankruptcy Rule 9014 and Rule 54(b) of the Federal Rules of Civil Procedure, as made applicable by Bankruptcy Rule 7054, the Court expressly finds that there is no just reason for delay in the implementation of this Lonafarnib/Lambda Sale Order in accordance with the Lonafarnib/Lambda APAs, waives any stay, and expressly directs entry of judgment as set forth herein.
- E. Marketing Process. The Debtors and their professionals adequately marketed the Transferred Assets to all Potential Bidders in accordance with the Bid Procedures Order. The sale process set forth in the Bid Procedures Order afforded all Potential Bidders (as defined in the Bid Procedures, attached as Exhibit 1 to the Bid Procedures Order and revised in that *Notice of Filing of Revised Bidding Procedures*, filed on April 15, 2024 [Docket No. 119]), as modified by the *Revised Notice of Sale, Bid Procedures, Auction, and Sale Hearing* [Docket No. 331], as further modified by the *Further Revised Notice of Bid Deadlines* [Docket No. 422], a full, fair, and

Case 24-80040-sgj11 Doc **898-6** Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit 6A Prage 42 off 1205

Assets and participate in the sale process. On August 2, 2024 the Debtors filed the Lonafarnib/Lambda Sale Notice that cancelled the Auctions. The value provided by the Purchaser pursuant to the Lonafarnib/Lambda APAs, which reflect (i) the final bid of a Base Price in the amount of \$5,200,000 plus the Purchaser Cure Amounts and Assumed Liabilities (collectively, the "Lonafarnib Purchase Price") and (ii) the final bid of a Base Price in the amount of \$1,000,000 plus the Purchaser Cure Amounts and Assumed Liabilities (collectively, the "Lambda Purchase Price" and together with the Lonafarnib Purchase Price, the "Purchase Price") provides the greatest value to the Debtors for the Transferred Assets. The Purchase Price constitutes the highest and best bid for the Transferred Assets. The marketing process was robust and sufficiently tested the market to determine the highest and best offer for the Transferred Assets.

- F. <u>Purchaser Protections</u>. The Purchaser Protections contained in the Lambda APA and the Lonafarnib APA (i) were necessary to preserve the value of the Debtors' estates by inducing the Purchaser to enter into the Lambda APA and the Lonafarnib APA, and (ii) are in compliance with the Bid Procedures and authorized by the Bid Procedures Order.
- G. <u>Sale Hearing</u>. This Court conducted the Lonafarnib/Lambda Sale Hearing on August 20, 2024, at which time this Court considered the Motion, the evidence and testimony presented, and the statements and argument of counsel, as applicable, in support of the Motion, the Lonafarnib/Lambda APAs, and the Lonafarnib/Lambda Sale Transactions. Except as otherwise expressly provided in this Lonafarnib/Lambda Sale Order, all objections to the Lonafarnib/Lambda Sale Transactions and the relief requested in the Motion, whether timely or untimely and whether written or made orally at the Lonafarnib/Lambda Sale Hearing, if any, were heard and considered by this Court. All such objections, if any, were either overruled by this

Case 24-80040-sgj11 Doc **898-6** Filed **08/25/25** Entered **08/25/25 20:22:25** Desc Exhibit 6A Prage 43 off 1205

Court, are resolved by the terms hereof or by separate agreement between the objecting party and the Debtors, or were adjourned or withdrawn as a result of an agreement between the objecting party and the Debtors.

- H. <u>Sound Business Purpose</u>. The Debtors have demonstrated good, sufficient, and sound business purposes and justifications for consummation of the Lonafarnib/Lambda Sale Transactions pursuant to the Lonafarnib/Lambda APAs and all other agreements, instruments, certificates, and other documents to be entered into or delivered by any party in connection with the Lonafarnib/Lambda Sale Transactions, including, without limitation, any assumption and assignment agreements entered into in connection therewith (collectively, the "<u>Transaction Documents</u>"), outside of the ordinary course of business and in accordance with the requirements of section 363(b) of the Bankruptcy Code. Consummation of the Lonafarnib/Lambda Sale Transactions prior to and not as part of a chapter 11 plan is (i) justified under the circumstances, (ii) an appropriate exercise of the Debtors' business judgment, and (iii) in the best interests of the Debtors, their estates, and their creditors.
- I. Following a robust marketing process, the Lonafarnib/Lambda APAs and the Purchase Price constitute the highest and best offer for the Transferred Assets. No other person, or group of persons, has offered to purchase the Transferred Assets for an amount that would give greater value to the Debtors than the value provided by the Purchase Price. The Lonafarnib/Lambda Sale Transactions are the best means available to the Debtors to maximize the return to their creditors and limit the losses to counterparties to the Assigned Contracts. No alternative to the Lonafarnib/Lambda Sale Transactions exists that would provide a greater value to the Debtors, their creditors, or other parties in interest.

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 40 of 205

- J. Approval of the Lonafarnib/Lambda Sale Transactions is necessary to maximize the value the Debtors' estates will receive for the Transferred Assets. It is important to the Debtors' customers and suppliers that the transition from the Debtors to the Purchaser occurs smoothly and without unnecessary delay, so that any customer and vendor issues may be minimized. It is also important that the Lonafarnib/Lambda Sale Transactions be consummated as expeditiously as possible to avoid any disruption to the ongoing development of the Lonafarnib/Lambda Assets and so there is no uncertainty about the future of the Lonafarnib/Lambda Assets.
- K. Accordingly, the sale of the Transferred Assets pursuant to sections 105(a), 363, and 365 of the Bankruptcy Code upon the terms and conditions set forth in the Lonafarnib/Lambda APAs is the optimal means to create value for the benefit of the Debtors' estates. The Lonafarnib/Lambda Sale Transactions maximize the value of the Transferred Assets because the Transferred Assets are being sold as part of a going concern, thereby preserving the continuity and remaining goodwill value associated with the Transferred Assets. Unless the sale is concluded expeditiously, as provided for in the Motion and the Lonafarnib/Lambda APAs, creditor recoveries may be substantially diminished.
- L. <u>Fair Purchase Price</u>. The Purchase Price provided by the Purchaser (i) is fair and adequate; (ii) constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia (including the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act, and similar laws); and (iii) will provide an equal or greater recovery for the Debtors' creditors than would be provided by any other available alternative. The terms of the Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions are fair and

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 45 of 205

reasonable under the circumstances of the Debtors' chapter 11 cases, and the Debtors' determination to proceed with such transaction constitutes a valid and sound exercise of the Debtors' business judgment.

- Adequate and Reasonable Notice. As evidenced by the affidavits of service filed M. with this Court [Docket Nos. 42, 114, 128, 140, 319, 320, 324, [•]], and based upon the record of the Lonafarnib/Lambda Sale Hearing, (i) due, proper, timely, adequate, and sufficient notice of the Motion, the Lonafarnib/Lambda Sale Notice, the Lonafarnib/Lambda Sale Hearing, the Lonafarnib/Lambda APAs, and the Lonafarnib/Lambda Sale Transactions has been provided to all parties in interest, (ii) such notice was and is good, sufficient, and appropriate under the circumstances, and reasonably calculated to reach and apprise all holders of Liens, claims, encumbrances, and other Interests (as defined herein), including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities, and was provided in accordance with the applicable requirements of the Bankruptcy Code, the Bankruptcy Rules, the Bankruptcy Local Rules, the Complex Case Procedures, and the procedural due process requirements of the United States Constitution, and (iii) no other or further notice of the Motion, the Lonafarnib/Lambda Sale Hearing, the Lonafarnib/Lambda APAs, the Lonafarnib/Lambda Sale Transactions, or of the entry of this Lonafarnib/Lambda Sale Order is necessary or shall be required.
- N. In accordance with the Bid Procedures Order, the Debtors filed with this Court and served the Cure Notice and Amended Cure Notice, containing (i) the list of all Contracts to potentially be assigned in connection with the Lonafarnib/Lambda Sale Transactions, (ii) information necessary and appropriate to provide notice of the relevant proposed assumption and assignment of Potentially Assigned Contracts (as defined in the Cure Notice and Amended

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 40 of 205

Cure Notice) and rights thereunder, (iii) Cure Amounts, where applicable, and (iv) the procedures for objecting thereto, on all counterparties to such Potentially Assigned Contracts and any party that has requested notice pursuant to Bankruptcy Rule 2002 ("Rule 2002 Notice List"), and caused such notice to be published on the website of the Debtors' noticing agent, Kurtzman Carson Consultants LLC dba Verita Global ("Verita") [Docket No. 313]. The Cure Notice and Amended Cure Notice (a) included the Debtors' good faith calculation of the Cure Amounts with respect to each Potentially Assigned Contract; (b) stated that assumption or assignment of any Potentially Assigned Contract is not guaranteed and is subject to this Court's approval; and (c) prominently displayed the deadline to file a Cure Objection (as defined herein). The service and provision of the Cure Notice and Amended Cure Notice were good, sufficient, and appropriate under the circumstances, and no other or further notice need be given.

- O. The Debtors also served the Assignment Notice on the counterparties to the Assigned Contracts, which contained (i) the list of Assigned Contracts selected by the Purchaser, (ii) information necessary and appropriate to provide notice of the relevant proposed assumption and assignment of the Assigned Contracts and rights thereunder, (iii) the Cure Amounts, and (iv) the procedures for requesting adequate assurance of future performance. The Debtors also served the [Notice of Lonafarnib/Lambda Sale Hearing] [Docket No. [•]] (the "Notice of Lonafarnib/Lambda Sale Hearing") on the counterparties to the Assigned Contracts. The service and provision of the Assignment Notice and Notice of Lonafarnib/Lambda Sale Hearing was good, sufficient, and appropriate under the circumstances, and no other or further notice need be given in connection with the assumption and assignment of the Assigned Contracts.
- P. A reasonable opportunity to object and to be heard with respect to the sale of the Transferred Assets, the assumption and assignment of the Assigned Contracts, and the

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 43 of 205

determination of defaults and Cure Amounts related thereto, as well as the Lonafarnib/Lambda APAs and the entry of this Lonafarnib/Lambda Sale Order, has been given to all interested Persons.

Good Faith Purchaser. The Debtors, the Purchaser, and their respective Q. principals, counsel, and advisors have negotiated, proposed, and entered into the Lonafarnib/Lambda APAs, the Transaction Documents, and each of the transactions contemplated therein in good faith, without collusion and from arm's-length bargaining positions. The Purchaser is a "good faith purchaser" and is acting in good faith within the meaning of section 363(m) of the Bankruptcy Code in connection with the Lonafarnib/Lambda Sale Transactions and, as such, is entitled to all the protections afforded thereby. The Purchaser has proceeded in good faith in all respects. The terms of the Lonafarnib/Lambda Sale Transactions, including the Purchase Price, were not controlled by any agreement among Potential Bidders and neither the Debtors nor the Purchaser have engaged in collusion or any conduct that would cause or permit the Lonafarnib/Lambda APAs to be challenged, avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy Code or any other law of the United States, any state, territory, possession thereof, or the District of Columbia, or any other applicable law. The Lonafarnib/Lambda APAs were not entered into for the purpose of hindering, delaying, or defrauding creditors under the Bankruptcy Code or under laws of the United States, any state, territory, or possession, or the District of Columbia, or any other applicable law. Neither the Debtors nor the Purchaser entered into the Lonafarnib/Lambda APAs or are consummating the Lonafarnib/Lambda Sale Transactions with any fraudulent or otherwise improper purpose. The Purchaser is not an "insider" or "affiliate" of any of the Debtors, as those terms are defined in section 101 of the Bankruptcy Code, and no common identity of incorporators, directors, or controlling stockholders exists between the Purchaser and the Debtors.

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 48 of 205

- R. The Lonafarnib/Lambda Sale Transactions, which include the sale of the Transferred Assets pursuant to the Lonafarnib/Lambda APAs and all covenants in and conditions thereto, is an integrated transaction, meaning that each component is an essential part of every other component and that the Lonafarnib/Lambda Sale Transactions can be consummated only if all of the components are consummated. Accordingly, each component of the Lonafarnib/Lambda Sale Transactions is subject to, and is protected by, the provisions of section 363(m) of the Bankruptcy Code.
- S. Sale Free and Clear under Section 363(f). The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions without entry of this Lonafarnib/Lambda Sale Order approving the Lonafarnib/Lambda Sale Transactions pursuant to section 363(f) of the Bankruptcy Code. Except as expressly provided otherwise in the Lonafarnib/Lambda APAs or this Lonafarnib/Lambda Sale Order, the Debtors have satisfied the standard set forth in section 363(f) of the Bankruptcy Code for selling the Transferred Assets free and clear of all of the following (collectively, "Interests"): Liens (including Permitted Liens), claims (including, but not limited to, those that constitute a "claim" as defined in section 101(5) of the Bankruptcy Code), encumbrances, obligations, liabilities, pledges, charges, demands, guarantees, actions, suits, defenses, deposits, credits, allowances, options, rights, restrictions, limitations, contractual commitments, rights of first refusal, rights of setoff or recoupment, royalties, hypothecations, preferences, debts, easements, suits, licenses, rights of recovery, judgments, orders and decrees of any court or foreign or domestic governmental entity, taxes (including foreign, state, and local taxes), covenants, indentures, instruments, leases, claims for reimbursement or subrogation, contribution, indemnity or exoneration, encumbrances, or interests of any kind or nature whatsoever against the Debtors,

or any of the Transferred Assets, including, without limitation, any debts arising under or out of, in connection with, or in any way relating to, any acts or omissions, obligations, demands, guaranties, rights, contractual commitments, restrictions, product liability claims, environmental liabilities, employment or labor law claims or liabilities, employee pension or benefit plan claims, multiemployer benefit plan claims, retiree healthcare or life insurance claims or claims for taxes of or against the Debtors or against any property of the Debtors, claims arising under state or federal antitrust laws, any indemnification claim or liabilities relating to any act or omission of the Debtors or any other person prior to the Closing Date or any Excluded Liabilities, any derivative, vicarious, transferee or successor liability claims, alter ego claims, de facto merger claims, rights or causes of action (whether known or unknown, legal or equitable, contingent, matured or unmatured, contingent or non-contingent, liquidated or unliquidated, choate or inchoate, filed or unfiled, scheduled or unscheduled, perfected or unperfected, allowed or disallowed, noticed or unnoticed, recorded or unrecorded, material or non-material, statutory or non-statutory, and asserted or unasserted), whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases (other than the Assumed Liabilities), whether imposed by agreement, understanding, law, equity or otherwise, including without limitation (i) those Interests that purport to give to any party a right or option to effect a setoff against or any forfeiture, modification, or termination of the Debtors' interests in the Transferred Assets, or any similar rights, if any, (ii) those Interests arising under all mortgages, deeds of trust, security interests, conditional sale or other title retention agreements, pledges, hypothecations, liens, judgments, demands, encumbrances, rights of first refusal or charges of any land or nature, if any, (iii) those Interests that are Excluded Liabilities as set forth in the Lonafarnib/Lambda APAs, (iv) those Interests held by the Prepetition Term Loan Lenders (as defined in the Final Cash Collateral Order) as provided

in the order entered by the Court at Docket No. 161 (the "Final Cash Collateral Order"), and (v) those Interests arising under or out of, in connection with, or in any way related to the Debtors or any of the Debtors' predecessors, Affiliates, or representatives, any of the Debtors' interests in the Transferred Assets, or the operation of any of the Debtors' businesses before the applicable Closing Date, including, without limitation, Interests based on successor liability, transferee liability, derivative liability, vicarious liability, de facto merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. Each holder of an Interest in the Transferred Assets (a) has, subject to the terms and conditions of this Lonafarnib/Lambda Sale Order, consented or shall be deemed to have consented to the relief requested in the Motion and with respect to the Lonafarnib/Lambda Sale Transactions, (b) could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest, or (c) otherwise falls within the provisions of section 363(f) of the Bankruptcy Code. Those holders of Interests that did not object to, or withdrew their objections, if any, to, the relief requested in the Motion, the Lonafarnib/Lambda APAs, the Lonafarnib/Lambda Sale Transactions, the Cure Notice, the Amended Cure Notice, or the Assignment Notices are deemed to have consented to the relief requested in the Motion, including, without limitation, the sale of the Transferred Assets and the assumption and assignment of the Assigned Contracts to the Purchaser, pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests that did object that have an Interest in the Transferred Assets could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest pursuant to section 363(f)(5) of the Bankruptcy Code or fall within one or more of the other subsections of 363(f) of the Bankruptcy Code and, therefore, are adequately protected by having their Interests that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Lonafarnib/Lambda Sale Transactions ultimately

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit & Page **5**1 of **2**05

attributable to the property in which they have an Interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Lonafarnib/Lambda Sale Transactions, subject to any defenses of the Debtors.

T. Except as expressly provided otherwise in the Lonafarnib/Lambda APAs or this Lonafarnib/Lambda Sale Order, neither the Purchaser nor any of the Purchasers' Affiliates (including any subsidiary of the Purchaser, any person or entity that could be treated as a single employer with the Purchaser pursuant to Section 4001(b) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") or Section 414(b), (c), (m) or (o) of the Internal Revenue Code of 1986, as amended ("IRC"), and any of their respective managed funds or accounts, any of their respective lenders or investors, and, in each case of the foregoing, each of their respective former, current, or future, shareholders, equity holders, owners, members, managers, employees, representatives, officers, limited or general partners, directors, agents, professionals, successors, affiliates, or permitted assignees) (collectively with the Purchaser, the "Purchaser Group") shall be responsible for any Interests, including in respect of, based on, relating to, or arising under, without limitation, the following: (i) any labor, collective bargaining, or employment agreements; (ii) any mortgages, deeds of trust, or security interests; (iii) any intercompany loans and receivables between one or more of the Seller and any Debtor; (iv) any pension, multiemployer (as such term is defined in Section 3(37) or Section 4001(a)(3) of ERISA), health or welfare plan participation or benefit trust, compensation or other employee benefit plans, agreements, practices and programs (including any Employee Benefit Plan) of or related to any of the Debtors or any of the Debtors' Affiliates or predecessors or any current or former employees of any of the foregoing, including, without limitation, any pension plan of any of the Debtors or any multiemployer plan to which the Debtors have at any time contributed to or had any liability

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 52 of 205

or potential liability; (v) the Debtors' business operations or cessation thereof; (vi) any litigation involving one or more of the Debtors; (vii) any other employee, worker's compensation, occupational disease or unemployment or temporary disability related claim, including, without limitation, claims that might otherwise arise under or pursuant to (a) ERISA, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the Multi-Employer Pension Plan Amendments Act of 1980, including all amendments thereto, (f) the Worker Adjustment and Retraining Notification Act of 1988 or any similar state or local law ("WARN"), (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, including, without limitation, the requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the IRC and of any similar state law (collectively, "COBRA"), (i) the National Labor Relations Act, (j) the Age Discrimination and Employment Act of 1967 and Age Discrimination in Employment Act, as amended, (k) state harassment, discrimination, or retaliation laws, (l) state unemployment compensation laws or any other similar state laws, or (m) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors, or relating to any wages, benefits, employment, or termination of employment with any or all Debtors or any of their predecessors; (viii) any liabilities arising under any Environmental Laws with respect to any assets owned or operated by any of the Debtors or any corporate predecessor of any of the Debtors at any time prior to the applicable Closing Date; (ix) any product liability law; (x) any antitrust laws; (xi) any bulk sales or similar law; (xii) any tax statutes or ordinances, including, without limitation, the IRC; and (xiii) any Excluded Liabilities.

U. <u>No Successor, Transferee, or Similar Liability</u>. Except for the Assumed Liabilities, as expressly set forth in the Lonafarnib/Lambda APAs or this Lonafarnib/Lambda Sale

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 59 of 205

Order, the Purchaser has not expressly or impliedly assumed any obligation of the Debtors, or any other party, with respect to the Interests and the Excluded Liabilities, whether at law or in equity, whether by payment, setoff, recoupment, or otherwise, directly or indirectly, and whether from the Transferred Assets or otherwise, including, without limitation, based on successor, transferee, derivative, or vicarious liability.

- V. The Lonafarnib/Lambda Sale Transactions described by the Lonafarnib/Lambda APAs and the Transaction Documents does not amount to a consolidation, merger, or de facto merger of the Purchaser and any of the Debtors and/or any of the Debtors' estates.
- W. There is no continuity between the Purchaser and any of the Debtors. The Purchaser is not holding itself out to the public as a continuation of any of the Debtors or their respective estates, businesses, or operations. The Purchaser is not a mere continuation of any of the Debtors or their respective estates, businesses, or operations. There is no common identity between any of the Debtors and the Purchaser. The Purchaser does not constitute a successor or a successor in interest to any of the Debtors or their estates.
- X. The Purchaser and the Debtors are not entering into the Lonafarnib/Lambda APAs and Transaction Documents or consummating the Lonafarnib/Lambda Sale Transactions for the fraudulent purpose of escaping liability for the Debtors' obligations or to defraud creditors in any way.
- Y. Sale Free and Clear and Continuation of Existing Approvals Required by the Purchaser. The Purchaser expressly negotiated for the protection of obtaining the Transferred Assets free and clear of all Interests, including, without limitation, any potential successor liability claims (other than the Assumed Liabilities). The total consideration to be provided under the Lonafarnib/Lambda APAs reflects the Purchaser's reliance on this Lonafarnib/Lambda Sale Order

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 20 of 205

to provide it, pursuant to sections 105(a), 363, and 365 of the Bankruptcy Code, with title to and possession of the Transferred Assets free and clear of all Interests of any kind or nature whatsoever (including, without limitation, any potential successor liability claims (other than the Assumed Liabilities)). The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions if the sale of the Transferred Assets to the Purchaser and the assumption and assignment of the Assigned Contracts to the Purchaser by the Debtors were not free and clear of all Interests of any kind or nature whatsoever (other than the Assumed Liabilities), as contemplated by this Lonafarnib/Lambda Sale Order, or if the Purchaser would, or in the future could, be liable for any of the Interests, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities. The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions if the Purchaser would not be authorized, as of the Closing Date, to operate under or renew any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the terms and conditions of the Lonafarnib/Lambda APAs); if such licenses, permits, registrations, and governmental authorizations or approvals would not be deemed to have been transferred to the Purchaser as of the Closing Date; or if existing licenses or permits applicable to the business would not remain active and in place for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred.

Z. <u>Assumption and Assignment of the Assigned Contracts</u>. The Assumption and Assignment of the Assigned Contracts are integral to the Lonafarnib/Lambda APAs, do not constitute unfair discrimination, are in the best interests of the Debtors, their estates and creditors, and all other parties in interest, and are based on the reasonable exercise of sound business

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 25 of 205

judgment by the Debtors. At the Closing and pursuant to Section 365 of the Bankruptcy Code and this Lonafarnib/Lambda Sale Order, the Debtors shall assume and, subject to the terms in the Lonafarnib/Lambda APAs, assign to the Purchaser, and Purchaser shall take assignment from the Debtors of, the Assigned Contracts.

AA. On or after the Closing Date, the Purchaser will pay all Purchaser Cure Amounts for Assigned Contracts (if any), and the Seller will pay all Seller Cure Amounts (if any) for Assigned Contracts, including with respect to the Assigned Contracts proposed to be resolved after the Closing Date in accordance with Paragraph 19 hereof. Accordingly, the Debtors or the Purchaser, as applicable, will have, to the extent necessary, (i) cured any default existing prior to the Closing with respect to the Assigned Contracts, and (ii) provided compensation, if any, to each counterparty to an Assigned Contract for any actual pecuniary loss to such party resulting from a default prior to the Closing with respect to the Assigned Contract with such counterparty, all within the meaning of sections 365(b)(1)(A) and 365(f)(2)(A) of the Bankruptcy Code.

BB. Pursuant to section 365(f) of the Bankruptcy Code, each Assigned Contract required to be assumed and assigned under the Lonafarnib/Lambda APAs shall be assigned and transferred to, and remain in full force and effect for the benefit of, the Purchaser, in accordance with their respective terms, notwithstanding any provision in such contract or other restrictions prohibiting its assignment or transfer. No section of any of the Assigned Contracts that would directly or indirectly prohibit, restrict, or condition the assumption or assignment of any of the Assigned Contracts or would permit termination or modification of such Assigned Contracts, or rights and obligations thereunder, by a party other than the Debtors, on account of assignment of such shall have any force or effect in connection with the Transferred Assets.

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 20 of 205

CC. The assumption and assignment of the Assigned Contracts (i) are necessary to sell the Transferred Assets to the Purchaser, (ii) allow the Debtors to sell the Transferred Assets to the Purchaser as a going concern, (iii) limit the losses suffered by counterparties to the Assigned Contracts, and (iv) maximize the recoveries to other creditors of the Debtors by limiting the number of claims against the Debtors' estates by avoiding the rejection of the Assigned Contracts. For these reasons, the Debtors have exercised sound business judgment in assuming and assigning the Assigned Contracts and such assumption and assignment is in the best interests of the Debtors' estates.

DD. Adequate Assurance of Future Performance. Counterparties to the Assigned Contracts were provided with the Assignment Notice and had the opportunity to request and review information with respect to the Purchaser's adequate assurance of future performance (see Lonafarnib/Lambda Sale Notice). No counterparties to Assigned Contacts filed any objections to the Purchaser's ability to provide adequate assurance of future performance as contemplated under sections 365(b)(1)(C), 365(b)(3) (to the extent applicable) and 365(f)(1) of the Bankruptcy Code an "Adequate Assurance Objection") prior to the Lonafarnib/Lambda Sale Hearing. Counterparties to Assigned Contracts that failed to timely file an Adequate Assurance Objection are hereby forever barred from objecting to the assumption and assignment of Assigned Contracts on the grounds of a failure to provide adequate assurance of future performance. Based on evidence adduced at the Lonafarnib/Lambda Sale Hearing and based on the record in these chapter 11 cases, to the extent necessary, the Debtors have satisfied the requirements of section 365 of the Bankruptcy Code, including sections 365(b)(l)(A), 365(b)(l)(B), 365(b)(l)(C), 365(b)(3) (to the extent applicable) and 365(f) of the Bankruptcy Code, in connection with the sale and assumption and assignment of the Assigned Contracts to the extent provided under the

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 23 of 205

Lonafarnib/Lambda APAs. Accordingly, subject to payment of the Cure Amounts, the Assigned Contracts may be assumed by the Debtors and assigned to the Purchaser as provided under the Lonafarnib/Lambda APAs and this Lonafarnib/Lambda Sale Order.

- EE. <u>Lonafarnib/Lambda Sale Order Required by the Purchaser.</u> Entry of this Lonafarnib/Lambda Sale Order approving the Lonafarnib/Lambda APAs is a requirement of the Lonafarnib/Lambda APAs and such requirement is a reasonable and appropriate condition precedent to the Purchaser's consummation of the Lonafarnib/Lambda Sale Transactions.
- FF. Transferred Assets Property of the Estates. The Transferred Assets constitute property of the selling Debtors' estates and title thereto is vested in the selling Debtors' estates within the meaning of section 541(a) of the Bankruptcy Code. The selling Debtors have all title, interest, and/or rights in the Transferred Assets required to transfer and to convey the Transferred Assets to the Purchaser, as required by the Lonafarnib/Lambda APAs.
- GG. <u>Corporate Authority</u>. Subject to the entry of this Lonafarnib/Lambda Sale Order, (i) the Debtors have full corporate power and authority to perform all of their obligations under the Lonafarnib/Lambda APAs and the Transaction Documents, and the Debtors' prior execution and delivery of, and performance of obligations under, the Lonafarnib/Lambda APAs and the Transaction Documents is hereby ratified, (ii) the Debtors have all of the corporate power and authority necessary to consummate the Lonafarnib/Lambda Sale Transactions, (iii) the Debtors have taken all corporate actions necessary to authorize, approve, execute, and deliver the Lonafarnib/Lambda APAs and the Transaction Documents and to consummate the Lonafarnib/Lambda Sale Transactions, except for the closing conditions expressly provided in the Lonafarnib/Lambda APAs and the Transaction Documents, and (iv) no consents or approvals are required to consummate the Lonafarnib/Lambda Sale Transactions or otherwise perform the

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 28 of 205

obligations under the Lonafarnib/Lambda APAs or the Transaction Documents, except for the closing conditions expressly provided herein or therein.

- HH. <u>Sale in Best Interests</u>. The relief requested in the Motion and set forth in this Lonafarnib/Lambda Sale Order is in the best interests of the Debtors, their respective creditors, estates, and all other parties in interest in the Debtors' chapter 11 cases.
- II. **Prompt Consummation**. To maximize the value of the Transferred Assets, it is essential that the Lonafarnib/Lambda Sale Transactions occur within the timeframe set forth in the Lonafarnib/Lambda APAs. Time is of the essence in consummating the Lonafarnib/Lambda Sale Transactions. Accordingly, there is cause to lift the stays established by Bankruptcy Rules 6004 and 6006 with regards to the Lonafarnib/Lambda Sale Transactions and the assignment of the Assigned Contracts.

NOW, THEREFORE, IT IS ORDERED THAT:

- 1. <u>Motion Is Granted</u>. The Motion and the relief requested therein, and entry into and performance under the Lonafarnib/Lambda APAs, is GRANTED and APPROVED, as set forth herein.
- 2. Objections Overruled. Except as stated otherwise herein, all objections to, or reservation of rights regarding, the relief requested in the Motion, the entry of this Lonafarnib/Lambda Sale Order, or the relief granted herein, including, without limitation, any objections to the assumption or assignment of the Assigned Contracts (including Cure Amounts related thereto) or relating to the cure of any defaults under any of the Assigned Contracts or to the assumption and assignment of any of the Assigned Contracts to the Purchaser by the Debtors, that have not been withdrawn, waived, settled, or adjourned as provided in Paragraphs 17-19 below or otherwise, or that have not otherwise been resolved pursuant to the terms hereof are hereby

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit A Page **29** of **2**05

denied and overruled on the merits with prejudice. All Persons that failed to timely object, or withdrew their objections, to the Motion or the entry of this Lonafarnib/Lambda Sale Order are deemed to consent to the relief granted herein for all purposes, including, without limitation, pursuant to section 363(f)(2) of the Bankruptcy Code.

- 3. <u>Notice</u>. Notice of the Motion, the Lonafarnib/Lambda Sale Hearing, and the assumption and assignment of Assigned Contracts was adequate, appropriate, fair, and equitable under the circumstances and complied in all respects with section 102(1) of the Bankruptcy Code and Bankruptcy Rules 2002, 6004, and 6006, and the Bankruptcy Local Rules, and as such no further or other notice is required.
- 4. Approval and Authorization. The sale of the Transferred Assets to the Purchaser on the terms and conditions contained in the Lonafarnib/Lambda APAs and the Transaction Documents, including, without limitation, the Closing of the Lonafarnib/Lambda Sale Transactions as required by the Lonafarnib/Lambda APAs, is hereby approved in all respects pursuant to sections 105(a), 363(b), 363(f), 363(m), and 365 of the Bankruptcy Code and Bankruptcy Rules 6004 and 6006. Pursuant to sections 105, 363, and 365 of the Bankruptcy Code, the Debtors are authorized to perform all obligations under and make all payments required by the Lonafarnib/Lambda APAs and the Transaction Documents as and when due thereunder without further order of this Court. The Debtors, the Purchaser, and each of their respective officers, employees, and agents are hereby authorized to (i) execute the Lonafarnib/Lambda APAs and the Transaction Documents, including the Lonafarnib/Lambda APAs, and any prior execution of such agreements, documents, and instruments, including the Transaction Documents, is hereby ratified, (ii) perform all obligations under the Lonafarnib/Lambda APAs and the Transaction Documents, to consummate each of the foregoing, including, without limitation, deeds, assignments, and other

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 26 of 205

instruments of transfer, and to consummate the Lonafarnib/Lambda Sale Transactions, and any prior performance of such obligations or any prior consummation of such Lonafarnib/Lambda Sale Transactions is hereby ratified, (iii) assume and assign the Assigned Contracts to the Purchaser, and (iv) take all other and further actions as may be reasonably necessary to consummate and implement the Lonafarnib/Lambda Sale Transactions and to perform all obligations under the Lonafarnib/Lambda APAs and the Transaction Documents and the consummation thereof, without any further corporate action or order of this Court. The Purchaser shall not be obligated to proceed with the Closing under the Lonafarnib/Lambda APAs until all conditions precedent to its obligation to do so thereunder have been satisfied or waived.

- 5. No Sub Rosa Plan. The sale of the Transferred Assets, including, without limitation, the assignment of the Assigned Contracts, pursuant to the Lonafarnib/Lambda APAs outside a chapter 11 plan neither impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms of the Debtors' subsequent chapter 11 plan. Neither the Lonafarnib/Lambda APAs nor the Lonafarnib/Lambda Sale Transactions constitute a sub rosa chapter 11 plan.
- 6. <u>Valid Transfer</u>. As of the Closing, the consummation of the Lonafarnib/Lambda Sale Transactions shall effect a legal, valid, and enforceable sale and transfer of the Transferred Assets to the Purchaser, and shall vest the Purchaser with all legal, equitable, and beneficial right, title, and interest in and to the Transferred Assets free and clear of all Interests of any kind or nature whatsoever. The Lonafarnib/Lambda APAs and the Transaction Documents are valid and binding contracts between the Debtors and the Purchaser and shall be enforceable pursuant to their terms. The Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions themself, and the consummation thereof, shall be specifically enforceable

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 07 of 205

against and binding upon (without posting any bond) the Debtors and their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Assigned Contracts, the Purchaser and its respective successors and assigns, any chapter 11 trustee appointed in these chapter 11 cases or any chapter 7 trustee appointed upon a conversion of these chapter 11 cases to cases under chapter 7 of the Bankruptcy Code, and shall not be subject to rejection or avoidance by the foregoing parties or any other Person.

7. Free and Clear. Except as expressly provided for in the Lonafarnib/Lambda APAs or this Lonafarnib/Lambda Sale Order, pursuant to sections 105(a), 363(b), 363(f), 365(b), and 365(f) of the Bankruptcy Code, the Debtors are authorized and directed to transfer the Transferred Assets to the Purchaser and, upon the Closing, other than the Purchaser's assumption of the Assumed Liabilities and the Purchaser's obligations under the Lonafarnib/Lambda APAs and the Assigned Contracts, the Purchaser shall have and take title to and possession of the Transferred Assets free and clear of and shall have no obligation with respect to all Interests (other than the Assumed Liabilities) of any kind or nature whatsoever, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities; de facto merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. All holders of Interests fall within one or more of the subsections of section 363(f) of the Bankruptcy Code and are adequately protected by having their Interests attach to the net proceeds ultimately received by the Debtors and attributable to the Transferred Assets against or in which such Interests are asserted, subject to the terms of such Interests, with the same validity, force, and effect, and in the same order of priority that such Interests now have against the

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 02 of 205

Transferred Assets or their proceeds as of Closing, subject to any rights, claims, and defenses the Debtors or their estates, as applicable, may possess with respect thereto, in addition to any limitations on the use of such proceeds pursuant to any provision of this Lonafarnib/Lambda Sale Order. This Lonafarnib/Lambda Sale Order: (a) is and shall be effective as a determination that, other than Assumed Liabilities or as otherwise provided herein, upon the applicable Closing in accordance with the Lonafarnib/Lambda APAs, all claims of any kind or nature whatsoever existing as to Transferred Assets, and any tax liability, prior to the applicable Closing have been unconditionally released, discharged, and terminated, and that the conveyances described herein have been effected, with such Interests and liens attaching in order of priority to the proceeds of the Lonafarnib/Lambda Sale Transactions, and (b) is and shall be binding upon and shall authorize all entities, including without limitation all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies or units, governmental departments or units, secretaries of state, federal, state and local officials and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to the Transferred Assets conveyed to the Purchaser. All recorded Interests against the Transferred Assets from their records, official and otherwise, shall be deemed stricken upon the Closing in accordance with the Lonafarnib/Lambda APAs and the terms of this Lonafarnib/Lambda Sale Order without the need for further action on the part of either the Purchaser or the Seller. The conditions of section 363(f) of the Bankruptcy Code have been satisfied in full; therefore, the Debtor is authorized and directed to sell the Transferred Assets free and clear of any liens, claims, and/or interests (other than the Assumed Liabilities).

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 2**0:**2**2:25 Desc Exhibit **A** Page **29** of **2**05

- 8. Those holders of Interests or claims who did not object (or who ultimately withdrew their objections, if any) to the Lonafarnib/Lambda Sale Transactions are deemed to have consented pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests or claims who did object that have an interest in the Transferred Assets fall within one or more of sections 363(f)(1), 363(f)(3), 363(f)(4), or 363(f)(5) of the Bankruptcy Code and are therefore adequately protected by having their Interests or claims that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Lonafarnib/Lambda Sale Transactions ultimately attributable to the property in which they have an interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Lonafarnib/Lambda Sale Transactions, subject to any defenses of the Debtors.
- 9. Release of Interests. Any and all Persons that have filed a financing statement, mortgage, mechanic's lien, *lis pendens*, or other document or agreement evidencing an Interest against or in the Transferred Assets shall deliver to the Debtors prior to the Closing, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, releases, and/or any other similar documents necessary for the purpose of documenting all Interests that such Person has against or in the Transferred Assets. For any Person who has not delivered such termination statements to the Debtors prior to the Closing, (i) the Debtors and/or the Purchaser are hereby authorized to execute and file such statements, instruments, releases, and/or other similar documents on behalf of such Person with respect to the Transferred Assets, (ii) the Purchaser is hereby authorized to file, register, or otherwise record a certified copy of this Lonafarnib/Lambda Sale Order that, once filed, registered, or otherwise recorded, shall constitute conclusive evidence of the release of all Interests of any kind or nature against or in the Transferred Assets, and (iii) the Purchaser may seek in this Court, or any other

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 60 of 205

court of appropriate jurisdiction, to compel the appropriate parties to execute termination statements, instruments of satisfaction, releases, and/or other similar documents with respect to all Interests that such Person has against or in the Transferred Assets. This Lonafarnib/Lambda Sale Order is deemed to be in recordable form sufficient to be placed in the filing or recording system of each and every federal, state, or local government agency, department, or office. Notwithstanding the foregoing, the provisions of this Lonafarnib/Lambda Sale Order authorizing the sale and assignment of the Transferred Assets free and clear of all Interests shall be self-executing, and neither the Debtors nor the Purchaser shall be required to execute or file releases, termination statements, assignments, consents, or other instruments in order to effectuate, consummate, and implement the provisions of this Lonafarnib/Lambda Sale Order.

- 10. <u>Surrender of Transferred Assets</u>. All Persons that are presently or on the Closing Date may be in possession of some or all of the Transferred Assets are directed to surrender possession of such Transferred Assets to the Purchaser as of the Closing Date.
- Closing Date, to operate under any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the terms of the Lonafarnib/Lambda APAs), and all such licenses, permits, registrations, and governmental authorizations or any other approvals are deemed to have been, and hereby are, directed to be transferred to the Purchaser as of the Closing Date. All existing licenses or permits applicable to the business shall remain active, in place, and, as applicable, shall be renewed for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred in accordance with applicable administrative procedures. To the maximum extent permitted by section 525(a) of the Bankruptcy Code, no governmental unit (as defined in

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 65 of 205

Bankruptcy Code § 101(27)) or any representative thereof may revoke or suspend, or in any way challenge or fail to consent to any renewal of any permit or license relating to the operation of the Transferred Assets because of the filing or pendency of the Debtors' chapter 11 cases or the consummation of the Lonafarnib/Lambda Sale Transactions.

12. **Injunction.** All Persons are hereby prohibited and enjoined from taking any action that would adversely affect or interfere with, or that would be inconsistent with, the ability of the Debtors to sell and transfer the Transferred Assets to the Purchaser in accordance with the terms of the Lonafarnib/Lambda APAs, the Transaction Documents, or this Lonafarnib/Lambda Sale Order. Except as expressly permitted by the Lonafarnib/Lambda APAs with respect to Assumed Liabilities or this Lonafarnib/Lambda Sale Order, all Persons (and their respective successors and assigns), including, without limitation, all holders of claims or Interests, lenders, debt security holders, governmental, tax and regulatory authorities, parties to executory contracts and unexpired leases, creditors, contract counterparties, customers, landlords, licensors, employees and former employees, litigation claimants, pension plans, labor unions, trade creditors, and other Persons holding Interests of any kind or nature whatsoever against or in the Debtors or the Transferred Assets (whether known or unknown, legal or equitable, matured or unmatured, contingent or noncontingent, liquidated or unliquidated, asserted or unasserted, whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases, whether imposed by agreement, understanding, law, equity, or otherwise), arising under or out of, in connection with, or in any way relating to, the Debtors, the operation of the Debtors' businesses prior to the Closing, the Transferred Assets, or the transfer of the Transferred Assets to the Purchaser (including, without limitation, any rights or claims based on any successor, transferee, derivative, or vicarious liabilities), shall be and hereby are forever barred, estopped, and permanently enjoined from

asserting, prosecuting, or otherwise pursuing any Interests against the Purchaser, any of its Affiliates, officers, directors, members, partners, principals, or shareholders, any of their respective representatives, successors, designees, or assigns, the property of the foregoing, and the Transferred Assets transferred to the Purchaser or interests of the Debtors in such Transferred Assets (other than the Assumed Liabilities). Following the Closing, no holder of an Interest against the Debtors shall interfere with the Purchaser's title to or use and enjoyment of the Debtors' former interests in the Transferred Assets, including, without limitation, taking any of the following actions with respect to or based on any Interest relating to the Transferred Assets or the transfer of the Transferred Assets to the Purchaser (other than Assumed Liabilities): (a) commencing or continuing in any manner any action or other proceeding against the Purchaser or its successors or assigns, assets or properties; (b) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order against the Purchaser or its successors or assigns, assets, or properties; (c) creating, perfecting, or enforcing any Interest against the Purchaser, its successors or assigns, assets (including the Transferred Assets), or properties; (d) asserting any Interest as a setoff, right of subrogation, or recoupment of any kind against any obligation due Purchaser or its successors or assigns: (e) commencing or continuing any action in any manner or place that does not comply or is inconsistent with the provisions of this Lonafarnib/Lambda Sale Order or the agreements or actions contemplated or taken in respect thereof; (f) interfering with, preventing, restricting, prohibiting, or otherwise enjoining the consummation of the Sale Transactions; or (g) enforcing any provision of any Assigned Contract that prohibits, restricts or conditions, or which purports to terminate or modify, or permits a party other than the Debtors to terminate or modify, any such Assigned Contract, or any right or obligation under such Assigned Contract, because of the assumption and assignment of such Assigned Contract by the Debtors to the

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 63 of 205

Purchaser. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Lonafarnib/Lambda Sale Order, any provision in an Assigned Contract, any other document, or any applicable law that purports to prohibit, restrict, modify or otherwise impair assignment of the Assigned Contracts or the Purchaser's ability to utilize the Transferred Assets in Purchaser's business is hereby void and of no force and effect with respect to the Lonafarnib/Lambda Sale Transactions, including without limitation any provision that (a) terminates or modifies any right or obligation of the Purchaser under such Assigned Contract; (b) cross-defaults to or from any other lease or executory contract that is not an Assigned Contract; (c) contains operating covenants or "go-dark" provisions that would purport to terminate or modify any Assigned Contract before assumption and assignment to the Purchaser; (d) requires a third party's consent prior to assignment of the Assigned Contract to the Purchaser; or (e) restricts the Purchaser's use or assignment of any licenses or similar permits if transferred.

- General Assignment. As of the Closing, this Lonafarnib/Lambda Sale Order shall be construed and shall constitute for any and all purposes a full and complete general assignment, conveyance, and transfer of the Transferred Assets and/or a bill of sale or assignment transferring indefeasible title and interest in the Transferred Assets, including the Assigned Contracts, to the Purchaser. Each and every federal, state, and local governmental agency or department is hereby authorized and directed to accept any and all documents and instruments necessary and appropriate to consummate the Lonafarnib/Lambda Sale Transactions and to reflect the effectiveness of the Lonafarnib/Lambda Sale Transactions.
- 14. **No Successor, Transferee, or Similar Liability**. The Purchaser, its Affiliates, and any of their respective officers, directors, members, partners, principals, employees, independent

contractors, and shareholders (or equivalent) and any of their respective representatives, agents, predecessors, successors, or assigns shall not be and shall not be deemed, as a result of the consummation of the Lonafarnib/Lambda Sale Transactions or otherwise, (i) to be a successor of, successor employer of, successor entity of, to have successorship obligations relating to, or to otherwise be deemed a successor, to the Debtors or the Debtors' estates, including with respect to any labor, employment, employee, personnel, or worker related matter, law, or agreement, including any collective bargaining agreement, works council agreement, union agreement, area labor agreement, multiemployer agreement, project labor agreement, construction agreement, contractor agreement, building agreement, regional agreement, work standards agreement, or other labor Contract (collectively, a "Collective Bargaining Agreement"), any employee benefit plans, any defined benefit pension plan, or any multiemployer plans, and the Purchaser and/or its Affiliates, as applicable, shall instead be, and be deemed to be, a new employer, including with respect to, among other things, any and all federal or state unemployment laws, including the Fair Labor Standards Act, any employee wage and hour law, privacy law, worker classification law, minimum wage law, overtime law, compensation or benefit law, meal or rest break law, time keeping law, employee record or documentation law, workers compensation law, unemployment compensation or tax law, or any other similar federal or state law (provided that the Purchaser shall pay employee-related liabilities solely to the extent expressly included in the Assumed Liabilities); (ii) to have any common law successorship liability in relation to any Collective Bargaining Agreement, union, multiemployer organization, employee benefit plan, or multiemployer plan, including with respect to withdrawal liability or contribution obligations; (iii) to have, de facto or otherwise, merged or consolidated with or into any of the Debtors or any of the Debtors' estates, (iv) to be the successor of or a successor employer (as defined under COBRA and applicable

regulations thereunder, common law, or otherwise) to the Debtors; (v) to have a common identity with the Debtors; (vi) to be an alter ego, joint employer, single employer, a continuation or substantial continuation, or to be holding itself out as a mere continuation, of any of the Debtors or their respective estates, or any enterprise of any of the Debtors, (vii) to be liable for any acts or omissions of the Seller or any of the other Debtors in connection with any Collective Bargaining Agreement, personnel, worker, employee, independent contractor, the conduct of the business, or the operation, funding, or administration of the employee benefit plans or multiemployer plans or arising under or related to the Transferred Assets other than as expressly set forth in the Lonafarnib/Lambda APAs; (viii) to have any successor liability, transferee liability, derivative liability, vicarious liability, or any similar theories of any kind or character including, without limitation, under any theory of foreign, federal, state, or local antitrust, environmental, successor, tax, ERISA, assignee or transferee liability, labor, product liability, employment, de facto merger, substantial continuity, or other law, rule, regulation, or doctrine, whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated; (ix) except as expressly set forth in the Lonafarnib/Lambda APAs, to have any successor liability, transferee liability, derivative, liability, vicarious liability, for any similar theories of any kind or character including under any pending, threatened, or potential claim, litigation, arbitration, settlement, investigation, fact circumstance, or event disclosed in the Transaction Documents; in each case whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated, except to the extent solely and expressly provided for in the Lonafarnib/Lambda APAs. The Purchaser shall not assume, or be deemed to assume, or in any way be responsible for any liability or obligation of any of the Debtors and/or their respective

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 36 of 205

estates, or any of their predecessors or Affiliates. The so-called "bulk sales," "bulk transfer," or other similar laws shall be waived in all necessary jurisdictions, including those relating to Taxes. Except as expressly set forth in the Lonafarnib/Lambda APAs with respect to Assumed Liabilities, the Purchaser, its Affiliates, officers, directors, members, partners, principals, and shareholders (or equivalent) and any of their respective representatives, successors, or assigns, or the Transferred Assets shall have no liability or responsibility whatsoever with respect to, or be required to satisfy in any manner, whether at law or in equity, whether by payment, setoff or otherwise, directly or indirectly (w) any Interest against the Debtors or against an insider of the Debtors, (x) any Interest or Excluded Liabilities, (y) the Debtors except as expressly set forth in the Lonafarnib/Lambda APAs and the Transaction Documents.

- 15. Good Faith of the Purchaser. The Lonafarnib/Lambda Sale Transactions specified in the Lonafarnib/Lambda APAs are undertaken by the Purchaser without collusion and in good faith, as that term is defined in section 363(m) of the Bankruptcy Code, and, accordingly, neither the reversal nor modification on appeal of the authorization provided in this Lonafarnib/Lambda Sale Order to consummate the sale shall affect the validity of the Lonafarnib/Lambda Sale Transactions, including, without limitation, the assumption and assignment of the Assigned Contracts, unless such authorization and consummation of the sale are duly and properly stayed pending such appeal. The Purchaser is a good faith purchaser within the meaning of section 363(m) of the Bankruptcy Code and, as such, is entitled to the full protections of section 363(m) of the Bankruptcy Code.
- 16. No Avoidance of the Lonafarnib/Lambda APAs. Neither the Debtors nor the Purchaser have engaged in any conduct that would cause or permit the Lonafarnib/Lambda APAs to be avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit & Page 37 of **2**05

Code. Accordingly, the Lonafarnib/Lambda APAs and the Lonafarnib/Lambda Sale Transactions shall not be avoidable under section 363(n) of the Bankruptcy Code, and no party shall be entitled to any damages or other recovery pursuant to section 363(n) of the Bankruptcy Code in respect of the Lonafarnib/Lambda APAs or the Lonafarnib/Lambda Sale Transactions. Specifically, the Purchaser has not acted in a collusive manner with any person or entity and the Purchase Price was not controlled by any agreement among bidders.

17. Payment of Cure Amounts and Cure Dispute Resolution. All defaults or other obligations of the Debtors under the Assigned Contracts arising prior to the Closing (without giving effect to any acceleration clauses or any default provisions of the kind specified in section 365(b)(2) of the Bankruptcy Code) as to which no objections were interposed, or as to which an objection was interposed but which do not remain pending as of the date of this Lonafarnib/Lambda Sale Order, are deemed satisfied by the payment of the proposed amount necessary, if any, to cure all monetary defaults, if any, under such Assigned Contract in those amounts set forth in the Assignment Notice, and which were satisfied, or shall be satisfied as soon as practicable. For all Assigned Contracts listed on the Assignment Notice for which the Cure Notice or Amended Cure Notice was served, the Purchaser is authorized and directed to pay all Purchaser Cure Amounts, and the Seller is authorized and directed to pay all Seller Cure Amounts, as soon as practicable after the Closing. Any non-Debtor counterparty to an Assigned Contract that has not filed an objection on or before the deadline as set forth in the relevant Cure Notice or Amended Cure Notice, or received an informal extension by the Debtors, shall be barred from objecting or asserting monetary or non-monetary defaults with respect to any such Assigned Contract other than the applicable amount set forth in the Assignment Notice, and such Assigned

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 38 of 205

Contract shall be deemed assumed by the Debtors and assigned to the Purchaser on the Closing Date.

- 18. With respect to the Assigned Contracts, subject to the terms of the Lonafarnib/Lambda APAs, and subject to the entry of this Lonafarnib/Lambda Sale Order, Purchaser shall make provision for the payment of the Purchaser Cure Amounts, and Seller shall make provision for the payment of the Seller Cure Amounts, in cash at Closing. The Purchaser's promise to perform the obligations under the Assigned Contracts arising after their assumption and assignment to the Purchaser shall constitute adequate assurance of future performance within the meaning of sections 365(b) and 365(f)(2) of the Bankruptcy Code. On the Closing Date, subject in all respects to the terms of this Lonafarnib/Lambda Sale Order, the Purchaser shall be deemed to be substituted for the Seller (and/or any other Debtor, to the extent any of them hold any rights, title, or interests in any of the Assigned Contracts) as a party to the applicable Assigned Contracts.
- 19. In the event of an objection by a Contract counterparty to the Cure Amount with regard to any Contract (such contract, a "Disputed Contract"), such Disputed Contract may be conditionally assumed and assigned, with the consent of the Purchaser, pending the entry of a Disputed Contract Order (as defined below). In the event a Disputed Contract remains unresolved as of the Closing Date, Seller shall either settle the objection of such party or shall litigate such objection under procedures as established by the Bankruptcy Court. In no event shall the Seller settle a Cure Amount objection with regard to any potential Assigned Contract without the express written consent (such consent not to be unreasonably withheld) of Purchaser (with an email consent being sufficient). Upon entry of an Order of the Bankruptcy Court (if necessary) determining any Cure Amount and authorizing the assumption and assignment to Purchaser of such Disputed Contract after the Closing, which order shall be in form and substance acceptable

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 39 of 205

to Purchaser (a "Disputed Contract Order"), Purchaser shall have the option to designate the Disputed Contract as an Assigned Contract or an Excluded Contract (regardless of whether such contract was identified on the Contracts List). If Purchaser elects to designate the Disputed Contract as an Excluded Contract, (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract. If Purchaser elects to designate the Disputed Contract as an Assigned Contract, such Disputed Contract shall be deemed an Assigned Contract for all purposes hereunder and, for the avoidance of doubt, Purchaser shall assume the Disputed Contract and shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Disputed Contract and (if applicable) Seller shall be responsible for paying all related Seller Cure Amounts, which such Cure Amount shall be made as soon as practicable after the Purchaser elects to assume the Disputed Contract. If Purchaser does not designate such Disputed Contract as either an Excluded Contract or an Assigned Contract within five (5) Business Days after the date of the Disputed Contract Order (or such later date as agreed by the Seller and Purchaser), (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under this Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract.

20. <u>Determination of Cure Amounts</u>. Unless a counterparty to any Assigned Contract has filed a timely Cure Objection which remains subject to an unresolved Cure Dispute as of the entry of this Lonafarnib/Lambda Sale Order, the Cure Amounts set forth on the Assignment Notice shall constitute findings of this Court and shall be final and binding on the

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 74 of 205

counterparties to the Assigned Contracts and their successors and designees upon the Closing and shall not be subject to further dispute or audit based on performance prior to the time of assumption and assignment, irrespective of the terms and conditions of such Assigned Contracts. Each counterparty to an Assigned Contract (other than a counterparty who filed a timely Cure Objection) shall be forever barred, estopped, and permanently enjoined from (i) asserting against the Purchaser or its property (including, without limitation, the Transferred Assets), any default arising prior to or existing as of the Closing, or any counterclaim, defense, recoupment, setoff, or any other Interest asserted or assertable against the Debtors (except as otherwise provided herein), and (ii) imposing or charging against the Purchaser or its Affiliates, any accelerations, assignment fees, increases, or any other fees or charges as a result of the Debtors' assumption and assignment to the Purchaser of the Assigned Contracts in connection with the Lonafarnib/Lambda Sale Transactions approved by this Lonafarnib/Lambda Sale Order. To the extent a counterparty to any of the Assigned Contracts received notice of the Debtors' proposed Cure Amount and fails to file a Cure Objection by the applicable deadline, such party shall be deemed to have (a) consented to the assumption and assignment of the applicable Assigned Contract and the payment of the Cure Amount provided in the Assignment Notices and (b) waived any right to assert or collect any other cure amount or enforce any default that may arise or have arisen prior to or as of the Closing.

21. <u>Cross-Over Contracts</u>. From the Agreement Date until the Plan Consummation Date, the Debtors shall not, and shall cause its Affiliates not, to reject, amend, modify, sell, assign, license, transfer, convey, deliver or otherwise divest its interests in the Cross-Over Contracts in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's rights or obligations under the Lonafarnib APA, or Purchaser's ability to Develop or Commercialize any Lonafarnib Antiviral Products.

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 45 of 205

- 22. Except for those Cross-Over Contracts rejected, transferred, assigned or terminated by the Debtors without violating Paragraph 21 of this Lonafarnib/Lambda Sale Order, the Debtors shall, upon Purchaser's written request, transfer and assign, and hereby transfers and assigns, automatically and without further notice, to the Purchaser, each other Cross-Over Contract, effective on the date that is the earliest to occur of (a) the date that each and every Cross-Over Contract Benefited Party of such Cross-Over Contract obtains (i) a new agreement with the applicable counterparty of such Cross-Over Contract for substantially the same services as those then being provided to Seller by such counterparty under such Cross-Over Contract, or (ii) an agreement with a Third Party such that such services then being provided under such Cross-Over Contract to such Cross-Over Benefited Party are no longer needed by such Cross-Over Benefited Party, (b) the date Purchaser and all Cross-Over Contract Benefited Parties of such Cross-Over Contract agree to such transfer and assignment of such Cross-Over Contract, and (c) the date all Cross-Over Benefited Parties are no longer receiving any services under such Cross-Over Contract; and upon such transfer and assignment, such Cross-Over Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib APA. Purchaser shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Cross-Over Contract, and (if applicable) Seller shall be responsible for paying all associated Seller Cure Amounts.
- 23. Notwithstanding the foregoing Paragraphs 21 and 22, (x) the IQVIA Contracts shall be Assigned Contracts upon the occurrence of the Satisfactory IQVIA Cure Resolution, and (y) the Cross-Over Contracts that are not IQVIA Contracts (the "Other Cross-Over Contracts") shall be Assigned Contracts upon the occurrence of the Satisfactory Other Cure Resolution, provided that if the Satisfactory IQVIA Cure Resolution does not occur by the Plan Consummation Date, the IQVIA Contracts shall be Excluded Contracts, and if the Satisfactory Other Cure Resolution does

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 72 of 205

not occur by the Plan Consummation Date, the Other Cross-Over Contracts shall be Excluded Contracts. In the event that a Cross-Over Contract becomes an Excluded Contract, Seller shall use commercially reasonable efforts to preserve the Transferred Data, including the Global Safety Databases, and fully transfer and transition the Transferred Data and Transferred Regulatory Information to Purchaser, and shall not instruct the counterparties to the IQVIA Contracts to delete or remove the Transferred Data from the Global Safety Databases.

24. **Previously Unknown Contracts.** If at any time, prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, it is discovered that a Contract material to the operation of the Business should have been identified on the Assumption Notice but was not so listed (any such Contract, a "Previously Unknown Contract"), Seller shall, promptly following the discovery thereof (but in no event later than five (5) Business Days following the discovery thereof), notify Purchaser in writing of such Previously Unknown Contract and provide Purchaser with a copy of such Previously Unknown Contract and the Cure Amount (if any) in respect thereof. Purchaser shall thereafter deliver written notice to Seller (email being sufficient), no later than ten (10) Business Days following such notice of such Previously Unknown Contract from Seller, if Purchaser elects for such Previously Unknown Contract to be an Assigned Contract. If Purchaser elects for a Previously Unknown Contract to be an Assigned Contract in accordance with this Paragraph 24, then to the extent not previously filed and served, Seller shall file and serve an assignment and assumption notice on the Contract counterparty to such Previously Unknown Contract (a "Supplemental Assignment Notice") notifying such Contract counterparty of Seller's intention to assume and assign to Purchaser such Previously Unknown Contract, including the proposed Cure Amount (if any). Such notice shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 73 of 205

assignment of the Contract to Purchaser (the "Supplemental Assignment Notice Objection Deadline"). Following expiration of the Supplemental Assignment Notice Objection Deadline and, if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib/Lambda APAs and this Lonafarnib/Lambda Sale Order. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under the Lonafarnib/Lambda APAs.

25. Previously Excluded Contract. At any time prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, Purchaser may elect to take an assignment of any Excluded Contract that has not yet been assumed and assigned pursuant to an order of the Bankruptcy Court (a "Previously Excluded Contract") by sending a written notice to Seller (email being sufficient) of such election. If Purchaser elects for a Previously Excluded Contract to be an Assigned Contract in accordance with this Paragraph 25, then to the extent not previously filed and served, Seller shall file and serve a Supplemental Assignment Notice on the Contract counterparty to such Previously Excluded Contract. Such Supplemental Assignment Notice Objection Deadline shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment Notice Objection Deadline and if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of

Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 78 of 205

such an order, such Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib/Lambda APAs and this Lonafarnib/Lambda Sale Order, and, subject to paragraphs 21-23 of this Lonafarnib/Lambda Sale Order and Section 7.15 of the Lonafarnib APA with respect to Cross-Over Contracts, the Purchaser shall be responsible for satisfying or paying any Cure Amounts or other Liabilities with respect to such Contract, whether or not such Cure Amounts or other Liabilities exceed the Purchaser Cure Amounts. For the avoidance of doubt, the Cross-Over Contracts are not Previously Excluded Contracts. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under the Lonafarnib/Lambda APAs.

- 26. <u>Purchaser Protections.</u> The Purchaser shall be entitled to the Purchaser Protections under the Lambda APA and the Lonafarnib APA, as applicable.
- 27. Ipso Facto Clauses Ineffective. Upon the Debtors' assumption and assignment of the Assigned Contracts to the Purchaser pursuant to this Lonafarnib/Lambda Sale Order and the payment of the Cure Amounts in accordance with this Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs, no default shall exist under any Assigned Contract and no counterparty to any such Assigned Contract shall be permitted to declare or enforce a default by the Debtors or the Purchaser thereunder or otherwise take action against the Purchaser as a result of any Debtor's financial condition, change in control, bankruptcy, or failure to perform any of its obligations under the applicable Assigned Contract. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Lonafarnib/Lambda Sale Order, any provision in a Assigned Contract that prohibits or conditions, whether directly or indirectly, the assignment of such Assigned Contract (including, without limitation, the granting of an Interest therein) or allows the counterparty thereto to terminate, recapture, impose any

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 2**0:**2**2:25 Desc Exhibit A Page **79** of **2**05

penalty, condition on renewal or extension, or modify any term or condition upon such assignment shall be deemed an unenforceable anti-assignment provision that is void and of no force and effect with respect to the Lonafarnib/Lambda Sale Transactions as approved by this Lonafarnib/Lambda Sale Order. The failure of the Debtors or the Purchaser to enforce at any time one or more terms or conditions of any Assigned Contract shall not be a waiver of such terms or conditions or of the Debtors' or the Purchaser's right, as applicable, to enforce every term and condition of such Assigned Contract.

28. Binding Effect. This Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs shall be binding upon and shall govern the acts of all entities, including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other Persons who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to any of the Transferred Assets. The terms and provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, and this Lonafarnib/Lambda Sale Order shall be binding in all respects upon the Debtors and their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Assigned Contracts, the Purchaser and its respective successors and assigns, and any and all third parties, notwithstanding any subsequent appointment of any trustee, examiners, "responsible persons" or other fiduciaries (collectively, the "Trustee") of the Debtors under any chapter of the Bankruptcy Code, as to which Trustee such Case 24-80040-sgj11 Doc 898-6 Filed 08/05/25 Entered 08/05/25 20:22:05 Desc Exhibit & Page 86 of 205

terms and provisions likewise shall be binding, and the Lonafarnib/Lambda APAs (including the Assigned Contracts) shall not be subject to rejection or avoidance under any circumstances.

- 29. Release, Discharge, and Termination of Interests. This Lonafarnib/Lambda Sale Order shall be effective as a determination that, on the Closing, all Interests of any kind or nature whatsoever existing prior to the Closing have been unconditionally released, discharged, and terminated solely as to the Transferred Assets (other than the Assumed Liabilities), and that the conveyances described herein have been effected.
- 30. No Material Modifications. The Lonafarnib/Lambda APAs and the Transaction Documents may be modified, amended, or supplemented by the Debtors and the Purchaser, in a writing signed by such parties, and in accordance with the terms thereof, without further order of this Court; provided, that (i) any such modification, amendment, or supplement does not have a material adverse effect on the Debtors' estates or its creditors, and (ii) has been agreed to between the Seller and the Purchaser (with respect to the Seller, such consent not to be unreasonably withheld) and approved by the Prepetition Term Loan Administrative Agent. Any material modification, amendment, or supplement to the Lonafarnib/Lambda APAs and the Transaction Documents adversely affecting the Debtors' estates must be filed on the docket and served on all interested parties. Interested parties shall have five business days to file an objection to any such material modification, amendment, or supplement. If no objections are received within five business days or the Court overrules such filed objections, the modified Lonafarnib/Lambda APAs and Transaction Documents shall be effective.
- 31. <u>Subsequent Orders and Plan Provisions</u>. Nothing contained in any chapter 11 plan confirmed in the Debtors' chapter 11 cases or any subsequent order of this Court, including, without limitation, any order confirming any such chapter 11 plan, any order authorizing

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 87 of 205

the sale of assets of the Debtors pursuant to any section of the Bankruptcy Code, and any order approving wind-down or dismissal of any Debtor's chapter 11 case or any subsequent chapter 7 case shall change, supersede, abrogate, nullify, restrict, or conflict with the provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, or this Lonafarnib/Lambda Sale Order, or in any way prevent or interfere with the consummation or performance of the Lonafarnib/Lambda Sale Transactions.

- 32. **Failure to Specify Provisions**. The failure to specify or include any particular provisions of the Lonafarnib/Lambda APAs or the Transaction Documents in this Lonafarnib/Lambda Sale Order shall not diminish or impair the effectiveness of such provisions, it being the intent of this Court that the Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions be authorized and approved in their entirety.
- 33. <u>Automatic Stay</u>. The automatic stay pursuant to section 362 of the Bankruptcy Code is hereby lifted solely to the extent necessary to (i) allow the Purchaser to deliver any notice provided for in the Lonafarnib/Lambda APAs and the Transaction Documents, and (ii) allow the Purchaser to take any and all actions permitted under the Lonafarnib/Lambda APAs and the Transaction Documents in accordance with the terms and conditions thereof. The automatic stay imposed by section 362 of the Bankruptcy Code shall be modified solely to the extent necessary to implement the preceding sentence, and this Court shall retain exclusive jurisdiction over any and all disputes with respect thereto.
- Bankruptcy Rules Satisfied or Waived. The requirements set forth in Bankruptcy Rules 6004 and 6006 have been satisfied or are otherwise deemed to be waived. As provided by Bankruptcy Rule 9014, the terms of this Lonafarnib/Lambda Sale Order shall be effective and enforceable immediately upon entry and shall not be subject to stay provisions

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 82 of 205

contained in Bankruptcy Rules 6004(h) and 6004(d). Time is of the essence in closing the Lonafarnib/Lambda Sale Transactions and the Debtors and the Purchaser intend to close the sale as soon as possible.

- Lonafarnib/Lambda APAs. To the extent anything contained in this Lonafarnib/Lambda Sale Order conflicts with a provision in the Lonafarnib/Lambda APAs or Transaction Documents, this Lonafarnib/Lambda Sale Order shall govern and control. Notwithstanding the foregoing, nothing in this Lonafarnib/Lambda Sale Order shall modify or waive any closing conditions or termination rights in the Lonafarnib/Lambda APAs, and all such conditions and rights shall remain in full force and effect in accordance with their terms.
- 36. <u>Provisions Nonseverable and Mutually Dependent</u>. The provisions of this Lonafarnib/Lambda Sale Order, the Lonafarnib/Lambda APAs, and the Transaction Documents are non-severable and mutually dependent.
- 37. **Retention of Jurisdiction**. This Court shall retain exclusive jurisdiction to, among other things, interpret, implement, and enforce the terms and provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, and this Lonafarnib/Lambda Sale Order, and each of the agreements executed in connection therewith to which the Debtors are a party or which has been assigned to the Purchaser by the Debtors, and to adjudicate, if necessary, any and all disputes concerning or relating in any way to the Lonafarnib/Lambda Sale Transactions. This Court retains jurisdiction to compel delivery of the Transferred Assets, to protect the Purchaser and its assets, including the Transferred Assets, against any Interests or successor or transferee liability and to enter orders, as appropriate, pursuant to sections 105(a), 363, or 365 (or other applicable sections) of the Bankruptcy Code necessary to transfer the Transferred Assets and the Assigned Contracts

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 89 of 205

to the Purchaser. In the event this Court abstains from exercising or declines to exercise jurisdiction with respect to any matter referenced in this paragraph or is without jurisdiction, such abstention, refusal, or lack of jurisdiction shall have no effect upon and shall not control, prohibit, or limit the exercise of jurisdiction of any other court having competent jurisdiction with respect to any such matter.

- 38. The Purchaser has standing to seek to enforce any terms of this Lonafarnib/Lambda Sale Order, the Lonafarnib/Lambda APAs, and the Transaction Documents in this Court or any other court with competent jurisdiction.
- 39. All time periods set forth in this Lonafarnib/Lambda Sale Order shall be calculated in accordance with Bankruptcy Rule 9006(a).

Submitted By:

SIDLEY AUSTIN LLP

Thomas Califano (TX Bar No. 24122825) William Curtin (admitted *pro hac vice*) Anne Wallice (admitted *pro hac vice*) 787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Charles Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000 Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Exhibit 1

Lonafarnib APA

Execution Version

LONAFARNIB ASSET PURCHASE AGREEMENT

by and between

EIGER INNOTHERAPEUTICS, INC., as Purchaser,

and

EIGER BIOPHARMACEUTICALS, INC., as Seller

Dated as of August 1, 2024

Table of Contents

| ARTICLE 1. | DEFINED TERMS | |
|------------|--|----|
| 1.1 | Defined Terms | |
| 1.2 | Other Definitional and Interpretive Matters | |
| ARTICLE 2. | THE PURCHASE AND SALE; CLOSING | |
| 2.1 | Purchase and Sale | |
| 2.2 | Excluded Assets | |
| 2.3 | Assumption of Liabilities | 18 |
| 2.4 | Excluded Liabilities | |
| 2.5 | Excluded Contracts | |
| 2.6 | Nontransferable Assets and Liabilities | |
| 2.7 | Closing | 19 |
| 2.8 | Closing Deliveries of the Parties | |
| 2.9 | Purchase Price; Assumed Liabilities; Deposits | |
| 2.10 | Transfer Taxes | |
| 2.11 | Allocation of Purchase Price | |
| 2.12 | Escrow Accounts | |
| 2.13 | Tax Withholding | |
| ARTICLE 3. | REPRESENTATIONS AND WARRANTIES OF THE SELLER | |
| 3.1 | Organization, Good Standing and Other Matters | 22 |
| 3.2 | Authority and Enforceability | |
| 3.3 | No Conflict; Required Filings and Consents | |
| 3.4 | Compliance With Laws | |
| 3.5 | Permits | |
| 3.6 | Litigation | |
| 3.7 | Real Property | |
| 3.8 | Assigned Contracts | |
| 3.9 | Financial Statements | |
| 3.10 | Absence of Material Developments. | |
| 3.11 | Customers and Suppliers. | |
| 3.12 | Intellectual Property. | |
| 3.13 | Taxes | |
| 3.14 | Product Liability | |
| 3.15 | Product Warranties; Product Return | |
| 3.16 | Brokers and Finders | |
| 3.17 | Virology Studies | |
| 3.18 | No Other Representations or Warranties | |
| ARTICLE 4. | REPRESENTATIONS AND WARRANTIES OF PURCHASER | |
| 4.1 | Organization, Good Standing and Other Matters | |
| 4.2 | Authority and Enforceability | |
| 4.3 | No Conflict: Required Filings and Consents | |
| 4.4 | Financing | |
| 4.5 | Solvency | |
| 4.6 | Litigation | |
| 4.7 | Brokers and Finders | |
| 4.8 | Non-Reliance of Purchaser; No Other Representations and Warranties | |
| 4.9 | No Other Representations or Warranties | |
| ARTICLE 5. | BANKRUPTCY COURT MATTERS | |
| 5.1 | Competing Transaction | |
| 5.2 | Bankruptcy Court Filings | 29 |

| 5.3 | Assumption of Assigned Contracts | 30 |
|-------------|---|----|
| 5.4 | Disputed Contracts. | |
| 5.5 | Previously Unknown and Previously Excluded Contracts | 31 |
| ARTICLE 6. | PRE-CLOSING COVENANTS | 32 |
| 6.1 | Conduct of Business | 32 |
| 6.2 | Access to Information; Confidentiality | 34 |
| 6.3 | Efforts to Consummate | |
| 6.4 | Notices and Consents | 35 |
| 6.5 | Regulatory Matters | |
| 6.6 | Public Announcements | |
| 6.7 | Update of Schedules; Knowledge of Breach | 36 |
| ARTICLE 7. | POST-CLOSING COVENANTS | |
| 7.1 | Access to Information; Books and Records | 36 |
| 7.2 | Post-Closing Receipt and Possession of Assets | |
| 7.3 | Tax Matters | |
| 7.4 | Wrong Pockets. | |
| 7.5 | Purchased Intellectual Property and Purchased Product Information | 38 |
| ARTICLE 8. | CONDITIONS PRECEDENT | 43 |
| 8.1 | Conditions to Each Party's Obligation | |
| 8.2 | Conditions to Obligations of Purchaser | |
| 8.3 | Conditions to Obligations of the Seller | |
| 8.4 | Waiver of Condition; Frustration of Conditions | |
| 8.5 | Delivery of a Notice of Readiness to Close | |
| ARTICLE 9. | TERMINATION | |
| 9.1 | Events of Termination | |
| 9.2 | Effect of Termination | 45 |
| 9.3 | Termination Fee and Expense Reimbursement | 46 |
| ARTICLE 10. | GENERAL PROVISIONS | |
| 10.1 | Survival of Representations, Warranties and Covenants | 47 |
| 10.2 | Entire Agreement | 47 |
| 10.3 | Amendment; No Waiver | 47 |
| 10.4 | Severability; Specific Versus General Provisions | 47 |
| 10.5 | Expenses and Obligations | 48 |
| 10.6 | Notices | 48 |
| 10.7 | Counterparts | 49 |
| 10.8 | Governing Law | 49 |
| 10.9 | Submission to Jurisdiction; Consent to Service of Process | 49 |
| 10.10 | Waiver of Jury Trial | 49 |
| 10.11 | Rights Cumulative | 50 |
| 10.12 | Assignment | 50 |
| 10.13 | Specific Enforcement; Remedies | 50 |
| 10.14 | Third-Party Beneficiaries | 50 |
| 10.15 | No Personal Liability of Directors, Officers and Owners | |
| 10.16 | General Release | |
| 10.17 | Legal Representation | |
| | | |

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 89 of 205

EXHIBITS

Exhibit A Form of Bill of Sale and Assignment and Assumption Agreement

Exhibit B Escrow Agreement

Exhibit C Form of Intellectual Property Assignment Agreement

SCHEDULES

| Schedule 1.1(a) | Permitted Liens |
|-----------------|------------------------------------|
| Schedule 2.1(a) | Assigned Contracts |
| Schedule 2.1(c) | Transferred Regulatory Information |
| Schedule 2.1(h) | Raw Materials and Inventory |

Schedule 2.1(h) Raw Materials and Inventory
Schedule 2.11(a) Preliminary Allocation Schedule

Schedule 3.3 Seller Conflict; Required Filings and Consents

Schedule 3.10 Absence of Material Developments

Schedule 3.11 Customers and Suppliers

Schedule 3.12(a) Intellectual Property Registrations
Schedule 3.12(b) Intellectual Property Agreements

Schedule 3.14 Product Liability

Schedule 3.18 Inventory Locations and Storage Contracts

Schedule 4.3 Purchaser Conflict; Required Filings and Consents

Schedule 6.1 Conduct of Business Schedule 6.4 Notices and Consents Schedule 7.6 Transition Activities

Schedule 7.10(d)(1) Pharmacovigilance Services Prior to Lonafarnib IND Transfer Date Schedule 7.10(d)(2) Pharmacovigilance Services After Lonafarnib IND Transfer Date

Schedule 7.15 Cross-Over Contracts

LONAFARNIB ASSET PURCHASE AGREEMENT

THIS LONAFARNIB ASSET PURCHASE AGREEMENT (this "Agreement"), dated as of August 1, 2024 (the "Agreement Date") is entered into by and between Eiger InnoTherapeutics, Inc., a Delaware corporation ("Purchaser") and Eiger BioPharmaceuticals, Inc., a Delaware corporation (the "Seller").

RECITALS

WHEREAS, on April 1, 2024 (the "Petition Date") the Seller and certain of its Affiliates filed voluntary petitions for relief under chapter 11 of Title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the Northern District of Texas (the "Bankruptcy Court"), thereby commencing Chapter 11 cases (collectively, the "Bankruptcy Cases");

WHEREAS, the Seller is a debtor-in-possession under the Bankruptcy Code and manages its properties and assets pursuant to Sections 1107(a) and 1108 of the Bankruptcy Code;

WHEREAS, the Seller is engaged in the Business and owns, directly or indirectly, all of the Transferred Assets;

WHEREAS, the Seller desires to sell (or cause to be sold) to Purchaser, and Purchaser desires to purchase from the Seller, all of the Transferred Assets Free and Clear, and the Seller desires Purchaser to assume, and Purchaser desires to assume from the Seller, all of the Assumed Liabilities, in each case upon the terms and subject to the conditions hereof, pursuant to a Sale Order and Sections 105(a), 363 and 365 of the Bankruptcy Code and Rules 6004 and 6006 of the Federal Rules of Bankruptcy Procedure;

WHEREAS, the transactions contemplated by this Agreement and the Related Documents are subject to approval by the Bankruptcy Court and will only be consummated pursuant, among other things, to the Sale Order to be entered in the Bankruptcy Cases; and

WHEREAS, concurrently with the execution of this Agreement, Purchaser shall deposit (or cause to be deposited) an aggregate amount equal to the Deposit Escrow Amount into an escrow account (the "Deposit Escrow Account") to be established and maintained by Escrow Agent pursuant to the Escrow Agreement.

NOW, **THEREFORE**, in consideration of the premises and the mutual representations, warranties, covenants, agreements and conditions set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE 1. DEFINED TERMS

- 1.1 **Defined Terms**. The following terms shall have the following meanings in this Agreement:
- "Action" means any action, proceeding, arbitration or litigation (whether civil, criminal or administrative) commenced, brought, conducted or heard by or before any Governmental Authority or arbitrator.
 - "AEs" has the meaning set forth in Section 7.10(a).
- "Affiliate" of any particular Person means any other Person directly or indirectly controlling, controlled by or under common control with such particular Person. For purposes of this Agreement, the direct or indirect ownership of fifty percent (50%) or more of the outstanding voting securities of a Person, or the right to receive fifty percent (50%) or more of the profits or earnings of a Person shall be deemed to constitute control. Such other relationship as in fact results in actual control over the management, business and affairs of a Person shall also be deemed to constitute control.
 - "Agreement" has the meaning set forth in the preamble.

- "Agreement Date" has the meaning set forth in the preamble.
- "Allocation Schedule" has the meaning set forth in Section 2.11(a).
- "Alternate Transaction" has the meaning set forth in Section 9.1(b).
- "Applicable Law" means, with respect to any Person, any federal, provincial, state, local law, ordinance, principle of common law, code, regulation or statute applicable to such Person or such Person's subsidiaries or to any of their respective securities, assets, properties or businesses.
- "Asset Taxes" means any Taxes with respect to the ownership or operation of the Transferred Assets other than (a) Taxes based on net or gross income, and (b) Transfer Taxes.
 - "Assigned Contracts" has the meaning set forth in Section 2.1(a).
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - "Assumption Notice" has the meaning set forth in Section 5.3(a).
 - "Attorney-Client Information" has the meaning set forth in Section 10.17.
 - "Auction" has the meaning set forth in Section 5.2(h).
 - "Avexitide Buyer" means Amylyx Pharmaceuticals, Inc.
- "Avoidance Actions" means any and all avoidance, recovery, subordination, or other claims, actions, rights, or remedies that may be brought by or on behalf of the Seller or its estate or other authorized parties in interest under the Bankruptcy Code or applicable non-bankruptcy law, including, but not limited to, actions or remedies under sections 510, 542, 543, 544, 545, and 547 through and including 553 of the Bankruptcy Code.
- "Back-Up Bid" means the second highest or otherwise best bid if the successful bidder fails to consummate its bid in accordance with the Bid Procedures.
- "Back-up Termination Date" means the first to occur of (a) thirty (30) days after the entry of the Sale Order, (b) consummation of the Transactions with the winning bidder at the Auction, and (c) October 1, 2024.
 - "Bankruptcy Cases" has the meaning set forth in the Recitals.
 - "Bankruptcy Code" has the meaning set forth in the Recitals.
 - "Bankruptcy Court" has the meaning set forth in the Recitals.
 - "Base Price" means \$5,200,000.
- "Bid Procedures" means those certain bidding procedures for the sale of the Seller's assets approved by the Bankruptcy Court as filed at Docket No. 119.
- "Bid Procedures Order" means that certain Order entered by the Bankruptcy Court at Docket No. 94 approving the Bid Procedures.
- "Bill of Sale and Assignment and Assumption Agreement" means the bill of sale and assignment and assumption agreement, dated as of the Closing Date, by and between the Seller and Purchaser, in substantially the form attached hereto as Exhibit A and acceptable to Purchaser.
 - "Biorasi Contract" means any Contract with Biorasi LLC.
- "BMS License Agreement" means that certain License Agreement, dated April 20, 2016, between the Seller and Bristol-Myers Squibb Company.

- "Business" means the business as presently conducted of the Seller Group related to the Development, Manufacture, and Commercialization of Lonafarnib Antiviral Products in the Lonafarnib Antiviral Field in the Territory.
- "Business Books and Records" means the records and files relating to any Licensed Product in any field (including the Lonafarnib Antiviral Field and Progeria Field) in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter, including without limitation (i) supplier and vendor lists, (ii) promotional materials, and (iii) other business records required to be transferred to Purchaser under Applicable Law. For clarity, Business Books and Records shall exclude Regulatory Information and Data.
- "Business Day" means any day other than (a) a Saturday, Sunday or federal holiday or (b) a day on which commercial banks in San Francisco, California are authorized or required to be closed.
- "Business Intellectual Property" means all Owned Intellectual Property Assets together with all other Intellectual Property used in, held for use in, or necessary for the conduct of the Business.
 - "Closing" has the meaning set forth in Section 2.7.
 - "Closing Date" has the meaning set forth in Section 2.7.
 - "Code" means the Internal Revenue Code of 1986, as amended, or any successor law.
 - "Commercialization" has the meaning given to it in the Merck License Agreement.
 - "Competing Bid" has the meaning set forth in Section 5.1.
- "Confidentiality Agreement" means that certain Confidentiality Agreement, dated as of April 4, 2024, by and between the Seller and Purchaser.
 - "Consent" means any consent, approval, authorization, waiver or license.
- "Contract" means any written agreement, mortgage, indenture, lease (whether for real or personal property), contract or subcontract.
 - "Contracts List" has the meaning set forth in Section 2.1(a).
 - "Contracting Parties" has the meaning set forth in Section 10.15.
- "Cross-Over Contract Benefited Party" means, with respect to any Cross-Over Contract, the Zokinvy Buyer, the Avexitide Buyer, and/or the Lambda Buyer, as applicable, that benefits, or whose products purchased from Seller or any of its Affiliates benefit, from such Cross-Over Contract.
 - "Cross-Over Contracts" has the meaning set forth in Section 7.15.
- "Cure Amounts" means any and all costs, expenses or actions with respect to defaults existing as of the Petition Date that Purchaser or the Seller, as applicable, are required to pay or perform to assume any of the Assigned Contracts pursuant to section 365(b)(1)(A) and (B) of the Bankruptcy Code or as otherwise agreed between Purchaser or the Seller, as applicable, and the counterparty to an Assigned Contract.
- "Data" means (a) any and all clinical, preclinical, non-clinical, toxicology, chemistry, biology, animal, CMC, safety, and other data, databases, information, batch records, laboratory records, and all other data and information, and (b) any and all global and country safety databases, in each case (a) and (b) that relate to any Licensed Compound or Licensed Product in any field (including the Lonafarnib Antiviral Field and Progeria Field), any other Transferred Asset, or any Assumed Liability that is in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the

Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter.

- "Deposit Escrow Account" has the meaning set forth in the Recitals.
- "Deposit Escrow Amount" means \$260,000.
- "Designation Deadline" has the meaning set forth in Section 5.3(b).
- "Determined Cure Amounts" means all Cure Amounts for Assigned Contracts, as determined by a final order of the Bankruptcy Court.
 - "Development" or "Develop" has the meaning given to it in the Merck License Agreement.
 - "Disputed Contract" has the meaning set forth in Section 5.4.
 - "Disputed Contract Order" has the meaning set forth in Section 5.4.
- "Enforceability Exceptions" means applicable bankruptcy, insolvency, reorganization, moratorium, receivership and similar Applicable Laws affecting the enforcement of creditors' rights generally and general equitable principles.
- "Environmental Laws" means any Applicable Law relating to pollution or protection of the environment or worker health and safety (in respect of exposure to Hazardous Substances), including such Applicable Laws relating to the use, treatment, storage, disposal, Release or transportation of Hazardous Substances.
 - "Escrow Agent" means Kurtzman Carson Consultants LLC.
- "Escrow Agreement" means the escrow agreement by and between the Seller and the Escrow Agent attached hereto as Exhibit B.
 - "Excluded Assets" has the meaning set forth in Section 2.2.
- "Excluded Books and Records" means the following originals and copies of those books and records, documents, data and information (in whatever form maintained) of the Seller Group and the Business: (i) all corporate minute books (and other similar corporate records) and stock records of the Seller Group, (ii) any books and records relating to the Excluded Assets or (iii) any books, records or other materials that any member of the Seller Group (x) is required by Applicable Law to retain (copies of which, to the extent permitted by Applicable Law, will be made available to Purchaser upon Purchaser's reasonable request), (y) reasonably believes is necessary to enable it to prepare and/or file Tax Returns (copies of which will be made available to Purchaser upon Purchaser's reasonable request) or (z) are prohibited by Applicable Law from delivering to Purchaser.
 - "Excluded Contracts" has the meaning set forth in Section 2.5.
 - "Excluded Liabilities" has the meaning set forth in Section 2.4.
- "Existing Manufacturing Contract" means any Assigned Contract under which the Seller or any of its Affiliates Manufactured or has Manufactured any Licensed Compounds or Lonafarnib Antiviral Products, as identified on Schedule 2.1(a).
 - "Existing Manufacturing Contract Interim Term" has the meaning set forth in Section 7.11(a).
- "Existing Manufacturing Contract Transfer Date" means, with respect to an Existing Manufacturing Contract, the date that is the earlier to occur of (a) November 3, 2024, (b) the date that the Zokinvy Buyer obtains a new agreement for substantially the same services as those provided to Seller by the counterparty under such Existing Manufacturing Contract prior to May 3, 2024, and (c) the date

Purchaser and the Zokinvy Buyer agree to arrangements for the supply of Licensed Progeria Product under the Existing Manufacturing Contracts following the assignment thereof to Purchaser.

- "Expense Reimbursement" means the reimbursement by the Seller of Purchaser's actual and reasonable out-of-pocket legal, accounting, and other third-party advisory or service costs and expenses incurred in connection with the Transactions, as evidenced by invoice(s) provided to the Seller, on the terms and subject to the conditions of Section 9.3.
 - "FDA" means the United States Food and Drug Administration.
- "FD&C Act" means the United States Federal Food, Drug and Cosmetic Act, as amended, and any rules, regulations, and requirements promulgated thereunder.
 - "Field" has the meaning given to it in the Merck License Agreement.
- "Final Order" means an Order, judgment or other decree of the Bankruptcy Court or any other Governmental Authority of competent jurisdiction that has not been reversed, vacated, modified or amended, is not stayed and remains in full force and effect; provided that such Order shall be considered a Final Order only after the time period for third parties seeking appeal has expired without the filing of any appeal or motion for reconsideration.
- "Free and Clear" means free and clear of all Liens and Excluded Liabilities (other than the Permitted Liens and the Assumed Liabilities) to the maximum extent permitted by Section 363(f) of the Bankruptcy Code.
- "GAAP" means generally accepted accounting principles in the United States as of the Agreement Date.
- "General Business Books and Records" means, excluding Transferred Business Books and Records and Business Books and Records that exclusively relate to the Licensed Progeria Product, any and all Business Books and Records that relate to any Licensed Product.
- "General Licensed Product Data" means, excluding Transferred Data and Licensed Progeria Product Data, any and all Data that relate to any Licensed Product.
- "General Licensed Product Regulatory Information" means, excluding Transferred Regulatory Information and Licensed Progeria Product Regulatory Information, any and all Regulatory Information that relate to any Licensed Product.
- "Global Safety Databases" means the databases established and owned or controlled (including via license) by Seller or any of its Affiliates, including such databases that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level) that contain the totality of all current and historic safety data and information with respect to the Licensed Product, including AEs, received, collected, used, held for use by or on behalf of Seller or its Affiliates or the Zokinvy Buyer or its Affiliates (including by or on behalf of any contractors or other service providers acting on its or their behalf, directly or indirectly, at any level), or pursuant to the Merck License Agreement or Merck Pharmacovigilance Agreement for drug surveillance, pharmacovigilance, and regulatory safety reporting purposes, including the global safety database that is the central repository of all such safety data and information worldwide and any and all local or territory databases of such safety data and information with respect to a particular country, region, jurisdiction, or territory.
- "Global Safety Database Contracts" means any and all Contracts by and between Seller or any of its Affiliates and a Third Party service provider under which any part of the Global Safety Databases is stored or administered, including the Contracts identified as Global Safety Database Contracts on Schedule 7.15.

- "Goods" has the meaning set forth in Section 3.14.
- "Governmental Authority" means any domestic or foreign national, provincial, state, multi-state or municipal or other local government, any subdivision, agency, commission or authority thereof, any court (including the Bankruptcy Court), tribunal, or any quasi-governmental or private body exercising any regulatory or taxing authority thereunder (including the IRS and the FDA).
- "Hazardous Substances" means any substances, materials or wastes which are defined as or included in the definition of "hazardous substances", "hazardous wastes", "hazardous materials", "toxic substances", "pollutants" or "contaminants" under any Environmental Law, including any petroleum or refined petroleum products, radioactive materials, friable asbestos or polychlorinated biphenyls.
- "IND" means (i) an Investigational New Drug application filed with the FDA in accordance with the FD&C Act, and (ii) any of its foreign equivalents as filed with the applicable Regulatory Authorities in other countries or regulatory jurisdictions in the world, as applicable, in each case ((i) and (ii)), including all supplements, amendments, variations, extensions, and renewals thereof that may be filed with respect to the foregoing.
- "Intellectual Property" means any and all intellectual property and proprietary rights in any jurisdiction throughout the world, including rights arising from the following: (i) patents and patent applications, design rights, industrial design registrations and applications therefor, divisions, continuations, continuations-in-part, reissues, substitutes, renewals, registrations, confirmations, reexaminations, extensions and any provisional applications, and any foreign or international equivalent of any of the foregoing; (ii) trademarks (whether registered, unregistered or applied for), service marks, trade dress, service names, trade names, brand names, product names, slogans, logos, business names, corporate names, and other source or business identifiers, all registrations and applications for registration thereof, and, in each case, together with all of the goodwill associated therewith; (iii) works of authorship, copyrights and all registrations and applications for registration thereof; (iv) trade secrets and Know-How; (v) rights in formulae, methods, techniques, processes, assembly procedures, software, software code (in any form, including source code and executable or object code), subroutines, test results, test vectors, user interfaces, protocols, schematics, specifications, drawings, prototypes, molds and models, and other forms of technology (whether or not embodied in any tangible form and including all tangible embodiments of the foregoing), and (vi) social media accounts, social media identifiers, internet domain name registrations.
- "Intellectual Property Assignment Agreement" means the assignment agreement, dated as of the Closing Date, by and between the Seller and Purchaser, in substantially the form attached hereto as Exhibit C and acceptable to Purchaser.
- "Intellectual Property Registrations" means, as to any Owned Intellectual Property Assets, any issuance, registration, application or other filing by, to or with any Governmental Authority or authorized private registrar in any jurisdiction, including domain names, registered trademarks and copyrights, issued and reissued patents and pending applications for any of the foregoing.
 - "Inventory" has the meaning set forth in Section 2.1(h).
- "IQVIA Contract" means any Contract with IQVIA Biotech LLC, IQVIA Clinical AB, IQVIA RDS INC., or Novella Clinical LLC, or any of their Affiliates.
 - "IRS" means the United States Internal Revenue Service.
 - "Joint Ownership Agreement" has the meaning set forth in Section 7.13.
- "Know-How" means all technical, scientific, manufacturing, and other know-how and information, trade secrets, knowledge, technology, means, methods, processes, practices, formulae, instructions, skills, techniques, procedures, experiences, ideas, technical assistance, designs, drawings, assembly procedures, computer programs, apparatuses, specifications, data, results and other material, including: biological, chemical, pharmacological, toxicological, pharmaceutical, physical and analytical, pre-clinical, clinical,

safety, manufacturing and quality control data and information, including study designs and protocols; assays; stability reports, production records, test methods, certificates of analyses, development reports, quality and technical agreements, and supplier audit reports and biological methodology; in each case (whether or not confidential, proprietary, patented or patentable) in written, electronic or any other tangible or intangible form now known or hereafter developed.

"Knowledge" means (a) with regard to the Seller, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Seller's chief executive officer, chief financial officer, and general counsel, in each case as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate) and (b) with regard to Purchaser, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Purchaser's chief executive officer as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate).

"Lambda Buyer" means the purchaser of the Seller assets associated with any "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.

"Law Firm" means Sidley Austin LLP and its successors.

"Letter of Authorization" has the meaning set forth in Section 7.9(c).

"Liabilities" means debts, liabilities, duties, obligations or commitments of any nature whatsoever, whether direct or indirect, asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured or otherwise, whenever or however arising (including whether arising out of any Contract or in a tort claim based on negligence or strict liability).

"Licensed Compound" has the meaning given to it in the Merck License Agreement.

"Licensed Product" has the meaning given to it in the Merck License Agreement.

"Licensed Product Data" means any and all Data that relates to any Licensed Product.

"Licensed Product Regulatory Information" means any and all Regulatory Information that relates to any Licensed Product.

"Licensed Progeria Product" has the meaning given to it in the Merck License Agreement.

"Licensed Progeria Product Data" means any and all Data that exclusively relate to the Licensed Progeria Product.

"Licensed Progeria Product Regulatory Information" means any and all Regulatory Information that exclusively relate to the Licensed Progeria Product.

"Lien" means all forms of lien (including mechanic's, contractor's or other similar liens arising under or relating to the provision of goods or services on or to any Transferred Assets, and liens arising under the Bankruptcy Code), encumbrance, defect or irregularity in title, pledge, mortgage, deed of trust, deed to secure debt, security interest, charge, transfer restriction or similar agreement or encumbrance, including any dedication under any gathering, transportation, treating, processing, fractionating, purchase, sale or similar agreements, or any other rights granted or consensual as or against any Transferred Assets including but not limited to easements, encroachments, rights of first refusal, options, or any other interest or right in property that constitutes a lien or interest within the definition or adjudication of such terms under Section 101(37) of the Bankruptcy Code.

"Lonafarnib Antiviral Field" means the Field, excluding the Progeria Field. For the avoidance of doubt, the Lonafarnib Antiviral Field includes the Lonafarnib HDV Field.

"Lonafarnib Antiviral Products" means any and all Licensed Products for use in the Lonafarnib Antiviral Field, excluding the Licensed Progeria Product for use in the Progeria Field.

- "Lonafarnib HDV Field" means the use of the Licensed Compound or Licensed Product for the treatment of Hepatitis D virus infections, including the treatment of patients co-infected with Hepatitis D virus and either or both of Hepatitis C virus and Hepatitis B virus.
- "Lonafarnib HDV Products" means any and all Lonafarnib Antiviral Products for use in the Lonafarnib HDV Field.
- "Lonafarnib IND" means any and all INDs owned or controlled by Seller or its Affiliates for Lonafarnib Antiviral Products anywhere in the world, including IND # 110,877 for the Lonafarnib HDV Product.
- "Lonafarnib IND Transfer Date" means the date on which the transfer of all Lonafarnib INDs by Seller or its Affiliates to Purchaser under this Agreement is complete such that Purchaser is considered the holder of all Lonafarnib INDs by the applicable Regulatory Authority.
 - "Manufacture" has the meaning given to it in the Merck License Agreement.

"Material Adverse Effect" means a material adverse effect on the business, financial condition or results of operations of the Business (including the Transferred Assets and Assumed Liabilities) taken as a whole; provided, however, that none of the following shall be deemed (either alone or in combination) to constitute, and none of the following shall be taken into account in determining whether there has been or may be, a Material Adverse Effect: (a) any change in, or effects arising from or relating to, general business or economic conditions affecting any industry in which the Business operates; (b) any change in, or effects arising from or relating to, the United States or foreign economies, or securities, banking or financial markets in general, or other general business, banking, financial or economic conditions (including (i) any disruption in any of the foregoing markets, (ii) debt defaults or other restructuring events of any country with respect to which bondholders take a discount to the debt of any country or any increases in the interest rates for any country's debt, (iii) any change in currency exchange rates, (iv) any decline or rise in the price of any security, commodity, contract or index and (v) any increased cost, or decreased availability, of capital or pricing or terms related to any financing for the Transactions); (c) any change from, or effects arising from or relating to, the occurrence, escalation or material worsening of any act of God or other calamity, natural disaster, pandemic or disease, outbreak, hostility, act of war, sabotage, cyber-attack or terrorism or military action; (d) any action taken by Purchaser or its Affiliates with respect to the Transactions or with respect to the Business; (e) any action taken, or failed to be taken, by the Seller at the request of or with the consent of Purchaser or otherwise in compliance with the terms of this Agreement or any change from, or effects arising from or relating to, Purchaser's failure to consent to any action restricted by Section 6.1; (f) any change in, or effects arising from or relating to changes in, Applicable Law or accounting rules (including GAAP) or any interpretation thereof; (g) the failure of the Business to meet any of its projections, forecasts, estimates, plans, predictions, performance metrics or operating statistics or the inputs into such items (whether or not shared with Purchaser or its Affiliates or representatives); (h) national or international political, labor or social conditions; (i) the public announcement of, entry into or pendency of, actions required or contemplated by or performance of obligations under, this Agreement and the Transactions or the identity of the parties to this Agreement; (j) the sale of any assets other than the Transferred Assets to any third parties by a member of the Seller Group or any of their Affiliates; (k) any effect arising or resulting from or related to the filing of the Bankruptcy Cases; (1) any action required to be taken under any Applicable Law or Order or any existing Contract by which any member of the Seller Group's (or any of their properties) are bound; (m) seasonal changes in the results of operations of the Seller Group; (n) any epidemic, pandemic, outbreak of disease or other public health emergency (including COVID-19) or any escalation or worsening of any such conditions or (o) any objections made in the Bankruptcy Court to this Agreement, the Transactions, the Sale Order or the reorganization, any orders of the Bankruptcy Court and any actions or omissions of the Seller in compliance with any order of the Bankruptcy Court and the assumption or rejection of any Assigned Contract; except in the cause of clauses (a) through (c), (h) and (n), to the extent such conditions, events, changes, crises and disasters, as applicable, do not have a material

and disproportionate impact on the Business, taken as a whole, compared to other industry participants (in which case, only the extent of such disproportionate effect shall be taken into account when determining whether there is a Material Adverse Effect).

- "Merck" means Merck Sharp & Dohme Corp. (successor-in-interest of Schering Corporation).
- "Merck License Agreement" means that certain License Agreement, dated September 3, 2010, by and between the Seller and Merck, and any and all amendments or supplements thereto, including that certain First Amendment, dated January 18, 2011, Amendment to License Agreement, dated June 11, 2013, Amendment #2 to License Agreement, dated November 20, 2014, Amendment #3 to License Agreement, dated March 6, 2015, Amendment #4 to License Agreement, dated June 9, 2015, Amendment #5 to License Agreement, dated December 17, 2015, Amendment #6 to License Agreement, dated May 15, 2018, and Amendment #7 to License Agreement, dated November 3, 2020.
- "Merck Pharmacovigilance Agreement" means the Safety Agreement, dated February 24, 2021, by and between the Seller and Merck, including any and all amendments, termination agreement or memo of understanding related thereto.
- "Merck Side Letter" means the letter agreement, dated as of the Closing Date, by and between the Seller, Purchaser and Merck, in a form reasonably acceptable to Purchaser.
- "NDA" means, with respect to a pharmaceutical product, a New Drug Application submitted to the FDA in accordance with the FD&C Act, and the rules and regulations promulgated thereunder, or any analogous application or submission with any Regulatory Authority outside of the United States.
 - "Non-Transferred Asset" has the meaning set forth in Section 2.6(a).
 - "Nonparty Affiliates" has the meaning set forth in Section 10.15.
 - "Notice of Readiness to Close" has the meaning set forth in Section 8.5.
- "Order" means any award, decision, injunction, judgment, ruling or verdict entered, issued, made or rendered by any Governmental Authority or arbitrator.
- "Organizational Documents" means (a) the articles or certificates of incorporation and the bylaws of a corporation, (b) the partnership agreement and any statement of partnership of a general partnership, (c) the limited partnership agreement and the certificate of limited partnership of a limited partnership, (d) the operating or limited liability company agreement and the certificate of formation of a limited liability company, (e) any charter, joint venture agreement or similar document adopted or filed in connection with the creation, formation or organization of a Person not described in clauses (a) through (d), and (f) any amendment to or equivalent of any of the foregoing.
 - "Outside Date" has the meaning set forth in Section 9.1(i).
- "Owned Intellectual Property Assets" means the Intellectual Property owned or purported to be owned by any member of the Seller Group that is used in, held for use in, or related to, the conduct of the Business as currently conducted or proposed to be conducted, including any Intellectual Property related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.
- "**Permit**" means all permits, authorizations, certificates, franchises, consents and other approvals from any Governmental Authority.
- "Permitted Liens" means (a) Liens for Taxes, assessments or other governmental charges not yet due and payable or being contested in good faith by appropriate proceedings; (b) mechanics', carriers', workers', repairers' and other similar Liens arising or incurred in the ordinary course of business for obligations that are not overdue or are being contested in good faith by appropriate proceedings; (c) zoning,

entitlement and building regulations and land use restrictions; (d) purchase money Liens and Liens securing rental payments under capital lease arrangements; (e) Liens arising under leases of property or equipment in favor of the owner thereof; (f) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security; (g) deposits to secure the performance of bids, Contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business; (h) licenses of Intellectual Property granted in the ordinary course of business; (i) Liens arising under or created by this Agreement or any of the Related Documents; (j) Liens arising in the ordinary course of business which would not reasonably be expected to have a Material Adverse Effect; and (k) Liens set forth on Schedule 1.1(a).

"Person" means any individual, partnership, joint venture, limited liability company, corporation, firm, trust, association, unincorporated organization, governmental authority or agency, or any other entity not specifically listed herein.

"Personal Information" means any information in the possession or control of the Seller Group (solely as related to the Business) about an identifiable individual other than the name, title or business address, business email address or telephone number of any employee of the Seller Group.

"Petition Date" has the meaning set forth in the Recitals.

"Plan Consummation Date" means the date on which the Seller Group's plan in the Bankruptcy Cases is substantially consummated.

"Pre-Closing Tax Period" means any taxable period ending on or prior to the Closing Date and the portion of any Straddle Period through the Closing Date.

"Preliminary Allocation Schedule" has the meaning set forth in Section 2.11(a).

"Previously Excluded Contract" has the meaning set forth in Section 5.5(b).

"Previously Unknown Contract" has the meaning set forth in Section 5.5(a).

"Progeria Field" has the meaning given to it in the Merck License Agreement.

"Provision" has the meaning set forth in Section 10.4.

"Public Health Measures" means any closures, "shelter-in-place," "stay at home," workforce reduction, social distancing, shut down, closure, curfew or other restrictions or any other Applicable Law, Orders, directives, guidelines or recommendations issued by any Governmental Authority, the Centers for Disease Control and Prevention, the World Health Organization, or any industry group in connection with COVID-19 or any other epidemic, pandemic, or outbreak of disease, or in connection with or in response to any other public health conditions.

"Purchase Price" means the Base Price plus the aggregate amount of Purchaser Cure Amounts.

"Purchaser" has the meaning set forth in the preamble.

"Purchaser Cure Amounts" means, with respect each Assigned Contract, the Determined Cure Amounts as follows: (a) if Purchaser does not assume any Cross-Over Contract, then up to \$180,000 in the aggregate, (b) if Purchaser assumes the IQVIA Contracts, then up to \$2,180,000, (c) if Purchaser assumes the Biorasi Contracts and the IQVIA Contracts, then up to \$2,380,000, or (d) if Purchaser assumes the Biorasi Contracts but not the IQVIA Contracts, then up to \$380,000 in the aggregate.

"Purchaser Group Members" has the meaning set forth in Section 10.17.

"Purchaser Releasing Party" has the meaning set forth in Section 10.16(b).

"Purchaser Schedules" has the meaning set forth in ARTICLE 4.

"Purchaser's FDA Transfer Letters" means the letters from Purchaser to FDA in form and substance acceptable to Purchaser, notifying FDA of the acceptance of the transfer from the Seller to Purchaser of all of Seller's right, title and interest in the Lonafarnib IND.

"PV Services Stop Date" has the meaning set forth in Section 7.10(d).

"Regulatory Applications" means (a) the single application or set of applications for approval and/or pre-market approval to Manufacture and sell commercially a pharmaceutical therapeutic product submitted to the FDA including, without limitation, any related registrations with or notifications to the FDA, and (b) any foreign equivalents to such applications filed with any other national or supranational Regulatory Authority in the Territory, and (c) all supplements and amendments that may be filed with respect to any of the foregoing.

"Regulatory Approval" means any and all approvals (including pricing or pricing reimbursement approvals), licenses, registrations, or authorizations of any federal, national, multinational, state, provincial or local regulatory agency, department, bureau or other governmental entity necessary for the Manufacture, use, storage, import, export, transport, promotion, marketing or sale of a Licensed Product in any field (including the Lonafarnib Antiviral Field and Progeria Field) in the applicable country.

"Regulatory Authority" means any United States federal, state, or local government, or any foreign government, or political subdivision thereof, or any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any governmental arbitrator or arbitral body with responsibility for granting licenses or approvals, including Regulatory Approvals, necessary for the marketing and sale of the Licensed Product in the applicable country in the Territory.

"Regulatory Information" means any filings, submissions, applications, data, reports or correspondence, including, without limitation, dossiers, manufacturing data, drug master files, inspection reports, adverse event files and complaint files, with any Governmental Authority that relate to any Licensed Compound or Licensed Product in any field (including the Lonafarnib Antiviral Field and Progeria Field), including any (a) INDs, Regulatory Applications, Regulatory Approvals, drug master files, applications for designation as an "Orphan Product" under the Orphan Drug Act, applications for designation as a humanitarian use device or a breakthrough device, for Fast Track or Breakthrough Therapy Designation, Accelerated Approval or Priority Review or for a Special Protocol Assessment or all other filings (including Regulatory Approval applications and counterparts to any of the foregoing in any country or region), (b) all supplements and amendments to any of the foregoing, and (c) all data and other information contained in, and correspondence relating to, any of the foregoing, in each case of any of the foregoing items listed in this definition, in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter.

"Related Claims" means all claims or causes of action (whether in contract or tort, in law or in equity, or granted by statute or otherwise) that may be based upon, arise out of or relate to this Agreement, the Related Documents and any other document or instrument delivered pursuant to this Agreement or the Related Documents, or the negotiation, execution, termination, validity, interpretation, construction, enforcement, performance or nonperformance of this Agreement or the Related Documents or otherwise arising from the Transactions or the relationship between the parties (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with, or as an inducement to enter into, this Agreement or the Related Documents).

"Related Documents" means the Escrow Agreement, the Bill of Sale and Assignment and Assumption Agreement, Intellectual Property Assignment Agreement, Sublicense Agreement, and Merck Side Letter; *provided, however*, that the Escrow Agreement, the Bill of Sale and Assignment and Assumption Agreement, Intellectual Property Assignment Agreement, Sublicense Agreement, and Merck Side Letter shall not be a Related Document solely for purposes of applying the provisions in ARTICLE 10 to the extent, and only to the extent, that any such document expressly conflicts with ARTICLE 10.

"Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing into the environment of any Hazardous Substances.

"Sale Order" means an Order of the Bankruptcy Court issued pursuant to sections 105(a), 363 and 365 of the Bankruptcy Code in form and substance acceptable to Purchaser and the Seller, approving this Agreement and all of the terms and conditions hereof and approving and authorizing the Seller to consummate the Transactions contemplated hereby Free and Clear and containing a finding that Purchaser has acted in "good faith" within the meaning of Section 363(m) of the Bankruptcy Code.

"Satisfactory IQVIA Cure Resolution" has the meaning set forth in Section 7.15(c).

"Satisfactory Other Cure Resolution" has the meaning set forth in Section 7.15(c).

"Schedules" has the meaning set forth in ARTICLE 3.

"Seller" has the meaning set forth in the preamble.

"Seller Access Contact" has the meaning set forth in Section 6.2(a).

"Seller Cure Amounts" means, with respect to Assigned Contracts, any Determined Cure Amounts that are not the then-applicable Purchaser Cure Amounts.

"Seller Financial Statements" has the meaning set forth in Section 3.9.

"Seller Group" means the Seller and each of its Affiliates.

"Seller Group Members" has the meaning set forth in Section 10.17.

"Seller Group Taxes" means any (i) Liability of Seller Group for Taxes, (ii) any Liability for Asset Taxes attributable to any Pre-Closing Tax Period, and (iii) any Liability of Seller Group for the unpaid Taxes of any Person under Treasury Regulation §1.1502-6 (or any similar provision of state, local, or non-U.S. law), as a transferee or successor, by contract, or otherwise.

"Seller Permits" has the meaning set forth in Section 3.5.

"Seller Releasing Party" has the meaning set forth in Section 10.16(a).

"Seller's FDA Transfer Letters" means the letters from the Seller to FDA in form and substance acceptable to Purchaser, notifying FDA of the transfer from the Seller to Purchaser of all of Seller's rights in the Lonafarnib IND.

"Solvent" when used with respect to any Person, means that, as of any date of determination, (a) the fair salable value (determined on a going concern basis) of its assets and property will, as of such date, exceed the amounts required to pay its debts as they become absolute and mature, as of such date, (b) such Person will have adequate capital to carry on its business and (c) such Person will be able to pay its debts as they become absolute and mature, in the ordinary course of business, taking into account the timing of and amounts of cash to be received by it and the timing of and amounts of cash to be payable on or in respect of its indebtedness.

"Specific Provision" has the meaning set forth in Section 10.4.

"Storage Contract" means each Contract (or portion thereof) with a Third Party pursuant to which any Inventory are held for storage or other activities.

- "Straddle Period" means any taxable year or other taxable period beginning on or before and ending after the Closing Date.
- "Sublicense Agreement" means the Sublicense Agreement, dated as of the Closing Date, by and between the Seller and Purchaser, in a form reasonably acceptable to the Seller and Purchaser.
 - "Supplemental Assignment Notice" has the meaning set forth in Section 5.5(a).
- "Supplemental Assignment Notice Objection Deadline" has the meaning set forth in Section 5.5(a).
- "Tax" means any tax of any kind whatsoever (including any income tax, franchise tax, branch profits tax, capital gains tax, value-added tax, unclaimed property, escheat, sales tax, use tax, property tax, transfer tax, payroll tax, social security tax or withholding tax), and any related fine, penalty, interest, or addition to tax with respect thereto, imposed, assessed or collected by or under the authority of any Governmental Authority.
- "Tax Return" means any return (including any information return), report, statement, schedule, notice, form, or other document or information (whether in tangible, electronic or other form), including any amendments, schedules attachments, supplements, appendices and exhibits thereto, filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority in connection with the determination, assessment, collection, or payment, of any Tax.
 - "Termination Fee" means a fee equal to \$36,000.
 - "Territory" has the meaning given to it in the Merck License Agreement.
 - "Third Party" means any Person other than a Contracting Party or its Affiliates.
- "Trademark" means, collectively, trademarks, service marks trade names, slogans, logos, trade dress or other similar source or origin identifiers (whether statutory or common law, whether registered or unregistered), together with all (a) registrations and applications for any of the foregoing, (b) extensions or renewals thereof, (c) goodwill (if any) connected with use thereof or symbolized thereby, and (d) rights and privileges arising under Applicable Law with respect to any of the foregoing.
- "Transactions" means the transactions contemplated by this Agreement and the Related Documents.
 - "Transfer Taxes" has the meaning set forth in Section 2.10.
 - "Transferred Assets" has the meaning set forth in Section 2.1.
 - "Transferred Business Books and Records" has the meaning set forth in Section 2.1(d).
- "Transferred Data" means any and all Data that (a) are owned or purported to be owned by the Seller or its Affiliates (including all such Data held by or on behalf of Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level)) and (b) exclusively relate to any Lonafarnib Antiviral Product, including any Data related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.
- "Transferred Materials" means the Transferred Data, Transferred Regulatory Information, Transferred Studies, Transferred Business Books and Records, and Inventory.
- "Transferred Regulatory Information" means any and all Regulatory Information that (a) are owned or purported to be owned by the Seller or its Affiliates (including all such Regulatory Information held by or on behalf of Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level)) and

(b) exclusively relate to any Lonafarnib Antiviral Product, including any Regulatory Information related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.

"Transferred Studies" all clinical, preclinical, and non-clinical studies to the extent on-going as of the Closing being conducted by or on behalf of Seller or any of its Affiliates related to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field, including without limitation the virology studies being conducted by Seller in collaboration with (a) INSERM U1110, Université de Strasbourg, France and (b) U1111, Centre International de Recherche en Infectiologie, Lyon, France, team HepVir (each, a "Virology Collaborator", and such studies, the "Virology Studies") and any studies related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.

"Transition Materials" means all Licensed Product Data, Licensed Product Regulatory Information, Transferred Studies, Business Books and Records, and Inventory.

"Virology Collaborator" has the meaning set forth in the definition of Transferred Studies.

"Virology Collaborator Confirmation Letters" means letters of confirmation from each Virology Collaborator in form and substance acceptable to Purchaser confirming that the Virology Studies are ongoing and have not been interrupted, suspended, or delayed and that all payments payable to such Virology Collaborator in connection with the relevant Virology Study has been duly and timely paid in full.

"Virology Studies" has the meaning set forth in the definition of Transferred Studies.

"Zokinvy Buyer" means Sentynl Therapeutics, Inc.

"Zokinvy Buyer Agreement" means an agreement between Purchaser and the Zokinvy Buyer regarding coordination relevant to the Development, Manufacture, and Commercialization of the Lonafarnib Antiviral Products by Purchaser and the Licensed Progeria Product by the Zokinvy Buyer.

"Zokinvy Buyer-Eiger Agreement" means that particular Asset Purchase Agreement entered into between Seller and the Zokinvy Buyer, dated March 31, 2024, under which the Seller sold certain assets to the Zokinvy Buyer related to the use of the Licensed Progeria Product in the Progeria Field.

"Zokinvy Dossier" means the complete regulatory dossier of the Zokinvy Product, including without limitation (a) all INDs, NDAs, and equivalent foreign applications or registrations for the Zokinvy Product or for Regulatory Approval of the Zokinvy Product (including all modules thereof, and amendments, updates, or supplements thereto); (b) all Regulatory Approvals and any other technical, medical and scientific registrations, authorizations and approvals (including approvals of NDAs or foreign equivalents, supplements and amendments, pre- and post- approvals, pricing and reimbursement approvals, and labeling approvals) of any Regulatory Authority necessary for or applicable to the development (including the conduct of clinical trials), manufacture, distribution, marketing, promotion, offer for sale, use, import, reimbursement, export or sale of the Zokinvy Product in any regulatory jurisdiction, together with all related correspondence to or from any Regulatory Authority and all documents referenced in the complete regulatory chronology for each NDA or foreign equivalent, including the drug master file (if any), IND, NDA and supplemental NDA, or foreign equivalents; and (c) all data and other information contained or referenced in any of (a) or (b) above.

"Zokinvy Product" means the pharmaceutical product containing lonafarnib as its active pharmaceutical ingredient and sold under the trademark Zokinvy®.

1.2 Other Definitional and Interpretive Matters.

- (a) Unless otherwise expressly provided, for purposes of this Agreement and the Related Documents, the following rules of interpretation shall apply:
 - (i) Calculation of Time Period. All references to a day or days shall be deemed to refer to a calendar day or days, as applicable, unless otherwise specifically provided. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.
 - (ii) Dollars. Any reference to \$ shall mean U.S. dollars, which is the currency used for all purposes in this Agreement and the Related Documents. The specification of any dollar amount in the representations and warranties or otherwise in this Agreement, the Related Documents or the Schedules is not intended and shall not be deemed to be an admission or acknowledgement of the materiality of such amounts or items, nor shall the same be used in any dispute or controversy between the parties hereto to determine whether any obligation, item or matter (whether or not described herein or included in any schedule) is or is not material for purposes of this Agreement, the Related Documents or the Schedules.
 - (iii) Exhibits/Schedules. The Exhibits and Schedules to this Agreement are an integral part of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Schedule shall be deemed to have been disclosed on each other Schedule. Disclosure of any item on any Schedule shall not constitute an admission or indication that any such item is required to be disclosed, or that such item or matter is material or has resulted in or will result in a Material Adverse Effect or that the included items or actions are not in the ordinary course of business. No disclosure on a Schedule relating to a possible breach or violation of any Contract, Applicable Law or Order shall be construed as an admission or indication that a breach or violation exists or has actually occurred. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.
 - (iv) Gender and Number. Any reference to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.
 - (v) Headings. The provision of a table of contents, the division of this Agreement or Related Documents into articles, sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement or Related Document, as applicable. Unless otherwise specified, all references in this Agreement to any "Section" or other subdivision are to the corresponding section or subdivision are to the corresponding section or other subdivision are to the corresponding section or subdivision of such Related Document.
 - (vi) Herein. The words such as "herein," "hereinafter," "hereof" and "hereunder" that are used in this Agreement refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires. Uses of such words in the Related Documents shall refer to such Related Document as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
 - (vii) Or. The word "or" shall be construed in the inclusive sense of "and/or" unless otherwise specified.
 - (viii) Including. The word "including" or any variation thereof means (unless the context of its usage otherwise requires) "including, without limitation" and shall not be

construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

- (ix) Successors. A reference to any party to this Agreement, any Related Document or any other agreement or document shall include such party's successors and permitted assigns.
- (x) Legislation. A reference to any legislation or to any provision of any legislation shall include any amendment thereto, and any modification or re-enactment thereof, any legislative provision substituted therefor and all regulations and statutory instruments issued thereunder or pursuant thereto.
- (xi) Reflected On or Set Forth In. An item arising with respect to a specific representation or warranty shall be deemed to be "reflected on" or "set forth in" a balance sheet or financial statement, to the extent any such phrase appears in such representation or warranty, if (a) there is a reserve, accrual or other similar item underlying a number on such balance sheet or financial statement that relates to the subject matter of such representation, (b) such item is otherwise specifically set forth on the balance sheet or financial statement or (c) such item is set forth in the notes to the balance sheet or financial statement.
- (xii) Made Available. Any reference in this Agreement to "made available" means a document or other item of information that was provided or made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions.
- (b) All representations and warranties set forth in this Agreement or the Related Documents are contractual in nature only and subject to the sole and exclusive remedies set forth herein. No Person is asserting the truth of any representation and warranty set forth in this Agreement or the Related Documents; rather, the parties have agreed that should any representations and warranties of any party prove untrue, the other parties shall have the specific rights and remedies herein specified as the exclusive remedy therefor, but that no other rights, remedies or causes of action (whether in law or in equity or whether in contract or in tort or otherwise) are permitted to any party hereto as a result of the untruth of any such representation and warranty. The phrase "to Seller's Knowledge" and phrases of similar import or effect are used herein to qualify and limit the scope of any representation or warranty in which they appear and are not affirmations of any Person's "superior knowledge" that the representation or warranty in which they are used is true.
- (c) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and the Related Documents and, in the event an ambiguity or question of intent or interpretation arises, this Agreement and the Related Documents shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement and the Related Documents. The parties hereto agree that changes from earlier drafts to the final version of this Agreement do not necessarily imply that the party agreeing to such change is agreeing to a change in meaning (as the party agreeing to such change may believe the change is stylistic and non-substantive); consequently, no presumption should exist by virtue of a change from a prior draft.

ARTICLE 2. THE PURCHASE AND SALE; CLOSING

- 2.1 **Purchase and Sale**. Upon the terms and subject to the conditions set forth in this Agreement, the Sublicense Agreement, the Merck Side Letter, and the Sale Order, at the Closing, in exchange for an aggregate payment from Purchaser to the Seller equal to the Purchase Price, Purchaser shall purchase, assume and accept from the Seller, and the Seller shall sell, transfer, assign, convey and deliver (or shall cause the sale, transfer, assignment, conveyance and delivery) to Purchaser, Free and Clear (except for Permitted Liens), all of the rights, title and interests in, to and under the following assets and interests used in the Business as the same shall exist on the Closing Date (and, subject to Section 7.11, with respect to the Existing Manufacturing Contracts, on the applicable Existing Manufacturing Contract Transfer Date) (collectively, the "**Transferred Assets**"):
 - (a) (i) subject to the ensuing clause (ii), all Contracts that are listed on Schedule 2.1(a) (as such Schedule may be amended pursuant to the terms of this Agreement, the "Contracts List"), (ii) on the applicable Existing Manufacturing Contract Transfer Date automatically and without further notice, the Existing Manufacturing Contracts, and (iii) all other Contracts that are Assigned Contracts pursuant to Sections 5.3(b), 5.4, 5.5 and 7.15, including all rights, interests, credits, prepaid charges and expenses, deferred charges, advance payments, deposits, and prepaid items of Seller related thereto (collectively, the "Assigned Contracts");
 - (b) the Owned Intellectual Property Assets, including the Intellectual Property Registrations listed on Schedule 3.12(a), as may be amended or supplemented with the agreement of the Seller at the request of Purchaser at any time prior to the Closing; *provided, however*, that any and all filing or transfer fees due to any Third Party (including any Governmental Authority) incurred by either party in connection with the transfer of such Intellectual Property Registrations shall be borne and paid by Purchaser;
 - (c) the Transferred Regulatory Information, including the information and documents listed on Schedule 2.1(c), as may be amended or supplemented at the request of Purchaser at any time prior to the Closing; provided, however, that the Seller may retain copies of such Transferred Regulatory Information or may retain originals of the Transferred Regulatory Information and instead provide Purchaser with copies to the extent permissible under Applicable Laws and shall maintain the confidentiality thereof in accordance with the terms of the Confidentiality Agreement as Confidential Information, except Seller will be deemed the "Recipient" and Purchaser will be deemed "Company under the Confidentiality Agreement and Seller will be obligated to keep such Confidential Information from being disclosed for an indefinite period of time, mutatis mutandis unless otherwise required to be disclosed under Applicable Law, including by a Governmental Authority; provided, further, that the Parties shall cooperate in good faith to effectuate the assignments and transfer of the Transferred Regulatory Information with any applicable Governmental Authority, including duly executing and delivering, or causing to be duly executed and delivered, such instruments (including the filing of such assignments, agreements and documents) as may be necessary in order to affect such assignment and transfer of the Transferred Regulatory Information from the Seller to Purchaser;
 - (d) the Business Books and Records exclusively relating to any Lonafarnib Antiviral Product (including any Business Books and Records related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement, but excluding records and files not reasonably separable from documents and databases that do not relate exclusively to any Lonafarnib Antiviral Product or any Transferred Materials) ("**Transferred Business Books and Records**"); *provided, however*, that the Seller may retain copies of the Transferred Business Books and Records and shall maintain the confidentiality thereof in accordance with the terms of the Confidentiality Agreement as Confidential Information, except Seller will be deemed the "Recipient" and

Purchaser will be deemed "Company under the Confidentiality Agreement and Seller will be obligated to keep such Confidential Information from being disclosed for an indefinite period of time, *mutatis mutandis* unless otherwise required to be disclosed under Applicable Law, including by a Governmental Authority; *provided, further*, that such Transferred Business Books and Records shall include solely such records created or acquired during the last three (3) years; *provided, further*, that the Seller will make available, or cause to be made available, to Purchaser copies of Business Books and Records that are not Transferred Business Books and Records, that are in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter, and the Seller is permitted to redact or remove any extraneous or unrelated confidential or proprietary information in furtherance of such obligation, in each case such that Purchaser is able to conduct the Business and Develop, Manufacture, and Commercialize the Lonafarnib Antiviral Products in the Lonafarnib Antiviral Field in the Territory as contemplated by this Agreement;

- (e) all rights to receive mail and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group relating to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field (including any such mail and other correspondence and communications (including electronic mail) from the FDA or any other Governmental Authority, customers, advertisers, suppliers, distributors, agents and others) and payments with respect to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field; *provided*, *however*, that with respect rights to receive mail and other and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group that is not exclusively relating to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field, such rights will be non-exclusive;
- (f) all of the Seller Group's rights, claims or causes of action, whether class, individual or otherwise in nature, under contract or in law or in equity, against third parties relating to the assets, properties, business or operations of the Seller Group with respect to the Business, the Transferred Assets and the Assumed Liabilities (including all guaranties, warranties, indemnities and similar rights in favor of the Seller Group or any their Affiliates to the extent solely related to the Transferred Assets or the Assumed Liabilities), in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date, except for such rights, claims and causes of action related to the Excluded Assets or Excluded Liabilities;
- (g) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, choses in action, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent used in or held for use for the Transferred Assets listed in clauses (a) through (f) above or the Assumed Liabilities;
- (h) all right, title and interest in and to (i) any raw materials (including work in process, buffer stock held by vendors, dies and active pharmaceutical ingredients inventory, reference standards and materials, and all components and materials used in the Manufacture of any Lonafarnib Antiviral Product), finished goods and other inventory of all Lonafarnib Antiviral Products in the possession or control of, otherwise held by or on behalf of (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level), or owned by the Seller Group; and (ii) all good and marketable unbroken lots of packaged finished goods inventory of all Lonafarnib Antiviral Product in the possession or control of, or otherwise held by or on behalf of (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level), the Seller Group as of Closing, regardless of where located,

and all rights to receive refunds, rebates or credits in connection therewith (for the avoidance of doubt, the Transferred Assets also include all manufactured product, packaging material, compounds and any other similar assets relating to any Lonafarnib Antiviral Product, and any assets that are under manufacture); in each case including the raw materials, reference standards and materials, and inventory listed in Schedule 2.1(h), as may be amended or supplemented at the request of Purchaser at any time prior to the Closing (collectively, "Inventory");

- (i) all Transferred Data;
- (i) all Transferred Studies;
- (k) all advertising, marketing, market research, sale and promotional files and materials (including any television, radio and print content and materials), pricing lists, consulting deliverables and other related literature, catalogs, point of sale materials and website content, including all Intellectual Property therein, relating to any Transferred Asset and Assumed Liability that are within the Seller Group's control or reasonably accessible to the Seller Group; and
- (1) to the extent not covered above, any goodwill associated with or symbolized by any of the foregoing Transferred Assets described in clauses (a) through (k) above and any properties, rights and interests of every kind and nature, whether tangible or intangible, real, personal or mixed, known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, wherever located, associated with or appurtenant to the above-referenced Transferred Assets.
- 2.2 **Excluded Assets**. Notwithstanding the provisions of Section 2.1 or anything to the contrary herein, any and all assets, rights and properties of the Seller Group that are not specifically identified in Section 2.1 as Transferred Assets, including the following (collectively, the "**Excluded Assets**"), shall be retained by the Seller Group, and Purchaser and its designees shall acquire no right, title or interest in the Excluded Assets in connection with the Transaction:
 - (a) all (i) cash and cash equivalents, wherever located, including bank balances and bank accounts or safe deposit boxes, monies in the possession of any banks, savings and loans or trust companies and similar cash items, (ii) escrow monies and deposits in the possession of landlords and utility companies, and (iii) investment securities and other short- and medium-term investments;
 - (b) all records, documents or other information exclusively relating to current or former employees of the Seller Group that are not hired by Purchaser, and any materials to the extent containing information about any employee, disclosure of which would violate Applicable Law or such employee's reasonable expectation of privacy;
 - (c) any interest of the Seller Group under this Agreement or the Related Documents, including the right to receive the Purchase Price and to enforce the Seller's rights and remedies thereunder:
 - (d) all Excluded Contracts (including all prepaid assets relating to the Excluded Contracts), other than the Assigned Contracts, to which any member of the Seller Group or any of their respective Affiliates is a party;
 - (e) any (i) Attorney-Client Information arising from communications prior to the Closing Date between a member of the Seller Group (including any one or more officers, directors or stockholders of such Seller Group member), on the one hand, and its counsel, on the other hand, and (ii) claims under any director and officer, errors and omissions, fiduciary and commercial crime insurance policies; and
 - (f) any rights of the Seller Group to Tax refunds (or credits for overpayment of Taxes in lieu of a refund) attributable to any Pre-Closing Tax Period;

- (g) all Permits (including applications therefor and any trade or import/export Permits) that (i) are not materially related to the Business or (ii) are not transferable to Purchaser under Applicable Law;
 - (h) the Excluded Books and Records;
- (i) any assets not otherwise designated as Transferred Assets or from time to time designated by the parties hereto as Excluded Assets;
- (j) all accounts receivable, intercompany obligations and other amounts receivable by the Seller Group;
 - (k) the Avoidance Actions;
- (l) all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group (including all guaranties, warranties, indemnities and similar rights in favor of the Sellers Group or any of their Affiliates) to the extent arising under the Bankruptcy Code or relating to any of the Excluded Assets or Excluded Liabilities, in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date; and
- (m) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent exclusively related to or exclusively used in or held for use for the Excluded Assets listed in clauses (a) through (l) above.

Notwithstanding anything to the contrary contained in this Agreement or any of the other Related Documents, Purchaser acknowledges and agrees that all of the following are also Excluded Assets, and all right, title and interest in and to all Excluded Assets shall be retained by the Seller Group and shall remain the property of the Seller Group (and shall expressly be excluded from the sale, transfer, assignment and conveyance to Purchaser hereunder), and neither Purchaser nor any of its Affiliates shall have any interest therein: (x) all records and reports prepared or received by the Seller Group or any of their Affiliates in connection with the sale of the Business and the Transactions, including all analyses relating to the Business or Purchaser so prepared or received; and (y) all confidentiality agreements with prospective purchasers of the Business or any portion thereof and all bids and expressions of interest received from third parties with respect thereto.

- 2.3 **Assumption of Liabilities**. On the terms and subject to the conditions set forth in this Agreement, Purchaser shall, effective as of the Closing, assume and agree to pay, discharge and perform in accordance with their terms the following Liabilities of the Seller Group arising from or related to the Business or the Transferred Assets as the same shall exist on the Closing Date arising only after the Closing Date (collectively, the "**Assumed Liabilities**"), including:
 - (a) all Liabilities relating to the Transferred Assets other than the Assigned Contracts (which are addressed in Section 2.3(b)) solely to the extent such Liabilities relate to and arise in periods following the Closing;
 - (b) subject to Section 2.4, all Liabilities arising under the Assigned Contracts other than the Existing Manufacturing Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Amounts;
 - (c) subject to Section 2.4, all Liabilities arising under each Existing Manufacturing Contract solely to the extent such Liabilities relate to and arise (i) in connection with the transition activities under Section 7.6 performed by the Seller pursuant to Purchaser's instructions following the Closing and before the applicable Existing Manufacturing Contract Transfer Date and (ii) in periods

following the applicable Existing Manufacturing Contract Transfer Date, and all of the Purchaser Cure Amounts; and

- (d) all Taxes for which Purchaser is liable pursuant to this Agreement.
- 2.4 **Excluded Liabilities**. Notwithstanding Section 2.3, Purchaser is assuming only the Assumed Liabilities of the Seller Group and will not assume or be liable for any Excluded Liabilities (including Seller Group Taxes), and the Seller Group shall retain and shall be responsible for, all Liabilities that are not Assumed Liabilities, including all Liabilities related to Excluded Assets or any other Liabilities of the Business (all such Liabilities not being assumed herein referred to as the "**Excluded Liabilities**"). The Excluded Liabilities shall exclude any amounts payable or due to Merck for the assignment by Seller to Purchaser of the Merck License Agreement, respectively, whether arising in periods before or following the Closing, which shall be solely borne by Purchaser.
- 2.5 **Excluded Contracts**. Purchaser is electing to purchase only the Assigned Contracts, and Purchaser is not purchasing any other Contract of the Seller Group (any such other Contract an "**Excluded Contract**"). The Excluded Contracts shall constitute Excluded Assets and shall not be included in the Transferred Assets for any purposes of this Agreement and Purchaser shall not have any obligation to satisfy or pay any Cure Amounts or other Liabilities with respect to Excluded Contracts.

2.6 Nontransferable Assets and Liabilities.

- (a) Notwithstanding any other provision of this Agreement to the contrary, this Agreement shall not constitute an agreement to assign or transfer any Transferred Asset or any claim, right or benefit arising thereunder or resulting therefrom if an attempted assignment or transfer thereof, without the Consent of a third party (including any Governmental Authority) (after giving effect to the Sale Order or any other applicable order of the Bankruptcy Court that effects such transfer without any required Consents), would constitute a breach or other contravention thereof or a violation of Applicable Law (each, a "Non-Transferred Asset").
- If, on the Closing Date, any third-party Consent is not obtained for a Non-Transferred Asset, or if an attempted transfer or assignment thereof would be ineffective or a violation of Applicable Law, then, until any requisite consent is obtained therefor and the same is transferred and assigned to Purchaser or its designee, each such Non-Transferred Asset shall be held by the Seller as agent for Purchaser, and the Seller shall, to the extent permitted by Applicable Law, provide to Purchaser the benefits and Purchaser shall assume the obligations and bear the economic burdens associated with such Non-Transferred Asset. The Seller and Purchaser shall use commercially reasonable efforts to enter into agreements (including subcontracting, sublicensing or subleasing, if permitted) by which (i) the Seller shall, at Purchaser's sole expense, without interruption of the Business, provide Purchaser with the economic and operational equivalent of obtaining the requisite third-party Consent and assigning the applicable Non-Transferred Asset to Purchaser (including, with the prior written consent of Purchaser, enforcing for the benefit of Purchaser, and at Purchaser's sole expense, all claims or rights arising thereunder) and (ii) Purchaser shall perform, at its sole expense, the obligations and assume the economic burdens of the Seller or its Affiliates to be performed after the Closing with respect to such Non-Transferred Asset. Purchaser shall promptly, upon receipt of a written request therefor from the Seller, reimburse the Seller for all monies paid by the Seller on Purchaser's behalf in connection with any Assumed Liability not assigned or transferred to Purchaser pursuant to this Section 2.6.
- 2.7 Closing. The closing of the Transactions (the "Closing") will take place remotely by electronic exchange of documents on the date (the "Closing Date") that is the second (2nd) Business Day after the date on which all of the conditions set forth in ARTICLE 8 (excluding conditions that, by their terms, are to be satisfied at the Closing, but subject to the satisfaction or waiver of all such conditions at the Closing), have been satisfied or waived by the party hereto entitled the benefit of the same, unless another time or date is agreed to in writing by the parties hereto. Except as otherwise set forth herein, all proceedings

to be taken and all documents to be executed and delivered by all parties hereto at the Closing will be deemed to have been taken and executed simultaneously and no proceedings will be deemed to have been taken nor documents executed or delivered until all have been taken, executed, and delivered.

2.8 Closing Deliveries of the Parties. On the Closing Date (except as otherwise indicated):

- (a) Purchaser and the Seller shall execute and deliver the Bill of Sale and Assignment and Assumption Agreement;
- (b) Purchaser and the Seller shall execute and deliver the Intellectual Property Assignment Agreement;
 - (c) Purchaser and the Seller shall execute and deliver the Sublicense Agreement;
- (d) Purchaser and the Seller shall transmit Purchaser's FDA Transfer Letter and the Seller's FDA Transfer Letters, respectively, to the FDA and shall take any other actions reasonably necessary to effect the transfer of the Lonafarnib IND from the Seller to Purchaser;
- (e) Purchaser shall deliver, or cause to be delivered, to the Seller or the applicable Person each of the following:
 - (i) a certificate, dated as of the Closing Date, executed by or on behalf of Purchaser as to the satisfaction of the conditions set forth in Section 8.3(a) and Section 8.3(b); and
 - (ii) payment of the closing payments set forth in Section 2.9; and
- (f) Purchaser and the Seller shall deliver, or cause to be delivered, to Purchaser, the Seller or the applicable Person the Merck Side Letter duly executed by Merck, Purchaser, and the Seller; and
- (g) the Seller shall deliver, or cause to be delivered, to Purchaser or the applicable Person each of the following:
 - (i) a certificate, dated as of the Closing Date, executed by or on behalf of the Seller as to the satisfaction of the conditions set forth in Section 8.2(a) and Section 8.2(b); and
 - (ii) an IRS Form W-9 with respect to the Seller, duly completed and executed.
- (h) The "Closing" as defined in that certain Lambda Asset Purchase Agreement, dated the date hereof, by and between the Seller and Purchaser takes place on the Closing Date of this Agreement.

2.9 Purchase Price; Assumed Liabilities; Deposits.

- (a) At the Closing, upon the terms and subject to the conditions set forth herein, in full consideration for the sale, transfer, conveyance, assignment and delivery of the Transferred Assets to Purchaser and assumption of the Assumed Liabilities by Purchaser, Purchaser shall (i) pay to the Seller an aggregate amount equal to the Purchase Price *minus* the Deposit Escrow Amount, which shall be released to the Seller by the Escrow Agent pursuant to Section 2.9(c), by irrevocable wire transfer of immediately available funds in accordance with payment instructions delivered by the Seller to Purchaser prior to the Closing; and (ii) assume the Assumed Liabilities.
- (b) At the Closing, on the terms and subject to the conditions set forth in this Agreement, Purchaser will assume and become responsible for the Assumed Liabilities. Purchaser agrees to pay, perform, honor, and discharge, or cause to be paid, performed, honored and discharged, all Assumed Liabilities in a timely manner in accordance with the terms hereof, including paying or causing to be paid, at or prior to the Closing, all Purchaser Cure Amounts for Assumed Contracts. Seller agrees to pay all Seller Cure Amounts for Assumed Contracts at or prior to the Closing.

- (c) The Deposit Escrow Amount shall be distributed as follows:
- (i) if the Closing shall occur, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available funds, to an account designated by the Seller to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to the Seller at Closing and credited against the amount required to be paid by Purchaser to the Seller at Closing in accordance with Section 2.9(a);
- (ii) if this Agreement is terminated by the Seller pursuant to Section 9.1(g), (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available funds, to an account designated by the Seller to the Escrow Agent and (B) the Deposit Escrow Amount, which shall constitute liquidated damages (and not a penalty), shall be delivered to the Seller within two (2) Business Days following delivery of such joint written instruction; or
- (iii) if this Agreement is validly terminated for any reason in accordance with the terms of this Agreement other than (x) by the Seller pursuant to Section 9.1(g) or (y) if Purchaser forfeits the Deposit Escrow Amount to the Seller pursuant to Section 8.5, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to Purchaser, by irrevocable wire transfer of immediately available funds, to an account designated by Purchaser to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to Purchaser within two (2) Business Days following delivery of such joint written instruction.

Any issue regarding the entitlement to the Deposit Escrow Amount shall be determined by the Bankruptcy Court, and Purchaser consents to the jurisdiction of the Bankruptcy Court for any issue related to this Agreement.

2.10 **Transfer Taxes**. Purchaser shall be solely responsible for, and shall indemnify, defend, and hold harmless the Seller Group for, any transfer, documentary, sales, use, excise, stock transfer, value-added, stamp, recording, registration and other similar taxes, levies and fees (including any penalties, fines and interest), together with any conveyance fees, recording charges and other similar fees and charges, incurred in connection with this Agreement and the Transactions (collectively, "**Transfer Taxes**"). Purchaser and the Seller shall cooperate in good faith to minimize, to the extent permissible under Applicable Law, the amount of any Transfer Taxes due with respect to the Transactions.

2.11 Allocation of Purchase Price.

(a) The Purchase Price (including all other amounts treated as consideration for U.S. federal income tax purposes) and Assumed Liabilities shall be allocated as set forth on Schedule 2.11(a)(the "Preliminary Allocation Schedule"). Within ninety (90) days following the final determination of the Purchase Price, Purchaser shall deliver to the Seller a schedule allocating the Purchase Price (and all other amounts treated as consideration for U.S. federal income tax purposes) among the Transferred Assets (the "Allocation Schedule"). The Allocation Schedule shall be reasonable and shall be prepared in accordance with the Preliminary Allocation Schedule, and Purchaser and the Seller shall negotiate in good faith to resolve disputed items, if any, in the Allocation Schedule as promptly as practicable. If Purchaser and the Seller are unable to reach agreement with respect to the Allocation Schedule within thirty (30) days after the delivery of the Allocation Schedule by Purchaser to

the Seller, the parties shall be entitled to use their own Purchase Price allocations for Tax reporting purposes.

- (b) To the extent Purchaser and the Seller agree on the Allocation Schedule pursuant to Section 2.11(a), Purchaser and the Seller shall (i) timely file all Tax Returns required to be filed in connection with the Allocation Schedule, and (ii) prepare and file all Tax Returns and determine all Taxes in a manner consistent with the Allocation Schedule, except as may be required by Applicable Law and except as may be necessary to reflect adjustments to the Allocation Schedule resulting from post-Closing payments or events. Purchaser, on the one hand, and the Seller, on the other hand, shall notify the other if it receives notice that any Governmental Authority proposes any allocation different from Allocation Schedule.
- 2.12 **Escrow Accounts**. At the Closing, the Deposit Escrow Amount shall be used to satisfy a portion of the payment obligations of Purchaser pursuant to Section 2.9(c), otherwise the Deposit Escrow Amount shall be released to Purchaser or the Seller pursuant to Section 2.9(c). Upon the final release of all of the Deposit Escrow Amount pursuant to the terms of this Agreement and the Escrow Agreement, the Escrow Agreement shall automatically terminate. Any fees owed to the Escrow Agent and obligations under the Escrow Agreement shall be borne by Purchaser. The Deposit Escrow Amount shall be held in trust for the benefit of the Seller and shall not be subject to any encumbrance, attachment, trustee process or any other judicial process of any creditor of any party hereto, and shall be held and disbursed solely for the purposes of and in accordance with the terms of this Agreement and the Escrow Agreement.
- 2.13 **Tax Withholding**. Notwithstanding anything in this Agreement to the contrary, Purchaser shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any Person such amounts as it is required to deduct and withhold from such Person with respect to the making of such payment under the Code and the rules and regulations promulgated thereunder, or any provision of any Applicable Law relating to Taxes; *provided, however*, that Purchaser shall (i) provide commercially reasonable notice to the Person prior to such deduction and withholding and (ii) afford the Person a reasonable opportunity to provide any additional information, forms or certifications to establish an exemption from, or obtain a reduced rate of, withholding. To the extent that amounts are so withheld and properly remitted by Purchaser, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to such Person in respect of which such deduction and withholding was made by Purchaser.

ARTICLE 3. REPRESENTATIONS AND WARRANTIES OF THE SELLER

Except as disclosed in a document herewith delivered by the Seller to Purchaser (the "Schedules"), the Seller hereby makes the representations and warranties contained in this ARTICLE 3 to Purchaser. Organization, Good Standing and Other Matters. Each member of the Seller Group is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has, subject to the necessary authority of the Bankruptcy Court, the requisite corporate power and authority to operate the Business and necessary to own, lease or operate the properties and assets owned, leased or operated by it to carry on the Business as now being conducted, except where the failure to be so duly organized, validly existing and in good standing, or to have such power and authority, would not, individually or in the aggregate, have a Material Adverse Effect. Each member of the Seller Group is duly qualified to do business as a foreign company in each jurisdiction in which the nature of the Business as currently conducted by it or the property owned or leased by it makes such qualification necessary, except where the failure to be so qualified would not, individually or in the aggregate, have a Material Adverse Effect.

3.2 **Authority and Enforceability**. Subject to Bankruptcy Court approval, the Seller has all requisite power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to

consummate the Transactions. The execution, delivery and performance of this Agreement and the each of the Related Documents to which the Seller is (or at Closing, will be) a party thereto, and the consummation by the Seller of the Transactions, has been duly authorized and approved by all necessary limited liability company action on the part of the Seller and are subject to the approval of the Bankruptcy Court. This Agreement has been, and each Related Document will be, at or prior to the Closing, duly executed and delivered by the Seller and, assuming the due execution and delivery by the other parties hereto or thereto, and subject to the approval of the Bankruptcy Court, constitutes a valid and binding obligation of the Seller, enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.

- No Conflict; Required Filings and Consents. Except (a) such filings as may be required in connection with the Transfer Taxes described in Section 2.10 and (b) as otherwise set forth on Schedule 3.3, the execution and delivery of this Agreement by the Seller does not and the execution and delivery of the Related Documents by the Seller will not, and the consummation of the Transactions hereby and thereby will not (i) violate the provisions of the Organizational Documents of any member of the Seller Group, (ii) subject to the entry of the Sale Order, violate any Applicable Law or Order to which any member of the Seller Group is subject or by which its properties or assets are bound, (iii) require any member of the Seller Group to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date (except as required by the Bankruptcy Code or the Sale Order), (iv) subject to the entry of the Sale Order, result in a breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any Assigned Contract or (v) subject to the entry of the Sale Order, result in the imposition or creation of any Lien upon or with respect to any of the assets or properties of the Seller Group; excluding from the foregoing clauses (ii) through (v) any Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, have a Material Adverse Effect.
- **Compliance With Laws.** To the Seller's Knowledge, (i) the Seller Group is conducting the Business in compliance in all material respects with all material Applicable Laws applicable to the Business and (ii) no member of the Seller Group has received any written notice since the Petition Date of any material violations of any material Applicable Law applicable to their conduct of the Business. As of the Agreement Date, the Seller has and, to the Seller's Knowledge, has obtained all permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals of the FDA or any other Governmental Authority, currently used in, necessary for and material to the Development, Manufacture, and Commercialization of all Lonafarnib Antiviral Products in the Lonafarnib Antiviral Field as presently conducted, all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals are included in the Transferred Assets and Seller has made available to Purchaser true and complete copies of all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. As of the Agreement Date, neither Seller nor, to the Seller's Knowledge, any other Person has received any communication from any Governmental Authority that threatens to withdraw or suspend any such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. Seller has filed with the applicable Governmental Authority all required filings, declarations, listings, registrations, reports or submissions, including adverse event reports, necessary for and material to the Development, Manufacture, and Commercialization of the Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field as presently conducted. All relevant filings, declarations, listings, registrations, reports or submissions were in material compliance with Applicable Law when filed, and no deficiencies have been asserted by any Governmental Authority with respect to any such filings, declarations, listing, registrations, reports or submissions. As of the Agreement Date, the Seller has not received or been subject to: (1) any FDA Form 483s directly relating to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field; (2) any FDA notices of adverse findings relating to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field; or (3) any warning letters or other correspondence from the FDA or any other

Governmental Authority in which the FDA or such other Governmental Authority asserted that the actions of Seller, with respect to any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field, were not in compliance with Applicable Laws. There has not been any occurrence of any product recall, market withdrawal or replacement, or post-sale warning conducted by or on behalf of the Seller concerning any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field or, to the Seller's Knowledge, any product recall, market withdrawal or replacement conducted by or on behalf of any entity as a result of any alleged defect in any Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field.

- 3.5 **Permits**. To the Seller's Knowledge, (i) the Seller Group possess all material Permits required for the operation of the Business as currently conducted (the "**Seller Permits**") and (ii) no member of the Seller Group has received as of the Agreement Date any written notice of any cancellation, suspension, revocation, invalidation or non-renewal of any Permit since the Petition Date.
- 3.6 **Litigation**. As of the Agreement Date, there is no Action pending or, to the Seller's Knowledge, formally threatened in writing, against any member of the Seller Group before any Governmental Authority that would have a Material Adverse Effect or affect the Transferred Assets in any material respect after the entry of the Sale Order, if determined adversely and after taking into effect applicable insurance coverage.
 - 3.7 **Real Property**. The Seller Group does not own any real property.
- Assigned Contracts. With respect to the Assigned Contracts, (i) except as a result of, or arising in connection with, the filing of the Bankruptcy Cases, no member of the Seller Group has received any written notice of any default or event that (with due notice or lapse of time or both) would constitute a default by the applicable member of the Seller Group under any Assigned Contract, other than defaults that have been cured or waived in writing or would not reasonably be expected to have a Material Adverse Effect, (ii) to the Seller's Knowledge, each Assigned Contract is a legal, valid and binding obligation of the applicable member of the Seller Group and is in full force and effect (except to the extent subject to, and limited by, the Enforceability Exceptions), (iii) to the Seller's Knowledge, no other party to any Assigned Contract is (with or without the lapse of time or the giving of notice, or both) in material breach of or in material default under any Assigned Contract and (iv) to the Seller's Knowledge, no member of the Seller Group has provided or received any notice of any intention to terminate any Assigned Contract. The Seller has made available to Purchaser true, correct and complete copies of each of the Assigned Contracts listed on Schedule 2.1(a), together with all amendments thereto.
- Report on Form 10-K filed with SEC on April 8, 2024 (the "Seller Financial Statements") have been prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements or, in the case of unaudited financial statements, except as permitted by the SEC on Form 10-K under the Exchange Act, and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments), have been prepared on a consistent basis throughout the periods covered thereby and presents fairly in all respects the financial condition of the Seller as of such dates and the results of operations of Seller for such periods, and are consistent with the books and records of Seller (which books and records are correct and complete in all material respects).
- 3.10 **Absence of Material Developments**. Except as disclosed on Schedule 3.10, since the Petition Date, there has occurred no fact, event, condition, change or circumstance which has had or would reasonably be expected to have a Material Adverse Effect.
- 3.11 **Customers and Suppliers.** Except as disclosed on Schedule 3.11(a), to the Knowledge of the Seller, since the Petition Date, no customer has or has threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions, purchasing materials, products or services from the Business. Except as disclosed on Schedule 3.11(b), to the Knowledge of the Seller, no supplier has or has

threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions. supplying materials, products or services to the Business.

3.12 **Intellectual Property.**

- (a) A true, correct and complete list of all Intellectual Property Registrations included in the Owned Intellectual Property Assets is set forth on Schedule 3.12(a), including the Trademarks and domain names pertaining to Lonafarnib Antiviral Products in the Lonafarnib Antiviral Field that are owned by the Seller or its Affiliates as of the Agreement Date.
- (b) The Seller Group exclusively owns all Owned Intellectual Property Assets. Except as set forth on Schedule 3.12(b), and no member of the Seller Group is a party to, or bound by, (i) any license, royalty agreement, or other agreement relating to the use of any material Business Intellectual Property (other than non-exclusive licenses grant to a member of the Seller Group for commercially available, unmodified, off-the-shelf software licensed for aggregate annual fees of less than \$50,000), and (ii) agreements pursuant to which a member of the Seller Group settled any action, litigation, suit or other judicial or administrative proceeding, claim, assertion, or threat with respect to any material Business Intellectual Property, including settlement agreements, coexistence agreements, and consent agreements.
- (c) Other than with respect to Excluded Contracts or Assigned Contracts that Purchaser does not ultimately assume, no current or former Affiliate, partner, director, stockholder, officer, member, manager, employee, consultant or contractor of the Seller Group will, after giving effect to the Transactions, own, license or retain any Owned Intellectual Property Assets.
- (d) All material Intellectual Property Registrations remain pending or in full force and effect and have not expired or been abandoned or cancelled. To Seller's Knowledge, no interference, opposition, reissue, reexamination, or other proceeding is or has been pending or threatened, in which the scope, validity, or enforceability of any material Owned Intellectual Property Assets is being, has been challenged.
- (e) To the Knowledge of the Seller, the conduct of the Business does not infringe, misappropriate or otherwise violate in any material respect any Person's Intellectual Property.
- (f) To the Knowledge of the Seller's, no Person is currently infringing, misappropriating or otherwise violating any material Owned Intellectual Property Assets.
- (g) The Seller Group has taken commercially reasonable steps to safeguard and maintain the confidentiality of all trade secrets that constitute Owned Intellectual Property Assets, including by using good faith efforts to require all Persons having access thereto to execute written non-disclosure agreements.
- (h) The Seller Group complies with all Applicable Laws, internal policies and contractual obligations relating to privacy, data protection and cybersecurity.
- 3.13 **Taxes.** The Seller Group has timely filed all Tax Returns that it was required to file with respect to Transferred Assets. All such Tax Returns were correct and complete in all material respects. All Taxes owed by the Seller Group (whether or not shown or required to be shown on any Tax Return) with respect to Transferred Assets have been paid. There are no Liens on any of the Transferred Assets that arose in connection with any failure (or alleged failure) to pay any Tax. There is no dispute, examination, judicial proceeding or claim concerning any Taxes of the Seller Group with respect to the Transferred Assets.
- 3.14 **Product Liability.** Except as disclosed on Schedule 3.14, within the three (3) year period prior to the Closing Date there has not been any, and as of the Closing Date there is no pending, material litigation commenced against any member of the Seller Group relating to the sale, distribution or use of any

item sold or used in the Business (the "Goods"), including litigation with respect to product liability or recall claims.

- 3.15 **Product Warranties; Product Returns.** Except for warranties arising solely pursuant to Applicable Law or in the ordinary course of business, (a) no member of the Seller Group has made any material warranties, express or implied, written or oral, to any third party with respect to any of the Goods within the three (3) year period prior to the Closing Date, and (b) there is no, and within the three (3) year period prior to the Closing Date there has not been any, material litigation pending or, to the Seller's Knowledge, threatened with respect to any such warranty.
- 3.16 **Brokers and Finders**. Except for SSG Advisors, LLC, the Seller has not, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.
- 3.17 **Virology Studies**. Each Virology Study is on-going, has been conducted in a professional manner, in accordance with industry standards, and in compliance with all Applicable Laws, there has not been any interruption, suspension, or delay in the conduct of each such Virology Study, and all payments payable to each Virology Collaborator in connection with each such Virology Study has been duly and timely paid in full.
- 3.18 **Inventory**. To Seller's Knowledge, the Inventory consists of all materials used to Manufacture or otherwise incorporated into the Licensed Product (including raw materials and active pharmaceutical ingredients) and inventory of Licensed Product exclusively owned by the Seller and its Affiliates as of the Closing Date. As of the Closing Date, Schedule 3.18 identifies the location of all Inventory and sets forth a complete and accurate list of all Storage Contracts and provides reasonable details with respect to the Inventory subject to each such Storage Contract.
- 3.19 **No Other Representations or Warranties.** Except for the representations and warranties contained in this ARTICLE 3 and the Related Documents, the Seller does not, nor do any other Persons on behalf of the Seller, make any other express or implied representation or warranty with respect to itself, the Business, the Transferred Assets or the Assumed Liabilities, or with respect to any other information provided to Purchaser or its representatives, and the Seller disclaims any other representations or warranties, whether made by or on behalf of the Seller or any other Person. The Seller will not, and no other Persons will, have or be subject to any Liability to Purchaser or any other Person resulting from the distribution to Purchaser, or Purchaser's use of, any such information, including any information, documents, projections, forecasts or other material made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever (electronic or otherwise) or otherwise in expectation of the Transactions.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as disclosed in a document herewith delivered by Purchaser to the Seller (the "**Purchaser Schedules**"), Purchaser hereby makes the representations and warranties contained in this ARTICLE 4 to the Seller.

4.1 **Organization, Good Standing and Other Matters**. Purchaser is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has all requisite corporate power or other entity power and authority to own its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed to conduct its business as currently conducted and is in good standing in each jurisdiction in which the location of the property owned, leased or operated by it or the nature of its business makes such qualification necessary, except where the failure

to be so qualified or licensed would not, individually or in the aggregate, materially impair or delay Purchaser's ability to consummate the Transactions.

- Authority and Enforceability. Purchaser has all requisite corporate power or other entity power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party, and the consummation of the Transactions, have been duly authorized and approved by its board of directors (or equivalent governing body) and no other action on the part of Purchaser or its members is necessary to authorize the execution, delivery and performance of this Agreement and the Related Documents by Purchaser and the consummation of the Transactions. This Agreement has been, and each Related Document will be at or prior to Closing, duly executed and delivered by Purchaser and, assuming the due execution and delivery by the other parties hereto or thereto, constitutes a valid and binding obligation of Purchaser enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.
- 4.3 **No Conflict: Required Filings and Consents.** Except (a) such filings as may be required in connection with the Transfer Taxes described in Section 2.10 and (b) as set forth on Schedule 4.3, the execution and delivery of this Agreement and of the Related Documents and the consummation of the Transactions by Purchaser will not (i) violate the provisions of its Organizational Documents, (ii) violate any Applicable Law or Order to which it is subject or by which any of its properties or assets are bound, (iii) require it to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date, (iv) result in a material breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any material Contract to which it is a party or (v) result in the imposition or creation of any Lien upon or with respect to any of its assets or properties; excluding from the foregoing clauses (ii) through (v) Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, (A) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (B) otherwise prevent, hinder or delay the consummation of the Transactions.
- 4.4 **Financing**. Purchaser has, and at the Closing will have, (a) sufficient internal funds (without giving effect to any unfunded financing regardless of whether any such financing is committed) available to pay the Purchase Price in accordance with the terms hereof and any other payments required hereunder and any expenses incurred or required to be paid by Purchaser in connection with the Transactions, and (b) the resources and capabilities (financial or otherwise) to perform its obligations hereunder and under the Related Documents. Purchaser has not incurred any obligation, commitment, restriction, or Liability of any kind, which would impair or adversely affect such resources and capabilities.
- 4.5 **Solvency**. Purchaser is not entering into this Agreement with the intent to hinder, delay or defraud either present or future creditors. Immediately after giving effect to all of the Transactions, including the making of the payments contemplated by Section 2.9, and assuming satisfaction of the conditions to Purchaser's obligation to consummate the Transactions as set forth herein, the accuracy of the representations and warranties of Purchaser set forth herein and the performance by Purchaser of its obligations hereunder in all material respects, Purchaser will be Solvent.
- 4.6 **Litigation**. There is no Action pending or, to Purchaser's Knowledge, formally threatened against Purchaser or involving any of its properties or assets that would be reasonably be expected to

- (a) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (b) otherwise prevent, hinder, or delay the consummation of the Transactions.
- 4.7 **Brokers and Finders**. None of Purchaser or its Affiliates have, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.

4.8 Non-Reliance of Purchaser; No Other Representations and Warranties.

- Except for the specific representations and warranties expressly made by the Seller in ARTICLE 3 and Related Documents as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, and the representations and warranties made by the Seller in the Sublicense Agreement or any Related Document, Purchaser acknowledges and agrees that (i) the Seller is not making and have not made any representation or warranty, expressed or implied, at law or in equity, in respect of the Business, the Transferred Assets, the Assumed Liabilities, or any of its operations, prospects or condition (financial or otherwise), including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any Liabilities. the prospects of the Business, the effectiveness or the success of any operations, or the accuracy or completeness of any confidential information memoranda, documents, projections, material or other information (financial or otherwise) regarding the Business furnished to Purchaser or its representatives or made available to Purchaser and its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever, and (ii) no officer, director, manager, stockholder, agent, Affiliate, advisor, representative or employee of the Seller Group has any authority, express or implied, to make any representations, warranties or agreements not specifically set forth in ARTICLE 3 and subject to the limited remedies herein provided, or any representations, warranties or agreements not specifically set forth in the Sublicense Agreement or any Related Document.
- (b) Other than the specific representations and warranties expressly set forth in ARTICLE 3 as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, and the representations and warranties made by the Seller in the Sublicense Agreement or any Related Document, Purchaser specifically disclaims that it is relying upon or has relied upon any such other representations or warranties that may have been made by any Person, and acknowledges and agrees that the Seller and the Seller's Affiliates have specifically disclaimed and do hereby specifically disclaim, and shall not have or be subject to any Liability for reliance on any such other representation or warranty made by any Person. Purchaser specifically waives any obligation or duty by the Seller or the Seller's Affiliates to make any disclosures of fact not required to be disclosed pursuant to the specific representations and warranties expressly set forth in ARTICLE 3 or in the Sublicense Agreement or any Related Document and disclaim reliance on any information not specifically required to be provided or disclosed pursuant to the specific representations and warranties set forth in ARTICLE 3 or in the Sublicense Agreement or any Related Document.
- (c) Purchaser is acquiring the Business, the Transferred Assets and the Assumed Liabilities subject only to the specific representations and warranties expressly set forth in ARTICLE 3 as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, and the representations and warranties expressly set forth in the Sublicense Agreement or any Related Document.
- 4.9 **No Other Representations or Warranties.** Except for the representations and warranties contained in this ARTICLE 4, neither Purchaser nor any other Person on behalf of Purchaser makes any other express or implied representation or warranty with respect to Purchaser or with respect to any other information provided to the Seller or its representatives, and Purchaser disclaims any other representations

or warranties, whether made by Purchaser or any of its Affiliates, officers, directors, employees, agents or representatives.

ARTICLE 5. BANKRUPTCY COURT MATTERS

5.1 **Competing Transaction**. This Agreement is subject to approval by the Bankruptcy Court and the consideration by the Seller of higher or better competing bids in respect of all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) in accordance with the terms of the Bid Procedures Order (each, a "**Competing Bid**"). From the Agreement Date (and any prior time) and until the Closing, the Seller is permitted to, and to cause its representatives to, initiate contact with, solicit or encourage submission of any inquiries, proposals or offers by, any Person (in addition to Purchaser and its Affiliates and representatives) in connection with any sale or other disposition of the Transferred Assets. In addition, the Seller shall have the authority to respond to any inquiries or offers to purchase all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) and perform any and all other acts related thereto which are required under the Bankruptcy Code, the Bid Procedures Order or other Applicable Law, including supplying information relating to the Business and the assets of the Seller Group to prospective purchasers.

5.2 Bankruptcy Court Filings.

- Subject to its right to pursue a Competing Bid in accordance with the Bid Procedures Order, the Seller shall diligently pursue the entry by the Bankruptcy Court of the Sale Order, which Sale Order shall provide for the transfer of the Transferred Assets and the Assumed Liabilities to Purchaser free from all successor or transferee Liability to the fullest extent permitted by Section 363 of the Bankruptcy Code. The Seller shall comply (or obtain an Order from the Bankruptcy Court waiving compliance) with all requirements under the applicable provisions of the Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the Local Bankruptcy Rules for the Bankruptcy Court in obtaining the entry of the Sale Order. The Seller further covenants and agrees that, after entry by the Bankruptcy Court of the Sale Order, and provided that the Sale Order becomes a Final Order, the terms of any other proposed order submitted by the Seller to the Bankruptcy Court shall not conflict with, supersede, abrogate, nullify or restrict the terms of this Agreement, or in any way prevent or interfere with the consummation or performance of the Transactions. Purchaser agrees that it will promptly take such actions as are reasonably requested by the Seller to assist in obtaining entry of the Sale Order, including by furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Purchaser under this Agreement and demonstrating that Purchaser is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. In the event, if the entry of the Sale Order shall be appealed, the Seller and Purchaser shall use their respective commercially reasonable efforts to defend such appeal.
- (b) Seller shall use commercially reasonable efforts to provide Purchaser with a reasonable opportunity to review and comment upon all motions, applications, and supporting papers relating to the transactions contemplated by this Agreement prepared by Seller or any Affiliates (including forms of orders and notices to interested parties) prior to the filing thereof in the Bankruptcy Cases; provided that the foregoing shall not require the Seller to take any action that would, in Seller's reasonable business judgment, threaten to harm the overall value to be produced by the Seller's in-court sale process.
- (c) The form of Sale Order submitted by the Seller to the Bankruptcy Court for approval shall be in a form and substance reasonably acceptable to Purchaser.
- (d) Seller shall not seek any modification to the Bid Procedures, Bid Procedures Order, or Sale Order by the Bankruptcy Court that are materially adverse to Purchaser without the prior written consent of Purchaser, which such consent shall not be unreasonably withheld.

- (e) Each of Purchaser and Seller will promptly take such actions as are reasonably requested by the other party to assist in obtaining entry of the Sale Order, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for purposes, among others, of providing necessary assurances of performance by Seller of its obligations under this Agreement and demonstrating that Purchaser is a good faith buyer under section 363(m) of the Bankruptcy Code.
- (f) Seller shall use commercially reasonable efforts to provide appropriate notice of the hearings on the Sale Order to all Persons entitled to notice, including, but not limited to, all Persons that have asserted Liens on the Transferred Assets, all parties to the Assigned Contracts and all taxing authorities in jurisdictions applicable to Seller and as otherwise required by the Bankruptcy Code and bankruptcy rules.
- (g) Within five (5) Business Days of the Auction (subject to the Bankruptcy Court's availability), if Purchaser is the successful bidder at the Auction (or if there is no Auction), Seller will seek entry of the Sale Order by the Bankruptcy Court.
- (h) The Seller and Purchaser agree that, in the event that Purchaser is not the winning bidder at an auction undertaken pursuant to the Bid Procedures Order (the "Auction"), and (i) Purchaser submits the Back-Up Bid at the Auction or (ii) the terms of this Agreement are deemed to constitute a Back-Up Bid, then Purchaser shall be obligated to promptly consummate the Transactions upon the terms and conditions as set forth herein, including the payment of the Purchase Price as the same may be increased by Purchaser at the Auction; provided that the Seller gives written notice to Purchaser on or before the Back-up Termination Date, stating that the Seller (A) failed to consummate the sale of the Transferred Assets with the winning bidder, and (B) terminated the purchase agreement with the winning bidder.

5.3 Assumption of Assigned Contracts.

- (a) On June 4, 2024, the Seller filed (or caused to be filed) a notice of assumption (the "Assumption Notice") with the Bankruptcy Court and served such notice on each counterparty to a Contract listed thereon. The Assumption Notice identified all Contracts that the Seller and Purchaser believe may be assumed and assigned in connection with the sale of the Transferred Assets and set forth a good faith estimate of the amount of Cure Amounts applicable to each such Contract (and if no Cure Amount is estimated to be applicable with respect to any particular Contract, the amount of such Cure Amount designated for such Contract shall be "\$0.00"). In accordance with the Bid Procedures Order, the Seller reserves the right to supplement such list of Contracts and provide additional notice of assumption.
- (b) On or before the date that is three (3) Business Days before the Closing Date (the "Designation Deadline"), Purchaser shall provide to the Seller a Contracts List, which shall identify all Contracts that Purchaser elects to have assumed and assigned to Purchaser on the Closing Date (and with respect to the Existing Manufacturing Contracts, assumed and assigned to Purchaser which will be automatically effective as of the applicable Existing Manufacturing Contract Transfer Date without further notice). For the avoidance of doubt, Purchaser shall be entitled, with the agreement of Seller, to add, or, in the sole discretion of Purchaser, to remove, any Contracts from the Contracts List at any time (or multiple times) prior to the Designation Deadline by providing to the Seller by email a copy of the amended Contracts List. Any Contracts List that Purchaser delivers to the Seller prior to the Designation Deadline shall be deemed to replace and supersede any Contracts List that Purchaser had previously delivered to Seller. For the avoidance of doubt, only those Contracts that are identified on the Contracts List as of the Designation Deadline shall constitute Assigned Contracts and will be assumed by the Seller and assigned to Purchaser pursuant to the Sale Order. The Seller shall file such motions or pleadings as may be appropriate or necessary to assume and assign the Assigned Contracts and to determine the amount of the Cure Amounts; provided that nothing herein shall preclude the Seller from filing one or

more motions to reject any Contracts that are not identified on the Contracts List as of the Designation Deadline.

- (c) Notwithstanding any provision in this Agreement to the contrary, a Contract shall not be an Assigned Contract hereunder and shall not be assigned to, or assumed by, Purchaser to the extent that such Contract is (i) deemed rejected under Section 365 of the Bankruptcy Code, (ii) the subject of an objection to assignment or assumption or requires the consent of any Governmental Authority or other third party (other than, and in addition to, the Bankruptcy Court) in order to permit the assumption and assignment by the applicable Seller to Purchaser of such Contract pursuant to Section 365 of the Bankruptcy Code, and such objection has not been resolved or such consent has not been obtained prior to the thirtieth (30th) day following the Closing Date (as such period may be extended by mutual agreement of Seller and Purchaser), or (iii) terminated by any party thereto other than Seller, or terminates or expires by its terms, on or prior to such time as it is to be assumed by and assigned to Purchaser as an Assigned Contract hereunder and is not continued or otherwise extended upon assumption. In no event shall the failure to assign to Purchaser any Contract in accordance with subsections (i) through (iii) above reduce the Purchase Price payable to Seller or constitute a failure to satisfy the conditions precedent of Seller under Section 8.3.
- (d) Subject to the terms of Section 2.5, Section 2.8, Section 5.3(a) and Section 5.3(b), and subject to the entry of an order (which may be the Sale Order) of the Bankruptcy Court authorizing the assignment to Purchaser of the Assigned Contracts, Purchaser shall make provision for the payment of the Purchaser Cure Amounts for Assumed Contracts, and Seller shall make provision for the payment of the Seller Cure Amounts for Assumed Contracts, in cash at Closing in accordance with the Sale Order.
- (e) Notwithstanding any provision in this Agreement to the contrary, from and after the date of the Assumption Notice through the Closing Date, the Seller will not reject or take any action (or fail to take any action that would result in rejection by operation of Applicable Law) to reject, withdraw, repudiate or disclaim any Assigned Contract unless (i) Purchaser has provided its prior written consent; or (ii) Purchaser has removed such Assigned Contract from the list of Assigned Contracts.
- **Disputed Contracts.** In the event of an objection by a Contract counterparty to the Cure Amount asserted by Seller with regard to any Contract on the Contract List (such contract, a "Disputed Contract"), Seller shall either settle the objection of such party or shall litigate such objection under procedures as established by the Bankruptcy Court. In no event shall the Seller settle a Cure Amount objection with regard to any potential Assigned Contract without the express written consent (such consent not to be unreasonably withheld) of Purchaser (with an email consent being sufficient). In the event that a dispute regarding the Cure Amounts with respect to a Contract has not been resolved as of the Closing, the parties shall nonetheless remain obligated to consummate the transactions contemplated by this Agreement. Upon entry of an Order of the Bankruptcy Court (if necessary) determining any Cure Amount and authorizing the assumption and assignment to Purchaser of such Disputed Contract after the Closing, which order shall be in form and substance acceptable to Purchaser (a "Disputed Contract Order"), Purchaser shall have the option to designate the Disputed Contract as an Assigned Contract or an Excluded Contract (regardless of whether such contract was identified on the Contracts List). If Purchaser elects to designate the Disputed Contract as an Excluded Contract, (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the Sale Order and this Agreement, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract. If Purchaser elects to designate the Disputed Contract as an Assigned Contract, such Disputed Contract shall be deemed an Assigned Contract for all purposes hereunder and, for the avoidance of doubt, Purchaser shall assume the Disputed Contract and shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Disputed Contract; and (if applicable) Seller shall be responsible for paying all related Seller Cure Amounts; provided, however, that if Purchaser does not designate such Disputed Contract as either an Excluded Contract or an Assigned Contract within five (5) Business Days after the date of the Disputed Contract Order (or such later date as agreed by the Seller and Purchaser), (a) such

Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the Sale Order and this Agreement, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract.

5.5 Previously Unknown and Previously Excluded Contracts.

(a) If at any time, prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, it is discovered that a Contract material to the operation of the Business should have been identified on the Assumption Notice but was not so listed (any such Contract, a "Previously Unknown Contract"), Seller shall, promptly following the discovery thereof (but in no event later than five (5) Business Days following the discovery thereof), notify Purchaser in writing of such Previously Unknown Contract and provide Purchaser with a copy of such Previously Unknown Contract and the Cure Amount (if any) in respect thereof. Purchaser shall thereafter deliver written notice to Seller (email being sufficient), no later than ten (10) Business Days following such notice of such Previously Unknown Contract from Seller, if Purchaser elects for such Previously Unknown Contract to be an Assigned Contract. If Purchaser elects for a Previously Unknown Contract to be an Assigned Contract in accordance with this Section, then to the extent not previously filed and served, Seller shall file and serve an assignment and assumption notice on the Contract counterparty to such Previously Unknown Contract (a "Supplemental Assignment Notice") notifying such Contract counterparty of Seller's intention to assume and assign to Purchaser such Previously Unknown Contract, including the proposed Cure Amount (if any). Such notice shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser (the "Supplemental Assignment Notice Objection Deadline"). Following expiration of the Supplemental Assignment Notice Objection Deadline and, if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under this Agreement and the Sale Order. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under this Agreement.

At any time prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, Purchaser may elect to take an assignment of any Excluded Contract that has not yet been assumed and assigned pursuant to an order of the Bankruptcy Court (a "Previously Excluded Contract") by sending a written notice to Seller (email being sufficient) of such election. If Purchaser elects for a Previously Excluded Contract to be an Assigned Contract in accordance with this Section, then to the extent not previously filed and served, Seller shall file and serve a Supplemental Assignment Notice on the Contract counterparty to such Previously Excluded Contract. Such Supplemental Assignment Notice Objection Deadline shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser. Following expiration of the Supplemental Assignment Notice Objection Deadline and if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under this Agreement and the Sale Order and, subject to Section 7.15 with respect to Cross-Over Contracts, the Purchaser shall be responsible for satisfying or paying any Cure Amounts or other Liabilities with respect to such Contract, whether or not such Cure Amounts or other Liabilities exceed the Purchaser Cure Amounts. For the avoidance of doubt, the Cross-Over Contracts are not Previously Excluded Contracts. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under this Agreement.

(c) Seller and Purchaser agree that the Sale Order shall contain language approving the assumption and assignment procedures with respect to Disputed Contracts, Previously Unknown Contracts and Previously Excluded Contracts as set forth in Sections 5.3(b), 5.4 and 5.5 hereof.

ARTICLE 6. PRE-CLOSING COVENANTS

- Conduct of Business. Except (i) as set forth on Schedule 6.1, (ii) as may be approved by Purchaser (which approval will not be unreasonably withheld, delayed or conditioned; *provided, however*, that the consent of Purchaser shall be deemed to have been given if Purchaser does not object within forty-eight (48) hours after written request for such consent is provided by the Seller to Purchaser), (iii) for actions taken or omitted to be taken by any member of the Seller Group in response to any Public Health Measure, or (iv) as is otherwise permitted, contemplated or required by this Agreement, any Assigned Contract, by Applicable Laws or by order of the Bankruptcy Court, from the Agreement Date through the earlier of the Closing Date or the termination of this Agreement in accordance with its terms:
 - (a) The Seller Group shall use their commercially reasonable efforts to carry on the Business in all material respects in the ordinary course of business as it has been conducted since the Petition Date; and
 - (b) The Seller shall not, and shall cause its Affiliates not to:
 - (i) sell, license, abandon or otherwise dispose of any material asset or property constituting Transferred Assets other than, in each case, in the ordinary course of business or for the purpose of disposing of obsolete or worthless assets;
 - (ii) except in the ordinary course of business, acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of any business or any corporation, partnership or other business organization or otherwise acquire any assets (except inventory), that as of the Closing would constitute Transferred Assets, except for the acquisition of assets in the ordinary course of business;
 - (iii) change its present accounting methods or principles in any material respect, except as required by GAAP or Applicable Law;
 - (iv) make or change any Tax election, change an annual accounting period, adopt or change any Tax accounting method, file any amended Tax Return, enter into any closing agreement, settle any material Tax claim or assessment or surrender any right to claim a refund of Taxes, other than in the ordinary course of business or as required by the Code or Applicable Law, and in each case that could have a material effect on the amount of Taxes due from the Business or due as a result of the Transferred Assets for a taxable period (or portion thereof) beginning after the Closing Date;
 - (v) compromise or settle any material litigation relating to the Business or cancel or compromise any material claim or waive or release any material right that, in each case, is related to the Business or a Transferred Asset;
 - (vi) encumber, transfer, abandon, allow to lapse, fail to prosecute or maintain, exclusively license, or otherwise dispose of any material Business Intellectual Property or Regulatory Approvals, except, in each case, other than in the ordinary course of business and other than the expiration of the statutory term of any Intellectual Property;
 - (vii) materially modify, materially breach, repudiate, reject, or terminate any Assigned Contract, or waive, release or assign any material rights or claims under any Assigned Contract:

- (viii) grant, impose or suffer to be imposed any Lien upon any of the Transferred Assets other than Permitted Liens or Liens that will be cured prior to the Closing; and
- (ix) authorize, agree or otherwise commit, whether or not in writing, to do any of the foregoing.
- (c) Notwithstanding anything to the contrary, nothing contained in this Agreement shall give Purchaser or any of its Affiliates, directly or indirectly, any right to control or direct the Business, assets and operations prior to the Closing. Prior to the Closing, the Seller shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its Business, assets and operations, subject to Purchaser's consent rights and Seller's obligations, in each case as expressly set forth in this Agreement.

6.2 Access to Information; Confidentiality.

- From the Agreement Date until the earlier of the Closing Date and the termination of this Agreement, the Seller shall grant Purchaser and its representatives (at Purchaser's sole cost and expense) reasonable access, during normal business hours and upon reasonable notice (and in the event of a facility visit request, at least forty-eight (48) hours prior notice), and subject to any limitations resulting from any Public Health Measures, to the personnel, facilities, book and records of the Seller Group related to the Business or the Transferred Assets that are in the possession of, owned by, or under the control (including via license) of the Seller Group, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller Group (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level); provided, however, that (i) all requests for access shall be directed to such other person(s) as the Seller may designate in writing from time to time (the "Seller Access Contact"), (ii) such activities do not unreasonably interfere with the ongoing business or operations of the Seller Group, (iii) the Seller shall have the right to have one or more of its representatives present at all times during any visits, examinations, discussions or contacts contemplated by this Section 6.2(a), (iv) Purchaser shall have no right to perform invasive or subsurface investigations or conduct any sampling or analysis of environmental media of the nature commonly referred to as a "Phase II Environmental Investigation," such as any soil or groundwater testing, (v) such access or related activities would not cause a violation of any agreement to which any Seller Group Member is a party, (vi) no Personal Information shall be disclosed or used other than in compliance with applicable privacy law and (vii) nothing herein shall require any member of the Seller Group or their representatives to furnish to Purchaser or provide Purchaser with access to information that (A) is subject to an attorney-client or an attorney work-product privilege, (B) legal counsel for the Seller Group reasonably concludes may give rise to antitrust or competition law issues or violate a protective order or otherwise may not be disclosed pursuant to Applicable Law (including any Public Health Measure) or (C) would cause significant competitive harm to the Seller Group if the Transactions are not consummated.
- (b) Notwithstanding anything to the contrary contained in this Agreement, from the Agreement Date until the Closing Date, Purchaser shall not, and shall cause its representatives not to, have any contact or discussions concerning any member of the Seller Group, the Business, the Transaction or any other matters with any lender, borrower, creditor, guarantor, business partner, bank, landlord, tenant, supplier, customer, employee, manager, franchisee, distributer, noteholder, independent contractor, consultant or other material business relation of any Seller Group Member, in each case, without the prior written consent of the Seller Access Contact (which consent may be withheld in the Seller's sole discretion and, if given, may be conditioned on the Seller Access Contact or his or her designee having the right to participate in any meeting or discussion); provided, however, that no such consent is required for Purchaser to exercise its rights or perform its obligations under Sections 7.9, 7.10, 7.11, 7.12, 7.13, 7.14, 7.15, and 7.16, to contact Merck in connection with the Merck Side Letter or Sublicense Agreement, to contact any of the counterparties to any Existing Manufacturing Contract or

any Cross-Over Contracts, or contact any of the buyers of Seller's assets that are beneficiaries of such Cross-Over Contracts, and Purchaser is hereby authorized to engage in such contact and discussions.

- (c) Any information provided to or obtained by Purchaser or its representatives, including pursuant to this Section 6.2 is confidential information and subject to the terms of, and the restrictions contained in, the Confidentiality Agreement. Purchaser agrees to be bound by and comply with the provisions set forth in the Confidentiality Agreement as if such provisions were set forth herein, and such provisions are hereby incorporated herein by reference. Effective upon (and only upon) the Closing, the Confidentiality Agreement shall automatically terminate and none of the parties thereto shall have any further Liability or obligation thereunder except with respect to any confidential information provided to or obtained by Purchaser or its representatives concerning the Seller Group, which information shall remain subject to the terms and conditions of the Confidentiality Agreement after the Closing Date. If this Agreement is terminated prior to Closing for any reason, the duration of the confidentiality of the Confidentiality Agreement shall be deemed extended, without any further action by the parties, for a period of time equal to the period of time elapsed between the date such Confidentiality Agreement was initially signed and the date of termination of this Agreement.
- 6.3 Efforts to Consummate. Except as otherwise provided in this Agreement, each of the parties hereto agrees to use its commercially reasonable efforts to cause the Closing to occur as soon as possible after the Agreement Date, including satisfying the conditions precedent set forth in ARTICLE 8 applicable to such party including (a) defending against any Actions, judicial or administrative, challenging this Agreement or the consummation of the Transactions, (b) seeking to have any preliminary injunction, temporary restraining order, stay or other legal restraint or prohibition entered or imposed by any court or other Governmental Authority that is not yet final and non-appealable vacated or reversed, and (c) and executing any additional instruments reasonably requested by another party hereto (without cost or expense to the executing party) necessary to carry out the Transactions and to fully carry out the purposes of this Agreement; provided, however, that, for purposes of "commercially reasonable efforts" standard as required by this Section 6.3, Section 6.4 or Section 6.5, neither the Seller nor its Affiliates or representatives shall be required to offer or grant any accommodation or concession (financial or otherwise) to any third party or to otherwise expend any money or suffer any detriment, to expend any money to remedy any breach of any representation or warranty hereunder, to commence any Action, to waive or surrender any right, to modify any agreement (including any Assigned Contract) or to provide financing to Purchaser for the consummation of the Transactions.
- Notices and Consents. Reasonably promptly following the execution of this Agreement, the Seller will give, or cause to be given, applicable notices to third parties and thereafter will use commercially reasonable efforts (as limited by Section 6.3) to obtain the third-party consents set forth on Schedule 6.4; provided, however, that no representation, warranty, covenant or agreement of the Seller shall be breached or deemed breached, and no condition shall be deemed not satisfied, as a result of (a) the failure to obtain any such third-party consent (unless such consent is part of a closing condition of Seller), (b) any termination of a Contract as a result of the failure to obtain such third-party consent (unless such consent is part of a closing condition of Seller) or (c) any Action commenced or threatened by or on behalf of any Person arising out of or relating to the failure to obtain any such consent or any such termination; provided, further, that nothing in this Section 6.4 shall require the Seller to expend any money or grant any concessions to obtain any such third-party consent (unless Purchaser provides the funds for or reimburses the Seller for such payment).

6.5 **Regulatory Matters.**

(a) Purchaser and the Seller will establish a mutually acceptable and prompt communication and interaction process to ensure the orderly transfer of the Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States. Promptly after Closing, the parties shall file with the FDA, and any other relevant Governmental

Authority all information required in order to transfer the Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, including the information required pursuant to 21 C.F.R. § 314.72, or any successor regulation thereto, any authorization letters or notices, and letters of acceptance. Seller shall file the information required of a former owner, and Purchaser shall file the information required of a new owner, at each party's own expense. Both Purchaser and the Seller also agree to use all commercially reasonable efforts to take any actions required by the Governmental Authority or other government/health agencies to effect the transfer of the Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, and hereby further agree to cooperate with each other in order to effectuate the foregoing transfer of the Lonafarnib IND. The parties agree to use all commercially reasonable efforts to complete the filing of the transfer of the Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States within ten (10) days from the Closing Date. The Seller may retain an archival copy of the Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States, including supplements and records that are required to be kept under 21 C.F.R. § 314.81 or other similar regulation.

- (b) From and after the Closing Date until the Seller is dissolved, the Seller shall cooperate with Purchaser in preparing, disclosing and providing any relevant records, reports, responses or any other documentation that are required to be made, maintained and reported pursuant to the Governmental Authority. The parties agree to use their commercially reasonable efforts to take any other actions required by the FDA or any other Governmental Authority to effect the transaction.
- (c) Until the completion of the transfer of the Lonafarnib IND to Purchaser, the Seller shall take all reasonably necessary or advisable actions to maintain the relevant Lonafarnib IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States.
- extent required by any Applicable Law or Action (including the Bankruptcy Cases), neither Purchaser nor the Seller shall, and Purchaser and the Seller shall cause their respective Affiliates and representatives not to, directly or indirectly, issue any press release or public announcement of any kind without the prior written consent of Purchaser and the Seller; *provided, however*, that the Seller and its Affiliates may make announcements from time to time to their respective employees, customers, suppliers, and other business relations and otherwise as the Seller may reasonably determine is necessary to comply with Applicable Law or the requirements of this Agreement or any other agreement to which any Seller Group Member or any such Affiliate is a party. Purchaser and the Seller shall cooperate in good faith to prepare a joint press release to be issued on the Closing Date, the terms of which shall be mutually agreed upon by the parties.
- 6.7 **Update of Schedules; Knowledge of Breach**. From time to time prior to the Closing, the Seller may supplement or amend the Schedules with respect to any matter first arising after the Agreement Date that would have been required to be set forth or described in such Schedules. Any such supplemental or amended disclosure shall not be deemed to have cured any such breach of representation or warranty for purposes of determining whether or not the conditions set forth in Section 8.2(a) have been satisfied. From and after the Closing, references to the Schedules shall be references to the Schedules as supplemented, modified and/or updated. If, prior to the Closing, Purchaser shall have reason to believe that any breach of a representation or warranty of the Seller has occurred (other than through notice from the Seller), Purchaser shall promptly so notify the Seller, in reasonable detail. Nothing in this Agreement, including this Section 6.7, shall imply that the Seller is making any representation or warranty as of any date other than the Closing Date (other than representations and warranties that are expressly made as of an earlier date).

ARTICLE 7. POST-CLOSING COVENANTS

7.1 Access to Information; Books and Records. From and after the Closing, Purchaser and its Affiliates shall (i) afford the Seller Group and their respective representatives reasonable access, during normal business hours, upon reasonable advance notice and under reasonable circumstances, to the books and records of Purchaser and the Business shall permit the Seller Group and their respective representatives to examine and copy such books and records to the extent reasonably requested by such party and (ii) cause their representatives to furnish all information reasonably requested by any member of the Seller Group or their representatives in connection with financial or regulatory reporting, audit, third party litigation, preparing or filing of any Tax Return or the defense of any Tax claim or assessment or any other business purpose; provided, however, that nothing in this Section 7.1 shall require Purchaser or its Affiliates to furnish to the Seller Group or their respective representatives any material that is subject to an attorneyclient or solicitor-client privilege or an attorney or solicitor work-product privilege or which may not be disclosed pursuant to Applicable Law. For a period of six (6) years following the Closing Date, or such longer period as may be required by Applicable Law or necessitated by applicable statutes of limitations, Purchaser shall, and shall cause its Affiliates to, maintain all such books and records in the jurisdiction in which such books and records were located prior to the Closing Date and shall not destroy, alter or otherwise dispose of any such books and records. On and after the end of such period, Purchaser shall, and shall cause its Affiliates to, provide the Seller with at least ten Business Days prior written notice before destroying, altering or otherwise disposing any such books and records, during which period the Seller may elect to take possession, at its own expense, of such books and records.

7.2 Post-Closing Receipt and Possession of Assets.

- (a) After the Closing Date, the Seller shall transfer promptly to Purchaser from time to time (but in any event on a monthly basis) any payments constituting Transferred Assets received by the Seller. After the Closing Date, Purchaser shall transfer promptly to the Seller, from time to time (but in any event on a monthly basis), any payments constituting Excluded Assets, including any accounts receivable constituting Excluded Assets, received by Purchaser after the Closing.
- (b) In the event that, after the Closing Date, Purchaser receives or otherwise is in possession of any other Excluded Asset, Purchaser shall promptly notify the Seller of its receipt or possession of such other Excluded Asset and transfer, at the Seller's expense, such Excluded Asset to the Seller. In the event that, after the Closing Date, the Seller receives or otherwise is in possession of any other Transferred Asset, the Seller shall promptly notify Purchaser of its receipt or possession of such other Transferred Asset and transfer, at Purchaser's expense (unless the Seller was required to transfer such Transferred Asset to Purchaser at Closing, in which case, and without limitation of any other remedies available to Purchaser, such transfer will be at the Seller's expense), such Transferred Asset to Purchaser.

7.3 Tax Matters.

- (a) All Taxes with respect to the income or operations of the Business or the ownership of the Transferred Assets that relate to any Straddle Period shall be apportioned between Seller and Purchaser as follows: (i) in the case of ad valorem or other property Taxes, on a per diem basis; and (ii) in the case of income, sales and use and withholding Taxes, employment Taxes, or other Taxes based on or measured by income, receipts or profits, as determined from the closing of the books and records of Seller and the Business at the close of business on the Closing Date.
- (b) After the Closing Date, Purchaser and Seller shall furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance (including access to books, records, work papers and Tax Returns for Pre-Closing Tax Periods) relating to the Business or the Transferred Assets as is reasonably necessary for the preparation of any Tax Return, claim for refund or audit, and the prosecution or defense of any claim, suit or proceeding relating to any proposed Tax adjustment. Upon reasonable notice, Seller and Purchaser shall make its employees and facilities available on a mutually convenient basis to provide reasonable explanation of any documents or

information provided hereunder. The other party hereto shall promptly (and in no event later than 30 days after receipt of the request) provide the requested information. The requesting party shall indemnify the other party for any out-of-pocket expenses incurred by such party in connection with providing any information or documentation pursuant to this Section 7.3(b). Any information obtained under this Section 7.3(b) shall be kept confidential, except as otherwise reasonably may be necessary in connection with the filing of Tax Returns or claims for refund or in conducting any Tax audit, dispute or contest.

7.4 Wrong Pockets.

- (a) <u>Assets</u>. If either Purchaser or Seller becomes aware that any of the Transferred Assets has not been transferred to Purchaser or that any of the Excluded Assets has been transferred to Purchaser, it shall promptly notify the other and the parties shall, as soon as reasonably practicable, ensure that such property is transferred, at the expense of Seller and with any necessary prior third party consent or approval, to (i) Purchaser, in the case of any Transferred Asset that was not transferred to Purchaser at the Closing; or (ii) Seller, in the case of any Excluded Asset that was transferred to Purchaser at the Closing.
- (b) <u>Payments</u>. If, on or after the Closing, either party shall receive any payments or other funds due to the other pursuant to the terms of this Agreement or any Related Document, then the party receiving such funds shall, within 30 days after receipt of such funds, forward such funds to the proper party. The parties acknowledge and agree there is no right of offset regarding such payments and a party may not withhold funds received from third parties for the account of the other party in the event there is a dispute regarding any other issue under this Agreement.
- 7.5 **Purchased Intellectual Property and Purchased Product Information**. Promptly following the Closing, at Purchaser's sole cost and expense, Seller shall take such further actions and execute such further documents as may be necessary or reasonably requested by Purchaser to effectuate, evidence and perfect the assignment and transfer of the Owned Intellectual Property Assets and Regulatory Approvals to Purchaser, including making such filings with any Governmental Authorities as may be required to transfer the Owned Intellectual Property Assets and Regulatory Approvals to Purchaser or to further the prosecution, issuance or maintenance of the Owned Intellectual Property Assets and Regulatory Approvals.
- 7.6 **Delivery of Transition Materials; Transition Activities**. The Seller will, as soon as reasonably practicable after the Closing Date, (a) in any event within seven (7) Business Days after the Closing Date, effect the delivery of a complete and true copy of the Zokinvy Dossier as of such date of delivery and all Licensed Product Data, Licensed Product Regulatory Information, and Business Books and Records, and (b) within thirty (30) days after the Closing Date, (i) effect the delivery of all Inventory in accordance with Purchaser's instructions at Purchaser's cost and all other Transition Materials not otherwise delivered to Purchaser, and (ii) use commercially reasonable efforts to perform, and cooperate with Purchaser regarding, the transition activities set forth on Schedule 7.6.

7.7 Licenses.

(a) <u>To Seller</u>. From and after the Closing, subject to the terms and conditions of this Agreement, including Purchaser's retained rights in Section 7.8 related to the Licensed Compound and Licensed Product in the Lonafarnib Antiviral Field, Purchaser hereby grants to Seller, during the period from the Closing until and expiring on completion of the wind-up of Seller, a non-exclusive, sublicensable (solely to a permitted sublicensee under the Merck License Agreement), royalty-free license, under Purchaser's rights to the Transferred Regulatory Information and Transferred Data to Develop, make, have made, use, import, export, Commercialize, sell, offer for sale, and market the Licensed Progeria Product in the Progeria Field in the Territory.

(b) <u>To Purchaser</u>. From and after the Closing, subject to the terms and conditions of this Agreement, including Seller's retained rights in Section 7.8 related to the Licensed Compound and Licensed Product in the Progeria Field, Seller hereby grants to Purchaser, (i) a perpetual, irrevocable, non-exclusive, sublicensable (solely to a permitted sublicensee under the Sublicense Agreement), royalty-free license, under Seller's rights to the Licensed Progeria Product Regulatory Information, General Licensed Product Regulatory Information, Licensed Progeria Product Data, and General Licensed Product Data to Develop, make, have made, use, import, export, Commercialize, sell, offer for sale, and market the Lonafarnib Antiviral Product in the Lonafarnib Antiviral Field in the Territory, and (ii) a perpetual, irrevocable, non-exclusive, sublicensable, royalty-free license under Seller's rights to the Business Books and Records that are not Transferred Business Books and Records to conduct the Business.

7.8 **Retained Rights; Covenants.** From and after the Closing:

- (a) Seller acknowledges and agrees that as between the parties, subject to Section 7.7 and Section 7.9, Purchaser retains any and all other rights under the Transferred Regulatory Information and Transferred Data (i) to the extent necessary to perform any of Purchaser's obligations hereunder, and (ii) that are outside the scope of the license granted to Seller under Section 7.7(a), including, for the avoidance of doubt, the right to Develop, make, have made, use, import, export, Commercialize, sell, offer for sale, and market the Licensed Compound and Licensed Product in the Lonafarnib Antiviral Field in the Territory.
- (b) Seller shall not grant any Third Party any license or right under any Transferred Regulatory Information and Transferred Data, other than as expressly permitted by this Agreement or as required to fulfill its obligations under the Zokinvy Buyer-Eiger Agreement or the Merck License Agreement. Any breach of this Section 7.8 by Seller shall be deemed a material breach of this Agreement.
- (c) Purchaser acknowledges and agrees that as between the parties, subject to Section 7.7 and Section 7.9, Seller retains any and all other rights under the Licensed Progeria Product Regulatory Information, General Licensed Product Regulatory Information, Licensed Progeria Product Data, and General Licensed Product Data (i) to the extent necessary to perform any of Purchaser's obligations hereunder, and (ii) that are outside the scope of the license granted to Purchaser under Section 7.7(b), including, for the avoidance of doubt, the right to Develop, make, have made, use, import, export, Commercialize, sell, offer for sale, and market the Licensed Compound and Licensed Product in the Progeria Field in the Territory.
- (d) Purchaser shall not grant any Third Party any license or right under any Licensed Progeria Product Regulatory Information, General Licensed Product Regulatory Information, Licensed Progeria Product Data, and General Licensed Product Data, other than as expressly permitted by this Agreement or as required to fulfill its obligations under the Zokinvy Buyer Agreement or the Sublicense Agreement.

7.9 **Right of Reference**. From and after the Closing:

(a) The Seller and its Affiliates shall grant, and hereby grant, and shall use reasonable efforts to cause its licensees and sublicensees of the Licensed Progeria Product to grant, to Purchaser and its Affiliates an irrevocable, perpetual, fully paid-up right to reference and access and receive a copy of, and shall provide, and shall use reasonable efforts to cause such licensees and sublicensees to provide, to Purchaser and its Affiliates, (i) the Regulatory Information, Regulatory Application(s), and Regulatory Approval(s) related to any Licensed Product in any field (including the

Lonafarnib Antiviral Field and Progeria Field), and (ii) all data included or referenced in such Regulatory Information, Regulatory Application(s), and Regulatory Approval(s), in each case (i) and (ii), in the possession of, owned by, or under the control (including via license) of Seller or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level), to the extent necessary or reasonably useful for Purchaser to Develop, Manufacture, and obtain and maintain Regulatory Approvals for, Commercialize the Lonafarnib Antiviral Products in the Lonafarnib Antiviral Field in the Territory, and comply with its obligations under Applicable Laws and to Regulatory Authorities and investigators with respect thereto; provided, however, that (A) such right of reference shall be used solely for exercising its license and rights and performing its obligations under the Sublicense Agreement and the Merck Side Letter and (B) all information that is subject to the right of reference shall be treated by Purchaser and its Affiliates, as between the parties, as confidential information of the Seller and its Affiliates under the Confidentiality Agreement.

- (b) Purchaser and its Affiliates shall grant, and hereby grant, and shall use reasonable efforts to cause its licensees and sublicensees of the Lonafarnib Antiviral Products to grant, to the Seller and its Affiliates an irrevocable, perpetual right, fully paid-up right to reference and access and receive a copy of, and shall provide, and shall use reasonable efforts to cause such licensees and sublicensees to provide, to Seller and its Affiliates, (i) the Regulatory Information, Regulatory Application(s), and Regulatory Approval(s) related to any Licensed Product in any field (including the Lonafarnib Antiviral Field and Progeria Field), and (ii) all data included or referenced in such Regulatory Information, Regulatory Application(s), and Regulatory Approval(s), in each case (i) and (ii), in the possession of, owned by, or under the control (including via license) of Purchaser or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of Purchaser or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of Purchaser or any of its Affiliates, directly or indirectly, at any level), to the extent necessary or reasonably useful for Seller to Develop, Manufacture, and obtain and maintain Regulatory Approvals for, Commercialize the Licensed Progeria Products in the Progeria Field in the Territory, and comply with its obligations under Applicable Laws and to Regulatory Authorities and investigators with respect thereto; provided, however, that (A) such right of reference shall be used solely for exercising its license and rights and performing its obligations under the Merck License Agreement, as retained by the Seller and its Affiliates and (B) all information that is subject to the right of reference shall be treated by the Seller and its Affiliates, as between the parties, as confidential information of Purchaser and its Affiliates under the Confidentiality Agreement.
- (c) Within thirty (30) days after the Closing Date, the party and its Affiliates granting a right of reference and other rights under this Section 7.9 will provide, and will cause its applicable licensees and sublicensees (to the extent such party and/or its Affiliates have the right to cause such licensee or sublicensee to do so) to provide, a signed statement to the other party and its Affiliates and applicable licensees or sublicensees that they may rely on, in support of the approval of Regulatory Applications and Regulatory Approval(s) controlled by them, and provide the applicable Regulatory Authority access to (i) such Regulatory Applications and Regulatory Approval(s) and (ii) the underlying data, including raw data, controlled by them included or referenced in such Regulatory Applications and Regulatory Approval(s) (such letter, a "Letter of Authorization"). Each party and its Affiliates will take such actions as may be reasonably requested by the other party and its Affiliates, including providing copies of Regulatory Applications and Regulatory Approval(s) and related data and providing letters of authorization or other documentation, to give effect to the intent of this Section 7.9 and to give the other party and its Affiliates the benefit of such party's and its Affiliates' Regulatory Applications, Regulatory Approval(s), and the underlying data, including raw data, included or referenced therein, as provided herein. The party and its Affiliates granting a right of reference and other rights under this Section 7.9

will bear its own costs and expenses associated with providing, or causing its applicable licensees and sublicensees to provide to, the other party and its Affiliates with such right of reference and other rights.

7.10 **Pharmacovigilance**. From and after the Closing:

- Purchaser with all adverse events ("AEs") for Licensed Products, Lonafarnib HDV Products, and Licensed Progeria Products to the extent not previously provided to Purchaser that are in the possession of, owned by, or under the control (including via license) of Seller or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level). In addition to the foregoing, Seller shall transfer to Purchaser in an agreed upon format, all relevant information (sufficient for Purchaser to comply with its obligations to Regulatory Authorities and investigators) regarding AEs that have been observed during any clinical trials conducted with Licensed Products, Lonafarnib HDV Products, and Licensed Progeria Products prior to the Closing Date that are in the possession of, owned by, or under the control (including via license) of Seller or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level).
- (b) Each party (with respect to Seller, to the extent permitted to do so under the Zokinvy Buyer-Eiger Agreement and the Merck License Agreement, and with respect to Purchaser, to the extent permitted to do so under the Zokinvy Buyer Agreement and the Sublicense Agreement) shall (i) notify the other party of all information coming into its possession concerning AEs associated with commercial or clinical uses, studies, investigations or tests with Licensed Products, Lonafarnib HDV Products, or Licensed Progeria Products, in the Territory, as applicable, involving Licensed Products, Lonafarnib HDV Products, or Licensed Progeria Products, as applicable, and (ii) forward to the other party, completed AE case reports associated with commercial or clinical uses, studies, investigations or tests with Licensed Products, Lonafarnib HDV Products, or Licensed Progeria Products, as applicable, within five (5) Business Days for any death/fatal-life threatening assessed AEs or, within ten (10) Business Days for all other serious AEs, to assure such party remains in compliance with investigator notifications in its respective field. Such AE information should be sent to Seller via email at dapelian@eigerbio.com or sent to Purchaser via email at the email address notified by Purchaser in writing, as applicable. Within thirty (30) days of the Closing, the parties shall enter into a separate written pharmacovigilance agreement with respect to the Licensed Progeria Products and other Licensed Products, as applicable, to enable the parties to fulfill their respective regulatory reporting obligations under Applicable Laws.
- (c) Without limiting any party's rights or obligations in the foregoing, at the request of a party, the other party shall provide the requesting party with all materials, data, information or other documents necessary in form and substance to allow the requesting party to comply with its obligations under Section 5.3 of the Merck License Agreement (in the case of Seller as the requesting party) and under the Sublicense Agreement (in the case of Purchaser as the requesting party).
- (d) Prior to the Lonafarnib IND Transfer Date, Seller will comply with its obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lonafarnib INDs. After the Lonafarnib IND Transfer Date, Purchaser will comply with its obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lonafarnib INDs. The Seller Group shall, at no cost to Purchaser, (i) maintain and administer the Global Safety

Databases itself and through its Third Party service provider under the Global Safety Database Contracts, (ii) provide Purchaser the pharmacovigilance services set forth on Schedule 7.10(d)(1) from the Closing Date until the Lonafarnib IND Transfer Date, and (iii) provide Purchaser the pharmacovigilance services set forth on Schedule 7.10(d)(2) from the Lonafarnib IND Transfer Date until the date that is ninety (90) days after the Closing Date, which services in clauses (ii) and (iii) will be provided by Seller as Purchaser's service provider and under the reasonable direction and supervision of Purchaser in order to assist Purchaser as reasonably necessary to comply with its obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lonafarnib INDs. Notwithstanding the foregoing, in each case of the foregoing clauses (i), (ii), and (iii), the Seller Group will no longer be required to perform such activities following the earlier of (A) the Plan Consummation Date and (B) the date on which ownership or administration of the Global Safety Databases has been fully transferred and transitioned to Purchaser, the Zokinvy Buyer, and/or a Third Party service provider, as mutually agreed by the Purchaser and the Zokinvy Buyer (such earlier date, the "PV Services Stop Date"). Until the PV Services Stop Date, the Seller Group shall not without Purchaser's prior written consent, (y) sell, assign, license, transfer, convey, deliver or otherwise divest its interests in any of the Global Safety Database Contracts to a Third Party, or amend or modify any of the Global Safety Database Contracts, in each case, in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's ability to access, receive, or be provided data from the Global Safety Databases, Purchaser's rights or obligations under this Agreement, or Purchaser's ability to Develop or Commercialize any Lonafarnib Antiviral Products, or (z) undertake any action that would constitute a material breach of, or reduce the Seller Group's rights under, any Global Safety Database Contract.

7.11 **Existing Manufacturing Contracts**. From and after the Closing:

For each Existing Manufacturing Contract, which assignment to Purchaser will become effective as of the applicable Existing Manufacturing Contract Transfer Date, during the period of time beginning on the Closing Date and ending on the Existing Manufacturing Contract Transfer Date for the applicable Existing Manufacturing Contract (the "Existing Manufacturing Contract Interim Term"), to the extent permitted to do so under Applicable Law. including any Order or Final Order, Seller shall retain each such Existing Manufacturing Contract in full force and effect until the applicable Existing Manufacturing Contract Transfer Date, including as necessary to (i) grant and provide the benefit to Purchaser of Seller's rights under such Existing Manufacturing Contracts, and (ii) delegate Seller's obligations under such Existing Manufacturing Contract to Purchaser, in each case (i) and (ii), for Purchaser to fully exercise Purchaser's rights and perform Purchaser's obligations pursuant to this Agreement, the Sublicense Agreement, or the Merck Side Letter, conduct the Business, and use and exploit the Transferred Assets. During the Existing Manufacturing Contract Interim Term, Purchaser hereby agrees to be bound by and comply with, and agrees to cause its Affiliates to be bound by and comply with, all of the terms, conditions, obligations, and any restriction of rights, applicable to a sublicensee of Seller under the Existing Manufacturing Contracts. Seller will use reasonable efforts to not, and to ensure that its Affiliates do not (A) sell, assign, transfer, convey, deliver or otherwise divest its interests in any of the Existing Manufacturing Contracts to a Third Party in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's rights or obligations under this Agreement or Purchaser's ability to Commercialize any Lonafarnib Antiviral Products, (B) amend any of the Existing Manufacturing Contracts in a manner that adversely affects the rights granted to Purchaser under this Agreement or Purchaser's ability to Commercialize any Lonafarnib Antiviral Products, or (C) undertake any action that would constitute a material breach of, and allow the Third Party that is a party to any Existing Manufacturing Contract to terminate, any Existing Manufacturing Contract, in each case, with respect to any Lonafarnib Antiviral Product.

- (b) Seller will provide to Purchaser, within thirty (30) days following the end of a Calendar Quarter, an invoice for each preceding Calendar Quarter, which will include all fees, costs and expenses incurred by Seller in connection with, or other amounts due under, any Existing Manufacturing Contract to the extent such fees, costs, expenses or amounts relate to any Lonafarnib Antiviral Product or any Third Party services provided under such Existing Manufacturing Contracts to the extent related to any Lonafarnib Antiviral Product. Purchaser will pay each invoice no later than thirty (30) days after receipt. If Purchaser fails to pay the full amount of any invoice within such thirty (30) day period, then Seller may, upon reasonable notice to Purchaser, suspend its obligations hereunder to provide any and all services or other benefits under such Existing Manufacturing Contracts until such time as all invoices have been paid in full.
- (c) As of each Existing Manufacturing Contract Date, except with respect to rights exercised by Seller on behalf of, or obligations delegated to, Purchaser pursuant to Section 7.11(a), or payments invoiced to Purchaser under Section 7.11(b) that have not been paid in full, the representations and warranties of Seller under Section 3.8 solely with respect to the Existing Manufacturing Contracts shall be true and correct in all respects as of applicable Existing Manufacturing Contract Date as though made at and as of such time.
- 7.12 **Zokinvy Buyer Agreement**. Following the Closing, Purchaser shall negotiate in good faith with the Zokinvy Buyer a Zokinvy Buyer Agreement which addresses the following matters: (a) the determination and allocation of Cross-Field Sales (as defined in the Merck License Agreement); (b) a safety data exchange agreement for the exchange of safety data relating to the Zokinvy Product and Lonafarnib Antiviral Products and responsibility for maintaining the Global Safety Databases; (c) a grant by Purchaser to the Zokinvy Buyer of a license to the Transferred Regulatory Information and Transferred Data to replace the license granted to Seller under Section 7.7, (d) a license and right of reference to, and right to access and receive copies of, the INDs and NDAs, including all modules thereof, related to the Zokinvy Product and all data related thereto directly from the Zokinvy Buyer, and letters of authorization in furtherance thereof; (e) a co-existence agreement for trademarks containing the word "Eiger"; and (f) supply by Purchaser to the Zokinvy Buyer of the Zokinvy Product under Purchaser's rights under the Existing Manufacturing Contracts after the Existing Manufacturing Contract Transfer Date for such Existing Manufacturing Contract.
- Licensed Product Data, and General Business Books and Records. As of the Closing Date, Seller shall assign, and hereby assigns, its entire right, title, and interest in and to all General Licensed Product Regulatory Information, General Licensed Product Data, and General Business Books and Records to Purchaser and the Zokinvy Buyer as equal joint owners, subject to and conditioned solely upon the agreement of Purchaser and the Zokinvy Buyer with respect to such joint ownership (the "Joint Ownership Agreement"). Such assignment shall be effective automatically and without further notice immediately and solely upon the effectiveness of the Joint Ownership Agreement. Following the Closing, Purchaser shall use good faith efforts to negotiate with the Zokinvy Buyer the Joint Ownership Agreement. Unless otherwise required under Applicable Law, including any Order or Final Order, Seller shall not grant any Third Party any license or right to, or sell, transfer, encumber, or distribute, any of its right, title, or interest in any General Licensed Product Regulatory Information, General Licensed Product Data, or General Business Books and Records without the prior written agreement of both Seller and the Zokinvy Buyer.

Any breach of this Section 7.13 by Seller shall be deemed a material breach of this Agreement and entitle the Purchaser to seek specific performance.

7.14 **Virology Collaborator Confirmation Letters**. Within thirty (30) days after the Closing Date, the Seller and Purchaser shall use commercially reasonable efforts to obtain a Virology Collaborator Confirmation Letter from each Virology Collaborator.

7.15 Cross-Over Contracts.

- (a) From the Agreement Date until the Plan Consummation Date, the Seller Group shall not, and shall cause its Affiliates not, to reject, amend, modify, sell, assign, license, transfer, convey, deliver or otherwise divest its interests in any of the agreements on Schedule 7.15 (the "Cross-Over Contracts") in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's rights or obligations under this Agreement, or Purchaser's ability to Develop or Commercialize any Lonafarnib Antiviral Products.
- (b) Except for those Cross-Over Contracts rejected, transferred, assigned or terminated by the Seller Group without violating Section 7.15(a), the Seller Group shall, upon Purchaser's written request, transfer and assign, and hereby transfers and assigns, automatically and without further notice, to the Purchaser, each other Cross-Over Contract, effective on the date that is the earliest to occur of (a) the date that each and every Cross-Over Contract Benefited Party of such Cross-Over Contract obtains (i) a new agreement with the applicable counterparty of such Cross-Over Contract for substantially the same services as those then being provided to Seller by such counterparty under such Cross-Over Contract, or (ii) an agreement with a Third Party such that such services then being provided under such Cross-Over Contract to such Cross-Over Benefited Party are no longer needed by such Cross-Over Benefited Party, (b) the date Purchaser and all Cross-Over Contract Benefited Parties of such Cross-Over Contract agree to such transfer and assignment of such Cross-Over Contract, and (c) the date all Cross-Over Benefited Parties are no longer receiving any services under such Cross-Over Contract; and upon such transfer and assignment, such Cross-Over Contract shall be deemed an Assigned Contract for all purposes under this Agreement; the Purchaser shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Cross-Over Contract; and (if applicable) Seller shall be responsible for paying all related Seller Cure Amounts.
- Notwithstanding the foregoing Sections 7.15(a) and 7.15(c), (x) the IQVIA Contracts shall be Assigned Contracts upon the occurrence of the Satisfactory IQVIA Cure Resolution, and (y) the Cross-Over Contracts that are not IOVIA Contracts (the "Other Cross-Over Contracts") shall be Assigned Contracts upon the occurrence of the Satisfactory Other Cure Resolution, provided that if the Satisfactory IQVIA Cure Resolution does not occur by the Plan Consummation Date, the IQVIA Contract shall be Excluded Contracts, and if the Satisfactory Other Cure Resolution does not occur by the Plan Consummation Date, the Other Cross-Over Contracts shall be Excluded Contracts. "Satisfactory IQVIA Cure Resolution" means a resolution of the cure objection with respect to the IQVIA Contracts that provides for a Cure Amount of no greater than \$2,000,000 or that is otherwise acceptable to Purchaser in its sole discretion. "Satisfactory Other Cure Resolution" means a resolution of the cure objections with respect to the Other Cross-Over Contracts that provides for a Cure Amount either (i) in the aggregate with the Cure Amount of the Satisfactory IQVIA Cure Resolution, of no greater than \$2,000,000, or (ii) if in excess of (i), that is otherwise acceptable to Purchaser in its sole discretion. In the event that the a Cross-Over Contract becomes an Excluded Contract, Purchaser agrees to use commercially reasonable efforts to preserve the Transferred Data, including the Global Safety Databases, and fully transfer and transition the Transferred Data and Transferred Regulatory Information to Purchaser, and shall not instruct the counterparties to the IQVIA Contracts to delete or remove the Transferred Data from the Global Safety Databases.

- (d) For the avoidance of doubt, after the Closing Date, when any Cross-Over Contract becomes an Assumed Contract, the Seller and Purchaser shall each promptly pay or cause to be paid all Purchaser Cure Amounts and Seller Cure Amounts (if any) for such Assumed Contract.
- 7.16 **Confirmation Letters.** Within seven (7) Business Days after Purchaser's written request provided to Seller after the Closing, Seller and its Affiliates shall provide a letter of confirmation to Purchaser for delivery by Purchaser to any Person that possesses or otherwise holds any Transferred Assets that confirms that Purchaser acquired and is the exclusive owner of the relevant Transferred Assets held by such Person in a form reasonably acceptable to Purchaser. Purchaser will have the right to provide any such letter of confirmation to any Person.

ARTICLE 8. CONDITIONS PRECEDENT

- 8.1 **Conditions to Each Party's Obligation**. The respective obligations of the parties hereto to effect the Transactions are subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller and Purchaser), at or prior to the Closing, of the following conditions:
 - (a) **No Injunctions or Restraints**. No Order or Applicable Law preventing the consummation of the Transactions shall be in effect.
 - (b) **Sale Order**. The Bankruptcy Court shall have entered the Sale Order and such Sale Order shall be a Final Order (unless such Final Order requirement is waived by the Seller and Purchaser in their respective sole discretion).
- 8.2 **Conditions to Obligations of Purchaser**. The obligations of Purchaser to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by Purchaser), at or prior to the Closing, of the following conditions:
 - (a) Representations and Warranties. Each of the representations and warranties of the Seller set forth in ARTICLE 3 shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality", "Material Adverse Effect" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not have, individually or in the aggregate, a Material Adverse Effect.
 - (b) **Performance of Covenants and Obligations**. The Seller shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated expressly by the terms of this Agreement or caused by the Transactions.
- (c) **Effective Assignment of Contracts**. The Bankruptcy Court shall have entered an order (which may be the Sale Order) approving the assumption and assignment to Purchaser of the Assigned Contracts, which order shall be a Final Order and in full force and effect and in a form and substance satisfactory to Purchaser.
- (d) **Closing Deliverables.** The Seller shall have delivered to Purchaser the closing deliveries required to be delivered by the Seller pursuant to Section 2.8(a), Section 2.8(b), Section 2.8(c), Section 2.8(d), Section 2.8(f), and Section 2.8(g).

- 8.3 **Conditions to Obligations of the Seller**. The obligation of the Seller to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller), at or prior to the Closing, of the following conditions:
 - (a) **Representations and Warranties**. Each of the representations and warranties of Purchaser set forth in ARTICLE 4 shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not, individually or in the aggregate, (i) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (ii) otherwise prevent, hinder or delay the consummation of the Transactions.
 - (b) **Performance of Covenants and Obligations of Purchaser**. Purchaser shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated by the terms of this Agreement or caused by the Transactions.
 - (c) Closing Deliverables. Purchaser shall have delivered to the Seller the closing deliveries required to be delivered by Purchaser pursuant to Section 2.8(a), Section 2.8(b), Section 2.8(c), Section 2.8(d), Section 2.8(e), and Section 2.8(f).
- 8.4 **Waiver of Condition; Frustration of Conditions**. All conditions to the Closing shall be deemed to have been satisfied or waived from and after the Closing. Neither Purchaser nor the Seller may rely on the failure of any condition set forth in this ARTICLE 8, as applicable, to be satisfied if such failure was caused by such party's failure to use, as required by this Agreement, its reasonable best efforts to consummate the Transactions.
- 8.5 **Delivery of a Notice of Readiness to Close**. At any time after the Seller's satisfaction of its conditions to Closing in accordance with the terms of Section 8.1 and Section 8.3 of this Agreement, the Seller may deliver a notice to Purchaser (a "**Notice of Readiness to Close**"). Purchaser shall have three (3) Business Days from delivery of a Notice of Readiness to Close to satisfy its conditions to Closing in accordance with the terms of Section 8.1 and Section 8.2 of this Agreement and consummate the Transactions. If Purchaser does not satisfy its conditions to Closing and consummate the Transaction within three (3) Business Days, Purchaser shall forfeit the entire Deposit Escrow Amount to the Seller.

ARTICLE 9. TERMINATION

- 9.1 **Events of Termination**. Notwithstanding anything to the contrary, this Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing:
 - (a) by mutual written consent of Purchaser and the Seller;
 - (b) automatically, upon (i) the consummation of a sale or other disposition of all or substantially all of the Transferred Assets to a Person other than Purchaser (each, an "Alternate Transaction"), (ii) if, at close of the Auction, Purchaser's bid has not been selected as either the winning bid or the Back-Up Bid or (iii) if, at the close of the Auction, Purchaser's bid was selected as the Back-Up Bid, upon the consummation of a Competing Bid or Alternate Transaction;
 - (c) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if the Bankruptcy Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code;
 - (d) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if Purchaser is not selected as having the winning bid or Back-Up Bid at Auction, if any;

- (e) by Purchaser if the Seller (i) withdraws the motion for the Sale Order, or publicly announces its intention to withdraw such motion, (ii) moves to voluntarily dismiss the Bankruptcy Cases, (iii) moves for conversion of the Bankruptcy Cases to Chapter 7 of the Bankruptcy Code, or (iv) moves for appointment of an examiner with expanded powers pursuant to Section 1104 of the Bankruptcy Code or a trustee in the Bankruptcy Cases;
- (f) by Purchaser, by written notice from Purchaser to the Seller, if there has been a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement, such that the conditions in Section 8.1 or Section 8.2 are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by the Seller prior to the earlier of (i) twenty (20) Business Days after receipt of written notice from Purchaser requesting such breach be cured or (ii) the Outside Date; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 9.1(f) shall not be available to Purchaser if the failure of Purchaser to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement;
- (g) by the Seller, by written notice from the Seller to Purchaser, if there has been a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement, such that the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by Purchaser prior to the earlier of (i) 20 Business Days after receipt of written notice from the Seller requesting such breach be cured or (ii) the Outside Date; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 9.1(g) shall not be available to the Seller if the failure of the Seller to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement;
- (h) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if any Governmental Authority of competent jurisdiction shall have issued an Order, enacted any Applicable Law or taken any other action restraining, enjoining or otherwise prohibiting the consummation of the Transactions and, in the case of Orders and other actions, such Order or other action shall have become Final Orders; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 9.1(h) shall not be available to the party seeking to terminate if any action of such party or any failure of such party to act has contributed to such Order or other action and such action or failure constitutes a breach of this Agreement;
- (i) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if the Closing has not occurred on or prior to October 5, 2024 (the "Outside Date"); provided, however, that the party exercising the right to terminate this Agreement pursuant to this Section 9.1(i) shall not have been responsible for such failure of the Closing to occur through a breach or inaccuracy of a covenant, representation or warranty contained in this Agreement (it being understood, acknowledged, and agreed that if Seller is unable to provide any required Closing deliverable of Seller, then Seller shall be deemed to have been responsible for such failure of the Closing for purposes of this Section 9.1(i)); or
- (j) by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a), Section 9.3(b), and Section 9.3(c).

9.2 Effect of Termination.

(a) In the event that this Agreement shall be terminated pursuant to Section 9.1, (a) Purchaser and its representatives shall promptly return all documents, work papers and other materials of

the Seller including any confidential information and (b) all further obligations of the parties hereto under this Agreement shall terminate without further Liability or obligation to the other parties hereto; *provided, however*, that, notwithstanding the foregoing, the Liabilities and obligations under (i) the Confidentiality Agreement, and (ii) Section 2.9(c), Section 6.2(c), this Section 9.2, Section 9.3, and ARTICLE 10 shall continue in full force and effect.

(b) Notwithstanding anything to the contrary in this Agreement, in the event of valid termination of this Agreement pursuant to Section 9.1, (i) Seller's Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement shall be capped at an amount equal to the Deposit Escrow Amount, and Purchaser shall be entitled to all remedies available at law or in equity, including payment of the Termination Fee and Expense Reimbursement pursuant to Section 9.3, and (ii) Purchaser's Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement shall be capped at an amount equal to the Deposit Escrow Amount and Seller shall be entitled to all remedies available at law or in equity, including payment of the Deposit Escrow Amount pursuant to Section 2.9(c).

9.3 Termination Fee and Expense Reimbursement.

- Subject to limitations set forth in the Bid Procedures Order, in consideration of Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof, and the identification and quantification of assets to be included in the Transferred Assets, and to compensate Purchaser as a stalking-horse bidder, the Seller shall pay in cash to Purchaser, by wire transfer of immediately available funds to the account specified by Purchaser to the Seller in writing, an amount equal to the Termination Fee in the event that this Agreement is terminated pursuant to any of Sections 9.1(b)-(f) or 9.1(h)-(i) in which case the Termination Fee shall be due and payable simultaneously with any termination of this Agreement; provided that Purchaser shall not be entitled to the fee described in this Section 9.3(a) to the extent Purchaser is in material breach of this Agreement at the time this Agreement is terminated pursuant to Sections 9.1(b)-(f) or 9.1(h)-(i) if Seller has provided notice of such material breach to Purchaser and such material breach has remained uncured for more than five (5) Business Days after Purchaser's receipt of such notice. The Seller's obligation to pay the Termination Fee pursuant to this Section 9.3(a) shall survive termination of this Agreement and shall constitute an administrative expense of the Seller under section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind, including those specified in section 503(b) or 507(b) of the Bankruptcy Code.
- Subject to limitations set forth in the Bid Procedures Order, in (b) consideration of Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof, and the identification and quantification of assets to be included in the Transferred Assets, if this Agreement is terminated in accordance with the terms set forth in any of Sections 9.1(b)-(f) or 9.1(h)-(i), then the Seller shall pay to Purchaser in cash not later than two (2) Business Days following receipt of documentation supporting the request for reimbursement of costs, fees and expenses, the Expense Reimbursement, in an amount not to exceed \$224,000, by wire transfer of immediately available funds to an account specified by Purchaser to the Seller in writing; provided that Purchaser shall not be entitled to the fee described in this Section 9.3(b) to the extent Purchaser is in material breach of this Agreement at the time this Agreement is terminated pursuant to Sections 9.1(b)-(f) or 9.1(h)-(i) if Seller has provided notice of such material breach to Purchaser and such material breach has remained uncured for more than five (5) Business Days after Purchaser's receipt of such notice. The Seller's obligation to pay the Expense Reimbursement pursuant to this Section 9.3(b) shall survive termination of this Agreement and shall constitute an administrative expense of Seller under section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind, including those specified in section 503(b) or 507(b) of the Bankruptcy Code.

(c) The Seller agrees and acknowledges that Purchaser's due diligence, efforts, negotiation, and execution of this Agreement have involved substantial investment of management time and have required significant commitment of financial, legal, and other resources by Purchaser, and that such due diligence, efforts, negotiation, and execution have provided value to the Seller and, in the Seller's reasonable business judgment, is necessary for the preservation of the value of the Seller's estate. The Seller further agrees and acknowledges that the Termination Fee and the Expense Reimbursement are not a penalty, but rather represent liquidated damages that are reasonable in relation to Purchaser's efforts, Purchaser's lost opportunities from pursuing the Transactions, and the magnitude of the Transactions. The provision of the Termination Fee and the Expense Reimbursement is an integral part of this Agreement, without which Purchaser would not have entered into this Agreement.

ARTICLE 10. GENERAL PROVISIONS

- 10.1 Survival of Representations, Warranties and Covenants. All covenants and agreements contained in this Agreement that by their term are to be performed in whole or in part, or which prohibit actions, subsequent to Closing shall, solely to the extent such covenants and agreements are to be performed, or prohibit actions, subsequent to Closing, survive the Closing in accordance with their terms until fully performed or satisfied. All other covenants and agreements contained herein, and all representations and warranties contained herein or in any certificated deliveries hereunder shall not survive Closing and shall therefor terminate, including any Action for damages in respect of any breach or inaccuracy thereof. Notwithstanding the foregoing, the provisions of Section 2.9(c), Section 6.2, Section 9.2, this ARTICLE 10 and the Confidentiality Agreement shall survive the Closing. For the avoidance of doubt, nothing in this Section 10.1 shall affect the survival of the covenants or representations or warranties of Seller under the Sublicense Agreement or its related agreements.
- Entire Agreement. This Agreement, including the Exhibits and Schedules hereto, the 10.2 Confidentiality Agreement and the Related Documents, contain the entire understanding of the parties hereto with respect to the subject matter contained herein and therein. This Agreement supersedes all prior and contemporaneous agreements, arrangements, contracts, discussions, negotiations, undertakings and understandings (including any letters of intent or term sheets), whether written or oral, among the parties with respect to such subject matter (other than, for the avoidance of doubt, the Confidentiality Agreement and the Related Documents) or any prior course of dealings. The parties hereto have voluntarily agreed to define their rights, Liabilities and obligations respecting the Transactions exclusively in contract pursuant to the express terms and conditions of this Agreement, the Confidentiality Agreement and the Related Documents, and the parties hereto expressly disclaim that they are owed any duties or entitled to any remedies not expressly set forth in this Agreement, the Confidentiality Agreement and the Related Documents. Furthermore, the parties each hereby acknowledge that this Agreement, the Confidentiality Agreement and the Related Documents embody the justifiable expectations of sophisticated parties derived from arm's-length negotiations, and all parties to this Agreement, the Confidentiality Agreement and the Related Documents specifically acknowledge that no party has any special relationship with another party that would justify any expectation beyond that of an ordinary purchaser and an ordinary seller in an arm'slength transaction. The sole and exclusive remedies for any Related Claims shall be those remedies available at law or in equity for breach of contract only (as such contractual remedies have been further limited or excluded pursuant to the express terms of this Agreement); and the parties hereby agree that neither party hereto shall have any remedies or cause of action (whether in contract or in tort or otherwise) of any statements, communications, disclosures, failures to disclose, representations or warranties not set forth in this Agreement.
- 10.3 **Amendment; No Waiver**. This Agreement and the Related Documents may be amended, supplemented or changed, and any provision hereof or thereof can be waived, only by a written instrument making specific reference to this Agreement (and, if applicable, the Related Documents) executed by the party against whom enforcement of any such amendment, supplement, modification or waiver is sought.

The waiver by any party of a breach of any provision of this Agreement or the Related Documents shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any party to exercise, and no delay in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall a single or partial exercise of such right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy.

- 10.4 Severability; Specific Versus General Provisions. Whenever possible, each provision of this Agreement and the Related Documents shall be interpreted in such manner as to be effective and valid under Applicable Law, but if any term or other provision of this Agreement or the Related Documents is invalid, illegal, or incapable of being enforced by any Applicable Law or public policy, all other terms or provisions of this Agreement and the Related Documents shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, in whole or in part, such term or provision is hereby deemed modified to give effect to the original written intent of the parties to the greatest extent consistent with being valid and enforceable under Applicable Law. No party hereto shall assert, and each party shall cause its respective Affiliates or related parties not to assert, that this Agreement or any part hereof is invalid, illegal or unenforceable. Notwithstanding anything to the contrary, to the extent that a representation, warranty, covenant or agreement of the Seller contained in this Agreement or the Schedules (each, a "Provision") addresses a particular issue with specificity (a "Specific Provision"), and no breach by the Seller exists under such Specific Provision, the Seller shall not be deemed to be in breach of any other Provision (with respect to such issue) that addresses such issue with less specificity than the Specific Provision, and if such Specific Provision is qualified or limited by the Seller's Knowledge, or in any other manner, no other Provision shall supersede or limit such qualification in any manner.
- 10.5 **Expenses and Obligations**. Except as otherwise provided in this Agreement, all costs and expenses incurred by the parties hereto in connection with the Transactions, including the costs, expenses and disbursements of counsel and accountants, shall be borne solely and entirely by the party that has incurred such expenses; *provided*, *however*, that Purchaser shall pay, or promptly reimburse the Seller for, any filing fees which relate to any required governmental filing or notification and Purchaser shall pay any Transfer Taxes.
- 10.6 **Notices**. All notices, consents, waivers, and other communications under this Agreement or the Related Documents shall be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery, (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service, (c) if delivered by electronic mail (unless the sender receives an automated message that the email has not been delivered) on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day) and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or email addresses set forth below (or to such other addresses as a party may designate by notice to the other parties in accordance with this Section 10.6):

If to Purchaser:

Eiger InnoTherapeutics, Inc. 2061 Webster Street Palo Alto, CA 94301 Attn: Dr. Jeffrey Glenn Email: jsglenn@stanford.edu

with a copy to (which will not constitute notice):
Goodwin Procter LLP

The New York Times Building 620 Eighth Avenue
New York, New York 10018
Attn: Kizzy Jarashow, Maggie Wong, and David Chen
email: kjarashow@goodwinlaw.com; mwong@goodwinlaw.com;

dchen@goodwinlaw.com

If to the Seller:

Eiger BioPharmaceuticals, Inc. 2100 Ross Avenue Dallas, Texas 75201 Attn: David Apelian, Chief Executive Officer

Email: dapelian@eigerbio.com

with a copy to (which will not constitute notice):

Sidley Austin LLP 2021 McKinney Ave., Suite 2000 Dallas, TX 75201

Attention: Thomas R. Califano, William E. Curtin and Anne G. Wallice

Email: tom.califano@sidley.com, wcurtin@sidley.com, and anne.wallice@sidley.com

- 10.7 **Counterparts**. This Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by electronic transmission in .PDF format, or other agreed format shall be sufficient to bind the parties to the terms and conditions of this Agreement. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement or any Related Document, shall be disregarded in determining the party's intent or the effectiveness of such signature.
- 10.8 **Governing Law.** This Agreement, the Related Documents and all Related Claims shall be governed by the internal laws of the State of Delaware (including its statute of limitations), without giving effect to any choice or conflict of law principles or rules that would cause the application of the Applicable Laws of any other jurisdiction.

10.9 Submission to Jurisdiction; Consent to Service of Process.

Without limiting any party's right to appeal any Order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to interpret and/or enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any Related Document, any breach or default hereunder or thereunder, or the Transactions, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 10.6; provided, however, that if the Bankruptcy Cases have closed, the parties agree to irrevocably submit to the exclusive jurisdiction of the United States District Court for the Northern District of Texas over all Related Claims, and each party hereto hereby irrevocably agrees that all Related Claims may be heard and determined in such courts. The parties hereto hereby irrevocably and unconditionally waive, to the fullest extent permitted by Applicable Law, any objection which they may now or hereafter have to the laying of venue of any such Related Claim brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Applicable Law.

- (b) Each of the parties hereto hereby consents to process being served by any party to this Agreement in any Related Claim by the delivery of a copy thereof in accordance with the provisions of Section 10.6 (other than by email) along with a notification that service of process is being served in conformance with this Section 10.9(b). Nothing in this Agreement will affect the right of any party to serve process in any other manner permitted by Applicable Law.
- 10.10 Waiver of Jury Trial. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY EXPRESSLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY PROCEEDING OR RELATED CLAIM BROUGHT BY OR AGAINST IT, DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS.
- 10.11 **Rights Cumulative**. All rights and remedies of each of the parties under this Agreement and the Related Documents will be cumulative, and the exercise of one or more rights or remedies will not preclude the exercise of any other right or remedy available under this Agreement, the Related Documents or Applicable Law.
- 10.12 **Assignment**. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors by operation of law and permitted assigns of the parties hereto. No assignment of this Agreement or any of the rights, interests or obligations under this Agreement may be made by any party hereto at any time, whether or not by operation of law, without the prior written consent of the Seller and Purchaser, and any attempted assignment without the required consent shall be void; *provided, however*, that (a) Purchaser may assign (i) any of its rights or delegate any of its duties under this Agreement to any of its Affiliates, and (ii) its rights, but not its duties, under this Agreement to any of its financing sources and (b), the Seller may assign any of its rights or delegate any of its duties under this Agreement (i) to any of its Affiliates, (ii) to any creditor or group of creditors pursuant to an order of the Bankruptcy Court entered in the Bankruptcy Cases, including Seller's rights to payment hereunder and rights and ability to enforce the terms of this Agreement and (iii) for collateral security purposes to any lender of the Seller or its Affiliates; *provided, further, however*, that, in each case, such assignment shall not release Purchaser from its obligations under this Agreement and the Seller shall have no obligation to pursue remedies against any assignee of Purchaser before proceeding against Purchaser for any breach of Purchaser's obligations hereunder.
- 10.13 **Specific Enforcement; Remedies.** The parties hereto agree that irreparable damage (for which monetary relief, even if available, would not be an adequate remedy) would occur in the event that any of the provisions of this Agreement were not performed by the parties hereto in accordance with their specific terms or were otherwise breached. It is accordingly agreed that (i) Purchaser, on the one hand, and the Seller, on the other hand, shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of competent jurisdiction without proof of damages or otherwise and that this shall include the right of the Seller to cause Purchaser to fully perform the terms of this Agreement to the fullest extent permissible pursuant to this Agreement and Applicable Laws and to thereafter cause this Agreement and the Transactions to be consummated on the terms and subject to the conditions thereto set forth in this Agreement, and (ii) the right of specific performance and other equitable relief is an integral part of the Transactions and without that right, neither the Seller nor Purchaser would have entered into this Agreement. Remedies shall be cumulative and not exclusive and shall be in addition to any other remedies which any party may have under this Agreement. Each of the parties hereto hereby (A) waives any defenses in any action for specific performance, including the defense that a remedy at law would be adequate, (B) waives any requirement under any Applicable Law to post a bond or other security as a prerequisite to obtaining equitable relief and (C) agrees not to assert that a remedy of specific performance or other equitable relief is unenforceable, invalid, contrary to law or

inequitable for any reason, and not to assert that a remedy of monetary damages would provide an adequate remedy or that the parties otherwise have an adequate remedy at law. Notwithstanding anything to the contrary, in no event shall this Section 10.13 be used, alone or together with any other provision of this Agreement, to require the Seller to remedy any breach of any representation or warranty of the Seller.

- 10.14 **Third-Party Beneficiaries**. Except as set forth in ARTICLE 2 (with respect to the Seller), Section 10.15 (with respect to the Nonparty Affiliates), Section 10.16 (with respect to the released parties identified therein), Section 10.17 (with respect to the Sellers' Group Members) and the next sentence, nothing in this Agreement, express or implied, is intended to confer upon any Person other than the parties hereto any rights or remedies of any nature whatsoever under or by reason of this Agreement. From and after the Closing, all of the Persons identified as third-party beneficiaries in the first sentence of this Section 10.14 shall be entitled to enforce such provisions and to avail themselves of the benefits of any remedy for any breach of such provisions, all to the same extent as if such Persons were parties to this Agreement. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with this Agreement without notice or Liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any party hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the Agreement Date or as of any other date.
- 10.15 No Personal Liability of Directors, Officers and Owners. All Related Claims may be made only against (and are those solely of) the entities that are expressly identified as parties in the preamble to this Agreement (the "Contracting Parties"). No Person who is not a Contracting Party, including any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any Contracting Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any of the foregoing (collectively, "Nonparty Affiliates"), shall have any Liability pursuant to any Related Claim; and, to the maximum extent permitted by Applicable Law, each Contracting Party hereby waives and releases all such Liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Applicable Law, (a) each Contracting Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at Applicable Law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose Liability of a Contracting Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to this Agreement or the Related Documents.

10.16 General Release.

(a) Effective as of the Closing, the Seller, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "Seller Releasing Party"), hereby fully, irrevocably and unconditionally releases and forever discharges Purchaser and its respective past and present directors, managers, officers, employees, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, any and all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Seller Releasing Party

has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Seller Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.

(b) Effective as of the Closing, Purchaser, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "Purchaser Releasing Party"), hereby fully, irrevocably and unconditionally releases and forever discharges the Seller, the Seller's Affiliates and its and their respective past and present directors, managers, officers, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Purchaser Releasing Party has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Purchaser Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.

10.17 **Legal Representation**. Purchaser and the Seller acknowledge and agree that the Law Firm has represented the Seller Group in connection with the negotiation, preparation, execution, delivery and performance of this Agreement and the Related Documents and the consummation of the Transactions, and that the Seller, its Affiliates and its partners, officers, directors and representatives (the "Seller Group Members") have a reasonable expectation that the Law Firm will represent them in connection with any Action involving any Seller Group Member, on the one hand, and Purchaser or any of its Affiliates and representatives (the "Purchaser Group Members"), on the other hand, arising under this Agreement, the Related Documents or the Transactions. Purchaser hereby, on behalf of itself and the other Purchaser Group Members, irrevocably: (a) acknowledges and agrees that any attorney-client privilege, solicitor-client privilege, work product or other attorney-client or solicitor-client confidential information ("Attorney-Client Information") arising from communications prior to the Closing between the Seller (including any one or more officers, directors or stockholders of such Seller), on the one hand, and the Law Firm, on the other hand, is not included in the property, rights, privileges, powers, franchises and other interests that are possessed by or vested in the Business or the Transferred Assets, that any such Attorney-Client Information shall be deemed property of, and controlled solely by, such Seller for the benefit and on behalf of the Seller Group Members and, upon request, convey and transfer any Attorney-Client Information to the Seller; (b) acknowledge and agree that the Seller Group Members shall have the right to retain, or cause the Law Firm to retain, any such documentation or information in the possession of the Law Firm or such Seller Group Members at the Closing; (c) agree not to access, retain or use any documentation or information constituting Attorney-Client Information and that no Purchaser Group Member shall have any right to waive any attorney-client privilege or other right to confidentiality with respect to such Attorney-Client Information; (d) disclaim the right to assert a waiver by any Seller Group Member with regard to the attorney-client privilege, solicitor-client privilege or other right to confidentiality with respect to such Attorney-Client Information solely due to the fact that such documentation or information is physically in the possession of Purchaser after the Closing; (e) consent to the Law Firm's representation after the Closing of any Seller Group Member in any Action that may relate to a Purchaser Group Member or the Transactions and consent to and waive any conflict of interest arising therefrom without the need for any future waiver or consent; and (f) consent to the disclosure by the Law Firm to any Seller Group Member of any documentation or information obtained by the Law Firm during the course of its representation of Seller or any Affiliate prior to the Closing, whether related to this Agreement, the Related Documents, the Transactions or otherwise, whether or not such disclosure is made prior to or after the Closing and whether

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 140 of 205

or not the documentation or information disclosed is subject to any attorney-client privilege, solicitor-client privilege or confidentiality obligation to any Seller Group Member, any Affiliate of the Seller or any other Person. In the event that any Action arises after the Closing between any Purchaser Group Member and a Person other than a Seller Group Member, such Purchaser Group Member shall not disclose any documentation or information that is subject to an attorney-client privilege or other rights of confidentiality referenced in this Section 10.17 without the prior written consent of the applicable Seller; *provided, however*, that if such Purchaser Group Member is required by judicial order or other legal process to make such disclosure, such Purchaser Group Member shall promptly notify the applicable Seller in writing of such requirement (without making disclosure) and shall provide such Seller with such cooperation and assistance as shall be necessary to enable such Seller to prevent disclosure by reason of such attorney-client privilege, solicitor-client privilege or other rights of confidentiality. This Section 10.17 is for the benefit of the Seller Group Members and such Persons are intended third-party beneficiaries of this Section 10.17.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 143 of 205

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

PURCHASER:

EIGER INNOTHERAPEUTICS, INC.

DocuSigned by:

Name: Dr. Jeffrey Glenn

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

SELLER:

EIGER BIOPHARMACEUTICALS, INC.

DocuSigned by:

Name: David Apelian

Title: Chief Executive Officer

Exhibit 2

Lambda APA

Execution Version

LAMBDA ASSET PURCHASE AGREEMENT

by and between

EIGER INNOTHERAPEUTICS, INC., as Purchaser,

and

EIGER BIOPHARMACEUTICALS, INC., as Seller

Dated as of August 1, 2024

TABLE OF CONTENTS

| ARTICLE 1. | DEFINED TERMS | 1 |
|------------|--|----|
| 1.1 | Defined Terms | |
| 1.2 | Other Definitional and Interpretive Matters | 12 |
| ARTICLE 2. | THE PURCHASE AND SALE; CLOSING | 14 |
| 2.1 | Purchase and Sale | 14 |
| 2.2 | Excluded Assets | 17 |
| 2.3 | Assumption of Liabilities | 18 |
| 2.4 | Excluded Liabilities | 18 |
| 2.5 | Excluded Contracts | |
| 2.6 | Nontransferable Assets and Liabilities | 19 |
| 2.7 | Closing | 19 |
| 2.8 | Closing Deliveries of the Parties | |
| 2.9 | Purchase Price; Assumed Liabilities; Deposits | |
| 2.10 | Transfer Taxes | |
| 2.11 | Allocation of Purchase Price | 21 |
| 2.12 | Escrow Accounts | 21 |
| 2.13 | Tax Withholding | |
| ARTICLE 3. | REPRESENTATIONS AND WARRANTIES OF THE SELLER | |
| 3.1 | Organization, Good Standing and Other Matters | |
| 3.2 | Authority and Enforceability | |
| 3.3 | No Conflict; Required Filings and Consents | |
| 3.4 | Compliance With Laws | |
| 3.5 | Permits | |
| 3.6 | Litigation | |
| 3.7 | Real Property | |
| 3.8 | Assigned Contracts | |
| 3.9 | Financial Statements | |
| 3.10 | Absence of Material Developments | |
| 3.11 | Customers and Suppliers | |
| 3.12 | Intellectual Property | |
| 3.13 | Taxes | |
| 3.14 | Product Liability | |
| 3.15 | Product Warranties; Product Return | |
| 3.16 | Brokers and Finders | |
| 3.17 | Inventory | 26 |
| 3.18 | No Other Representations or Warranties | |
| ARTICLE 4. | REPRESENTATIONS AND WARRANTIES OF PURCHASER | |
| 4.1 | Organization, Good Standing and Other Matters | 26 |
| 4.2 | Authority and Enforceability | |
| 4.3 | No Conflict: Required Filings and Consents | |
| 4.4 | Financing | |
| 4.5 | Solvency | |
| 4.6 | Litigation | |
| 4.7 | Brokers and Finders | |
| 4.8 | Non-Reliance of Purchaser; No Other Representations and Warranties | |
| 4.9 | No Other Representations or Warranties | |
| ARTICLE 5. | BANKRUPTCY COURT MATTERS | |
| 5.1 | Competing Transaction | |
| 5.2 | Bankruptcy Court Filings | |
| 5.3 | Assumption of Assigned Contracts | |
| | | |

| 5.4 | Disputed Contracts | 31 |
|-------------|---|----|
| 5.5 | Previously Unknown and Previously Excluded Contracts | |
| ARTICLE 6. | PRE-CLOSING COVENANTS | 32 |
| 6.1 | Conduct of Business | 32 |
| 6.2 | Access to Information; Confidentiality | 34 |
| 6.3 | Efforts to Consummate | 35 |
| 6.4 | Notices and Consents | |
| 6.5 | Regulatory Matters | 35 |
| 6.6 | Public Announcements | |
| 6.7 | Update of Schedules; Knowledge of Breach | 36 |
| ARTICLE 7. | POST-CLOSING COVENANTS | 36 |
| 7.1 | Access to Information; Books and Records | 36 |
| 7.2 | Post-Closing Receipt and Possession of Assets | 37 |
| 7.3 | Tax Matters | 37 |
| 7.4 | Wrong Pockets | 38 |
| 7.5 | Purchased Intellectual Property and Purchased Product Information | |
| 7.6 | Delivery of Transferred Materials; Transition Activities | 38 |
| 7.7 | License | 38 |
| ARTICLE 8. | CONDITIONS PRECEDENT | 40 |
| 8.1 | Conditions to Each Party's Obligation | 40 |
| 8.2 | Conditions to Obligations of Purchaser | 40 |
| 8.3 | Conditions to Obligations of the Seller | 41 |
| 8.4 | Waiver of Condition; Frustration of Conditions | 41 |
| 8.5 | Delivery of a Notice of Readiness to Close | 41 |
| ARTICLE 9. | TERMINATION | |
| 9.1 | Events of Termination | 42 |
| 9.2 | Effect of Termination | 43 |
| 9.3 | Expense Reimbursement | |
| ARTICLE 10. | GENERAL PROVISIONS | |
| 10.1 | Survival of Representations, Warranties and Covenants | |
| 10.2 | Entire Agreement | |
| 10.3 | Amendment; No Waiver | |
| 10.4 | Severability; Specific Versus General Provisions | |
| 10.5 | Expenses and Obligations | |
| 10.6 | Notices | |
| 10.7 | Counterparts | |
| 10.8 | Governing Law | |
| 10.9 | Submission to Jurisdiction; Consent to Service of Process | |
| 10.10 | Waiver of Jury Trial | |
| 10.11 | Rights Cumulative | |
| 10.12 | Assignment | |
| 10.13 | Specific Enforcement; Remedies | |
| 10.14 | Third-Party Beneficiaries | |
| 10.15 | No Personal Liability of Directors, Officers and Owners | |
| 10.16 | General Release | |
| 10.17 | Legal Representation | 49 |

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 159 of 205

EXHIBITS

Exhibit A Form of Bill of Sale and Assignment and Assumption Agreement

Exhibit B Escrow Agreement

Exhibit C Form of Intellectual Property Assignment Agreement

SCHEDULES

Schedule 1.1(a) Permitted Liens Schedule 2.1(a) Assigned Contracts

Schedule 2.1(c) Transferred Regulatory Information Schedule 2.1(h) Raw Materials and Inventory Schedule 2.11(a) Preliminary Allocation Schedule

Schedule 3.3 Seller Conflict; Required Filings and Consents

Schedule 3.10 Absence of Material Developments

Schedule 3.11 Customers and Suppliers

Schedule 3.12(a) Intellectual Property Registrations
Schedule 3.12(b) Intellectual Property Agreements

Schedule 3.14 Product Liability

Schedule 3.17 Inventory Locations and Storage Contracts

Schedule 4.3 Purchaser Conflict; Required Filings and Consents

Schedule 6.1 Conduct of Business Schedule 6.4 Notices and Consents Schedule 7.6 Transition Activities

Schedule 7.8(d)(1) Pharmacovigilance Services Prior to Lambda IND Transfer Date Schedule 7.8(d)(2) Pharmacovigilance Services After Lambda IND Transfer Date

LAMBDA ASSET PURCHASE AGREEMENT

THIS LAMBDA ASSET PURCHASE AGREEMENT (this "Agreement"), dated as of August 1, 2024 (the "Agreement Date") is entered into by and between Eiger InnoTherapeutics, Inc., a Delaware corporation ("Purchaser") and Eiger BioPharmaceuticals, Inc., a Delaware corporation (the "Seller").

RECITALS

WHEREAS, on April 1, 2024 (the "Petition Date") the Seller and certain of its Affiliates filed voluntary petitions for relief under chapter 11 of Title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the Northern District of Texas (the "Bankruptcy Court"), thereby commencing Chapter 11 cases (collectively, the "Bankruptcy Cases");

WHEREAS, the Seller is a debtor-in-possession under the Bankruptcy Code and manages its properties and assets pursuant to Sections 1107(a) and 1108 of the Bankruptcy Code;

WHEREAS, the Seller is engaged in the Business and owns, directly or indirectly, all of the Transferred Assets;

WHEREAS, the Seller desires to sell (or cause to be sold) to Purchaser, and Purchaser desires to purchase from the Seller, all of the Transferred Assets Free and Clear, and the Seller desires Purchaser to assume, and Purchaser desires to assume from the Seller, all of the Assumed Liabilities, in each case upon the terms and subject to the conditions hereof, pursuant to a Sale Order and Sections 105(a), 363 and 365 of the Bankruptcy Code and Rules 6004 and 6006 of the Federal Rules of Bankruptcy Procedure;

WHEREAS, the transactions contemplated by this Agreement and the Related Documents are subject to approval by the Bankruptcy Court and will only be consummated pursuant, among other things, to the Sale Order to be entered in the Bankruptcy Cases; and

WHEREAS, concurrently with the execution of this Agreement, Purchaser shall deposit (or cause to be deposited) an aggregate amount equal to the Deposit Escrow Amount into an escrow account (the "Deposit Escrow Account") to be established and maintained by Escrow Agent pursuant to the Escrow Agreement.

NOW, **THEREFORE**, in consideration of the premises and the mutual representations, warranties, covenants, agreements and conditions set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE 1. DEFINED TERMS

- 1.1 **Defined Terms**. The following terms shall have the following meanings in this Agreement:
- "Action" means any action, proceeding, arbitration or litigation (whether civil, criminal or administrative) commenced, brought, conducted or heard by or before any Governmental Authority or arbitrator.
 - "AEs" has the meaning set forth in Section 7.8(a).
- "Affiliate" of any particular Person means any other Person directly or indirectly controlling, controlled by or under common control with such particular Person. For purposes of this Agreement, the direct or indirect ownership of fifty percent (50%) or more of the outstanding voting securities of a Person, or the right to receive fifty percent (50%) or more of the profits or earnings of a Person shall be deemed to constitute control. Such other relationship as in fact results in actual control over the management, business and affairs of a Person shall also be deemed to constitute control.
 - "Agreement" has the meaning set forth in the preamble.

- "Agreement Date" has the meaning set forth in the preamble.
- "Allocation Schedule" has the meaning set forth in Section 2.11(a).
- "Alternate Transaction" has the meaning set forth in Section 9.1(b).
- "Applicable Law" means, with respect to any Person, any federal, provincial, state, local law, ordinance, principle of common law, code, regulation or statute applicable to such Person or such Person's subsidiaries or to any of their respective securities, assets, properties or businesses.
- "Asset Taxes" means any Taxes with respect to the ownership or operation of the Transferred Assets other than (a) Taxes based on net or gross income, and (b) Transfer Taxes.
 - "Assigned Contracts" has the meaning set forth in Section 2.1(a).
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - "Assumption Notice" has the meaning set forth in Section 5.3(a).
 - "Attorney-Client Information" has the meaning set forth in Section 10.17.
 - "Auction" has the meaning set forth in Section 5.2(h).
 - "Avexitide Buyer" means Amylyx Pharmaceuticals, Inc.
- "Avoidance Actions" means any and all avoidance, recovery, subordination, or other claims, actions, rights, or remedies that may be brought by or on behalf of the Seller or its estate or other authorized parties in interest under the Bankruptcy Code or applicable non-bankruptcy law, including, but not limited to, actions or remedies under sections 510, 542, 543, 544, 545, and 547 through and including 553 of the Bankruptcy Code.
- "Back-Up Bid" means the second highest or otherwise best bid if the successful bidder fails to consummate its bid in accordance with the Bid Procedures.
- "Back-up Termination Date" means the first to occur of (a) thirty (30) days after the entry of the Sale Order, (b) consummation of the Transactions with the winning bidder at the Auction, and (c) October 1, 2024.
 - "Bankruptcy Cases" has the meaning set forth in the Recitals.
 - "Bankruptcy Code" has the meaning set forth in the Recitals.
 - "Bankruptcy Court" has the meaning set forth in the Recitals.
 - "Base Price" means \$1,000,000.
- "Bid Procedures" means those certain bidding procedures for the sale of the Seller's assets approved by the Bankruptcy Court as filed at Docket No. 119.
- "Bid Procedures Order" means that certain Order entered by the Bankruptcy Court at Docket No. 94 approving the Bid Procedures.
- "Bill of Sale and Assignment and Assumption Agreement" means the bill of sale and assignment and assumption agreement, dated as of the Closing Date, by and between the Seller and Purchaser, in substantially the form attached hereto as Exhibit A and acceptable to Purchaser.
 - "BMS" means Bristol-Myers Squibb Company.
- "BMS License Agreement" means that certain License Agreement, dated April 20, 2016, between the Seller and BMS.

- "Business" means the business as presently conducted of the Seller Group related to the Development, Manufacture, and Commercialization of Licensed Compounds and Licensed Products in the Field in the Territory.
- "Business Books and Records" means the records and files relating to the Business in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter, including without limitation (i) supplier and vendor lists, (ii) promotional materials, and (iii) other business records required to be transferred to Purchaser under Applicable Law. For clarity, Business Books and Records shall exclude Regulatory Information and Data.
- "Business Day" means any day other than (a) a Saturday, Sunday or federal holiday or (b) a day on which commercial banks in San Francisco, California are authorized or required to be closed.
- "Business Intellectual Property" means all Owned Intellectual Property Assets together with all other Intellectual Property used in, held for use in, or necessary for the conduct of the Business.
 - "Closing" has the meaning set forth in Section 2.7.
 - "Closing Date" has the meaning set forth in Section 2.7.
 - "Code" means the Internal Revenue Code of 1986, as amended, or any successor law.
 - "Commercialization" has the meaning given to it in the BMS License Agreement.
 - "Competing Bid" has the meaning set forth in Section 5.1.
- "Confidentiality Agreement" means that certain Confidentiality Agreement, dated as of April 4, 2024, by and between the Seller and Purchaser.
 - "Consent" means any consent, approval, authorization, waiver or license.
- "Contract" means any written agreement, mortgage, indenture, lease (whether for real or personal property), contract or subcontract.
 - "Contracts List" has the meaning set forth in Section 2.1(a).
 - "Contracting Parties" has the meaning set forth in Section 10.15.
- "Cure Amounts" means any and all costs, expenses or actions with respect to defaults existing as of the Petition Date that Purchaser or the Seller, as applicable, are required to pay or perform to assume any of the Assigned Contracts pursuant to section 365(b)(1)(A) and (B) of the Bankruptcy Code or as otherwise agreed between Purchaser or the Seller, as applicable, and the counterparty to an Assigned Contract.
- "Data" means (a) any and all clinical, preclinical, non-clinical, toxicology, chemistry, biology, animal, CMC, safety, and other data, databases, information, batch records, laboratory records, and all other data and information, and (b) any and all global and country safety databases, in each case (a) and (b), that relate to any Licensed Compound or Licensed Product, any other Transferred Asset, or any Assumed Liability that is in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter.
 - "Deposit Escrow Account" has the meaning set forth in the Recitals.
 - "Deposit Escrow Amount" means \$50,000.

- "Designation Deadline" has the meaning set forth in Section 5.3(b).
- "Determined Cure Amounts" means all Cure Amounts for Assigned Contracts, as determined by a final order of the Bankruptcy Court.
 - "Development" or "Develop" has the meaning given to it in the BMS License Agreement.
 - "Disputed Contract" has the meaning set forth in Section 5.4.
 - "Disputed Contract Order" has the meaning set forth in Section 5.4.
 - "EMA" means the European Medicines Agency, or any successor agency thereto.
- "Enforceability Exceptions" means applicable bankruptcy, insolvency, reorganization, moratorium, receivership and similar Applicable Laws affecting the enforcement of creditors' rights generally and general equitable principles.
- "Environmental Laws" means any Applicable Law relating to pollution or protection of the environment or worker health and safety (in respect of exposure to Hazardous Substances), including such Applicable Laws relating to the use, treatment, storage, disposal, Release or transportation of Hazardous Substances.
 - "Escrow Agent" means Kurtzman Carson Consultants LLC.
- "Escrow Agreement" means the escrow agreement by and between the Seller and the Escrow Agent attached hereto as Exhibit B.
 - "Excluded Assets" has the meaning set forth in Section 2.2.
- "Excluded Books and Records" means the following originals and copies of those books and records, documents, data and information (in whatever form maintained) of the Seller Group and the Business: (i) all corporate minute books (and other similar corporate records) and stock records of the Seller Group, (ii) any books and records relating to the Excluded Assets or (iii) any books, records or other materials that any member of the Seller Group (x) is required by Applicable Law to retain (copies of which, to the extent permitted by Applicable Law, will be made available to Purchaser upon Purchaser's reasonable request), (y) reasonably believes is necessary to enable it to prepare and/or file Tax Returns (copies of which will be made available to Purchaser upon Purchaser's reasonable request) or (z) are prohibited by Applicable Law from delivering to Purchaser.
 - "Excluded Contracts" has the meaning set forth in Section 2.5.
 - "Excluded Liabilities" has the meaning set forth in Section 2.4.
- "Existing Manufacturing Contract" means any Assigned Contract under which the Seller or any of its Affiliates Manufactured or has Manufactured any Licensed Compound or Licensed Product, as identified on Schedule 2.1(a).
- "Expense Reimbursement" means the reimbursement by the Seller of Purchaser's actual and reasonable out-of-pocket legal, accounting, and other third-party advisory or service costs and expenses incurred in connection with the Transactions, as evidenced by invoice(s) provided to the Seller, on the terms and subject to the conditions of Section 9.3.
 - "FDA" means the United States Food and Drug Administration.
- "FD&C Act" means the United States Federal Food, Drug and Cosmetic Act, as amended, and any rules, regulations, and requirements promulgated thereunder.
 - "Field" has the meaning given to it in the BMS License Agreement.
- "Final Order" means an Order, judgment or other decree of the Bankruptcy Court or any other Governmental Authority of competent jurisdiction that has not been reversed, vacated, modified or

amended, is not stayed and remains in full force and effect; provided that such Order shall be considered a Final Order only after the time period for third parties seeking appeal has expired without the filing of any appeal or motion for reconsideration.

"Free and Clear" means free and clear of all Liens and Excluded Liabilities (other than the Permitted Liens and the Assumed Liabilities) to the maximum extent permitted by Section 363(f) of the Bankruptcy Code.

"GAAP" means generally accepted accounting principles in the United States as of the Agreement Date.

"Global Safety Databases" means the databases established and owned or controlled (including via license) by Seller or any of its Affiliates, including such databases that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level) that contain the totality of all current and historic safety data and information with respect to the Licensed Product, including AEs, received, collected, used, held for use by or on behalf of Seller or its Affiliates (including by or on behalf of any contractors or other service providers acting on its or their behalf, directly or indirectly, at any level), or pursuant to the BMS License Agreement for drug surveillance, pharmacovigilance, and regulatory safety reporting purposes, including the global safety database that is the central repository of all such safety data and information worldwide and any and all local or territory databases of such safety data and information with respect to a particular country, region, jurisdiction, or territory.

"Global Safety Database Contracts" means any and all Contracts by and between Seller or any of its Affiliates and a Third Party service provider under which any part of the Global Safety Databases is stored or administered, including the Contracts identified as Global Safety Database Contracts on Schedule 7.9.

"Goods" has the meaning set forth in Section 3.14.

"Governmental Authority" means any multi-national, national, federal, state, local, municipal, provincial, county, or other political subdivision, agency or other body, domestic or foreign or other government authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, court, tribunal or other entity).

"Hazardous Substances" means any substances, materials or wastes which are defined as or included in the definition of "hazardous substances", "hazardous wastes", "hazardous materials", "toxic substances", "pollutants" or "contaminants" under any Environmental Law, including any petroleum or refined petroleum products, radioactive materials, friable asbestos or polychlorinated biphenyls.

"IND" means (i) an Investigational New Drug application filed with the FDA in accordance with the FD&C Act, and (ii) any of its foreign equivalents as filed with the applicable Regulatory Authorities in other countries or regulatory jurisdictions in the world, as applicable, in each case ((i) and (ii)), including all supplements, amendments, variations, extensions, and renewals thereof that may be filed with respect to the foregoing.

"Intellectual Property" means any and all intellectual property and proprietary rights in any jurisdiction throughout the world, including rights arising from the following: (i) patents and patent applications, design rights, industrial design registrations and applications therefor, divisions, continuations, continuations-in-part, reissues, substitutes, renewals, registrations, confirmations, reexaminations, extensions and any provisional applications, and any foreign or international equivalent of any of the foregoing; (ii) trademarks (whether registered, unregistered or applied for), service marks, trade dress, service names, trade names, brand names, product names, slogans, logos, business names, corporate names, and other source or business identifiers, all registrations and applications for registration thereof,

and, in each case, together with all of the goodwill associated therewith; (iii) works of authorship, copyrights and all registrations and applications for registration thereof; (iv) trade secrets and Know-How; (v) rights in formulae, methods, techniques, processes, assembly procedures, software, software code (in any form, including source code and executable or object code), subroutines, test results, test vectors, user interfaces, protocols, schematics, specifications, drawings, prototypes, molds and models, and other forms of technology (whether or not embodied in any tangible form and including all tangible embodiments of the foregoing), and (vi) social media accounts, social media identifiers, internet domain name registrations.

"Intellectual Property Assignment Agreement" means the assignment agreement, dated as of the Closing Date, by and between the Seller and Purchaser, in substantially the form attached hereto as Exhibit C and acceptable to Purchaser.

"Intellectual Property Registrations" means, as to any Owned Intellectual Property Assets, any issuance, registration, application or other filing by, to or with any Governmental Authority or authorized private registrar in any jurisdiction, including domain names, registered trademarks and copyrights, issued and reissued patents and pending applications for any of the foregoing.

"Inventory" has the meaning set forth in Section 2.1(h).

"IRS" means the United States Internal Revenue Service.

"Know-How" means all technical, scientific, manufacturing, and other know-how and information, trade secrets, knowledge, technology, means, methods, processes, practices, formulae, instructions, skills, techniques, procedures, experiences, ideas, technical assistance, designs, drawings, assembly procedures, computer programs, apparatuses, specifications, data, results and other material, including: biological, chemical, pharmacological, toxicological, pharmaceutical, physical and analytical, pre-clinical, clinical, safety, manufacturing and quality control data and information, including study designs and protocols; assays; stability reports, production records, test methods, certificates of analyses, development reports, quality and technical agreements, and supplier audit reports and biological methodology; in each case (whether or not confidential, proprietary, patented or patentable) in written, electronic or any other tangible or intangible form now known or hereafter developed.

"Knowledge" means (a) with regard to the Seller, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Seller's chief executive officer, chief financial officer, and general counsel, in each case as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate) and (b) with regard to Purchaser, the actual knowledge, without any implication of verification or investigation concerning such knowledge, of Purchaser's chief executive officer as of the Agreement Date (or, with respect to a certificate delivered pursuant to this Agreement, as of the date of delivery of such certificate).

"Lambda IND" means any and all INDs owned or controlled by Seller or its Affiliates for any Licensed Product anywhere in the world, including IND # 154118.

"Lambda IND Transfer Date" means the date on which the transfer of all Lambda INDs by Seller or its Affiliates to Purchaser under this Agreement is complete such that Purchaser is considered the holder of all Lambda INDs by the applicable Regulatory Authority.

"Law Firm" means Sidley Austin LLP and its successors.

"Liabilities" means debts, liabilities, duties, obligations or commitments of any nature whatsoever, whether direct or indirect, asserted or unasserted, known or unknown, absolute or contingent, accrued or unaccrued, matured or unmatured or otherwise, whenever or however arising (including whether arising out of any Contract or in a tort claim based on negligence or strict liability).

"Licensed Compound" has the meaning given to it in the BMS License Agreement.

"Licensed Product" has the meaning given to it in the BMS License Agreement.

"Lien" means all forms of lien (including mechanic's, contractor's or other similar liens arising under or relating to the provision of goods or services on or to any Transferred Assets, and liens arising under the Bankruptcy Code), encumbrance, defect or irregularity in title, pledge, mortgage, deed of trust, deed to secure debt, security interest, charge, transfer restriction or similar agreement or encumbrance, including any dedication under any gathering, transportation, treating, processing, fractionating, purchase, sale or similar agreements, or any other rights granted or consensual as or against any Transferred Assets including but not limited to easements, encroachments, rights of first refusal, options, or any other interest or right in property that constitutes a lien or interest within the definition or adjudication of such terms under Section 101(37) of the Bankruptcy Code.

"Manufacture" means all activities related to the manufacturing of a pharmaceutical product, or any ingredient thereof, including but not limited to test method development and stability testing, formulation, process development, manufacturing scale-up, manufacturing, quality assurance/quality control development, quality control testing (including in-process release and stability testing), packaging, shipment and release of product or any component or ingredient thereof, quality assurance activities related to manufacturing and release of product or any component or ingredient thereof, and regulatory activities related to all of the foregoing.

"Material Adverse Effect" means a material adverse effect on the business, financial condition or results of operations of the Business (including the Transferred Assets and Assumed Liabilities) taken as a whole; provided, however, that none of the following shall be deemed (either alone or in combination) to constitute, and none of the following shall be taken into account in determining whether there has been or may be, a Material Adverse Effect: (a) any change in, or effects arising from or relating to, general business or economic conditions affecting any industry in which the Business operates; (b) any change in, or effects arising from or relating to, the United States or foreign economies, or securities, banking or financial markets in general, or other general business, banking, financial or economic conditions (including (i) any disruption in any of the foregoing markets, (ii) debt defaults or other restructuring events of any country with respect to which bondholders take a discount to the debt of any country or any increases in the interest rates for any country's debt, (iii) any change in currency exchange rates, (iv) any decline or rise in the price of any security, commodity, contract or index and (v) any increased cost, or decreased availability, of capital or pricing or terms related to any financing for the Transactions); (c) any change from, or effects arising from or relating to, the occurrence, escalation or material worsening of any act of God or other calamity, natural disaster, pandemic or disease, outbreak, hostility, act of war, sabotage, cyberattack or terrorism or military action; (d) any action taken by Purchaser or its Affiliates with respect to the Transactions or with respect to the Business; (e) any action taken, or failed to be taken, by the Seller at the request of or with the consent of Purchaser or otherwise in compliance with the terms of this Agreement or any change from, or effects arising from or relating to, Purchaser's failure to consent to any action restricted by Section 6.1; (f) any change in, or effects arising from or relating to changes in, Applicable Law or accounting rules (including GAAP) or any interpretation thereof; (g) the failure of the Business to meet any of its projections, forecasts, estimates, plans, predictions, performance metrics or operating statistics or the inputs into such items (whether or not shared with Purchaser or its Affiliates or representatives); (h) national or international political, labor or social conditions; (i) the public announcement of, entry into or pendency of, actions required or contemplated by or performance of obligations under, this Agreement and the Transactions or the identity of the parties to this Agreement; (i) the sale of any assets other than the Transferred Assets to any third parties by a member of the Seller Group or any of their Affiliates; (k) any effect arising or resulting from or related to the filing of the Bankruptcy Cases; (1) any action required to be taken under any Applicable Law or Order or any existing Contract by which any member of the Seller Group's (or any of their properties) are bound; (m) seasonal changes in the results of operations of the Seller Group; (n) any epidemic, pandemic, outbreak of disease or other public health emergency (including COVID-19) or any escalation or worsening of any such conditions or (o) any objections made in the Bankruptcy Court to this Agreement, the Transactions, the Sale Order or the reorganization, any orders of the Bankruptcy Court and any actions or omissions of the Seller in compliance with any order of the

Bankruptcy Court and the assumption or rejection of any Assigned Contract; except in the cause of clauses (a) through (c), (h) and (n), to the extent such conditions, events, changes, crises and disasters, as applicable, do not have a material and disproportionate impact on the Business, taken as a whole, compared to other industry participants (in which case, only the extent of such disproportionate effect shall be taken into account when determining whether there is a Material Adverse Effect).

"Merck License Agreement" means that certain License Agreement, dated September 3, 2010, by and between the Seller and Merck, and any and all amendments or supplements thereto, including that certain First Amendment, dated January 18, 2011, Amendment to License Agreement, dated June 11, 2013, Amendment #2 to License Agreement, dated November 20, 2014, Amendment #3 to License Agreement, dated March 6, 2015, Amendment #4 to License Agreement, dated June 9, 2015, Amendment #5 to License Agreement, dated December 17, 2015, Amendment #6 to License Agreement, dated May 15, 2018, and Amendment #7 to License Agreement, dated November 3, 2020.

"Non-Transferred Asset" has the meaning set forth in Section 2.6(a).

"Nonparty Affiliates" has the meaning set forth in Section 10.15.

"Notice of Readiness to Close" has the meaning set forth in Section 8.5.

"Order" means any award, decision, injunction, judgment, ruling or verdict entered, issued, made or rendered by any Governmental Authority or arbitrator.

"Organizational Documents" means (a) the articles or certificates of incorporation and the bylaws of a corporation, (b) the partnership agreement and any statement of partnership of a general partnership, (c) the limited partnership agreement and the certificate of limited partnership of a limited partnership, (d) the operating or limited liability company agreement and the certificate of formation of a limited liability company, (e) any charter, joint venture agreement or similar document adopted or filed in connection with the creation, formation or organization of a Person not described in clauses (a) through (d), and (f) any amendment to or equivalent of any of the foregoing.

"Outside Date" has the meaning set forth in Section 9.1(i).

"Owned Intellectual Property Assets" means the Intellectual Property owned or purported to be owned by any member of the Seller Group that is used in, held for use in, or related to, the conduct of the Business as currently conducted or proposed to be conducted, excluding any Intellectual Property related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.

"Permit" means all permits, authorizations, certificates, franchises, consents and other approvals from any Governmental Authority.

"Permitted Liens" means (a) Liens for Taxes, assessments or other governmental charges not yet due and payable or being contested in good faith by appropriate proceedings; (b) mechanics', carriers', workers', repairers' and other similar Liens arising or incurred in the ordinary course of business for obligations that are not overdue or are being contested in good faith by appropriate proceedings; (c) zoning, entitlement and building regulations and land use restrictions; (d) purchase money Liens and Liens securing rental payments under capital lease arrangements; (e) Liens arising under leases of property or equipment in favor of the owner thereof; (f) pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security; (g) deposits to secure the performance of bids, Contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business; (h) licenses of Intellectual Property granted in the ordinary course of business; (i) Liens arising under or created by this Agreement or any of the Related Documents; (j) Liens arising in the ordinary

course of business which would not reasonably be expected to have a Material Adverse Effect; and (k) Liens set forth on Schedule 1.1(a).

- "**Person**" means any individual, partnership, joint venture, limited liability company, corporation, firm, trust, association, unincorporated organization, governmental authority or agency, or any other entity not specifically listed herein.
- "Personal Information" means any information in the possession or control of the Seller Group (solely as related to the Business) about an identifiable individual other than the name, title or business address, business email address or telephone number of any employee of the Seller Group.
 - "Petition Date" has the meaning set forth in the Recitals.
- "Plan Consummation Date" means the date on which the Seller Group's plan in the Bankruptcy Cases is substantially consummated.
- "PMDA" means the Japanese Pharmaceutical and Medical Device Agency or its successor, or Ministry of Health, Labour and Welfare.
- "Pre-Closing Tax Period" means any taxable period ending on or prior to the Closing Date and the portion of any Straddle Period through the Closing Date.
 - "Preliminary Allocation Schedule" has the meaning set forth in Section 2.11(a).
 - "Previously Excluded Contract" has the meaning set forth in Section 5.5(b).
 - "Previously Unknown Contract" has the meaning set forth in Section 5.5(a).
 - "Provision" has the meaning set forth in Section 10.4.
- "Public Health Measures" means any closures, "shelter-in-place," "stay at home," workforce reduction, social distancing, shut down, closure, curfew or other restrictions or any other Applicable Law, Orders, directives, guidelines or recommendations issued by any Governmental Authority, the Centers for Disease Control and Prevention, the World Health Organization, or any industry group in connection with COVID-19 or any other epidemic, pandemic, or outbreak of disease, or in connection with or in response to any other public health conditions.
 - "Purchase Price" means the Base Price plus the aggregate amount of Purchaser Cure Amounts.
 - "Purchaser" has the meaning set forth in the preamble.
- "Purchaser Cure Amounts" means, with respect to Assigned Contracts, all Determined Cure Amounts up to \$269,000 in the aggregate.
 - "Purchaser Group Members" has the meaning set forth in Section 10.17.
 - "Purchaser Releasing Party" has the meaning set forth in Section 10.16(b).
 - "Purchaser Schedules" has the meaning set forth in ARTICLE 4.
- "Purchaser's FDA Transfer Letters" means the letters from Purchaser to FDA in form and substance acceptable to Purchaser, notifying FDA of the acceptance of the transfer from the Seller to Purchaser of all of Seller's right, title and interest in the Lambda IND.
 - "PV Services Stop Date" has the meaning set forth in Section 7.8(d).
- "Regulatory Applications" means (a) the single application or set of applications for approval and/or pre-market approval to Manufacture and sell commercially a pharmaceutical therapeutic product submitted to the FDA including, without limitation, any related registrations with or notifications to the FDA, and (b) any foreign equivalents to such applications filed with any other national or supranational

Regulatory Authority in the Territory, and (c) all supplements and amendments that may be filed with respect to any of the foregoing.

"Regulatory Approval" means any and all approvals (including pricing or pricing reimbursement approvals), licenses, registrations, or authorizations of any federal, national, multinational, state, provincial or local regulatory agency, department, bureau or other governmental entity necessary for the Manufacture, use, storage, import, export, transport, promotion, marketing or sale of a Licensed Compound or Licensed Product in the applicable country.

"Regulatory Authority" means any national or supranational governmental authority, including the FDA, PMDA or EMA, that has responsibility in countries in the Territory over the Development and/or Commercialization of the Licensed Compounds and/or Licensed Products.

"Regulatory Information" means any filings, submissions, applications, data, reports or correspondence, including, without limitation, dossiers, manufacturing data, drug master files, inspection reports, adverse event files and complaint files, with any Governmental Authority that relate to any Licensed Compound or Licensed Product, including any (a) INDs, Regulatory Applications, Regulatory Approvals, drug master files, applications for designation as an "Orphan Product" under the Orphan Drug Act, applications for designation as a humanitarian use device or a breakthrough device, for Fast Track or Breakthrough Therapy Designation, Accelerated Approval or Priority Review or for a Special Protocol Assessment or all other filings (including Regulatory Approval applications and counterparts to any of the foregoing in any country or region), (b) all supplements and amendments to any of the foregoing, and (c) all data and other information contained in, and correspondence relating to, any of the foregoing, in each case of any of the foregoing items listed in this definition, in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter.

"Related Claims" means all claims or causes of action (whether in contract or tort, in law or in equity, or granted by statute or otherwise) that may be based upon, arise out of or relate to this Agreement, the Related Documents and any other document or instrument delivered pursuant to this Agreement or the Related Documents, or the negotiation, execution, termination, validity, interpretation, construction, enforcement, performance or nonperformance of this Agreement or the Related Documents or otherwise arising from the Transactions or the relationship between the parties (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with, or as an inducement to enter into, this Agreement or the Related Documents).

"Related Documents" means the Escrow Agreement, the Bill of Sale and Assignment and Assumption Agreement, and Intellectual Property Assignment Agreement; provided, however, that the Escrow Agreement, the Bill of Sale and Assignment and Assumption Agreement, and Intellectual Property Assignment Agreement shall not be a Related Document solely for purposes of applying the provisions in ARTICLE 10 to the extent, and only to the extent, that any such document expressly conflicts with ARTICLE 10.

"Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing into the environment of any Hazardous Substances.

"Sale Order" means an Order of the Bankruptcy Court issued pursuant to sections 105(a), 363 and 365 of the Bankruptcy Code in form and substance acceptable to Purchaser and the Seller, approving this Agreement and all of the terms and conditions hereof and approving and authorizing the Seller to consummate the Transactions contemplated hereby Free and Clear and containing a finding that Purchaser has acted in "good faith" within the meaning of Section 363(m) of the Bankruptcy Code.

- "Schedules" has the meaning set forth in ARTICLE 3.
- "Seller" has the meaning set forth in the preamble.
- "Seller Access Contact" has the meaning set forth in Section 6.2(a).
- "Seller Cure Amounts" means, with respect to Assigned Contracts, any Determined Cure Amounts exceeding \$269,000 in the aggregate.
 - "Seller Financial Statements" has the meaning set forth in Section 3.9.
 - "Seller Group" means the Seller and each of its Affiliates.
 - "Seller Group Members" has the meaning set forth in Section 10.17.
- "Seller Group Taxes" means any (i) Liability of Seller Group for Taxes, (ii) any Liability for Asset Taxes attributable to any Pre-Closing Tax Period, and (iii) any Liability of Seller Group for the unpaid Taxes of any Person under Treasury Regulation §1.1502-6 (or any similar provision of state, local, or non-U.S. law), as a transferee or successor, by contract, or otherwise.
 - "Seller Permits" has the meaning set forth in Section 3.5.
 - "Seller Releasing Party" has the meaning set forth in Section 10.16(a).
- "Seller's FDA Transfer Letters" means the letters from the Seller to FDA in form and substance acceptable to Purchaser, notifying FDA of the transfer from the Seller to Purchaser of all of Seller's rights in the Lambda IND.
- "Solvent" when used with respect to any Person, means that, as of any date of determination, (a) the fair salable value (determined on a going concern basis) of its assets and property will, as of such date, exceed the amounts required to pay its debts as they become absolute and mature, as of such date, (b) such Person will have adequate capital to carry on its business and (c) such Person will be able to pay its debts as they become absolute and mature, in the ordinary course of business, taking into account the timing of and amounts of cash to be received by it and the timing of and amounts of cash to be payable on or in respect of its indebtedness.
 - "Specific Provision" has the meaning set forth in Section 10.4.
- "Storage Contract" means each Contract (or portion thereof) with a Third Party pursuant to which any Inventory are held for storage or other activities.
- "Straddle Period" means any taxable year or other taxable period beginning on or before and ending after the Closing Date.
 - "Supplemental Assignment Notice" has the meaning set forth in Section 5.5(a).
- "Supplemental Assignment Notice Objection Deadline" has the meaning set forth in Section 5.5(a).
- "Tax" means any tax of any kind whatsoever (including any income tax, franchise tax, branch profits tax, capital gains tax, value-added tax, unclaimed property, escheat, sales tax, use tax, property tax, transfer tax, payroll tax, social security tax or withholding tax), and any related fine, penalty, interest, or addition to tax with respect thereto, imposed, assessed or collected by or under the authority of any Governmental Authority.
- "Tax Return" means any return (including any information return), report, statement, schedule, notice, form, or other document or information (whether in tangible, electronic or other form), including any amendments, schedules attachments, supplements, appendices and exhibits thereto, filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority in connection with the determination, assessment, collection, or payment, of any Tax.

- "Territory" has the meaning given to it in the BMS License Agreement.
- "Third Party" means any Person other than a Contracting Party or its Affiliates.
- "Trademark" means, collectively, trademarks, service marks trade names, slogans, logos, trade dress or other similar source or origin identifiers (whether statutory or common law, whether registered or unregistered), together with all (a) registrations and applications for any of the foregoing, (b) extensions or renewals thereof, (c) goodwill (if any) connected with use thereof or symbolized thereby, and (d) rights and privileges arising under Applicable Law with respect to any of the foregoing.
- "Transactions" means the transactions contemplated by this Agreement and the Related Documents.
 - "Transfer Taxes" has the meaning set forth in Section 2.10.
 - "Transferred Assets" has the meaning set forth in Section 2.1.
 - "Transferred Business Books and Records" has the meaning set forth in Section 2.1(d).
- "Transferred Data" means any and all Data that (a) are owned or purported to be owned by the Seller or its Affiliates (including all such Data held by or on behalf of Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level)) and (b) relate to any Licensed Compound or Licensed Product, excluding any Data related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.
- "Transferred Materials" means the Transferred Data, Transferred Regulatory Information, Transferred Studies, Transferred Business Books and Records, and Inventory.
- "Transferred Regulatory Information" means any and all Regulatory Information that (a) are owned or purported to be owned by the Seller or its Affiliates (including all such Regulatory Information held by or on behalf of Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf the Seller or any of its Affiliates, directly or indirectly, at any level)) and (b) relate to any Licensed Compound or Licensed Product, excluding any Regulatory Information related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.
- "Transferred Studies" all clinical, preclinical, and non-clinical studies to the extent on-going as of the Closing being conducted by or on behalf of Seller or any of its Affiliates related to any Licensed Compound or Licensed Product, excluding any clinical, preclinical, and non-clinical studies related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement.

1.2 Other Definitional and Interpretive Matters.

- (a) Unless otherwise expressly provided, for purposes of this Agreement and the Related Documents, the following rules of interpretation shall apply:
 - (i) Calculation of Time Period. All references to a day or days shall be deemed to refer to a calendar day or days, as applicable, unless otherwise specifically provided. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

- (ii) Dollars. Any reference to \$ shall mean U.S. dollars, which is the currency used for all purposes in this Agreement and the Related Documents. The specification of any dollar amount in the representations and warranties or otherwise in this Agreement, the Related Documents or the Schedules is not intended and shall not be deemed to be an admission or acknowledgement of the materiality of such amounts or items, nor shall the same be used in any dispute or controversy between the parties hereto to determine whether any obligation, item or matter (whether or not described herein or included in any schedule) is or is not material for purposes of this Agreement, the Related Documents or the Schedules.
- (iii) Exhibits/Schedules. The Exhibits and Schedules to this Agreement are an integral part of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Schedule shall be deemed to have been disclosed on each other Schedule. Disclosure of any item on any Schedule shall not constitute an admission or indication that any such item is required to be disclosed, or that such item or matter is material or has resulted in or will result in a Material Adverse Effect or that the included items or actions are not in the ordinary course of business. No disclosure on a Schedule relating to a possible breach or violation of any Contract, Applicable Law or Order shall be construed as an admission or indication that a breach or violation exists or has actually occurred. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.
- (iv) Gender and Number. Any reference to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.
- (v) Headings. The provision of a table of contents, the division of this Agreement or Related Documents into articles, sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement or Related Document, as applicable. Unless otherwise specified, all references in this Agreement to any "Section" or other subdivision are to the corresponding section or subdivision of this Agreement, and all references in a Related Document to any "Section" or other subdivision are to the corresponding section or subdivision of such Related Document.
- (vi) Herein. The words such as "herein," "hereinafter," "hereof" and "hereunder" that are used in this Agreement refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires. Uses of such words in the Related Documents shall refer to such Related Document as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
- (vii) Or. The word "or" shall be construed in the inclusive sense of "and/or" unless otherwise specified.
- (viii) Including. The word "including" or any variation thereof means (unless the context of its usage otherwise requires) "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

- (ix) Successors. A reference to any party to this Agreement, any Related Document or any other agreement or document shall include such party's successors and permitted assigns.
- (x) Legislation. A reference to any legislation or to any provision of any legislation shall include any amendment thereto, and any modification or re-enactment thereof, any legislative provision substituted therefor and all regulations and statutory instruments issued thereunder or pursuant thereto.
- (xi) Reflected On or Set Forth In. An item arising with respect to a specific representation or warranty shall be deemed to be "reflected on" or "set forth in" a balance sheet or financial statement, to the extent any such phrase appears in such representation or warranty, if (a) there is a reserve, accrual or other similar item underlying a number on such balance sheet or financial statement that relates to the subject matter of such representation, (b) such item is otherwise specifically set forth on the balance sheet or financial statement or (c) such item is set forth in the notes to the balance sheet or financial statement.
- (xii) Made Available. Any reference in this Agreement to "made available" means a document or other item of information that was provided or made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions.
- (b) All representations and warranties set forth in this Agreement or the Related Documents are contractual in nature only and subject to the sole and exclusive remedies set forth herein. No Person is asserting the truth of any representation and warranty set forth in this Agreement or the Related Documents; rather, the parties have agreed that should any representations and warranties of any party prove untrue, the other parties shall have the specific rights and remedies herein specified as the exclusive remedy therefor, but that no other rights, remedies or causes of action (whether in law or in equity or whether in contract or in tort or otherwise) are permitted to any party hereto as a result of the untruth of any such representation and warranty. The phrase "to Seller's Knowledge" and phrases of similar import or effect are used herein to qualify and limit the scope of any representation or warranty in which they appear and are not affirmations of any Person's "superior knowledge" that the representation or warranty in which they are used is true.
- (c) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and the Related Documents and, in the event an ambiguity or question of intent or interpretation arises, this Agreement and the Related Documents shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement and the Related Documents. The parties hereto agree that changes from earlier drafts to the final version of this Agreement do not necessarily imply that the party agreeing to such change is agreeing to a change in meaning (as the party agreeing to such change may believe the change is stylistic and non-substantive); consequently, no presumption should exist by virtue of a change from a prior draft.

ARTICLE 2. THE PURCHASE AND SALE; CLOSING

2.1 **Purchase and Sale**. Upon the terms and subject to the conditions set forth in this Agreement and the Sale Order, at the Closing, in exchange for an aggregate payment from Purchaser to the

Seller equal to the Purchase Price, Purchaser shall purchase, assume and accept from the Seller, and the Seller shall sell, transfer, assign, convey and deliver (or shall cause the sale, transfer, assignment, conveyance and delivery) to Purchaser, Free and Clear (except for Permitted Liens), all of the rights, title and interests in, to and under the following assets and interests used in the Business as the same shall exist on the Closing Date (collectively, the "Transferred Assets"):

- (a) (i) all Contracts that are listed on Schedule 2.1(a) (as such Schedule may be amended pursuant to the terms of this Agreement, the "Contracts List"), and (ii) all other Contracts that are Assigned Contracts pursuant to Section 5.3(b), 5.4 and 5.5, including all rights, interests, credits, prepaid charges and expenses, deferred charges, advance payments, deposits, and prepaid items of Seller related thereto (collectively, the "Assigned Contracts");
- (b) the Owned Intellectual Property Assets, including the Intellectual Property Registrations listed on Schedule 3.12(a), as may be amended or supplemented with the agreement of the Seller at the request of Purchaser at any time prior to the Closing; *provided*, *however*, that any and all filing or transfer fees due to any Third Party (including any Governmental Authority) incurred by either party in connection with the transfer of such Intellectual Property Registrations shall be borne and paid by Purchaser;
- the Transferred Regulatory Information, including the information and (c) documents listed on Schedule 2.1(c), as may be amended or supplemented at the request of Purchaser at any time prior to the Closing; provided, however, that the Seller may retain copies of such Transferred Regulatory Information or may retain originals of the Transferred Regulatory Information and instead provide Purchaser with copies to the extent permissible under Applicable Laws and shall maintain the confidentiality thereof in accordance with the terms of the Confidentiality Agreement as Confidential Information, except Seller will be deemed the "Recipient" and Purchaser will be deemed "Company under the Confidentiality Agreement and Seller will be obligated to keep such Confidential Information from being disclosed for an indefinite period of time, mutatis mutandis unless otherwise required to be disclosed under Applicable Law, including by a Governmental Authority; provided, further, that the Parties shall cooperate in good faith to effectuate the assignments and transfer of the Transferred Regulatory Information with any applicable Governmental Authority, including duly executing and delivering, or causing to be duly executed and delivered, such instruments (including the filing of such assignments, agreements and documents) as may be necessary in order to affect such assignment and transfer of the Transferred Regulatory Information from the Seller to Purchaser;
- the Business Books and Records exclusively relating to the Business or any Licensed Compound or Licensed Product (including any Business Books and Records related to a combination of the "Licensed Compound" or "Licensed Product" as such terms are defined in the Merck License Agreement with the "Licensed Compound" or "Licensed Product" as such terms are defined in the BMS License Agreement, but excluding records and files not reasonably separable from documents and databases that do not relate exclusively to any the Business or Licensed Compound or Licensed Product and excluding any Business Books and Records related to a combination with any "Licensed Compound" or "Licensed Product" as such terms are defined in that the Merck License Agreement) ("Transferred Business Books and Records"); provided, however, that the Seller may retain copies of the Transferred Business Books and Records and shall maintain the confidentiality thereof in accordance with the terms of the Confidentiality Agreement as Confidential Information, except Seller will be deemed the "Recipient" and Purchaser will be deemed "Company under the Confidentiality Agreement and Seller will be obligated to keep such Confidential Information from being disclosed for an indefinite period of time, mutatis mutandis unless otherwise required to be disclosed under Applicable Law, including by a Governmental Authority; provided, further, that such Transferred Business Books and Records shall include solely such records created or acquired during the last three (3) years; provided, further, that the Seller will make

available, or cause to be made available, to Purchaser copies of Business Books and Records that are not Transferred Business Books and Records that relate to the Business or any Licensed Compound or Licensed Product but are not exclusively related thereto, that are in the possession of, owned by, or under the control (including via license) of the Seller or any of its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level) as of the Closing Date or at any time thereafter, and the Seller is permitted to redact or remove any extraneous or unrelated confidential or proprietary information in furtherance of such obligation, in each case such that Purchaser is able to conduct the Business and Develop, Manufacture, and Commercialize any Licensed Compound or Licensed Product as contemplated by this Agreement;

- (e) all rights to receive mail and other correspondences and communications (including electronic mail) addressed to Seller or any other member of the Seller Group relating to any Licensed Compound or Licensed Product (including any such mail and other correspondence and communications (including electronic mail) from the FDA or any other Governmental Authority, customers, advertisers, suppliers, distributors, agents and others) and payments with respect to any Licensed Compound or Licensed Product;
- (f) all of the Seller Group's rights, claims or causes of action, whether class, individual or otherwise in nature, under contract or in law or in equity, against third parties relating to the assets, properties, business or operations of the Seller Group with respect to the Business, the Transferred Assets and the Assumed Liabilities (including all guaranties, warranties, indemnities and similar rights in favor of the Seller Group or any their Affiliates to the extent solely related to the Transferred Assets or the Assumed Liabilities), in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date, except for such rights, claims and causes of action related to the Excluded Assets or Excluded Liabilities;
- (g) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, choses in action, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent used in or held for use for the Transferred Assets listed in clauses (a) through (f) above or the Assumed Liabilities:
- all right, title and interest in and to (i) any raw materials (including work in process, buffer stock held by vendors, dies and active pharmaceutical ingredients inventory, reference standards and materials, and all components and materials used in the Manufacture of any Licensed Compound or Licensed Product), finished goods and other inventory of all Licensed Compound and Licensed Product in the possession or control of, otherwise held by or on behalf of (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level), or owned by the Seller Group; and (ii) all good and marketable unbroken lots of packaged finished goods inventory of all Licensed Compound and Licensed Product in the possession or control of, or otherwise held by or on behalf of (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level), the Seller Group as of Closing, regardless of where located, and all rights to receive refunds, rebates or credits in connection therewith (for the avoidance of doubt, the Transferred Assets also include all manufactured product, packaging material, compounds and any other similar assets relating to any Licensed Compound or Licensed Product, and any assets that are under manufacture); in each case including the raw materials, reference standards and materials, and inventory listed in Schedule 2.1(h), as may be amended or supplemented at the request of Purchaser at any time prior to the Closing (collectively, "Inventory");

- (i) all Transferred Data:
- (i) all Transferred Studies;
- (k) all advertising, marketing, market research, sale and promotional files and materials (including any television, radio and print content and materials), pricing lists, consulting deliverables and other related literature, catalogs, point of sale materials and website content, including all Intellectual Property therein, relating to any Transferred Asset and Assumed Liability that are within the Seller Group's control or reasonably accessible to the Seller Group; and
- (l) to the extent not covered above, any goodwill associated with or symbolized by any of the foregoing Transferred Assets described in clauses (a) through (k) above and any properties, rights and interests of every kind and nature, whether tangible or intangible, real, personal or mixed, known or unknown, fixed or unfixed, accrued, absolute, contingent or otherwise, wherever located, associated with or appurtenant to the above-referenced Transferred Assets.
- 2.2 **Excluded Assets**. Notwithstanding the provisions of Section 2.1 or anything to the contrary herein, any and all assets, rights and properties of the Seller Group that are not specifically identified in Section 2.1 as Transferred Assets, including the following (collectively, the "**Excluded Assets**"), shall be retained by the Seller Group, and Purchaser and its designees shall acquire no right, title or interest in the Excluded Assets in connection with the Transaction:
- (a) all (i) cash and cash equivalents, wherever located, including bank balances and bank accounts or safe deposit boxes, monies in the possession of any banks, savings and loans or trust companies and similar cash items, (ii) escrow monies and deposits in the possession of landlords and utility companies, and (iii) investment securities and other short- and medium-term investments;
- (b) all records, documents or other information exclusively relating to current or former employees of the Seller Group that are not hired by Purchaser, and any materials to the extent containing information about any employee, disclosure of which would violate Applicable Law or such employee's reasonable expectation of privacy;
- (c) any interest of the Seller Group under this Agreement or the Related Documents, including the right to receive the Purchase Price and to enforce the Seller's rights and remedies thereunder;
- (d) all Excluded Contracts (including all prepaid assets relating to the Excluded Contracts), other than the Assigned Contracts, to which any member of the Seller Group or any of their respective Affiliates is a party;
- (e) any (i) Attorney-Client Information arising from communications prior to the Closing Date between a member of the Seller Group (including any one or more officers, directors or stockholders of such Seller Group member), on the one hand, and its counsel, on the other hand, and (ii) claims under any director and officer, errors and omissions, fiduciary and commercial crime insurance policies; and
- (f) any rights of the Seller Group to Tax refunds (or credits for overpayment of Taxes in lieu of a refund) attributable to any Pre-Closing Tax Period;
- (g) all Permits (including applications therefor and any trade or import/export Permits) that (i) are not materially related to the Business or (ii) are not transferable to Purchaser under Applicable Law;

- (h) the Excluded Books and Records;
- (i) any assets not otherwise designated as Transferred Assets or from time to time designated by the parties hereto as Excluded Assets;
- (j) all accounts receivable, intercompany obligations and other amounts receivable by the Seller Group;
 - (k) the Avoidance Actions;
- (l) all of the Seller Group's rights, claims or causes of action against third parties relating to the assets, properties, business or operations of the Seller Group (including all guaranties, warranties, indemnities and similar rights in favor of the Sellers Group or any of their Affiliates) to the extent arising under the Bankruptcy Code or relating to any of the Excluded Assets or Excluded Liabilities, in each case, whether arising by way of counterclaim or otherwise, and whether arising out of transactions occurring prior to, on or after the Closing Date; and
- (m) all prepaid expenses, claims, deposits, prepayments, refunds, causes of action, demands, actions, suits, rights of recovery, rights under guarantees, warranties, indemnities and all similar rights against third parties, rights of setoff and rights of recoupment, in each case, to the extent exclusively related to or exclusively used in or held for use for the Excluded Assets listed in clauses (a) through (l) above.

Notwithstanding anything to the contrary contained in this Agreement or any of the other Related Documents, Purchaser acknowledges and agrees that all of the following are also Excluded Assets, and all right, title and interest in and to all Excluded Assets shall be retained by the Seller Group and shall remain the property of the Seller Group (and shall expressly be excluded from the sale, transfer, assignment and conveyance to Purchaser hereunder), and neither Purchaser nor any of its Affiliates shall have any interest therein: (x) all records and reports prepared or received by the Seller Group or any of their Affiliates in connection with the sale of the Business and the Transactions, including all analyses relating to the Business or Purchaser so prepared or received; and (y) all confidentiality agreements with prospective purchasers of the Business or any portion thereof and all bids and expressions of interest received from third parties with respect thereto.

- 2.3 **Assumption of Liabilities**. On the terms and subject to the conditions set forth in this Agreement, Purchaser shall, effective as of the Closing, assume and agree to pay, discharge and perform in accordance with their terms the following Liabilities of the Seller Group arising from or related to the Business or the Transferred Assets as the same shall exist on the Closing Date arising only after the Closing Date (collectively, the "**Assumed Liabilities**"), including:
- (a) all Liabilities relating to the Transferred Assets other than the Assigned Contracts (which are addressed in Section 2.3(b)) solely to the extent such Liabilities relate to and arise in periods following the Closing;
- (b) subject to Section 2.4, all Liabilities arising under the Assigned Contracts other than the Existing Manufacturing Contracts solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Amounts;
- (c) subject to Section 2.4, all Liabilities arising under each Existing Manufacturing Contract solely to the extent such Liabilities relate to and arise in periods following the Closing, and all of the Purchaser Cure Amounts; and

- (d) all Taxes for which Purchaser is liable pursuant to this Agreement.
- 2.4 **Excluded Liabilities**. Notwithstanding Section 2.3, Purchaser is assuming only the Assumed Liabilities of the Seller Group and will not assume or be liable for any Excluded Liabilities (including Seller Group Taxes), and the Seller Group shall retain and shall be responsible for, all Liabilities that are not Assumed Liabilities, including all Liabilities related to Excluded Assets or any other Liabilities of the Business (all such Liabilities not being assumed herein referred to as the "**Excluded Liabilities**"). The Excluded Liabilities shall exclude any amounts payable or due to BMS for the assignment by Seller to Purchaser of the BMS License Agreement, respectively, whether arising in periods before or following the Closing, which shall be solely borne by Purchaser.
- 2.5 **Excluded Contracts**. Purchaser is electing to purchase only the Assigned Contracts, and Purchaser is not purchasing any other Contract of the Seller Group (any such other Contract an "**Excluded Contract**"). The Excluded Contracts shall constitute Excluded Assets and shall not be included in the Transferred Assets for any purposes of this Agreement and Purchaser shall not have any obligation to satisfy or pay any Cure Amounts or other Liabilities with respect to Excluded Contracts.

2.6 Nontransferable Assets and Liabilities.

- (a) Notwithstanding any other provision of this Agreement to the contrary, this Agreement shall not constitute an agreement to assign or transfer any Transferred Asset or any claim, right or benefit arising thereunder or resulting therefrom if an attempted assignment or transfer thereof, without the Consent of a third party (including any Governmental Authority) (after giving effect to the Sale Order or any other applicable order of the Bankruptcy Court that effects such transfer without any required Consents), would constitute a breach or other contravention thereof or a violation of Applicable Law (each, a "Non-Transferred Asset").
- If, on the Closing Date, any third-party Consent is not obtained for a Non-Transferred Asset, or if an attempted transfer or assignment thereof would be ineffective or a violation of Applicable Law, then, until any requisite consent is obtained therefor and the same is transferred and assigned to Purchaser or its designee, each such Non-Transferred Asset shall be held by the Seller as agent for Purchaser, and the Seller shall, to the extent permitted by Applicable Law, provide to Purchaser the benefits and Purchaser shall assume the obligations and bear the economic burdens associated with such Non-Transferred Asset. The Seller and Purchaser shall use commercially reasonable efforts to enter into agreements (including subcontracting, sublicensing or subleasing, if permitted) by which (i) the Seller shall, at Purchaser's sole expense, without interruption of the Business, provide Purchaser with the economic and operational equivalent of obtaining the requisite third-party Consent and assigning the applicable Non-Transferred Asset to Purchaser (including, with the prior written consent of Purchaser, enforcing for the benefit of Purchaser, and at Purchaser's sole expense, all claims or rights arising thereunder) and (ii) Purchaser shall perform, at its sole expense, the obligations and assume the economic burdens of the Seller or its Affiliates to be performed after the Closing with respect to such Non-Transferred Asset. Purchaser shall promptly, upon receipt of a written request therefor from the Seller, reimburse the Seller for all monies paid by the Seller on Purchaser's behalf in connection with any Assumed Liability not assigned or transferred to Purchaser pursuant to this Section 2.6.
- 2.7 **Closing**. The closing of the Transactions (the "**Closing**") will take place remotely by electronic exchange of documents on the date (the "**Closing Date**") that is the second (2nd) Business Day after the date on which all of the conditions set forth in ARTICLE 8 (excluding conditions that, by their terms, are to be satisfied at the Closing, but subject to the satisfaction or waiver of all such conditions at the Closing), have been satisfied or waived by the party hereto entitled the benefit of the same, unless another time or date is agreed to in writing by the parties hereto. Except as otherwise set forth herein, all proceedings to be taken and all documents to be executed and delivered by all parties hereto at the Closing

will be deemed to have been taken and executed simultaneously and no proceedings will be deemed to have been taken nor documents executed or delivered until all have been taken, executed, and delivered.

- 2.8 **Closing Deliveries of the Parties**. On the Closing Date (except as otherwise indicated):
- (a) Purchaser and the Seller shall execute and deliver the Bill of Sale and Assignment and Assumption Agreement;
- (b) Purchaser and the Seller shall execute and deliver the Intellectual Property Assignment Agreement;
- (c) Purchaser and the Seller shall transmit Purchaser's FDA Transfer Letter and the Seller's FDA Transfer Letters, respectively, to the FDA and shall take any other actions reasonably necessary to effect the transfer of the Lambda IND from the Seller to Purchaser;
- (d) Purchaser shall deliver, or cause to be delivered, to the Seller or the applicable Person each of the following:
 - (i) a certificate, dated as of the Closing Date, executed by or on behalf of Purchaser as to the satisfaction of the conditions set forth in Section 8.3(a) and Section 8.3(b); and
 - (ii) payment of the closing payments set forth in Section 2.9;
- (e) the Seller shall deliver, or cause to be delivered, to Purchaser or the applicable Person an IRS Form W-9 with respect to the Seller, duly completed and executed; and
- (f) the Seller shall deliver, or cause to be delivered, to the Seller or the applicable Person a certificate, dated as of the Closing Date, executed by or on behalf of the Seller as to the satisfaction of the conditions set forth in Section 8.2(a) and Section 8.2(b).
- (g) The "Closing" as defined in that certain Lonafarnib Asset Purchase Agreement, dated the date hereof, by and between the Seller and Purchaser takes place on the Closing Date of this Agreement.

2.9 Purchase Price; Assumed Liabilities; Deposits.

- (a) At the Closing, upon the terms and subject to the conditions set forth herein, in full consideration for the sale, transfer, conveyance, assignment and delivery of the Transferred Assets to Purchaser and assumption of the Assumed Liabilities by Purchaser, Purchaser shall (i) pay to the Seller an aggregate amount equal to the Purchase Price *minus* the Deposit Escrow Amount, which shall be released to the Seller by the Escrow Agent pursuant to Section 2.9(c), by irrevocable wire transfer of immediately available funds in accordance with payment instructions delivered by the Seller to Purchaser prior to the Closing; and (ii) assume the Assumed Liabilities.
- (b) At the Closing, on the terms and subject to the conditions set forth in this Agreement, Purchaser will assume and become responsible for the Assumed Liabilities. Purchaser agrees to pay, perform, honor, and discharge, or cause to be paid, performed, honored and discharged, all Assumed Liabilities in a timely manner in accordance with the terms hereof, including paying or causing to be paid, at or prior to the Closing, all Purchaser Cure Amounts. Seller agrees to pay all Seller Cure Amounts at or prior to the Closing.
 - (c) The Deposit Escrow Amount shall be distributed as follows:
 - (i) if the Closing shall occur, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available

funds, to an account designated by the Seller to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to the Seller at Closing and credited against the amount required to be paid by Purchaser to the Seller at Closing in accordance with Section 2.9(a);

- (ii) if this Agreement is terminated by the Seller pursuant to Section 9.1(g), (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to the Seller, by irrevocable wire transfer of immediately available funds, to an account designated by the Seller to the Escrow Agent and (B) the Deposit Escrow Amount, which shall constitute liquidated damages (and not a penalty), shall be delivered to the Seller within two (2) Business Days following delivery of such joint written instruction; or
- (iii) if this Agreement is validly terminated for any reason in accordance with the terms of this Agreement other than (x) by the Seller pursuant to Section 9.1(g) or (y) if Purchaser forfeits the Deposit Escrow Amount to the Seller pursuant to Section 8.5, (A) the Seller and Purchaser shall deliver a joint written instruction to the Escrow Agent in accordance with the Escrow Agreement instructing the Escrow Agent to release from the Deposit Escrow Account the entire Deposit Escrow Amount to Purchaser, by irrevocable wire transfer of immediately available funds, to an account designated by Purchaser to the Escrow Agent, and (B) the Deposit Escrow Amount shall be delivered to Purchaser within two (2) Business Days following delivery of such joint written instruction.

Any issue regarding the entitlement to the Deposit Escrow Amount shall be determined by the Bankruptcy Court, and Purchaser consents to the jurisdiction of the Bankruptcy Court for any issue related to this Agreement.

2.10 **Transfer Taxes**. Purchaser shall be solely responsible for, and shall indemnify, defend, and hold harmless the Seller Group for, any transfer, documentary, sales, use, excise, stock transfer, value-added, stamp, recording, registration and other similar taxes, levies and fees (including any penalties, fines and interest), together with any conveyance fees, recording charges and other similar fees and charges, incurred in connection with this Agreement and the Transactions (collectively, "**Transfer Taxes**"). Purchaser and the Seller shall cooperate in good faith to minimize, to the extent permissible under Applicable Law, the amount of any Transfer Taxes due with respect to the Transactions.

2.11 Allocation of Purchase Price.

- (a) The Purchase Price (including all other amounts treated as consideration for U.S. federal income tax purposes) and Assumed Liabilities shall be allocated as set forth on Schedule 2.11(a) (the "Preliminary Allocation Schedule"). Within ninety (90) days following the final determination of the Purchase Price, Purchaser shall deliver to the Seller a schedule allocating the Purchase Price (and all other amounts treated as consideration for U.S. federal income tax purposes) among the Transferred Assets (the "Allocation Schedule"). The Allocation Schedule shall be reasonable and shall be prepared in accordance with the Preliminary Allocation Schedule, and Purchaser and the Seller shall negotiate in good faith to resolve disputed items, if any, in the Allocation Schedule as promptly as practicable. If Purchaser and the Seller are unable to reach agreement with respect to the Allocation Schedule within thirty (30) days after the delivery of the Allocation Schedule by Purchaser to the Seller, the parties shall be entitled to use their own Purchase Price allocations for Tax reporting purposes.
- (b) To the extent Purchaser and the Seller agree on the Allocation Schedule pursuant to Section 2.11(a), Purchaser and the Seller shall (i) timely file all Tax Returns required to be filed in connection with the Allocation Schedule, and (ii) prepare and file all Tax Returns and determine all Taxes in a manner consistent with the Allocation Schedule, except as may be required by Applicable Law

and except as may be necessary to reflect adjustments to the Allocation Schedule resulting from post-Closing payments or events. Purchaser, on the one hand, and the Seller, on the other hand, shall notify the other if it receives notice that any Governmental Authority proposes any allocation different from Allocation Schedule.

- 2.12 **Escrow Accounts**. At the Closing, the Deposit Escrow Amount shall be used to satisfy a portion of the payment obligations of Purchaser pursuant to Section 2.9(c), otherwise the Deposit Escrow Amount shall be released to Purchaser or the Seller pursuant to Section 2.9(c). Upon the final release of all of the Deposit Escrow Amount pursuant to the terms of this Agreement and the Escrow Agreement, the Escrow Agreement shall automatically terminate. Any fees owed to the Escrow Agent and obligations under the Escrow Agreement shall be borne by Purchaser. The Deposit Escrow Amount shall be held in trust for the benefit of the Seller and shall not be subject to any encumbrance, attachment, trustee process or any other judicial process of any creditor of any party hereto, and shall be held and disbursed solely for the purposes of and in accordance with the terms of this Agreement and the Escrow Agreement.
- 2.13 **Tax Withholding**. Notwithstanding anything in this Agreement to the contrary, Purchaser shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any Person such amounts as it is required to deduct and withhold from such Person with respect to the making of such payment under the Code and the rules and regulations promulgated thereunder, or any provision of any Applicable Law relating to Taxes; *provided, however*, that Purchaser shall (i) provide commercially reasonable notice to the Person prior to such deduction and withholding and (ii) afford the Person a reasonable opportunity to provide any additional information, forms or certifications to establish an exemption from, or obtain a reduced rate of, withholding. To the extent that amounts are so withheld and properly remitted by Purchaser, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to such Person in respect of which such deduction and withholding was made by Purchaser.

ARTICLE 3. REPRESENTATIONS AND WARRANTIES OF THE SELLER

Except as disclosed in a document herewith delivered by the Seller to Purchaser (the "Schedules"), the Seller hereby makes the representations and warranties contained in this ARTICLE 3 to Purchaser. Organization, Good Standing and Other Matters. Each member of the Seller Group is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has, subject to the necessary authority of the Bankruptcy Court, the requisite corporate power and authority to operate the Business and necessary to own, lease or operate the properties and assets owned, leased or operated by it to carry on the Business as now being conducted, except where the failure to be so duly organized, validly existing and in good standing, or to have such power and authority, would not, individually or in the aggregate, have a Material Adverse Effect. Each member of the Seller Group is duly qualified to do business as a foreign company in each jurisdiction in which the nature of the Business as currently conducted by it or the property owned or leased by it makes such qualification necessary, except where the failure to be so qualified would not, individually or in the aggregate, have a Material Adverse Effect.

3.2 Authority and Enforceability. Subject to Bankruptcy Court approval, the Seller has all requisite power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement and the each of the Related Documents to which the Seller is (or at Closing, will be) a party thereto, and the consummation by the Seller of the Transactions, has been duly authorized and approved by all necessary limited liability company action on the part of the Seller and are subject to the approval of the Bankruptcy Court. This Agreement has been, and each Related Document will be, at or prior to the Closing, duly executed and delivered by the Seller and, assuming the due execution and delivery by the other parties hereto or thereto,

and subject to the approval of the Bankruptcy Court, constitutes a valid and binding obligation of the Seller, enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.

- No Conflict; Required Filings and Consents. Except (a) such filings as may be required in connection with the Transfer Taxes described in Section 2.10 and (b) as otherwise set forth on Schedule 3.3, the execution and delivery of this Agreement by the Seller does not and the execution and delivery of the Related Documents by the Seller will not, and the consummation of the Transactions hereby and thereby will not (i) violate the provisions of the Organizational Documents of any member of the Seller Group, (ii) subject to the entry of the Sale Order, violate any Applicable Law or Order to which any member of the Seller Group is subject or by which its properties or assets are bound, (iii) require any member of the Seller Group to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date (except as required by the Bankruptcy Code or the Sale Order), (iv) subject to the entry of the Sale Order, result in a breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any Assigned Contract or (v) subject to the entry of the Sale Order, result in the imposition or creation of any Lien upon or with respect to any of the assets or properties of the Seller Group; excluding from the foregoing clauses (ii) through (v) any Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, have a Material Adverse Effect.
- 3.4 Compliance With Laws. To the Seller's Knowledge, (i) the Seller Group is conducting the Business in compliance in all material respects with all material Applicable Laws applicable to the Business and (ii) no member of the Seller Group has received any written notice since the Petition Date of any material violations of any material Applicable Law applicable to their conduct of the Business. As of the Agreement Date, the Seller has and, to the Seller's Knowledge, has obtained all permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals of the FDA or any other Governmental Authority, currently used in, necessary for and material to the Development, Manufacture, and Commercialization of all Licensed Compounds and Licensed Products as presently conducted, all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals are included in the Transferred Assets and Seller has made available to Purchaser true and complete copies of all such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. As of the Agreement Date, neither Seller nor, to the Seller's Knowledge, any other Person has received any communication from any Governmental Authority that threatens to withdraw or suspend any such permits, licenses, certifications, registrations, qualifications, authorizations, consents or approvals. Seller has filed with the applicable Governmental Authority all required filings, declarations, listings, registrations, reports or submissions, including adverse event reports, necessary for and material to the Development, Manufacture, and Commercialization of Licensed Compounds and Licensed Products as presently conducted. All relevant filings, declarations, listings, registrations, reports or submissions were in material compliance with Applicable Law when filed, and no deficiencies have been asserted by any Governmental Authority with respect to any such filings, declarations, listing, registrations, reports or submissions. As of the Agreement Date, the Seller has not received or been subject to: (1) any FDA Form 483s directly relating to any Licensed Compound or Licensed Product; (2) any FDA notices of adverse findings relating to any Licensed Compounds or Licensed Product; or (3) any warning letters or other correspondence from the FDA or any other Governmental Authority in which the FDA or such other Governmental Authority asserted that the actions of Seller, with respect to any Licensed Compounds or Licensed Product, were not in compliance with Applicable Laws. There has not been any occurrence of any product recall, market withdrawal or replacement, or post-sale warning conducted by or on behalf of the Seller concerning any Licensed Compounds or Licensed Product or, to the Seller's Knowledge, any product recall, market withdrawal or replacement conducted by or on behalf of any entity as a result of any alleged defect in any

Licensed Compounds or Licensed Product.

- 3.5 **Permits**. To the Seller's Knowledge, (i) the Seller Group possess all material Permits required for the operation of the Business as currently conducted (the "Seller Permits") and (ii) no member of the Seller Group has received as of the Agreement Date any written notice of any cancellation, suspension, revocation, invalidation or non-renewal of any Permit since the Petition Date.
- 3.6 **Litigation**. As of the Agreement Date, there is no Action pending or, to the Seller's Knowledge, formally threatened in writing, against any member of the Seller Group before any Governmental Authority that would have a Material Adverse Effect or affect the Transferred Assets in any material respect after the entry of the Sale Order, if determined adversely and after taking into effect applicable insurance coverage.
 - 3.7 **Real Property**. The Seller Group does not own any real property.
- 3.8 **Assigned Contracts**. With respect to the Assigned Contracts, (i) except as a result of, or arising in connection with, the filing of the Bankruptcy Cases, no member of the Seller Group has received any written notice of any default or event that (with due notice or lapse of time or both) would constitute a default by the applicable member of the Seller Group under any Assigned Contract, other than defaults that have been cured or waived in writing or would not reasonably be expected to have a Material Adverse Effect, (ii) to the Seller's Knowledge, each Assigned Contract is a legal, valid and binding obligation of the applicable member of the Seller Group and is in full force and effect (except to the extent subject to, and limited by, the Enforceability Exceptions), (iii) to the Seller's Knowledge, no other party to any Assigned Contract is (with or without the lapse of time or the giving of notice, or both) in material breach of or in material default under any Assigned Contract and (iv) to the Seller's Knowledge, no member of the Seller Group has provided or received any notice of any intention to terminate any Assigned Contract. The Seller has made available to Purchaser true, correct and complete copies of each of the Assigned Contracts listed on Schedule 2.1(a), together with all amendments thereto.
- Report on Form 10-K filed with SEC on April 8, 2024 (the "Seller Financial Statements") have been prepared in accordance with GAAP (except as may be indicated in the notes to such financial statements or, in the case of unaudited financial statements, except as permitted by the SEC on Form 10-K under the Exchange Act, and except that the unaudited financial statements may not contain footnotes and are subject to normal and recurring year-end adjustments), have been prepared on a consistent basis throughout the periods covered thereby and presents fairly in all respects the financial condition of the Seller as of such dates and the results of operations of Seller for such periods, and are consistent with the books and records of Seller (which books and records are correct and complete in all material respects).
- 3.10 **Absence of Material Developments**. Except as disclosed on Schedule 3.10, since the Petition Date, there has occurred no fact, event, condition, change or circumstance which has had or would reasonably be expected to have a Material Adverse Effect.
- 3.11 **Customers and Suppliers.** Except as disclosed on Schedule 3.11(a), to the Knowledge of the Seller, since the Petition Date, no customer has or has threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions, purchasing materials, products or services from the Business. Except as disclosed on Schedule 3.11(b), to the Knowledge of the Seller, no supplier has or has threatened to stop or decrease the rate of, or as a result of the Bankruptcy Cases or the Transactions. supplying materials, products or services to the Business.

3.12 **Intellectual Property**.

- (a) A true, correct and complete list of all Intellectual Property Registrations included in the Owned Intellectual Property Assets is set forth on Schedule 3.12(a), including the Trademarks and domain names pertaining to Licensed Products that are owned by the Seller or its Affiliates as of the Agreement Date.
- (b) The Seller Group exclusively owns all Owned Intellectual Property Assets. Except as set forth on Schedule 3.12(b), and no member of the Seller Group is a party to, or bound by, (i) any license, royalty agreement, or other agreement relating to the use of any material Business Intellectual Property (other than non-exclusive licenses grant to a member of the Seller Group for commercially available, unmodified, off-the-shelf software licensed for aggregate annual fees of less than \$50,000), and (ii) agreements pursuant to which a member of the Seller Group settled any action, litigation, suit or other judicial or administrative proceeding, claim, assertion, or threat with respect to any material Business Intellectual Property, including settlement agreements, coexistence agreements, and consent agreements.
- (c) Other than with respect to Excluded Contracts or Assigned Contracts that Purchaser does not ultimately assume, no current or former Affiliate, partner, director, stockholder, officer, member, manager, employee, consultant or contractor of the Seller Group will, after giving effect to the Transactions, own, license or retain any Owned Intellectual Property Assets.
- (d) All material Intellectual Property Registrations remain pending or in full force and effect and have not expired or been abandoned or cancelled. To Seller's Knowledge, no interference, opposition, reissue, reexamination, or other proceeding is or has been pending or threatened, in which the scope, validity, or enforceability of any material Owned Intellectual Property Assets is being, has been challenged.
- (e) To the Knowledge of the Seller, the conduct of the Business does not infringe, misappropriate or otherwise violate in any material respect any Person's Intellectual Property.
- (f) To the Knowledge of the Seller's, no Person is currently infringing, misappropriating or otherwise violating any material Owned Intellectual Property Assets.
- (g) The Seller Group has taken commercially reasonable steps to safeguard and maintain the confidentiality of all trade secrets that constitute Owned Intellectual Property Assets, including by using good faith efforts to require all Persons having access thereto to execute written non-disclosure agreements.
- (h) The Seller Group complies with all Applicable Laws, internal policies and contractual obligations relating to privacy, data protection and cybersecurity.
- 3.13 **Taxes.** The Seller Group has timely filed all Tax Returns that it was required to file with respect to Transferred Assets. All such Tax Returns were correct and complete in all material respects. All Taxes owed by the Seller Group (whether or not shown or required to be shown on any Tax Return) with respect to Transferred Assets have been paid. There are no Liens on any of the Transferred Assets that arose in connection with any failure (or alleged failure) to pay any Tax. There is no dispute, examination, judicial proceeding or claim concerning any Taxes of the Seller Group with respect to the Transferred Assets.
- 3.14 **Product Liability.** Except as disclosed on Schedule 3.14, within the three (3) year period prior to the Closing Date there has not been any, and as of the Closing Date there is no pending, material litigation commenced against any member of the Seller Group relating to the sale, distribution or use of any

item sold or used in the Business (the "Goods"), including litigation with respect to product liability or recall claims.

- 3.15 **Product Warranties; Product Returns.** Except for warranties arising solely pursuant to Applicable Law or in the ordinary course of business, (a) no member of the Seller Group has made any material warranties, express or implied, written or oral, to any third party with respect to any of the Goods within the three (3) year period prior to the Closing Date, and (b) there is no, and within the three (3) year period prior to the Closing Date there has not been any, material litigation pending or, to the Seller's Knowledge, threatened with respect to any such warranty.
- 3.16 **Brokers and Finders**. Except for SSG Advisors, LLC, the Seller has not, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.
- 3.17 **Inventory**. To Seller's Knowledge, the Inventory consists of all materials used to Manufacture or otherwise incorporated into the Licensed Product (including raw materials and active pharmaceutical ingredients) and inventory of Licensed Product exclusively owned by the Seller and its Affiliates as of the Closing Date. As of the Closing Date, Schedule 3.17 identifies the location of all Inventory and sets forth a complete and accurate list of all Storage Contracts and provides reasonable details with respect to the Inventory subject to each such Storage Contract.
- 3.18 **No Other Representations or Warranties**. Except for the representations and warranties contained in this ARTICLE 3 and the Related Documents, the Seller does not, nor do any other Persons on behalf of the Seller, make any other express or implied representation or warranty with respect to itself, the Business, the Transferred Assets or the Assumed Liabilities, or with respect to any other information provided to Purchaser or its representatives, and the Seller disclaims any other representations or warranties, whether made by or on behalf of the Seller or any other Person. The Seller will not, and no other Persons will, have or be subject to any Liability to Purchaser or any other Person resulting from the distribution to Purchaser, or Purchaser's use of, any such information, including any information, documents, projections, forecasts or other material made available to Purchaser or its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever (electronic or otherwise) or otherwise in expectation of the Transactions.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as disclosed in a document herewith delivered by Purchaser to the Seller (the "**Purchaser Schedules**"), Purchaser hereby makes the representations and warranties contained in this ARTICLE 4 to the Seller.

- 4.1 **Organization, Good Standing and Other Matters**. Purchaser is duly organized, validly existing and in good standing under the Applicable Laws of its jurisdiction of organization and has all requisite corporate power or other entity power and authority to own its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed to conduct its business as currently conducted and is in good standing in each jurisdiction in which the location of the property owned, leased or operated by it or the nature of its business makes such qualification necessary, except where the failure to be so qualified or licensed would not, individually or in the aggregate, materially impair or delay Purchaser's ability to consummate the Transactions.
- 4.2 **Authority and Enforceability**. Purchaser has all requisite corporate power or other entity power and authority to execute and deliver this Agreement and each of the Related Documents to which it is (or at Closing, will be) a party and to perform its obligations hereunder and thereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement and each of the Related

Documents to which it is (or at Closing, will be) a party, and the consummation of the Transactions, have been duly authorized and approved by its board of directors (or equivalent governing body) and no other action on the part of Purchaser or its members is necessary to authorize the execution, delivery and performance of this Agreement and the Related Documents by Purchaser and the consummation of the Transactions. This Agreement has been, and each Related Document will be at or prior to Closing, duly executed and delivered by Purchaser and, assuming the due execution and delivery by the other parties hereto or thereto, constitutes a valid and binding obligation of Purchaser enforceable against it in accordance with its respective terms, except to the extent that such enforceability may be subject to, and limited by, the Enforceability Exceptions.

- 4.3 **No Conflict: Required Filings and Consents.** Except (a) such filings as may be required in connection with the Transfer Taxes described in Section 2.10 and (b) as set forth on Schedule 4.3, the execution and delivery of this Agreement and of the Related Documents and the consummation of the Transactions by Purchaser will not (i) violate the provisions of its Organizational Documents, (ii) violate any Applicable Law or Order to which it is subject or by which any of its properties or assets are bound, (iii) require it to obtain any Consent, or give any notice to, or make any filing with, any Governmental Authority on or prior to the Closing Date, (iv) result in a material breach of or constitute a default (with or without due notice or lapse of time or both), give rise to any right of termination, cancellation or acceleration under, or require the Consent of any third party to, any material Contract to which it is a party or (v) result in the imposition or creation of any Lien upon or with respect to any of its assets or properties; excluding from the foregoing clauses (ii) through (v) Consents, approvals, notices and filings the absence of which, and violations, breaches, defaults, rights of acceleration, cancellation or termination, and Liens, the existence of which would not, individually or in the aggregate, (A) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (B) otherwise prevent, hinder or delay the consummation of the Transactions.
- 4.4 **Financing**. Purchaser has, and at the Closing will have, (a) sufficient internal funds (without giving effect to any unfunded financing regardless of whether any such financing is committed) available to pay the Purchase Price in accordance with the terms hereof and any other payments required hereunder and any expenses incurred or required to be paid by Purchaser in connection with the Transactions, and (b) the resources and capabilities (financial or otherwise) to perform its obligations hereunder and under the Related Documents. Purchaser has not incurred any obligation, commitment, restriction, or Liability of any kind, which would impair or adversely affect such resources and capabilities.
- 4.5 **Solvency**. Purchaser is not entering into this Agreement with the intent to hinder, delay or defraud either present or future creditors. Immediately after giving effect to all of the Transactions, including the making of the payments contemplated by Section 2.9, and assuming satisfaction of the conditions to Purchaser's obligation to consummate the Transactions as set forth herein, the accuracy of the representations and warranties of Purchaser set forth herein and the performance by Purchaser of its obligations hereunder in all material respects, Purchaser will be Solvent.
- 4.6 **Litigation**. There is no Action pending or, to Purchaser's Knowledge, formally threatened against Purchaser or involving any of its properties or assets that would be reasonably be expected to

- (a) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (b) otherwise prevent, hinder, or delay the consummation of the Transactions.
- 4.7 **Brokers and Finders**. None of Purchaser or its Affiliates have, directly or indirectly, entered into any agreement with any Person that would obligate the Seller to pay any commission, brokerage fee or "finder's fee" in connection with the Transactions.

4.8 Non-Reliance of Purchaser; No Other Representations and Warranties.

- Except for the specific representations and warranties expressly made by the Seller in ARTICLE 3 and Related Documents as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, Purchaser acknowledges and agrees that (i) the Seller is not making and have not made any representation or warranty, expressed or implied, at law or in equity, in respect of the Business, the Transferred Assets, the Assumed Liabilities, or any of its operations, prospects or condition (financial or otherwise), including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any Liabilities, the prospects of the Business, the effectiveness or the success of any operations, or the accuracy or completeness of any confidential information memoranda, documents, projections, material or other information (financial or otherwise) regarding the Business furnished to Purchaser or its representatives or made available to Purchaser and its representatives in any "data rooms," "virtual data rooms," management presentations or in any other form in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever, and (ii) no officer, director, manager, stockholder, agent, Affiliate, advisor, representative or employee of the Seller Group has any authority, express or implied, to make any representations, warranties or agreements not specifically set forth in ARTICLE 3 and subject to the limited remedies herein provided.
- (b) Other than the specific representations and warranties expressly set forth in ARTICLE 3 as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement, Purchaser specifically disclaims that it is relying upon or has relied upon any such other representations or warranties that may have been made by any Person, and acknowledges and agrees that the Seller and the Seller's Affiliates have specifically disclaimed and do hereby specifically disclaim, and shall not have or be subject to any Liability for reliance on any such other representation or warranty made by any Person. Purchaser specifically waives any obligation or duty by the Seller or the Seller's Affiliates to make any disclosures of fact not required to be disclosed pursuant to the specific representations and warranties expressly set forth in ARTICLE 3 and disclaim reliance on any information not specifically required to be provided or disclosed pursuant to the specific representations and warranties set forth in ARTICLE 3.
- (c) Purchaser is acquiring the Business, the Transferred Assets and the Assumed Liabilities subject only to the specific representations and warranties expressly set forth in ARTICLE 3 as further limited by the specifically bargained-for exclusive remedies as set forth in ARTICLE 9 of this Agreement.
- 4.9 **No Other Representations or Warranties.** Except for the representations and warranties contained in this ARTICLE 4, neither Purchaser nor any other Person on behalf of Purchaser makes any other express or implied representation or warranty with respect to Purchaser or with respect to any other information provided to the Seller or its representatives, and Purchaser disclaims any other representations or warranties, whether made by Purchaser or any of its Affiliates, officers, directors, employees, agents or representatives.

ARTICLE 5. BANKRUPTCY COURT MATTERS

5.1 **Competing Transaction**. This Agreement is subject to approval by the Bankruptcy Court and the consideration by the Seller of higher or better competing bids in respect of all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) in accordance with the terms of the Bid Procedures Order (each, a "**Competing Bid**"). From the Agreement Date (and any prior time) and until the Closing, the Seller is permitted to, and to cause its representatives to, initiate contact with, solicit or encourage submission of any inquiries, proposals or offers by, any Person (in addition to Purchaser and its Affiliates and representatives) in connection with any sale or other disposition of the Transferred Assets. In addition, the Seller shall have the authority to respond to any inquiries or offers to purchase all or any part of the Transferred Assets (whether in combination with other assets of the Seller Group or otherwise) and perform any and all other acts related thereto which are required under the Bankruptcy Code, the Bid Procedures Order or other Applicable Law, including supplying information relating to the Business and the assets of the Seller Group to prospective purchasers.

5.2 Bankruptcy Court Filings.

- Subject to its right to pursue a Competing Bid in accordance with the Bid Procedures Order, the Seller shall diligently pursue the entry by the Bankruptcy Court of the Sale Order, which Sale Order shall provide for the transfer of the Transferred Assets and the Assumed Liabilities to Purchaser free from all successor or transferee Liability to the fullest extent permitted by Section 363 of the Bankruptcy Code. The Seller shall comply (or obtain an Order from the Bankruptcy Court waiving compliance) with all requirements under the applicable provisions of the Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the Local Bankruptcy Rules for the Bankruptcy Court in obtaining the entry of the Sale Order. The Seller further covenants and agrees that, after entry by the Bankruptcy Court of the Sale Order, and provided that the Sale Order becomes a Final Order, the terms of any other proposed order submitted by the Seller to the Bankruptcy Court shall not conflict with, supersede, abrogate, nullify or restrict the terms of this Agreement, or in any way prevent or interfere with the consummation or performance of the Transactions. Purchaser agrees that it will promptly take such actions as are reasonably requested by the Seller to assist in obtaining entry of the Sale Order, including by furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Purchaser under this Agreement and demonstrating that Purchaser is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. In the event, if the entry of the Sale Order shall be appealed, the Seller and Purchaser shall use their respective commercially reasonable efforts to defend such appeal.
- (b) Seller shall use commercially reasonable efforts to provide Purchaser with a reasonable opportunity to review and comment upon all motions, applications, and supporting papers relating to the transactions contemplated by this Agreement prepared by Seller or any Affiliates (including forms of orders and notices to interested parties) prior to the filing thereof in the Bankruptcy Cases; provided that the foregoing shall not require the Seller to take any action that would, in Seller's reasonable business judgment, threaten to harm the overall value to be produced by the Seller's in-court sale process.
- (c) The form of Sale Order submitted by the Seller to the Bankruptcy Court for approval shall be in a form and substance reasonably acceptable to Purchaser.
- (d) Seller shall not seek any modification to the Bid Procedures, Bid Procedures Order, or Sale Order by the Bankruptcy Court that are materially adverse to Purchaser without the prior written consent of Purchaser, which such consent shall not be unreasonably withheld.
- (e) Each of Purchaser and Seller will promptly take such actions as are reasonably requested by the other party to assist in obtaining entry of the Sale Order, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for purposes, among

others, of providing necessary assurances of performance by Seller of its obligations under this Agreement and demonstrating that Purchaser is a good faith buyer under section 363(m) of the Bankruptcy Code.

- (f) Seller shall use commercially reasonable efforts to provide appropriate notice of the hearings on the Sale Order to all Persons entitled to notice, including, but not limited to, all Persons that have asserted Liens on the Transferred Assets, all parties to the Assigned Contracts and all taxing authorities in jurisdictions applicable to Seller and as otherwise required by the Bankruptcy Code and bankruptcy rules.
- (g) Within five (5) Business Days of the Auction (subject to the Bankruptcy Court's availability), if Purchaser is the successful bidder at the Auction (or if there is no Auction), Seller will seek entry of the Sale Order by the Bankruptcy Court.
- (h) The Seller and Purchaser agree that, in the event that Purchaser is not the winning bidder at an auction undertaken pursuant to the Bid Procedures Order (the "Auction"), and (i) Purchaser submits the Back-Up Bid at the Auction or (ii) the terms of this Agreement are deemed to constitute a Back-Up Bid, then Purchaser shall be obligated to promptly consummate the Transactions upon the terms and conditions as set forth herein, including the payment of the Purchase Price as the same may be increased by Purchaser at the Auction; provided that the Seller gives written notice to Purchaser on or before the Back-up Termination Date, stating that the Seller (A) failed to consummate the sale of the Transferred Assets with the winning bidder, and (B) terminated the purchase agreement with the winning bidder.

5.3 Assumption of Assigned Contracts.

- (a) On June 4, 2024, the Seller filed (or caused to be filed) a notice of assumption (the "Assumption Notice") with the Bankruptcy Court and served such notice on each counterparty to a Contract listed thereon. The Assumption Notice identified all Contracts that the Seller and Purchaser believe may be assumed and assigned in connection with the sale of the Transferred Assets and set forth a good faith estimate of the amount of Cure Amounts applicable to each such Contract (and if no Cure Amount is estimated to be applicable with respect to any particular Contract, the amount of such Cure Amount designated for such Contract shall be "\$0.00"). In accordance with the Bid Procedures Order, the Seller reserves the right to supplement such list of Contracts and provide additional notice of assumption.
- (b) On or before the date that is three (3) Business Days before the Closing Date (the "Designation Deadline"), Purchaser shall provide to the Seller a Contracts List, which shall identify all Contracts that Purchaser elects to have assumed and assigned to Purchaser on the Closing Date. For the avoidance of doubt, Purchaser shall be entitled, with the agreement of Seller, to add, or, in the sole discretion of Purchaser, to remove, any Contracts from the Contracts List at any time (or multiple times) prior to the Designation Deadline by providing to the Seller by email a copy of the amended Contracts List. Any Contracts List that Purchaser delivers to the Seller prior to the Designation Deadline shall be deemed to replace and supersede any Contracts List that Purchaser had previously delivered to Seller. For the avoidance of doubt, only those Contracts that are identified on the Contracts List as of the Designation Deadline shall constitute Assigned Contracts and will be assumed by the Seller and assigned to Purchaser pursuant to the Sale Order. The Seller shall file such motions or pleadings as may be appropriate or necessary to assume and assign the Assigned Contracts and to determine the amount of the Cure Amounts; provided that nothing herein shall preclude the Seller from filing one or more motions to reject any Contracts that are not identified on the Contracts List as of the Designation Deadline.
- (c) Notwithstanding any provision in this Agreement to the contrary, a Contract shall not be an Assigned Contract hereunder and shall not be assigned to, or assumed by, Purchaser to the extent that such Contract is (i) deemed rejected under Section 365 of the Bankruptcy Code, (ii) the

subject of an objection to assignment or assumption or requires the consent of any Governmental Authority or other third party (other than, and in addition to, the Bankruptcy Court) in order to permit the assumption and assignment by the applicable Seller to Purchaser of such Contract pursuant to Section 365 of the Bankruptcy Code, and such objection has not been resolved or such consent has not been obtained prior to the thirtieth (30th) day following the Closing Date (as such period may be extended by mutual agreement of Seller and Purchaser), or (iii) terminated by any party thereto other than Seller, or terminates or expires by its terms, on or prior to such time as it is to be assumed by and assigned to Purchaser as an Assigned Contract hereunder and is not continued or otherwise extended upon assumption. In no event shall the failure to assign to Purchaser any Contract in accordance with subsections (i) through (iii) above reduce the Purchase Price payable to Seller or constitute a failure to satisfy the conditions precedent of Seller under Section 8.3.

- (d) Subject to the terms of Section 2.5, Section 2.8, Section 5.3(a) and Section 5.3(b), and subject to the entry of an order (which may be the Sale Order) of the Bankruptcy Court authorizing the assignment to Purchaser of the Assigned Contracts, Purchaser shall make provision for the payment of the Purchaser Cure Amounts, and Seller shall make provision for the payment of the Seller Cure Amounts, in cash at Closing in accordance with the Sale Order.
- (e) Notwithstanding any provision in this Agreement to the contrary, from and after the date of the Assumption Notice through the Closing Date, the Seller will not reject or take any action (or fail to take any action that would result in rejection by operation of Applicable Law) to reject, withdraw, repudiate or disclaim any Assigned Contract unless (i) Purchaser has provided its prior written consent; or (ii) Purchaser has removed such Assigned Contract from the list of Assigned Contracts.
- **Disputed Contracts.** In the event of an objection by a Contract counterparty to the Cure Amount asserted by Seller with regard to any Contract on the Contract List (such contract, a "Disputed Contract"), Seller shall either settle the objection of such party or shall litigate such objection under procedures as established by the Bankruptcy Court. In no event shall the Seller settle a Cure Amount objection with regard to any potential Assigned Contract without the express written consent (such consent not to be unreasonably withheld) of Purchaser (with an email consent being sufficient). In the event that a dispute regarding the Cure Amounts with respect to a Contract has not been resolved as of the Closing, the parties shall nonetheless remain obligated to consummate the transactions contemplated by this Agreement. Upon entry of an Order of the Bankruptcy Court (if necessary) determining any Cure Amount and authorizing the assumption and assignment to Purchaser of such Disputed Contract after the Closing, which order shall be in form and substance acceptable to Purchaser (a "Disputed Contract Order"), Purchaser shall have the option to designate the Disputed Contract as an Assigned Contract or an Excluded Contract (regardless of whether such contract was identified on the Contracts List). If Purchaser elects to designate the Disputed Contract as an Excluded Contract, (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the Sale Order and this Agreement, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract. If Purchaser elects to designate the Disputed Contract as an Assigned Contract, such Disputed Contract shall be deemed an Assigned Contract for all purposes hereunder and, for the avoidance of doubt, Purchaser shall assume the Disputed Contract and shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Disputed Contract; and (if applicable) Seller shall be responsible for paying all related Seller Cure Amounts; provided, however, that if Purchaser does not designate such Disputed Contract as either an Excluded Contract or an Assigned Contract within five (5) Business Days after the date of the Disputed Contract Order (or such later date as agreed by the Seller and Purchaser), (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the

Sale Order and this Agreement, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract.

5.5 Previously Unknown and Previously Excluded Contracts.

- If at any time, prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, it is discovered that a Contract material to the operation of the Business should have been identified on the Assumption Notice but was not so listed (any such Contract, a "Previously Unknown Contract"), Seller shall, promptly following the discovery thereof (but in no event later than five (5) Business Days following the discovery thereof), notify Purchaser in writing of such Previously Unknown Contract and provide Purchaser with a copy of such Previously Unknown Contract and the Cure Amount (if any) in respect thereof. Purchaser shall thereafter deliver written notice to Seller (email being sufficient), no later than ten (10) Business Days following such notice of such Previously Unknown Contract from Seller, if Purchaser elects for such Previously Unknown Contract to be an Assigned Contract. If Purchaser elects for a Previously Unknown Contract to be an Assigned Contract in accordance with this Section, then to the extent not previously filed and served, Seller shall file and serve an assignment and assumption notice on the Contract counterparty to such Previously Unknown Contract (a "Supplemental Assignment Notice") notifying such Contract counterparty of Seller's intention to assume and assign to Purchaser such Previously Unknown Contract, including the proposed Cure Amount (if any). Such notice shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser (the "Supplemental Assignment Notice Objection Deadline"). Following expiration of the Supplemental Assignment Notice Objection Deadline and, if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under this Agreement and the Sale Order. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under this Agreement.
- At any time prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, Purchaser may elect to take an assignment of any Excluded Contract that has not yet been assumed and assigned pursuant to an order of the Bankruptcy Court (a "Previously Excluded Contract") by sending a written notice to Seller (email being sufficient) of such election. If Purchaser elects for a Previously Excluded Contract to be an Assigned Contract in accordance with this Section, then to the extent not previously filed and served, Seller shall file and serve a Supplemental Assignment Notice on the Contract counterparty to such Previously Excluded Contract. Such Supplemental Assignment Notice Objection Deadline shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser. Following expiration of the Supplemental Assignment Notice Objection Deadline and if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under this Agreement and the Sale Order. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under this Agreement.
- (c) Seller and Purchaser agree that the Sale Order shall contain language approving the assumption and assignment procedures with respect to Disputed Contracts, Previously Unknown Contracts and Previously Excluded Contracts as set forth in Sections 5.3(b), 5.4 and 5.5 hereof.

ARTICLE 6. PRE-CLOSING COVENANTS

- Conduct of Business. Except (i) as set forth on Schedule 6.1, (ii) as may be approved by Purchaser (which approval will not be unreasonably withheld, delayed or conditioned; *provided, however*, that the consent of Purchaser shall be deemed to have been given if Purchaser does not object within forty-eight (48) hours after written request for such consent is provided by the Seller to Purchaser), (iii) for actions taken or omitted to be taken by any member of the Seller Group in response to any Public Health Measure, or (iv) as is otherwise permitted, contemplated or required by this Agreement, any Assigned Contract, by Applicable Laws or by order of the Bankruptcy Court, from the Agreement Date through the earlier of the Closing Date or the termination of this Agreement in accordance with its terms:
- (a) The Seller Group shall use their commercially reasonable efforts to carry on the Business in all material respects in the ordinary course of business as it has been conducted since the Petition Date; and
 - (b) The Seller shall not, and shall cause its Affiliates not to:
 - (i) sell, license, abandon or otherwise dispose of any material asset or property constituting Transferred Assets other than, in each case, in the ordinary course of business or for the purpose of disposing of obsolete or worthless assets;
 - (ii) except in the ordinary course of business, acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of any business or any corporation, partnership or other business organization or otherwise acquire any assets (except inventory), that as of the Closing would constitute Transferred Assets, except for the acquisition of assets in the ordinary course of business;
 - (iii) change its present accounting methods or principles in any material respect, except as required by GAAP or Applicable Law;
 - (iv) make or change any Tax election, change an annual accounting period, adopt or change any Tax accounting method, file any amended Tax Return, enter into any closing agreement, settle any material Tax claim or assessment or surrender any right to claim a refund of Taxes, other than in the ordinary course of business or as required by the Code or Applicable Law, and in each case that could have a material effect on the amount of Taxes due from the Business or due as a result of the Transferred Assets for a taxable period (or portion thereof) beginning after the Closing Date;
 - (v) compromise or settle any material litigation relating to the Business or cancel or compromise any material claim or waive or release any material right that, in each case, is related to the Business or a Transferred Asset;
 - (vi) encumber, transfer, abandon, allow to lapse, fail to prosecute or maintain, exclusively license, or otherwise dispose of any material Business Intellectual Property or Regulatory Approvals, except, in each case, other than in the ordinary course of business and other than the expiration of the statutory term of any Intellectual Property;
 - (vii) materially modify, materially breach, repudiate, reject, or terminate any Assigned Contract, or waive, release or assign any material rights or claims under any Assigned Contract;

- (viii) grant, impose or suffer to be imposed any Lien upon any of the Transferred Assets other than Permitted Liens or Liens that will be cured prior to the Closing; and
- (ix) authorize, agree or otherwise commit, whether or not in writing, to do any of the foregoing.
- (c) Notwithstanding anything to the contrary, nothing contained in this Agreement shall give Purchaser or any of its Affiliates, directly or indirectly, any right to control or direct the Business, assets and operations prior to the Closing. Prior to the Closing, the Seller shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its Business, assets and operations, subject to Purchaser's consent rights and Seller's obligations, in each case as expressly set forth in this Agreement.

6.2 Access to Information; Confidentiality.

- From the Agreement Date until the earlier of the Closing Date and the termination of this Agreement, the Seller shall grant Purchaser and its representatives (at Purchaser's sole cost and expense) reasonable access, during normal business hours and upon reasonable notice (and in the event of a facility visit request, at least forty-eight (48) hours prior notice), and subject to any limitations resulting from any Public Health Measures, to the personnel, facilities, book and records of the Seller Group related to the Business or the Transferred Assets that are in the possession of, owned by, or under the control (including via license) of the Seller Group, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller Group (including by or on behalf of any contractors or other service providers acting on behalf of the Seller Group, directly or indirectly, at any level); provided, however, that (i) all requests for access shall be directed to such other person(s) as the Seller may designate in writing from time to time (the "Seller Access Contact"), (ii) such activities do not unreasonably interfere with the ongoing business or operations of the Seller Group, (iii) the Seller shall have the right to have one or more of its representatives present at all times during any visits, examinations, discussions or contacts contemplated by this Section 6.2(a), (iv) Purchaser shall have no right to perform invasive or subsurface investigations or conduct any sampling or analysis of environmental media of the nature commonly referred to as a "Phase II Environmental Investigation," such as any soil or groundwater testing, (v) such access or related activities would not cause a violation of any agreement to which any Seller Group Member is a party, (vi) no Personal Information shall be disclosed or used other than in compliance with applicable privacy law and (vii) nothing herein shall require any member of the Seller Group or their representatives to furnish to Purchaser or provide Purchaser with access to information that (A) is subject to an attorney-client or an attorney work-product privilege, (B) legal counsel for the Seller Group reasonably concludes may give rise to antitrust or competition law issues or violate a protective order or otherwise may not be disclosed pursuant to Applicable Law (including any Public Health Measure) or (C) would cause significant competitive harm to the Seller Group if the Transactions are not consummated.
- (b) Notwithstanding anything to the contrary contained in this Agreement, from the Agreement Date until the Closing Date, Purchaser shall not, and shall cause its representatives not to, have any contact or discussions concerning any member of the Seller Group, the Business, the Transaction or any other matters with any lender, borrower, creditor, guarantor, business partner, bank, landlord, tenant, supplier, customer, employee, manager, franchisee, distributer, noteholder, independent contractor, consultant or other material business relation of any Seller Group Member, in each case, without the prior written consent of the Seller Access Contact (which consent may be withheld in the Seller's sole discretion and, if given, may be conditioned on the Seller Access Contact or his or her designee having the right to participate in any meeting or discussion); provided, however, that no such consent is required for Purchaser to exercise its rights or perform its obligations under Sections 7.8, 7.9, and 7.10, to contact BMS

in connection with the BMS License Agreement, to contact any of the counterparties to any Existing Manufacturing Contract, and Purchaser is hereby authorized to engage in such contact and discussions.

- (c) Any information provided to or obtained by Purchaser or its representatives, including pursuant to this Section 6.2 is confidential information and subject to the terms of, and the restrictions contained in, the Confidentiality Agreement. Purchaser agrees to be bound by and comply with the provisions set forth in the Confidentiality Agreement as if such provisions were set forth herein, and such provisions are hereby incorporated herein by reference. Effective upon (and only upon) the Closing, the Confidentiality Agreement shall automatically terminate and none of the parties thereto shall have any further Liability or obligation thereunder except with respect to any confidential information provided to or obtained by Purchaser or its representatives concerning the Seller Group, which information shall remain subject to the terms and conditions of the Confidentiality Agreement after the Closing Date. If this Agreement is terminated prior to Closing for any reason, the duration of the confidentiality of the Confidentiality Agreement shall be deemed extended, without any further action by the parties, for a period of time equal to the period of time elapsed between the date such Confidentiality Agreement was initially signed and the date of termination of this Agreement.
- Efforts to Consummate. Except as otherwise provided in this Agreement, each of the parties hereto agrees to use its commercially reasonable efforts to cause the Closing to occur as soon as possible after the Agreement Date, including satisfying the conditions precedent set forth in ARTICLE 8 applicable to such party including (a) defending against any Actions, judicial or administrative, challenging this Agreement or the consummation of the Transactions, (b) seeking to have any preliminary injunction, temporary restraining order, stay or other legal restraint or prohibition entered or imposed by any court or other Governmental Authority that is not yet final and non-appealable vacated or reversed, and (c) and executing any additional instruments reasonably requested by another party hereto (without cost or expense to the executing party) necessary to carry out the Transactions and to fully carry out the purposes of this Agreement; provided, however, that, for purposes of "commercially reasonable efforts" standard as required by this Section 6.3, Section 6.4 or Section 6.5, neither the Seller nor its Affiliates or representatives shall be required to offer or grant any accommodation or concession (financial or otherwise) to any third party or to otherwise expend any money or suffer any detriment, to expend any money to remedy any breach of any representation or warranty hereunder, to commence any Action, to waive or surrender any right, to modify any agreement (including any Assigned Contract) or to provide financing to Purchaser for the consummation of the Transactions.
- Notices and Consents. Reasonably promptly following the execution of this Agreement, the Seller will give, or cause to be given, applicable notices to third parties and thereafter will use commercially reasonable efforts (as limited by Section 6.3) to obtain the third-party consents set forth on Schedule 6.4; provided, however, that no representation, warranty, covenant or agreement of the Seller shall be breached or deemed breached, and no condition shall be deemed not satisfied, as a result of (a) the failure to obtain any such third-party consent (unless such consent is part of a closing condition of Seller), (b) any termination of a Contract as a result of the failure to obtain such third-party consent (unless such consent is part of a closing condition of Seller) or (c) any Action commenced or threatened by or on behalf of any Person arising out of or relating to the failure to obtain any such consent or any such termination; provided, further, that nothing in this Section 6.4 shall require the Seller to expend any money or grant any concessions to obtain any such third-party consent (unless Purchaser provides the funds for or reimburses the Seller for such payment).

6.5 **Regulatory Matters**.

(a) Purchaser and the Seller will establish a mutually acceptable and prompt communication and interaction process to ensure the orderly transfer of the Lambda IND and other similar

regulatory approval and authorization documents for jurisdictions outside of the United States. Promptly after Closing, the parties shall file with the FDA, and any other relevant Governmental Authority all information required in order to transfer the Lambda IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, including the information required pursuant to 21 C.F.R. § 314.72, or any successor regulation thereto, any authorization letters or notices, and letters of acceptance. Seller shall file the information required of a former owner, and Purchaser shall file the information required of a new owner, at each party's own expense. Both Purchaser and the Seller also agree to use all commercially reasonable efforts to take any actions required by the Governmental Authority or other government/health agencies to effect the transfer of the Lambda IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States from the Seller to Purchaser, and hereby further agree to cooperate with each other in order to effectuate the foregoing transfer of the Lambda IND. The parties agree to use all commercially reasonable efforts to complete the filing of the transfer of the Lambda IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States within ten (10) days from the Closing Date. The Seller may retain an archival copy of the Lambda IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States, including supplements and records that are required to be kept under 21 C.F.R. § 314.81 or other similar regulation.

- (b) From and after the Closing Date until the Seller is dissolved, the Seller shall cooperate with Purchaser in preparing, disclosing and providing any relevant records, reports, responses or any other documentation that are required to be made, maintained and reported pursuant to the Governmental Authority. The parties agree to use their commercially reasonable efforts to take any other actions required by the FDA or any other Governmental Authority to effect the transaction.
- (c) Until the completion of the transfer of the Lambda IND to Purchaser, the Seller shall take all reasonably necessary or advisable actions to maintain the relevant Lambda IND and other similar regulatory approval and authorization documents for jurisdictions outside of the United States.
- extent required by any Applicable Law or Action (including the Bankruptcy Cases), neither Purchaser nor the Seller shall, and Purchaser and the Seller shall cause their respective Affiliates and representatives not to, directly or indirectly, issue any press release or public announcement of any kind without the prior written consent of Purchaser and the Seller; *provided, however*, that the Seller and its Affiliates may make announcements from time to time to their respective employees, customers, suppliers, and other business relations and otherwise as the Seller may reasonably determine is necessary to comply with Applicable Law or the requirements of this Agreement or any other agreement to which any Seller Group Member or any such Affiliate is a party. Purchaser and the Seller shall cooperate in good faith to prepare a joint press release to be issued on the Closing Date, the terms of which shall be mutually agreed upon by the parties.
- 6.7 **Update of Schedules; Knowledge of Breach**. From time to time prior to the Closing, the Seller may supplement or amend the Schedules with respect to any matter first arising after the Agreement Date that would have been required to be set forth or described in such Schedules. Any such supplemental or amended disclosure shall not be deemed to have cured any such breach of representation or warranty for purposes of determining whether or not the conditions set forth in Section 8.2(a) have been satisfied. From and after the Closing, references to the Schedules shall be references to the Schedules as supplemented, modified and/or updated. If, prior to the Closing, Purchaser shall have reason to believe that any breach of a representation or warranty of the Seller has occurred (other than through notice from the Seller), Purchaser shall promptly so notify the Seller, in reasonable detail. Nothing in this Agreement, including this Section 6.7, shall imply that the Seller is making any representation or warranty as of any date other than the Closing Date (other than representations and warranties that are expressly made as of an earlier date).

ARTICLE 7. POST-CLOSING COVENANTS

7.1 Access to Information; Books and Records. From and after the Closing, Purchaser and its Affiliates shall (i) afford the Seller Group and their respective representatives reasonable access, during normal business hours, upon reasonable advance notice and under reasonable circumstances, to the books and records of Purchaser and the Business shall permit the Seller Group and their respective representatives to examine and copy such books and records to the extent reasonably requested by such party and (ii) cause their representatives to furnish all information reasonably requested by any member of the Seller Group or their representatives in connection with financial or regulatory reporting, audit, third party litigation, preparing or filing of any Tax Return or the defense of any Tax claim or assessment or any other business purpose; provided, however, that nothing in this Section 7.1 shall require Purchaser or its Affiliates to furnish to the Seller Group or their respective representatives any material that is subject to an attorneyclient or solicitor-client privilege or an attorney or solicitor work-product privilege or which may not be disclosed pursuant to Applicable Law. For a period of six (6) years following the Closing Date, or such longer period as may be required by Applicable Law or necessitated by applicable statutes of limitations, Purchaser shall, and shall cause its Affiliates to, maintain all such books and records in the jurisdiction in which such books and records were located prior to the Closing Date and shall not destroy, alter or otherwise dispose of any such books and records. On and after the end of such period, Purchaser shall, and shall cause its Affiliates to, provide the Seller with at least ten Business Days prior written notice before destroying, altering or otherwise disposing any such books and records, during which period the Seller may elect to take possession, at its own expense, of such books and records.

7.2 Post-Closing Receipt and Possession of Assets.

- (a) After the Closing Date, the Seller shall transfer promptly to Purchaser from time to time (but in any event on a monthly basis) any payments constituting Transferred Assets received by the Seller. After the Closing Date, Purchaser shall transfer promptly to the Seller, from time to time (but in any event on a monthly basis), any payments constituting Excluded Assets, including any accounts receivable constituting Excluded Assets, received by Purchaser after the Closing.
- (b) In the event that, after the Closing Date, Purchaser receives or otherwise is in possession of any other Excluded Asset, Purchaser shall promptly notify the Seller of its receipt or possession of such other Excluded Asset and transfer, at the Seller's expense, such Excluded Asset to the Seller. In the event that, after the Closing Date, the Seller receives or otherwise is in possession of any other Transferred Asset, the Seller shall promptly notify Purchaser of its receipt or possession of such other Transferred Asset and transfer, at Purchaser's expense (unless the Seller was required to transfer such Transferred Asset to Purchaser at Closing, in which case, and without limitation of any other remedies available to Purchaser, such transfer will be at the Seller's expense), such Transferred Asset to Purchaser.

7.3 Tax Matters.

- (a) All Taxes with respect to the income or operations of the Business or the ownership of the Transferred Assets that relate to any Straddle Period shall be apportioned between Seller and Purchaser as follows: (i) in the case of ad valorem or other property Taxes, on a per diem basis; and (ii) in the case of income, sales and use and withholding Taxes, employment Taxes, or other Taxes based on or measured by income, receipts or profits, as determined from the closing of the books and records of Seller and the Business at the close of business on the Closing Date.
- (b) After the Closing Date, Purchaser and Seller shall furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance (including access to books, records, work papers and Tax Returns for Pre-Closing Tax Periods) relating to the Business

or the Transferred Assets as is reasonably necessary for the preparation of any Tax Return, claim for refund or audit, and the prosecution or defense of any claim, suit or proceeding relating to any proposed Tax adjustment. Upon reasonable notice, Seller and Purchaser shall make its employees and facilities available on a mutually convenient basis to provide reasonable explanation of any documents or information provided hereunder. The other party hereto shall promptly (and in no event later than 30 days after receipt of the request) provide the requested information. The requesting party shall indemnify the other party for any out-of-pocket expenses incurred by such party in connection with providing any information or documentation pursuant to this Section 7.3(b). Any information obtained under this Section 7.3(b) shall be kept confidential, except as otherwise reasonably may be necessary in connection with the filing of Tax Returns or claims for refund or in conducting any Tax audit, dispute or contest.

7.4 Wrong Pockets.

- (a) <u>Assets</u>. If either Purchaser or Seller becomes aware that any of the Transferred Assets has not been transferred to Purchaser or that any of the Excluded Assets has been transferred to Purchaser, it shall promptly notify the other and the parties shall, as soon as reasonably practicable, ensure that such property is transferred, at the expense of Seller and with any necessary prior third party consent or approval, to (i) Purchaser, in the case of any Transferred Asset that was not transferred to Purchaser at the Closing; or (ii) Seller, in the case of any Excluded Asset that was transferred to Purchaser at the Closing.
- (b) <u>Payments</u>. If, on or after the Closing, either party shall receive any payments or other funds due to the other pursuant to the terms of this Agreement or any Related Document, then the party receiving such funds shall, within 30 days after receipt of such funds, forward such funds to the proper party. The parties acknowledge and agree there is no right of offset regarding such payments and a party may not withhold funds received from third parties for the account of the other party in the event there is a dispute regarding any other issue under this Agreement.
- 7.5 **Purchased Intellectual Property and Purchased Product Information**. Promptly following the Closing, at Purchaser's sole cost and expense, Seller shall take such further actions and execute such further documents as may be necessary or reasonably requested by Purchaser to effectuate, evidence and perfect the assignment and transfer of the Owned Intellectual Property Assets and Regulatory Approvals to Purchaser, including making such filings with any Governmental Authorities as may be required to transfer the Owned Intellectual Property Assets and Regulatory Approvals to Purchaser or to further the prosecution, issuance or maintenance of the Owned Intellectual Property Assets and Regulatory Approvals.
- 7.6 **Delivery of Transferred Materials; Transition Activities**. The Seller will, as soon as reasonably practicable after the Closing Date, (a) in any event within seven (7) Business Days after the Closing Date, effect the delivery of all Transferred Data, Transferred Regulatory Information, and Business Books and Records, and (b) within thirty (30) days after the Closing Date, (i) effect the delivery of all Inventory in accordance with Purchaser's instructions at Purchaser's cost and all other Transferred Materials not otherwise delivered to Purchaser, and (ii) use commercially reasonable efforts to perform, and cooperate with Purchaser regarding, the transition activities set forth on Schedule 7.6.
- 7.7 **License**. From and after the Closing, subject to the terms and conditions of this Agreement, Seller hereby grants to Purchaser, a perpetual, irrevocable, non-exclusive, sublicensable, royalty-free license under Seller's rights to (a) the Business Books and Records that are not Transferred Business Books and Records, (b) the Data that are not Transferred Data, and (c) the Regulatory Information that are not Transferred Regulatory Information, in each case, to conduct the Business.

7.8 **Pharmacovigilance**. From and after the Closing:

- (a) Within thirty (30) days after the Closing Date, Seller shall provide Purchaser with all adverse events ("AEs") for Licensed Products to the extent not previously provided to Purchaser that are in the possession of, owned by, or under the control (including via license) of Seller or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates (including by or on behalf of any contractors or other service providers acting on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level). In addition to the foregoing, Seller shall transfer to Purchaser in an agreed upon format, all relevant information (sufficient for Purchaser to comply with its obligations to Regulatory Authorities and investigators) regarding AEs that have been observed during any clinical trials conducted with Licensed Products prior to the Closing Date that are in the possession of, owned by, or under the control (including via license) of Seller or its Affiliates, including such foregoing items that are controlled by Seller or its Affiliates as a result of being held by or on behalf of the Seller or any of its Affiliates, directly or indirectly, at any level).
- (b) Seller shall (i) notify Purchaser of all information coming into its possession concerning AEs associated with commercial or clinical uses, studies, investigations or tests with Licensed Products in the Territory involving Licensed Products, and (ii) forward to Purchaser, completed AE case reports associated with commercial or clinical uses, studies, investigations or tests with Licensed Products, within five (5) Business Days for any death/fatal-life threatening assessed AEs or, within ten (10) Business Days for all other serious AEs, to assure such party remains in compliance with investigator notifications in its respective field. Such AE information should be sent to Purchaser via email at the email address notified by Purchaser in writing, as applicable. Within thirty (30) days of the Closing, the parties shall enter into a separate written pharmacovigilance agreement with respect to the Licensed Products to enable Purchaser to fulfill its regulatory reporting obligations under Applicable Laws.
- (c) Without limiting Purchaser's rights or obligations in the foregoing, at the request of Purchaser, Seller shall provide Purchaser with all materials, data, information or other documents necessary in form and substance to allow Purchaser to comply with its obligations under the BMS License Agreement.
- Prior to the Lambda IND Transfer Date, Seller will comply with its (d) obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lambda INDs. After the Lambda IND Transfer Date, Purchaser will comply with its obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lambda INDs. The Seller Group shall, at no cost to Purchaser, (i) maintain and administer the Global Safety Databases itself and through its Third Party service provider under the Global Safety Database Contracts, (ii) provide Purchaser the pharmacovigilance services set forth on Schedule 7.8(d)(1) from the Closing Date until the Lambda IND Transfer Date, and (iii) provide Purchaser the pharmacovigilance services set forth on Schedule 7.8(d)(2) from the Lambda IND Transfer Date until the date that is ninety (90) days after the Closing Date, which services in clauses (ii) and (iii) will be provided by Seller as Purchaser's service provider and under the reasonable direction and supervision of Purchaser in order to assist Purchaser as reasonably necessary to comply with its obligations under Applicable Laws, including drug surveillance, safety data reporting, and other required pharmacovigilance activities, as the holder of the Lambda INDs. Notwithstanding the foregoing, in each case of the foregoing clauses (i), (ii), and (iii), the Seller Group will no longer be required to perform such activities following the earlier of (A) the Plan Consummation Date and (B) the date on which ownership or administration of the Global Safety Databases has been fully transferred and transitioned to Purchaser and/or a Third Party service provider designated by Purchaser (such earlier date,

- the "PV Services Stop Date"). Until the PV Services Stop Date, the Seller Group shall not without Purchaser's prior written consent, (y) sell, assign, license, transfer, convey, deliver or otherwise divest its interests in any of the Global Safety Database Contracts to a Third Party, or amend or modify any of the Global Safety Database Contracts, in each case, in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's ability to access, receive, or be provided data from the Global Safety Databases, Purchaser's rights or obligations under this Agreement, or Purchaser's ability to Develop or Commercialize any Licensed Products, or (z) undertake any action that would constitute a material breach of, or reduce the Seller Group's rights under, any Global Safety Database Contract.
- 7.9 **Confirmation Letters.** Within seven (7) Business Days after Purchaser's written request provided to Seller after the Closing, Seller and its Affiliates shall provide a letter of confirmation to Purchaser for delivery by Purchaser to any Person that possesses or otherwise holds any Transferred Assets that confirms that Purchaser acquired and is the exclusive owner of the relevant Transferred Assets held by such Person in a form reasonably acceptable to Purchaser. Purchaser will have the right to provide any such letter of confirmation to any Person.

ARTICLE 8. CONDITIONS PRECEDENT

- 8.1 Conditions to Each Party's Obligation. The respective obligations of the parties hereto to effect the Transactions are subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller and Purchaser), at or prior to the Closing, of the following conditions:
- (a) **No Injunctions or Restraints**. No Order or Applicable Law preventing the consummation of the Transactions shall be in effect.
- (b) **Sale Order**. The Bankruptcy Court shall have entered the Sale Order and such Sale Order shall be a Final Order (unless such Final Order requirement is waived by the Seller and Purchaser in their respective sole discretion).
- 8.2 Conditions to Obligations of Purchaser. The obligations of Purchaser to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by Purchaser), at or prior to the Closing, of the following conditions:
- (a) Representations and Warranties. Each of the representations and warranties of the Seller set forth in ARTICLE 3 shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality", "Material Adverse Effect" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not have, individually or in the aggregate, a Material Adverse Effect.
- (b) **Performance of Covenants and Obligations**. The Seller shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated expressly by the terms of this Agreement or caused by the Transactions.
- (c) **Effective Assignment of Contracts**. The Bankruptcy Court shall have entered an order (which may be the Sale Order) approving the assumption and assignment to Purchaser of the Assigned Contracts, which order shall be a Final Order and in full force and effect and in a form and substance satisfactory to Purchaser.

- (d) **Closing Deliverables**. The Seller shall have delivered to Purchaser the closing deliveries required to be delivered by the Seller pursuant to Section 2.8(a), Section 2.8(b), Section 2.8(c), Section 2.8(e), and Section 2.8(f).
- 8.3 **Conditions to Obligations of the Seller.** The obligation of the Seller to effect the Transactions is subject to the satisfaction (or, to the extent permitted by Applicable Law, waiver by the Seller), at or prior to the Closing, of the following conditions:
- (a) Representations and Warranties. Each of the representations and warranties of Purchaser set forth in ARTICLE 4 shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or words of similar import set forth therein) as of the Closing as though made at and as of such time (other than such representations and warranties as are made as of an earlier date, which shall be so true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not, individually or in the aggregate, (i) have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or (ii) otherwise prevent, hinder or delay the consummation of the Transactions.
- (b) **Performance of Covenants and Obligations of Purchaser**. Purchaser shall have performed or complied in all material respects with all obligations and covenants required to have been performed or complied with by it under this Agreement at or prior to the Closing, except to the extent of changes or developments contemplated by the terms of this Agreement or caused by the Transactions.
- (c) Closing Deliverables. Purchaser shall have delivered to the Seller the closing deliveries required to be delivered by Purchaser pursuant to Section 2.8(a), Section 2.8(b), Section 2.8(c), and Section 2.8(d).
- 8.4 **Waiver of Condition; Frustration of Conditions.** All conditions to the Closing shall be deemed to have been satisfied or waived from and after the Closing. Neither Purchaser nor the Seller may rely on the failure of any condition set forth in this ARTICLE 8, as applicable, to be satisfied if such failure was caused by such party's failure to use, as required by this Agreement, its reasonable best efforts to consummate the Transactions.
- 8.5 **Delivery of a Notice of Readiness to Close.** At any time after the Seller's satisfaction of its conditions to Closing in accordance with the terms of Section 8.1 and Section 8.3 of this Agreement, the Seller may deliver a notice to Purchaser (a "**Notice of Readiness to Close**"). Purchaser shall have three (3) Business Days from delivery of a Notice of Readiness to Close to satisfy its conditions to Closing in accordance with the terms of Section 8.1 and Section 8.2 of this Agreement and consummate the Transactions. If Purchaser does not satisfy its conditions to Closing and consummate the Transaction within three (3) Business Days, Purchaser shall forfeit the entire Deposit Escrow Amount to the Seller.

ARTICLE 9. TERMINATION

- 9.1 **Events of Termination**. Notwithstanding anything to the contrary, this Agreement may be terminated and the Transactions may be abandoned at any time prior to the Closing:
 - (a) by mutual written consent of Purchaser and the Seller;
- (b) automatically, upon (i) the consummation of a sale or other disposition of all or substantially all of the Transferred Assets to a Person other than Purchaser (each, an "Alternate Transaction"), (ii) if, at close of the Auction, Purchaser's bid has not been selected as either the winning

bid or the Back-Up Bid or (iii) if, at the close of the Auction, Purchaser's bid was selected as the Back-Up Bid, upon the consummation of a Competing Bid or Alternate Transaction;

- (c) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if the Bankruptcy Case is dismissed or converted to a case under chapter 7 of the Bankruptcy Code;
- (d) by Purchaser or the Seller by written notice to Purchaser or the Seller from the other, if Purchaser is not selected as having the winning bid or Back-Up Bid at Auction, if any;
- (e) by Purchaser if the Seller (i) withdraws the motion for the Sale Order, or publicly announces its intention to withdraw such motion, (ii) moves to voluntarily dismiss the Bankruptcy Cases, (iii) moves for conversion of the Bankruptcy Cases to Chapter 7 of the Bankruptcy Code, or (iv) moves for appointment of an examiner with expanded powers pursuant to Section 1104 of the Bankruptcy Code or a trustee in the Bankruptcy Cases;
- (f) by Purchaser, by written notice from Purchaser to the Seller, if there has been a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement, such that the conditions in Section 8.1 or Section 8.2 are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by the Seller prior to the earlier of (i) twenty (20) Business Days after receipt of written notice from Purchaser requesting such breach be cured or (ii) the Outside Date; *provided, however*, that the right to terminate this Agreement pursuant to this Section 9.1(f) shall not be available to Purchaser if the failure of Purchaser to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement;
- by the Seller, by written notice from the Seller to Purchaser, if there has been a breach or inaccuracy of a covenant, representation or warranty made by Purchaser in this Agreement, such that the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied and which breach is incapable of being cured or, if capable of being cured, has not been cured by Purchaser prior to the earlier of (i) 20 Business Days after receipt of written notice from the Seller requesting such breach be cured or (ii) the Outside Date; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 9.1(g) shall not be available to the Seller if the failure of the Seller to fulfill any of its obligations under this Agreement has been the primary cause of, or resulted in, such breach, or if the conditions in Section 8.1 or Section 8.3 are not capable of being satisfied because there is then a breach or inaccuracy of a covenant, representation or warranty made by the Seller in this Agreement;
- (h) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if any Governmental Authority of competent jurisdiction shall have issued an Order, enacted any Applicable Law or taken any other action restraining, enjoining or otherwise prohibiting the consummation of the Transactions and, in the case of Orders and other actions, such Order or other action shall have become Final Orders; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 9.1(h) shall not be available to the party seeking to terminate if any action of such party or any failure of such party to act has contributed to such Order or other action and such action or failure constitutes a breach of this Agreement;
- (i) by Purchaser or the Seller, by written notice from Purchaser or the Seller to the other, if the Closing has not occurred on or prior to October 5, 2024 (the "Outside Date"); provided, however, that the party exercising the right to terminate this Agreement pursuant to this Section 9.1(i) shall not have been responsible for such failure of the Closing to occur through a breach or inaccuracy of a

covenant, representation or warranty contained in this Agreement (it being understood, acknowledged, and agreed that if Seller is unable to provide any required Closing deliverable of Seller, then Seller shall be deemed to have been responsible for such failure of the Closing for purposes of this Section 9.1(i)); or

(j) by Purchaser by written notice to the Seller if the Bankruptcy Court does not approve the Bid Procedures Order without any material modifications (other than such modifications reasonably acceptable to Purchaser) to the protections to Purchaser set forth in Section 9.3(a) and Section 9.3(b).

9.2 **Effect of Termination**.

- (a) In the event that this Agreement shall be terminated pursuant to Section 9.1, (a) Purchaser and its representatives shall promptly return all documents, work papers and other materials of the Seller including any confidential information and (b) all further obligations of the parties hereto under this Agreement shall terminate without further Liability or obligation to the other parties hereto; *provided, however*, that, notwithstanding the foregoing, the Liabilities and obligations under (i) the Confidentiality Agreement, and (ii) Section 2.9(c), Section 6.2(c), this Section 9.2, Section 9.3, and ARTICLE 10 shall continue in full force and effect.
- (b) Notwithstanding anything to the contrary in this Agreement, in the event of valid termination of this Agreement pursuant to Section 9.1, (i) Seller's Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement shall be capped at an amount equal to the Deposit Escrow Amount, and Purchaser shall be entitled to all remedies available at law or in equity, including payment of the Expense Reimbursement pursuant to Section 9.3, and (ii) Purchaser's Liability hereunder for any and all breaches of this Agreement prior to such termination of this Agreement shall be capped at an amount equal to the Deposit Escrow Amount and Seller shall be entitled to all remedies available at law or in equity, including payment of the Deposit Escrow Amount pursuant to Section 2.9(c).

9.3 Expense Reimbursement.

- Subject to limitations set forth in the Bid Procedures Order, in consideration of Purchaser having expended considerable time and expense in connection with this Agreement and the negotiation thereof, and the identification and quantification of assets to be included in the Transferred Assets, if this Agreement is terminated in accordance with the terms set forth in any of Sections 9.1(b)-(f) or 9.1(h)-(i), then the Seller shall pay to Purchaser in cash not later than two (2) Business Days following receipt of documentation supporting the request for reimbursement of costs, fees and expenses, the Expense Reimbursement, in an amount not to exceed \$50,000, by wire transfer of immediately available funds to an account specified by Purchaser to the Seller in writing; provided that Purchaser shall not be entitled to the fee described in this Section 9.3(a) to the extent (i) Purchaser is in material breach of this Agreement at the time this Agreement is terminated pursuant to Section 9.1(b)-(f) or 9.1(h)-(i) and (ii) Seller has provided notice of such material breach to Purchaser and such material breach remained uncured for more than five (5) Business Days after Purchaser's receipt of such notice. The Seller's obligation to pay the Expense Reimbursement pursuant to this Section 9.3(a) shall survive termination of this Agreement and shall constitute an administrative expense of Seller under section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind, including those specified in section 503(b) or 507(b) of the Bankruptcy Code.
- (b) The Seller agrees and acknowledges that Purchaser's due diligence, efforts, negotiation, and execution of this Agreement have involved substantial investment of management time and have required significant commitment of financial, legal, and other resources by Purchaser, and that such due diligence, efforts, negotiation, and execution have provided value to the Seller and, in the Seller's reasonable business judgment, is necessary for the preservation of the value of the Seller's estate.

The Seller further agrees and acknowledges that the Expense Reimbursement is not a penalty, but rather represent liquidated damages that are reasonable in relation to Purchaser's efforts, Purchaser's lost opportunities from pursuing the Transactions, and the magnitude of the Transactions. The provision of the Expense Reimbursement is an integral part of this Agreement, without which Purchaser would not have entered into this Agreement.

ARTICLE 10. GENERAL PROVISIONS

- 10.1 Survival of Representations, Warranties and Covenants. All covenants and agreements contained in this Agreement that by their term are to be performed in whole or in part, or which prohibit actions, subsequent to Closing shall, solely to the extent such covenants and agreements are to be performed, or prohibit actions, subsequent to Closing, survive the Closing in accordance with their terms until fully performed or satisfied. All other covenants and agreements contained herein, and all representations and warranties contained herein or in any certificated deliveries hereunder shall not survive Closing and shall therefor terminate, including any Action for damages in respect of any breach or inaccuracy thereof. Notwithstanding the foregoing, the provisions of Section 2.9(c), Section 6.2, Section 9.2, this ARTICLE 10 and the Confidentiality Agreement shall survive the Closing.
- Entire Agreement. This Agreement, including the Exhibits and Schedules hereto, the Confidentiality Agreement and the Related Documents, contain the entire understanding of the parties hereto with respect to the subject matter contained herein and therein. This Agreement supersedes all prior and contemporaneous agreements, arrangements, contracts, discussions, negotiations, undertakings and understandings (including any letters of intent or term sheets), whether written or oral, among the parties with respect to such subject matter (other than, for the avoidance of doubt, the Confidentiality Agreement and the Related Documents) or any prior course of dealings. The parties hereto have voluntarily agreed to define their rights, Liabilities and obligations respecting the Transactions exclusively in contract pursuant to the express terms and conditions of this Agreement, the Confidentiality Agreement and the Related Documents, and the parties hereto expressly disclaim that they are owed any duties or entitled to any remedies not expressly set forth in this Agreement, the Confidentiality Agreement and the Related Documents. Furthermore, the parties each hereby acknowledge that this Agreement, the Confidentiality Agreement and the Related Documents embody the justifiable expectations of sophisticated parties derived from arm's-length negotiations, and all parties to this Agreement, the Confidentiality Agreement and the Related Documents specifically acknowledge that no party has any special relationship with another party that would justify any expectation beyond that of an ordinary purchaser and an ordinary seller in an arm'slength transaction. The sole and exclusive remedies for any Related Claims shall be those remedies available at law or in equity for breach of contract only (as such contractual remedies have been further limited or excluded pursuant to the express terms of this Agreement); and the parties hereby agree that neither party hereto shall have any remedies or cause of action (whether in contract or in tort or otherwise) of any statements, communications, disclosures, failures to disclose, representations or warranties not set forth in this Agreement.
- 10.3 Amendment; No Waiver. This Agreement and the Related Documents may be amended, supplemented or changed, and any provision hereof or thereof can be waived, only by a written instrument making specific reference to this Agreement (and, if applicable, the Related Documents) executed by the party against whom enforcement of any such amendment, supplement, modification or waiver is sought. The waiver by any party of a breach of any provision of this Agreement or the Related Documents shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any party to exercise, and no delay in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall a single or partial exercise of such

right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy.

- Severability; Specific Versus General Provisions. Whenever possible, each provision of this Agreement and the Related Documents shall be interpreted in such manner as to be effective and valid under Applicable Law, but if any term or other provision of this Agreement or the Related Documents is invalid, illegal, or incapable of being enforced by any Applicable Law or public policy, all other terms or provisions of this Agreement and the Related Documents shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, in whole or in part, such term or provision is hereby deemed modified to give effect to the original written intent of the parties to the greatest extent consistent with being valid and enforceable under Applicable Law. No party hereto shall assert, and each party shall cause its respective Affiliates or related parties not to assert, that this Agreement or any part hereof is invalid, illegal or unenforceable. Notwithstanding anything to the contrary, to the extent that a representation, warranty, covenant or agreement of the Seller contained in this Agreement or the Schedules (each, a "Provision") addresses a particular issue with specificity (a "Specific Provision"), and no breach by the Seller exists under such Specific Provision, the Seller shall not be deemed to be in breach of any other Provision (with respect to such issue) that addresses such issue with less specificity than the Specific Provision, and if such Specific Provision is qualified or limited by the Seller's Knowledge, or in any other manner, no other Provision shall supersede or limit such qualification in any manner.
- 10.5 **Expenses and Obligations**. Except as otherwise provided in this Agreement, all costs and expenses incurred by the parties hereto in connection with the Transactions, including the costs, expenses and disbursements of counsel and accountants, shall be borne solely and entirely by the party that has incurred such expenses; *provided*, *however*, that Purchaser shall pay, or promptly reimburse the Seller for, any filing fees which relate to any required governmental filing or notification and Purchaser shall pay any Transfer Taxes.
- 10.6 **Notices**. All notices, consents, waivers, and other communications under this Agreement or the Related Documents shall be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery, (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service, (c) if delivered by electronic mail (unless the sender receives an automated message that the email has not been delivered) on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day) and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or email addresses set forth below (or to such other addresses as a party may designate by notice to the other parties in accordance with this Section 10.6):

If to Purchaser:

Eiger InnoTherapeutics, Inc. 2061 Webster Street Palo Alto, CA 94301 Attn: Dr. Jeffrey Glenn Email: jsglenn@stanford.edu

with a copy to (which will not constitute notice):

Goodwin Procter LLP The New York Times Building 620 Eighth Avenue New York, New York 10018

Attn: Kizzy Jarashow, Maggie Wong, and David Chen

email: kjarashow@goodwinlaw.com; mwong@goodwinlaw.com; dchen@goodwinlaw.com

If to the Seller:

Eiger BioPharmaceuticals, Inc. 2100 Ross Avenue Dallas, Texas 75201 Attn: David Apelian, Chief Executive Officer

Email: dapelian@eigerbio.com

with a copy to (which will not constitute notice):

Sidley Austin LLP 2021 McKinney Ave., Suite 2000 Dallas, TX 75201

Attention: Thomas R. Califano, William E. Curtin and Anne G. Wallice

Email: tom.califano@sidley.com, wcurtin@sidley.com, and anne.wallice@sidley.com

- 10.7 **Counterparts**. This Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by electronic transmission in .PDF format, or other agreed format shall be sufficient to bind the parties to the terms and conditions of this Agreement. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement or any Related Document, shall be disregarded in determining the party's intent or the effectiveness of such signature.
- 10.8 **Governing Law.** This Agreement, the Related Documents and all Related Claims shall be governed by the internal laws of the State of Delaware (including its statute of limitations), without giving effect to any choice or conflict of law principles or rules that would cause the application of the Applicable Laws of any other jurisdiction.

10.9 Submission to Jurisdiction; Consent to Service of Process.

- Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to interpret and/or enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any Related Document, any breach or default hereunder or thereunder, or the Transactions, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 10.6; provided, however, that if the Bankruptcy Cases have closed, the parties agree to irrevocably submit to the exclusive jurisdiction of the United States District Court for the Northern District of Texas over all Related Claims, and each party hereto hereby irrevocably agrees that all Related Claims may be heard and determined in such courts. The parties hereto hereby irrevocably and unconditionally waive, to the fullest extent permitted by Applicable Law, any objection which they may now or hereafter have to the laying of venue of any such Related Claim brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Applicable Law.
- (b) Each of the parties hereto hereby consents to process being served by any party to this Agreement in any Related Claim by the delivery of a copy thereof in accordance with the provisions of Section 10.6 (other than by email) along with a notification that service of process is being

served in conformance with this Section 10.9(b). Nothing in this Agreement will affect the right of any party to serve process in any other manner permitted by Applicable Law.

- 10.10 Waiver of Jury Trial. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY EXPRESSLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY PROCEEDING OR RELATED CLAIM BROUGHT BY OR AGAINST IT, DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATED DOCUMENTS OR ANY RELATED CLAIMS.
- 10.11 **Rights Cumulative**. All rights and remedies of each of the parties under this Agreement and the Related Documents will be cumulative, and the exercise of one or more rights or remedies will not preclude the exercise of any other right or remedy available under this Agreement, the Related Documents or Applicable Law.
- 10.12 Assignment. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors by operation of law and permitted assigns of the parties hereto. No assignment of this Agreement or any of the rights, interests or obligations under this Agreement may be made by any party hereto at any time, whether or not by operation of law, without the prior written consent of the Seller and Purchaser, and any attempted assignment without the required consent shall be void; *provided, however*, that (a) Purchaser may assign (i) any of its rights or delegate any of its duties under this Agreement to any of its Affiliates, and (ii) its rights, but not its duties, under this Agreement to any of its financing sources and (b), the Seller may assign any of its rights or delegate any of its duties under this Agreement (i) to any of its Affiliates, (ii) to any creditor or group of creditors pursuant to an order of the Bankruptcy Court entered in the Bankruptcy Cases, including Seller's rights to payment hereunder and rights and ability to enforce the terms of this Agreement and (iii) for collateral security purposes to any lender of the Seller or its Affiliates; *provided, further, however*, that, in each case, such assignment shall not release Purchaser from its obligations under this Agreement and the Seller shall have no obligation to pursue remedies against any assignee of Purchaser before proceeding against Purchaser for any breach of Purchaser's obligations hereunder.
- **Specific Enforcement; Remedies.** The parties hereto agree that irreparable damage (for which monetary relief, even if available, would not be an adequate remedy) would occur in the event that any of the provisions of this Agreement were not performed by the parties hereto in accordance with their specific terms or were otherwise breached. It is accordingly agreed that (i) Purchaser, on the one hand, and the Seller, on the other hand, shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of competent jurisdiction without proof of damages or otherwise and that this shall include the right of the Seller to cause Purchaser to fully perform the terms of this Agreement to the fullest extent permissible pursuant to this Agreement and Applicable Laws and to thereafter cause this Agreement and the Transactions to be consummated on the terms and subject to the conditions thereto set forth in this Agreement, and (ii) the right of specific performance and other equitable relief is an integral part of the Transactions and without that right, neither the Seller nor Purchaser would have entered into this Agreement. Remedies shall be cumulative and not exclusive and shall be in addition to any other remedies which any party may have under this Agreement. Each of the parties hereto hereby (A) waives any defenses in any action for specific performance, including the defense that a remedy at law would be adequate, (B) waives any requirement under any Applicable Law to post a bond or other security as a prerequisite to obtaining equitable relief and (C) agrees not to assert that a remedy of specific performance or other equitable relief is unenforceable, invalid, contrary to law or inequitable for any reason, and not to assert that a remedy of monetary damages would provide an adequate remedy or that the parties otherwise have an adequate remedy at law. Notwithstanding anything to the

contrary, in no event shall this Section 10.13 be used, alone or together with any other provision of this Agreement, to require the Seller to remedy any breach of any representation or warranty of the Seller.

- 10.14 **Third-Party Beneficiaries**. Except as set forth in ARTICLE 2 (with respect to the Seller), Section 10.15 (with respect to the Nonparty Affiliates), Section 10.16 (with respect to the released parties identified therein), Section 10.17 (with respect to the Sellers' Group Members) and the next sentence, nothing in this Agreement, express or implied, is intended to confer upon any Person other than the parties hereto any rights or remedies of any nature whatsoever under or by reason of this Agreement. From and after the Closing, all of the Persons identified as third-party beneficiaries in the first sentence of this Section 10.14 shall be entitled to enforce such provisions and to avail themselves of the benefits of any remedy for any breach of such provisions, all to the same extent as if such Persons were parties to this Agreement. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance with this Agreement without notice or Liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any party hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the Agreement Date or as of any other date.
- 10.15 No Personal Liability of Directors, Officers and Owners. All Related Claims may be made only against (and are those solely of) the entities that are expressly identified as parties in the preamble to this Agreement (the "Contracting Parties"). No Person who is not a Contracting Party, including any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any Contracting Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, representative or assignee of, or any financial advisor or lender to, any of the foregoing (collectively, "Nonparty Affiliates"), shall have any Liability pursuant to any Related Claim; and, to the maximum extent permitted by Applicable Law, each Contracting Party hereby waives and releases all such Liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by Applicable Law, (a) each Contracting Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at Applicable Law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose Liability of a Contracting Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to this Agreement or the Related Documents.

10.16 General Release.

(a) Effective as of the Closing, the Seller, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "Seller Releasing Party"), hereby fully, irrevocably and unconditionally releases and forever discharges Purchaser and its respective past and present directors, managers, officers, employees, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, any and all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Seller Releasing Party has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the

Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Seller Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.

- (b) Effective as of the Closing, Purchaser, on behalf of itself, its Affiliates and each of their respective successors and assigns (each of the foregoing, a "Purchaser Releasing Party"), hereby fully, irrevocably and unconditionally releases and forever discharges the Seller, the Seller's Affiliates and its and their respective past and present directors, managers, officers, agents, stockholders, members, representatives and Affiliates from and against, and covenants that it will not (directly or indirectly) assert any claim or proceeding of any kind before any Governmental Authority based upon, all claims, Actions, causes of action, suits, rights, agreements, Liabilities and demands whatsoever and all consequences thereof, known or unknown, actual or potential, suspected or unsuspected, fixed or contingent, both in law and in equity, whether existing as of the Closing or arising thereafter, that a Purchaser Releasing Party has or may have, now or in the future, arising out of, relating to, or resulting from any act or omission, error, negligence, breach of contract, tort, violation of law, matter or cause whatsoever from the beginning of time to the Closing Date. The foregoing sentence shall not be deemed to be a release or waiver by a Purchaser Releasing Party of any Action it may have under this Agreement or any of the other Related Documents.
- 10.17 **Legal Representation**. Purchaser and the Seller acknowledge and agree that the Law Firm has represented the Seller Group in connection with the negotiation, preparation, execution, delivery and performance of this Agreement and the Related Documents and the consummation of the Transactions, and that the Seller, its Affiliates and its partners, officers, directors and representatives (the "Seller Group **Members**") have a reasonable expectation that the Law Firm will represent them in connection with any Action involving any Seller Group Member, on the one hand, and Purchaser or any of its Affiliates and representatives (the "Purchaser Group Members"), on the other hand, arising under this Agreement, the Related Documents or the Transactions. Purchaser hereby, on behalf of itself and the other Purchaser Group Members, irrevocably: (a) acknowledges and agrees that any attorney-client privilege, solicitor-client privilege, work product or other attorney-client or solicitor-client confidential information ("Attorney-Client Information") arising from communications prior to the Closing between the Seller (including any one or more officers, directors or stockholders of such Seller), on the one hand, and the Law Firm, on the other hand, is not included in the property, rights, privileges, powers, franchises and other interests that are possessed by or vested in the Business or the Transferred Assets, that any such Attorney-Client Information shall be deemed property of, and controlled solely by, such Seller for the benefit and on behalf of the Seller Group Members and, upon request, convey and transfer any Attorney-Client Information to the Seller; (b) acknowledge and agree that the Seller Group Members shall have the right to retain, or cause the Law Firm to retain, any such documentation or information in the possession of the Law Firm or such Seller Group Members at the Closing; (c) agree not to access, retain or use any documentation or information constituting Attorney-Client Information and that no Purchaser Group Member shall have any right to waive any attorney-client privilege or other right to confidentiality with respect to such Attorney-Client Information; (d) disclaim the right to assert a waiver by any Seller Group Member with regard to the attorney-client privilege, solicitor-client privilege or other right to confidentiality with respect to such Attorney-Client Information solely due to the fact that such documentation or information is physically in the possession of Purchaser after the Closing; (e) consent to the Law Firm's representation after the Closing of any Seller Group Member in any Action that may relate to a Purchaser Group Member or the Transactions and consent to and waive any conflict of interest arising therefrom without the need for any future waiver or consent; and (f) consent to the disclosure by the Law Firm to any Seller Group Member of any documentation or information obtained by the Law Firm during the course of its representation of Seller or any Affiliate prior to the Closing, whether related to this Agreement, the Related Documents, the Transactions or otherwise, whether or not such disclosure is made prior to or after the Closing and whether or not the documentation or information disclosed is subject to any attorney-client privilege, solicitor-client privilege or confidentiality obligation to any Seller Group Member, any Affiliate of the Seller or any other Person. In the event that any Action arises after the Closing between any Purchaser Group Member and a Person other than a Seller Group Member, such Purchaser Group Member shall not disclose any

Case 24-80040-sgj11 Doc **89**8-6 Filed 0**8**/25/2**5** Entered 0**8**/25/2**5 20**:**22**:25 Desc Exhibit & Page 269 of 205

documentation or information that is subject to an attorney-client privilege or other rights of confidentiality referenced in this Section 10.17 without the prior written consent of the applicable Seller; *provided, however*, that if such Purchaser Group Member is required by judicial order or other legal process to make such disclosure, such Purchaser Group Member shall promptly notify the applicable Seller in writing of such requirement (without making disclosure) and shall provide such Seller with such cooperation and assistance as shall be necessary to enable such Seller to prevent disclosure by reason of such attorney-client privilege, solicitor-client privilege or other rights of confidentiality. This Section 10.17 is for the benefit of the Seller Group Members and such Persons are intended third-party beneficiaries of this Section 10.17.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

Case 24-80040-sgj11 Doc 898-6 Filed 08/25/25 Entered 08/25/25 20:22:25 Desc Exhibit & Page 200 of 205

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

PURCHASER:

EIGER INNOTHERAPEUTICS, INC.

DocuSigned by:

Name: Dr. Jeffrey Glenn

Title: Founding President

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

SELLER:

EIGER BIOPHARMACEUTICALS, INC.

DocuSigned by:

By:____{0F1FCA681024459...}

Name: David Apelian

Title: Chief Executive Officer

Exhibit 7

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825)

William E. Curtin (admitted pro hac vice)

Anne G. Wallice (admitted pro hac vice)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Attorneys for the Debtors and Debtors in Possession

SIDLEY AUSTIN LLP

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF HEARING FOR AUGUST 20, 2024 SETTING

PLEASE TAKE NOTICE that a hearing has been scheduled for August 20, 2024 at 9:30 a.m. (prevailing Central Time) before the Honorable Chief Judge Stacey G. C. Jernigan to consider the following matters:

- 1. Debtors' Motion for Entry of an Order Estimating Claim of Innovatus Life Sciences Lending Fund I, LP for the Purposes of Establishing Sufficient Reserves to Unimpair Claim [Docket No. 488] (the "Estimation Motion").
- 2. Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Ave., Dallas, Texas 75201.

Casse2248800409sggj111 DDocc85897 Filitelc084125225 Hintereclc084125225191347295 DDess C MainEloloicutn7entPage agef 3 of 4

Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Docket No. 490] (the "Lonafarnib/Lambda Sale Motion").

PLEASE TAKE FURTHER NOTICE that, parties may participate in the hearing either in person or by an audio and video connection.

PLEASE TAKE FURTHER NOTICE that, parties who wish to appear and participate in the Hearing by videoconference may do so via WebEx at the following link:

LINK: https://us-courts.webex.com/meet/jerniga

Desktop or Mobile Application: Meeting Number 2304-154-2638

PLEASE TAKE FURTHER NOTICE that, the WebEx hearing Instructions may be obtained from Judge Jernigan's hearing/calendar site: https://www.txnb.uscourts.gov/judgesinfo/hearing-dates/chief-judge-jernigans-hearing-dates. Parties should review the Webex instructions prior to the hearing.

• **Dial-In:** 650.479.3207

• **Access Code:** 2304-154-2638

PLEASE TAKE FURTHER NOTICE that, hearing appearances must be made electronically in advance of electronic hearings. To make your appearance, click the "Electronic Appearance" link on Judge Jernigan's homepage: https://www.txnb.uscourts.gov/content/chiefjudge-stacey-g-c-jernigan. Select the case name, complete the required fields and click "Submit" to complete your appearance.

PLEASE TAKE FURTHER NOTICE that, copies of the foregoing pleadings may be obtained (i) at the website established by the Debtors' notice and claims agent, Verita Global f/k/a Kurtzman Carson Consultants LLC at https://veritaglobal.net/Eiger, (ii) from the Court's website http://www.txnb.uscourts.gov via ECF/Pacer, or (iii) upon request to the undersigned.

Casse2248800409sgijj11 DDooc85/897 Filitelc084125225 Hintereclc084125225191347295 DDess MainEloloichitm7enPagle 4ggef 3 of 4

Dated: August 14, 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Casse2248800440ssjýj.11. DDocc85897 Ffilitelc084125225 Hintereclc0841252251191347295 DDessc MainEloloichitm7en1Pagle5ggf 45 of 4

Certificate of Service

I certify that on August 14, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ Thomas R. Califano

Thomas R. Califano

Exhibit 8



CLERK, U.S. BANKRUPTCY COURT NORTHERN DISTRICT OF TEXAS

ENTERED

THE COURT'S DOCKET

The following constitutes the ruling of the court and has the force and effect therein described.

Signed August 21, 2024

United States Bankruptcy Judge

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

REVISED ORDER (I) AUTHORIZING
THE SALE OF THE LONAFARNIB AND LAMBDA ASSETS FREE
AND CLEAR OF LIENS, CLAIMS, ENCUMBRANCES, AND OTHER INTERESTS,
(II) AUTHORIZING THE ASSUMPTION AND ASSIGNMENT OF EXECUTORY
CONTRACTS AND UNEXPIRED LEASES, (III) GRANTING THE PURCHASER THE
PROTECTIONS AFFORDED TO A GOOD FAITH PURCHASER, (IV) APPROVING
PURCHASER PROTECTIONS IN CONNECTION WITH THE SALE OF THE
LONAFARNIB AND LAMBDA ASSETS, AND (V) GRANTING RELATED RELIEF

_

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A), and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

Casse2248800449ssgij111 DDooc85588 Filitelc084225225 Hintercelc084225225161447525 DDess c MairEl2hoibuir6en1Pagle 3cpef 32of 51

Upon consideration of the Motion² of the debtors and debtors in possession in the abovecaptioned chapter 11 cases (collectively, the "Debtors") for entry of an order (this "Revised Lonafarnib/Lambda Sale Order"), pursuant to sections 105(a), 363, and 365 of title 11 of the United States Code (the "Bankruptcy Code") and Rules 2002, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), Rule 9013-1 of the Bankruptcy Local Rules for the Northern District of Texas (the "Bankruptcy Local Rules"), and Section E of the *Procedures* for Complex Chapter 11 Cases in the Northern District of Texas (the "Complex Case Procedures"), authorizing (a) the Debtors' sale of certain of their property free and clear of liens, claims, encumbrances, and interests on the terms set forth in that certain (i) Asset Purchase Agreement by and between Eiger InnoTherapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated as of August 1, 2024 for the sale of the Lonafarnib Assets (the "Lonafarnib APA") attached hereto as **Exhibit A** and (ii) Asset Purchase Agreement by and between Eiger InnoTherapeutics, Inc., as Purchaser, and Eiger BioPharmaceuticals, Inc., as Seller, Dated as of August 1, 2024 for the sale of the Lambda Assets (the "Lambda APA" and together with the Lonafarnib APA, the "Lonafarnib/Lambda APAs") attached hereto as **Exhibit B**; (b) the assumption and assignment of the Assigned Contracts in connection with the Lonafarnib/Lambda APAs; and (c) granting related relief, all as more fully set forth in the Motion; and this Court having previously entered the Order (I)(A) Approving the Bid Procedures;

⁻

² Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the *Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Docket No. 490] (the "Motion") or the Lonafarnib/Lambda APAs. If there are any inconsistencies between the defined terms in the Lonafarnib/Lambda APAs and the Motion, the defined terms of the Lonafarnib/Lambda APAs shall control.*

Casse2248800449sggj111 DDooc85588 Filiteck084225225 Hinteccck084225225161447525 DDessc MairEpholoutrisen1Page alger 32of 51

(B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; and (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order"); and the Debtors having filed the Notice of Cancellation of Auction(s), Designation of Winning Bid for the Lonafarnib Sale Transaction, and Transition To Private Sale Process for Lonafarnib/Lambda Sale Transactions [Docket No. 489] (the "Lonafarnib/Lambda Sale Notice") selecting Eiger InnoTherapeutics, Inc. ("Inno") as the highest and best bidder for the Transferred Assets (the "Purchaser") pursuant to the Lonafarnib/Lambda APAs with the Purchaser having submitted the highest and best offer for the Transferred Assets to be sold to the Purchaser as identified in the Lonafarnib/Lambda APAs, as reflected in the Lonafarnib/Lambda APAs and as from time to time may be amended in accordance with this Revised Lonafarnib/Lambda Sale Order or further order of this Court, including by the Lonafarnib/Lambda APAs, pursuant to which the Debtors have agreed, among other things, to sell the Transferred Assets to the Purchaser, including the Assigned Contracts that will be assumed and assigned to the Purchaser on the terms and conditions set forth in the Lonafarnib/Lambda APAs inclusive of the Lonafarnib Purchase Price (the "Lonafarnib Sale Transaction") and Lambda Purchase Price (the "Lonafarnib Sale Transaction" and together with the Lonafarnib Sale Transaction, the "Lonafarnib/Lambda Sale Transactions"); and the Debtors having filed the Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired

Casse2248800449sggj111 DDocc85888 Filiteck084225225 Hinteccck084225225161447525 DDessc MairEpholoutrisen1Page 5.cgef 4.20f 5.1

Leases in Connection with the Remaining Sale Transaction(s) [Docket No. 313] (the "Cure Notice") and the Amended Notice of Cure Amounts and Potential Assumption and Assignment of Executory Contracts and Unexpired Leases in Connection with the Remaining Assets Sale Transaction(s) [Docket No. 351] (the "Amended Cure Notice") and served the Lonafarnib Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. A] and the Lambda Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. B] (the "Assignment Notice"); and this Court having conducted the hearing to consider approval of the Lonafarnib/Lambda Sale Transactions (the "Lonafarnib/Lambda Sale Hearing"), at which time all interested parties were offered an opportunity to be heard with respect to the Lonafarnib/Lambda Sale Transactions; and this Court having reviewed and considered (i) the Motion and the exhibits thereto, (ii) the First Day Declaration [Docket No. 27], (iii) and the Declaration of J. Scott Victor in Support of the Debtors' Emergency Motion for the Entry of an Order (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Docket No. 491], and the arguments and representations of counsel made, and the evidence proffered or adduced at the Lonafarnib/Lambda Sale Hearing; and it appearing that due and proper notice of the Motion, the Lonafarnib/Lambda APAs, the Cure Notice, the Amended Cure Notice, and the Assignment Notice having been provided; and all objections, if any, to approval of the Lonafarnib/Lambda Sale Transactions having been withdrawn, resolved (including by separate agreement between the objecting party and the Debtors), adjourned, or overruled as provided in this Revised Lonafarnib/Lambda Sale Order; and

Casse2248800409ssjýj 11 DDocc85888 Filitiek k 108422525 Hintereck 108422522516144075025 DDess c Mair Etholobitr Bent Pag & Goef 52 of 51

it appearing entry of this Revised Lonafarnib/Lambda Sale Order and consummation of the Lonafarnib/Lambda Sale Transactions are in the best interests of the Debtors, their estates and creditors, and all parties in interest in these chapter 11 cases; and upon the record of the Lonafarnib/Lambda Sale Hearing and these chapter 11 cases; and after due deliberation thereon; and sufficient cause appearing therefor,

IT IS HEREBY FOUND AND DETERMINED THAT:

- A. <u>Fed. R. Bankr. P. 7052</u>. The findings and conclusions set forth herein constitute this Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such. This Court's findings shall also include any oral findings of fact and conclusions of law made by this Court during or at the conclusion of the Lonafarnib/Lambda Sale Hearing. To the extent of any conflict, the oral rulings control.
- B. <u>Jurisdiction and Venue</u>. This Court has jurisdiction over the Motion and the Lonafarnib/Lambda Sale Transactions described therein, and in the Lonafarnib/Lambda APAs, including, without limitation, the sale of the Transferred Assets, pursuant to 28 U.S.C. §§ 157 and 1334. Venue for these Chapter 11 Cases is proper pursuant to 28 U.S.C. § 1408. This Court may enter a final order consistent with Article III of the United States Constitution. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- C. <u>Statutory Predicates</u>. The statutory authorization for the relief granted herein is found in sections 105(a), 363, and 365 of the Bankruptcy Code; Bankruptcy Rules 2002, 6004,

Casse2248800409sggj.11 DDocc85888 Fill Helc 084225225 Hintered to 84225225161447525 DDess c Mair Exhibiting en Pag Pager 625 f 51

and 9014; Rule 9013-1 of the Bankruptcy Local Rules; and Section E of the Complex Case Procedures.

D. This Revised Lonafarnib/Lambda Sale Order constitutes a final and appealable order within the meaning of 28 U.S.C. § 158(a). Time is of the essence in closing the Lonafarnib/Lambda Sale Transactions referenced herein, and the Debtors and the Purchaser intend to close the Lonafarnib/Lambda Sale Transactions as soon as practicable in accordance with the Lonafarnib/Lambda APAs, and there is no just reason for delay in the implementation of this Revised Lonafarnib/Lambda Sale Order. Specifically, the Lonafarnib/Lambda Sale Transactions must be approved and consummated promptly in accordance with the Lonafarnib/Lambda APAs to preserve the viability of the business in the hands of the Purchaser as a going concern, and to maximize the value to the Debtors, their estates, their creditors, and all other parties in interest. Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), and to any extent necessary under Bankruptcy Rule 9014 and Rule 54(b) of the Federal Rules of Civil Procedure, as made applicable by Bankruptcy Rule 7054, the Court expressly finds that there is no just reason for delay in the implementation of this Revised Lonafarnib/Lambda Sale Order in accordance with the Lonafarnib/Lambda APAs, waives any stay, and expressly directs entry of judgment as set forth herein.

E. <u>Marketing Process</u>. The Debtors and their professionals adequately marketed the Transferred Assets to all Potential Bidders in accordance with the Bid Procedures Order. The sale process set forth in the Bid Procedures Order afforded all Potential Bidders (as defined in the Bid Procedures, attached as <u>Exhibit 1</u> to the Bid Procedures Order and revised in that *Notice of Filing of Revised Bidding Procedures*, filed on April 15, 2024 [Docket No. 119]), as modified by the *Revised Notice of Sale, Bid Procedures, Auction, and Sale Hearing* [Docket No. 331], as further

Casse2248800449sggj111 DDocc85888 Filitieck084225225 Hintercek084225225161447525 DDessc MairEpholoutrisen1Page 8 gef 320f 51

modified by the Further Revised Notice of Bid Deadlines [Docket No. 422], a full, fair, and reasonable opportunity to submit a higher or otherwise better offer to purchase the Transferred Assets and participate in the sale process. On August 2, 2024 the Debtors filed the Lonafarnib/Lambda Sale Notice that cancelled the Auctions. The value provided by the Purchaser pursuant to the Lonafarnib/Lambda APAs, which reflect (i) the final bid of a Base Price in the amount of \$5,200,000 plus the Purchaser Cure Amounts and Assumed Liabilities (collectively, the "Lonafarnib Purchase Price") and (ii) the final bid of a Base Price in the amount of \$1,000,000 plus the Purchaser Cure Amounts and Assumed Liabilities (collectively, the "Lambda Purchase Price" and together with the Lonafarnib Purchase Price, the "Purchase Price") provides the greatest value to the Debtors for the Transferred Assets. The Purchase Price constitutes the highest and best bid for the Transferred Assets. The marketing process was robust and sufficiently tested the market to determine the highest and best offer for the Transferred Assets.

- F. <u>Purchaser Protections</u>. The Purchaser Protections contained in the Lambda APA and the Lonafarnib APA (i) were necessary to preserve the value of the Debtors' estates by inducing the Purchaser to enter into the Lambda APA and the Lonafarnib APA, and (ii) are in compliance with the Bid Procedures and authorized by the Bid Procedures Order.
- G. <u>Sale Hearing</u>. This Court conducted the Lonafarnib/Lambda Sale Hearing on August 20, 2024, at which time this Court considered the Motion, the evidence and testimony presented, and the statements and argument of counsel, as applicable, in support of the Motion, the Lonafarnib/Lambda APAs, and the Lonafarnib/Lambda Sale Transactions. Except as otherwise expressly provided in this Revised Lonafarnib/Lambda Sale Order, all objections to the Lonafarnib/Lambda Sale Transactions and the relief requested in the Motion, whether timely or untimely and whether written or made orally at the Lonafarnib/Lambda Sale Hearing, if any, were

Casse2248800449sggj111 DDooc85888 Filitiele0084225225 Hintercelc084225225161447525 DDessc MairEpholoutrisentPage sign 520f 51

heard and considered by this Court. All such objections, if any, were either overruled by this Court, are resolved by the terms hereof or by separate agreement between the objecting party and the Debtors, or were adjourned or withdrawn as a result of an agreement between the objecting party and the Debtors.

- H. <u>Sound Business Purpose</u>. The Debtors have demonstrated good, sufficient, and sound business purposes and justifications for consummation of the Lonafarnib/Lambda Sale Transactions pursuant to the Lonafarnib/Lambda APAs and all other agreements, instruments, certificates, and other documents to be entered into or delivered by any party in connection with the Lonafarnib/Lambda Sale Transactions, including, without limitation, any assumption and assignment agreements entered into in connection therewith and any agreement entered into or to be entered into by the Purchaser with Merck Sharp & Dohme LLC ("Merck") contemplated by the Lonafarnib APA (collectively, the "Transaction Documents"), outside of the ordinary course of business and in accordance with the requirements of section 363(b) of the Bankruptcy Code. Consummation of the Lonafarnib/Lambda Sale Transactions prior to and not as part of a chapter 11 plan is (i) justified under the circumstances, (ii) an appropriate exercise of the Debtors' business judgment, and (iii) in the best interests of the Debtors, their estates, and their creditors.
- I. Following a robust marketing process, the Lonafarnib/Lambda APAs and the Purchase Price constitute the highest and best offer for the Transferred Assets. No other person, or group of persons, has offered to purchase the Transferred Assets for an amount that would give greater value to the Debtors than the value provided by the Purchase Price. The Lonafarnib/Lambda Sale Transactions are the best means available to the Debtors to maximize the return to their creditors and limit the losses to counterparties to the Assigned Contracts. No

alternative to the Lonafarnib/Lambda Sale Transactions exists that would provide a greater value to the Debtors, their creditors, or other parties in interest.

- J. Approval of the Lonafarnib/Lambda Sale Transactions is necessary to maximize the value the Debtors' estates will receive for the Transferred Assets. It is important to the Debtors' customers and suppliers that the transition from the Debtors to the Purchaser occurs smoothly and without unnecessary delay, so that any customer and vendor issues may be minimized. It is also important that the Lonafarnib/Lambda Sale Transactions be consummated as expeditiously as possible to avoid any disruption to the ongoing development of the Lonafarnib/Lambda Assets and so there is no uncertainty about the future of the Lonafarnib/Lambda Assets.
- K. Accordingly, the sale of the Transferred Assets pursuant to sections 105(a), 363, and 365 of the Bankruptcy Code upon the terms and conditions set forth in the Lonafarnib/Lambda APAs is the optimal means to create value for the benefit of the Debtors' estates. The Lonafarnib/Lambda Sale Transactions maximize the value of the Transferred Assets because the Transferred Assets are being sold as part of a going concern, thereby preserving the continuity and remaining goodwill value associated with the Transferred Assets. Unless the sale is concluded expeditiously, as provided for in the Motion and the Lonafarnib/Lambda APAs, creditor recoveries may be substantially diminished.
- L. <u>Fair Purchase Price</u>. The Purchase Price provided by the Purchaser (i) is fair and adequate; (ii) constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia (including the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act, and similar laws); and (iii) will provide an equal or greater recovery for the Debtors' creditors

Casse2**24880040**9ssgij**11**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainExholicum8enPage **1.0**162of 51

than would be provided by any other available alternative. The terms of the Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions are fair and reasonable under the circumstances of the Debtors' chapter 11 cases, and the Debtors' determination to proceed with such transaction constitutes a valid and sound exercise of the Debtors' business judgment.

M. Adequate and Reasonable Notice. As evidenced by the affidavits of service filed with this Court [Docket Nos. 42, 114, 128, 140, 319, 320, 324, 509, 512], and based upon the record of the Lonafarnib/Lambda Sale Hearing, (i) due, proper, timely, adequate, and sufficient notice of the Motion, the Lonafarnib/Lambda Sale Notice, the Lonafarnib/Lambda Sale Hearing, the Lonafarnib/Lambda APAs, and the Lonafarnib/Lambda Sale Transactions has been provided to all parties in interest, (ii) such notice was and is good, sufficient, and appropriate under the circumstances, and reasonably calculated to reach and apprise all holders of Liens, claims, encumbrances, and other Interests (as defined herein), including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities, and was provided in accordance with the applicable requirements of the Bankruptcy Code, the Bankruptcy Rules, the Bankruptcy Local Rules, the Complex Case Procedures, and the procedural due process requirements of the United States Constitution, and (iii) no other or further notice of the Motion, the Lonafarnib/Lambda Sale Hearing, the Lonafarnib/Lambda APAs, the Lonafarnib/Lambda Sale Transactions, or of the entry of this Revised Lonafarnib/Lambda Sale Order is necessary or shall be required.

N. In accordance with the Bid Procedures Order, the Debtors filed with this Court and served the Cure Notice and Amended Cure Notice, containing (i) the list of all Contracts to potentially be assigned in connection with the Lonafarnib/Lambda Sale Transactions,

Casse2248800499sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibiting en Pag 12 12 20 152 of 51

(ii) information necessary and appropriate to provide notice of the relevant proposed assumption and assignment of Potentially Assigned Contracts (as defined in the Cure Notice and Amended Cure Notice) and rights thereunder, (iii) Cure Amounts, where applicable, and (iv) the procedures for objecting thereto, on all counterparties to such Potentially Assigned Contracts and any party that has requested notice pursuant to Bankruptcy Rule 2002 ("Rule 2002 Notice List"), and caused such notice to be published on the website of the Debtors' noticing agent, Kurtzman Carson Consultants LLC dba Verita Global ("Verita") [Docket No. 313]. The Cure Notice and Amended Cure Notice (a) included the Debtors' good faith calculation of the Cure Amounts with respect to each Potentially Assigned Contract; (b) stated that assumption or assignment of any Potentially Assigned Contract is not guaranteed and is subject to this Court's approval; and (c) prominently displayed the deadline to file a Cure Objection (as defined herein). The service and provision of the Cure Notice and Amended Cure Notice were good, sufficient, and appropriate under the circumstances, and no other or further notice need be given.

O. The Debtors also served the Assignment Notice on the counterparties to the Assigned Contracts, which contained (i) the list of Assigned Contracts selected by the Purchaser, (ii) information necessary and appropriate to provide notice of the relevant proposed assumption and assignment of the Assigned Contracts and rights thereunder, (iii) the Cure Amounts, and (iv) the procedures for requesting adequate assurance of future performance. The Debtors also served the [Notice of Lonafarnib/Lambda Sale Hearing] [Docket No. 509] (the "Notice of Lonafarnib/Lambda Sale Hearing") on the counterparties to the Assigned Contracts. The service and provision of the Assignment Notice and Notice of Lonafarnib/Lambda Sale Hearing was good, sufficient, and appropriate under the circumstances, and no other or further notice need be given in connection with the assumption and assignment of the Assigned Contracts.

Casse2248800499sgijj11 DDocc85888 Filitiek0084225225 Hintereck084225225161447525 DDessc MainElokoikuin6en1Pagle746eo11926f51

P. A reasonable opportunity to object and to be heard with respect to the sale of the Transferred Assets, the assumption and assignment of the Assigned Contracts, and the determination of defaults and Cure Amounts related thereto, as well as the Lonafarnib/Lambda APAs and the entry of this Revised Lonafarnib/Lambda Sale Order, has been given to all interested Persons.

Q. Good Faith Purchaser. The Debtors, the Purchaser, and their respective principals, counsel, and advisors have negotiated, proposed, and entered into the Lonafarnib/Lambda APAs, the Transaction Documents, and each of the transactions contemplated therein in good faith, without collusion and from arm's-length bargaining positions. The Purchaser is a "good faith purchaser" and is acting in good faith within the meaning of section 363(m) of the Bankruptcy Code in connection with the Lonafarnib/Lambda Sale Transactions and, as such, is entitled to all the protections afforded thereby. The Purchaser has proceeded in good faith in all respects. The terms of the Lonafarnib/Lambda Sale Transactions, including the Purchase Price, were not controlled by any agreement among Potential Bidders and neither the Debtors nor the Purchaser have engaged in collusion or any conduct that would cause or permit the Lonafarnib/Lambda APAs to be challenged, avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy Code or any other law of the United States, any state, territory, possession thereof, or the District of Columbia, or any other applicable law. The Lonafarnib/Lambda APAs were not entered into for the purpose of hindering, delaying, or defrauding creditors under the Bankruptcy Code or under laws of the United States, any state, territory, or possession, or the District of Columbia, or any other applicable law. Neither the Debtors nor the Purchaser entered into the Lonafarnib/Lambda APAs or are consummating the Lonafarnib/Lambda Sale Transactions with any fraudulent or otherwise improper purpose. The

Casse2**24880049**9sggj**11**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainElabrichingen Pagle 14e0**1.3**2of 51

Purchaser is not an "insider" or "affiliate" of any of the Debtors, as those terms are defined in section 101 of the Bankruptcy Code, and no common identity of incorporators, directors, or controlling stockholders exists between the Purchaser and the Debtors.

- R. The Lonafarnib/Lambda Sale Transactions, which include the sale of the Transferred Assets pursuant to the Lonafarnib/Lambda APAs and all covenants in and conditions thereto, is an integrated transaction, meaning that each component is an essential part of every other component and that the Lonafarnib/Lambda Sale Transactions can be consummated only if all of the components are consummated. Accordingly, each component of the Lonafarnib/Lambda Sale Transactions is subject to, and is protected by, the provisions of section 363(m) of the Bankruptcy Code.
- S. Sale Free and Clear under Section 363(f). The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions without entry of this Revised Lonafarnib/Lambda Sale Order approving the Lonafarnib/Lambda Sale Transactions pursuant to section 363(f) of the Bankruptcy Code. Except as expressly provided otherwise in the Lonafarnib/Lambda APAs or this Revised Lonafarnib/Lambda Sale Order, the Debtors have satisfied the standard set forth in section 363(f) of the Bankruptcy Code for selling the Transferred Assets free and clear of all of the following (collectively, "Interests"): Liens (including Permitted Liens), claims (including, but not limited to, those that constitute a "claim" as defined in section 101(5) of the Bankruptcy Code), encumbrances, obligations, liabilities, pledges, charges, demands, guarantees, actions, suits, defenses, deposits, credits, allowances, options, rights, restrictions, limitations, contractual commitments, rights of first refusal, rights of setoff or recoupment, royalties, hypothecations, preferences, debts, easements, suits, licenses, rights of recovery, judgments, orders and decrees of

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibiting en Pag 12 45 20 f 51

any court or foreign or domestic governmental entity, taxes (including foreign, state, and local taxes), covenants, indentures, instruments, leases, claims for reimbursement or subrogation, contribution, indemnity or exoneration, encumbrances, or interests of any kind or nature whatsoever against the Debtors, or any of the Transferred Assets, including, without limitation, any debts arising under or out of, in connection with, or in any way relating to, any acts or omissions, obligations, demands, guaranties, rights, contractual commitments, restrictions, product liability claims, environmental liabilities, employment or labor law claims or liabilities, employee pension or benefit plan claims, multiemployer benefit plan claims, retiree healthcare or life insurance claims or claims for taxes of or against the Debtors or against any property of the Debtors, claims arising under state or federal antitrust laws, any indemnification claim or liabilities relating to any act or omission of the Debtors or any other person prior to the Closing Date or any Excluded Liabilities, any derivative, vicarious, transferee or successor liability claims, alter ego claims, de facto merger claims, rights or causes of action (whether known or unknown, legal or equitable, contingent, matured or unmatured, contingent or non-contingent, liquidated or unliquidated, choate or inchoate, filed or unfiled, scheduled or unscheduled, perfected or unperfected, allowed or disallowed, noticed or unnoticed, recorded or unrecorded, material or nonmaterial, statutory or non-statutory, and asserted or unasserted), whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases (other than the Assumed Liabilities), whether imposed by agreement, understanding, law, equity or otherwise, including without limitation (i) those Interests that purport to give to any party a right or option to effect a setoff against or any forfeiture, modification, or termination of the Debtors' interests in the Transferred Assets, or any similar rights, if any, (ii) those Interests arising under all mortgages, deeds of trust, security interests, conditional sale or other title retention agreements, pledges,

Casse2248800499sggj111 DDocc85888 Fill Helc 084225225 Hintered 084225225161447525 DDess c Mair Exhibit with en Pag & 16 eo 152 of 51

hypothecations, liens, judgments, demands, encumbrances, rights of first refusal or charges of any land or nature, if any, (iii) those Interests that are Excluded Liabilities as set forth in the Lonafarnib/Lambda APAs, (iv) those Interests held by the Prepetition Term Loan Lenders (as defined in the Final Cash Collateral Order) as provided in the order entered by the Court at Docket No. 161 (the "Final Cash Collateral Order"), and (v) those Interests arising under or out of, in connection with, or in any way related to the Debtors or any of the Debtors' predecessors, Affiliates, or representatives, any of the Debtors' interests in the Transferred Assets, or the operation of any of the Debtors' businesses before the applicable Closing Date, including, without limitation, Interests based on successor liability, transferee liability, derivative liability, vicarious liability, de facto merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. Each holder of an Interest in the Transferred Assets (a) has, subject to the terms and conditions of this Revised Lonafarnib/Lambda Sale Order, consented or shall be deemed to have consented to the relief requested in the Motion and with respect to the Lonafarnib/Lambda Sale Transactions, (b) could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest, or (c) otherwise falls within the provisions of section 363(f) of the Bankruptcy Code. Those holders of Interests that did not object to, or withdrew their objections, if any, to, the relief requested in the Motion, the Lonafarnib/Lambda APAs, the Lonafarnib/Lambda Sale Transactions, the Cure Notice, the Amended Cure Notice, or the Assignment Notices are deemed to have consented to the relief requested in the Motion, including, without limitation, the sale of the Transferred Assets and the assumption and assignment of the Assigned Contracts to the Purchaser, pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests that did object that have an Interest in the Transferred Assets could be compelled in a legal or equitable proceeding to accept money satisfaction of such Interest pursuant

Casse2248800409ssgijj.11 DDoor:85888 Filitiek0084225225 Hintereck008422522516144075025 DDess c MainExholicum8en1Page2167eo1620f51

to section 363(f)(5) of the Bankruptcy Code or fall within one or more of the other subsections of 363(f) of the Bankruptcy Code and, therefore, are adequately protected by having their Interests that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Lonafarnib/Lambda Sale Transactions ultimately attributable to the property in which they have an Interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Lonafarnib/Lambda Sale Transactions, subject to any defenses of the Debtors.

T. Except as expressly provided otherwise in the Lonafarnib/Lambda APAs or this Revised Lonafarnib/Lambda Sale Order, neither the Purchaser nor any of the Purchasers' Affiliates (including any subsidiary of the Purchaser, any person or entity that could be treated as a single employer with the Purchaser pursuant to Section 4001(b) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") or Section 414(b), (c), (m) or (o) of the Internal Revenue Code of 1986, as amended ("IRC"), and any of their respective managed funds or accounts, any of their respective lenders or investors, and, in each case of the foregoing, each of their respective former, current, or future, shareholders, equity holders, owners, members, managers, employees, representatives, officers, limited or general partners, directors, agents, professionals, successors, affiliates, or permitted assignees) (collectively with the Purchaser, the "Purchaser Group") shall be responsible for any Interests, including in respect of, based on, relating to, or arising under, without limitation, the following: (i) any labor, collective bargaining, or employment agreements; (ii) any mortgages, deeds of trust, or security interests; (iii) any intercompany loans and receivables between one or more of the Seller and any Debtor; (iv) any pension, multiemployer (as such term is defined in Section 3(37) or Section 4001(a)(3) of ERISA), health or welfare plan participation or benefit trust, compensation or other employee benefit plans,

Casse2248800409ssjójj11 DDocc85888 Filitelc084225245 EEnteredc084225251614407505 DDessc MainExholiciún6en Pag & 16601520 f 51

agreements, practices and programs (including any Employee Benefit Plan) of or related to any of the Debtors or any of the Debtors' Affiliates or predecessors or any current or former employees of any of the foregoing, including, without limitation, any pension plan of any of the Debtors or any multiemployer plan to which the Debtors have at any time contributed to or had any liability or potential liability; (v) the Debtors' business operations or cessation thereof; (vi) any litigation involving one or more of the Debtors; (vii) any other employee, worker's compensation, occupational disease or unemployment or temporary disability related claim, including, without limitation, claims that might otherwise arise under or pursuant to (a) ERISA, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the Multi-Employer Pension Plan Amendments Act of 1980, including all amendments thereto, (f) the Worker Adjustment and Retraining Notification Act of 1988 or any similar state or local law ("WARN"), (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, including, without limitation, the requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the IRC and of any similar state law (collectively, "COBRA"), (i) the National Labor Relations Act, (j) the Age Discrimination and Employment Act of 1967 and Age Discrimination in Employment Act, as amended, (k) state harassment, discrimination, or retaliation laws, (l) state unemployment compensation laws or any other similar state laws, or (m) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors, or relating to any wages, benefits, employment, or termination of employment with any or all Debtors or any of their predecessors; (viii) any liabilities arising under any Environmental Laws with respect to any assets owned or operated by any of the Debtors or any corporate predecessor of any of the Debtors at any time prior to the applicable Closing Date; (ix) any product liability law; (x) any antitrust laws;

Casse2248800409sgijj11 DDocc85888 Filitiek0084225225 Hintereck0842252251614407505 DDess C MairExhibitin6enPage 1.9eo 1.52 of 51

(xi) any bulk sales or similar law; (xii) any tax statutes or ordinances, including, without limitation,

the IRC; and (xiii) any Excluded Liabilities.

U. No Successor, Transferee, or Similar Liability. Except for the Assumed

Liabilities, as expressly set forth in the Lonafarnib/Lambda APAs or this Revised

Lonafarnib/Lambda Sale Order, the Purchaser has not expressly or impliedly assumed any

obligation of the Debtors, or any other party, with respect to the Interests and the Excluded

Liabilities, whether at law or in equity, whether by payment, setoff, recoupment, or otherwise,

directly or indirectly, and whether from the Transferred Assets or otherwise, including, without

limitation, based on successor, transferee, derivative, or vicarious liability.

V. The Lonafarnib/Lambda Sale Transactions described by the Lonafarnib/Lambda

APAs and the Transaction Documents does not amount to a consolidation, merger, or de facto

merger of the Purchaser and any of the Debtors and/or any of the Debtors' estates.

W. There is no continuity between the Purchaser and any of the Debtors. The

Purchaser is not holding itself out to the public as a continuation of any of the Debtors or their

respective estates, businesses, or operations. The Purchaser is not a mere continuation of any of

the Debtors or their respective estates, businesses, or operations. There is no common identity

between any of the Debtors and the Purchaser. The Purchaser does not constitute a successor or

a successor in interest to any of the Debtors or their estates.

X. The Purchaser and the Debtors are not entering into the Lonafarnib/Lambda APAs

and Transaction Documents or consummating the Lonafarnib/Lambda Sale Transactions for the

fraudulent purpose of escaping liability for the Debtors' obligations or to defraud creditors in any

way.

Y. Sale Free and Clear and Continuation of Existing Approvals Required by the

Purchaser. The Purchaser expressly negotiated for the protection of obtaining the Transferred Assets free and clear of all Interests, including, without limitation, any potential successor liability claims (other than the Assumed Liabilities). The total consideration to be provided under the Lonafarnib/Lambda APAs reflects the Purchaser's reliance on this Revised Lonafarnib/Lambda Sale Order to provide it, pursuant to sections 105(a), 363, and 365 of the Bankruptcy Code, with title to and possession of the Transferred Assets free and clear of all Interests of any kind or nature whatsoever (including, without limitation, any potential successor liability claims (other than the Assumed Liabilities)). The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions if the sale of the Transferred Assets to the Purchaser and the assumption and assignment of the Assigned Contracts to the Purchaser by the Debtors were not free and clear of all Interests of any kind or nature whatsoever (other than the Assumed Liabilities), as contemplated by this Revised Lonafarnib/Lambda Sale Order, or if the Purchaser would, or in the future could, be liable for any of the Interests, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities. The Purchaser would not have entered into the Lonafarnib/Lambda APAs and would not consummate the Lonafarnib/Lambda Sale Transactions if the Purchaser would not be authorized, as of the Closing Date, to operate under or renew any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the terms and conditions of the Lonafarnib/Lambda APAs); if such licenses, permits, registrations, and governmental authorizations or approvals would not be deemed to have been transferred to the Purchaser as of the Closing Date; or if existing licenses or

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibitum 8en Pag (Pagleo 26) 26 2 of 51

permits applicable to the business would not remain active and in place for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred.

Assignment of the Assigned Contracts are integral to the Lonafarnib/Lambda APAs, do not constitute unfair discrimination, are in the best interests of the Debtors, their estates and creditors, and all other parties in interest, and are based on the reasonable exercise of sound business judgment by the Debtors. At the Closing and pursuant to Section 365 of the Bankruptcy Code and this Revised Lonafarnib/Lambda Sale Order, the Debtors shall assume and, subject to the terms in the Lonafarnib/Lambda APAs, assign to the Purchaser, and Purchaser shall take assignment from the Debtors of, the Assigned Contracts.

AA. On or after the Closing Date, the Purchaser will pay all Purchaser Cure Amounts for Assigned Contracts (if any), and the Seller will pay all Seller Cure Amounts (if any) for Assigned Contracts, including with respect to the Assigned Contracts proposed to be resolved after the Closing Date in accordance with Paragraph 19 hereof. Accordingly, the Debtors or the Purchaser, as applicable, will have, to the extent necessary, (i) cured any default existing prior to the Closing with respect to the Assigned Contracts, and (ii) provided compensation, if any, to each counterparty to an Assigned Contract for any actual pecuniary loss to such party resulting from a default prior to the Closing with respect to the Assigned Contract with such counterparty, all within the meaning of sections 365(b)(1)(A) and 365(f)(2)(A) of the Bankruptcy Code.

BB. Pursuant to section 365(f) of the Bankruptcy Code, each Assigned Contract required to be assumed and assigned under the Lonafarnib/Lambda APAs shall be assigned and transferred to, and remain in full force and effect for the benefit of, the Purchaser, in accordance with their respective terms, notwithstanding any provision in such contract or other restrictions

prohibiting its assignment or transfer. No section of any of the Assigned Contracts that would directly or indirectly prohibit, restrict, or condition the assumption or assignment of any of the Assigned Contracts or would permit termination or modification of such Assigned Contracts, or rights and obligations thereunder, by a party other than the Debtors, on account of assignment of such shall have any force or effect in connection with the Transferred Assets.

CC. The assumption and assignment of the Assigned Contracts (i) are necessary to sell the Transferred Assets to the Purchaser, (ii) allow the Debtors to sell the Transferred Assets to the Purchaser as a going concern, (iii) limit the losses suffered by counterparties to the Assigned Contracts, and (iv) maximize the recoveries to other creditors of the Debtors by limiting the number of claims against the Debtors' estates by avoiding the rejection of the Assigned Contracts. For these reasons, the Debtors have exercised sound business judgment in assuming and assigning the Assigned Contracts and such assumption and assignment is in the best interests of the Debtors' estates.

DD. Adequate Assurance of Future Performance. Counterparties to the Assigned Contracts were provided with the Assignment Notice and had the opportunity to request and review information with respect to the Purchaser's adequate assurance of future performance (*see* Lonafarnib/Lambda Sale Notice). No counterparties to Assigned Contacts filed any objections to the Purchaser's ability to provide adequate assurance of future performance as contemplated under sections 365(b)(l)(C), 365(b)(3) (to the extent applicable) and 365(f)(1) of the Bankruptcy Code (each, an "Adequate Assurance Objection") prior to the Lonafarnib/Lambda Sale Hearing. Counterparties to Assigned Contracts that failed to timely file an Adequate Assurance Objection are hereby forever barred from objecting to the assumption and assignment of Assigned Contracts on the grounds of a failure to provide adequate assurance of future performance. Based

Casse2248800409sgijj111 DDocc85888 Filitiek0084225225 Hintereck0842252251614407525 DDessc MainElokoikuin6en1Pagle266e02526f51

on evidence adduced at the Lonafarnib/Lambda Sale Hearing and based on the record in these chapter 11 cases, to the extent necessary, the Debtors have satisfied the requirements of section 365 of the Bankruptcy Code, including sections 365(b)(l)(A), 365(b)(l)(B), 365(b)(l)(C), 365(b)(3) (to the extent applicable) and 365(f) of the Bankruptcy Code, in connection with the sale and assumption and assignment of the Assigned Contracts to the extent provided under the Lonafarnib/Lambda APAs. Accordingly, subject to payment of the Cure Amounts, the Assigned Contracts may be assumed by the Debtors and assigned to the Purchaser as provided under the Lonafarnib/Lambda APAs and this Revised Lonafarnib/Lambda Sale Order.

- EE. Revised Lonafarnib/Lambda Sale Order Required by the Purchaser. Entry of this Revised Lonafarnib/Lambda Sale Order approving the Lonafarnib/Lambda APAs is a requirement of the Lonafarnib/Lambda APAs and such requirement is a reasonable and appropriate condition precedent to the Purchaser's consummation of the Lonafarnib/Lambda Sale Transactions.
- FF. <u>Transferred Assets Property of the Estates</u>. The Transferred Assets constitute property of the selling Debtors' estates and title thereto is vested in the selling Debtors' estates within the meaning of section 541(a) of the Bankruptcy Code. The selling Debtors have all title, interest, and/or rights in the Transferred Assets required to transfer and to convey the Transferred Assets to the Purchaser, as required by the Lonafarnib/Lambda APAs.
- GG. <u>Corporate Authority</u>. Subject to the entry of this Revised Lonafarnib/Lambda Sale Order, (i) the Debtors have full corporate power and authority to perform all of their obligations under the Lonafarnib/Lambda APAs and the Transaction Documents, and the Debtors' prior execution and delivery of, and performance of obligations under, the Lonafarnib/Lambda APAs and the Transaction Documents is hereby ratified, (ii) the Debtors have all of the corporate

power and authority necessary to consummate the Lonafarnib/Lambda Sale Transactions, (iii) the Debtors have taken all corporate actions necessary to authorize, approve, execute, and deliver the Lonafarnib/Lambda APAs and the Transaction Documents and to consummate the Lonafarnib/Lambda Sale Transactions, except for the closing conditions expressly provided in the Lonafarnib/Lambda APAs and the Transaction Documents, and (iv) no consents or approvals are required to consummate the Lonafarnib/Lambda Sale Transactions or otherwise perform the obligations under the Lonafarnib/Lambda APAs or the Transaction Documents, except for the closing conditions expressly provided herein or therein.

- HH. <u>Sale in Best Interests</u>. The relief requested in the Motion and set forth in this Revised Lonafarnib/Lambda Sale Order is in the best interests of the Debtors, their respective creditors, estates, and all other parties in interest in the Debtors' chapter 11 cases.
- II. **Prompt Consummation**. To maximize the value of the Transferred Assets, it is essential that the Lonafarnib/Lambda Sale Transactions occur within the timeframe set forth in the Lonafarnib/Lambda APAs. Time is of the essence in consummating the Lonafarnib/Lambda Sale Transactions. Accordingly, there is cause to lift the stays established by Bankruptcy Rules 6004 and 6006 with regards to the Lonafarnib/Lambda Sale Transactions and the assignment of the Assigned Contracts.

NOW, THEREFORE, IT IS ORDERED THAT:

- 1. <u>Motion Is Granted</u>. The Motion and the relief requested therein, and entry into and performance under the Lonafarnib/Lambda APAs, is GRANTED and APPROVED, as set forth herein.
- 2. <u>Objections Overruled</u>. Except as stated otherwise herein, all objections to, or reservation of rights regarding, the relief requested in the Motion, the entry of this Revised

Casse2**24880049**9sgijj**1**11 DDoc:8**5**588 Filitiek0**08422**52**2**5 <u>Hinterec</u>k0**8422**52**2**516144075**2**5 DDessc MainExholicum8en1Page**25**eo**2**520f 51

Lonafarnib/Lambda Sale Order, or the relief granted herein, including, without limitation, any objections to the assumption or assignment of the Assigned Contracts (including Cure Amounts related thereto) or relating to the cure of any defaults under any of the Assigned Contracts or to the assumption and assignment of any of the Assigned Contracts to the Purchaser by the Debtors, that have not been withdrawn, waived, settled, or adjourned as provided in Paragraphs 17-19 below or otherwise, or that have not otherwise been resolved pursuant to the terms hereof are hereby denied and overruled on the merits with prejudice. All Persons that failed to timely object, or withdrew their objections, to the Motion or the entry of this Revised Lonafarnib/Lambda Sale Order are deemed to consent to the relief granted herein for all purposes, including, without limitation, pursuant to section 363(f)(2) of the Bankruptcy Code.

- 3. <u>Notice</u>. Notice of the Motion, the Lonafarnib/Lambda Sale Hearing, and the assumption and assignment of Assigned Contracts was adequate, appropriate, fair, and equitable under the circumstances and complied in all respects with section 102(1) of the Bankruptcy Code and Bankruptcy Rules 2002, 6004, and 6006, and the Bankruptcy Local Rules, and as such no further or other notice is required.
- 4. <u>Approval and Authorization</u>. The sale of the Transferred Assets to the Purchaser on the terms and conditions contained in the Lonafarnib/Lambda APAs and the Transaction Documents, including, without limitation, the Closing of the Lonafarnib/Lambda Sale Transactions as required by the Lonafarnib/Lambda APAs, is hereby approved in all respects pursuant to sections 105(a), 363(b), 363(f), 363(m), and 365 of the Bankruptcy Code and Bankruptcy Rules 6004 and 6006. Pursuant to sections 105, 363, and 365 of the Bankruptcy Code, the Debtors are authorized to perform all obligations under and make all payments required by the Lonafarnib/Lambda APAs and the Transaction Documents as and when due thereunder without

Casse2248800449sgijj11 Dooc85588 Filitiek008422525 Hintereck084225225161447525 Desse MainElokoikuin6en1Pagle266e02526f51

further order of this Court. The Debtors, the Purchaser, and each of their respective officers, employees, and agents are hereby authorized to (i) execute the Lonafarnib/Lambda APAs and the Transaction Documents, including the Lonafarnib/Lambda APAs, and any prior execution of such agreements, documents, and instruments, including the Transaction Documents, is hereby ratified, (ii) perform all obligations under the Lonafarnib/Lambda APAs and the Transaction Documents, to consummate each of the foregoing, including, without limitation, deeds, assignments, and other instruments of transfer, and to consummate the Lonafarnib/Lambda Sale Transactions, and any prior performance of such obligations or any prior consummation of such Lonafarnib/Lambda Sale Transactions is hereby ratified, (iii) assume and assign the Assigned Contracts to the Purchaser, and (iv) take all other and further actions as may be reasonably necessary to consummate and implement the Lonafarnib/Lambda Sale Transactions and to perform all obligations under the Lonafarnib/Lambda APAs and the Transaction Documents and the consummation thereof, without any further corporate action or order of this Court. The Purchaser shall not be obligated to proceed with the Closing under the Lonafarnib/Lambda APAs until all conditions precedent to its obligation to do so thereunder have been satisfied or waived.

- 5. No Sub Rosa Plan. The sale of the Transferred Assets, including, without limitation, the assignment of the Assigned Contracts, pursuant to the Lonafarnib/Lambda APAs outside a chapter 11 plan neither impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms of the Debtors' subsequent chapter 11 plan. Neither the Lonafarnib/Lambda APAs nor the Lonafarnib/Lambda Sale Transactions constitute a sub rosa chapter 11 plan.
- 6. <u>Valid Transfer</u>. As of the Closing, the consummation of the Lonafarnib/Lambda Sale Transactions shall effect a legal, valid, and enforceable sale and transfer of the Transferred

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibitum 8en Pag (Pag (exage) 252) f 51

Assets to the Purchaser, and shall vest the Purchaser with all legal, equitable, and beneficial right, title, and interest in and to the Transferred Assets free and clear of all Interests of any kind or nature whatsoever. The Lonafarnib/Lambda APAs and the Transaction Documents are valid and binding contracts between the Debtors and the Purchaser and shall be enforceable pursuant to their terms. The Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions themself, and the consummation thereof, shall be specifically enforceable against and binding upon (without posting any bond) the Debtors and their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Assigned Contracts, the Purchaser and its respective successors and assigns, any chapter 11 trustee appointed in these chapter 11 cases or any chapter 7 trustee appointed upon a conversion of these chapter 11 cases to cases under chapter 7 of the Bankruptcy Code, and shall not be subject to rejection or avoidance by the foregoing parties or any other Person.

7. Free and Clear. Except as expressly provided for in the Lonafarnib/Lambda APAs or this Revised Lonafarnib/Lambda Sale Order, pursuant to sections 105(a), 363(b), 363(f), 365(b), and 365(f) of the Bankruptcy Code, the Debtors are authorized and directed to transfer the Transferred Assets to the Purchaser and, upon the Closing, other than the Purchaser's assumption of the Assumed Liabilities and the Purchaser's obligations under the Lonafarnib/Lambda APAs and the Assigned Contracts, the Purchaser shall have and take title to and possession of the Transferred Assets free and clear of and shall have no obligation with respect to all Interests (other than the Assumed Liabilities) of any kind or nature whatsoever, including, without limitation, rights or claims based on any successor, transferee, derivative, or vicarious liabilities; de facto

Casse2**24880049**9ssjýj**11**1 DDoc3**5**588 Filitelo0**8422**52**2**5 Eintereclo0**8422**52**2**516144075**2**5 DDessc MairExholiciún6en1Page286eo23651

merger, continuation or continuity, or any similar theories under applicable state or federal law or otherwise. All holders of Interests fall within one or more of the subsections of section 363(f) of the Bankruptcy Code and are adequately protected by having their Interests attach to the net proceeds ultimately received by the Debtors and attributable to the Transferred Assets against or in which such Interests are asserted, subject to the terms of such Interests, with the same validity, force, and effect, and in the same order of priority that such Interests now have against the Transferred Assets or their proceeds as of Closing, subject to any rights, claims, and defenses the Debtors or their estates, as applicable, may possess with respect thereto, in addition to any limitations on the use of such proceeds pursuant to any provision of this Revised Lonafarnib/Lambda Sale Order. This Revised Lonafarnib/Lambda Sale Order: (a) is and shall be effective as a determination that, other than Assumed Liabilities or as otherwise provided herein, upon the applicable Closing in accordance with the Lonafarnib/Lambda APAs, all claims of any kind or nature whatsoever existing as to Transferred Assets, and any tax liability, prior to the applicable Closing have been unconditionally released, discharged, and terminated, and that the conveyances described herein have been effected, with such Interests and liens attaching in order of priority to the proceeds of the Lonafarnib/Lambda Sale Transactions, and (b) is and shall be binding upon and shall authorize all entities, including without limitation all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies or units, governmental departments or units, secretaries of state, federal, state and local officials and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to the Transferred Assets conveyed to the Purchaser. All recorded Interests against the

Casse2**24880040**9ssgj**j11**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainExholicum8en1Page2**9**eo**25**20f 51

Transferred Assets from their records, official and otherwise, shall be deemed stricken upon the Closing in accordance with the Lonafarnib/Lambda APAs and the terms of this Revised Lonafarnib/Lambda Sale Order without the need for further action on the part of either the Purchaser or the Seller. The conditions of section 363(f) of the Bankruptcy Code have been satisfied in full; therefore, the Debtor is authorized and directed to sell the Transferred Assets free and clear of any liens, claims, and/or interests (other than the Assumed Liabilities).

- 8. Those holders of Interests or claims who did not object (or who ultimately withdrew their objections, if any) to the Lonafarnib/Lambda Sale Transactions are deemed to have consented pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests or claims who did object that have an interest in the Transferred Assets fall within one or more of sections 363(f)(1), 363(f)(3), 363(f)(4), or 363(f)(5) of the Bankruptcy Code and are therefore adequately protected by having their Interests or claims that constitute interests in the Transferred Assets, if any, attach solely to the proceeds of the Lonafarnib/Lambda Sale Transactions ultimately attributable to the property in which they have an interest, in the same order of priority and with the same validity, force, and effect that such holders had prior to the Lonafarnib/Lambda Sale Transactions, subject to any defenses of the Debtors.
- 9. **Release of Interests**. Any and all Persons that have filed a financing statement, mortgage, mechanic's lien, *lis pendens*, or other document or agreement evidencing an Interest against or in the Transferred Assets shall deliver to the Debtors prior to the Closing, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, releases, and/or any other similar documents necessary for the purpose of documenting all Interests that such Person has against or in the Transferred Assets. For any Person who has not delivered such termination statements to the Debtors prior to the Closing, (i) the

Debtors and/or the Purchaser are hereby authorized to execute and file such statements, instruments, releases, and/or other similar documents on behalf of such Person with respect to the Transferred Assets, (ii) the Purchaser is hereby authorized to file, register, or otherwise record a certified copy of this Revised Lonafarnib/Lambda Sale Order that, once filed, registered, or otherwise recorded, shall constitute conclusive evidence of the release of all Interests of any kind or nature against or in the Transferred Assets, and (iii) the Purchaser may seek in this Court, or any other court of appropriate jurisdiction, to compel the appropriate parties to execute termination statements, instruments of satisfaction, releases, and/or other similar documents with respect to all Interests that such Person has against or in the Transferred Assets. This Revised Lonafarnib/Lambda Sale Order is deemed to be in recordable form sufficient to be placed in the filing or recording system of each and every federal, state, or local government agency, department, or office. Notwithstanding the foregoing, the provisions of this Revised Lonafarnib/Lambda Sale Order authorizing the sale and assignment of the Transferred Assets free and clear of all Interests shall be self-executing, and neither the Debtors nor the Purchaser shall be required to execute or file releases, termination statements, assignments, consents, or other instruments in order to effectuate, consummate, and implement the provisions of this Revised Lonafarnib/Lambda Sale Order.

- 10. <u>Surrender of Transferred Assets</u>. All Persons that are presently or on the Closing Date may be in possession of some or all of the Transferred Assets are directed to surrender possession of such Transferred Assets to the Purchaser as of the Closing Date.
- 11. <u>Continuation of Existing Approvals</u>. The Purchaser shall be authorized, as of the Closing Date, to operate under any license, permit, registration, and governmental authorization or approval of the Debtors with respect to the Transferred Assets (subject, in each case, to the

Casse2**24880040**9ssgij**j1**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainExholicum8enPagePage35

terms of the Lonafarnib/Lambda APAs), and all such licenses, permits, registrations, and governmental authorizations or any other approvals are deemed to have been, and hereby are, directed to be transferred to the Purchaser as of the Closing Date. All existing licenses or permits applicable to the business shall remain active, in place, and, as applicable, shall be renewed for the Purchaser's benefit until either new licenses and permits are obtained or existing licenses and permits are transferred in accordance with applicable administrative procedures. To the maximum extent permitted by section 525(a) of the Bankruptcy Code, no governmental unit (as defined in Bankruptcy Code § 101(27)) or any representative thereof may revoke or suspend, or in any way challenge or fail to consent to any renewal of any permit or license relating to the operation of the Transferred Assets because of the filing or pendency of the Debtors' chapter 11 cases or the consummation of the Lonafarnib/Lambda Sale Transactions.

12. Injunction. All Persons are hereby prohibited and enjoined from taking any action that would adversely affect or interfere with, or that would be inconsistent with, the ability of the Debtors to sell and transfer the Transferred Assets to the Purchaser in accordance with the terms of the Lonafarnib/Lambda APAs, the Transaction Documents, or this Revised Lonafarnib/Lambda Sale Order. Except as expressly permitted by the Lonafarnib/Lambda APAs with respect to Assumed Liabilities or this Revised Lonafarnib/Lambda Sale Order, all Persons (and their respective successors and assigns), including, without limitation, all holders of claims or Interests, lenders, debt security holders, governmental, tax and regulatory authorities, parties to executory contracts and unexpired leases, creditors, contract counterparties, customers, landlords, licensors, employees and former employees, litigation claimants, pension plans, labor unions, trade creditors, and other Persons holding Interests of any kind or nature whatsoever against or in the Debtors or the Transferred Assets (whether known or unknown, legal or equitable, matured or unmatured,

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibitum 8en Pag Page 852 of 51

contingent or non-contingent, liquidated or unliquidated, asserted or unasserted, whether arising prior to or subsequent to the commencement of the Debtors' chapter 11 cases, whether imposed by agreement, understanding, law, equity, or otherwise), arising under or out of, in connection with, or in any way relating to, the Debtors, the operation of the Debtors' businesses prior to the Closing, the Transferred Assets, or the transfer of the Transferred Assets to the Purchaser (including, without limitation, any rights or claims based on any successor, transferee, derivative, or vicarious liabilities), shall be and hereby are forever barred, estopped, and permanently enjoined from asserting, prosecuting, or otherwise pursuing any Interests against the Purchaser, any of its Affiliates, officers, directors, members, partners, principals, or shareholders, any of their respective representatives, successors, designees, or assigns, the property of the foregoing, and the Transferred Assets transferred to the Purchaser or interests of the Debtors in such Transferred Assets (other than the Assumed Liabilities). Following the Closing, no holder of an Interest against the Debtors shall interfere with the Purchaser's title to or use and enjoyment of the Debtors' former interests in the Transferred Assets, including, without limitation, taking any of the following actions with respect to or based on any Interest relating to the Transferred Assets or the transfer of the Transferred Assets to the Purchaser (other than Assumed Liabilities): (a) commencing or continuing in any manner any action or other proceeding against the Purchaser or its successors or assigns, assets or properties; (b) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order against the Purchaser or its successors or assigns, assets, or properties; (c) creating, perfecting, or enforcing any Interest against the Purchaser, its successors or assigns, assets (including the Transferred Assets), or properties; (d) asserting any Interest as a setoff, right of subrogation, or recoupment of any kind against any obligation due Purchaser or its successors or assigns: (e) commencing or continuing any action in any manner or place that does

not comply or is inconsistent with the provisions of this Revised Lonafarnib/Lambda Sale Order or the agreements or actions contemplated or taken in respect thereof; (f) interfering with, preventing, restricting, prohibiting, or otherwise enjoining the consummation of the Sale Transactions; or (g) enforcing any provision of any Assigned Contract that prohibits, restricts or conditions, or which purports to terminate or modify, or permits a party other than the Debtors to terminate or modify, any such Assigned Contract, or any right or obligation under such Assigned Contract, because of the assumption and assignment of such Assigned Contract by the Debtors to the Purchaser. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Revised Lonafarnib/Lambda Sale Order, any provision in an Assigned Contract, any other document, or any applicable law that purports to prohibit, restrict, modify or otherwise impair assignment of the Assigned Contracts or the Purchaser's ability to utilize the Transferred Assets in Purchaser's business is hereby void and of no force and effect with respect to the Lonafarnib/Lambda Sale Transactions, including without limitation any provision that (a) terminates or modifies any right or obligation of the Purchaser under such Assigned Contract; (b) cross-defaults to or from any other lease or executory contract that is not an Assigned Contract; (c) contains operating covenants or "go-dark" provisions that would purport to terminate or modify any Assigned Contract before assumption and assignment to the Purchaser; (d) requires a third party's consent prior to assignment of the Assigned Contract to the Purchaser; or (e) restricts the Purchaser's use or assignment of any licenses or similar permits if transferred.

13. <u>General Assignment</u>. As of the Closing, this Revised Lonafarnib/Lambda Sale Order shall be construed and shall constitute for any and all purposes a full and complete general assignment, conveyance, and transfer of the Transferred Assets and/or a bill of sale or assignment

Casse2**24880049**9sgijj**11**1 DDoc:8**5:8**88 Filitiek0**08422522**5 <u>Hinterce</u>0**08422522**516144075**2**5 DDessc MainElohikuin6en1Pagle734e08**5**20f 51

transferring indefeasible title and interest in the Transferred Assets, including the Assigned Contracts, to the Purchaser. Each and every federal, state, and local governmental agency or department is hereby authorized and directed to accept any and all documents and instruments necessary and appropriate to consummate the Lonafarnib/Lambda Sale Transactions and to reflect the effectiveness of the Lonafarnib/Lambda Sale Transactions.

14. No Successor, Transferee, or Similar Liability. The Purchaser, its Affiliates, and any of their respective officers, directors, members, partners, principals, employees, independent contractors, and shareholders (or equivalent) and any of their respective representatives, agents, predecessors, successors, or assigns shall not be and shall not be deemed, as a result of the consummation of the Lonafarnib/Lambda Sale Transactions or otherwise, (i) to be a successor of, successor employer of, successor entity of, to have successorship obligations relating to, or to otherwise be deemed a successor, to the Debtors or the Debtors' estates, including with respect to any labor, employment, employee, personnel, or worker related matter, law, or agreement, including any collective bargaining agreement, works council agreement, union agreement, area labor agreement, multiemployer agreement, project labor agreement, construction agreement, contractor agreement, building agreement, regional agreement, work standards agreement, or other labor Contract (collectively, a "Collective Bargaining Agreement"), any employee benefit plans, any defined benefit pension plan, or any multiemployer plans, and the Purchaser and/or its Affiliates, as applicable, shall instead be, and be deemed to be, a new employer, including with respect to, among other things, any and all federal or state unemployment laws, including the Fair Labor Standards Act, any employee wage and hour law, privacy law, worker classification law, minimum wage law, overtime law, compensation or benefit law, meal or rest break law, time keeping law, employee record or documentation law, workers compensation law, unemployment

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibitum 8en Pag Page 35e08520 f 51

compensation or tax law, or any other similar federal or state law (provided that the Purchaser shall pay employee-related liabilities solely to the extent expressly included in the Assumed Liabilities); (ii) to have any common law successorship liability in relation to any Collective Bargaining Agreement, union, multiemployer organization, employee benefit plan, or multiemployer plan, including with respect to withdrawal liability or contribution obligations; (iii) to have, de facto or otherwise, merged or consolidated with or into any of the Debtors or any of the Debtors' estates, (iv) to be the successor of or a successor employer (as defined under COBRA and applicable regulations thereunder, common law, or otherwise) to the Debtors; (v) to have a common identity with the Debtors; (vi) to be an alter ego, joint employer, single employer, a continuation or substantial continuation, or to be holding itself out as a mere continuation, of any of the Debtors or their respective estates, or any enterprise of any of the Debtors, (vii) to be liable for any acts or omissions of the Seller or any of the other Debtors in connection with any Collective Bargaining Agreement, personnel, worker, employee, independent contractor, the conduct of the business, or the operation, funding, or administration of the employee benefit plans or multiemployer plans or arising under or related to the Transferred Assets other than as expressly set forth in the Lonafarnib/Lambda APAs; (viii) to have any successor liability, transferee liability, derivative liability, vicarious liability, or any similar theories of any kind or character including, without limitation, under any theory of foreign, federal, state, or local antitrust, environmental, successor, tax, ERISA, assignee or transferee liability, labor, product liability, employment, de facto merger, substantial continuity, or other law, rule, regulation, or doctrine, whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated; (ix) except as expressly set forth in the Lonafarnib/Lambda APAs, to have any successor liability, transferee liability, derivative, liability,

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibitum 8en Pag Page 36e08520 f 51

vicarious liability, for any similar theories of any kind or character including under any pending, threatened, or potential claim, litigation, arbitration, settlement, investigation, fact circumstance, or event disclosed in the Transaction Documents; in each case whether known or unknown as of the Closing Date, whether now existing or hereafter arising, whether asserted or unasserted, fixed or contingent, liquidated or unliquidated, except to the extent solely and expressly provided for in the Lonafarnib/Lambda APAs. The Purchaser shall not assume, or be deemed to assume, or in any way be responsible for any liability or obligation of any of the Debtors and/or their respective estates, or any of their predecessors or Affiliates. The so-called "bulk sales," "bulk transfer," or other similar laws shall be waived in all necessary jurisdictions, including those relating to Taxes. Except as expressly set forth in the Lonafarnib/Lambda APAs with respect to Assumed Liabilities, the Purchaser, its Affiliates, officers, directors, members, partners, principals, and shareholders (or equivalent) and any of their respective representatives, successors, or assigns, or the Transferred Assets shall have no liability or responsibility whatsoever with respect to, or be required to satisfy in any manner, whether at law or in equity, whether by payment, setoff or otherwise, directly or indirectly (w) any Interest against the Debtors or against an insider of the Debtors, (x) any Interest or Excluded Liabilities, (y) the Debtors except as expressly set forth in the Lonafarnib/Lambda APAs and the Transaction Documents.

15. <u>Good Faith of the Purchaser</u>. The Lonafarnib/Lambda Sale Transactions specified in the Lonafarnib/Lambda APAs are undertaken by the Purchaser without collusion and in good faith, as that term is defined in section 363(m) of the Bankruptcy Code, and, accordingly, neither the reversal nor modification on appeal of the authorization provided in this Revised Lonafarnib/Lambda Sale Order to consummate the sale shall affect the validity of the Lonafarnib/Lambda Sale Transactions, including, without limitation, the assumption and

Casse2**24880049**9sgijj**11**1 DDoc:8**5:8**88 Filitiek0**08422522**5 <u>Hinterce</u>0**08422522**516144075**2**5 DDessc MainElokoikuin6en1Pagle7367e08620f51

assignment of the Assigned Contracts, unless such authorization and consummation of the sale are duly and properly stayed pending such appeal. The Purchaser is a good faith purchaser within the meaning of section 363(m) of the Bankruptcy Code and, as such, is entitled to the full protections of section 363(m) of the Bankruptcy Code.

- 16. No Avoidance of the Lonafarnib/Lambda APAs. Neither the Debtors nor the Purchaser have engaged in any conduct that would cause or permit the Lonafarnib/Lambda APAs to be avoided or costs and damages to be imposed under section 363(n) of the Bankruptcy Code. Accordingly, the Lonafarnib/Lambda APAs and the Lonafarnib/Lambda Sale Transactions shall not be avoidable under section 363(n) of the Bankruptcy Code, and no party shall be entitled to any damages or other recovery pursuant to section 363(n) of the Bankruptcy Code in respect of the Lonafarnib/Lambda APAs or the Lonafarnib/Lambda Sale Transactions. Specifically, the Purchaser has not acted in a collusive manner with any person or entity and the Purchase Price was not controlled by any agreement among bidders.
- 17. Payment of Cure Amounts and Cure Dispute Resolution. All defaults or other obligations of the Debtors under the Assigned Contracts arising prior to the Closing (without giving effect to any acceleration clauses or any default provisions of the kind specified in section 365(b)(2) of the Bankruptcy Code) as to which no objections were interposed, or as to which an objection was interposed but which do not remain pending as of the date of this Revised Lonafarnib/Lambda Sale Order, are deemed satisfied by the payment of the proposed amount necessary, if any, to cure all monetary defaults, if any, under such Assigned Contract in those amounts set forth in the Assignment Notice, and which were satisfied, or shall be satisfied as soon as practicable. For all Assigned Contracts listed on the Assignment Notice for which the Cure Notice or Amended Cure Notice was served, the Purchaser is authorized and directed to pay all

Casse2**24880049**9sgijj**11**1 DDoc:8**5:5**88 Filitiek0**08422**52**2**5 <u>Hintereck</u>0**08422522**516144075**2**5 DDessc MainExholicum8en1Page**28**6e0**8**32of 51

Purchaser Cure Amounts, and the Seller is authorized and directed to pay all Seller Cure Amounts, as soon as practicable after the Closing. Any non-Debtor counterparty to an Assigned Contract that has not filed an objection on or before the deadline as set forth in the relevant Cure Notice or Amended Cure Notice, or received an informal extension by the Debtors, shall be barred from objecting or asserting monetary or non-monetary defaults with respect to any such Assigned Contract other than the applicable amount set forth in the Assignment Notice, and such Assigned Contract shall be deemed assumed by the Debtors and assigned to the Purchaser on the Closing Date.

- 18. With respect to the Assigned Contracts, subject to the terms of the Lonafarnib/Lambda APAs, and subject to the entry of this Revised Lonafarnib/Lambda Sale Order, Purchaser shall make provision for the payment of the Purchaser Cure Amounts, and Seller shall make provision for the payment of the Seller Cure Amounts, in cash at Closing. The Purchaser's promise to perform the obligations under the Assigned Contracts arising after their assumption and assignment to the Purchaser shall constitute adequate assurance of future performance within the meaning of sections 365(b) and 365(f)(2) of the Bankruptcy Code. On the Closing Date, subject in all respects to the terms of this Revised Lonafarnib/Lambda Sale Order, the Purchaser shall be deemed to be substituted for the Seller (and/or any other Debtor, to the extent any of them hold any rights, title, or interests in any of the Assigned Contracts) as a party to the applicable Assigned Contracts.
- 19. In the event of an objection by a Contract counterparty to the Cure Amount with regard to any Contract (such contract, a "<u>Disputed Contract</u>"), such Disputed Contract may be conditionally assumed and assigned, with the consent of the Purchaser, pending the entry of a Disputed Contract Order (as defined below). In the event a Disputed Contract remains unresolved

Casse2248800409ssgjj111 DDoor:85888 Filitield084225225 Hintered008422522516144075025 DDess c MainEndoileutn6en1Pagle789e08820f51

as of the Closing Date, Seller shall either settle the objection of such party or shall litigate such objection under procedures as established by the Bankruptcy Court. In no event shall the Seller settle a Cure Amount objection with regard to any potential Assigned Contract without the express written consent (such consent not to be unreasonably withheld) of Purchaser (with an email consent being sufficient). Upon entry of an Order of the Bankruptcy Court (if necessary) determining any Cure Amount and authorizing the assumption and assignment to Purchaser of such Disputed Contract after the Closing, which order shall be in form and substance acceptable to Purchaser (a "Disputed Contract Order"), Purchaser shall have the option to designate the Disputed Contract as an Assigned Contract or an Excluded Contract (regardless of whether such contract was identified on the Contracts List). If Purchaser elects to designate the Disputed Contract as an Excluded Contract, (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract for all purposes under the Revised Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract. If Purchaser elects to designate the Disputed Contract as an Assigned Contract, such Disputed Contract shall be deemed an Assigned Contract for all purposes hereunder and, for the avoidance of doubt, Purchaser shall assume the Disputed Contract and shall be responsible for paying the associated Purchaser Cure Amount (if any) with respect to such Disputed Contract and (if applicable) Seller shall be responsible for paying all related Seller Cure Amounts, which such Cure Amount shall be made as soon as practicable after the Purchaser elects to assume the Disputed Contract. If Purchaser does not designate such Disputed Contract as either an Excluded Contract or an Assigned Contract within five (5) Business Days after the date of the Disputed Contract Order (or such later date as agreed by the Seller and Purchaser), (a) such Disputed Contract shall automatically be deemed to be an Excluded Contract Casse2**24880040**9ssgj**j11**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainExholicum8enPage**240**eo**89**2of 51

for all purposes under this Revised Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs, and (b) Purchaser shall not be obligated to pay any Cure Amount or liabilities associated with such Disputed Contract.

20. **Determination of Cure Amounts.** Unless a counterparty to any Assigned Contract has filed a timely Cure Objection which remains subject to an unresolved Cure Dispute as of the entry of this Revised Lonafarnib/Lambda Sale Order, the Cure Amounts set forth on the Assignment Notice shall constitute findings of this Court and shall be final and binding on the counterparties to the Assigned Contracts and their successors and designees upon the Closing and shall not be subject to further dispute or audit based on performance prior to the time of assumption and assignment, irrespective of the terms and conditions of such Assigned Contracts. Each counterparty to an Assigned Contract (other than a counterparty who filed a timely Cure Objection) shall be forever barred, estopped, and permanently enjoined from (i) asserting against the Purchaser or its property (including, without limitation, the Transferred Assets), any default arising prior to or existing as of the Closing, or any counterclaim, defense, recoupment, setoff, or any other Interest asserted or assertable against the Debtors (except as otherwise provided herein), and (ii) imposing or charging against the Purchaser or its Affiliates, any accelerations, assignment fees, increases, or any other fees or charges as a result of the Debtors' assumption and assignment to the Purchaser of the Assigned Contracts in connection with the Lonafarnib/Lambda Sale Transactions approved by this Revised Lonafarnib/Lambda Sale Order. To the extent a counterparty to any of the Assigned Contracts received notice of the Debtors' proposed Cure Amount and fails to file a Cure Objection by the applicable deadline, such party shall be deemed to have (a) consented to the assumption and assignment of the applicable Assigned Contract and the payment of the Cure Amount provided in the Assignment Notices and (b) waived any right to

Casse2**24880040**9ssgij**11**1 DDoor8**55**88 Filitiek0**08422522**5 <u>Hintereck</u>0**08422522**516144075**2**5 DDess c MainElokoikuin6en1Paq**P44**teo4162of 51

assert or collect any other cure amount or enforce any default that may arise or have arisen prior to or as of the Closing.

- 21. <u>Cross-Over Contracts</u>. From the Agreement Date until the Plan Consummation Date, the Debtors shall not, and shall cause its Affiliates not, to reject, amend, modify, sell, assign, license, transfer, convey, deliver or otherwise divest its interests in the Cross-Over Contracts in a manner that adversely affects, or would reasonably be expected to adversely affect, Purchaser's rights or obligations under the Lonafarnib APA, or Purchaser's ability to Develop or Commercialize any Lonafarnib Antiviral Products.
- 22. Except for those Cross-Over Contracts rejected, transferred, assigned or terminated by the Debtors without violating Paragraph 21 of this Revised Lonafarnib/Lambda Sale Order, the Debtors shall, upon Purchaser's written request, transfer and assign, and hereby transfers and assigns, automatically and without further notice, to the Purchaser, each other Cross-Over Contract, effective on the date that is the earliest to occur of (a) the date that each and every Cross-Over Contract Benefited Party of such Cross-Over Contract obtains (i) a new agreement with the applicable counterparty of such Cross-Over Contract for substantially the same services as those then being provided to Seller by such counterparty under such Cross-Over Contract, or (ii) an agreement with a Third Party such that such services then being provided under such Cross-Over Contract to such Cross-Over Benefited Party are no longer needed by such Cross-Over Benefited Party, (b) the date Purchaser and all Cross-Over Contract Benefited Parties of such Cross-Over Contract agree to such transfer and assignment of such Cross-Over Contract, and (c) the date all Cross-Over Benefited Parties are no longer receiving any services under such Cross-Over Contract; and upon such transfer and assignment, such Cross-Over Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib APA. Purchaser shall be responsible for

Casse2248800409sggj111 DDocc85888 Fill Helc 084225225 Hintered 0842252251614407525 DDess c Main Exhibit with en Pag & 42 eo 4152 of 51

paying the associated Purchaser Cure Amount (if any) with respect to such Cross-Over Contract, and (if applicable) Seller shall be responsible for paying all associated Seller Cure Amounts.

- 23. Notwithstanding the foregoing Paragraphs 21 and 22, (x) the IQVIA Contracts shall be Assigned Contracts upon the occurrence of the Satisfactory IQVIA Cure Resolution, and (y) the Cross-Over Contracts that are not IQVIA Contracts (the "Other Cross-Over Contracts") shall be Assigned Contracts upon the occurrence of the Satisfactory Other Cure Resolution, provided that if the Satisfactory IQVIA Cure Resolution does not occur by the Plan Consummation Date, the IQVIA Contracts shall be Excluded Contracts, and if the Satisfactory Other Cure Resolution does not occur by the Plan Consummation Date, the Other Cross-Over Contracts shall be Excluded Contracts. In the event that a Cross-Over Contract becomes an Excluded Contract, Seller shall use commercially reasonable efforts to preserve the Transferred Data, including the Global Safety Databases, and fully transfer and transition the Transferred Data and Transferred Regulatory Information to Purchaser, and shall not instruct the counterparties to the IQVIA Contracts to delete or remove the Transferred Data from the Global Safety Databases.
- 24. Previously Unknown Contracts. If at any time, prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, it is discovered that a Contract material to the operation of the Business should have been identified on the Assumption Notice but was not so listed (any such Contract, a "Previously Unknown Contract"), Seller shall, promptly following the discovery thereof (but in no event later than five (5) Business Days following the discovery thereof), notify Purchaser in writing of such Previously Unknown Contract and provide Purchaser with a copy of such Previously Unknown Contract and the Cure Amount (if any) in respect thereof. Purchaser shall thereafter deliver written notice to Seller (email being sufficient), no later than ten (10) Business Days following such notice of such

Previously Unknown Contract from Seller, if Purchaser elects for such Previously Unknown Contract to be an Assigned Contract. If Purchaser elects for a Previously Unknown Contract to be an Assigned Contract in accordance with this Paragraph 24, then to the extent not previously filed and served, Seller shall file and serve an assignment and assumption notice on the Contract counterparty to such Previously Unknown Contract (a "Supplemental Assignment Notice") notifying such Contract counterparty of Seller's intention to assume and assign to Purchaser such Previously Unknown Contract, including the proposed Cure Amount (if any). Such notice shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser (the "Supplemental Assignment Notice Objection Deadline"). Following expiration of the Supplemental Assignment Notice Objection Deadline and, if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib/Lambda APAs and this Revised Lonafarnib/Lambda Sale Order. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under the Lonafarnib/Lambda APAs.

25. **Previously Excluded Contract**. At any time prior to the earlier of confirmation of a plan in the Chapter 11 Cases or entry of an order dismissing the Chapter 11 Cases, Purchaser may elect to take an assignment of any Excluded Contract that has not yet been assumed and assigned pursuant to an order of the Bankruptcy Court (a "Previously Excluded Contract") by sending a written notice to Seller (email being sufficient) of such election. If Purchaser elects for a Previously Excluded Contract to be an Assigned Contract in accordance with this Paragraph 25,

Casse2248800449sgijj111 DDocc85888 Filitiek0084225225 Hintereck084225225161447525 DDessc MainElokoikuin8en1PaqeP44e041520f51

then to the extent not previously filed and served, Seller shall file and serve a Supplemental Assignment Notice on the Contract counterparty to such Previously Excluded Contract. Such Supplemental Assignment Notice Objection Deadline shall state that such Contract counterparty shall have fourteen (14) days to object to the assumption and assignment of the Contract to Purchaser. Following expiration of the Supplemental Assignment Notice Objection Deadline and if no objections are received, Seller shall submit a proposed order (in form and substance reasonably acceptable to Purchaser) to the Bankruptcy Court under certification of counsel authorizing the assumption and assignment of such Contract to Purchaser and, upon the entry of such an order, such Contract shall be deemed an Assigned Contract for all purposes under the Lonafarnib/Lambda APAs and this Revised Lonafarnib/Lambda Sale Order, and, subject to paragraphs 21-23 of this Revised Lonafarnib/Lambda Sale Order and Section 7.15 of the Lonafarnib APA with respect to Cross-Over Contracts, the Purchaser shall be responsible for satisfying or paying any Cure Amounts or other Liabilities with respect to such Contract, whether or not such Cure Amounts or other Liabilities exceed the Purchaser Cure Amounts. For the avoidance of doubt, the Cross-Over Contracts are not Previously Excluded Contracts. If such Contract counterparty objects to the proposed assumption and assignment, the Contract at issue shall be deemed a Disputed Contract for all purposes under the Lonafarnib/Lambda APAs.

26. **Bristol-Myers Squibb ("BMS") License Agreement**. From and after the closing in connection with the Lambda APA, BMS shall be deemed to have consented to the assumption and assignment of the BMS License Agreement (as defined in the Lambda APA) to the Purchaser, and the BMS License Agreement shall be deemed assumed and assigned to Purchaser, with Purchaser being substituted for Debtor Eiger BioPharmaceuticals, Inc. under such BMS License

Agreement for all purposes thereunder. Notwithstanding anything to the contrary set forth in this Revised Lonafarnib/Lambda Sale Order or the Lambda APA:

- i. The assumption and assignment of the BMS License Agreement to Purchaser pursuant to the Lambda APA shall not constitute any Sublicense (as defined in the BMS License Agreement) or assignment of rights to the BMS Patents, the Licensed Compounds and/or Licensed Products (each, as defined in the BMS License Agreement) under Section 8.3 of the BMS License Agreement, and shall not require Debtor Eiger BioPharmaceuticals, Inc. or Purchaser to pay or owe BMS any share of Sublicense Revenues (as defined in the BMS License Agreement) received by Debtor Eiger BioPharmaceuticals, Inc. in connection with such assumption and assignment;
- ii. except as provided in clause i. above, neither the Revised Lonafarnib/Lambda Sale Order nor the Lambda APA shall be deemed to modify any rights, intellectual property licenses, benefits or other obligations (including but not limited to any royalty, revenue sharing, milestone payment obligations, or indemnification obligations) owed to Bristol-Myers Squibb Company ("BMS") under the BMS License Agreement; and
- iii. except as provided in clause i. above, from and after the closing in connection with the Lambda APA, Purchaser and BMS shall each be (a) obligated to continue to perform as set forth in and pursuant to the terms of the BMS License Agreement, and (b) bound by the terms of the BMS License Agreement; provided, however, that from and after the closing with respect to the Lambda APA, the Purchaser and BMS may consensually enter into a side-letter or other agreement (a "Modification Agreement") modifying any provisions, rights, benefits, obligations and/or requirements under the terms of the BMS License Agreement, but unless and until such Modification Agreement shall have been mutually executed by Purchaser and BMS, all provisions, rights, benefits, obligations and requirements related to the BMS License Agreement shall remain in full force and effect.
- 27. <u>Purchaser Protections.</u> The Purchaser shall be entitled to the Purchaser Protections under the Lambda APA and the Lonafarnib APA, as applicable.
- 28. <u>Ipso Facto Clauses Ineffective</u>. Upon the Debtors' assumption and assignment of the Assigned Contracts to the Purchaser pursuant to this Revised Lonafarnib/Lambda Sale Order and the payment of the Cure Amounts in accordance with this Revised Lonafarnib/Lambda Sale

Casse2248800449sggj111 DDocc85888 Fill Helc084225225 Hintereck084225225161447525 DDessc Main Exhibiting en Pag 12 46 eo 45 20 f 51

Order and the Lonafarnib/Lambda APAs, no default shall exist under any Assigned Contract and no counterparty to any such Assigned Contract shall be permitted to declare or enforce a default by the Debtors or the Purchaser thereunder or otherwise take action against the Purchaser as a result of any Debtor's financial condition, change in control, bankruptcy, or failure to perform any of its obligations under the applicable Assigned Contract. For the avoidance of doubt, and without limiting the generality of the foregoing or the operability of any other relief obtained pursuant to this Revised Lonafarnib/Lambda Sale Order, any provision in a Assigned Contract that prohibits or conditions, whether directly or indirectly, the assignment of such Assigned Contract (including, without limitation, the granting of an Interest therein) or allows the counterparty thereto to terminate, recapture, impose any penalty, condition on renewal or extension, or modify any term or condition upon such assignment shall be deemed an unenforceable anti-assignment provision that is void and of no force and effect with respect to the Lonafarnib/Lambda Sale Transactions as approved by this Revised Lonafarnib/Lambda Sale Order. The failure of the Debtors or the Purchaser to enforce at any time one or more terms or conditions of any Assigned Contract shall not be a waiver of such terms or conditions or of the Debtors' or the Purchaser's right, as applicable, to enforce every term and condition of such Assigned Contract.

29. <u>Binding Effect</u>. This Revised Lonafarnib/Lambda Sale Order and the Lonafarnib/Lambda APAs shall be binding upon and shall govern the acts of all entities, including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other Persons who may be required by operation of law, the duties of their office, or contract, to accept, file, register, or otherwise record or release any documents or instruments, or who may be required to report or

Casse2**24880040**9ssgij**j1**1 DDocc8**5**888 Filitiek0**08422**52**2**5 Eintereck0**8422**52**2**516144075**2**5 DDessc MainENdrichingen Page 47eo4525 51

insure any title or state of title in or to any of the Transferred Assets. The terms and provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, and this Revised Lonafarnib/Lambda Sale Order shall be binding in all respects upon the Debtors and their respective Affiliates and subsidiaries and such parties' successors and assigns, the Debtors' estates, all creditors thereof (whether known or unknown), all holders of equity interests in any Debtor, holders of Interests in, against, or on all or any portion of the Transferred Assets, all non-Debtor parties to the Assigned Contracts, the Purchaser and its respective successors and assigns, and any and all third parties, notwithstanding any subsequent appointment of any trustee, examiners, "responsible persons" or other fiduciaries (collectively, the "Trustee") of the Debtors under any chapter of the Bankruptcy Code, as to which Trustee such terms and provisions likewise shall be binding, and the Lonafarnib/Lambda APAs (including the Assigned Contracts) shall not be subject to rejection or avoidance under any circumstances.

Merck License Agreement. Other than as expressly set forth in any agreement entered into or to be entered into by the Purchaser with Merck contemplated by the Lonafarnib APA, nothing herein or in any other Transaction Document shall constitute a determination or modification of or alter, impair or otherwise affect the rights, title, and interest, whether legal, equitable or contractual, of Merck in, to, and under the Merck License Agreement (as defined in the Lonafarnib/Lambda APAs), including Merck's rights with respect to any payments made or to be made to Merck under the Merck License Agreement or pursuant to any agreement entered into or to be entered into by the Purchaser with Merck contemplated by the Lonafarnib APA. Payments made to Merck in connection with the Lonafarnib/Lambda Sale Transactions shall not be subject to surcharge, reduction, set-off, claw-back, disgorgement or avoidance for any reason. Nothing contained in any Chapter 11 plan confirmed by the Debtors or in any subsequent order of this

Casse2248800409sgijj111 DDocc85888 Filitiek0084225225 Hintereck0842252251614407525 DDessc MainElokoikuin6en1PageP446e0432of51

Court, including any order confirming any plan, any order authorizing the sale of assets of the Debtors pursuant to any section of the Bankruptcy Code or any order approving wind-down or dismissal of any Debtor's Chapter 11 case or any subsequent Chapter 7 case shall change, supersede, abrogate, nullify, restrict or conflict with or in any way prevent or interfere with the consummation or performance of any agreement entered into or to be entered into by the Purchaser with Merck contemplated by the Lonafarnib APA.

- Assumed Liabilities), and that the conveyances described herein have been effected. This Revised Lonafarnib, and that the conveyances described herein have been effected.
- 32. No Material Modifications. The Lonafarnib/Lambda APAs and the Transaction Documents may be modified, amended, or supplemented by the Debtors and the Purchaser, in a writing signed by such parties, and in accordance with the terms thereof, without further order of this Court; *provided*, that (i) any such modification, amendment, or supplement does not have a material adverse effect on the Debtors' estates or its creditors, and (ii) has been agreed to between the Seller and the Purchaser (with respect to the Seller, such consent not to be unreasonably withheld) and approved by the Prepetition Term Loan Administrative Agent. Any material modification, amendment, or supplement to the Lonafarnib/Lambda APAs and the Transaction Documents adversely affecting the Debtors' estates must be filed on the docket and served on all interested parties. Interested parties shall have five business days to file an objection to any such material modification, amendment, or supplement. If no objections are received within five

Casse2248800409sgijj111 DDocc85888 Filitiek0084225225 Hintereck0842252251614407525 DDessc MainElokoikuin6en1PageP449e04820f51

business days or the Court overrules such filed objections, the modified Lonafarnib/Lambda APAs and Transaction Documents shall be effective.

- 33. <u>Subsequent Orders and Plan Provisions</u>. Nothing contained in any chapter 11 plan confirmed in the Debtors' chapter 11 cases or any subsequent order of this Court, including, without limitation, any order confirming any such chapter 11 plan, any order authorizing the sale of assets of the Debtors pursuant to any section of the Bankruptcy Code, and any order approving wind-down or dismissal of any Debtor's chapter 11 case or any subsequent chapter 7 case shall change, supersede, abrogate, nullify, restrict, or conflict with the provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, or this Revised Lonafarnib/Lambda Sale Order, or in any way prevent or interfere with the consummation or performance of the Lonafarnib/Lambda Sale Transactions.
- 34. **Failure to Specify Provisions**. The failure to specify or include any particular provisions of the Lonafarnib/Lambda APAs or the Transaction Documents in this Revised Lonafarnib/Lambda Sale Order shall not diminish or impair the effectiveness of such provisions, it being the intent of this Court that the Lonafarnib/Lambda APAs, the Transaction Documents, and the Lonafarnib/Lambda Sale Transactions be authorized and approved in their entirety.
- 35. Automatic Stay. The automatic stay pursuant to section 362 of the Bankruptcy Code is hereby lifted solely to the extent necessary to (i) allow the Purchaser to deliver any notice provided for in the Lonafarnib/Lambda APAs and the Transaction Documents, and (ii) allow the Purchaser to take any and all actions permitted under the Lonafarnib/Lambda APAs and the Transaction Documents in accordance with the terms and conditions thereof. The automatic stay imposed by section 362 of the Bankruptcy Code shall be modified solely to the

Casse2**24880049**9sgijj**1**11 DDoc:8**5**588 Filitiek0**0842**2525 <u>Hinterec</u>k0**8422**52**2**51614**4**75**2**5 DDessc MainElokoikuin6en1Pagle7**50**e04**9**20f 51

extent necessary to implement the preceding sentence, and this Court shall retain exclusive jurisdiction over any and all disputes with respect thereto.

- Bankruptcy Rules Satisfied or Waived. The requirements set forth in Bankruptcy Rules 6004 and 6006 have been satisfied or are otherwise deemed to be waived. As provided by Bankruptcy Rule 9014, the terms of this Revised Lonafarnib/Lambda Sale Order shall be effective and enforceable immediately upon entry and shall not be subject to stay provisions contained in Bankruptcy Rules 6004(h) and 6004(d). Time is of the essence in closing the Lonafarnib/Lambda Sale Transactions and the Debtors and the Purchaser intend to close the sale as soon as possible.
- Lonafarnib/Lambda APAs. To the extent anything contained in this Revised Lonafarnib/Lambda Sale Order conflicts with a provision in the Lonafarnib/Lambda APAs or Transaction Documents, this Revised Lonafarnib/Lambda Sale Order shall govern and control. Notwithstanding the foregoing, nothing in this Revised Lonafarnib/Lambda Sale Order shall modify or waive any closing conditions or termination rights in the Lonafarnib/Lambda APAs, and all such conditions and rights shall remain in full force and effect in accordance with their terms.
- 38. **Provisions Nonseverable and Mutually Dependent**. The provisions of this Revised Lonafarnib/Lambda Sale Order, the Lonafarnib/Lambda APAs, and the Transaction Documents are non-severable and mutually dependent.
- 39. **Retention of Jurisdiction**. This Court shall retain exclusive jurisdiction to, among other things, interpret, implement, and enforce the terms and provisions of the Lonafarnib/Lambda APAs, the Transaction Documents, and this Revised Lonafarnib/Lambda Sale Order, and each of

Casse2248800409sggj111 DDocc85888 Fillielc08422525 Hintereclc08422525161447505 DDessc MairExhibitur8enPage 54.ex562x151

the agreements executed in connection therewith to which the Debtors are a party or which has been assigned to the Purchaser by the Debtors, and to adjudicate, if necessary, any and all disputes concerning or relating in any way to the Lonafarnib/Lambda Sale Transactions. This Court retains jurisdiction to compel delivery of the Transferred Assets, to protect the Purchaser and its assets, including the Transferred Assets, against any Interests or successor or transferee liability and to enter orders, as appropriate, pursuant to sections 105(a), 363, or 365 (or other applicable sections) of the Bankruptcy Code necessary to transfer the Transferred Assets and the Assigned Contracts to the Purchaser. In the event this Court abstains from exercising or declines to exercise jurisdiction with respect to any matter referenced in this paragraph or is without jurisdiction, such abstention, refusal, or lack of jurisdiction shall have no effect upon and shall not control, prohibit, or limit the exercise of jurisdiction of any other court having competent jurisdiction with respect to any such matter.

- 40. The Purchaser has standing to seek to enforce any terms of this Revised Lonafarnib/Lambda Sale Order, the Lonafarnib/Lambda APAs, and the Transaction Documents in this Court or any other court with competent jurisdiction.
- 41. All time periods set forth in this Revised Lonafarnib/Lambda Sale Order shall be calculated in accordance with Bankruptcy Rule 9006(a).

Submitted By:

SIDLEY AUSTIN LLP

Thomas Califano (TX Bar No. 24122825) William Curtin (admitted *pro hac vice*) Anne Wallice (admitted *pro hac vice*) 787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Charles Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000 Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Exhibit 9

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825)

William E. Curtin (admitted *pro hac vice*)

Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

Attorneys for the Debtors and Debtors in Possession

SIDLEY AUSTIN LLP

Charles M. Persons (TX Bar No. 24060413)

2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: cpersons@sidley.com

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al. 1

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF CLOSING OF LONAFARNIB/LAMBDA SALE TRANSACTIONS

On April 5, 2024, the Court entered the Order (I)(A) Approving the Bid Procedures; (B) Authorizing the Debtors to Select Sentynl Therapeutics, Inc. as the Zokinvy Stalking Horse Purchaser & Approving Bid Protections; (C) Approving the Bid Protections Relating to the Remaining Assets Stalking Horse Purchaser(s), if any; (D) Establishing Bid Deadlines, Auction(s), and Sale Hearing(s); (E) Approving the Form and Manner of Sale Notice; (F) Approving Assignment and Assumption Procedures; (G) Approving the Form and Manner of Potential Assumption and Assignment Notice; (II)(A) Authorizing the Sale of the Assets Free and Clear; and (B) Approving the Assumption and Assignment of Designated Contracts; And (III) Granting Related Relief [Docket No. 94] (the "Bid Procedures Order"), which, among other things, establishes key dates and deadlines related to the Auction for, and the Sale of, the Assets.²

The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

Capitalized terms used but not defined herein shall have the meanings given to them in the Bid Procedures Motion, Bid Procedures Order, Revised Bid Procedures, Revised Lonafarnib/Lambda Sale Order, and Lonafarnib/Lambda APAs.

Casse2248800409sgijj.11. DDooc863869 Filitelc094025225 Hintereclc0940252251.014497445 DDessc Mai Extribitorius nentrage Page 200 f 5

On April 8, 2024, and June 3, 2024, the Debtors served the *Notice of Sale, Bid Procedures, Auction, and Sale Hearing* on all known parties in interest. *See* Docket Nos. 128, 320.

On April 15, 2024, the Debtors filed the *Notice of Filing of Revised Bidding Procedures* [Docket No. 119], which included the revised Bid Procedures (the "<u>Bid Procedures</u>") attached thereto as <u>Exhibit A</u>.

On June 12, 2024, the Debtors filed and served the *Revised Notice of Sale*, *Bid Procedures*, *Auction*, *and Sale Hearing* [Docket No. 331] on all known parties in interest. *See* Docket Nos. 374, 431.

On July 13, 2024, Debtors filed the *Further Revised Notice of Bid Deadlines* [Docket No. 422], which included revised dates and deadlines related to the Bid Deadline for the Lonafarnib sale transaction (the "Lonafarnib Sale Transaction") and the Lambda sale transaction (the "Lambda Sale Transaction").

On August 2, 2024, Debtors filed the *Notice of Cancellation of Auction(s)*, *Designation of Winning Bid for the Lonafarnib Sale Transaction, and Transition To Private Sale Process for Lonafarnib/Lambda Sale Transactions* [Docket No. 489] (the "Lonafarnib/Lambda Sale Notice") selecting the Purchaser as the highest and best bidder for the Lonafarnib/Lambda Assets, and served the Lonafarnib Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. A] and the Lambda Assigned Contracts and Cure Amounts [Lonafarnib/Lambda Sale Notice, Ex. B] (the "Assignment Notice").

On August 5, 2024, Debtors filed the *Debtors' Emergency Motion for the Entry of an Order* (I) Authorizing the Sale of the Lonafarnib and Lambda Assets Free and Clear of Liens, Claims, Encumbrances, and Other Interests, (II) Authorizing the Assumption and Assignment of Executory Contracts and Unexpired Leases, (III) Granting the Purchaser the Protections Afforded to a Good Faith Purchaser, (IV) Approving Purchaser Protections in Connection With the Sale of the Lonafarnib and Lambda Assets, and (V) Granting Related Relief [Docket No. 490] (the "Motion"), which included as Exhibit A the proposed order for the sale of the Lonafarnib/Lambda Assets, and asset purchase agreements for the sale of the Lonafarnib/Lambda Assets (the "Lonafarnib/Lambda APAs") attached as Exhibit 1 and Exhibit 2 to the proposed sale order.

On August 19, 2024, the Debtors filed the *Notice of Revised Proposed Form of Lonafarnib/Lambda Sale Order* [Docket No. 540], which included as <u>Exhibit A</u> the proposed revised order for the sale of the Lonafarnib/Lambda Assets.

On August 21, 2024, the Court entered an order [Docket No. 558] (the "<u>Revised Lonafarnib/Lambda Sale Order</u>") authorizing and approving entry into the Lonafarnib/Lambda APAs and the Lonafarnib/Lambda Sale Transactions contemplated thereunder.

On September 3, 2024, the Closing occurred in accordance with the Lonafarnib/Lambda APAs and the Revised Lonafarnib/Lambda Sale Order. Attached as **Exhibit A** and **Exhibit B** are the final lists of Assigned Contracts pursuant to the Lonafarnib/Lambda APAs.

Copies of the Lonafarnib/Lambda APAs, as well as all related filings and exhibits, are available by: (i) visiting the website of the Debtors' claims, noticing, and solicitation agent,

Casse2248800440ssjýj.11. DDocc86869 Ffilitelc094025225 Hintereclc09402522511014497445 DDessc Mai Extribitorius nentragePage 200f 5

Kurtzman Carson Consultants LLC dba Verita Global ("Verita") at https://www.veritaglobal.net/Eiger, (ii) (888)733-1544 (Toll-Free) or (310 751-2638 (International), and/or (iii) emailing https://www.veritaglobal.net/Eiger/inquiry or (iv) for a fee via PACER at https://ecf.txnb.uscourts.gov/.

[Remainder of page intentionally left blank]

Casse2248800409sgijj111 DDooc863869 Filitelc094025225 Hintereclc09402522511014497445 DDess Mailin xibidoitu 9ne ni Page Page 200 f 5

Dated: September 4, 2024

Dallas, Texas

SIDLEY AUSTIN LLP

/s/ Thomas R. Califano

Thomas R. Califano (TX Bar No. 24122825) William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: tom.califano@sidley.com

wcurtin@sidley.com anne.wallice@sidley.com

and

Charles M. Persons (TX Bar No. 24060413) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400 Email: cpersons@sidley.com

Attorneys for the Debtors and Debtors in Possession

Certificate of Service

I certify that on September 4, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ *Thomas R. Califano* Thomas R. Califano

Exhibit A

Final Lonafarnib Assigned Contracts List

Lonafarnib Assigned Contracts¹

| Asset | Counterparty | Description of Contract |
|------------|---|---|
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | LNF/RTV FDC Tablet Dev. Change Order #7 to E141-8598, dated January 23, 2018 |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Amendment No. 2 to the Master Services and Clinical Manufacture Agreement, dated May 29, 2019 |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Master Services and Clinical Manufacture Agreement, dated May 16, 2016 |
| Lonafarnib | BIORASI, LLC | Master Services Agreement, dated June 23, 2020 |
| Lonafarnib | BIORASI, LLC | Statement of Work #157-1, dated July 10, 2020, as governed by Master Services Agreement, dated June 23, 2020 |
| Lonafarnib | BIORASI, LLC | Change Order 1 to Statement of Work #157-1, dated July 23, 2021 |
| Lonafarnib | BIORASI, LLC | Change Order 2 to Statement of Work #157-1, dated December 21, 2021 |
| Lonafarnib | BIORASI, LLC | Change Order 3 to Statement of Work #157-1, dated January 30, 2023 |
| Lonafarnib | BIORASI, LLC | Change Order 4 to Statement of Work #157-1, dated August 25, 2023 |
| Lonafarnib | Corden Pharma Colorado | Change Order #6 to Statement of Work # 2, dated May 19, 2021 |
| Lonafarnib | Corden Pharma Colorado | Statement of Work 6, dated April 17, 2023 |
| Lonafarnib | Corden Pharma Colorado; Corden Pharma International GmbH | Change Order 1 to the Statement of Work 6, dated April 26, 2023 |
| Lonafarnib | Cyprotex US, LLC | Proposal for Analysis of Active Metabolites of Lonafarnib (LNF): MH17 and HM21, dated May 6, 2019 |
| Lonafarnib | Fisher Clinical Services GmbH | Quote 214873 Order 8 Version 3 20220225, dated February 25, 2022 |
| Lonafarnib | Fisher Clinical Services, Inc. | Quote PSG-A-1051277.v3 20220225, dated February 25, 2022 |

-

¹ Existing Manufacturing Contracts, if any, are identified by the * symbol.

| Asset | Counterparty | Description of Contract |
|------------|--|---|
| Lonafarnib | Fisher Clinical Services U.K. Limited | LNF/RTV with and w/o Alfa Labeling Kits Quote PSG-A- 1007765.v1 20190514, dated May 14, 2019 |
| Lonafarnib | INTRINSIK CORP | Statement of Work #8, dated July 9, 2022, as governed by Master Services Agreement, dated March 6, 2020 |
| Lonafarnib | LONZA BEND, INC. | Amendment No. 1 to the Commercial Supply Agreement, dated March 9, 2023 |
| Lonafarnib | LONZA BEND, INC. | Amendment No. 2 to the Commercial Supply Agreement, dated January 1, 2024 |
| Lonafarnib | LONZA BEND, INC. | Change Order 8 to Statement of Work E141-8598, dated November 12, 2018 |
| Lonafarnib | LONZA BEND, INC. | Statement of Work PN-166560, dated April 10, 2023 |
| Lonafarnib | Lonza Bend; Patheon Canada | Total Transportation Management ("TTM") Freight Quote, dated August 16, 2021 |
| Lonafarnib | Lonza Pharma & BioTech | Validation Proposal, dated 6 April 2020 |
| Lonafarnib | ² Patheon, Inc. | Solely to the extent related to the 25mg strength, XRPD Change of Scope COS-55-R0 to Proposal No. P-TRP-114750-R2, dated May 15, 2023 |
| Lonafarnib | Patheon, Inc. | Project Proposal # C-TRC-270507- R4, dated September 27, 2021 |
| Lonafarnib | Patheon, Inc. | Change of Scope # C-TRC-270507- R4-COS-01-R0, dated January 30, 2023 |
| Lonafarnib | Patheon, Part of Thermo Fischer Scientific; Element Toronto | Element Quote 20-012-162900 Revision 1, dated April 20, 2020 |
| Lonafarnib | PharmaDirections, Inc | WKO-EIG-879 Ad hoc Consulting, dated October 29, 2014 |
| Lonafarnib | PharmaDirections, Inc | Amendment # 1 to WKO-EIG-879, dated June 10, 2015 |

_

² Any Contracts with TFS Entities (as defined in Schedule 3.3) shall be on this Schedule 2.1(a) solely to the extent related to the 25mg strength (but not for 50mg strength, 75mg strength or an AVX injection), and all other Contracts with TFS Entities shall be removed and shall not be deemed on this Schedule 2.1(a).

| Asset | Counterparty | Description of Contract |
|------------|---|---|
| Lonafarnib | PharmaDirections, Inc | Amendment # 2 to WKO-EIG-879, dated January 1, 2019 |
| Lonafarnib | Q SQUARED SOLUTIONS HOLDINGS, LLC | Work Order, dated October 20, 2023, under that certain Master Laboratory Services Agreement, dated May 3, 2019 |
| Lonafarnib | Trialog Clinical Trials Ltd | Study Protocol No.: EIG-LNF-011, dated July 18, 2019 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2019120, dated August 14, 2019 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #201989, dated December 3, 2019 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020017, dated January 27, 2020 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020082, dated March 30, 2020 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020191, dated July 28, 2020 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020201, dated August 9, 2020 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2020348, dated December 31, 2020 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2021-028, dated January 25, 2021 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #2021-210, dated June 8, 2021 |
| Lonafarnib | Trialog Clinical Trials Ltd | Study Protocol No.: SCRC20042, dated June 7, 2021 |
| Lonafarnib | Trialog Clinical Trials Ltd | Price Quote #20221259, dated July 20, 2022 |
| Lonafarnib | Patheon, Inc. | Master Manufacturing Services Agreement, dated January 9, 2020* |
| Lonafarnib | Patheon, Inc. | Quality Agreement, dated January 31, 2020* |
| Lonafarnib | LONZA BEND, INC. (f.k.a. Bend Research, Inc.) | Commercial Manufacturing Services and Supply Agreement, dated October 9, 2019* |

| Asset | Counterparty | Description of Contract |
|------------|--------------------------------------|--------------------------------------|
| Lonafarnib | LONZA BEND, INC. | Commercial Quality Agreement, |
| Lonaramio | (f.k.a. Bend Research, | dated October 17, 2019, as amended |
| | Inc.) | by Amendment No. 1 to Quality |
| | , | Agreement, dated February 15, |
| | | 2023* |
| Lonafarnib | LONZA BEND, INC. | Quality Agreement for Commercial |
| | (f.k.a. Bend Research, | Manufacture of Product, dated |
| | Inc.) | November 1, 2023* |
| | | |
| Lonafarnib | CordenPharma | Master Services Agreement, dated |
| | | March 22, 2016* |
| | | |
| Lonafarnib | CordenPharma | Commercial Quality Agreement, |
| Lonaramo | Cordem narma | dated February 19, 2020* |
| | | |
| Lonafarnib | Fisher Clinical | Master Services Agreement, dated |
| Lonaranno | Services, Inc. | May 6, 2016* |
| | Bervices, Inc. | , 171dy 0, 2010 |
| Lonafarnib | Fisher Clinical | First Amendment and Restated |
| Lonararmo | Services, Inc. | Quality Agreement, dated February |
| | Services, file. | 23, 2021* |
| | | , |
| Lonafarnib | General Synco, Inc. | Quotation GLS q-Eiger-JJ- |
| | | 20220622-300kg, dated June 15, 2022* |
| Lonafarnib | GLSynthesis Inc. | Quotation, dated August 16, 2018* |
| Lonaramio | OLSynthesis me. | Quotation, dated August 10, 2010 |
| Lonafarnib | GLSynthesis Inc. | Quotation, dated November 14, |
| Lonaramio | OLSynthesis inc. | 2018* |
| Lonafarnib | INCEDM 111110 | |
| Lonaranno | INSERM U1110, Université de | Project Proposal 1 |
| | Strasbourg, France | |
| Lonafarnib | U1111, Centre | Project Proposal V-2023-03-16, |
| | International de | dated March 16, 2023 |
| | Recherche en | |
| | Infectiologie, Lyon, | |
| | France, team HepVir | |
| N/A | Eiger Group | Asset Purchase Agreement, dated |
| | International, Inc. | December 8, 2010 |
| Lonafarnib | EZUS LYON | Research Agreement, dated February |
| | (Subsidiary of the | 15, 2024. |
| | Universite Claude | |
| | Bernard Lyon 1), | |
| | Subsidiary of the | |
| | Universite Claude Bernard Lyon 1, | |
| | Centre National de la | |
| | Centre ryanonai de la | |

Case 24-80040-sgj11 Doc 636-9 Filed 09/25/25 Entered 09/25/25 10:49:05 Desc Exhibiti9A PRage 13 of 30

| Asset | Counterparty | Description of Contract |
|------------|-------------------------|--------------------------------|
| | Recherche | |
| | Scientifique, Ecole | |
| | Nomale Superieure de | |
| | Lyon, and Inserm | |
| | Transfert SA | |
| Lonafarnib | SATT Conectus | Sponsored Research Agreement, |
| | Alsace, University of | dated January 12, 2024. |
| | Strasbourg, French | |
| | National Institute of | |
| | Health and Medical | |
| | Research, and Institute | |
| | for Viral and Liver | |
| | Diseases | |

Schedule 7.15 Cross-Over Contracts^{3 4}

| Asset | Counterparty | Description of Contract | |
|------------|-------------------|--------------------------------|--|
| Lonafarnib | IQVIA RDS INC. | Work Order | |
| | | #KZA43736, dated May | |
| | | 8, 2019 | |
| Lonafarnib | IQVIA RDS INC. | Change Order 1 to WO | |
| | | #KZA43736, dated | |
| | | March 26, 2020 | |
| Lonafarnib | IQVIA RDS INC. | General Services | |
| | | Agreement for Emerging | |
| | | Biotech Clients, dated | |
| | | October 15, 2018 | |
| Lonafarnib | IQVIA RDS INC. | Change Order 3 to | |
| | | MZA58497 | |
| Lonafarnib | IQVIA RDS INC. | Change Order 5 to | |
| | | MZA58497 | |
| Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 1 | |
| Lonafarnib | IQVIA Biotech LLC | Change Notification No. | |
| | | 1 | |
| Lonafarnib | IQVIA Biotech LLC | Change Notification No. | |
| | | 2 | |
| Lonafarnib | IQVIA Biotech LLC | Change Notification No. | |
| | | 3 | |
| Lonafarnib | IQVIA Biotech LLC | Change Notification No. | |
| | | 4 | |
| Lonafarnib | IQVIA Biotech LLC | Change Proposal No. 15 | |
| Lonafarnib | Novella Clinical | Master Services | |
| | LLC | Agreement, dated | |
| | | January 15, 2016 | |
| Lonafarnib | Novella Clinical | Statement of Work, dated | |
| | LLC | April 8, 2016 | |
| Lonafarnib | Novella Clinical | Statement of Work, dated | |
| | LLC | July 19, 2016 | |
| Lonafarnib | Novella Clinical | Change Proposal 1 | |
| | LLC | | |
| Lonafarnib | Novella Clinical | Change Proposal 2 | |
| | LLC | | |
| Lonafarnib | Novella Clinical | Change Proposal 3 | |
| | LLC | | |
| Lonafarnib | Novella Clinical | Change Proposal 4 | |
| | LLC | | |

_

³ Contracts with IQVIA RDS INC., IQVIA Biotech LLC and Novella Clinical LLC are Global Safety Database Contracts.

⁴ Notwithstanding anything to the contrary set forth in the Agreement, the Contracts with Accenture, LLP (a) may be assumed by Purchaser only pursuant to Section 7.15(b) of the Agreement once all conditions are met and Purchaser makes a request, and (b) may not be assumed by Purchaser pursuant to 7.15(c) of the Agreement under any circumstances.

| Lonafarnib | Novella Clinical LLC | Change Proposal 5 | |
|------------|-------------------------|---|--|
| Lonafarnib | Novella Clinical LLC | Change Proposal 6 | |
| Lonafarnib | Novella Clinical LLC | Change Proposal 7 | |
| Lonafarnib | Novella Clinical LLC | Change Proposal 8 | |
| Lonafarnib | Accenture, LLP | Amendment One to the Master Services Agreement, dated May 25, 2018 | |
| Lonafarnib | Accenture, LLP | Change Order 3 to SOW 3, dated June 15, 2022 | |
| Lonafarnib | Accenture, LLP | Change Order Form No. 9 to SOW 5, dated December 13, 2021 | |
| Lonafarnib | Accenture, LLP | Scope of Work 4, dated March 2, 2016 | |
| Lonafarnib | Accenture, LLP | Scope of Work 5, dated November 7, 2017 | |
| Lonafarnib | Accenture, LLP | Master Services Agreement, dated March 2, 2016 | |

Exhibit B

Final Lambda Assigned Contracts List

Lambda Assigned Contracts¹

| Asset | Counterparty | Description of Contract | |
|--------|---------------------------------------|--|--|
| Lambda | BECTON, DICKINSON AND COMPANY | Quote #20200427 re: Pharmaceutical Products, dated April 27, 2020 | |
| Lambda | BECTON, DICKINSON AND COMPANY | Quote #20200513, dated May 13, 2020 | |
| Lambda | BECTON, DICKINSON AND COMPANY | Quote PS-CPQ-636, dated July 7, 2021 | |
| Lambda | BECTON, DICKINSON AND COMPANY | Quote #20220224 re: Pharmaceutical Products, dated February 24, 2022 | |
| Lambda | BIORASI, LLC | Statement of Work #157-2, dated April 16, 2021, as governed by Master Services Agreement, dated June 23, 2020 | |
| Lambda | BIORASI, LLC | Change Order 1 to Statement of Work #157-2, dated December 29, 2021 | |
| Lambda | BIORASI, LLC | Change Order 2 to Statement of Work #157-2, dated December 20, 2021 | |
| Lambda | BIORASI, LLC | Statement of Work #157-3, dated January 21, 2021, as governed by Master Services Agreement, dated June 23, 2020 | |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | Assignment and Assumption Agreement, dated May 25, 2016 | |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | Common Stock Purchase Agreement, dated April 20, 2016 | |
| Lambda | BRISTOL-MYERS SQUIBB COMPANY | License Agreement, dated April 20, 2016 | |
| Lambda | Eurofins Biopharma Product Testing | Quotation # HEY2PH220237-01 re: Establishment of a Method for Free PEG by HPLC-CAD, dated May 26, 2022 | |
| Lambda | Eurofins Biopharma Product Testing | Quotation # HEY2PH220237-02 re: Establishment of a Method for Free PEG by HPLC-CAD, dated November 9, 2022 | |
| Lambda | Eurofins BioPharma Product Testing | Quotation # VFK8PH210375-01 re: FBS Qualification for for Lambda-1 (Python), dated September 14, 2021 | |

¹ Existing Manufacturing Contracts, if any, are identified by the * symbol.

| Asset | Counterparty | Description of Contract | |
|--------|--|---|--|
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #W9MYPH200689-02, dated December 4, 2020 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #W9MYPH200689-05, dated April 18, 2022 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #NQ-0143063, dated December 5, 2016 | |
| Lambda | Eurofins Lancaster Laboratories (Lancaster PA) | Quotation #NQ-0148470, dated March 15, 2017 | |
| Lambda | FISHER BIOSERVICES, INC. | Statement of Work # OPP-009068, dated May 23, 2016 | |
| Lambda | FISHER BIOSERVICES, INC. | Amendment #1 to SOW # OPP- 009068, dated June 27, 2016 | |
| Lambda | FISHER BIOSERVICES, INC. | Amendment #2-R6 to SOW # OPP-009068, dated February 3, 2017 | |
| Lambda | Fisher Clinical Services Inc. | CO 1 to PSG-A-1073971 (PSG-A-1076893) 20230524 | |
| Lambda | Fisher Clinical Services Inc. | Quote 20160517 | |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A- 1043137.v1 20210812 | |
| Lambda | Fisher Clinical Services Inc. | Quote 20160927 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 20170221 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Change Order 1 20161116 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Order 7 20170629 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 55799 Order 8 20180918 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 58040 20161206 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 62278 20180309 | |
| Lambda | Fisher Clinical Services Inc. | Quote FCS 68128 Order 1 Version 1 20190720 | |

| Asset | Counterparty | Description of Contract |
|--------|----------------------------------|------------------------------------|
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1007253 V2 20190508 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1009306.V1 20190619 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1037571.v1 20210414 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1037572.v1_20210420 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1037587.v1 20210420 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1038183.v1 20210422 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1040820.V3 20210713 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1041275.v4 20210819 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1044658.v1 20210902 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1045926.v1 20210927 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1053812.v2 20220323 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1056697.v1 20221205 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1060127.v1 20221205 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1072645 v1 20230420 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1073971 20230429 |
| Lambda | Fisher Clinical Services Inc. | Quote PSG-A-1086699.v1 20231128 |
| Lambda | Fisher Clinical Services Inc. | Quote re: Protocol No. EIG-LMD-001 |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1037570.v2 20210423 |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1045938.v1 20210927 |
| Lambda | Fisher Clinical Services Inc. | Quote-PSG-A-1069905.v1 20230204 |

| Asset | Counterparty | Description of Contract | |
|--------|--|---|--|
| Lambda | Fujifilm Diosynth | Stability Studies Termination, Accountability and Reconciliation Memo, dated December 13, 2023 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Scope of Work #9, dated November 15, 2020 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Change Order 1 to SOW9, dated February 11, 2021 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Master Bioprocessing Services Agreement, dated September 22, 2016 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Change Order 5 re: MCB and WCB Bioassay Characterization, dated April 4, 2017 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Change Order 6 re: Establishment of Degraded SEC and Degraded Issi- Asp, CEX & RP Purity Assay Controls, dated July 31, 2017 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Change Order 6 re: Positional Isomer Feasibility, dated February 2, 2021 | |
| Lambda | FUJIFILM DIOSYNTH BIOCTECHNOLOGI ES USA, INC. | Scope of Work 23, dated March 8, 2023 | |
| Lambda | Curia New Jersey, LLC | Laboratory Service Fee Quotation Q-87184-20220825-1431, dated September 22, 2022 | |
| Lambda | INTRINSIK CORP | Statement of Work 9, dated October 12, 2022, as governed by Master Services Agreement, dated March 6, 2020 | |
| Lambda | Intrinsik Health Sciences Inc. | Proposal Re: Canadian Regulatory Services for Phase II Study for PEG- Interferon Lambda, dated March 4, 2016 | |
| Lambda | KRYOCAL, LLC DBA KYROSPHERE | Statement of Understanding, dated February 28, 2018 | |
| Lambda | Patheon Manufacturing Services LLC | Project Proposal (P-MNC-101564-R3), effective July 29, 2016 | |

| Asset | Counterparty | Description of Contract | |
|--------|--|--|--|
| Lambda | Patheon UK Limited | Change of Scope COS-17-R0 to P- MNC- 101564-R3_20220324 | |
| Lambda | Patheon UK Limited | Change of Scope COS-P-MNC- 101564-R3- COS-08-R3_20210309 | |
| Lambda | Patheon UK Limited | Change of Scope: P-MNC-101564-R4-COS-19-R0, dated October 21, 2022 | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 20 P-MNC-101564-R4_20220722 | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 24 P-MNC-101564-R4_20230209 | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope Patheon UK_COS 29-R0 to P-MNC-101564- R4_20240213 | |
| Lambda | Patheon UK Limited; Fisher Clinical Services | Change of Scope: Prefilled Syringes Patheon UK_P-MNC-101564-R4- COS-23-R0_20220922 | |
| Lambda | Thermo Fisher Scientific; Patheon UK Limited, Part of Thermo Fisher Scientific | Quotation #220328-01-SF, dated March 28, 2022 | |
| Lambda | Total Transport Management | Netherlands Hub Freight Quote, dated March 15, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Study Protocol No.: SCRC20006 Agreement, dated April 14, 2020 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, REF #20221093, dated March 3, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, REF #20221258, dated July 20, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, REF #20221259, dated July 20, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, REF #20221437, dated December 7, 2022 | |
| Lambda | Trialog Clinical Trials Ltd | Price Quotation for EIG-LMD-002, REF #20231153, dated February 21, 2023 | |

SIDLEY AUSTIN LLP

William E. Curtin (admitted *pro hac vice*) Anne G. Wallice (admitted *pro hac vice*)

787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: wcurtin@sidley.com

anne.wallice@sidley.com

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: tom.califano@sidley.com

Attorneys for the Debtors and Debtors in Possession

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

In re:

EIGER BIOPHARMACEUTICALS, INC., et al.¹

Debtors.

Chapter 11

Case No. 24-80040 (SGJ)

(Jointly Administered)

NOTICE OF OCCURRENCE OF EFFECTIVE DATE OF FIFTH AMENDED JOINT PLAN OF LIQUIDATION OF EIGER BIOPHARMACEUTICALS, INC. AND ITS DEBTOR AFFILIATES PURSUANT TO CHAPTER 11 OF THE BANKRUPTCY CODE

TO ALL CREDITORS, INTEREST HOLDERS, AND OTHER PARTIES IN INTEREST:

PLEASE TAKE NOTICE that, on September 5, 2024, the United States Bankruptcy Court for the Northern District of Texas (the "Court") entered the Order Approving the Debtors' Amended Disclosure Statement and Confirming the Fifth Amended Joint Plan of Liquidation of Eiger Biopharmaceuticals, Inc. and its Debtor Affiliates [Docket No. 639] (the "Confirmation Order") confirming the Fifth Amended Joint Plan of Liquidation of Eiger BioPharmaceuticals, Inc. and its Debtor Affiliates Pursuant to Chapter 11 of the Bankruptcy Code [Docket No. 635-1] (as may be altered, amended, modified, or supplemented from time to time, including all exhibits and schedules thereto, the "Plan").²

¹ The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Avenue, Dallas, Texas 75201.

² Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to them in the Plan.

PLEASE TAKE FURTHER NOTICE that, on September 30, 2024, the Effective Date of the Plan occurred. Each of the conditions precedent to consummation enumerated in Article IX of the Plan have been satisfied or waived in accordance with the Plan and the Confirmation Order.

PLEASE TAKE FURTHER NOTICE that the Court has approved certain release, exculpation, injunction, and related provisions in Article IX of the Plan.

PLEASE TAKE FURTHER NOTICE that, on the Effective Date, except as otherwise set forth in the Plan or Confirmation Order, the terms of the Plan became effective and enforceable and deemed binding upon the Debtors, the Plan Administrator, the Liquidation Trustee, any and all Holders of Claims and Interests (regardless of whether such Holders of Claims or Interests accepted or rejected, or were deemed to have accepted or rejected, the Plan), all Entities that are parties to or subject to the settlements, compromises, releases, and injunctions described in the Plan, each Entity acquiring property under the Plan or the Confirmation Order, and any and all non-Debtor parties to Executory Contracts and/or Unexpired Leases with the Debtors.

PLEASE TAKE FURTHER NOTICE that, pursuant to the Plan and the Confirmation Order, the deadline for filing requests for payment of Administrative Claims (other than Professional Compensation Claims) shall be 5:00 p.m. prevailing Central Time on the date that is thirty (30) days after the Effective Date, unless otherwise ordered by the Court.

PLEASE TAKE FURTHER NOTICE that, pursuant to the Article V of the Plan and the Confirmation Order, except as otherwise provided in the Plan or Confirmation Order or otherwise identified on the Schedule of Assumed Executory Contracts and Unexpired Leases, all Executory Contracts and/or Unexpired Leases not otherwise assumed, assumed and assigned, or rejected pursuant to an order of the Bankruptcy Court, will be deemed rejected, in accordance with the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code.

PLEASE TAKE FURTHER NOTICE that all Proofs of Claim with respect to Claims arising from the rejection of Executory Contracts and/or Unexpired Leases, pursuant to the Plan or Confirmation Order, if any, must be Filed with the Notice and Claims Agent and served upon counsel to the Plan Administrator and the Liquidating Trustee within thirty (30) days of the Effective Date. Any Claims arising from the rejection of an Executory Contract or Unexpired Lease pursuant to the Plan and Confirmation Order that are not timely Filed within thirty (30) days of the Effective Date will be disallowed automatically, forever barred from assertion, and shall not be enforceable against, as applicable, the Debtors, the Estates, the Plan Administrator, or the Liquidating Trustee, or any of their respective assets and properties.

PLEASE TAKE FURTHER NOTICE that the Plan and the Confirmation Order contain other provisions that may affect your rights. You are encouraged to review the Plan and the Confirmation Order in their entirety.

PLEASE TAKE FURTHER NOTICE that copies of the pleadings may be obtained (i) at the website established by the Debtors' notice and claims agent, Verita Global f/k/a Kurtzman Carson Consultants, LLC at https://veritaglobal.net/Eiger, (ii) from the Court's website at https://www.txnb.uscourts.gov via ECF/Pacer, or (iii) upon request to the undersigned.

THIS NOTICE IS BEING SENT TO YOU FOR INFORMATIONAL PURPOSES ONLY. IF YOU HAVE QUESTIONS WITH RESPECT TO YOUR RIGHTS UNDER THE PLAN OR ABOUT ANYTHING STATED HEREIN OR IF YOU WOULD LIKE TO OBTAIN ADDITIONAL INFORMATION, PLEASE CONTACT THE NOTICE AND CLAIMS AGENT.

[Remainder of page intentionally left blank.]

Cosse 4-8-80404-9-gg/gill 1 Doo 8-588-10 File 16-0 9/4/2/2/4 5 Efficience 0 9/4/2/2/4 2 0:5-3:0:05 Desce Mai Extradoutura nt Page 25 ef 46 of 5

Dated: September 30, 2024

Dallas, Texas

/s/ Thomas R. Califano

SIDLEY AUSTIN LLP

Thomas R. Califano (TX Bar No. 24122825) 2021 McKinney Avenue, Suite 2000

Dallas, Texas 75201

Telephone: (214) 981-3300 Facsimile: (214) 981-3400

Email: tom.califano@sidley.com

and

William E. Curtin (admitted pro hac vice) Anne G. Wallice (admitted pro hac vice) 787 Seventh Avenue New York, NY 10019

Telephone: (212) 839-5300 Facsimile: (212) 839-5599

Email: wcurtin@sidley.com

anne.wallice@sidley.com

Attorneys for the Debtors and Debtors in Possession

C6555242-48900404-05-gi_0j_11_1 DDoo 835851.0 Filfeite 0 9743/272425 Efficience 0 974/272425 (1:15:31:17:17)5 DEccessor Maifix Dividuo in 12 Central Page 360 eff 560 f 5

Certificate of Service

I certify that on September 30, 2024, I caused a copy of the foregoing document to be served by the Electronic Case Filing System for the United States Bankruptcy Court for the Northern District of Texas.

/s/ Thomas R. Califano
Thomas R. Califano

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

| In re: | § | Chapter II |
|--|---|-------------------------|
| | § | |
| EIGER BIOPHARMACEUTICALS, INC., et al. 1 | § | Case No. 24-80040 (SGJ) |
| | Š | , |
| Debtors. | § | (Jointly Administered) |

DECLARATION OF JAMES VOLLINS IN SUPPORT OF THE LIQUIDATING TRUSTEE AND THE PLAN ADMINISTRATOR'S LIMITED RESPONSE TO MOTION OF SENTYNL THERAPEUTICS, INC TO (I) ENFORCE THE ZOKINVY SALE ORDER AND (II) FOR CONTEMPT AGAINST EIGER INNOTHERAPEUTICS, INC.

I, James Vollins, pursuant to section 1726 of title 28 of the United States Code, hereby declare that the following is true to the best of my knowledge, information, and belief:

- 1. I worked as General Counsel, Chief Compliance Officer, and Corporate Secretary of Eiger Biopharmaceuticals, Inc. ("<u>Eiger</u>" or the "<u>Debtors</u>") from April 2023 through September 30, 2024 (the effective date of the Debtors' Plan of Liquidation).
- 2. While working for Eiger, I was a member of the team managing negotiation and due diligence with Sentynl Therapeutics, Inc. ("Sentynl") concerning the sale of certain assets related to the FDA approved drug Zokinvy (lonafarnib). The Sentynl representatives involved in the negotiations and due diligence included Michael Hercz, Senior Vice President, General Counsel & Chief Compliance Officer.
- 3. Sentynl's interest in acquiring assets related to Zokinvy pre-dated Eiger's Chapter 11 filing by more than six (6) months. Sentynl and Eiger participated in many months of pre-transaction due diligence. As part of the pre-transaction due diligence process, Eiger shared with

¹ The Debtors in these chapter 11 cases, together with the last four digits of each Debtor's federal tax identification number, are: Eiger BioPharmaceuticals, Inc. (1591); EBPI Merger Inc. (9986); EB Pharma LLC (8352); Eiger BioPharmaceuticals Europe Limited (N/A); and EigerBio Europe Limited (N/A). The Debtors' service address is 2100 Ross Ave., Dallas, Texas 75201.

Sentynl certain manufacturing and quality agreements, including, but not limited to, Eiger's agreements with Lonza Bend Research ("Lonza") and Corden Pharma ("Corden").

- 4. In connection with the filing for Chapter 11 protection, Eiger evaluated the potential for reorganizing its business around the drug development program evaluating lonafarnib for the treatment of HDV and/or selling assets related to that drug development program to a third party. Given the potential options for the lonafarnib/HDV program, Eiger had a reasonable and material economic interest in retaining certain agreements and/or rights in agreements concerning lonafarnib for HDV. Eiger contracted with several of the same vendors for certain services related to the commercialization and manufacture of Zokinvy (lonafarnib) and lonafarnib for HDV.
- 5. On March 28, 2024, I emailed draft sublicense schedules to Sentynl's General Counsel, Mr. Hercz, for his review. Draft Schedule 3.7 included the Lonza and Corden contracts as proposed contracts to be "retained" by Eiger. The draft schedule included a drafting note to Sentynl reflecting Eiger's willingness to discuss how rights and obligations under those manufacturing agreements could be assigned or assumed by Sentynl if Sentynl wanted to acquire or assume any of Eiger's rights in those agreements.

"Note to Sentynl: These/manufacturing storage agreements concern Zokinvy and various development programs. Eiger is open to discussing an optimal manner to assign the portions of the agreements related to Zokinvy, add Sentynl as a party for Zokinvy only, or manage as retained agreement for reasonable period post sale."

6. I understood that Sentynl intended to negotiate new direct contracts with the third party vendors who provided services to Eiger under the "Retained Agreements" and/or new contracts with different manufacturers for substantially the same services after the effective date of the Zokinvy sale transaction.

- 7. Sentynl agreed that Eiger should manage certain manufacturing agreements, including, but not limited to, Eiger's agreements with Lonza and Corden as "Retained Agreements" until the earlier of the date that Sentynl obtained a new agreement for substantially the same services as those provided by the counterparty under the Retained Agreement or six (6) months from the closing date of the sale transaction.
- 8. Schedule 3.7 of the final sublicense agreement executed between Eiger and Sentynl identified the Lonza and Corden agreements as "Retained Agreements" that were not purchased or assumed by Sentynl.
- 9. Sentynl never asked me or to my knowledge anyone else employed at Eiger to amend the Lonza agreements or otherwise waive any exclusivity provision included in that agreement.
- 10. At no time prior to my last day employed by Eiger (September 30, 2024) did Sentynl notify me or, to my knowledge, anyone else employed at Eiger, that the assignment of the Lonza or Corden agreements to Eiger Innotherapeutics, Inc. ("Inno") would adversely affect, or would reasonably be expected to adversely affect, Sentynl's ability to commercialize Zokinvy.
- 11. On several occasions, I solicited Sentynl's interest in acquiring Eiger's assets related to lonafarnib for HDV, citing the potential value of the drug under development and the potential positive synergies of Sentynl acquiring all Eiger's assets relative to lonafarnib. I did not advise Sentynl's General Counsel, Mr. Hercz, of the "potential for a third party purchaser to improperly use the lonafarnib assets (including the Retained Agreements) to interfere with Sentynl's use and enjoyment of the Zokinvy assets it purchased 'free and clear'" as is claimed in Sentynl's motion to enforce the sale order and hold Inno in contempt.

Cases & 4-4-800404-9-gigijil 1 DDoo & 3789-1.1 Filfeite 10 30/4/20/20/25 Editertente 10 30/4/20/20/25-15:10:4:74:305 Descessor Mai Extribito itu 11:10 to 12 per 145 of 4

Pursuant to 28 U.S.C. § 1746, I declare under penalty of perjury under the laws of the

United States of America that the foregoing is true and correct to the best of my knowledge and

belief.

Dated: March 27, 2025

By: /s/ James Vollins

James Vollins

Filed Under Seal

Filed Under Seal

Filed Under Seal